CONTRACTOR CONTRACTOR

BEFORE THE PUBLIC UTILITIES COMMISSION OF THE STATE OF CALIFORNIA

Application of Southern California Edison Company (U 338-E) For Authority to Increase Its Authorized Revenues for Electric Service In 2025, Among Other Things, and to Reflect That Increase in Rates.

Application 23-05-010 (Filed May 12, 2023) **FILED** 07/15/24 04:59 PM A2305010

OPENING BRIEF OF THE UTILITY REFORM NETWORK

PUBLIC VERSION

David Cheng, Staff Attorney Hayley Goodson, Managing Attorney Thomas Long, Director of Legal Strategy Robert Finkelstein, General Counsel Matthew Freedman, Staff Attorney Katy Morsony, Staff Attorney Elise Torres, Staff Attorney Isabella Montoya, Utility Justice Legal Fellow

THE UTILITY REFORM NETWORK

360 Grand Avenue, #150 Oakland, CA 94610 Phone: (415) 929-8876 E-mail: <u>dcheng@turn.org</u>

July 15, 2024

TABLE OF CONTENTS

			1
POI	LICY		2
3.1	The C	Commission Should Take Steps To Ensure Transparency In Light of the	e
Growi	ng Relian	ce on Machine Learning-Based Modeling Methods.	2
3.2	The C	Commission Must Reject SCE's Proposal to Treat Recorded 2023 Capit	tal
Expen	ditures as	Per Se Reasonable and Prudent, Even Where Opposed by Other Partie	es 2
ÁFI	FORDAB	BILITY AND EQUITY	2
4.1	Affor	dability	2
	4.1.1	The Commission Should Find that Current Levels of Energy Rates and	nd
		Bills Are Not Affordable for Many Low-Income Customers Despite	Low
		Income Assistance Programs	2
	4.1.2	The Commission Should Find that Authorizing SCE's GRC Request	ed
		Increases Will Decrease Affordability for Customers	3
	4.1.3	The Record Clearly Demonstrates that SCE's "Commitment" to	
		Affordability Is Little More Than Lip Service to the Catch Word	4
	4.1.4	The Commission Should Weed Out Spending Requests that Provide	
		Minimal Benefits While Enriching Shareholders and Only Approve	Thos
		Costs that SCE Has Demonstrated Are Both Necessary and Affordab	ole fo
		California Customers	4
4.2	Equity	y Issues	4
	4.2.1	Utility Bill Unaffordability Increases the Likelihood of Homelessnes	s 4
	4.2.2	Frontline Communities: Known Harms and Unknown Benefits of SC	CE's
		Transportation Electrification Grid Readiness Plan Forecast	5
RIS	K-INFO	RMED STRATEGY AND BUSINESS PLAN	5
5.1	Clima	te Change Policy	5
5.2	Enviro	onmental And Social Justice Goals	5
5.3	Quant	titative Risk Modeling	5
	5.3.1	Overview of RSEs and Risk Scores	6
	532	The Commission Developed Quantitative Risk Analysis as a Useful	Tool
	5.5.2		6
	0.0.2	to Prioritize Risk Reduction Spending	0
	5.3.3	to Prioritize Risk Reduction Spending TURN's Testimony Supplements the RSE Information Presented by	SCE
	5.3.3	to Prioritize Risk Reduction Spending TURN's Testimony Supplements the RSE Information Presented by and Presents the SCE and TURN Information in a Useful Format	SCE
	5.3.3 5.3.4	to Prioritize Risk Reduction Spending TURN's Testimony Supplements the RSE Information Presented by and Presents the SCE and TURN Information in a Useful Format SCE Undervalues Quantitative Risk Analysis	SCE
DIS	5.3.3 5.3.4 TRIBUT	to Prioritize Risk Reduction Spending TURN's Testimony Supplements the RSE Information Presented by and Presents the SCE and TURN Information in a Useful Format SCE Undervalues Quantitative Risk Analysis	SCE 6 6
DIS 6.1	5.3.3 5.3.4 5 TRIBUT Infras	to Prioritize Risk Reduction Spending TURN's Testimony Supplements the RSE Information Presented by and Presents the SCE and TURN Information in a Useful Format SCE Undervalues Quantitative Risk Analysis TON GRID tructure Replacement	SCE 6 6 7 7
DIS 6.1	5.3.3 5.3.4 5.3.4 5.3.4 5.3.4 5.3.4 5.3.4 5.3.4 5.3.4 5.3.4 5.3.4 5.3.4 5.3.4 5.3.4 5.3.4 5.3.4 5.3.4 5.3.4 5.5.2	to Prioritize Risk Reduction Spending TURN's Testimony Supplements the RSE Information Presented by and Presents the SCE and TURN Information in a Useful Format SCE Undervalues Quantitative Risk Analysis TON GRID tructure Replacement	SCE 6 6 7 7 7
DIS 6.1	5.3.3 5.3.4 5.3.4 5.3.4 5.3.4 5.3.4 5.3.4 5.3.4 5.3.4 5.3.4 5.3.4 5.3.4 5.3.4 5.3.4 5.3.4 5.3.4 5.3.4 5.5.2	to Prioritize Risk Reduction Spending TURN's Testimony Supplements the RSE Information Presented by and Presents the SCE and TURN Information in a Useful Format SCE Undervalues Quantitative Risk Analysis TON GRID tructure Replacement Overhead Conductor Program Underground Cable Replacement	SCE 6 6 7 7 7 7
DIS 6.1 6.2	5.3.3 5.3.4 5.3.4 5.3.4 5.3.4 5.3.4 5.3.4 5.3.4 5.3.4 5.3.4 5.3.4 5.3.4 5.3.4 5.3.4 5.3.4 5.3.4 5.3.4 5.3.4 5.5.2 5.3.4 5.5.2 5.3.4 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.2 5.5.5 5.5.5 5.5.5 5.5.5 5.5.5 5.5.5.5 5.5.5.5 5.5.5.5 5.5.5.5.5.5.5 5.5.5.5.5.5.5.5.5.5.5.5.5.5.5.5.5.5.5.5.	to Prioritize Risk Reduction Spending TURN's Testimony Supplements the RSE Information Presented by and Presents the SCE and TURN Information in a Useful Format SCE Undervalues Quantitative Risk Analysis TON GRID tructure Replacement Overhead Conductor Program Underground Cable Replacement ction And Maintenance, And Capital-Related Expense	SCE 6 7 7 7 8 8

7.	MET	ER ACTIVITIES	. 86
	7.1	Meter O&M	. 86
	7.2	Meter Capital	. 86
8.	TRA	NSMISSION GRID	. 86
	8.1	Transmission Grid O&M	. 86
	8.2	Transmission Grid Capital Expenditures	. 86
	8.3	Transmission Infrastructure Replacement	. 86
9.	SUBS	STATION	. 86
	9.1	Substation O&M	. 86
	9.2	Substation Capital	. 86
	9.3	Substation Infrastructure Replacement	. 86
		9.3.1 Circuit Breaker Replacement Program	. 87
		9.3.2 Substation Transformer Bank Replacement Program	. 92
10.	GRII	D MODERNIZATION, GRID TECHNOLOGY, AND ENERGY STORAGE	2 95
	10.1	Grid Modernization	. 95
	10.2	Grid Technology Assessments, Pilots, And Adoption	. 99
	10.3	Energy Storage	. 99
11.	LOA	D GROWTH, TRANSMISSION PROJECTS, AND ENGINEERING	102
	11.1	Load Growth	102
		11.1.1 SCE's Load Growth Forecast is Too High Due to Erroneous Assumption	ons
		in the TEGR that Inflate Anticipated Demand	108
		11.1.2 The Commission Should Not Authorize Costs Based on SCE's TEGR	
		Load Growth Forecast Due to Misalignment with IEPR	116
		11.1.3 The Rate Increases Resulting from SCE's Inflated Load Growth Foreca	ist
		Will Disproportionately Burden Disadvantaged Customers and Discour	age
		Electrification	122
		11.1.4 Private Market Parties Claims of Energization Delays are Not Due to	
		SCE's Lack of Capital	125
		11.1.5 TURN's Project Specific Forecast Reductions are Reasonable and	
		Supported by the Record	127
	11.2	Transmission Projects	135
	11.3	Engineering O&M	135
12.	NEW	SERVICE CONNECTIONS AND CUSTOMER REQUESTED SYSTEM	
	MOD	DIFICATIONS	135
	12.1	New Service Connections	135
		12.1.1 The Commission Should Adopt TURN's Residential Gross Meter Set	100
		Forecast for the Test Year.	138
		12.1.2 The Commission Should Adopt TURN's Commercial Gross Meter Set	1 5 0
		Forecast for the Test Year.	150
		12.1.3 The Commission Should Adopt TURN's Agricultural Gross Meter Set	1 ~ 1
		Forecast for the Test Year.	151
		12.1.4 The Commission Should Direct SCE to Include All Information and Ra	ιW ·
		Data Supporting Its Customer and New Service Connections Forecasts	1n
	12.2	Its Direct Testimony Workpapers in Future GRCs	152
	12.2	Customer Kequested System Modifications	154

		12.2.1 Rule 20 Conversions	154
13.	POLE	S	154
	13.1	Poles O&M	154
	13.2	Poles Capital	154
14.	VEGE	TATION MANAGEMENT	154
	14.1	Inspections Program: The Commission Should Ensure that the Utility Deliver t	the
	Promised	Benefits of Remote Sensing to its Ratepayers.	157
		14.1.1 SCE's Proposal Will Charge Ratepayers for both Ground and Remote	
		Sensing Inspections with No Timeline for Delivering the Efficiencies	
		Promised by LiDAR Technology.	158
		14.1.2 SCE's "Alternative" Proposal Relies on Inflated Escalation Costs to	
		Create an Artificially High Forecast	161
	14.2	Routine Line Clearing: Inflated Escalation and Discretionary Costs Should be	
	Removed	I from SCE's Routine Line Clearing Budget	162
		14.2.1 TURN Adjusts Routine Line Clearing Unit Costs to Remove	
		Unnecessarily High Escalation.	164
		14.2.2 TURN Adjusts Unit Costs Downward to Account for the Utility Relyin	g
		on Elevated Costs the Commission Approved as a Temporary Measure.	
			165
		14.2.3 Expanded Line Clearing Costs Should be Removed from the Forecast.	167
	14.3	Dead, Dying, Diseased Tree Removal: The Dead, Dying, and Diseased Tree	
	Removal	Budget Should be Reduced to Remove Inflated Contract Inflation and Expansion)n
	of the Pro	ogram.	171
	14.4	Hazard Tree Management Program: SCE Fails to Demonstrate that the Hazard	
	Tree Mar	agement Program Is a Cost-Effective and Reasonable Use of Ratepayer Funds.	172
	14.5	Seasonal Patrols/AOC/Emergent Work	175
	14.6	Structure Brushing	175
	14.7	Environmental Support For Vegetation Management	175
	14.8	Wildfire Mitigation Vegetation Management Technology Solutions	175
15.	WILD	OFIRE MANAGEMENT	175
	15.1	Overview.	175
	15.2	15.2.1. Symmetry of SCE's Drop and TUDN's Decommendations	175
		15.2.1 Summary of SCE's Proposal and TURN's Recommendations	1/3
		15.2.2 Covered Conductor Reduces Significant whome Risk, Call Be Elinand	eu
		With Effecting Technologies, is Much More Cost-Effective than Undergrounding, and Can Ba Danlayed Much More Quickly	177
		15.2.3 SCE's Proposed Massive Investment in Undergrounding Would Page P	1//
		15.2.5 SCE's Floposed Massive investment in Ondergrounding would be a F	001
		Has Already Been Accomplished With Covered Conductor, at Patenay	or
		Expense	182
		15.2.4 SCE Wrongly Relies on Its Flawed Designation of So-Called 'Severe R	Pick
		Areas' (SR A) to Define the Scone of Its Undergrounding Proposal	184
		15.2.5. SCE Places Inadequate Emphasis on Cost-Effectiveness in Its Proposal	for
		How to Spend Ratenaver Money	194
		15.2.6 TURN's Recommendation	197
			- / /

		15.2.7 Response to SCE's Rebuttal Testimony	206
	15.3	Emergent Technology And Inspections And Remediations	212
	15.4	PSPS And Other Wildfire Activities	212
16.	T&D	OTHER COSTS AND OTHER OPERATING REVENUE	212
	16.1	T&D Other Costs	212
	16.2	T&D Other Operating Revenues	212
17.	CUST	FOMER SERVICE OPERATIONS	212
	17.1	Billing And Payments	212
	17.2	Customer Contacts	212
	17.3	Customer Service Re-Platform	213
	17.4	Customer Service-Related Other Operating Revenues	213
	17.5	Billing Practices And Policies	213
18.	BUSI	NESS CUSTOMER SERVICES	213
	18.1	Business Customer Services	213
	18.2	Communications, Education, And Outreach	213
19.	CUST	FOMER PROGRAMS AND SERVICE	213
	19.1	Customer Experience Management	213
	19.2	Customer Programs Management	213
20.	BUSI	NESS CONTINUATION	214
	20.1	Planning, Continuity, And Governance	214
	20.2	All Hazards Assessment, Mitigation and Analytics Seismic Resiliency for	Non-
	Electric	Facilities	214
21.	EME	RGENCY MANAGEMENT	220
	21.1	Training, Drills And Exercises	220
	21.2	Emergency Preparedness And Response	220
	21.3	Storm Response	220
22.	CYBI	ERSECURITY	220
	22.1	Cybersecurity Delivery	220
	22.2	Grid Modernization Cybersecurity	220
•••	22.3	Software License & Maintenance	220
23.		SICAL SECURITY	220
24.	GENI	ERATION	220
	24.1	Hydro	220
			220
	24.2	24.1.2 Hydro O&M	222
	24.2	Fossil Fuel (Including Mountainview and Peakers)	220
		24.2.1 Mountainview Capital	220
		24.2.2 Peakers Capital	234
	24.2	24.2.5 Decommissioning Accruais and Contingency	237
	24.3		240
	24.4	24.4.1 O RM posts	241
		24.4.1 UCIVI CUSIS	243
	24.5	24.4.2 Capital costs	231
	24.3	Vataillia	2/1
		24.3.1 Solal Calport project	213

		24.5.2 Battery Control System Upgrade	. 283
	24.6	Nuclear / Palo Verde Nuclear Generating Station	. 285
		24.6.1 Historical overforecasting justifies adjustments to PVGS ratemaking	. 286
		24.6.2 The Commission should continue to require shareholders to pay for 50)%
		of Nuclear Energy Institute dues	. 289
25.	ENER	RGY PROCUREMENT.	. 301
	25.1	Energy Procurement O&M	. 301
	25.2	Energy Procurement Capital	. 301
26.	ENTE	CRPRISE TECHNOLOGY	. 301
201	26.1	Technology Planning Design And Support	301
	26.2	Technology Delivery	. 301
	26.3	Digital And Process Transformation	. 301
	26.4	Service Management Office And Operations	. 301
27.	OPER	ATING UNIT CAPITALIZED SOFTWARE	. 302
28.	ENTE	CRPRISE PLANNING AND GOVERNANCE (NON-INSURANCE)	. 302
	28.1	Financial Oversight And Transactional Processing	. 302
	28.2	Legal	302
	28.3	Business And Financial Planning	302
	28.4	Supply Chain Management And Supplier Diversity And Development	306
29.	INSU	RANCE	. 306
_>,	29.1	Liability Insurance (Wildfire)	. 306
	29.2	Liability Insurance (Non-Wildfire)	. 307
	29.3	Property Insurance	. 308
30.	EMPI	OYEE BENEFITS, TRAINING AND SUPPORT	. 308
	30.1	Employee Support	. 309
	30.2	Employee Benefits & Programs	. 309
		30.2.1 Short-Term Incentive Program (STIP) and Executive Incentive	
		Compensation Program (EICP)	. 309
		30.2.2 Long-Term Incentive Program (LTIP)	. 324
		30.2.3 401(k) Savings Plan	. 325
		30.2.4 Medical Program	. 330
		30.2.5 Pensions	. 331
		30.2.6 Post-Retirement Benefits Other than Pensions (PBOPs)	. 343
	30.3	Employee Training	. 345
		30.3.1 Training Seat Time and Delivery – Transmission and Distribution	. 345
31.	ΤΟΤΑ	AL COMPENSĂTION STUDY	. 359
32.	ENVI	RONMENTAL SERVICES	. 359
	32.1	Environmental Services O&M	. 359
	32.2	Environmental Services Capital	. 362
	32.3	SDG&E Request For SONGS-Related Cost Recovery Re: Marine Mitigation	. 363
33.	AUDI	T SERVICES	. 363
34.	ETHI	CS AND COMPLIANCE	. 363
35.	SAFE	TY PROGRAMS	. 363
36.	ENTE	RPRISE OPERATIONS	. 363
	36.1	Transportation Services Department	. 363
		V	

	36.2	Facilities And Land Operations	. 363
		36.2.1 Edison Training Academy (formerly, T&D Training Center)	. 365
		36.2.2 Vehicle Maintenance Facilities	. 373
		36.2.3 Alhambra Regional Operations Facility Renovations	. 377
		36.2.4 Westminster Combined Facility Renovations	. 380
		36.2.5 San Jacinto Laydown Yard	. 382
37.	POLI	CY, EXTERNAL ENGAGEMENT, AND RATEMAKING	. 386
	37.1	Develop And Manage Policy And Initiatives	. 386
	37.2	Education, Safety, And Operations	. 386
	37.3	Professional Education And Development	. 388
		37.3.1 The Commission should reduce SCE's forecast for EEI dues	. 388
		37.3.2 The Commission should disallow ratepayer funding for the California	
		Taxpayer Association	. 394
	37.4	Ratemaking Cost Recovery Business Planning Element	. 395
38.	RESU	ILTS OF OPERATIONS	. 395
	38.1	Results Of Operations	. 395
	38.2	CPUC-Jurisdictional Revenue Requirement	. 395
	38.3	GRC Ratemaking Proposals, Including Memorandum And Balancing Accoun 395	its
		38.3.1 Memorandum and Balancing Accounts	. 395
		38.3.2 Other GRC Ratemaking Proposals	. 425
	38.4	Forecasts Of Sales, Customers, And New Meter Connections	. 431
	38.5	Present Rate Revenue	. 433
	38.6	Cost Escalation	. 433
	38.7	Other Operating Revenues (Excluding Non-Tariffed Products And Services).	. 433
	38.8	Other Operating Revenues - Non-Tariffed Products And Services	. 433
		38.8.1 SCE's Lack of Auditable Records Along with the Inherent Conflict of	•
		Interest in the Determination of Incremental Costs Should Result in th	e
		Finding that SCE Has Not Met Its Burden of Proof	. 433
		38.8.2 The Commission Should Order SCE to Maintain Auditable "But For"	
		Tests and Time Logs at Shareholder Expense	. 436
		38.8.3 The Unreasonable Sharing Mechanism and Arbitrary Determination of	f
		Incremental Costs Have Allowed Shareholders to Achieve Astronomic	cal
		Levels of Profitability Under the NTP&S Program	. 438
		38.8.4 SCE's Usage of Ratepayer Funded Assets Raises Questions About Overbuilding of Capacity for Non-Utility Uses	. 439
		38.8.5 The Commission Needs to Perform a Comprehensive Review of the	
		NTP&S Program, Including the Outdated Sharing Mechanism that Ha	S
		Unreasonably Enriched Shareholders at Ratepayers' Expense	. 442
	38.9	Operation And Maintenance Expense Forecast	. 445
	38.10	Overhead Allocation	. 445
	38.11	Reinvestments In Utility-Owned Generation Resources	. 445
39.	RATE	E BASE	. 445
	39.1	Plant In Service, Reserves, And Depreciation Expense	. 445

		39.1.1 A Utility Cannot Receive Cost Recovery for Capital Expenditures	
		Recorded in Memorandum Accounts Until the Commission Reviews	,
		Those Costs for Reasonableness and Determines that They Were	
		Prudently Incurred.	447
		39.1.2 The Commission Should Direct SCE to Exclude from Plant Within it	ts RO
		Model All Costs Recorded to Memorandum Accounts that the	
		Commission Has Not Found Reasonable for Recovery	449
	39 2	Working Capital (Excluding Customer Deposits)	450
		39.2.1 Income Tax Lag	450
	393	Customer Deposits	453
	39.4	Taxes	455
40.	SCE	ASSET DEPRECIATION STUDY	455
101	40.1	T&D Net Salvage	456
	10.1	40.1.1 General Principles of Net Salvage	457
		40.1.2 TURN's Net Salvage Recommendation – The Commission Should I	imit
		Increases Consistent With The "Gradualism" Employed In Recent G	RC
		Decisions	458
		40.1.3 After Ignoring the Concept of Gradualism In Its Depreciation Study	SCE
		Raised Off-Rejected Or Baseless Arguments Against It in Rebuttal	460
	40.2	T&D Average Service Life	466
	40.2	40.2.1. TURN's Analysis And Recommendations Are Firmly Based On SCH	400 7's
		Retirement Data and Produce Reasonable Curves and Lives	466
		40.2.2. The Criticisms Raised in SCE's Rebuttal Lack Adequate Evidentiary	400
		Analytical Support	472
	40.3	Small Hydro Decommissioning	477
	40.4	Generation Decommissioning Escalation	480
	40.5	Solar PV	483
	40.6	Fuel Cell Generation	483
	40.7	Miscellaneous/Other	483
41	POST	T TEST VEAR RATEMAKING	483
111	41.1	Attrition Is Intended to Provide Some Reasonable Relief to Shareholders Be	tween
	General	Rate Cases Not Guarantee Company Farnings	485
	General	41.1.1. SCE Confuses Attrition with Cost of Service Ratemaking	486
	41.2	O&M Attrition Should Be Based on General Inflation (CPI-II)	489
	11.2	41.2.1 Using a Broad Index for O&M Escalation Reasonably Addresses	109
		Inflation-Related Cost Pressures While Providing the Utility an Incer	ntive
		to Manage its Operations as Efficiently as Possible	490
	413	Capital Attrition Should Be Based on Seven Years of Recorded Capital Add	litions
	With an	Exception for Wildfire Mitigation	494
	with an	41.3.1 SCE's Critiques of TURN's Capital Attrition Mechanism for Non-	דעד
		Wildfire Should Be Dismissed	496
		41.3.2 Wildfire Canital Attrition Should Re Based on Attrition Vear Budget	498
		41.3.3 SCF's Other Budget Based Attrition Proposals Should Be Denied	۵۲۲ .c. ۵۹ <u>۸</u>
		11.5.5 SOL 5 Other Dudger Dused Multion I toposals bilouid De Dellied	470

41.4	If the Commission Adopts SCE's Basic Capital Attrition Mechanism, th	e
Commis	ssion Should Apply the National Version of Capital Escalation Indices and	Reject
SCE's 7	est Year Capital Additions "Lag" Adjustment.	501
	41.4.1 Test Year Capital Additions Should Be Escalated Using National	lly-Based
	Rather Than Regionally-Based Capital Indices.	501
	41.4.2 SCE's Proposed Adjustment to Mitigate the "Lag" in Capital Ex	penditures
	Closing to Plant is Inappropriate.	503
41.5	The Z-Factor Threshold and Deductible Should Be Increased to Reflect	Inflation
Since Its	s Initial Adoption for SCE.	505
RESI	DENTIAL DISCONNECTIONS AND ARREARAGES	507
COM	IPLIANCE REQUIREMENTS	507
ACC	ESSIBILITY ISSUES	507
RESU	ULTS OF FINANCIAL EXAMINATION BY CAL ADVOCATES	507
GRC	UPDATE PHASE	507
CON	CLUSION	507
	41.4 Commis SCE's T 41.5 Since Its RESI COM ACC RESI GRC CON	 41.4 If the Commission Adopts SCE's Basic Capital Attrition Mechanism, th Commission Should Apply the National Version of Capital Escalation Indices and SCE's Test Year Capital Additions "Lag" Adjustment. 41.4.1 Test Year Capital Additions Should Be Escalated Using National Rather Than Regionally-Based Capital Indices. 41.4.2 SCE's Proposed Adjustment to Mitigate the "Lag" in Capital Exp Closing to Plant is Inappropriate. 41.5 The Z-Factor Threshold and Deductible Should Be Increased to Reflect Since Its Initial Adoption for SCE. RESIDENTIAL DISCONNECTIONS AND ARREARAGES. COMPLIANCE REQUIREMENTS. ACCESSIBILITY ISSUES. RESULTS OF FINANCIAL EXAMINATION BY CAL ADVOCATES. GRC UPDATE PHASE. CONCLUSION.

LIST OF TABLES AND FIGURES (WIP)

TARIFS
TABLES Table 1: SCE Territory Percent of Renter Occupied Housing by County
Table 2: Public Participation Hearing Comments Regarding Concerns of Homelessness52
Table 3. TURN's proposed distribution automation upgrades and associated savings 98
Table 4. SCE Proposed LDES Capital Expenditures
Table 5: Summary of SCE and TURN Recommendations 103
Table 6: Summary of SCE and TURN Recommendations by Project Area 107
Table 7: SCE's Table II-9: Routine Line Clearing
Table 8: Mileage and Costs of Grid Hardening (\$ thousands) – TURN vs. SCE 205
Table 9: Summary of TURN Employee Benefits & Training Recommendations (\$1,000s)
Table 10: Summary of TURN's Recommendation Regarding SCE's STIP Cost Forecast
(\$1,000s)
Table 11: Comparison of STIP/EICP Forecast, SCE versus TURN
Table 12: Itemized Comparison of SCE and TURN's Training Seat Time Forecast
(1,000s of Constant 2022\$)

FIGURES

Figure 1. SCE proposed investments in distribution automation	96
Figure 2: Comparing Statewide Light-Duty EV Forecasts	109
Figure 3: Comparing Statewide Medium- and Heavy-Duty EV Forecasts	111
Figure 4: Comparing Medium- and Heavy-Duty EV Forecasts in SCE Territory	112
Figure 5: SCE Forecast Unit Cost of Undergrounding vs. Covered Conductor	179

	Figure 6: Risk Spend Efficiency of Undergrounding vs. Covered Conductor	180
	Figure 7: Risk Spend Efficiency of Undergrounding vs. Covered Conductor	181
	Figure 8: Wildfire Risk Remaining After Grid Hardening and Fast Curve Settings (201	8-
2028).		183
	Figure 9: SCE Decision Process for Grid Hardening	186
	Figure 10: Absolute Risk per Mile Values of Circuit Segments in "Severe Risk Areas"	
		188
	Figure 11: Risk Spend Efficiency of Undergrounding	198
	Figure 12: Absolute Risk Reduction of Grid Hardening Proposals – TURN vs. SCE	200

TABLE OF AUTHORITIES

295
119
229
229

Other Authorities

Cal. Pub. Util. Code § 451	
Cal. Pub. Util. Code § 454	
D.01-06-047	
D.01-10-031	
D.03-12-059	
D.04-05-055	
D.04-07-022	passim
D.05-12-040	
D.06-05-016	
D.06-11-048	

D.07-03-044	
D.09-03-025	passim
D.09-06-027	
D.09-06-049	
D.10-02-032	
D.10-04-028	
D.10-06-031	
D.11-05-018	
D.11-09-017	
D.12-11-051	passim
D.12-12-030	
D.13-05-010	passim
D.14-08-032	passim
D.14-12-025	
D.15-11-021	passim
D.16-08-018	
D.18-02-004	
D.18-04-012	
D.18-07-025	
D.18-12-014	
D.19-05-020	passim
D.19-09-025	
D.19-09-051	passim

D.20-07-038	
D.20-12-029	
D.21-01-002	
D.21-05-003	
D.21-08-026	
D.21-08-036	passim
D.21-10-036	
D.21-11-036	
D.22-06-032	
D.22-11-007	
D.22-12-027	passim
D.23-05-003	
D.23-05-013	
D.23-11-069	passim
D.24-01-004	
D.24-03-008	
D.24-03-042	
D.24-04-042	
D.24-05-037	
D.24-05-064	
D.24-06-025	
D.84-09-120	
D.85-08-046	

D.85-12-076	
D.85-12-108	
D.89-02-074	
D.90-09-088	
D.92-08-036	
D.92-12-057	
D.92497	
D.93-12-043	
D.94-03-048	
D.96-12-066	
D.97-07-054	
Resolution E-3238	
Resolution SPD-17	
Rules	

Rule 13.11	1
------------	---

OPENING BRIEF OF THE UTILITY REFORM NETWORK

1. INTRODUCTION/SUMMARY OF RECOMMENDATIONS

Pursuant to Rule 13.11 of the Commission's Rules of Practice and Procedure, The Utility Reform Network ("TURN") respectfully submits this opening brief in the Test Year 2025 General Rate Case ("GRC") of Southern California Edison Company ("SCE"). In the Sections that follow, TURN provides numerous recommendations regarding the proposals of SCE which are necessary to ensure that the rates adopted by the Commission in this proceeding are just and reasonable.¹ Moreover, TURN's recommendations will promote bill affordability for the millions of California households who depend on SCE to furnish essential electric utility services.

SUMMARY OF RECOMMENDATIONS

Section 2 – Legal Standard

Section 3 – Policy

- **3.1** The Commission should take steps to ensure transparency in light of the growing reliance on Machine Learning-based modeling methods.
- **3.2** The Commission must reject SCE's proposal to treat recorded 2023 capital expenditures as *per se* reasonable and prudent, even where opposed by other parties.

Section 4 – Affordability and Equity

4.1 Affordability

- The Commission should find that current levels of energy rates and bills are not affordable for many low-income customers despite low-income assistance programs.
- The Commission should find that authorizing SCE's requested increases will decrease affordability relative to its current levels.

¹ TURN reserves the right to address issues in its reply brief that TURN does not address here.

- The Commission should maintain SCE's disconnection cap of 4% going forward.
- The Commission should require SCE to include the "normalized" revenue requirement for its proposals going forward that removes the temporary reductions from capital investments.
- The Commission should weed out spending requests that provide minimal benefit and only approve those costs that SCE has demonstrated are both necessary for safe and reliable service and affordable for California customers.

4.2 Equity Issues

- Unaffordable utility bills can lead to utility disconnections, eviction and homelessness. During an electric rate affordability crisis and homelessness crisis, the Commission must consider the impact of rate increases on access to housing and carefully evaluate whether the increases are just and reasonable.
- The Commission should reject SCE's Transportation Electrification Grid Readiness (TEGR) plan load growth capital expenditures forecast that is based on a flawed modeling approach and would unnecessarily increase electric bills without producing benefits for most customers, especially low-income and vulnerable residential customers.

Section 5 – Risk-Informed Strategy and Business Plan

5.1 Climate Change Policy5.2 Environmental and Social Justice Goals

5.3 Quantitative Risk Modeling

Section 6 – Distribution Grid

6.1 Infrastructure Replacement

6.1.1 Overhead Conductor Program

- The Commission should deny funding for the entire program other than the Accelerated OCP (AOCP) element. SCE's proposal to spend upwards of \$330 million annually (from 2025-2028) would be unduly burdensome on ratepayers, and unlikely to achieve the projected reliability and safety benefits.
 - If the Commission chooses to authorize funding for more than the Accelerated OCP element, the funding should be tied to small-gauge conductor replacements (634 miles of the 1,680 miles SCE has proposed).
 - The Commission should deny ratepayer funding for replacements using bare conductor.

- The Commission should direct SCE to assess the benefits and costs of alternatives such as replacements of splice, connector or tap equipment as a lower-cost alternative to conductor replacement.
- Going forward, SCE must take steps to ensure that its reliance on machine learning (ML) models are less opaque, such as sharing the assumptions, testing multiple models, showing confidence intervals and ranges in results, and demonstrating the final model's performance against other potential models.

6.1.2 Underground Cable Replacement

• The Commission should authorize funding based on 800 miles of replacement, rather than SCE's proposed 1,600 miles, as the reduced number of miles will still enable achievement of 60-70% of the safety and reliability benefits according to SCE's models. The resulting funding for the 2023-2025 period on a forecast basis is \$65.15 million, as compared to SCE's forecast of \$114.1 million for that period.

6.2 Inspection and Maintenance, and Capital-related Expense

- The Commission should direct SCE to reevaluate and, as appropriate, scale back its deployment of and reliance on Inspect App for routine and compliance-based inspections.
- The Commission should direct SCE in its next GRC showing to analyze and address any trends in preventive or corrective maintenance and the impacts on associated capital replacement activities, broken out by each category of Distribution Infrastructure Replacement activity. This review should consider, at minimum, any trends in maintenance completion as indicated by "closed notifications."

6.3 Safety and Reliability Investment Incentive Mechanism

Section 7 – Meter Activities

7.1 Meter O&M7.2 Meter Capital

Section 8 – Transmission Grid

8.1 Transmission Grid O&M8.2 Transmission Grid Capital Expenditures8.3 Transmission Infrastructure Replacement

Section 9 – Substation

9.1 Substation O&M

9.2 Substation Capital

9.3 Substation Infrastructure Replacement

- <u>Circuit Breaker Replacement Program:</u>
- The Commission should authorize \$153.13 million rather than SCE's forecast of \$164.29 million for the 2023-'25 period, a reduction of approximately 6.8%.
- The Commission should direct SCE to target replacement of only those circuit breakers that are deemed to be in "poor" or "very poor" condition according to SCE's arbitrary, non-linear Health Index.
- The Commission should direct SCE in its next GRC to: Present a more detailed analysis of unit costs, and break out its circuit breaker proposals into more granular voltage classes, rather than relying on two broad classes; substantiate the accuracy of its chosen health index, or to modify that index to comport with TURN's suggestion of a linear, unbiased index; and develop a designated age threshold approach for replacements. The Commission should also encourage SCE to engage in a stakeholder-involved process in order to develop such thresholds for equipment included in SCE-02, Vol. 02 ("Substations").

9.3.1 Substation Transformer Bank Replacement

- The Commission should authorize funding of \$152.93 million rather than the \$182.00 million for the 2023-'25 period, a reduction of approximately 6.9%.
- The Commission should direct SCE to target replacement of only those substation transformers that are deemed to be in "poor" or "very poor" condition according to SCE's non-linear Health Index.
- The Commission should direct SCE in its next GRC to: develop a designated age threshold replacement approach for replacements; engage in a stakeholder-involved process in order to develop such thresholds for substation transformer banks and other categories of equipment; and present unit cost and annual forecast information at a more granular level for assets like circuit breakers and transformers, such as by specific voltage classes rather than "A-bank" and "B-bank" transformers.

Section 10 – Grid Modernization, Grid Technology, and Energy Storage

10.1 Grid Modernization

• The Commission should adopt TURN's proposal, which is not only more affordable than SCE's proposal (savings of \$168 million to \$170 million) but also more cost-effective (BCR of 8.5 to 9.5 compared with 4.6 to 6.8 for SCE's proposal).

10.2 Grid Technology Assessments, Pilots, and Adoption 10.3 Energy Storage

• The Commission should reject SCE's proposal for LDES capital expenditures of \$18.730 million for 2027 and \$37.977 million for 2028. In addition, any approval for LDES

investments be accompanied by annual progress reports that summarize the status of the project, funds expended, and lessons learned.

Section 11 – Load Growth, Transmission Projects, and Engineering

11.1 Load Growth

- The Commission should reject SCE's \$1,031 million capital expenditure forecast for 2023-2028 associated with the Transportation Electrification Grid Readiness (TEGR) plan given SCE's flawed modeling approach used to develop the TEGR and significant locational and technical uncertainties associated with load forecasting, particularly for new technology groups such as medium- and heavy-duty electric vehicles.
- The Commission should adopt TURN's project specific reductions totaling \$249.762 million to SCE's Base forecast for 2023-2028 to remove projects that have been cancelled, are delayed and duplicative, insufficiently scoped, or fail to demonstrate cost worthiness.

11.2 Transmission Projects 11.3 Engineering O&M

Section 12 – New Service Connections and Customer Requested System Modifications

12.1 New Service Connections

- The Commission should adopt TURN's forecast of residential, commercial, and agricultural new connections, instead of SCE's.
- The Commission should direct SCE to include housing completions in its models in the next GRC, in addition to or instead of housing starts, if SCE choses to forecast residential customers and new meters based on housing.
- The Commission should require SCE to provide all support for its customer and new connections forecasts in future GRCs in its direct testimony workpapers, including but not limited to modeling inputs.

12.2 Customer Requested System Modifications

• The Commission adopt revised forecast for Rule 20A Conversions in its rebuttal testimony.

Section 13 – Poles

13.1 Poles O&M13.2 Poles Capital

Section 14 – Vegetation Management

14.1 Inspections Program

- The Commission should reject SCE's budget request and instead adopt TURN's proposed budget for Inspections of \$52.122 m for 2025.
- The Commission should reject the SCE request to fund two types of simultaneous inspections and instead direct the utility to more efficiently manage the transition to remote sensing.
- The Commission should reject SCE's alternative Traditional Ground Inspections proposal.

14.2 Routine Line Clearing

- TURN recommends that the Commission reject the SCE proposal and instead adopt an adjusted forecast of \$213.776 m.
- SCE's escalation percentage is inflated, and the Commission should instead adopt TURN's 2% escalation factor.
- SCE's inflated unit cost for routine line clearing should be adjusted downward to provide the promised efficiencies to ratepayers and to account for the impact of deep trims on unit costs.
- The Commission should reject Expanded Line Clearing costs as this program does not provide benefits that exceed its costs.

14.3 Dead, Dying, and Diseased Tree Removal

- TURN recommends that the Commission adopt a forecast of \$25.108 m for dead, dying, and diseased tree removal.
- SCE's escalation percentage is inflated, and the Commission should instead adopt TURN's 5% escalation factor.
- SCE has provided no evidence of additional increased tree mortality, and the Commission should reject the request to expand the program.

14.4 Hazard Tree Management Program

• TURN recommends that the Commission reject SCE's proposed Hazard Tree Management Program. The program does not provide benefits that exceed the costs of the program and is not a reasonable use of ratepayer funds.

- 14.1 Seasonal Patrols/AOC/Emergent Work
- 14.2 Structure Brushing
- 14.3 Environmental Support for Vegetation Management
- 14.4 Wildfire Mitigation Vegetation Management Technology Solutions

Section 15 – Wildfire Management

15.1 Overview 15.2 Grid Hardening

- The Commission should approve a grid hardening forecast for the 2025-2028 period of 177 overhead miles converted to undergrounding and 1,651 miles insulated with covered conductor, as shown in the table in Section 15.2.6.3.
- For all 2025-2028 undergrounding projects, SCE should conduct the location-specific analysis described in Section 15.2.6.2 and should only implement projects where the analysis shows that undergrounding is the best alternative for that location.
- Ratepayers shall not be required to fund more than 177 overhead miles in 2025-2028. If SCE cannot justify undergrounding of 177 miles under the location-specific analysis and therefore undergrounds fewer miles, overhead hardening should be deployed on those miles. To the extent that SCE performs less than 177 miles of undergrounding and replaces those miles with overhead hardening, at the end of the rate case period, the difference in costs, based on the forecast unit costs adopted in this decision, should be refunded to ratepayers via the one-way balancing account that TURN recommends in Section 38.3 of this brief.
- The one-way balancing account that TURN recommends in Section 38.3 of this brief should have a separate one-way subaccount for undergrounding expenditures.
- The Commission should require SCE to submit an annual accountability report, similar to the report required in D.23-11-069. The report should require SCE to provide the results of its location-specific analysis for each undergrounding project it opted to pursue (including incomplete projects) in the preceding year. Consistent with the report required in D.23-11-069, SCE's annual report should also include information on completed undergrounding projects, including costs, unit costs, and overhead to underground conversion ratio information.

15.3 Emergent Technology and Inspections and Remediations 15.4 PSPS and Other Wildfire Activities

Section 16 – T&D Other Costs and Other Operating Revenue

16.1 T&D Other Costs

16.2 T&D Other Operating Revenues

Section 17 – Customer Service Operations

17.1 Billing and Payments

• The Commission should adopt all of the individual recommendations included in Ex. SCE-25, recognizing that they constitute an integrated agreement supported by TURN, Cal Advocates, and SCE in its entirety.

17.2 Customer Contacts

• The Commission should adopt all of the individual recommendations included in Ex. SCE-29, recognizing that they constitute an integrated agreement supported by TURN, Cal Advocates, and SCE in its entirety.

17.3 Customer Service Re-Platform17.4 Customer Service-Related Other Operating Revenues17.5 Billing Practices and Policies

Section 18 – Business Customer Services

18.1 Business Customer Services

• The Commission should adopt all of the individual recommendations included in Ex. SCE-26, recognizing that they constitute an integrated agreement supported by TURN, Cal Advocates, Walmart, and SCE in its entirety.

18.2 Communications, Education, and Outreach

Section 19 – Customer Programs and Service

19.1 Customer Experience Management 19.2 Customer Programs Management

• The Commission should adopt all of the individual recommendations included in Ex. SCE-28, recognizing that they constitute an integrated agreement supported by TURN, Cal Advocates, and SCE in its entirety.

Section 20 – Business Continuation

20.1 Planning, Continuity, and Governance

20.2 All Hazards Assessment, Mitigation, and Analytics

• The Commission should adopt TURN's forecast for SCE's seismic retrofitting activities at non-electric facilities.

Section 21 – Emergency Management

- 21.1 Training, Drills and Exercises21.2 Emergency Preparedness and Response
- 21.3 Storm Response

Section 22 – Cybersecurity

22.1 Cybersecurity Delivery22.2 Grid Modernization Cybersecurity22.3 Software License & Maintenance

Section 23 – Physical Security

Section 24 – Generation

- For hydro capital, adopt TURN's proposed forecast for 2024-2028 which was accepted by SCE in its rebuttal testimony but disallow \$10 million relating to San Gorgonio decommissioning to reflect overcollections by SCE for work not performed in 5 previous GRC cycles.
- For hydro O&M, reduce SCE's hydro O&M forecast by \$0.911 million (\$2022) to reflect the use of a longer historical period (2016-2020) for calculating base year non-labor O&M and to account for expected delays in work due to the later anticipated issuance of new federal licenses for the Big Creek and Kaweah facilities.
- For Mountainview capital, reduce SCE's capital expenditure forecast of \$17.692 million between 2023-2028 to reflect the removal of three capital projects from SCE's forecast, reduce allowable recovery by 25% for the Inlet Flow Distribution Grid project, and allow SCE to recover costs for the Turbine Generator Improvement program via a one-way balancing account with excess costs tracked in a memorandum account.
- For Peakers capital, reduce SCE's capital forecast by \$2 million (\$1 million in 2025 and \$1 million in 2026) to reflect TURN's recommendation regarding the timing of relay replacements.
- For fossil generation decommissioning accruals, adopt the use of a 15% contingency (instead of SCE's assumed 20%), a change that would reduce decommissioning estimates by \$13.167 million for Mountainview and \$6.020 million for the Peakers.

- For fuel cells, the Commission should deny SCE any rate of return on \$0.299 million in unrecovered capital due to the projects having been removed from service at the end of 2022.
- For solar O&M, reduce SCE's forecast by \$2.75 million to reflect a 50% disallowance of lease payments and a lower expected escalation rate to reflect more reasonable inflation assumptions.
- For solar capital, reduce SCE's capital forecast by \$40.65 million (2023-2028) by adopting a 50% disallowance of decommissioning capital costs and the use a 10% decommissioning cost contingency. Additionally, TURN recommends a \$125 million disallowance (2023-2030) to SCE's revenue requirements to reflect a 50% disallowance of net book costs for prematurely retired facilities and no rate of return on any such costs allowed to be recovered in rates. SCE should be required to identify the amount of any stranded distribution plant associated with the retired solar facilities and be permitted to recover only 50% of that amount with no rate of return over 6 years.
- For Catalina capital, reduce SCE's capital forecast by \$3.858 million (2023-2028) based on the removal of the solar carport project, the Battery Control System project, and repurposing of the microturbine space. Additionally, enforce a permanent disallowance of solar carport project costs due to SCE's violation of the Settlement Agreement adopted in D.22-11-007. The solar carports should not be assumed to be in service until early 2026.
- For Palo Verde O&M, reduce non-labor O&M by 6% to correct for sustained historic overforecasting and track costs in balancing account with overspending limited to 110% of the forecast value. Additionally, reduce non-labor O&M by \$0.144 million by enforcing the longstanding requirement that 50% of Nuclear Energy Institute trade association dues be paid by shareholders.

Section 25 – Energy Procurement

25.1 Energy Procurement O&M 25.2 Energy Procurement Capital

Section 26 – Enterprise Technology

26.1 Technology Planning, Design, and Support26.2 Technology Delivery26.3 Digital and Process Transformation26.4 Service Management Office and Operations

Section 27 – Operating Unit Capitalized Software

Section 28 – Enterprise Planning and Governance (Non-Insurance)

28.1 Financial Oversight and Transactional Processing 28.2 Legal

28.3 Business and Financial Planning

• The Commission should reduce SCE's forecast for Business Planning by \$3.073 because SCE has failed to justify its non-labor forecast in light of consistently declining non-labor costs and underspending. The Commission should adopt TURN's forecast of \$33.459 million, which includes SCE's request for an increase in labor but uses last recorded year for non-labor.

28.4 Supply Chain Management and Supplier Diversity and Development

Section 29 – Insurance

29.1 Liability Insurance (Wildfire)29.2 Liability Insurance (Non-Wildfire)

• The Commission should adopt the proposed stipulation reached by TURN, SCE and Cal Advocates.

29.3 Property Insurance

• The Commission should adopt the proposed stipulation reached by TURN, SCE and Cal Advocates.

Section 30 – Employee Benefits, Training and Support

30.1 Employee Support

• The Commission should find the stipulation TURN, Cal Advocates and SCE reached and presented in Exhibit SCE-31, reasonable in light of the testimony submitted, consistent with law, and in the public interest.

30.2 Employee Benefits & Programs

- The Commission should reduce SCE's \$120.406, Short-Term Incentive Program (STIP) forecast by \$46.958 million, resulting in a forecast of \$73.447 million by excluding 100% of the STIP costs for the Core Earnings, Capital Deployment and Clean Energy Transition goals and 50% of the STIP costs for the Covered Conductors and Operational Excellence (Catalyst) goals. This, in combination with no TURN objections to full funding of the remaining goals, results in 61% ratepayer responsibility for the STIP funding target. (Note: to implement this recommendation, a 39% reduction should be applied to the RO Model calculation, given that SCE's STIP forecast will ultimately depend on reductions that the Commission makes to the labor force in its GRC decision.)
- The Commission should order SCE to report each year of the rate-case cycle in a Tier 2 advice letter to show that it has made the forecasted conversion of STIP to Base Pay.

- The Commission should continue its policy of disallowing recovery of the Long-Term Incentive Program (LTIP) for the well-established reasons that the Commission has relied upon in past decisions, and reduce SCE's LTIP forecast by the full amount of \$22.017 million.
- The Commission should reduce the 401(k) Savings Plan nominal forecast of \$132.041 million down by \$5.146 million to \$126.895 million to ensure that the STIP-to-Base Pay conversion does not inappropriately result in unjustified, higher 401(k) costs for ratepayers. (Note: to implement this recommendation, a 4% reduction² should be applied to the RO Model calculation.)
- The Commission should reduce SCE's \$153.788 million Medical Programs nominal forecast downward by \$20.869 million for a forecast of \$132.919 million to remove the unreasonable and unsupported increase to SCE's Total Compensation caused by SCE's premium-sharing redesign proposal. (Note: to implement this recommendation, a 16% reduction should be applied to the RO Model calculation.)
- Pensions: The Commission should authorize pension expense of \$17 million, consistent with retaining the historical funding policy rather than adopting SCE's proposed new funding policy.
- PBOPs: The Commission should begin the process of exploring opportunities to put the more than \$1 billion of overcollected PBOP assets to other permissible uses that would benefit SCE's ratepayers, and direct SCE to report on its activities to-date in this regard.

30.3 Employee Training

- The Commission should reduce the Training Seat Time activity forecast by \$8.512 million from \$37.023 million to \$28.511 million to include the combined effects of lower expected training hours (for both existing and new-hire employees) and use a labor rate that disaggregates the overall average that SCE uses for all labor into one that differentiates between existing labor (which is naturally higher) and new hires (which are naturally lower). However, if the Commission agrees with SCE regarding the labor rate used in the Training Seat Time forecast, TURN's alternative forecast is \$29.204 million.
- The Commission should reduce the Training Delivery forecast by \$5.326 million from \$23.189 million to \$17.872 million to be consistent with the recommendation for lower Training Seat Time hours.
- Regardless of the Commission's decision on whether the particular forecast for Training Seat Time and Delivery is reasonable, the Commission should order SCE, either manually or dynamically within the RO Model, to reduce both the Training Delivery and Training Seat Time activities if the Commission makes reductions to any training-

² I.e., \$5.146M / \$126.895M.

impacted personnel on a prorated, percentage basis, that starts with the 2025 Test Year forecasts that the Commission adopts.

Section 31 – Total Compensation Study

Section 32 – Environmental Services

32.1. Environmental Services O&M

 The Commission should adopt the forecast of Environmental Services O&M recommended jointly by TURN, Cal Advocates, and SCE in Ex. SCE-30. The Commission should also clarify that the forecast authorized for Environmental Services O&M covers the costs and activities included in SCE's Environmental Services request presented in Exhibit SCE-06V06, none of which are eligible for tracking in SCE's VMBA or other wildfire mitigation accounts for potential future cost recovery.

32.2. Environmental Services Capital

• The Commission should adopt the forecast of Environmental Services Capital recommended jointly by TURN, Cal Advocates, and SCE in Ex. SCE-30.

32.3. SDG&E Request for SONGS-Related Cost Recovery re: marine mitigation

- Section 33 Audit Services
- Section 34 Ethics and Compliance
- Section 35 Safety Programs

Section 36 – Enterprise Operations

36.1 Transportation Services Department36.2 Facilities and Land Operations

- The Commission should deny SCE's third request for funding for the Edison Training Academy, given prior funding authorizations in the 2018 and 2021 GRCs and ongoing project delays. Alternatively, if the Commission concludes the funding for this project is appropriate, the Commission should disallow SCE's requested \$11 million contingency.
- The Commission should deny SCE's third request for funding for the Vehicle Maintenance Facilities project, given prior funding authorization in the 2018 and SCE's failure to meet the Commission's requirements for additional funding in the 2021 GRC. Alternatively, if the Commission concludes the funding for this project is appropriate, the Commission should disallow SCE's requested \$2 million contingency.

- The Commission should deny SCE's second request for funding for the Alhambra Regional Operations Facility Renovations, given the prior funding authorization in the 2021 GRC and ongoing project delays. Alternatively, if the Commission concludes the funding for this project is appropriate, the Commission should disallow SCE's requested \$4.810 million contingency.
- The Commission should deny SCE's second request for funding for the Westminster Combined Facility Renovations, given the prior funding authorization in the 2021 GRC and ongoing project delays. Alternatively, if the Commission concludes the funding for this project is appropriate, the Commission should disallow SCE's requested \$3.216 million contingency.
- The Commission should deny SCE's request for funding for the San Jacinto Laydown Yard, given ongoing project delays. Alternatively, if the Commission concludes the funding for this project is appropriate, the Commission should disallow SCE's requested \$1 million contingency.

Section 37 – Policy, External Engagement, and Ratemaking

37.1 Develop and Manage Policy and Initiatives37.2 Education, Safety, and Operations

• The Commission should reject SCE's forecast of \$7.630 million for Education, Safety and Operations because SCE underspent on this activity in every year from 2018-2023 and costs have consistently declined from 2019-2023. The more reasonable forecast is \$6.193 million, based on last recorded year (2022).

37.3 Professional Education and Development

• The Commission should reduce SCE's forecast for Professional Education and Development to remove a greater portion of EEI dues than already excluded by SCE (an additional \$0.770 million) and all dues for the California Taxpayers Association (\$0.042 million), and authorize a forecast of \$1.301 million.

37.4 Ratemaking Cost Recovery Business Planning Element

Section 38 – Results of Operations

38.1 Results of Operations38.2 CPUC-Jurisdictional Revenue Requirement

38.3 GRC Ratemaking Proposals, including Memorandum and Balancing Accounts

38.3.1 Memorandum and Balancing Accounts

- The Commission should take reasonable steps to reduce reliance on memorandum and balancing accounts, and to ensure that amounts permitted to be recorded in such accounts receive appropriate reasonableness review before any rate recovery is authorized.
- The Commission should adopt a \$10 million deductible that would routinely apply to new memorandum accounts.
- The Commission should reject SCE's proposed changes to the Wildfire Risk Mitigation Balancing Account/Grid Hardening Balancing Account, Vegetation Management Balancing Account, Electric Vehicle Infrastructure Memorandum Account, and Z-Factor Memorandum Account. The Commission should adopt TURN's proposed changes for each of these accounts.
- The Commission should reject SCE's proposals for the following new memorandum accounts: NextGen ERP SAP Memorandum Account, Advanced Metering Infrastructure 2.0 Memorandum Account, Cybersecurity Compliance Memorandum Account, and Historic Sporting Events Cost Tracking Memorandum Account.
- The Commission should deny SCE's request for rate recovery of the balance recorded in the Service Center Modernization Projects Memorandum Account due to the utility's failure to demonstrate the reasonableness of its above-authorized recorded costs.
- The Commission should reject SCE's proposal to seek rate recovery in this GRC of an amount for which rate recovery was specifically denied in the prior GRC as an abuse of the memorandum account mechanism, among other reasons.

38.3.2 SCE's Proposed "True-Up" of 2023 Recorded Capital Expenditures

• The Commission must reject SCE's proposal to treat recorded 2023 capital expenditures as *per se* recoverable even where opposed, in favor of continuation of the established practice of relying on the recorded figure only where there is no opposition to doing so.

38.4 Forecasts of Sales, Customers, and New Meter Connections

• The Commission should find that SCE's residential customer forecast relies on flawed regression analysis and an overly optimistic housing forecast and endorse TURN's forecast methodology instead.

38.5 Present Rate Revenue38.6 Cost Escalation38.7Other Operating Revenues (excluding Non-Tariffed Products and Services)

38.8 Other Operating Revenues – Non-Tariffed Products and Services

- The Commission should order SCE to maintain auditable "but for" tests and time logs at shareholder expense.
- The Commission should adjust the \$16.72 million initial threshold for NTP&S that was established 25 years ago to account for inflation.
- The Commission should authorize the NTP&S Program for two more years, and if SCE wishes to continue its NTP&S program, it should be required to file an application containing at a minimum the same information PG&E is required to submit.
- The Commission should perform a comprehensive review of the NTP&S Program, including the outdated sharing mechanism.

38.9 Operation and Maintenance Expense Forecast

- **38.10** Overhead Allocation
- 38.11 Reinvestments in Utility-Owned Generation Resources

Section 39 – Rate Base

39.1 Plant in Service, Reserves, and Depreciation Expense

• The Commission should direct SCE to exclude from plant within its RO Model all costs recorded to memorandum accounts that the Commission has not found reasonable for recovery.

39.2 Working Capital (Excluding Customer Deposits)

• The Commission should find that if SCE fails to owe cash federal and state taxes again during this GRC cycle, then working capital should be revised to reflect a federal tax lag of 365 days and a state tax lag of 290 days.

39.3 Customer Deposits

• The Commission should maintain its longstanding practice and treatment of SCE's customer deposits and adopt \$174 million as an offset to rate base.

39.4 Taxes

Section 40 – SCE Asset Depreciation Study

40.1 T&D Net Savage

- 40.1 The Commission should adopt TURN's recommended net salvage values, as they provide for increases that are consistent with the now-longstanding practice of applying gradualism for net salvage adjustments.
- The Commission should adopt TURN's recommended average service lives for the accounts TURN addressed, as they are based on a straightforward analysis without any need to rely on "statistically aged" or otherwise simulated data.

40.2 T&D Average Service Life 40.3 Small Hydro Decommissioning

• For small hydro decommissioning, prohibit SCE from accruing costs for any facility that has less than a 90% probability of being decommissioned.

40.4 Generation Decommissioning Escalation

• For both hydro and fossil decommissioning, accruals should be calculated using constant dollars (\$2028) rather than SCE's proposal to use nominal dollars.

40.5 Solar PV 40.6 Fuel Cell Generation 40.7 Miscellaneous/Other

Section 41 – Post Test Year Ratemaking

• The Commission should find that TURN's two-part attrition mechanism meets the objectives of attrition and more reasonably balances the interests of ratepayers and shareholders during the post-test year period than SCE's proposal. TURN's mechanism resembles the Commission's original approach to attrition and escalates O&M expense using CPI-U, while addressing capital attrition with a 7-year average of recorded capital additions.

41.5 Z-Factor Deductible

• The Commission should increase the Z Factor threshold and deductible to \$18 million to reflect inflation that has occurred since it was first adopted for SCE in the mid-1990s, or a slightly lesser amount for inflation since SCE first asked that the Z Factor be retained for cost of service ratemaking purposes.

Section 42 – Residential Disconnections and Arrearages

Section 43 – Compliance Requirements

Section 44 – Accessibility Issues

Section 45 – Results of Financial Examination by Cal Advocates

Section 46 – GRC Update Phase

Section 47 – Conclusion

2. LEGAL STANDARD- EVIDENTIARY STANDARDS AND BURDEN OF PROOF

The Commission is charged with ensuring that "[a]ll charges demanded or received by any public utility, ... shall be just and reasonable" and cannot approve a rate change "except upon a showing before the commission and a finding by the commission that the new rate is justified."³ In the test year 2009 GRC for SCE, the Commission succinctly described the utility's burden of proof that follows from these statutory mandates:

As the applicant, SCE must meet the burden of proving that it is entitled to the relief it is seeking in this proceeding. SCE has the burden of affirmatively establishing the reasonableness of all aspects of its application. Other parties do not have the burden of proving the unreasonableness of SCE showing. As the applicant in this rate case, SCE has the burden of proving that each of its proposals is reasonable.⁴

Thus, SCE has the burden of affirmatively establishing the reasonableness of all aspects of their application. This evidentiary burden is entirely the utility's; it is not up to intervenors to establish that the utility's forecast is unreasonable unless the Commission first determines that the utility has met its burden of proof with regard to that forecast.⁵ In a recent rehearing decision, the Commission made clear that the utility's failure to meet its ultimate burden of proof could appear as a failure to make an adequate direct showing, or where it has failed to overcome reasonable doubts raised in other parties' showings.⁶ And where the Commission is unable to

³ Cal. Pub. Util. Code Sections 451and 454.

⁴ D.09-03-025, p. 8 (citing Sections 451 and 454, and D.06-05-016 (SCE Test Year 2006 GRC)), p. 7.

⁵ D.21-11-036 (Rehearing decision in PG&E test yaer 2019 GT&S rate case), pp. 3-5.

⁶ See, e.g., D.09-03-025, p. 8; D.06-05-016, p. 7; D.01-10-031, pp. 8-9.

determine that the costs are just and reasonable, including because the utility failed to meet its burden of proof, it "can and must disallow those costs: that is, unjust or unreasonable costs must not be recovered in rates from ratepayers."⁷

The Commission's GRC review will also entail application of the "prudent manager standard" in order to determine whether SCE's costs are prudently incurred. It is a longstanding and well-established principle of California public utilities regulation that costs that result from a utility's imprudence are not reasonable under Section 451 and should not be recovered from ratepayers.⁸ As the Commission emphatically stated in D.84-09-120, "it would be unconscionable from a regulatory perspective to reward such imprudent activity by passing the resultant costs through to ratepayers."⁹

The "prudent manager standard" applies whenever the "necessity of [the utility's] actions

is called into question."¹⁰

Under the prudent manager standard, the Commission does not evaluate reasonableness based on hindsight but based on what the utility knew or should have known at the time it made its decision. [citation to SCE TY 2021 GRC Track 3 decision] This standard reaches not just the activities

⁹ D.84-09-120, 16 CPUC 2d 249, 283.

¹⁰ D.23-11-069 (PG&E TY 2023 GRC), pp. 25-26.

⁷ D.18-07-025 (Rehearing of decision denying SDG&E rate recovery of wildfire claims costs), p. 5, quoting D.14-06-007.

⁸ See, e.g., D.12-12-030, pp. 61, 87 (disallowing recovery of proposed pressure testing costs for pipe segments for which PG&E lacks required records and further disallowing proposed costs of gas system records search and organization costs); D.09-06-027, pp. 35-36 (in a rate case decision, excluding from rate base the costs of a retaining wall for which the water utility failed to meet its burden of proof of demonstrating that the work would have been necessary in the absence of the utility's imprudence); D.01-06-047, p. 2 (disallowing Southwest Gas Company gas procurement costs based on imprudent managerial action); D. 95-12-046, 1995 Cal. PUC Lexis 959 (disallowing costs associated with the utility's imprudent long-term commitment to interstate gas pipeline capacity); D.94-03-048, 53 CPUC 2d 452, 456 (holding that it is not reasonable to pass on to Southern California Edison ratepayers costs resulting from the Mojave Coal Plant accident); D.85-08-102, 18 CPUC 2d 700, 715-716 (holding that ratepayers are not responsible for bearing the consequences of PG&E's imprudence with respect to the Helms Pumped Storage Project).

and associated costs for which [the utility] seeks recovery here but extends to the actions or inactions that resulted in these activities being necessary.¹¹

Here, the standard will apply to determining the reasonableness of both recorded costs from periods preceding the 2025 test year, and also the cost forecasts for the 2025 test year and the rest of the rate case period to the extent the associated activities and spending are necessitated by utility imprudence. That is, to the extent spending in the GRC period was made necessary due to prior SCE actions or inactions that influenced the need for the spending, the prudent manager standard is implicated, and SCE must meet its burden of demonstrating the prudence of said actions or inactions.

In D.17-11-033, denying SDG&E rate recovery of costs related to the 2007 wildfires, the Commission described the prudence standard:

The Commission's standard for reasonableness reviews, reaffirmed in a series of decisions, is as follows: The term reasonable and prudent means that at a particular time any of the practices, methods and acts engaged in by a utility follows the exercise of reasonable judgment in light of the facts known or which should have been known at the time the decision was made. The act or decision is expected by the utility to accomplish the desired result at the lowest reasonable cost consistent with good utility practices. Good utility practices are based upon cost effectiveness, safety and expedition.¹⁷

Thus, in determining whether SCE met the prudent manager standard in this proceeding,

the Commission will need to take a number of factors into consideration. First and most obviously, it will need to assess the quality of the evidence put forward by the utility. In order to meet its burden the utility must present sufficient direct evidence in support of its rate recovery request. For example, simply presenting recorded balances in memorandum accounts would not

¹¹ Id.

establish the prudence of the underlying utility actions. Instead, SCE must provide sufficiently detailed evidence to justify rate recovery of recorded costs in excess of any amounts found reasonable in past decisions. Reliance on utility-produced scheduling and planning documents is insufficient, as such information "does not automatically establish the reasonableness" of the program or project spending in question.¹² The utility needs to demonstrate that the costs were incurred prudently and that it made best efforts to contain costs, with the cost overruns explained and demonstrated to be reasonable.

Similarly, to meet its prudence burden, SCE must show that its actions were at the <u>lowest</u> reasonable cost consistent with good utility practices. And a key element of good utility practices is cost effectiveness. Thus, making sure that SCE incurred the claimed costs wisely and cost-effectively is fundamental to the prudence standard that SCE must satisfy.

Finally, in applying the prudent manager standard, the Commission has held that it will expect the utility's managers to exercise "proportionately greater care" to decisions involving large amounts of money, greater levels of uncertainty, or high degrees of risk.¹³ Exercising greater care, however, is not the equivalent of simply doing more work; it is a recognition that each ratepayer dollar should be put to its best use.

As for the standard of proof in this GRC, the Commission currently requires utilities to meet the "preponderance of the evidence" standard of proof in rate cases.¹⁴ Under that standard,

¹² D.21-11-036 (Rehearing decision in PG&E test year 2019 GT&S rate case), pp. 10-11.

¹³ D.89-02-074, 31 CPUC 2d 236, 246. See also D.85-08-102, 18 CPUC 2d 700, 710-711 (where tasks undertaken are of such enormity to expose the utilities and potentially ratepayers to substantial financial risks, utilities must exercise "even greater care and managerial acumen" than would be called for in ordinary circumstances; rejecting view that "marginal" or "average" performance was required and holding the utility to a "good performance" standard).

¹⁴ D.14-12-025, p. 21.
the applicant must establish the reasonableness of every aspect of its request with evidence that, "when weighted with that opposed to it, has more convincing force and the greater probability of truth."¹⁵

3. POLICY

3.1 The Commission Should Take Steps To Ensure Transparency In Light of the Growing Reliance on Machine Learning-Based Modeling Methods.

The Commission is seeing a significant and growing reliance on machine learning (ML) models as the basis for forecasts of program activities and associated reliability and safety benefits.¹⁶ These models may range from simple Ordinary Least Squares (OLS) models to more complex supervised learning frameworks for classification and regression tasks such as Random Forests (a model used in this GRC application).

To get a flavor for what the use of these models looks like in just one program within the GRC context, the Commission should consider SCE's response to a TURN data request asking about a passage in the utility's workpapers describing the development of the scope of its proposed Overhead Conductor Program as part of its Infrastructure Replacement activities.¹⁷

The workpaper passage said,

The program scope was informed by a machine learning algorithm utilizing a multitude of mechanical, electrical and environmental attributes to estimate the risk and consequence of failure for each of the 500,000+ overhead conductor segments on the SCE distribution system.

¹⁵ D.21-08-036 (SCE TY 2021 GRC), pp. 9-10, *citing* D.08-12-058, p. 19 (citing Witkin, Calif. Evidence, 4th, Vol. 1, 187).

¹⁶ TURN discusses the ML models used for SCE's Overhead Conductor Program and Underground Cable Replacement Program in Sections 6.1.1 and 6.1.2 of this brief.

¹⁷ TURN-SCE-072, Q3

The output of the algorithm was then used to select 1,680 of the higher risk circuit miles population for replacement.¹⁸

TURN sought further information regarding not only the models used but also the models considered but <u>not</u> used. SCE's response is daunting to mere mortals, even those with extensive experience in the utility regulatory process. It includes a high level description of the ML models and sub-models used, and a reference to use of the "Receiver Operating Characteristic (ROC) and Area Under the Curve (AUC)" as "metrics used to determine model algorithm selection, feature selection, and hyperparameter tuning."¹⁹ When asked to specify the outcome variables for the ML model, SCE began with a relatively straightforward statement that even an attorney could understand: "The output of all of SCE's machine learning models is probability of failure for a given asset within an asset class." But the further explanation that followed was not so straightforward, at least to those not steeped in how such models are designed and operate:

The random forest algorithm statistically determines the probability based on constructing a collection or "ensemble" of decision trees. These decision trees are constructed by taking multiple bootstrap samples from historical data. For prediction, each of these decision trees casts a "vote" for whether an asset was a part of the failing or passing class in their bootstrapped data set. Due to the random sampling and sub-setting of data used for decision tree creation and because these trees create decision boundaries across many different variables, the ML model does not have a smooth, continuous functional representation.²⁰

¹⁸ Ex. TURN-05-E, Attachment 2, Response to TURN-07, Question 3(a-f), at pages 85 to 91 of 96 in the PDF file. The same material is contained in Ex. SCE-13, Vol. 1, Appendix A, pp. A11 to A17.

¹⁹ *Id.*, response to TURN-07, Question 3.b.

²⁰ Id., response to TURN-07, Question 3.c.

There are charts showing how each variable for the model is ranked, but the variables are listed in what appears to be coding format (with no description), and nothing approaching plain English.²¹

In sum, TURN submits the description of the information and process involved in this single instance of SCE relying on ML to support its infrastructure replacement proposal makes clear that the Commission, its staff, and interested parties can no longer rely on past practices to effectively assess the utility's showing. And perhaps the even more troubling part of SCE's description is its assertion that the utility considered other model algorithms, but determined "there was no appreciable accuracy improvement."²² If the Commission cedes to the utility the role of selecting which model's results go forward, without an effective process for comparing and contrasting the results of other models considered but not selected, the agency's regulatory oversight risks becoming compromised, as there may be no clear pathway for intervenors and Commission staff to understand or scrutinize these models.

The Commission has tools to ensure that the agency and interested parties have sufficient access to the full range of model inputs (data, assumptions), the process by which the utility selected the model it relies upon to the exclusion of other available alternatives, and the outcomes produced by not only the model the utility chose to rely upon, but also the others it considered or assessed. Public Utilities Code Sections 1821 and 1822 and Rules 10.3 and 10.4 of the Commission's Rules of Practice and Procedure are designed to avoid having the Commission's decisions rely on "black box" processes. However, those provisions date from years ago and were initially designed for simpler closed-form models which, in comparison to

²¹ Id., response to TURN-07, Question 3.d.

²² *Id.*, response to TURN-07, Question 3. b.

today's multivariate, sophisticated ML-based models, might now seem relatively rudimentary. Therefore, it is appropriate for the Commission to specify requirements for SCE's next GRC showing in order to ensure that the transparency of and access to its ML-based and similar analytical process achieves consistency with the longstanding standards.

First, to foster a transparent review process in future GRCs, the Commission should require SCE to submit working ML models following the traditional peer review protocol, and to do so no later than the outset of the GRC process. All such information should be provided as part of the Master Data Request response. This should include, at a minimum, access to the operational models (in Python, RStudio, or other applicable software), labeled datasets that the model is trained on, and the ability for stakeholders to run and test these models under varied assumptions to gauge each model's robustness and sensitivity, aligning with Section 1822(a).

Second, it is crucial that SCE provide not only the final outputs of these models but also details on the inputs, variable descriptions, test/train dataset split and the specific model used wherever ML models are used. This full disclosure should include a descriptive explanation of why the utility chose certain models over others, as well as a transparent analysis of the models' performance through metrics such as ROC, AUC, R² and other precision and accuracy metrics.

Finally, the Commission should encourage the utility to opt for simpler, more efficient models when the trade-off between model complexity and performance suggests that a less complex model provides comparable performance metrics without the opacity or computational demands of more complex, 'black box' models. . This approach would support easier verification and engagement by stakeholders and Commission staff, ensuring that the application of machine learning aids rather than obfuscates utility regulation and oversight.

By adopting a framework that includes ML modeling best practices, the Commission will not only enhance the accountability of SCE's use of advanced modeling techniques but move toward promoting a standard of openness that aligns with technological advancement and regulatory integrity.

3.2 The Commission Must Reject SCE's Proposal to Treat Recorded 2023 Capital Expenditures as *Per Se* Reasonable and Prudent, Even Where Opposed by Other Parties.

SCE's rebuttal testimony unveiled the utility's proposed treatment for recorded 2023 capital expenditures, which the utility argues should be subject to a "true-up" treatment across the board. Where the recorded amounts for 2023 are greater than the utility's forecast for that year, SCE would not be required to establish the reasonableness of the additional increment; rather, it would be enough to show that the utility had spent the money on assets that are used and useful. Reasons for disallowance of such costs would be limited to "imprudence," a term SCE's testimony does not define for this purpose, and any intervenor seeking such a disallowance would have to meet a "heavy burden of persuasion."²³

TURN more fully addresses the lack of merit to SCE's proposal in Section 38.3 of this brief (GRC Ratemaking Proposals). However, given the extreme policy shift reflected in aspects of SCE's proposal, particularly the proposed shift in the evidentiary burden, TURN thought it wise to flag the issue in the policy section of the brief as well.

²³ Ex. SCE-18, Vol. 1, pp. 114-117.

4. AFFORDABILITY AND EQUITY

4.1 Affordability

The Commission has repeatedly stated that "[w]ater, energy, and telecommunications services should be affordable,"²⁴ recognizing that "Californians rely on utility services, including electricity, gas, water, and telecommunications, to live and work."²⁵ Without energy services, families cannot conduct many of the necessary activities of life – refrigerate groceries, cook regular meals, bathe in warm water, or study and work at night with appropriate illumination. In Senate Bill 598 (2017), the California Legislature formally declared that living without basic gas or electric utility service "causes tremendous hardship and undue stress, including increased health risks to vulnerable populations."²⁶ Yet, there can be no doubt that due to high and increasing rates, energy services are not fully accessible to every Californian. The pace of increase in SCE's rates and bills is unsustainable not only from a policy standpoint, but from a moral one because the services SCE provides are life necessities.

Despite clear legislative and regulatory intent to enable universal access to essential energy services, SCE requests incremental base revenue requirement increases of \$1.7 billion (20%) in 2025,²⁷ \$620 million (6%) in 2026, \$660 million (6%) in 2027, and \$700 million (6%) in 2028.²⁸ This translates to more than \$10 billion additional funding from ratepayers over this 2025 GRC cycle. SCE's request is on top of increases of 16% that have already been authorized

https://www.cpuc.ca.gov/industries-and-topics/electrical-energy/affordability.

²⁴ CPUC website: Affordability Rulemaking available at:

²⁵ Order Instituting Rulemaking 18-07-006, p. 3.

²⁶ Senate Bill 598 (2017), Sec. 1(c).

²⁷ Ex. TURN-02E, p. 1, citing SCE 2025 GRC Application (A.23-05-010).

²⁸ Ex. TURN-02E, p. 1, citing SCE 2025 GRC Application (A.23-05-010).

since 2022. As shown below, based on SCE's request, by 2028 SCE's base revenues will have increased by nearly 70% cumulatively since 2022. These figures do not consider any past or future authorized increases, other revenue proceedings, or the impact of balancing account true ups and cost of capital adjustments.

SCE's GRC Revenues (2022-2028)²⁹

SCE's Historical and Proposed GRC Base Revenue Requirement

Values in \$000s unless												
indicated	Jan-22	Mar-22	Jan-23	Mar-23	Jun-23	Oct-23	Jan-24	2025 Req	2026 Req	2027 Req	2028 Req	Totals
	(1)	(1)	(2)	(2)	(3)	(4)	(4)	(5)	(5)	(5)	(5)	
GRC Base RRQ	\$ 7,259,200	7,259,220	7,792,632	7,702,857	7,633,108	7,633,108	8,582,245	10,266,672	10,885,338	11,548,972	12,253,484	\$ 68,588,543
Increase - \$	-	\$ 20	533,432	(89,775)	(69,749)	-	789,613	1,684,427	618,666	663,634	704,512	\$ 4,834,760
Inc. Increase-%	-	0.0%	7.3%	-1.2%	-0.9%	0.0%	10.3%	19.6%	6.0%	6.1%	6.1%	68.8%
Cum Increase -%	-	0.0%	7.3%	6.1%	5.2%	5.2%	16.0%	39.2%	47.8%	56.9%	66.6%	66.6%

<u>Sources:</u> (1) AL 4719-E p. 2. (2). AL E_4997-E p.3. (3) AL E_5041-E, p. 3

(4). AL E_5149-E-A p.5.
(5) SCE 2025 GRC Application, p. 7., Table 1

The Commission must not just address, but also arrest current trends of rising electricity

rates well in excess of the growth of household incomes, unprecedented capital spending levels,

and the proliferation of revenue adjustments that by their sheer volume and number make it

difficult to keep track of why rates are rising and affordability declining.

4.1.1 The Commission Should Find that Current Levels of Energy Rates and Bills Are Not Affordable for Many Low-Income Customers Despite Low-Income Assistance Programs

For SCE and for other California IOUs, 2019-2024 describes a period of truly astonishing

rate and revenue increases as shown below.

²⁹ Ex. TURN-02E, p. 2.

Recent History of Electricity Rate Increases³⁰



Notes:

IOU data from annual consolidated and true up advice letters.

* LADWP Rates based on assumed 500 kWh per month average usage and tier 1 total energy charge

(Source: https://rates.ladwp.com/UserFiles/Rate%20Summaries/Simplified%20Electric%20Rates/Simplified%20Electric%20Rates%20Jan%202019%20-%20Jun%202023.pdf)

The cumulative increase in SCE's system average bundled residential electricity rate from 2019 to 2023 is nearly 46%. By comparison, the cumulative rate of CPI-U over 2019-2023 has been roughly 21%³¹ for the Los Angeles area and 18%³² for California overall. This would indicate that from 2019 to 2023, SCE customers have faced a *real* increase in electricity costs of roughly 25%.³³

³⁰ Ex. TURN-02E, p. 3, citing Ex. SCE-07, Vol 4, Figure II-2, p. 11.

³¹ Ex. TURN-02E, p. 4, citing <u>https://www.dir.ca.gov/oprl/cpi/entireccpi.pdf</u>. Calculation: (310.4-263.0)/263.0=18%

³² Ex. TURN-02E, p. 4, citing <u>https://www.dir.ca.gov/oprl/cpi/entireccpi.pdf</u>. Calculation: (310.4-263.0)/263.0=18%

³³ Ex. TURN-02E, p. 4, citing US Bureau of Labor Statistics, Western Information Office Los Angeles-Long Beach-Anaheim data available at: <u>https://www.bls.gov/regions/west/news-</u> <u>release/consumerpriceindex_losangeles.htm</u>. (Calculation: Jan 2024 CPI - Jan 2019 CPI)/Jan 2019 CPI, (326.640-269.468)/269.468=21.2%

Among actions supporting affordability, SCE cites assistance programs such as CARE and FERA. SCE states that as of year-end 2020, over 25% of SCE residential accounts are receiving CARE subsidies to help pay their electric rates. This is out of about 28% eligible households across SCE's service territory. Nonetheless, roughly 150,000 of SCE's low-income customers are not receiving support from the CARE program even though eligible.³⁴ Based on 2022 data, only 12% of eligible households are receiving assistance from FERA, leaving about 350,000 eligible households unserved by CARE and FERA.³⁵ In addition to the customers eligible for CARE or FERA but unenrolled, many customers have incomes too high to qualify for energy assistance but too low to absorb the increases in electricity rates and bills without making tradeoffs among necessities. Thus, they fall though the CARE/FERA assistance gap.

4.1.1.1 CARE/FERA Income Gap Demonstrates that Low-Income Programs Do Not Fully Address the Need for Assistance.

Self-Sufficiency Standard (SSS) calculators are bottoms-up, budgetary estimates of the level of household income necessary to pay all household costs without the need for any public assistant program.³⁶ The SSS income is frequently greater than the highest income qualifying for CARE or FERA which are set at 200% and 250% of the state poverty level respectively.³⁷ Thus, comparing SSS income to CARE and FERA limits demonstrates what TURN calls the "Assistance Gap." At income levels in the Assistance Gap, households lack income sufficiency to pay all household costs timely each month, but earn too much to qualify for low-income

³⁴ Ex. TURN-02E, p. 12.

³⁵ Ex. TURN-02E, p. 13.

³⁶ Ex. TURN-02E, p. 13.

³⁷ Ex. TURN-02E, p. 13.

energy subsidy programs. The figure below shows that the CARE/FERA "Assistance Gaps" have widened since 2018. For example, in 2018 in Riverside County, the self-sufficiency income for a family of three people consisting of two adults and a school-aged child was about \$50,000, roughly the minimum to make ends meet every month, but not a great deal more. In 2018, \$50,000 in income still allowed that family to qualify for FERA, making electricity a bit less expensive, and freeing up that incremental saving for other necessities. In 2023, the self-sufficiency income is at \$96,000,³⁸ but the same family makes too much money to receive subsidy, even though it is still just getting by economically. This means that the family will have to make more trade-offs between essential services in order to remain within its finite budget. In this respect, all essential services that increase in real terms become less affordable.

(in \$)			2018					2023		
				2018	2018		2023			
			2018 FERA	CARE	FERA		CARE	2023 FERA		
		2018 CARE	Theshold	Income	income	2023 SSS	Theshold	Theshold	2023 CARE	2018 FERA
County	2018 SSS (1)	Theshold (2)	(2)	Gap	Gap	(3)	(4)	(4)	Income Gap	income Gap
Fresno	\$45,756	\$40,840	\$51,050	(\$4,916)	\$5,294	\$80,784	\$49,720	\$62,150	(\$31,064)	(\$18,634)
Imperial	\$45,908	\$40,840	\$51,050	(\$5,068)	\$5,142	\$80,088	\$49,720	\$62,150	(\$30,368)	(\$17,938)
Inyo	\$45,051	\$40,840	\$51,050	(\$4,211)	\$5,999	\$89,072	\$49,720	\$62,150	(\$39,352)	(\$26,922)
Kern	\$44,522	\$40,840	\$51,050	(\$3,682)	\$6,528	\$79,140	\$49,720	\$62,150	(\$29,420)	(\$16,990)
Kings	\$45,683	\$40,840	\$51,050	(\$4,843)	\$5,367	\$80,436	\$49,720	\$62,150	(\$30,716)	(\$18,286)
Los Angeles	\$59,936	\$40,840	\$51,050	(\$19,096)	(\$8,886)	\$106,116	\$49,720	\$62,150	(\$56,396)	(\$43,966)
Madera	\$47,938	\$40,840	\$51,050	(\$7,098)	\$3,112	\$84,804	\$49,720	\$62,150	(\$35,084)	(\$22,654)
Mono	\$55,414	\$40,840	\$51,050	(\$14,574)	(\$4,364)	\$99,896	\$49,720	\$62,150	(\$50,176)	(\$37,746)
Orange	\$64,584	\$40,840	\$51,050	(\$23,744)	(\$13,534)	\$114,951	\$49,720	\$62,150	(\$65,231)	(\$52,801)
Riverside	\$50,028	\$40,840	\$51,050	(\$9,188)	\$1,023	\$96,072	\$49,720	\$62,150	(\$46,352)	(\$33,922)
San Bernardino	\$49,020	\$40,840	\$51,050	(\$8,180)	\$2,030	\$91,212	\$49,720	\$62,150	(\$41,492)	(\$29,062)
Santa Barbara	\$58,008	\$40,840	\$51,050	(\$17,168)	(\$6,958)	\$120,449	\$49,720	\$62,150	(\$70,729)	(\$58,299)
Tulare	\$45,191	\$40,840	\$51,050	(\$4,351)	\$5,859	\$80,760	\$49,720	\$62,150	(\$31,040)	(\$18,610)
Tuolumne	\$53,734	\$40,840	\$51,050	(\$12,894)	(\$2,684)	\$91,477	\$49,720	\$62,150	(\$41,757)	(\$29,327)
Ventura	\$65,785	\$40,840	\$51,050	(\$24,945)	(\$14,735)	\$111,238	\$49,720	\$62,150	(\$61,518)	(\$49,088)

Increasing Households in the CARE/FERA "Assistance Gap"³⁹

³⁸ Ex. TURN-02E, p. 14.

³⁹ Ex. TURN-02E, p. 14, citing TURN DR 01.

4.1.1.2 Even Middle-Income Customers May Make Budget Trade-Offs to Pay for Essential Utility Services.

Similar to the CARE/FERA Assistance Gap, TURN compared SSS to the median income in each county in SCE's service territory. This comparison highlights areas where the median income, in all cases too high to receive CARE or FERA assistance, is actually *below* the SSS income for the area. This means that customers whose income approximates AR₅₀ are just getting by economically. The figure below shows the margin above SSS income at median income (roughly 50 percentile) and "moderate income" defined as 120% of median based on 2023 HUD data by California county.

(Monthly Income in \$)			2023		
Countr	2022 555 (1)	HUD Median	HUD Moderate	Median Margin	Moderate Income Margin Over SSS
Encene			free (2)		6910
Fresho	\$0,732	\$0,283	\$7,542	(\$449)	\$810
Imperial	\$6,674	\$6,283	\$7,542	(\$391)	\$868
Inyo	\$7,423	\$6,404	\$7,688	(\$1,019)	\$265
Kern	\$6,595	\$6,283	\$7,542	(\$312)	\$947
Kings	\$6,703	\$6,283	\$7,542	(\$420)	\$839
Los Angeles	\$8,843	\$7,367	\$8,754	(\$1,476)	(\$89)
Madera	\$7,067	\$6,283	\$7,542	(\$784)	\$475
Mono	\$8,325	\$7,183	\$8,621	(\$1,141)	\$296
Orange	\$9,579	\$9,583	\$11,500	\$4	\$1,921
Riverside	\$8,006	\$7,088	\$8,504	(\$919)	\$498
San Bernardino	\$7,601	\$7,088	\$8,504	(\$514)	\$903
Santa Barbara	\$10,037	\$8,046	\$9,658	(\$1,992)	(\$379)
Tulare	\$6,730	\$6,283	\$7,542	(\$447)	\$812
Tuolumne	\$7,623	\$7,329	\$8,796	(\$294)	\$1,173
Ventura	\$9,270	\$9,263	\$11,117	(\$7)	\$1,847

2023 Margin of Median and Moderate Income Above "Self-Sufficiency"⁴⁰

Sources:

 (1) Based on 2 adults and one school-aged child. SSS taken as the lower of MIT Living Wage Calculator or Family Budget Calculator (https://laborcenter.berkeley.edu/living-wage-and-self-sufficiency-tools-and-data/)
 (2) 3 person household data HUD Median and Moderate Income Data per 2023 State Income Limits, dated June 6, 2023, available at: https://www.hcd.ca.gov/sites/default/files/docs/grants-and-funding/income-limits-2023.pdf

In a number of counties, moderate income (120% of median) is insufficient! In Los

Angeles monthly income of nearly \$9,000 per month is "just making it" economically, right at

SSS. This underscores TURN's assertion that even at middle-income levels, there is little

margin in household budgets to absorb rate increases that are higher than inflation.

4.1.1.3 Disconnection Caps Should Be Maintained Indefinitely

As TURN has asserted in the Disconnections OIR (R18-07-005) and other proceedings

touching on the issues of affordability and the accessibility of essential energy services, the level

⁴⁰ Ex. TURN-02E, p. 15.

of IOU payment arrearages is an indication of the level of threat to customers' energy security. The Commission has also recognized: "The number of customers in arrears for non-payment of electric and gas bills is an indicator or a risk factor for customer disconnections and can be a proxy for the baseline number of people who could have their electricity or gas service disconnected."⁴¹ Since December 2018, the level of customer arrearages in dollars has increased nearly tenfold. Data shows that the number of customers in arrears has declined but the total dollars have increased indicating that those in arrears are further behind.

Changes in Residential Arrearages 2023 vs. 2018⁴²

	December-18	December-23	Dollar Change	% Change
Dollar Amount Res	\$82,623,852	\$862,053,149	\$779,429,297	943%
Non-CARE/FERA	\$54,456,112	\$513,347,981	\$458,891,869	843%
CARE/FERA/Medical Baseline	\$28,167,740	\$348,705,168	\$320,537,428	1138%
# of Res. Customers	1,149,389	922,406		
Non-CARE/FERA	636,092	543,100		
CARE/FERA/Medical Baseline	513,297	379,306		

Sources:

Data extracted from SCE Monthly Disconnect Data Report _012019 (R.1807005) and R1807005_SCE Monthly Disconnect Data Report December 2023

Prior to the Disconnections OIR (R.18-07-005), all California IOUs experienced a surge in disconnections from 2010 through 2018 when the disconnections cap was put into place, but the experience of SCE customers was notable in that its shutoff rates approached 10% of residential customers.⁴³ Shutoffs fell disproportionally on FERA customers (a 48% increase from 2010-2016 in SCE's service territory)⁴⁴ and customers whose income was too high to

⁴¹ R.18-07-005, Disconnections OIR, Attachment 1, pp. 1-14.

⁴² Ex. TURN-02E, p. 16.

⁴³ Ex. TURN-02E, p. 17.

⁴⁴ Ex. TURN-02E, p. 17.

qualify for CARE or FERA.⁴⁵ To help customers maintain energy security despite the trend of dramatically rising rates, the Commission should find in this proceeding that given the magnitude of SCE's request and the fact that rates are already unaffordable for some customers, its disconnection cap of 4% should remain in place indefinitely.

Absent a continuation of the disconnections cap, TURN is concerned that the volume of arrearages will incentivize SCE and other IOUs to increase disconnections activity again. This would be detrimental to the energy security of customers.

4.1.1.4 SCE's Ballooning Capital Spend Obscures the Full Price Tag During the Early Years and Exacerbate the Affordability Crisis Later

SCE proposes capital spending of nearly \$8 billion annually⁴⁶ for each of the four years of its 2025 GRC cycle. SCE projects that by 2028 this investment will increase its weighted average rate base by nearly 50% relative to 2023.⁴⁷ As a result, the equity return component of the revenue requirement will grow nearly \$1.5 billion by 2028.⁴⁸ SCE's capital investment produces an incremental 5 cents in annual earnings per share for every additional \$100 million of capital spending,⁴⁹ creating a powerful incentive for SCE to spend as much capital as it can apply to its operations.

⁴⁵ Ex. TURN-02E, p. 17.

⁴⁶ Ex. TURN-02E, p. 4, citing Ex. SCE-07 Vol2 (Results of Operations) WP 32.

⁴⁷ Ex. TURN-02E, p. 4, citing EIX Fourth-Quarter and Full-Year 2023 Financial Results, dated February 22, 2024, p 13, available at: <u>https://www.edison.com/investors/events-presentations</u>. %

⁴⁸ Ex. TURN-02E, p. 5.

⁴⁹ Ex. TURN-02E, p. 5, citing EIX Fourth-Quarter and Full-Year 2023 Financial Results, dated February 22, 2024, p 27, available at: <u>https://www.edison.com/investors/events-presentations</u>

SCE characterizes every dollar of its proposal as essential for safety, reliability and the advancement of California climate policy. But the Commission must scrutinize SCE's request with the recognition that there may be conflicting interest between shareholders and ratepayers. Also, customer affordability constrains the level of utility spending that is prudent – and even possible. "The Commission has emphasized that, 'a key element of finding a charge or rate just and reasonable is whether that charge or rate is affordable."⁵⁰ Given the enormous amounts of capital spending proposed, the Commission should heed its own admonishment. "Moreover, in considering the amount of funding to authorize the Commission must balance safety and reliability with affordability and reasonable rates."⁵¹

A powerful step in balancing potentially conflicting interests is to increase transparency surrounding the full cost of capital investment programs. Multi-year capital spending initiatives generate large amounts of bonus and accelerated depreciation. Due to the significant volume of tax timing differences for ratemaking versus book recovery, capital spending reduces revenue requirements in the early years of the investment's useful life.

The significantly lower cost recovery in early years, however, obscures the full price tag of the utility's proposal. The full ratepayer cost of the capital spending may not be apparent until well into and beyond the GRC cycle in which the spending was authorized. As TURN has raised before, this results in a "pancaking of capital" on ratepayers' bills for the capital expenditures as well as their associated equity return on rate base and taxes. As the Commission explained in its 2021 Report, *Utility Costs and Affordability of the Grid of the Future: An Evaluation of Electric Costs, Rates, and Equity Issues Pursuant to P.U. Code Section 913.1*:

⁵⁰ D.21-08-036, p.19 citing D.19-05-020 at 11.

⁵¹ D.21-08-036, p.41.

The growth in rates can be largely attributed to increases in capital additions driven by rising investments in transmission by PG&E and distribution by SCE and SDG&E. While the utilities have made major financial commitments to wildfire mitigation and transportation electrification, these costs have not been fully reflected in rates so far.⁵²

For example, SCE's expected *federal* tax depreciation adjustments (shown below) are

increasing over the GRC cycle due to increasing capital spending.

Estimated Revenue Req Reduction from Federal Tax Depreciation 53

Federal Depreciation Tax	Recorded		Estimated								
Adjustments (in \$000)	2022	2023	2024	2025	202	6 202	7 2028		Totals		
Federal Tax Depreciation	\$ 2,703,331	\$ 2,986,834	\$ 3,251,764	\$ 3,592,159	\$ 3,831,920	\$ 4,079,403	\$ 4,434,426	\$	24,879,837		
Federal Cash Effect	567,700	627,235	682,870	754,353	804,703	856,675	931,229	\$	5,224,766		

Source: WP SCE-07, Vol. 2, Book B, p. 40. (Table A-4.17); Federal cash effect assumes 21% corporate tax rate.

For the 2025 GRC cycle alone, TURN estimates there will be temporary revenue requirement reductions due to *federal tax depreciation* of nearly \$16 billion, all things being equal. This amount does not include state tax depreciation benefits and as such understates the magnitude of the potential reductions in revenue requirement. These depreciation timing differences, however, will "reverse" and ratepayers will have to pay them back through higher revenue requirements in the future. As long as capital expenditures remain large and increasing, new depreciation savings will offset the reversal of tax benefits in the revenue requirement, but rate base (and customer's obligation to fund return on rate base) will grow. And this will be reflected in higher general levels of utility revenues and rates.

"...rate base has a direct relationship with the return on rate base revenue requirement that is recovered from ratepayers. The return on rate base revenue requirement reflects the opportunity for the IOU to earn a profit. Return on rate

⁵² CPUC, "Utility Costs and Affordability of the Grid of the Future: An Evaluation of Electric Costs, Rates, and Equity Issues Pursuant to P.U. Code Section 913.1" (May 2021), p. 7.
⁵³ Ex. TURN-02E, p. 6.

base may represent a return to shareholders paid by ratepayers; however, having a set rate of return ensures that IOUs are able to raise sufficient capital to make improvements to infrastructure and provide safe and reliable service to all customers. On the flip side, by having a set rate of return, IOUs are inherently incentivized to make investments to drive an increase in their rate base and therefore, their profitability.⁵⁴

With SCE's 2025 proposal, total annual capital spending has increased from figures in the \$4 to \$5 billion range to planned spending that exceeds \$8 billion in a single year.⁵⁵ Given the proliferation of capital spending, it is important to increase cost transparency for all stakeholders.

In the illustration shown below (Figure 4), including only federal tax effects, SCE's

proposed capital spending would create an apparent reduction to revenue requirements of nearly

\$3.5 billion or an average of more than \$800 million per year, compared to normalized recovery.

Although ratepayers may not feel the effects of SCE's massive capital program during this GRC

cycle, SCE's shareholders are expected to enjoy 6 to 8% compound annual growth in rate base

(and earnings, all else equal).

Values in \$000s unless indicated	2022	2023	2024	2025	2026 (3)	2027 (3)	2028 (3)	Totals
GRC Base RRQ	\$ 7,259,200	7,792,632	8,582,245	10,266,672	10,885,338	11,548,972	12,253,484	\$ 68,588,543
Inc. Increase in RRQ	-	533,432	789,613	1,684,427	618,666	663,634	704,512	4,994,284
Plus Cash Effect of Tax								
Depreciation	567,700	627,235	682,870	754,353	804,703	856,675	931,229	5,224,766
Normalized GRC RRQ								
(w/o depreciation benefits)	\$ 7,826,900	\$ 8,419,867	\$ 9,265,115	\$ 11,021,025	\$ 11,690,041	\$ 12,405,647	\$13,184,713	73,813,309
Dollar Increase in RRQ	-	592,968	845,248	1,755,910	669,016	715,605	779,067	5,357,814

Estimated Revenue Red	Reduction	of State and	Federal Ta	ax Depreciation
------------------------------	-----------	--------------	------------	-----------------

Sources: (1) AL 4719-E p. 2. (2). AL E_5149-E-A p.5. (3) SCE 2025 GRC Application, p. 7., Table 1

⁵⁴ CPUC, "Utility Costs and Affordability of the Grid of the Future: An Evaluation of Electric Costs, Rates, and Equity Issues Pursuant to P.U. Code Section 913.1" (May 2021), p. 26

⁵⁵ Ex. TURN-02E, p. 7, citing EIX Fourth-Quarter and Full-Year 2023 Financial Results, dated February 22, 2024, p 11, available at: <u>https://www.edison.com/investors/events-presentations</u>.

TURN recommends that the Commission require SCE going forward to include the "normalized"⁵⁶ revenue requirement cost of its proposals in its application. As shown in TURN's illustration, this disclosure would simply include both the proposed base revenue requirement and the revenue requirement without the cash impact of lower taxes from the depreciation timing differences. This material is already included in the utility's testimony workpapers to support its income tax showing, and as such should not be onerous to produce. The annual tax depreciation information is publicly available. Requiring such a disclosure would be an important step in clarifying the long-term cost of large capital initiatives and the impacts on affordability over time.

4.1.2 The Commission Should Find that Authorizing SCE's GRC Requested Increases Will Decrease Affordability for Customers

SCE states that the Commission has not adopted a threshold as to what level of its metrics is unaffordable.⁵⁷ It asserts that in isolation, no specific metric or analysis can determine what is affordable for *all customers*.⁵⁸ TURN agrees. However, overall SCE believes that its proposed "increase in rates and bills will not unduly impact customers' ability to pay for essential electricity service."⁵⁹ SCE's conclusions about the affordability of its request are not credible.

⁵⁶ TURN recognized that the term "normalized" has specialized meanings in tax law and regulatory accounting. Here it is used to generally mean adding back the tax effects to the revenue requirement for purposes of alternative presentation.

⁵⁷ Ex. SCE-07, Vol 4 p. 1.

⁵⁸ Ex. SCE-07, Vol 4 p. 3.

⁵⁹ Ex. SCE-07, Vol 4 p. 3.

4.1.2.1 SCE's Energy Burden & Share of Wallet Analysis Is Misleading and Inaccurate

SCE claims that if the Commission adopts SCE's clean energy investments, by the early 2030s the average SCE household can expect to see a significant decrease in total energy bills.⁶⁰ However, an examination into the assumptions behind SCE's analysis reveals that it is not only inaccurate but also misleading – the creative gymnastics engaged by SCE necessary to produce the desired result is staggering. The Commission should fully reject SCE's misleading and inaccurate analysis.

First of all, SCE's analysis does not account for the cost of purchasing an EV in order for a customer to electrify his/her transportation costs – it assumes that all customers that electrify will receive a free EV.⁶¹ This is an absurd and misleading assumption, and the Commission should be alarmed by SCE's attempt to push its narrative that average SCE households can expect to see a significant decrease in total energy bills if its clean energy investments were adopted when it contains such an absurd assumption.

Second, SCE's analysis assumes that between 2028 and 2035, SCE's "delivery cost would continue to grow at the historical 10-year growth while the generation cost would be constant at approximately 14 cents per kWh."⁶² Given that SCE's requested increase of 23% if the largest increase sought by SCE in recent GRCs, a growth at historical 10-year historical average in the future is clearly not realistic. Furthermore, the constant generation cost of 14 cents per kWh is also absurdly unrealistic. Seven years ago in 2017, SCE's generation rate was

⁶⁰ Ex. SCE-01 V01, p. 35; Ex. SCE07 V04, p. 6.

⁶¹ 11 RT 1077:9 – 1088:10.

⁶² Ex. SCE-07 V04 WP, p. 8.

7.5 cents,⁶³ and it is now 14.9 cents in 2024,⁶⁴ an increase of nearly 100% over seven years. This shows that an assumption of a constant generation cost of 14 cents per kWh is far-fetched.

Given the unrealistic and misleading assumptions used by SCE to produce its energy burden analysis that asserts a significant decrease in total energy bills, the Commission should disregard SCE's analysis.

4.1.2.2 Affordability Ratio (AR)

The Affordability Ratio (AR) metric quantifies the percentage of a representative household's income that would be used to pay for an essential utility service after nondiscretionary expenses such as housing and other essential utility service charges are deducted from the household's income. The higher an AR, the less affordable the utility service."⁶⁵ SCE's AR metrics show a history of steady decrease in affordability for both AR₂₀ and AR₅₀ (see figure below). Current AR levels have decreased by an average of more than 40% for AR₂₀ and nearly 40% for AR₅₀ customers, a significant decrease in affordability since 2022. Concerningly, the current and projected AR₂₀ values are more than twice the 4% guideline targeted in the low-income Percent of Income Pilot program. This suggests that at least for low income AR₂₀, SCE's proposal would make already unaffordable rates less affordable.

⁶³ SCE AL 3659-E.

⁶⁴ SCE AL 5178-E.

⁶⁵ D.20-07-032.

SCE Historical AR Metrics⁶⁶

			AR20					AR50		
Electric Climate Zone	2020	2022	2023	2025	% Increase	2020	2022	2023	2025	% Increase
SCE 10	5.79%	5.94%	7.5%	7.9%	36%	1.90%	1.84%	2.5%	2.6%	36%
SCE 13	6.31%	8.58%	8.4%	9.0%	42%	2.43%	2.74%	3.3%	3.5%	42%
SCE 14	8.81%	8.60%	11.8%	12.5%	41%	2.54%	2.33%	3.4%	3.6%	41%
SCE 15	10.30%	17.60%	13.1%	13.8%	34%	2.92%	4.00%	3.7%	3.9%	33%
SCE 16	6.21%	7.21%	8.3%	8.8%	41%	2.06%	2.10%	2.8%	2.9%	41%
SCE 5	8.43%	10.31%	10.9%	11.9%	41%	2.23%	2.19%	2.9%	3.1%	40%
SCE 6	5.71%	5.85%	7.5%	8.2%	43%	1.27%	1.19%	1.7%	1.8%	40%
SCE 8	6.45%	6.39%	8.6%	9.4%	45%	1.39%	1.41%	1.9%	1.9%	40%
SCE 9	6.86%	7.80%	9.1%	10.3%	50%	1.60%	1.63%	2.2%	2.3%	41%
Average	6.60%	7.21%	8.7%	9.4%	43%	1.74%	1.76%	2.3%	2.4%	39%

4.1.2.3 Hours-at-Minimum Wage (HW) Metrics

SCE characterizes the increase in Hours-at-Minimum Wage ("HW") as "modest."67 A

summary of SCE's HW calculations is shown below.

Figure 11: Sum	mary of SCE HW	Metric Calculation	for Essential	Service Bill ⁶⁸
----------------	----------------	--------------------	---------------	----------------------------

				203	23							2028		
SCE Climate Zone	Estimated # of Housing Units	2023 Essential Usage Average Bill	Average Bill State (\$15.50/hr.)	Average Bill Los Angeles City (\$16.04/hr.)	Average Bill Los Angeles Co. Malibu City Santa Monica City (\$15.96/hr.)	Average Bill West Hollywood (\$17.50/hr.)	Average Bill Pasadena City (\$16.11/hr.)	E Av	2028 Essential Usage Verage Bill	Average Bill State (\$15.50/hr.)	Average Bill Los Angeles City (\$16.04/hr.)	Average Bill Los Angeles Co. Malibu City Santa Monica City (\$15.96/hr.)	Average Bill West Hollywood (\$17.50/hr.)	Average Bill Pasadena City (\$16.11/hr.)
10 hot (Sec 745)	1,347,156	\$ 139.58	9.0		8.7	-		\$	160.17	10.3		10.0		-
13 hot	166,335	\$ 153.45	9.9	-	-	-	-	\$	176.10	11.4	-	-	-	-
14 hot	466,812	\$ 141.33	9.1	-	8.9	-	-	\$	162.18	10.5	-	10.2	-	-
15 hot	235,147	\$ 201.42	13.0			-	-	\$	231.20	14.9	-		-	-
16 cool	150,603	\$ 137.15	8.8	8.6	8.6	-	8.5	\$	157.38	10.2	9.8	9.8	-	9.8
5 warm	7,859	\$ 189.26	12.2	-		-	-	\$	217.23	14.0			-	
6 cool	943,130	\$ 106.49	6.9	6.6	6.7	-	-	\$	122.17	7.9	7.6	7.7	-	-
8 cool	1,246,054	\$ 106.60	6.9	6.6	6.7	-	-	\$	122.29	7.9	7.6	7.7	-	-
9 warm	1,160,081	\$ 129.70	8.4	8.1	8.1	7.4	8.1	\$	148.83	9.6	9.3	9.3	8.5	9.2

⁶⁶ Ex. TURN-02E, p. 24.

- ⁶⁷ Ex. SCE-07 Vol 4, p. 3.
- ⁶⁸ Ex. TURN-02E, p. 25, citing TURN DR_01.

The hours that must be worked at minimum wage to pay an essential usage electricity bill range from 7 to 8 hours or about 5% of monthly income (assuming 130 hours paid each month). By 2028, this is projected to rise to around 9 hours per month or 7% of monthly income. Based on typical usage, electricity represents roughly 12% of monthly income assuming 130 hours per month of paid labor and the state minimum wage estimate. These percentages are substantially higher than the 4% benchmark (for both gas and electric commodities) that the Commission set for its percent of income pilot program in all cases.⁶⁹ They are also higher than the 5% of gross income spent on energy (energy burden threshold) considered as a guideline in past Commission deliberations.⁷⁰ And for context, in 2022, the United Way estimates workers in Riverside must work 61 hours to afford a one bedroom apartment and worker in Los Angeles must work 88 hours,⁷¹ before they can even begin to concern themselves with an electricity bill. Even adjusting for 2024 minimum wage of \$16.00 per hour, after housing costs SCE's request would require between 16% and 25% of the remaining paid hours each month.⁷² This does not consider food, transportation or other essential utility services which may also be require by the household. It is hard to imagine how households with competing cost pressures could find this level of bill affordable.

⁶⁹ Ex. TURN-02E, p. 25.

⁷⁰ D.21-10-012, p. 41.

⁷¹ Ex. TURN-02E, p. 25.

⁷² Ex. TURN-02E, p. 26.

Figure 12: Calculation	of HW Metric f	for 2023 Average	Typical Usage	Bills ⁷³
rigure 12. Calculation		tor 2025 Average	i ypicar Osage	DIIIS

			2023				
Hours Worked at Minimum Wage by SCE Climate Zones (in hours except as noted)			Avg. Bill State (\$15.50/hr.)	Avg. Bill Los Angeles City (\$16.04/hr.)	Avg. Bill Los Angeles Co. Malibu City	Average Bill West Hollywood (\$17.50/hr.)	Average Bill Pasadena City
	Тур	pical Bill	2	2	Santa	2	(\$16.11/hr.)
Maximum Minimum Wa	ige		\$15.50	\$16.04	\$15.95	\$17.50	\$16.11
10 hot (Sec 745)	\$	225.72	14.6	-	14.2	-	-
13 hot	\$	242.00	15.6	-	-	-	-
14 hot	\$	227.05	14.6	-	14.2	-	-
15 hot	\$	266.17	17.2	-	-	-	-
16 cool	\$	172.92	11.2	10.8	10.8	-	10.7
5 warm	\$	129.82	8.4	-	-	-	-
6 cool	\$	143.72	9.3	9.0	9.0	-	-
8 cool	\$	171.34	11.1	10.7	10.7	9.8	-
9 warm	\$	200.74	13.0	12.5	12.6	11.5	12.5
Wt. Avg. Typical Usage Bill at							
Various Levels of Minimum							
Wage	\$	196.02	12.6	12.2	12.3	11.2	12.2
Days at Minimum Wage (i	n da	ys)	1.7	1.6	1.6	1.5	1.6

Figure 13: Calculation of HW for 2028 Projected Average Typical Usage Bills⁷⁴

2028					
Hours Worked at Minimum Wage by SCE Climate Zones (in hours except as noted)	Туј	pical Bill	Average Bill State (\$15.50/hr.) ₂	Average Bill State (Escalated per CA law) ₃	Higher than Average Escalated
Maximum Minimum Wage			\$15.50	18.36	\$20.00
10 hot (Sec 745)	\$	259.10	16.7	14.1	13.0
13 hot	\$	277.80	17.9	15.1	13.9
14 hot	\$	260.63	16.8	14.2	13.0
15 hot	\$	305.56	19.7	16.6	15.3
16 cool	\$	198.46	12.8	10.8	9.9
5 warm	\$	148.97	9.6	8.1	7.4
6 cool	\$	164.92	10.6	9.0	8.2
8 cool	\$	196.64	12.7	10.7	9.8
9 warm	\$	230.42	<u>14.9</u>	12.5	<u>11.5</u>
Wt. Avg. Typical Usage					
Bill at Various Levels of Minimum Wage	\$	225.00	14.52	12.25	11.25
Days at Minimum Wage (in days)			1.9	1.6	1.5

⁷³ Ex. TURN-02E, p. 26, citing TURN DR_01.

⁷⁴ Ex. TURN-02E, p. 27, citing TURN DR_01.

4.1.3 The Record Clearly Demonstrates that SCE's "Commitment" to Affordability Is Little More Than Lip Service to the Catch Word

SCE claims that it is committed to keeping bills and rates affordable.⁷⁵ However, the Commission should find that the record clearly demonstrates that SCE's "commitment" to affordability is little more than lip service to the catch word.

Even though SCE claims that affordability was an important consideration for its GRC application,⁷⁶ SCE's CEO and President conceded that he or his team did not provide any guidelines or boundaries in terms of percentage or dollar increase to the rest of the company regarding how affordability should be considered.⁷⁷ SCE's CEO also conceded that each department could not measure the affordability of their respective GRC requested increases since the departments did not calculate affordability metrics,⁷⁸ and that the CEO or GRC team did not provide a range of increase that it wanted the individual departments or overall request to stay under.⁷⁹ Furthermore, SCE requests an increase of 23% in revenue requirement from 2024 to 2025, which results in approximately a 10% increase on the customer bills (the other bill components not addressed by this GRC could also increase further during this GRC cycle). Yet, SCE's CEO concedes that the average income for SCE customers is unlikely to grow by 10% between 2024 and 2025, ⁸⁰ let alone 23%. SCE's CEO further concedes that the increases requested by this GRC would make bills less affordable for the average SCE customer.⁸¹

⁷⁹ 4 RT 371:6-21.

⁷⁵ Ex. SCE-01 V01, p. 31; Ex. SCE-12, p. 5.

⁷⁶ 4 RT 365:6-14.

⁷⁷ 4 RT 365:22 – 366:18.

⁷⁸ 4 RT 367:8 – 368:7.

⁸⁰ 4 RT 379:17 – 380:2.

⁸¹ 4 RT 380:3 – 381:1.

The above evidence clearly demonstrates that as much as SCE would like to claim that it is committed to affordability and that affordability was an important consideration in this application, the reality is that the extent of its "commitment" seems to be little more than lip service – no guidance was provided to the company on how to consider or evaluate affordability, and the lead policy witness acknowledges that the increases requested in this GRC would make bills less affordable for the average SCE customers since average incomes are likely to grow at a slower pace than the increases sought by SCE.

4.1.4 The Commission Should Weed Out Spending Requests that Provide Minimal Benefits While Enriching Shareholders and Only Approve Those Costs that SCE Has Demonstrated Are Both Necessary and Affordable for California Customers

While the Commission cannot address all issues affecting utility bill affordability, there

are crucial factors under the Commission's direct control. The Commission has formally

acknowledged the affordability constraint placed on the rate of growth of utility spending by the

growth rate of household budgets. As the Commission stated in its decision resolving Southern

California Edison's (SCE) 2018 GRC:

Therefore, in every instance where SCE cannot establish by a preponderance of the evidence that a request is necessary to provide safe and reliable service, we deny their requests. We do so with a goal of limiting the annual increase in SCE's revenue requirements during this GRC period to, not double the growth in customer income, but rather a true alignment with no more than that growth rate. It is only by endeavoring to meet that goal, that we can begin to strive for greater affordability.⁸²

⁸² D.19-05-020, p. 20. (Emphasis added.)

This admonition holds equally true in this GRC. In this proceeding, the Commission will determine SCE's base revenue requirement over the next four years. As the Commission has previously recognized:

The Commission can, and does, address issues related to affordability in other proceedings, especially those focused on rate design, low income energy efficiency, and the design of the CARE discount program. However, those cases address how to deal with the backend - how to ameliorate the impact of high rates and bills through other programs and cost allocation. They do not address the underlying cause of the high bills. The primary drivers of high customer bills, even with relatively low consumption levels compared to other states, are the high revenue requirements and associated high electric rates. It is in this rate case that the Commission can actually mitigate the root of the problem by weeding out spending requests that provide minimal benefit from a safety and reliability perspective.⁸³

TURN urges the Commission to apply this fair and pragmatic view to SCE's 2025 GRC

request. Because it substantially exceeds inflation and makes no effort to meet such a constraint, SCE's request will only make bills less affordable. As California emerges from the economic impacts of a global pandemic, the Commission must recognize the limitations of household budgets to absorb such high levels of growth in energy costs and only approve those costs that SCE has demonstrated are both necessary for safe and reliable service *and affordable* for California customers.

4.2 Equity Issues

4.2.1 Utility Bill Unaffordability Increases the Likelihood of Homelessness

Utility disconnections occur most often due to customer non-payment. In California renters are generally responsible for maintaining energy utilities in their homes. According to the California Association of Realtors, the template lease agreement they suggest be used

⁸³ D.19-05-020, pp. 18-19 (quoting TURN's opening brief).

includes the statement, "RESIDENT agrees to pay for all utilities and/or services based upon occupancy of the premises except _____."⁸⁴ The form provides the option for landlords to grant exemptions for certain utilities, but the default expectation is that the tenant is responsible for maintaining active utility connection. This means that for more than 5.9 million Californians who live in rental housing, a majority of whom are people of color as shown in the figure below,⁸⁵ being disconnected from utilities could constitute a violation of their lease agreement.

Black, Pacific Islander, and Latinx Californians Are Most Likely to Live in Renter Households



Share of Individuals Who Live in Renter Households by Race/Ethnicity, 2019

Source: Budget Center analysis of US Census Bureau, American Community Survey data

⁸⁴ Ex. TURN-310.

⁸⁵ Ex. TURN-03, Figure 4, p. 7. Davalos, Monica, Sara Kimberlin, and Aureo Mequita. Issue Brief: California's 17 Million Renters Face Housing Instability and Inequity Before and After COVID-19. California Budget & Policy Center. 2021.

As SCE's rates have skyrocketed, increasing 16% since 2022⁸⁶ and far outpacing wage growth, more people than ever are facing affordability challenges that could result in disconnection, eviction, and homelessness.⁸⁷



If rental agreements stipulate that a tenant is responsible for paying the utilities, having utilities disconnected due to customer non-payment is considered a violation of the lease agreement and grounds for eviction. According to research produced by San Diego State University, utility assistance programs that offer subsidized utility rates help people remain housed.⁸⁸ This is at least partially because when people are not able to pay their electric and gas

⁸⁶ Ex. TURN-02, p. 1.

⁸⁷ See Ex. TURN-03, pp. 3-4, FN 4-8, for details regarding increases in homeless populations in SCE's service territory based on data from County Point in Time Reports.

⁸⁸ See Exhibit TURN-311, "The Impact of Utility Assistance on Keeping People Housed," Mounah Abdel-Samad, PhD and Naader Ho, December 2020.

bills, they are vulnerable to losing their housing. As shown in the Table below, between one third and more than one half of all housing units in every county that SCE serves are renter occupied, which are disproportionately people of color.⁸⁹ Accordingly, the threat of eviction due to unaffordable energy bills is something TURN urges the Commission to consider seriously.

County	Renter Occupied Housing Units
	Data from ACS 2022, Table A10060 (1 year estimate)
Tulare	41.51%
Kern	39.04%
Santa Barbara	49.24%
Ventura	34.51%
Los Angeles	54.56%
San Bernardino	37.4%
Riverside	31.44%
Orange	44.2%
Kings	42.78%
Imperial	41.29%
Mono	No data

Table 1: SCE Territory Percent of Renter Occupied Housing by County⁹⁰

⁸⁹ Ex. TURN-03, Figure 4, p. 7. Davalos, Monica, Sara Kimberlin, and Aureo Mequita. Issue Brief: California's 17 Million Renters Face Housing Instability and Inequity Before and After COVID-19. California Budget & Policy Center. 2021.

⁹⁰ Ex. TURN-03, p. 4, FN 9, referencing American Community Survey (2022) https://www.socialexplorer.com/3ca97b85b6/view.

Tuolumne	No data
Inyo	No data

For SCE customers who receive Section 8 housing vouchers, unaffordable energy bills not only mean potential eviction for their current residence, but also the loss of their housing voucher and expulsion from the Section 8 program.⁹¹ Renters utilizing the Section 8 housing voucher program are required to sign a standard lease agreement, like any other renter, in addition to Section 8 specific documents with the relevant Housing Authority. The same standard lease terms apply to tenants who are using a housing voucher as to those who are not, meaning that Section 8 recipients are also in violation of their lease, and subject to eviction, if they are disconnected due to non-payment. If a Section 8 recipient is evicted by their landlord, Section 8 rules stipulate that the housing voucher will be revoked. Losing the housing voucher means that this person no longer qualifies for the rental assistance necessary to afford housing, making it extremely difficult to secure new housing. In short, utility disconnections directly contribute to evictions and homelessness for all renters and may result in chronic homelessness for California's most vulnerable renters.

In all three Public Participation Hearings held in this proceeding, SCE customers came forward to share their desperate concerns about the likelihood that they, or their fellow community members, could become homeless due to utility disconnection and eviction.

⁹¹ See Exhibit TURN-312, "Section 8 and Subsidized Housing Part 03: Evictions and Terminations, California Courts Section 8 & Subsidized Housing" Webpage, available at https://www.courts.ca.gov/section8.htm.

Table 2: Public Participation Hearing Comments Regarding Concerns of
Homelessness

Date	Location of PPH	Name and Place of Residence	Page number, line number	Comment
3/5/24	Virtual	Annah Laux- Minjnares, Newhall, CA	Page 68, Lines11-14	It's not a joke when people are talking about being homeless or having to eat cat food or dog food because you can't afford the electricity in the summer months because it's so overwhelmingly hot.
3/5/24	Virtual	Gigi Jackson, Orange County, CA	Page 52, Lines 17-21	We're concerned about our homeless populations and at this pace, if those of us who are on fixed incomes continue at this pace, we're not going to be able to continue in our house or the apartments that we live in.
3/7/34	Docket card	Jacob Diaz, Long Beach	Online docket card. Page 44 of downloaded PDF	To allow rate increases for customers during a time of massive inflation, housing crisis, homeless crisis, we believe the commission would not be representing the will, interests and safety of Californians and energy consumers.
3/9/24	Docket card	Paul Calabrese, Victorville	Online docket card. Page 42 of downloaded PDF	Please stop increasing rates. I can barely afford rent/mortgage as it is. Now you want to increase rates for electricity. California needs to realize what it's doing to its people. You are going to have more homeless than ever before why can't you guys use your head?
3/20/24	Virtual	Linda Stevens, Long Beach, CA	Page 118, Lines 13-19	Last month, my electric bill was the highest it's ever been, \$456. This month, March, my bill is \$350. So, just to struggle to pay that, that's like \$800 in two months. Someone who's retired, on a fixed income, and without family help, I wouldn't be able to pay it. So, we wonder why a lot of people end up homeless, can't pay their bills.

4/10/24	Long Beach	Angie Reyes- English, City Council member in City of Hawthorne	Page 267, Lines 1-11	Keep in mind that California has 180,000 homeless individuals today. It is up six percent from 2023. I used to be homeless living in a car. That's fact. You've got COVID you just ended. You've got people that are in the rears. Come February 1, they were all due, all rents. People can't afford that. They're not going to afford that. They will never afford that. Therefore, you are going to have people pushed out into the streets again. Homelessness, again, will rise from six percent even higher.
4/10/24	Long Beach	Andrew Mandujano, Development Manager with Long Beach Residents Empowered	Page 285, Lines 12-20; Page 286, Lines 2-6	We know that the housing crisis is severe in Southern California, oftentimes, depending on terms of tenant's leases, the inability to afford utility bills can lead to an eviction as this continuing essential service may violate potential agreements. This is specifically concerning given the 2023 point-in-time homeless count found that 60 percent of all homeless individuals in Long Beach become homeless because of an eviction. So, this proposed rate increase doesn't just directly cause an increase in monthly bills on top of housing-cost-burdened individuals. They mainly directly cause homelessness and disastrous health impacts for our aging population.
4/10/24	Long Beach	Martha Pineda, Alhambra	Page 309, Lines 4-10 and 16-22	If you proceed to move forward with this rate increase, what you're saying or what Edison and everyone included will be saying is that they don't really care about the livelihood of these communities. Raising the utility rate will lead to higher default rates overall and an increased risk of homelessness. So, we know very well, you know, how severe the housing crisis is, and it will continue to get worse, if we treat housing and utilities as a luxury, instead of a human right. This rent [<i>rate</i>] increase would lead people to their

				inability to afford utility bills, which would lead to an eviction, which would lead to homelessness.
4/10/24	Long Beach	Melodie Cuevas, Long Beach	Page 310, Lines 10-22	In a already difficult and grueling economy, a \$26 increase per month, or \$312 annually, would hurt many families. Working in many low-income communities, many families cannot even afford the cost of living at the moment, going into severe debt just by simply living. Although \$26 may not seem like much, for many families, it is choosing between food and electricity. It is a very scary position to be in. I've worked with communities who have severe utility debt, and have advocated for utility efforts for many people who have either been evicted or are in jeopardy of being so.
6/6/24	Docket card	Gayle Nollau, Crestline	Online docket card. Page 87 of downloaded PDF	When does the greed stop. Between your company and the gas company more and more people are becoming homless. I get a rate proposal notice in almost every bill. This has to stop. Enough is enough. You need to hire more professional budget analysts or let anotuher company take over that's more efficient at running electric business that can lower rates. You are constantly sending out notifications requesting an increase in rates and I'm tired of it and can't afford it.

4.2.2 Frontline Communities: Known Harms and Unknown Benefits of SCE's Transportation Electrification Grid Readiness Plan Forecast

The Commission should reject the Transportation Electrification Grid Readiness (TEGR)

supplemental portion of SCE's load growth forecast that SCE claims is required to support

medium and heavy duty (MDHD) fleet electrification in Exhibit SCE-07, Vol. 4. TURN's

primary opposition to the TEGR plan forecast, as discussed in greater detail below in Section

11.1, is that SCE has not met its burden to establish this forecasted level of spending is necessary and reasonable. Additionally, TURN highlights the inherent inequity in these costs here.

SCE acknowledges that the "TEGR request is primarily driven by MDHD electrification, not residential ownership of LD EVs."⁹² At its core, these TEGR load growth projects benefit those private companies the distribution infrastructure upgrades are being made on behalf of, while harming families who already struggle to pay their electric utility bills. These companies will benefit from a savings on vehicle fuel and maintenance costs, as supported by SCE's own witness, R. Thomas,⁹³ and can boost their public image as companies invested in environmental responsibility, while residential ratepayers will receive increased monthly bills.

TURN believes the transition to transportation electrification needs to be done equitably and not at the expense of the most vulnerable ratepayers. The argument in Exhibit SCE-13, Vol. 7 that increased electric demand will eventually result in downward pressure on rates,⁹⁴ assumes that ratepayers will be able to weather the near-term financial storm. Unfortunately, many customers already struggle to pay their utility bills, and cannot afford any more rate increases. Additionally, higher bills deter people from consuming additional electricity, and in turn, deter them from switching to plug-in electric vehicles. This means the demand for electricity may not increase as quickly as SCE claims, thus pushing back the timeline for when this "downward pressure" will occur. In the meantime, people are left to struggle to avoid disconnection in the face of unaffordable bills. This affordability crisis makes it vitally important that every dollar of ratepayer spending go toward benefitting the public good.

⁹² Ex. SCE-13, V07, p. 34.

⁹³ Hearing Transcript, Volume 11, May 15, 2024, p. 1067: line numbers 2-17 (SCE/ Thomas).
⁹⁴ Ex. SCE-13, V07, p. 34.

Although SCE and NRDC claim that vulnerable front-line communities – communities near freeways, refineries, and other pollution sources – will receive air quality benefits from the electrification enabled by the proposed load growth projects, there has been no evidence of the specific benefits of SCE's TEGR load growth projects presented in this case to substantiate that claim. When TURN asked about this issue, SCE responded to a data request by citing to a 2022 study, "Quantifying the Air Quality Impacts of Decarb and Distributed Energy Programs in California,"⁹⁵ which models the potential air quality benefits of complete transportation electrification, across all vehicle classifications, statewide.⁹⁶ While the study does offer a breakout of impacts specific to the South Coast Air Basin (SoCAB), the model used only considers "complete removal of emissions from each sector."⁹⁷ This research does not offer any insight into the incremental impacts on air quality from projects like SCE's, which only support electrification of a small fraction of the vehicles on the road, not complete removal of emissions from the transportation sector.

SCE's TEGR investments will only produce air quality benefits if forecast vehicle populations emerge to make use of SCE's infrastructure. As discussed in Section 11.1, the forecasting methodology underlying the TEGR significantly overestimates MDHD EV population growth. Given the numerous other sources of pollution in SCE territory,⁹⁸ including

⁹⁵ Ex. TURN-309: Quantifying the Air Quality Impacts of Decarbonization and Distributed Energy Programs in California, Energy and Environmental Economics, Inc. & Advanced Power and Energy Program (funded by the CPUC), 2022.

⁹⁶ Ex. TURN-309, Quantifying the Air Quality Impacts of Decarb and Distributed Energy Programs in California, EEEI & APEP (funded by the CPUC), 2022, p. 5.

⁹⁷ Ex. TURN-309, Quantifying the Air Quality Impacts of Decarb and Distributed Energy Programs in California, EEEI & APEP (funded by the CPUC), 2022, p. 24.

⁹⁸ See Ex. TURN-03, Table 1, p. 12-13.

oil refineries, meat rendering plants, and other industries, it stands to reason that the electrification of only a fraction of on-road vehicles will create, at most, a very small improvement to air quality. The marginal degree of air quality improvement the TEGR load growth projects might provide has not been quantified and should not be assumed to be significant enough to justify the harms to vulnerable ratepayers resulting from the associated rate increase.

While the potential benefits of SCE's TEGR load growth projects are unknown, the harms from unaffordable bills are very well known. High energy bills often force people to sacrifice buying food and medicine, or using air conditioning during extreme heat events, so they can afford to pay their monthly bill. That is an untenable, and potentially unhealthy, decision that SCE ratepayers are forced into making. Customers who are unable to pay their bills face disconnection, accumulated debt, and potentially eviction and homelessness.⁹⁹ SCE's proposal is an example of spending that benefits select private corporations but harms residential ratepayers.

Given the extensive harms discussed above, TURN urges the Commission to minimize to reject all unnecessary rate increases. In the context of SCE's Load Growth forecast, this measn rejecting the inflated TEGR load growth capital expenditures forecast as discussed in Section 11.1 below. The Commission must be prudent in the evaluation of residential bill impacts relative to residential benefits. Prioritizing equity in utility spending is a critical component to combatting the state's homeless crisis and protecting the wellbeing of low-income customers, as well as Black, Indigenous and other People of Color (BIPOC) customers who have been historically marginalized.

⁹⁹ Ex. TURN-03, pp. 3-5.
5. RISK-INFORMED STRATEGY AND BUSINESS PLAN

5.1 Climate Change Policy

5.2 Environmental And Social Justice Goals

5.3 Quantitative Risk Modeling

This is the third GRC in which the utility is required to present the results of its quantitative risk modeling pursuant to the S-MAP Settlement adopted in D.18-12-014.¹⁰⁰ In order to assess the current state of risk on the system, the S-MAP framework requires the utility to calculate baseline risk scores for portions of its system with homogenous risk profiles, referred to as tranches. To assist in prioritizing the utility's proposed spending of ratepayer funds, it further requires the utility to calculate risk spend efficiency (RSE) values – risk reduction divided by cost –f or each risk mitigation program, both in the aggregate (i.e., for the entire proposed program) and broken down by tranche.

Like its predecessors in this cycle of GRCs, PG&E and the Sempra Utilities, SCE attempts to downplay the usefulness of the results of this required quantitative risk analysis in CPUC decision-making. SCE states that RSEs should not be the only factor on which the CPUC should base its funding decisions¹⁰¹ -- a statement that TURN does not dispute -- but SCE goes further and incorrectly contends that the quantitative analysis required by the CPUC should be given a limited role in decision-making because, among other things, it does not account for the company's expertise and judgment.¹⁰² In other words, SCE would prefer that the CPUC rely

¹⁰⁰ D.22-12-027 revised the MAVF framework adopted in D.18-12-014 and directed utilities to transition to a new Cost-Benefit Approach in their next RAMP/GRC cycle. Accordingly, the S-MAP framework adopted in the Settlement approved in D.18-12-014 applies to SCE in this case.

¹⁰¹ Ex. SCE-01, Vol. 2, p. 18.

¹⁰² Ex. SCE-15, Vol. 5, pp. 14-15. TURN rebuts this argument with respect to SCE's grid hardening proposal in Section 15.2.4.4 below.

more on the utility's qualitative arguments, rather than the rigorous output of the S-MAP framework, which takes advantage of the best available data *and* the utility's expertise and judgment.

TURN disagrees with utility efforts to discourage use of the S-MAP quantitative analysis to promote the best use of ratepayer funds. Quantitative risk analysis is a vital tool to ensure finite ratepayer funding is used reasonably and cost-effectively. TURN agrees with D.21-08-036 that "RSE calculations are *critical* for determining whether utilities are effectively allocating resources to initiatives that provide the greatest risk reduction benefits per dollar spent, thus ensuring responsible use of ratepayer funds."¹⁰³

The value and importance of the quantitative information furnished via the S-MAP framework was an issue in controversy in this case that influenced the parties' positions regarding certain wildfire grid hardening and vegetation management programs. With respect to grid hardening, rather than rely on the results of its S-MAP analysis, SCE's undergrounding proposal is predicated on its *qualitative* designation of "Severe Risk Areas," which conflicts with the results of the S-MAP framework and amounts to an attempted end-run around the CPUC's required quantitative analysis.¹⁰⁴ In addition, as discussed in Section 15.2.2, RSE analysis shows that covered conductor is far more cost-effective than undergrounding. This important information, along with considerable additional analysis presented by TURN, supports a more modest and targeted undergrounding program than SCE has proposed and a continued reliance on covered conductor as the centerpiece of SCE's grid hardening efforts in high fire threat

 ¹⁰³ D.21-08-036, p. 38, quoting Resolution WSD-002 (June 11, 2020), p. 20 (emphasis added).
¹⁰⁴ See Section 15.2.4 below.

districts (HFTD).¹⁰⁵ Moreover, SCE's quantitative S-MAP analysis shows that, following SCE's extensive and successful deployment of covered conductor in the riskiest locations, wildfire risk has been significantly reduced and is highly concentrated, further supporting a highly targeted undergrounding program.¹⁰⁶

With respect to vegetation management, as discussed in Section 14.3, the RSE and B/C Ratio information in the record show that two programs, expanded line clearing¹⁰⁷ and the Hazard Tree Management Program¹⁰⁸ are not cost-effective and should not be funded by ratepayers.

Before responding to SCE's attempt to minimize the value of quantitative risk modeling under the S-MAP framework, TURN provides two background sections regarding, first, the definition and purpose of RSEs (Section 5.3.1) and second, the history of how quantitative risk analysis and RSEs came to be a requirement in GRCs (Section 5.3.2). In Section 5.3.3, we discuss the supplemental information added by TURN's testimony to improve the record regarding the results of SCE's quantitative risk analysis. Finally, in Section 5.3.4, we discuss why SCE's undervaluing of the S-MAP quantitative risk analysis is contrary to Commission precedent and sound policy.

5.3.1 Overview of RSEs and Risk Scores

RSE is a way to quantify the cost-effectiveness of a risk mitigation activity.

Specifically, RSE is the estimated risk reduction from an activity divided by its estimated cost,

¹⁰⁵ TURN presents its full arguments for its grid hardening proposal in Section 15.2 below.

¹⁰⁶ See Sections 15.2.3 and 15.2.6 below.

¹⁰⁷ See Section 14.1.3 below.

¹⁰⁸ See Section 14.3 below.

where both the numerator and the denominator are expressed in present value terms.¹⁰⁹ RSEs enable a comparison of the cost-effectiveness of various proposed risk reduction activities, which allows those activities to be prioritized based on cost-effectiveness.¹¹⁰ In addition, as will be discussed in Section 5.3.3 below, RSEs calculated under the S-MAP Settlement can be expressed as Benefit-Cost (B/C) Ratios in which the risk reduction value in the numerator is expressed in terms of dollars. B/C Ratios provide additional information -- whether the monetary value of risk reduction benefits exceed the costs – that enables assessment of the cost-effectiveness of a mitigation on a stand-alone basis.¹¹¹ However, just because a proposal is "cost-effective" does not necessarily mean it should be adopted by the Commission – more cost-effective solutions (*i.e.*, with higher RSEs and B/C ratios) may be available or reductions in funding may be warranted if the proposal is not consistent with affordable rates.¹¹²

In the quantitative risk modeling mandated by the S-MAP Settlement, risk is calculated by a simple equation: Likelihood of the Risk Event (LoRE) times the Consequences of the Risk Event (CoRE)., *i.e.*, LoRE × CoRE.¹¹³ The S-MAP Settlement requires SCE's LoRE and CoRE estimates to be based on data, and where necessary, to be supplemented by subject matter expertise.¹¹⁴

¹⁰⁹ D.18-12-014 (adopting S-MAP Settlement), p. A-13, Row 25.

¹¹⁰ Ex. TURN-04 (Borden), p. 3.

¹¹¹ Ex. TURN-04, p. 3.

¹¹² *Id.*, p. 15.

¹¹³ S-MAP Settlement, p. A-11, Row 13.

¹¹⁴ S-MAP Settlement, p. A-18, Row 31. *See also* Section 15.2.4.4 regarding the role of utility subject matter expertise and judgment in the S-MAP framework.

To calculate the risk reduction from a proposed mitigation activity, one compares the level of risk, or risk score, prior to applying the mitigation to the risk score after the mitigation is implemented. The difference between the *pre-mitigation risk score* (calculated as pre-mitigation LoRE times CoRE) and the *post-mitigation risk score* (post-mitigation LoRE times CoRE) is the risk reduction, the numerator in the RSE calculation.¹¹⁵

The S-MAP Settlement requires risk reductions and RSEs to be calculated at the *tranche* level to give a more granular view of how mitigations will reduce risk.¹¹⁶ Row 14 of the Settlement explains how tranches are to be determined and requires that, for each risk, assets be grouped such that each tranche has a homogenous risk profile (*i.e.*, the same LoRE and CoRE).¹¹⁷ Appropriately granular tranches are important for comparing RSEs within a single proposed program because, within a group of assets addressed by a program, there can be a variety of factors affecting the level of risk. Tranche-level RSEs enable better targeting of ratepayer dollars to the most cost-effective activities.¹¹⁸

The quantitative methodology mandated by the S-MAP Settlement is thus a rigorous and methodical approach to marshalling the utility's data and expertise in order to quantify the current state of risk on the utility's system and the cost effectiveness of the utility's proposed risk reduction activities.

¹¹⁵ S-MAP Settlement, p. A-12, Rows 19, 22, and 23.

¹¹⁶ S-MAP Settlement, p. A-11, Row 14.

¹¹⁷ *Id*.

¹¹⁸ Ex. TURN-04, pp. 17-18.

5.3.2 The Commission Developed Quantitative Risk Analysis as a Useful Tool to Prioritize Risk Reduction Spending

Beginning a decade ago, the Commission has devoted considerable time and effort to developing a quantitative methodology for prioritizing utility spending based on cost-effectiveness.¹¹⁹ In D.14-12-025, the CPUC established the RAMP as a precursor to GRCs and specified that utility RAMP submissions should present "a prioritization of risk mitigation alternatives, in light of estimated mitigation costs to risk mitigation benefits (Risk Mitigated to Cost Ratio),"¹²⁰ which is another way of describing what has come be known as RSE. In further pursuit of the goal of prioritizing utility programs based on cost-effectiveness, D.16-08-018 adopted the Multi-Attribute Value Function (MAVF) approach for quantifying risk reduction, noting that the MAVF model "enables the computation of cost-effectiveness for different mitigations."¹²¹

The Commission's years-long efforts to adopt a methodology for quantifying the cost effectiveness of utility mitigation proposals came to fruition in D.18-12-014, which adopted the S-MAP Settlement. The large utilities and intervenors agreed in that Settlement to the detailed methodology for calculating RSE, described in the previous section, as well as the requirement that utility GRCs provide a ranking of mitigations by RSE.¹²² In adopting that Settlement and making it mandatory for the utility calculation of RSEs, the Commission stated that one of the goals achieved by the Settlement was to "use risk reduction per dollar spent to prioritize

¹¹⁹ Order Instituting Investigation (OIR) 13-11-006, p. 1, identified as a goal of the proceeding to "better equip decision makers with the necessary information to ensure that we prioritize safety while continuing our long-standing mandate to ensure that adopted rates are just and reasonable."

¹²⁰ D.14-12-025, p. 32. The decision states on page 41 that it adopts the elements of the "Refined Straw Proposal" listed on page 32, which include the material quoted in the text.

¹²¹ D.16-08-018, p. 115.

¹²² S-MAP Settlement, p. A-14, Row 26.

projects.¹²³ The Commission further explained that the adopted Settlement "demonstrates success" toward a more rigorous, quantitative method of risk assessment and risk prioritization and toward "providing information required to better understand the cost effectiveness of proposed mitigations.¹²⁴

Thus, the S-MAP Settlement's RSE methodology adopted in D.18-12-014 represents the Commission's success in its decade-long effort to fill a significant void in its decision-making record in GRCs – RSE information that enables the Commission to prioritize risk reduction proposals based on a quantitative measure of cost-effectiveness. The Commission's continued commitment to this effort is shown by the modifications to the S-MAP framework ordered in D.22-12-027, designed to enhance the usefulness of the cost-effectiveness information. Those modifications require risk reduction values to be calculated in dollars, enabling direct computation of cost-benefit ratios for each proposed mitigation.¹²⁵

5.3.3 TURN's Testimony Supplements the RSE Information Presented by SCE and Presents the SCE and TURN Information in a Useful Format

As required by D.18-12-014, SCE presented RSE results for most risk reduction programs it proposes in this case and a ranking of those programs by RSE.¹²⁶ TURN supplemented SCE's RSE information in two ways.

¹²³ D.18-12-014, pp. 12, 14 (item 8, "*" indicates goal was achieved by the Settlement).

¹²⁴ D.18-12-014, p. 44.

¹²⁵ D.22-12-027, pp. 25-26. SCE was not required to implement the enhancements ordered in D.22-12-027 in this GRC, but will be required to do so in its next RAMP submission. *Id.*, p. 63 (Ordering Paragraph 2).

¹²⁶ Ex. SCE-01, Vol. 2, p. 22.

First, TURN calculated alternative RSEs using different discount rate assumptions than those that underpin SCE's RSEs. TURN's expert, Eric Borden, explained the importance of discount rates to the RSE results:

The choice of the discount rate used to develop the present value of benefits and costs is important because the benefits and costs of a risk mitigation can accrue over many years, and at times, the benefits and costs accrue over different timescales relative to one another. An RSE is therefore sensitive to the discount rates used for benefits (risk reduction, in the numerator) and costs (dollars, in the denominator).¹²⁷

SCE used a 3 percent discount rate for benefits (risk reduction, the numerator of the RSE calculation), and a weighted average cost of capital (WACC) of 7.44 percent for costs (dollars, the denominator of the RSE calculation.¹²⁸

For reasons explained in Mr. Borden's testimony, TURN believes that it would be more reasonable to use a uniform discount rate in both the numerator and denominator and that this uniform discount rate should be the WACC.¹²⁹ TURN therefore recalculated SCE's RSEs in this way and included those alternative values in the RSE tables attached to TURN's testimony.¹³⁰ TURN's adjustment decreased SCE's RSE numerators and therefore the RSEs, the magnitude of the decrease dependent on the number of years of expected benefits for the mitigation in question. The RSE reductions ranged from 6 percent to 47 percent.¹³¹

TURN's presentation of alternative values to show the sensitivity of RSEs to different discount rates proved to be consistent with D.24-05-064, even though that decision was issued

¹²⁷ Ex. TURN-4, p. 4.

¹²⁸ *Id.*, p. 7.

¹²⁹ *Id.*, pp. 7-12.

¹³⁰ Ex. TURN-4, p. 12.

¹³¹ *Id.*, pp. 12-13.

after all testimony was submitted in this case. D.24-05-064 directed the utilities to present three discount rate scenarios for their future CBR calculations, with one of the scenarios being the same as the alternative approach TURN presented in this case.¹³²

TURN emphasizes that making this one adjustment to SCE's RSEs does not mean that TURN believes the revised RSEs are without flaws. For example, in the RAMP, TURN also expressed concern about the excessive implied value of a statistical life (VSL) on which SCE's risk scores and RSEs are based, which inflates SCE's RSEs.¹³³ SCE did not fix this problem in its GRC risk modeling. Accordingly, while TURN believes that the SCE and TURN RSE values can be highly useful to inform CPUC decision-making,¹³⁴ the Commission should recognize that most of SCE's RSEs are unduly inflated.

Second, TURN converted SCE's RSE values to Benefit/Cost (B/C) Ratios. TURN's

testimony explained that, based on SCE's MAVF parameters, this conversion is accomplished by dividing SCE's RSEs by 50.¹³⁵ A B/C Ratio below 1.0 indicates that costs exceed benefits and the mitigation is not cost-effective. As discussed in Section 14.3, SCE's results show this to be the case for two vegetation management programs. TURN cautions that a B/C Ratio above 1.0

¹³² D.24-05-064 (decision in Phase 3 of the Risk-Based Decision-Making Framework docket, R.20-07-013), pp. 102-103.

¹³³ SPD Staff Evaluation Report on SCE's 2022 RAMP, A.22-05-013, Nov. 10, 2022, Attachment 3 (TURN Comments), pp. 10-14 (hereinafter "SPD RAMP Evaluation). The SPD RAMP Evaluation and the party comments were made a part of the record of this case by ALJ ruling. Tr. Vol. 10, p. 996:20 – p. 997:3. The CPUC agreed that utilities RSEs were based on excessive implied SVLs in D.22-12-027, and ordered a correction to this flaw to be applied in the new Cost-Benefit Approach. D.22-12-027, pp. 35-36.

¹³⁴ Ex. TURN-4, p. 5.

¹³⁵ Ex. TURN-4, p. 14.

does not necessarily mean that an activity should be approved.¹³⁶ As discussed above, SCE's (and TURN's discount rate-adjusted) RSEs may be significantly inflated, which likewise inflates the B/C Ratios. In addition, where a competing alternative has a higher RSE and B/C Ratio, as in the case of covered conductor compared to undergrounding, the B/C ratio should not be dispositive.¹³⁷

TURN presented SCE's RSE results, TURN's alternative RSEs based on its recommended discount rate approach, and the B/C Ratio equivalents for both sets of RSEs in tables attached to TURN's testimony and in workpapers.¹³⁸ Both tables include RSEs and B/C Ratios for the aggregated program proposed by SCE, as well as for each tranche identified by SCE.¹³⁹ The first table¹⁴⁰ presents this information organized by risk category, and the second table¹⁴¹ sorts the information from highest to lowest RSEs, using SCE's RSE values. The lower ranked discretionary activities proposed by SCE are candidates for reduced or eliminated funding, as SCE's data shows they provide low risk reduction benefit per ratepayer dollar.¹⁴²

5.3.4 SCE Undervalues Quantitative Risk Analysis

As noted in the introduction to this section, SCE attempts to downplay the usefulness of the S-MAP quantitative analysis, viewing those results as just one factor among many other

¹³⁶ Ex. TURN-4, p. 1

¹³⁷ Id.

¹³⁸ Ex. TURN-4, p. 18 and Appendix A, Tables 2 and 3. TURN provides its Excel workpapers supporting these tables and presenting different views of the information in Ex. TURN-4-Atch1.

¹³⁹ Ex. TURN-4, p. 18. TURN's use of SCE's tranches does not mean that TURN agrees that SCE's tranches are correctly delineated and sufficiently granular. *Id.*, p. 18.

¹⁴⁰ Table 2 in Ex. TURN-4, App. A.

¹⁴¹ Table 3 in Ex. TURN-4, App. A.

¹⁴² Ex. TURN-4, p. 4.

qualitative considerations that must be taken into consideration.¹⁴³ While, of course,

Commission decisions should take into account all relevant considerations, the CPUC should reject the view that RSE results are just another data point. SCE's position is at odds with the Commission's decade-long effort – and continuing efforts in R.20-07-013 -- to develop the S-MAP quantitative risk methodology; the Commission would not commit so much time and effort to the initiative unless it were meant to make a transformative improvement to its decision-making tools.

In addition, SCE's reliance on *qualitative* analysis to justify its risk reduction proposals is exactly what the Commission has been trying to improve upon. Absent risk scores and RSEs, the record would be limited to program after program in which the utility's argument boils down to an assertion that the proposed activity is necessary or important for safety. Such qualitative prose discussions offer no means for the parties or the CPUC to compare how much risk the various proposals would reduce and to assess which proposals are more important or cost-effective than others. Moreover, the risk scores and RSEs that SCE seeks to downplay are, in accordance with the S-MAP Settlement, the tip of a deep and extensive pyramid of analysis – performed by SCE itself -- of all of the key factors in deciding whether a proposed risk reduction program should be funded.¹⁴⁴

For any discretionary program designed to reduce risk, the ultimate questions the CPUC must decide concern whether the program is needed, whether it is a worthwhile use of ratepayer funds, and, if it is a worthwhile program, whether the utility's proposed scope for the program is

¹⁴³ *E.g.*, Ex. SCE-01, Vol. 2, p. 17; Ex. SCE-15, Vol. 15, Part 2, pp. 10, 14.

¹⁴⁴ See Section 15.2.4.4.

reasonable and affordable in combination with other funding decisions. Central to all of these

issues are the very questions that are addressed by the S-MAP quantitative analysis, including:

- What is the risk event the program would address?
- What is the baseline level of the risk? In that baseline risk, what is the breakdown between likelihood and consequences of the risk event, *e.g.*, is it a high consequence–low likelihood risk or a low consequence-high likelihood risk?
- What are the drivers (causes) of the risk event?
- How effective is the proposed program at mitigating the risk drivers or otherwise reducing the risk?
- How much would the proposed program reduce the baseline risk and how does that risk reduction compare to the cost of the program? Is the program providing good risk reduction value for the money it would cost?
- How does the proposed program compare to alternative programs regarding risk reduction and cost?
- How much would the risk reduction benefit vary based on the portion of the system (i.e., tranche) that is being addressed? Would a more targeted program, or combination of alternatives, be a more efficient use of ratepayer funds?

The S-MAP Settlement's RSE methodology requires the utility to do a comprehensive

analysis of all of these key issues. The resulting values should thus be highly informative in the

CPUC's decision-making process.¹⁴⁵

¹⁴⁵ SCE also takes its argument too far in contending that the S-MAP framework *precludes* it from basing its proposals solely on quantitative risk analysis and RSEs. (Ex. SCE-15, Vol. 5, Part 2, p. 19.) This contention is based on a misreading of D.22-12-027. Rather than stating that numerous other qualitative and quantitative factors *must* be taken into account, D.22-12-027 makes the correct and much more limited point that RSEs and B-C ratios "need not" serve as the sole determinants of risk mitigations to fund. (D.22-12-027, p. 26.) This statement is fully consistent with Row 26 of the D.18-12-014 Settlement which provides that utilities are "not bound" to select mitigations based "solely" on RSE rankings, but requires utilities that deviate from RSE rankings to "explain whether and how" any non-RSE factors affected the utility's mitigation selections. D.22-12-027 states that it retains the language and direction of Row 26 (*id.*), which makes clear that RSE analysis should serve a key role in the selection and justification of mitigations, consistent with the CPUC's longstanding and continuing efforts to use quantitative risk analysis to prioritize spending.

6. **DISTRIBUTION GRID**

6.1 Infrastructure Replacement

SCE describes Distribution Infrastructure Replacement (DIR) as a continuous and necessary process involving the renewal and upgrading of fundamental components of the electricity distribution system, such as poles, transformers, cable and conductors, based on the probability and consequences of in-service failures.¹⁴⁶ SCE further notes that where in-service failures have minimal consequences, a run-to-failure strategy may be the preferred approach.¹⁴⁷ The Commission needs to balance, on the one hand, the need for such DIR and the safety and reliability benefits that it may reasonably expect to be achieved through well-supported programs and, on the other, the extraordinary cost levels that can attach to such programs. Ultimately, the determination of the necessity of the proposed investments rests largely on the modeling used to determine the probability of failures and the safety and reliability consequences in cases where no replacement occurs and potential failures ensue.

• Safety and Reliability Impacts of Pre-2018 DIR Spending Cuts May Be Overstated.

It is essential that the Commission closely and carefully review SCE's proposals, and only authorize funding to the extent the utility has established that its proposal relies on compelling evidence that is sufficiently transparent. The Commission must also keep in mind that SCE unilaterally chose to significantly scale back its DIR activities and spending levels in recent years, particularly in the 2021-2024 period. After spending approximately \$500 million per year during the 2014-2018 period, SCE's recorded DIR spending for 2021 and 2022 were in

 ¹⁴⁶ Ex. TURN-5-E, p. 35, citing Ex. SCE-02, Vol. 1, Pt. 2, p. 4; Ex. SCE-02, Vol. 1, Part 2, p. 5.
¹⁴⁷ Ex. SCE-02, Vol. 1, Part 2, pp. 5, lines 22,23

the \$200 million per year range, with similar amounts forecasted for 2023 and 2024. For the 2025 test year and the remainder of the upcoming GRC period, SCE forecasts total spending in the \$700-\$800 million range.¹⁴⁸ Importantly, the record evidence demonstrates that SCE's customers did not experience significant declines in the utility's reliability or safety performance during the years of relatively lower spending. The linear fit of System Average Interruption Duration Index (SAIDI) figures reveal a near straight-line trend for the 2013-2022 period, whether viewed on a distribution-only or system-wide basis.¹⁴⁹ Similarly, the number of "wire-down" events was relatively stable during the 2018-2022 period, and the serious injuries and fatalities (SIFs) associated with such events was lower in 2018-2022 (two injuries total) as compared to 2013-2017 (ten injuries and five fatalities).¹⁵⁰

• SCE's Data and Modeling Lack Compelling Evidence to Justify the Utility's Very Substantial Proposed Increase in DIR Spending.

TURN's recommendations rely on three general arguments, in addition to the programspecific issues raised in the sections that follow. First, the Commission needs to recognize that a transmission level fault may cause poorer distribution-level reliability as measured by the SAIDI and SAIFI metrics.¹⁵¹ Secondly, SCE's approach to maintaining a steady-state replacement rate is based solely on a theoretical model lacking detailed, asset-specific historical data to substantiate an evidence-informed rate or frequency of replacement.¹⁵² Third, while TURN

¹⁴⁸ Ex. TURN-05-E, p. 36, Figure 14

¹⁴⁹ *Id.*, p. 38, Figure 17 2013-2022 SAIDI trend from SCE's 2022 Annual Reliability Report. The SAIDI trend for SCE's transmission operations slopes downwards, for reasons unrelated to SCE's distribution infrastructure.

¹⁵⁰ Ex. SCE-02, Vol. 1, Part 2, pp. 84 (Figure II-41) and 86 (Table II-21).

¹⁵¹ Ex. TURN-05-E, pp. 38-39.

¹⁵² *Id.*, pp. 39-40. It is worth noting that in SCE's test year 2021 GRC, the Commission noted that the utility there argued "that attempting to calculate a steady-state replacement rate for [infrastructure

supports SCE's use of advanced machine learning (ML) models, it is essential that these models operate with transparency and undergo independent peer reviews to ensure their methodologies are both understandable and verifiable.¹⁵³ Given the magnitude of the utility's proposed infrastructure replacement spending (\$3.6 billion in this GRC period alone) and modeling of safety and reliability implications, it is essential that SCE establish the effectiveness of its selected models, and provide stakeholders and, ultimately, the Commission with the means to understand and independently verify the models.¹⁵⁴

• Investigate Breadth of Unit Cost Calculation Issues

In addition, there is an overarching unit cost issue that may warrant further inquiry. When TURN identified the potential for unit cost estimates being higher than they should be due to a double escalation build into their calculation, SCE investigated and confirmed such issues with unit cost estimates in the "Circuit Breaker Replacement" and "Substation Transformer Bank Replacement" programs. TURN submits that further scrutiny is needed for the "Relays, Protection and Control Replacements Program" (TY 2025 forecast of \$72.3 million) and "Substation Rebuilds Program" (TY 2025 forecast of \$91.1 million).

6.1.1 Overhead Conductor Program

SCE proposes a massive expansion of its Overhead Conductor program, including ultimately targeting approximately 30,000 of non-High Fire Risk Area (HFRA) circuit miles out

replacement] planning purposes is fundamentally a 'practical impossibility' given the inherent uncertainties in forecasting a distribution asset's lifespan and would not provide meaningful information." D.21-08-036, p. 44, citing SCE's opening brief.

¹⁵³ TURN further addresses Machine Learning-related issues in the policy discussion of Section 3.2 of this brief.

¹⁵⁴ *Id.*, pp. 40-41.

of the utility's 40,000 primary overhead distribution circuit miles.¹⁵⁵ The utility proposes to reconductor approximately 420 circuit miles per year starting in 2025, a figure that is two to four times higher than any year since 2019.



Overhead Conductor Program Circuit Miles (2018-2022 recorded vs. 2023-2028 proposed)¹⁵⁶

And the associated cost is approximately \$333 million per year beginning in 2025, again a figure that outstrips any previous year's spending level.¹⁵⁷

¹⁵⁵ Ex. TURN-05-E, Figure 19.

¹⁵⁶ Ex. TURN-05-E, p. 45, Figure 20.

¹⁵⁷ *Id.*, p. 42, Figure 18.



SCE's unprecedented Overhead Conductor Program is not adequately supported.

The Commission should determine that SCE has not sufficiently justified its extensive reliance on reconductoring in non-HFRA territory during the upcoming GRC period, particularly without showing how it complements other cable and conductor replacement programs—such as the Cable-In-Conduit Replacement Program, Cable Life Extension Program, Underground Cable Replacement, and Worst Performing Circuit—that could collectively enhance the safety and reliability benefits attributed to the Overhead Conductor Program. Furthermore, of the 1,666 miles proposed for reconductoring during this GRC period, 1032 miles are "large gauge" wires for which the utility has failed to demonstrate that the associated safety and reliability benefits warrant including wires of that gauge, as opposed to the 634 miles of small gauge wire, the gauge that has historically been the focus of this project.¹⁵⁸ In addition, SCE's proposed

¹⁵⁸ Note that in its description of the OCP, SCE references 1,680 miles of circuit miles replacement. Based on TURN's calculations from TURN-SCE-072 (Q01 a,b), the total circuit miles sum to 1,666 miles

replacement of over 200 miles of bare wire conductor with like kind conductor, even while SCE suggests that bare wire conductor has disadvantages under such circumstances, makes no sense and should not be funded in rates.¹⁵⁹

• Cost-Benefit of OCP, accuracy of ML modeling, and an analysis of the synergy of OCP with other DIR programs is notably absent in SCE's analysis.

There are a number of reasons why the Commission should substantially scale back SCE's request for its non-HFRA Overhead Conductor Program in this GRC. First, the Commission needs to keep in mind that the \$330 million per year that SCE proposes to spend on overhead reconductoring activities covered in this GRC represents only a portion of the utility's total overhead reconductoring activities. That is, it represents something of an initial payment because if the Overhead Conductor Program is approved as requested, all 40,000 primary overhead circuit miles in SCE category would seem eligible for overhead reconductoring in subsequent GRCs.¹⁶⁰ Relatedly, there is a notable absence of any analysis of potential synergies between the Overhead Conductor Program (OCP) and other Distribution Infrastructure Replacement (DIR) cable and conduit replacement programs, such as the Cable-In-Conduit Replacement Program (CiC), the Cable Life Extension Program (Cle), and the Worst Performing Circuit (WPC). The Commission should adopt a scaled-back figure, rather than one that is based on such assumptions and omissions.

⁽¹⁰³² large gauge + 634 small gauge wires). Ex. TURN-05-E, Attachment 2, Response to TURN-SCE-072 (Q01 a,b).

¹⁵⁹ Ex. TURN-05-E, pp. 41-45.

¹⁶⁰ *Id.*, p. 44. (This includes 8,500 miles of Wildfire Covered Conductor Program (WCCP), as well as 1,500 HFRA miles not included in WCCP and 30,000 non-HFRA miles as part of OCP.)

Second, the Commission should be concerned with the proposed scale of the GRCproposed program. SCE's proposal to replace 1,680 miles of fully functional overhead conductors represents a 133% increase in forecasted costs compared to the 2017-2022 period, setting SCE down an exceedingly costly path without a sufficient demonstration that the program is clearly cost-effective and otherwise reasonable.¹⁶¹ SCE has presented no evidence that the impact such programs have on affordability of electricity service is even considered in the utility's decision-making process. Covered conductor is one of the costliest mitigation measures available to the utility and should be used only where less costly alternatives are inadequate. It is not just a matter of the cost of the covered conductor itself, which is approximately 1.3 times more expensive than bare conductor. The covered conductor is approximately 1.5 to 1.9 times heavier than bare conductor, increasing the likelihood that a reconductoring project will need to also include replacement of poles and other equipment sized for bare conductor but inadequate for covered conductor. There are also reduced ampacity issues and the potential for premature insulation breakdown that all suggest SCE's proposal to rely on covered conductor on a more widespread basis is not reasonable.162

Third, the reliability benefits of covered conductor as compared to bare conductor are likely exaggerated. As TURN's testimony demonstrated, SCE's reliability data is already improving with the current Infrastructure Replacement programs and recent far lower spending levels, a trend achieved without the proposed shift to covered conductor in non-HFRA areas.¹⁶³ SCE's attempt to counter that assertion through the illustration presented in Figure II-10 of its

¹⁶¹ *Id*.

¹⁶² *Id.*, pp. 45-46.

¹⁶³ *Id.*, pp. 46-47.

rebuttal testimony falls short, as it actually confirms TURN's claim that the SAIDI figure for overhead conductor and ancillary equipment ("Conductor / Splice / Connector / Tap") has been generally lower during the 2018-2022 period than the figures during the 2013-2017 period, while the SAIDI figures for "Other Overhead Equipment" drives the trend line in SCE's version of the table.¹⁶⁴ Below is the same Figure II-10, but with the trend line redrawn to reflect SAIDI for only the overhead conductor and ancillary equipment.



2013-2022 OH Equipment SAIDI, Excluding MED

And as a general matter, SCE's proposed new program scope includes in excess of 1,000 miles of large gauge conductor that were not included in the utility's 2022 RAMP filing leading up to this GRC. SCE has not adequately analyzed the safety and reliability benefits of this expansion, nor provided sufficient Risk Spend Efficiency (RSE) scores for it.¹⁶⁵ Despite TURN's opposition

¹⁶⁴ Ex. SCE-13, Vol. 1, p. 41, Figure II-10; "Other OH equipment" could be switches, transformers, capacitor banks. Ex. TURN-05-E, p. 47, citing response to TURN-072, Q. 5.c.

¹⁶⁵ *Id.*, p. 49. The Safety Policy Division (SPD) report on SCE's RAMP application labeled SCE's tranche analysis of its Contact with Energized Equipment (CEE) Risk – which only covers small gauge wire - as lacking reliability and financial consequences from the calculation of risk scores, and being not in compliance with the Settlement Agreement requirement regarding the level of granularity expected in the showing for this risk. SPD Staff Evaluation Report in A.22-05-013, p. 42.

to SCE's use of covered conductors in non-HFRA areas, if the Commission chooses to authorize the utility to proceed in this manner, it should also direct SCE to specifically target replacing small gauge wire, as this approach is better supported by historical data on wire-down incidents.¹⁶⁶



Fourth, SCE's exclusive reliance on Machine Learning-based models to assess failure possibilities and consequences as part of development of the proposed scope and prioritization is problematic. TURN's prepared testimony describes in some detail the two sub-models that comprise SCE's ML model and the shortcomings of SCE's approach.¹⁶⁷ Yet it was the model

¹⁶⁶ Ex. TURN-05-E, p. 50, Figure 24 (Large / Small / Unknown Gauge - related wire-down events by Primary Driver (2013- 2022)).

¹⁶⁷ TURN's testimony identified transparency issues (SCE's predictive model lacks detailed information about its algorithms, assumptions, model validity, and data, and validation and verification concerns (the model's effectiveness is questionable without rigorous validation against other potential models with better accuracy). Ex. TURN-05-E, p. 40.

that produced the 1,680 mile estimate of the Overhead Conductor Program's work during the 2025 GRC period. And as noted earlier, the scope of the program as proposed in this GRC is expanded beyond what was described in SCE's RAMP showing leading into the GRC.¹⁶⁸ TURN's testimony presented the results of applying an Ordinary Least Squares model using SCE-provided data, and determined that SCE's risk-reduction modeling suggests near-linear risk reduction for every unit increase in covered conductor length from 680 to 1,680 miles.¹⁶⁹ In contrast, SCE's previous modeling shows significantly diminishing returns from covered conductor in HFRAs, with the first approximately 500 miles showing the steepest safety and

¹⁶⁹ *Id.*, pp. 52-53.

TURN also raised a general critique regarding the cost-benefit imbalance. SCE's RSE scores for the OCP increased from an RSE score of 48 in the RAMP to184 in the GRC as detailed in Table II-24, page 98 of SCE-02, Vol. 1, Pt. 2. Part of this discrepancy may be due to the fact that the RSE / Benefit Cost Ratio (BCR) calculations in the RAMP filing considered only small gauge wires as mitigations, and large gauge wires were introduced to SCE's proposal post-RAMP, and without any supporting benefit-cost analysis included. *Id.*, pp. 56-57.

¹⁶⁸ Ex. TURN-05-E, p. 49.



reliability benefits as shown in the figure below.¹⁷⁰

TURN's recommendations for the Overhead Conductor Program (OCP) are as follows:

- The Commission should deny funding for the entire program other than the Accelerated OCP (AOCP) element. SCE's proposal to spend upwards of \$330 million annually (from 2025-2028) would be unduly burdensome on ratepayers, and unlikely to achieve the projected reliability and safety benefits.
- If the Commission chooses to authorize funding for more than the Accelerated OCP element, the funding should be tied to small-gauge conductor replacements (634 miles of the 1,680 miles SCE has proposed) consistent with the higher failure probability of such conductor and, by extension, the higher projected safety and reliability benefits from focusing on their replacement.
- The Commission should deny ratepayer funding for replacements using bare conductor. While SCE's proposal involves a relatively modest funding request, the plan to replace

¹⁷⁰ *Id.* The figure is from Ex. TURN-110, using data SCE provided in its response to TURN DR 125, Question 4.

208 miles of bare conductor with ratepayer funds in 2023-'24 contradicts their own arguments presented in the Overhead Conductor Program (OCP). The utility's data has previously shown that bare conductors are linked to energized wire-down events, which compromise both safety and reliability.

- The Commission should direct SCE to assess the benefits and costs of alternatives such as replacements of splice, connector or tap equipment as a lower-cost alternative to conductor replacement.
- Going forward, SCE must take steps to ensure that its reliance on machine learning (ML) models are less opaque, such as sharing the assumptions, testing multiple models, showing confidence intervals and ranges in results, and demonstrating the final model's performance against other potential models.¹⁷¹

The authorized spending for the 2023-2025 forecasts resulting from TURN's

recommendations would be \$111.76 million, as compared to SCE's proposal for \$458.451 million over that period.

6.1.2 Underground Cable Replacement

SCE's Underground Cable Replacement Program aims to preemptively address risks in

high-risk primary underground cables.¹⁷² SCE reports spending approximately \$6 million to \$18

million per year in the 2021-2024 period, but proposes to spend an average of approximately

\$100 million per year for the 2025-2028 period.¹⁷³ The cumulative amount for 2023-2028

represents a 37% increase to the cumulative recorded amount for 2017-2022.¹⁷⁴

¹⁷¹ TURN further addresses Machine Learning-related issues in the policy discussion of Section 3.2 of this brief.

¹⁷² Ex. SCE-02, Vol. 1, Part 2, pp. 28-29.

¹⁷³ Ex. TURN-05-E, p. 55, Figure 27.

¹⁷⁴ *Id.*, p. 37, Figure 15. The "% difference" figure in the table is stated as 14% between 2017-2022 vs. 2023-2028. \$416.45 million is \$112.51 million higher than the recorded figure of \$303.94 million, and \$112.51 million is approximately 34% of \$303.94 million.

• Root Cause Investigations in "Ghost" Cable Failures are necessary before authorizing funding for cable replacement.

SCE's proposed program scale for the Underground Cable Replacement Program again is tied to ML-based predictive analytics. The concerns with such an approach here are similar to those identified earlier with regard to the utility's Overhead Conductor Program, particularly about the lack of transparency surrounding the forecast.¹⁷⁵ The Commission should not authorize funding without first ensuring that the underlying predictive analytics accurately identify high-risk scenarios without leading to unnecessary replacements.¹⁷⁶

SCE's analysis is also questionable because the substantial majority of the cable failure causes tracked by the utility fall into the "other" or "unknown" categories. As Figure 28 of TURN's testimony (replicated below) illustrates, any analysis of cable failures tied to a specific cause is an analysis of only a very small proportion of the overall cable failures, and thus is of limited analytical value. The Commission should require SCE's showing to take a more nuanced approach with better information supporting a root cause analysis.¹⁷⁷

¹⁷⁵ Again, TURN's testimony identified transparency issues (SCE's predictive model lacks detailed information about its algorithms, assumptions, and data, making it hard to assess its accuracy) and validation and verification concerns (The model's effectiveness is questionable without rigorous validation, potentially leading to misclassification of risk levels). Ex. TURN-05-E, p. 40. Here there is also a cost-benefit imbalance, as the significant investment in SCE's aggressive replacement strategy may not result in proportional benefits given that the causes of more than 90% of cable failures is "unknown" or "other. *Id.*, p. 57, Figure 28. Finally, there are also concerns about the shift in methodology, that is, the shift from an age-based to a risk-based replacement model, a change that introduces complexities that could result in resource wastage or increased failure risks if not managed accurately. *Id.*, p. 56.

¹⁷⁶ Ex. TURN-05-E, p. 56.

¹⁷⁷ *Id.*, pp. 56-57.



TURN advocates for a data-driven age threshold for underground cable replacement.¹⁷⁸ Furthermore, when proposing replacements for equipment below the established age threshold, detailed justification should be mandatory. This justification should extend beyond the standard evidence required for such equipment, including additional documentation that demonstrates all preventative measures have been exhausted. It should also clearly establish that the need for replacement is not merely due to preventative or corrective maintenance issues or delays, which should be substantiated by specific notification orders.¹⁷⁹

TURN's recommendation for the Underground Cable Replacement Program is that the Commission authorize funding based on 800 miles of such replacement, rather than SCE's proposed 1,600 miles, as the reduced number of miles will still enable achievement of 60-70% of the safety and reliability benefits according to SCE's models. The resulting funding for the

¹⁷⁸ TURN's analysis suggests the existing 41-year threshold for mainline cable replacement might need to be lowered, based on historical failure rates and transparent age determination. Ex. TURN-05-E, pp. 57-58.

¹⁷⁹ *Id.*, p. 9

2023-2025 period on a forecast basis is \$65.15 million, as compared to SCE's forecast of \$114.1 million for that period.¹⁸⁰

6.2 Inspection And Maintenance, And Capital-Related Expense

TURN's analysis revealed a declining trend in the number of "closed notifications" for both preventive and breakdown maintenance.¹⁸¹ "Closed notifications" indicate the completion of identified maintenance work by SCE. If SCE maintained a rising or stable trend in completed maintenance notifications, it would suggest effective maintenance management, and the Commission could reasonably anticipate a reduction in the need for capital replacements.¹⁸² But instead, SCE's numbers for closed notifications for both preventive and breakdown maintenance reflect a generally downward trend over much if not all of the 2018-2022 period. Such a trend may suggest reduced levels of routine maintenance and, by extension, a need for increased capital expenditures.

• SCE's pattern of declining trend in maintenance notifications and Inspect App efficacy.

In its rebuttal testimony, SCE only challenged TURN's analysis on the basis of TURN having relied on "closed notifications" that reflected the year the maintenance need was <u>found</u> rather than the year the maintenance was <u>closed</u>.¹⁸³ In discovery, SCE provided TURN with the 2018-2022 data that reflected the year the maintenance was <u>closed</u>. The pattern TURN had identified in its testimony was virtually unchanged.¹⁸⁴

¹⁸⁰ *Id.*, pp. 55 and 58 (Figure 29).

¹⁸¹ Ex. TURN-05-E, pp. 9-10.

¹⁸² *Id.*, p. 10.

¹⁸³ Ex. SCE-13, Vol. 2, p. 28.

¹⁸⁴ Ex. TURN-100 (Response to TURN-115, Question 5 and TURN graphs based on the provided data for 2018-2022). SCE's data request response chose to add 2023 data as well, likely to permit the utility to

TURN also raised concerns regarding the extent of SCE's reliance on "Inspect App," a software used as part of SCE's inspection process that requires entry of responses to 80-161 survey questions and mandatory photographs for each structure, as part of its inspection process. The reliance on Inspect App rather than its predecessor required additional equipment and contractors to support the inspection process, and by SCE's calculations has tripled the amount of inspection time per structure.¹⁸⁵ TURN's analysis shows the claimed tripling of inspection time per structure is understated.¹⁸⁶ Furthermore, the benefit of this expanded use of Inspect App and the increased costs associated with the extra materials, consultants, and inspection time remains unsubstantiated.

TURN does not present a specific forecast adjustment tied to the reliance on Inspect App within SCE's Distribution Ground Inspections program, due to a lack of relevant supporting data supporting the use of Inspect App. However, the Commission should direct SCE to reevaluate its use of the more extensive (and expensive) inspection effort for routine and compliance-based inspections outside of the High Fire Threat District, due to its unproven efficacy for such purposes. and consider relying instead on the inspection methods that preceded the increased reliance on Inspect App beginning in 2020.

TURN's recommendations for Inspection and Maintenance and Capital-related Expense are:

• The Commission should direct SCE to reevaluate and, as appropriate, scale back its deployment of and reliance on Inspect App for routine and compliance-based inspections,

point out the "uptick" such data would create if it were included in TURN's graphs. Ragsdale, SCE, 4 RT 421, ll. 9-14. TURN does not disagree that the 2023 data creates an uptick from the 2022 data, but it also does not materially change the overall trend for the 2018-2022 period.

¹⁸⁵ Ex. TURN-05-E, p. 12.

¹⁸⁶ *Id.*, p. 14.

given the clear underestimation of time and resources needed for Inspect App over traditional inspection methods, especially outside of High Fire Threat Districts.

• The Commission should direct SCE in its next GRC showing to analyze and address any trends in preventive or corrective maintenance and the impacts on associated capital replacement activities, broken out by each category of Distribution Infrastructure Replacement activity. This review should consider, at minimum, any trends in maintenance completion as indicated by "closed notifications."

6.3 Safety And Reliability Investment Incentive Mechanism

7. METER ACTIVITIES

- 7.1 Meter O&M
- 7.2 Meter Capital

8. TRANSMISSION GRID

- 8.1 Transmission Grid O&M
- 8.2 Transmission Grid Capital Expenditures
- 8.3 Transmission Infrastructure Replacement

9. SUBSTATION

- 9.1 Substation O&M
- 9.2 Substation Capital

9.3 Substation Infrastructure Replacement

SCE's Substation Infrastructure Replacement Program focuses on pre-emptive

replacement of equipment and structures based on a determination they are in poor condition per

the utility's heath index, aged and obsolete.¹⁸⁷ SCE's direct testimony forecasted \$1.629 billion for substation infrastructure replacement from 2023 to 2028. The upward trajectory of such replacement expenditures is coupled with a downward trend in the volume of routine maintenance activities, which TURN submits must raise concerns with the Commission.¹⁸⁸ TURN recommends a relatively small reduction to SCE's forecast for this GRC period, and a directive to the utility to implement more stringent verification of replacement needs based on a more rigorous and transparent analysis of the various components of its health indices, unit cost calculations underlying its forecasts, and the actual condition of equipment.

9.3.1 Circuit Breaker Replacement Program

SCE's Circuit Breaker Replacement Program has displayed a pattern of recording high costs for the replacement work, even in years when it is replacing a far lower number of units. In order to better understand SCE's costs and practices, the utility needs to provide more granular data for the circuit breakers being replaced, particularly the varying voltage levels and their corresponding unit costs.¹⁸⁹

• SCE's incorrect unit cost calculations, non-linear Health Index scale, and faulty Weibull analysis.

SCE's records reveal approximately 70 in-service circuit breaker failures over the last thirteen years, for a failure rate of approximately 0.04% per year. TURN agrees that SCE should make reasonable efforts to avoid having such equipment fail in service, but the utility needs to develop a more effective replacement strategy to ensure reasonable costs. This is particularly so

¹⁸⁷ Ex. SCE-02, Vol. 5, p. 132.

¹⁸⁸ Ex. TURN-05-E, p. 16.

¹⁸⁹ Ex. TURN-05-E, pp. 16-17.

where, as here, SCE seeks funding to proactively replace 1,190 during this GRC period, or approximately 10% of the 13,000 circuit breakers currently in service.¹⁹⁰ For example, SCE fails to provide data regarding reliability-related or safety-related costs and impacts incurred due to circuit breaker failures, and does not address how its proposed program would reduce those costs or impacts. Similarly, the utility does not track benefits from its prior replacements in terms of improved safety or reliability. Rather than a deliberate, targeted replacement approach supported by data, SCE replaces equipment more as opportunities arise, an approach TURN's testimony likened to playing whack-a-mole.¹⁹¹

The analysis of circuit breaker replacements needs to keep in mind the various other layers of defense that serve to avoid catastrophic failures and ensure system reliability and safety. TURN's testimony identified two other major Substation Infrastructure programs (Relays, Protection and Controls Replacement, and Substation Rebuilds), both of which have substantial forecasted capital spending and neither of which TURN is challenging here.¹⁹² It is reasonable to expect there will be synergies among these various programs that may reasonably be expected to reduce the need for circuit breaker replacements. For example, upstream relay protection or fuse-based systems are designed to preemptively isolate a circuit breaker prior to a "catastrophic failure" occurring. Preventive measures like regular maintenance and thermal imaging, along with the unexplored synergies of SCE's major substation programs, may contribute significantly to system reliability and safety without the immediate need for costly

¹⁹⁰ Ex. TURN-05-E, p. 20.

¹⁹¹ Ex. SCE-13, Vol. 5, p. 37 (SCE refers to "operational efficiencies" as an umbrella term for replacements that are not justified based on its own health index); Ex. TURN-5-E, pp. 18-19.

¹⁹² Ex. TURN-5-E, p. 19. The test year 2025 forecasts for "Relays, Protection and Controls Replacement" and "Substation Rebuilds" are \$91.1 million and \$72.3 million, respectively.

replacements.¹⁹³ The forecast for circuit breaker replacements should reflect a more integrated analysis of the various programs and practices that will impact replacement needs, rather than assessing substation equipment replacements as if they were a stand-alone activity.

SCE's "health index" based positions are undercut by the utility's deployment of a nonlinear scale. Rather than rely on SCE's index as presented, the Commission should direct the utility to revise the underlying index to apply a linear scale going forward. Per its health index, SCE asserts that of the approximately 12,900 circuit breakers in service, 2,000 are in "Poor" or "Very Poor" condition. The Health Index developed by SCE integrates various parameters sulfur hexafluoride gas purity for SF6-based CBs, oil circuit breaker analysis for oil CBs, operational frequency, contact resistance tests, and overstress percentages. But the underlying calculations add up to 175% of the number of circuit breakers, rather than the 100% as one would expect for a weighted scoring scale.¹⁹⁴ Furthermore, the non-linear nature of SCE's scale results in an overestimation of the number in the "poor" and "very poor" categories.¹⁹⁵ Finally, SCE applies its health index in a non-uniform manner, as its recorded and forecasted replacements include a number of circuit breakers that fall into the "Very Good" to "Fair" condition categories, without any clear justification for those replacements.¹⁹⁶ SCE's implementation of a non-linear scale has led to an overestimation of equipment in poor and very poor conditions.

¹⁹⁶ Ex. TURN-05-E, p. 23.

¹⁹³ *Id.*, pp. 19-20.

¹⁹⁴ *Id.*, p. 21.

¹⁹⁵ Ex. TURN-107 (Response to TURN DR 115, Question 2 and associated TURN calculations). In the linear scale, the sum of 'Poor' and 'Very Poor' circuit breakers is 13.1%, and for transformers, the corresponding sum is 8.6%. Conversely, in the non-linear scale, circuit breakers classified as 'Poor' or 'Very Poor' sum to 16.3%, while transformers in these categories total 20.2%.

The Commission should also not rely on the results of SCE's Weibull reliability analysis. Weibull analysis is a statistical model used for forecasting failure rates and assessing product reliability based on historical data. Though the Weibull model is frequently used in reliability analysis, the outputs can only be as good as the inputs, and TURN's analysis demonstrated that SCE's model inputs were fundamentally flawed.¹⁹⁷ In addition, SCE used a relatively small data set (from 2016-2022) to predict failure rates extending for a twelve-year period (2023-2034), without justifying why it did not use pre-2016 available data. And SCE did not meaningfully evaluate any alternatives to the Weibull model.¹⁹⁸ In light of these elements, the Commission should not place any stock in SCE's Weibull-based analysis.

One clear area for improved efficacy of this program would be to target replacement of Oil and Air Magnetic circuit breakers, as the median age of that equipment is 59 years, a dramatic difference as compared to the median ages of 16 and 10 years for Gas and Vacuum circuit breakers, respectively.¹⁹⁹

SCE's analysis was not helped by the utility's reliance on erroneously escalated unit costs in its calculations. TURN's testimony had flagged the incorrect arithmetic underlying SCE's figures, due to the utility having applied double escalation. Correcting for this double escalation resulted in a 2% reduction to SCE's average cost figure for circuit breakers in the 220-500 kV voltage class, and a 46% reduction for the 2.4-115 kV voltage class.²⁰⁰ SCE's rebuttal

¹⁹⁷ Ex. TURN-05-E, p. 25. TURN described how SCE assumed that "replacement rates" were the same as "failure rates," as if all replacements had been the result of an in-service failure, when in fact there were relatively few in-service failures.

 ¹⁹⁸ *Id.* TURN's testimony identified lognormal and gamma models as alternatives that could potentially provide better estimates – or at least provide a relative comparison against Weibull-based point estimates.
¹⁹⁹ Ex. TURN-05-E, p. 25.

²⁰⁰ Ex. TURN-05-E, pp. 20-21.

testimony simply denied the error, without any indication that the utility had checked its calculations.²⁰¹ In response to a TURN data request, SCE appears to have re-checked its calculations, and acknowledged that it had incorrectly applied escalation.²⁰² Given that the double-escalation issue cropped up in two key programs, TURN recommends that the Commission direct SCE to broaden its review to <u>the remaining two programs (i.e.</u> "Relays, Protection and Control Replacements Program" and "Substation Rebuilds Program") included in its "Substation" volume, and either confirm that there was no double escalation for the other programs, or make the necessary adjustments to its forecast to remove such double escalation.

As a general matter, SCE's direct showing in a GRC should provide detailed workpapers that clearly demonstrate how project-level costs are derived and translate into unit costs. These should also show the flow of these costs from historical project level costs to unit costs and finally into the annual forecasts. This approach ensures basic transparency and allows for verification of unit cost assumptions. SCE should also demonstrate that it has more fully assessed options such as diagnostic testing, refurbishment and remedial work as less costly alternatives to replacement.²⁰³

TURN's recommendations for the Circuit Breaker Replacement Program are:²⁰⁴

²⁰⁴ *Id.*, p. 27.

²⁰¹ Ex. SCE-13, Vol. 5, p. 32.

²⁰² Ex. TURN-107 (SCE Response to TURN DR 115, Questions 1, 3 and 4), Response to Question 1. . However, SCE also took the opportunity to identify another error that it claimed caused its unit costs to be understated and included adjustments in the utility's favor. TURN did not have a sufficient opportunity to review SCE's additional information regarding purported corrections extending beyond the doubleescalation in unit cost issue that both parties had previously addressed in prepared direct and rebuttal testimony, which is not surprising where the additional information arrived at the outset of evidentiary hearings.

²⁰³ Ex. TURN-05-E, p. 20.

- The Commission should authorize \$153.13 million rather than SCE's forecast of \$164.29 million for the 2023-'25 period, a reduction of approximately 6.8%.
- The Commission should direct SCE to target replacement of only those circuit breakers that are deemed to be in "poor" or "very poor" condition according to SCE's arbitrary, non-linear Health Index. Circuit breakers in "very good," "good," and "fair" condition should not be replaced as part of the Infrastructure Replacement Program.
- The Commission should direct SCE in its next GRC to:
 - Present a more detailed analysis of unit costs, and break out its circuit breaker proposals into more granular voltage classes, rather than relying on two broad classes.
 - Substantiate the accuracy of its chosen health index, or to modify that index to comport with TURN's suggestion of a linear, unbiased index.
 - Develop a replacement approach that, like the 41-year threshold for Cable Replacement, establishes the need to replace substation equipment that fails earlier than a designated age threshold.²⁰⁵ The Commission should also encourage SCE to engage in a stakeholder-involved process in order to develop such thresholds for equipment included in SCE-02, Vol. 02 ("Substations"). The results of this effort should be reported and reflected in SCE's next GRC showing.

9.3.2 Substation Transformer Bank Replacement Program

TURN's analysis of SCE's proposed funding for the Substation Transformer Bank

Replacement Program addressed issues similar to those raised with regard to the Circuit Breaker

Replacement Program and discussed in the preceding section of this brief. In addition, after

serving its rebuttal testimony, SCE acknowledged that the double escalation error identified for

²⁰⁵ In the context of substations, while age is incorporated into SCE's health index, establishing a definitive age threshold is crucial. For plant that has not yet reached the established threshold, SCE must present a compelling demonstration to justify equipment replacements. TURN's analysis of the age distribution of transformers and circuit breakers suggests a median age of 9-23 years for circuit breakers and 12-24 years for transformers, indicating that the overall fleet is relatively new.

its circuit breaker unit cost calculations extended to its unit cost calculations here as well, and provided corrected figures.²⁰⁶

• Persistent Incorrect Unit Cost Calculations, Skewed Health Index, and Faulty Weibull Analysis.

TURN's prepared testimony summarized issues such as the absence of data that might support SCE's unit cost calculations, the use of a non-linear health index, and errors in SCE's calculation of the rate of replacement and its Weibull analysis.²⁰⁷ The current replacement strategy lacks sufficient justification in unit cost calculations and health index assessments, leading to potentially unnecessary replacements. Historical data does not support the urgency of replacing transformers with a median age well below industry standards for end-of-life, particularly those categorized as being in 'Good' or 'Fair' health.

TURN's analysis suggests a median age of 12 years for A-Bank transformers, and 24 years for B-Bank transformers, suggesting a reduced need for a preemptive replacement strategy given the longer lives expected for such equipment.²⁰⁸ And TURN explained that under SCE's approach the utility would include replacements of 94 transformers in "fair" or "good" condition under SCE's own non-linear health index, out of a total of 298.²⁰⁹

SCE justifies its use of the biased non-linear Health Index scale by citing Transformers Magazine, Vol. 7, Issue 4 from 2020 in its rebuttal.²¹⁰ The referenced table (titled "Table I – Health Index Reference Scale") suggests only "Poor" and "Very Poor" health condition

 ²⁰⁶ Ex. TURN-107 (SCE Response to TURN DR 115, Questions 1, 3 and 4), Response to Question 4.
²⁰⁷ Ex. TURN-05-E, pp. 30-31.

²⁰⁸ Ex. TURN-05-E, p. 32.

²⁰⁹ *Id*.

²¹⁰ SCE-13, Vol. 05, footnote 82.
transformers may be "at end-of-life," and only those in the "Poor" and "Very Poor" categories are likely to warrant preemptive replacement during the GRC period.

н	Condition	Description	Expected Lifetime
85 - 100	Very Good	Minor deterioration	≥ 15 years
70 - 85	Good	Significant deterioration	≥ 10 years
50 - 70	Fair	Widespread deterioration	Up to 10 years
30 - 50	Poor	Serious deterioration	≤ 3 years
0 - 30	Very Poor	Extensive deterioration	At end-of-life

TURN's recommendations for the Substation Transformer Bank Replacement Program

are:211

- The Commission should authorize funding of \$152.93 million rather than the \$182.00 million for the 2023-'25 period, a reduction of approximately 6.9%. This level of funding is consistent with replacement of 86 transformer banks from 2023-'25, as opposed to SCE's proposal of 98 for the same period.²¹²
- The Commission should direct SCE to target replacement of only those substation transformers that are deemed to be in "poor" or "very poor" condition according to SCE's non-linear Health Index. Transformers in "very good," "good," and "fair" condition should not be replaced as part of the Infrastructure Replacement Program.
- The Commission should direct SCE in its next GRC to:
 - The Commission should direct SCE to develop a replacement approach that, like the 41-year threshold for Cable Replacement, establishes the need to replace substation transformer banks above an age-based threshold that relies on historical data of in-service failures. For equipment that has not reached a specified age threshold, the utility should be required to demonstrate that all preventive, refurbishment, and other maintenance measures have been exhausted before justifying replacement.

²¹¹ *Id.*, pp. 33-34.

²¹² *Id.*, p. 34.

- The Commission should also encourage SCE to engage in a stakeholder-involved process in order to develop such thresholds for substation transformer banks and other categories of equipment. The results of this effort should be reported and reflected in SCE's next GRC showing.
- The Commission should direct SCE to present unit cost and annual forecast information at a more granular level for assets like circuit breakers and transformers, such as by specific voltage classes rather than "A-bank" and "B-bank" transformers. The more granular showing should at a minimum extend to unit costs, recorded and forecasted costs, and health condition assessments to provide a clearer understanding of asset management and investment needs.

10. GRID MODERNIZATION, GRID TECHNOLOGY, AND ENERGY STORAGE

10.1 Grid Modernization

SCE proposes to approximately double its investments in distribution automation, from

an average of \$31 million per year over the period 2018-2022 to an average of \$74 million per

year for the period 2025-2028.²¹³ This rapid increase in investment is shown in the figure below.

²¹³ Ex. TURN-06E, p. 2, citing SCE-02 Vol. 06, May 12, 2023, Table II-6, p. 35.



Figure 1. SCE proposed investments in distribution automation²¹⁴

SCE claims that this distribution automation will provide "system operators with additional visibility, situational awareness, and control,"²¹⁵ and will increase its ability to respond to dynamic grid conditions to maintain reliability and facilitate the ability of DERs to provide services to SCE's distribution system.²¹⁶

For the period 2025-2028, SCE proposes to invest \$295 million related to distribution automation, at a rate of about \$74 million per year for the 2025-2028 period. Although this represents more than a doubling in capital expenditures relative to recent years, SCE proposes to upgrade approximately the same number of circuits annually (approximately 100 circuits per year) as it has in recent years.

SCE asserts that its distribution automation investments would be cost-effective in terms of delivering reliability benefits to customers. However, these investments must be viewed in

²¹⁴ Ex. TURN-06E, p. 2, citing Ex. SCE-02 V06, p. 35.

²¹⁵ Ex. SCE-02 Vol. 06, p. 98.

²¹⁶ Ex. SCE-02 Vol. 06, p. 98.

light of the affordability crisis facing California ratepayers. While improved reliability is desirable, it must be balanced against cost. Given the massive rate increase that ratepayers are already facing, it is not the time to double spending on reliability. Instead, TURN recommends that SCE continue a measured approach to implementing distribution automation at a rate of approximately \$32 million per year and targeting the circuits with the highest benefit-cost ratios.

TURN developed a scaled back proposal by limiting the number of circuits upgraded to those with the highest net benefits in SCE's BCA workbook. Under TURN's proposal, the weighted average benefit-cost ratio under the 3/3 scheme would be increased to 9.5 and 9.3 for reliability-driven and DER-driven circuits, respectively, and would provide ratepayers with a savings of \$168 million relative to SCE's proposal. Alternatively, SCE could address substantially more circuits through implementing a 2/2 scheme. Under TURN's alternative proposal, SCE could upgrade 190 reliability-driven circuits and 60 DER circuits using a 2/2 scheme for \$170 million less. This alternative proposal has a BCR of 9.0 and 8.5 for reliabilitydriven and DER-driven circuits, respectively.

The table below summarizes TURN's primary proposal (3/3 scheme) and alternative proposal (2/2 scheme) relative to SCE's proposal. The results show how the benefit-cost ratio could be increased and costs be reduced by targeting a more limited number of high-impact circuits.

97

		SCE Proposal (3/3 scheme)	TURN Proposal (3/3 scheme)	TURN Alternative Proposal (2/2 scheme)
Reliability-	Number of Circuits	255	138	190
Driven	BCR	6.8	9.5	9.0
Upgrades	Capex Cost	\$211.5 million	\$100.2 million	\$100.5 million
DER-Driven	Number of Circuits	110	45	60
Upgrades	BCR	4.6	9.3	8.5
	Capex Cost	\$83.4 million	\$26.4 million	\$24.3 million
Total Capex (Reliability + DER)		\$295 million	\$127 million	\$125 million
Savings Relativ	ve to SCE Proposal		\$168 million	\$170 million

Table 3. TURN's proposed distribution automation upgrades and associated savings

Hence, the Commission should adopt TURN's proposal, which is not only more affordable than SCE's proposal (savings of \$168 million to \$170 million) but also more costeffective (BCR of 8.5 to 9.5 compared with 4.6 to 6.8 for SCE's proposal). The circuits that are candidates for upgrading under TURN's Proposal (3/3 scheme) and TURN's Alternative Proposal (2/2 scheme) are provided in Appendix B of Ex. TURN-06E.

Response to SCE's Rebuttal. In its rebuttal, SCE claims that its approach to demonstrating the cost-reasonableness of the proposed distribution automation investments consists entirely of a benefit-cost analysis that uses avoided customer outage costs as the benefit,²¹⁷ and that adopting TURN's proposal "would negatively impact customers and increase customer outage costs by \$940.8 million compared to SCE's proposal."²¹⁸ SCE's claim is

²¹⁷ Ex. SCE-13V06, p. 15.

²¹⁸ Ex. SCE-13V06, p. 15.

misleading as well as unreasonable and should be rejected by the Commission. First of all, the "avoided customer outage costs" asserted by SCE are *not* actual outage costs avoided by SCE customers, nor are they actual avoided costs in SCE's rates – they are calculated values using value of service ("VOS") metrics from a study performed by Nexant that was not reviewed by parties in this proceeding.²¹⁹ Furthermore, the study assumes VOS metrics that are highly inequitable – it values momentary outages for residential customers at \$4.59 and large customers at \$17,696.69,²²⁰ which is 3855 times a residential customer! Yet, residential customers pay more than 50% of distribution related revenue requirement per SCE's latest GRC Phase 2 decision.²²¹

Thus, while the VOS metrics could be used to aid the prioritization of SCE's grid modernization efforts, they should not be used as approximations for actual avoided customer outage costs. Hence, SCE's claim that customers would be negatively impacted by adopting TURN's proposal should be rejected.

10.2 Grid Technology Assessments, Pilots, And Adoption

10.3 Energy Storage

The focus of SCE's Long Duration Energy Storage (LDES) proposal is a Thermo-Chemical Energy Storage (TCES) system, which would provide 24-hour duration energy

²¹⁹ Ex. SCE-02 V06 WP, p. 70.

²²⁰ Ex. SCE-02 V06 WP, p. 70.

²²¹ D.22-08-001.

storage. SCE states that the purpose of these pilots is to accelerate the commercialization of nonlithium-ion storage technologies to facilitate greater renewable integration and resiliency.²²²

SCE seeks authorization for \$78.158 million in capital expenditures and \$0.15 million (normalized) for Test Year 2025 O&M expenses to conduct LDES pilots during the 2025-2028 time period:²²³

 2025
 2026
 2027
 2028*
 Total

 GRC Forecast
 \$9,196
 \$12,254
 \$18,730
 \$37,977
 \$78,158

Table 4. SCE Proposed LDES Capital Expenditures²²⁴

Although SCE had applied for \$140 million in additional funding from the US Department of Energy to support a 24-36 MW TCES pilot, it was not successful in attaining that funding.²²⁵ Because SCE did not receive federal funding, it proposes instead to conduct a smaller, 2 MW pilot TCES project in 2025-2026, and then "to fund technology enhancements to the preliminary TCES pilot and/or additional pilots of other LDES technologies"²²⁶ with the remaining funds during the 2027-2028 period.

Long-duration energy storage is likely to play an important role in the decarbonization of California's grid, and it will be important for utilities, including SCE, to become familiar with these technologies. However, TURN has two significant concerns with SCE's proposal:

²²² Ex. TURN-06E, p. 7, citing Ex. SCE-02 V06, p. 163.

²²³ Ex. SCE-02 Vol. 06, p. 172.

²²⁴ Ex. TURN-06E, p. 8, citing Ex. SCE-06 V06, p. 203.

²²⁵ Ex. TURN-06E, p. 8, citing Response to TURN-SCE-054 Q02.a.

²²⁶ Ex. TURN-06E, p. 8, citing SCE-02, Vol. 06, p. 204.

- The investments that SCE plans to undertake in 2027 and 2028 beyond the core TCES project are ill-defined and not adequately supported; and
- SCE has not proposed any reporting measures regarding its LDES investments in terms of spending or lessons learned.

In terms of SCE's proposed expenditures beyond the initial pilot, SCE simply notes that it plans to integrate technology enhancements to the TCES and/or conduct additional pilot projects in 2027-2028 that build on the results of the TCES pilot.²²⁷ SCE is vague in terms of what these investments would entail, explaining only that enhancements would "likely include additional energy storage capacity," and that additional pilots are "likely to be of the sodium flow and/or liquid metal battery types."²²⁸

SCE provides no detailed budget data to support its capital expenditure request of almost \$19 million for 2027 and \$38 million for 2028 for these enhancements and additional pilots. Further, SCE has not proposed a mechanism for SCE to return any potential underspend to customers, such as a one-way balancing account. Because SCE has not developed a clear scope or budget for its proposed long duration energy storage expenditures beyond the core TCES pilot, TURN recommends that the Commission reject SCE's proposal for LDES capital expenditures of \$18.730 million for 2027 and \$37.977 million for 2028.

In addition, TURN recommends that any approval for LDES investments be accompanied by annual progress reports that summarize the status of the project and funds

²²⁷ Ex. TURN-06E, p. 9, citing Ex. SCE-02, Vol. 06, p. 199.

²²⁸ Ex. TURN-06E, p. 9, citing Ex. SCE-02, Vol. 06, p. 204.

expended, and that SCE submit a report regarding lessons learned prior to its next GRC application scheduled to be filed in 2027.

In its rebuttal, SCE stated that it agrees with TURN's recommendations.²²⁹

11. LOAD GROWTH, TRANSMISSION PROJECTS, AND ENGINEERING

11.1 Load Growth

SCE's load growth request from 2025 to 2028 marks a significant increase over the previous GRC period that warrants careful scrutiny as these capital costs will impact SCE customers for decades. TURN's testimony, Exhibit TURN-07,²³⁰ provided detailed analysis and recommendations regarding SCE's forecasts for capital expenditures to support load growth in its 2025 GRC. In Exhibit SCE-13, Volume 7,²³¹ SCE presents a revised load growth capital expenditure forecast of \$3,139 million over the 2023-2028 period, and proposes a rebuttal position of \$920.781 million over 2023-2025, including \$442.890 million in the 2025 test year.²³² TURN proposes a forecast of \$660.922 million over 2023-2025, including \$237.339

The table below presents the costs forecast and proposed by SCE in its rebuttal testimony, alongside TURN's recommendations. TURN opposes all costs associated with the Transportation Electrification Grid Readiness Plan (TEGR), as well as certain project-specific

²²⁹ Ex. SCE-13 V06, pp. 46-47.

²³⁰ Ex. TURN-07E, Prepared Testimony of Sylvie Ashford Addressing Southern California Edison's Test Year 2025 General Rate Case Load Growth Investments, errata filed April 9 2024.

²³¹ This volume is broken down into three chapters on load growth, transmission projects, and engineering. TURN only addresses the subset of costs pertaining to the load growth chapter, but this does not imply support for the other portions of SCE's request.

²³² Ex. SCE-13, Vol. 7, p. 7.

costs based on analysis of SCE's workpapers. SCE's rebuttal position varies from its revised forecast in two respects. First, SCE is proposing adoption of its 2023 recorded capital expenditures rather than those forecast in opening testimony. This results in a lower request for load growth in 2023 overall. Second, SCE reduced its forecast for New Capacitors within the System Improvements Program in response to a mathematical calculation error highlighted by TURN, which also results in a lower request.²³³ As this correction addresses TURN's recommendation for the System Improvements Program, TURN does not discuss the project category in this brief.

Table 5: Summary of SCE²³⁴ and TURN²³⁵ Recommendations Load Growth, Capital Expenditures²³⁶ Nominal \$000s

Description	2023		2024		2025		2026		2027		2028		Total (2023-2025)		Total (2023-2028)	
SCE Revised Forecast	\$ 214,584	\$	299,893	\$	443,059	\$	658,550	\$	782,755	\$	739,820	\$	957,536	\$	3,138,661	
SCE Rebuttal Position	\$ 178,278	\$	299,614	\$	442,890		-		-		-	\$	920,781		-	
TURN Reductions	\$ 12,009	\$	42,299	\$	205,551	\$	305,461	\$	372,851	\$	342,929	\$	259,860	\$	1,281,101	
TEGR	\$ -	\$	32,674	\$	167,367	\$	237,357	\$	306,650	\$	287,289	\$	200,041	\$	1,031,338	
Project-Specific	\$ 12,009	\$	9,625	\$	38,184	\$	68,104	\$	66,201	\$	55,639	\$	59,818	\$	249,762	
TURN Recommendation	\$ 166,268	\$	257,314	\$	237,339		*		*		*	\$	660,922		*	

SCE is requesting 60% more for its 2025 test year capital expenditures compared to 2021, and anticipates spending more than double its load growth capital expenditures over 2025-2028 compared to the most recent GRC period.²³⁷ SCE's load growth request is primarily driven by projects under the Distribution Substation Plan (DSP) and Transmission Substation Plan

²³⁷ Ex. TURN-07E, pp. 4-5.

²³³ Ex. SCE-13, Vol. 7, p. 87.

²³⁴ SCE's recommendations are summarized in Ex. SCE-13, Vol. 7, p. 7 (Table II-6).

²³⁵ TURN's recommendations are summarized in Ex. TURN-07E, p. 2 (Table 1).

²³⁶ As described in Section 41 below and Ex. TURN-17 (Addressing Post-Test Year Ratemaking), TURN is not proposing budget-based attrition year funding for load growth capital expenditures. Reductions for 2026-2028 are displayed in the table solely to provide comparison with SCE's revised forecast.

(TSP).²³⁸ Whether or not an increase in load growth funding relative to the prior GRC period is reasonable, SCE has not demonstrated the reasonableness of its Transportation Electrification Grid Readiness (TEGR) project costs (which are more than \$1 billion in capital expenditures over the period²³⁹).

The Commission should not accept SCE's narrative that new sources of demand for electricity automatically equate to higher capital infrastructure costs. New load does not inherently mean new investments are necessary,²⁴⁰ and as SCE agrees, does not scale linearly with costs.²⁴¹ Given significant locational and technical uncertainties associated with load forecasting for new technology groups such as medium- and heavy-duty electric vehicles, there is a meaningful risk that poor infrastructure investments will inequitably harm overburdened ratepayers and increase retail rates, discouraging transportation electrification.²⁴² These risks and the affordability challenges facing SCE's ratepayers necessitate prudent spending on load growth capital projects.

Further, SCE's history of underspending in this category highlights the need for the Commission to ensure SCE has met its burden in justifying each dollar requested. SCE spent 10.9% less than it was authorized for the Load Growth Business Planning Element (BPE) in the 2021 test year.²⁴³ This variance was driven primarily by underspending of \$59 million, or 35.5%

²³⁸ Ex. TURN-07E, p. 5.

²³⁹ \$1.031 billion over 2023-2028, according to SCE's revised forecast in Ex. SCE-13, Vol. 07, which corresponds to a 2023-2025 rebuttal position of \$200.041 million.

²⁴⁰ Hearing Transcript, Volume 16, May 22, 2024, p. 1521 (Ashford/ TURN).

²⁴¹ Ex. SCE-13, Vol. 7, p. 69.

²⁴² Ex. TURN-07E, pp. 23-24.

²⁴³ Ex. TURN-07E, p. 28.

of total authorization, for the Distribution Substation Plan (DSP). SCE's 2023 recorded costs were also less than requested overall. SCE spent 46%, or \$223.244 million less than forecast across this volume,²⁴⁴ including significantly less (\$49.6 million, 85%) than forecast TSP costs.²⁴⁵ SCE even spent less (\$2.9 million, 8%) than forecast DSP costs, setting aside carryover costs.

The following sections will highlight that SCE has not met its burden to demonstrate that its proposed load growth capital expenditures forecast is just and reasonable due to the myriad of flawed assumptions it relied on when developing the TEGR. Where the Commission is unable to determine that costs are just and reasonable, including because the utility failed to meet its burden of proof, it "can and must disallow those costs: that is, unjust or unreasonable costs must not be recovered in rates from ratepayers."²⁴⁶ Accordingly, the Commission should adopt TURN's proposed reduction of \$259.860 million in capital expenditures from SCE's load growth recommendation over the 2023-2025 period, including a reduction of \$205.551 million or 46% of SCE's rebuttal position in the 2025 test year. Approximately 80% of this reduction in the 2025 test year is associated with SCE's supplemental TEGR, and 20% with other projects.

TURN removes costs for capital expenditures associated with SCE's TEGR plan because it was based on many flawed assumptions, including inputs from the 2020 Integrated Energy Policy Report (IEPR) and forecasts from the California Air Resources Board (CARB) 2020 Mobile Source Strategy that have not materialized in the past few years and conflict with the

²⁴⁴ Ex. SCE-11, p. B-2E3.

²⁴⁵ Ex. TURN-300, pp. 4-5.

²⁴⁶ D.18-07-025 (Rehearing of decision denying SDG&E rate recovery of wildfire claims costs), p. 5, quoting D.14-06-007.

California Energy Commission's (CEC's) updated 2022 and 2023 IEPR vintages. This reduction is reasonable because SCE did not employ the best available information in its forecasting, as evidenced by the demand inputs utilized in all other sections of its GRC application, and recorded data. TURN also recommends minimal reductions to non-TEGR projects based on project-specific concerns. These reductions are modest compared to those recommend by Cal Advocates, who demonstrated that aspects of SCE's "base" load growth forecast are also inflated due to reliance on inputs from the 2020 IEPR, making TURN's proposal to remove the costs of the TEGR even more reasonable. The following table presents SCE and TURN's respective forecasts for each component of the Load Growth BPE.

Table 6: Summary of SCE²⁴⁷ and TURN²⁴⁸ Recommendations by Project Area Load Growth, Capital Expenditures Nominal \$000s

Description		2023	2024	2025	2026	2027	2028	Total (2023-2025)	Total (2023-2028)	
Distribution Substation Plan (DSP)	SCE Revised Forecast	\$104,135	\$193,330	\$231,375	\$359,188	\$362,929	\$257,603	\$528,840	\$1,508,561	
	SCE Rebuttal Position	\$130,649	\$193,330	\$231,375	-	-	-	\$555,355	-	
	TURN Recommendation	\$122,621	\$155,121	\$143,717	-	-	-	\$421,458	-	
	TURN Reduction	\$8,028	\$38,210	\$87,659	\$98,880	\$159,441	\$116,440	\$133,896	\$508,657	
	SCE Revised Forecast	\$59,186	\$49,318	\$143,991	\$238,008	\$352,897	\$412,955	\$252,494	\$1,256,354	
Transmission	SCE Rebuttal Position	\$8,485	\$49,318	\$143,991	-	-	-	\$201,794	-	
Plan (TSP)	TURN Recommendation	\$4,504	\$45,228	\$45,840	-	-	-	\$95,572	-	
	TURN Reduction	\$3,981	\$4,090	\$98,151	\$186,673	\$193,363	\$206,150	\$106,222	\$692,407	
	SCE Revised Forecast	\$50,344	\$56,270	\$46,970	\$40,430	\$45,852	\$47,860	\$153,583	\$287,725	
System	SCE Rebuttal Position	\$38,205	\$55,990	\$46,800	-	-	-	\$140,995	-	
Programs	TURN Recommendation	\$38,205	\$55,990	\$46,800	-	-	-	\$140,995	-	
	TURN Reduction	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	
Land Rights Management	SCE Revised Forecast	\$920	\$975	\$983	\$1,015	\$1,030	\$1,062	\$2,877	\$5,984	
	SCE Rebuttal Position	\$939	\$975	\$983	-	-	-	\$2,896	-	
	TURN Recommendation	\$939	\$975	\$983	-	-	-	\$2,896	-	
	TURN Reduction	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	
	SCE Revised Forecast	\$0	\$0	\$19,742	\$19,908	\$20,047	\$20,339	\$19,742	\$80,037	
Driven	SCE Rebuttal Position	\$0	\$0	\$19,742	-	-	-	\$19,742	-	
Circuit Ties	TURN Recommendation	\$0	\$0	\$0	-	-	-	\$0	-	
	TURN Reduction	\$0	\$0	\$19,742	\$19,908	\$20,047	\$20,339	\$19,742	\$80,037	
	SCE Revised Forecast	\$214,584	\$299,893	\$443,059	\$658,550	\$782,755	\$739,820	\$957,536	\$3,138,661	
Totals	SCE Rebuttal Position	\$178,278	\$299,614	\$442,890	-	-	-	\$920,781	-	
Totals	TURN Recommendation	\$166,268	\$257,314	\$237,339	-	-	-	\$660,922	-	
	TURN Reduction	\$12,009	\$42,299	\$205,551	\$305,461	\$372,851	\$342,929	\$259,860	\$1,281,101	

Consistent with its position across the GRC, TURN is not recommending budget-based attrition year funding. Instead, "all non-wildfire related capital additions should be based on an escalation of the seven-year recorded average of non-wildfire related capital additions" for the post-test years.²⁴⁹ Further, TURN recommends that "the Commission should deny SCE's inappropriate request to augment its capital attrition mechanism with budgeted amounts for non-

²⁴⁹ Ex. TURN-17, p. 2.

²⁴⁷ SCE's recommendations are summarized in Ex. SCE-13, Vol 2, p. 7 (Table II-6). SCE does not propose budget-based attrition year funding except for two load growth projects detailed in Ex. SCE-07, Vol. 4.

²⁴⁸ TURN's recommendations are summarized in Ex. TURN-07E, p. 2 (Table 1). This table presents a lower recommendation in 2023 to reflect SCE's recorded underspending. As described in Ex. TURN-17, TURN does not propose budget-based attrition year funding. Reductions for 2026-2028 are displayed in the table solely to provide comparison with SCE's revised forecast.

wildfire mitigation projects."²⁵⁰ If, however, the Commission rules in favor of budget-based attrition year funding for one or more of SCE's load growth projects, TURN would propose incorporating the annual reductions to the load growth forecast identified in the table above.

11.1.1 SCE's Load Growth Forecast is Too High Due to Erroneous Assumptions in the TEGR that Inflate Anticipated Demand

SCE's "base" forecast for each project area in the Load Growth BPE is based on the California Energy Commission's 2020 Integrated Energy Policy Report (IEPR).²⁵¹ However, SCE went on to identify additional capital expenditures based on a supplemental Transportation Electrification Grid Readiness (TEGR) plan to reflect incremental new transportation and building electrification load after the state established climate policies that were not included in the 2020 IEPR.²⁵² The TEGR is a planning methodology developed without stakeholder input or review, based on outdated inputs from the 2020 IEPR, the 2020 CARB Mobile Source Strategy, and opaque customer information. As shown in Ex. TURN-07, significant future uncertainty around load location, technological and behavioral innovations create high risk for ratepayers.²⁵³ About 80% of the TEGR supplemental load comes from transportation electrification.²⁵⁴ Section IV (A) and (B) of Exhibit TURN-07 address in detail SCE's load forecasting methodology and the key contributors to SCE's unreasonably high load growth forecast. The key flaws with SCE's methodology are highlighted in the following subsections.

²⁵⁰ Ex. TURN-17, p. 9

²⁵¹ Ex. SCE-02, Vol. 7, p. 17.

²⁵² Ex. SCE-02, Vol. 7, p. 19.

²⁵³ Ex. TURN-07E, pp. 23-24.

²⁵⁴ Ex. WP SCE-02, Vol. 07 BkA, p. 94.

11.1.1.1 The EV Population Forecast SCE Relies on to Develop TEGR is Inaccurate

SCE based its electric vehicle population forecast on the CARB 2020 Mobile Source Strategy (MSS), an illustrative decarbonization pathway that has been higher by a wider margin than real-world population data each year since 2020.²⁵⁵ SCE suggests that this forecast is aligned with the latest 2022 and 2023 IEPR updates in years beyond the GRC window. TURN does not agree. SCE assumes an up to 25% larger population of light-duty EVs each year through 2031 as compared to the more recent 2022 and 2023 IEPR updates and assumes more light-duty EVs each year through the end of the forecast (2035) as compared to the 2023 IEPR.²⁵⁶



Figure 2: Comparing Statewide Light-Duty EV Forecasts²⁵⁷

²⁵⁵ Ex. TURN-07E, p. 7.

²⁵⁶ Ex. TURN-302, TURN-SCE-110_Q1, Figure II 2-5.

²⁵⁷ Forecasts provided in Ex. TURN-302 (SCE Response to TURN-SCE-110) and CEC actual data in Ex. TURN-303 (CEC EV Sales Dashboard Light-Duty and MD-HD 2022 & 2023 Sales Data).

As noted by TURN, Cal Advocates, and NRDC, SCE's medium- and heavy-duty (MDHD) forecast also diverges from recent IEPR updates.²⁵⁸ SCE assumes an up to 457% larger population of MDHD EVs each year through 2031 as compared to the most recent IEPR updates, and assumes more MDHD EVs through the end of the forecast (2035) than the 2022 IEPR.²⁵⁹ This includes an unreasonable starting point for the MDHD EV population of 23,453 vehicles in 2022, when the CEC vehicle dashboard reported there were actually just 2,186 such vehicles in California; in other words, SCE's TEGR assumed that there were nearly ten times more MDHD EVs on the road than actually existed statewide. Similarly, the TEGR forecast 32,287 MDHD vehicles in 2023, while the CEC dashboard reflects just 3,581.²⁶⁰

²⁵⁸ Ex. TURN-07, p. 9; Ex. CA-08, p. 19; NRDC-01E, p. 19.

²⁵⁹ Ex. TURN-302, TURN-SCE-110_Q1, Figure II 2-5.

²⁶⁰ Hearing Transcript, Volume 16, May 22, 2024, pages 1532-1533 (Ashford/ TURN), see also Ex. TURN-303.



Figure 3: Comparing Statewide Medium- and Heavy-Duty EV Forecasts²⁶¹

SCE's MDHD forecast diverged from reality well before SCE submitted its application. MDHD EVs are so few that SCE did not have any adoption or population data for its territory when preparing its application and had to invent a baseline territory share of the state's vehicles (33%) and possible load shapes.²⁶² SCE's Charge Ready Transport Program, which funds electrical system upgrades and charging infrastructure to support charging infrastructure for EV bus and truck fleets, has also seen very slow uptake, and SCE has previously over forecast EV adoption in applications to the Commission.²⁶³ SCE's own internal 2022 Q4 Sales Forecast, which it used for customer demand inputs "consistently throughout all parts of SCE's GRC testimony except for the Load Growth area," overlaps closely with the 2022 IEPR forecast and

²⁶¹ Forecasts provided in Ex. SCE-13, Vol. 7, p. B28 and CEC actual data in Ex. TURN-303. These forecasts include buses and are thus slightly higher than those originally presented in Ex. SCE-02, Vol. 7, and Ex. TURN-07.

²⁶² Ex. TURN-07E, p. 11.

²⁶³ Id., pp. 12-13.

assumes a significantly smaller EV population than the TEGR.²⁶⁴



Figure 4: Comparing Medium- and Heavy-Duty EV Forecasts in SCE Territory²⁶⁵

SCE has also suggested that incremental EV adoption is the critical factor for load growth planning, unlike cumulative adoption, and that SCE's incremental adoption forecast is lower than the latest IEPR updates in future years.²⁶⁶ First, however, cumulative load and incremental load are both relevant when assessing the need for capacity upgrades. In evidentiary hearings, SCE witness Mr. Esguerra acknowledged that "when we forecast we start off with a starting point year,"²⁶⁷ and when asked if that starting point affects planning for later years, stated that "it has some effect." ²⁶⁸ Second, incremental vehicle adoption is still misaligned between the TEGR and

²⁶⁴ Id., p. 12.

²⁶⁵ Excluding buses and applying SCE's assumed vehicle share for consistency. Forecasts are provided in Ex. TURN-07-Atch1 (SCE Response to TURN-SCE-006, Q1 and Q4) and CEC data in Ex. TURN-303.

²⁶⁶ Ex. SCE-13, Vol. 7, pp. 40-41.

²⁶⁷ Hearing Transcript, Volume 6, May 8, 2024, page 597, lines 5-10 (Esguerra/ SCE).

²⁶⁸ Hearing Transcript, Volume 6, May 8, 2024, page 601, line 9 (Esguerra/ SCE).

recent IEPR updates. The CEC vehicle population dashboard shows an incremental increase of 402,988 light-duty electric vehicles statewide and 1,395 MDHD electric vehicles statewide between end 2022 and end 2023.²⁶⁹ The TEGR assumes an incremental increase of 467,129 light-duty and 8,834 MDHD EVs in 2023.²⁷⁰ Thus, the TEGR presents a 533% higher incremental vehicle forecast for MDHD EVs, and a 16% higher incremental forecast for light-duty EVs in 2023 compared to the actual statewide population changes reported by the CEC.

11.1.1.2 Other Planning Assumptions in the TEGR Are Inaccurate

Besides too-high forecasts of electric vehicle population growth, SCE adopted other modelling assumptions that inflate the load growth from electrification anticipated by the TEGR compared to the 2022 IEPR.²⁷¹ These inputs include per-vehicle usage, time of use (load shapes), and the impacts of other demand side resources and load-modifying resources. The TEGR assumed significantly higher average per-vehicle consumption for light-duty and MDHD EVs than the 2022 IEPR and even SCE's own internal 2022 Q4 sales forecast.²⁷² SCE countered that this is due to different assumptions about the share of battery electric versus plug-in hybrid light-duty EVs, and medium- versus heavy-duty EVs, in the TEGR compared to the 2022 IEPR.²⁷³ However, forecasting an unreasonably large share of high usage vehicles is an assumption that inflates load expectations and further diverges SCE's expectations from reality. The TEGR assumes that just 2% of MDHD EV are medium-duty in 2023, rising to 32% in 2030;

²⁶⁹ Ex. TURN-303.

²⁷⁰ Ex. SCE-13, Vol. 7, p. B28.

 ²⁷¹ The 2023 IEPR public demand files were not released in time to incorporate into intervenor testimony.
 ²⁷² Ex. TURN-07, pp. 14-15.

²⁷³ Ex. SCE-13, Vol. 7, pp. 48-49.

the 2022 IEPR assumes that 81% of MDHD EV are medium-duty in 2023, decreasing to 60% in 2030.²⁷⁴

The TEGR also relies on load shapes for light-duty and MDHD electric vehicles that differ from the 2022 IEPR.²⁷⁵ These include less mid-day charging and higher nighttime peaks beginning at 9pm for MDHD vehicles, and moderately more on-peak charging for light-duty EVs.²⁷⁶ SCE emphasizes that peak coincidence, not simply load shapes, is what determines the need for local capacity upgrades. But as Cal Advocates explains: "This clustering of MDHD EV's which all begin charging at 9pm results in large, local power spikes which can skew forecasting models into calculating far more infrastructure overloads than a more distributed charging profile,"²⁷⁷ and as NRDC notes, "if enough MDHD trucks accumulate on the same circuit, then MDHD charging itself may become the source of the peak demand on that circuit."²⁷⁸ Data from SCE's medium- and heavy-duty charging sites to date evidence smoother aggregated load profiles and softer peaks than those assumed by the TEGR.²⁷⁹

SCE also neglects to consider the full load-mitigating impacts of other resources. The TEGR assumes the "mid" CEC 2020 IEPR demand scenarios for resources such as solar PV, energy storage, demand response, and time-of-use, with some upward adjustments, while using the "high" scenarios for electrification demand.²⁸⁰ However, California has seen increasingly

²⁷⁴ Ex. TURN-302 (SCE Response to TURN-SCE-110, Q9).

²⁷⁵ Ex. TURN-07E, pp. 16-18.

²⁷⁶ Ex. CA-08, p. 24.

²⁷⁷ Ex. CA-08, p. 25.

²⁷⁸ Ex. NRDC-01E, p. 20.

²⁷⁹ Ex. TURN-302 (SCE Response to TURN-SCE-110, Q2).

²⁸⁰ Ex. SCE-13, Vol. 7, pp. 55-57.

high investment in solar and storage, slower than expected population growth, and high electricity rates depressing demand, all of which exert a downward impact on the 2023 IEPR load forecast in contrast with SCE's TEGR assumptions.²⁸¹ Further, SCE does not anticipate any future load mitigation impacts from its ongoing investments in Grid Management Systems (GMS) and emerging technologies such as vehicle to grid (V2G) integration.²⁸²

SCE spent \$210.411 million over 2018-2022 on GMS, and forecasts another \$258.691 million over 2023-2028, but it has not begun to quantify the potential benefits of this program for mitigating overloads, even though GMS is designed to improve grid performance in a high DER context. The Commission should ensure that programs designed to reduce system strain and avoid other investments demonstrate those capabilities. While SCE explains that it does not consider GMS because the program is set to begin in 2027, outside "the time horizon of our analysis,"²⁸³ SCE stresses throughout its application that the bulk of its TEGR request is based on forecast grid needs in 2027 and 2028.²⁸⁴ Similarly, SCE explains that is has not assessed vehicle grid integration as part of its planning process, because the technology is too nascent.²⁸⁵ Yet this technology is sure to become relevant in the later years of SCE's planning, and Senate Bill 676 and D.20-12-029 require consideration of electric vehicle grid integration when dealing with transportation electrification related applications.²⁸⁶ The utility cannot have it both ways

²⁸¹ Ex. TURN-07E, pp. 19-20; Ex. NRDC-01E, p. 13.

²⁸² Ex. TURN-07E, p. 19.

²⁸³ Ex. SCE-13, Vol. 7, p. 58.

²⁸⁴ Ex. SCE-13, Vol. 7, p. 73.

²⁸⁵ Ex. SCE-13, Vol. 7, p. 57.

²⁸⁶ Ex. TURN-07E, p. 20.

and argue that it should be granted funding for investments based on uncertain future load, without consideration of emergent load-mitigating technologies.

11.1.1.3 Cal Advocates' Testimony Demonstrates SCE's Base Forecast is Also Inflated

SCE's Base Load Growth Forecast refers to the load growth that SCE estimates will occur within its service territory separate from the supplemental electric vehicle load growth that SCE forecasts in the TEGR.²⁸⁷ Exhibit Cal Advocates-07 identifies many instances where the Base or Baseline forecast for the Distribution Substation Plan and Transmission Substation Plan are inflated, due to its reliance on the 2020 IEPR.²⁸⁸ TURN's testimony on new customer service connections (Exhibit TURN-08) also identified that SCE overestimates customer growth, which further increases Base load growth expectations.²⁸⁹ Accordingly, TURN's proposal to authorize the Base load growth forecast, minus the project specific reductions discussed in Section 11.1.5 below, could potentially be too high. This should provide SCE with sufficient flexibility to cover reasonably expected emergent load growth for electric vehicles, even if its Base request does not include transportation forecasts.

11.1.2 The Commission Should Not Authorize Costs Based on SCE's TEGR Load Growth Forecast Due to Misalignment with IEPR

D.18-02-004 established that "the most recent IEPR system-level forecast is the most appropriate source for DER growth scenarios."²⁹⁰ TURN agrees with this concept as the most

²⁸⁷ Ex. CA-07, p. 3.

²⁸⁸ Id. pp. 11-18.

²⁸⁹ Ex. TURN-08E, p. 1.

²⁹⁰ D.18-02-004, p. 1

up to date forecast is most likely to be accurate. While SCE has repeatedly claimed that the 2020 IEPR was the most recently available load forecast that it could incorporate in time for its system planning process, SCE used more updated demand inputs throughout all other parts of SCE's GRC testimony.²⁹¹ The 2020 CARB Mobile Source Strategy, which SCE used as a vehicle population growth forecast, is an aspirational vision that was not intended for system planning and has diverged from reality each year since its inception.²⁹² According, TURN originally proposed that SCE revise its load growth forecast using inputs from, or aligned with, the 2022 IPER.

However, SCE's rebuttal testimony makes it clear SCE is unwilling and seemingly unable to, revise its forecast using the most recent data despite the plethora of evidence presented in this proceeding that shows key elements of SCE's TEGR forecast are significantly inflated and cannot be relied upon as a reasonable forecast. As TURN's Expert, Ms. Ashford testified, "...SCE has not demonstrated the reasonableness of costs above the Base" forecast.²⁹³ Just because SCE's planning process involves engineering and local knowledge does not provide cover writ large for any and all requests.²⁹⁴ Intervenors do not have the ability to repeat SCE's full planning process, nor have they been given all the requisite information to make recommendations in terms of the IEPR; "SCE did not provide intervenors with a complete proposal based on the 2020 IEPR, as its Base request removes transportation-related costs."²⁹⁵

²⁹¹ Ex. TURN-07E, pp. 11-12, referencing SCE Response to DR TURN-SCE-092, Q2.

²⁹² Ex. TURN-07E, pp. 7-8

²⁹³ Hearing Transcript, Volume 16, May 22, 2024, page 1529.

²⁹⁴ Hearing Transcript, Volume 16, May 22, 2024, page 1527.

²⁹⁵ Hearing Transcript, Volume 16, May 22, 2024, p.1528 (Ashford/ TURN).

As shown above in Sections 11.1.1.1-2 above, the TEGR forecast is not aligned with the most up-to-date data, including the CEC's 2022 and 2023 IEPR in regard to EV forecasts, especially for MD-HD vehicles, peak load, and other key drivers of load growth. Further, the recently released 2023 IEPR shows even lower peak load, slower customer growth and higher solar adoption to offset load growth needs.²⁹⁶ All of the flaws with SCE's TEGR forecast demonstrate that SCE has not met its burden to affirmatively demonstrate this aspect of its load growth forecast is just and reasonable. This evidentiary burden is entirely the utility's; other parties do not have the burden of proving the unreasonableness of the utility's forecasts or requests, though TURN and Cal Advocates have demonstrated the numerous flaws in the TEGR methodology.²⁹⁷ Accordingly, the Commission should reject the TEGR and remove the costs from SCE's load growth capital expenditures forecast, a reduction of \$1,031 million.

The Commission should not allow itself to be swayed by fear mongering that a failure to grant SCE's full load growth request is necessary to enable the State's energization initiatives. While transportation and building electrification initiatives will increase load, the impact on capital spending has not been substantiated by SCE's TEGR. Further, the Commission does not need to worry that removing the costs associated with SCE's TEGR forecast will leave SCE without sufficient capital to support energization requests. While TURN's proposed forecast is reasonable and should be sufficient to support energization and load growth in SCE's service territory, SB 410 provides the Investor Owned Utilities (IOUs) an additional opportunity to request incremental funding for energization projects in the unlikely event SCE's Base load growth forecast is exhausted prematurely.

²⁹⁶ Ex. TURN-07E, p. 21.

²⁹⁷ Ex. TURN-07E, pp. 6-23; Ex. CA-08, pp. 18-33.

11.1.2.1 SB 410 Provides an Alternative Funding Mechanism in the Unlikely Event SCE's GRC Authorized Forecast is Insufficient

Pursuant to Senate Bill 410 (Becker),²⁹⁸ which was signed into law in 2023, an IOU may request a ratemaking mechanism that will "track costs for energization projects placed in service after January 1, 2024, that exceed the costs included in the electrical corporation's annual authorized revenue requirement for energization, as established in the ... [IOUs GRC] or any other proceeding."²⁹⁹ The interim rate recovery authorized under SB 410 is limited to "energization projects" that enable customers to connect to the electrical distribution grid.³⁰⁰ The bill defines energization as:

connecting customers to the electrical distribution grid and establishing adequate electrical distribution capacity or upgrading electrical distribution or transmission capacity to provide electrical service for a new customer, or to provide upgraded electrical service to an existing customer. The determination of adequate electrical distribution capacity includes consideration of future load. "Energization" and "energize" do not include activities related to connecting electrical supply resources.³⁰¹

Accordingly, many of the same project costs included in SCE's load growth forecast BPE would be considered energization projects potentially subject to interim rate recovery under SB 410, if the additional spending incremental to the GRC authorization is necessary.

In rebuttal, SCE opposes Tesla's recommendation to pursue an SB 410 ratemaking mechanism on two counts. The utility argues that, first, being directed to file a request is premature because the funding avenue is intended to cover shortfalls in GRC funding, and

²⁹⁸ Senate Bill (SB) 410 (Becker), Stats. 2023, Ch. 394.

²⁹⁹ Cal. Pub. Util. Code §937(a)(1).

³⁰⁰ Cal. Pub. Util. Code §937(b)(1).

³⁰¹ Cal. Pub. Util. Code §931(b).

second, that the relevant section of the Public Utilities Code is set for repeal in 2027.³⁰² TURN agrees it would be premature for SCE to file an application to utilize the SB 410 ratemaking mechanism and is not recommending SCE do so. SCE's Base request is adequate to cover likely load growth expenditures, TURN references SB 410 to establish that SCE has an avenue to pursue additional funding if conditions change significantly between now and the end of 2026.

TURN and Cal Advocates' analyses demonstrate that SCE's Base load growth forecast should be sufficient to fund necessary and reasonably forecast load growth projects during the GRC period. However, the SB 410 ratemaking mechanism for incremental energization costs is an alternative recovery mechanism SCE can exercise if necessary. If SCE needs to utilize the SB

³⁰² Ex. SCE-13, Vol. 7, pp. 102-104.

³⁰³ D.24-07-008, Decision Authorizing a Ratemaking Mechanism for Energization Projects Pursuant to Senate Bill 410 (A.21-06-021), p. 2.

³⁰⁴ D.24-07-008, Decision Authorizing a Ratemaking Mechanism for Energization Projects Pursuant to Senate Bill 410 (A.21-06-021), p. 2.

410 mechanism, then the Commission should adopt a similar recovery mechanism as it recently authorized for PG&E, a memorandum account,³⁰⁵ subject to an annual cost cap.³⁰⁶

11.1.2.2 The Commission Should Reject NRDC's Two-Way Balancing Account Proposal

In rebuttal testimony, NRDC revised its original recommendation that the Commission authorize SCE to "establish a one-way balancing account for funds authorized to support energization of load growth related infrastructure"³⁰⁷ to recommending authorization of a two-way balancing account.³⁰⁸ NRDC states that it is revising its position "to ensure that SCE has enough funds to make necessary upgrades in light of uncertainty that cuts in both directions but with the greater risk stemming from insufficient investment."³⁰⁹ TURN agrees there is a lot of uncertainty associated with SCE's load growth forecast, however it is unreasonable and unjust to ratepayers to grant the level of flexibility provided by a two-way balancing account.

To TURN's knowledge, a two-way balancing account for such a significant forecast is unprecedented. It is also unnecessary in light of the interim funding mechanism provided under SB 410. In its first Decision implementing SB 410, the commission deemed a memorandum account as appropriate, despite PG&E's request for a balancing account, as addressed above in Section 11.1.2.1. The Commission should not depart for the standard GRC ratemaking

³⁰⁵ D.24-07-008, Decision Authorizing a Ratemaking Mechanism for Energization Projects Pursuant to Senate Bill 410 (A.21-06-021), pp. 40-41.

³⁰⁶ D.24-07-008, Decision Authorizing a Ratemaking Mechanism for Energization Projects Pursuant to Senate Bill 410 (A.21-06-021), pp. 50-52 & p. 87-88 FOF #25.

³⁰⁷ Ex. NRDC-01, p. 26.

³⁰⁸ Ex. NRDC-03, p. 10.

³⁰⁹ Id.

mechanism and should authorize a reasonable forecast for SCE for load growth capital expenditures.

11.1.3 The Rate Increases Resulting from SCE's Inflated Load Growth Forecast Will Disproportionately Burden Disadvantaged Customers and Discourage Electrification.

As addressed in Section 4 above, the entirety of SCE's GRC request, including the load growth capital expenditures forecast, represents a nearly 40% increase over March 2023 rates that could raise the typical Hot Climate Zone customer's bill almost \$60 each month by January 2028.³¹⁰ These rate impacts raise significant concerns for equity, particularly when evaluating customer burden versus benefit, and for incentivizing electric vehicle adoption to reach California's climate targets. As addressed in Exhibits TURN-03 and TURN-07, low-income households who pay a greater share of their income on electricity bills are disproportionately burdened by rate increases.³¹¹ Further, as TURN witness Ashford notes, "A significant portion of SCE's load growth investments are driven by anticipated commercial vehicle fleets, which, besides potentially positive local-level air quality impacts, are less likely to benefit residential customers in the near-term."³¹² Ms. Ashford's analysis of SCE's proposals further indicates that only a small number of the fleets SCE's load growth capital expenditures forecast is intended to enable electrification of fleets that provide public benefits, while a majority serve commercial purposes.³¹³ Accordingly, a significant portion of the infrastructure investments proposed in

³¹¹ Ex. TURN-07E, p. 24, referencing Borenstein et al., "Paying for Electricity in California: How Residential Rate Design Impacts Equity and Electrification." Next 10 and the Energy Institute at Haas, University of California, September 2021, p. 5. Available at https://www.next10.org/sites/default/files/2022-09/Next10-paying-for-electricity-final-comp.pdf
 ³¹² Id. at p. 25.

³¹⁰ See Ex. TURN-02 (Dowdell) for comparison of rate impacts by climate zone.

³¹³ Id.

SCE's load growth testimony will support EV charging by private corporations, but will be paid for by all ratepayers, including residential customers, many of whom are already struggling to afford their electric bills. Section 4.2 above discusses the equity implications of this.

Rate increases also make it more costly to operate electric vehicles and electrify buildings, counterproductively discouraging new customer vehicle adoption and building electrification. As noted in Exhibit TURN-07, "a 2021 study of electric vehicle adoption in California found that an electricity price increase of one cent per kilowatt hour was associated with a two percent decrease in vehicle purchases across different utility territories."³¹⁴ In regard to building electrification, Exhibit TURN-304 identifies high electricity prices and low natural gas prices as two of the key reasons why electric heat pump sales are slumping. Exhibit TURN-304 states,

... residential electricity rates in California are already about twice as high as the national average and still increasing. These high electricity rates make heat pumps and other forms of building electrification much more expensive.³¹⁵

While the Exhibit TURN-304 also identifies high interest rates as a reason heat pump sales are slumping, that factor cannot be addressed by the Commission in this proceeding. The Commission's decision in this proceeding will have a direct impact on SCE's electric rates which will impact how they compare to most SCE customers other residential fuel choice, natural gas.

In responding to TURN's concerns about the rate impacts of the TEGR, SCE discuss the potential for downward pressure on rates from transportation and building electrification. TURN

³¹⁴ Id. at p. 26, referencing Bushnell, J., Muehlegger, E., & Rapson, D. (2021). Do Electricity Prices Affect Electric Vehicle Adoption? UC Office of the President: University of California Institute of Transportation Studies. Available at http://dx.doi.org/10.7922/G29S1PB5 Retrieved from https://escholarship.org/uc/item/5f80503b.

³¹⁵ Ex. TURN-304, p. 3.

acknowledges the potential for downward pressure on rates from transportation and other beneficial electrification, but notes "(H)owever, ratepayers will not receive these benefits if the system investments to serve newly incurred load exceed this downward pressure."³¹⁶ NRDC also addresses the potential for downward pressure on rates and points to Cal Advocates' Distribution Grid Electrification Model (DGEM) study showing increased electrical load in SCE's service territory could decrease residential rates by \$0.02 per kWh by 2035.³¹⁷ NRDC neglects to mention the many caveats in Cal Advocate's DGEM study regarding downward pressure on rates. The DGEM study states:³¹⁸

Achieving this downward pressure on residential electricity rates is contingent upon five key model assumptions. Downward pressure on residential rates might not be achieved if:

6. EVs mostly charge in the evening, near peak hours (i.e., 6 p.m. to 10 p.m.), which would drive a higher peak load and, therefore, higher upgrade costs.

7. Electric rates rise to cover additional electrification programs, such as deploying EV chargers.

8. New feeders and substations are more expensive than the DGEM estimates.

9. Expected load growth due to electrification does not occur.

10. Utilities build more infrastructure than is needed or build infrastructure in the wrong locations because upgrade costs will be higher. (Citations omitted)

SCE's TEGR proposal raises concerns under the final category, creating significant risks of

building potentially unnecessary infrastructure, or infrastructure in the wrong locations.

³¹⁶ Id. at p. 26.

³¹⁷ Ex. NRDC-01, pp. 23-24.

³¹⁸ Ex. NRDC-01, pp. 24, FN 39, referencing Public Advocates Office, *Distribution Grid Electrification Model – Study and Report* (2023). See pp. 43-44 for a discussion of the relevant caveats, available at https://www.publicadvocates.cpuc.ca.gov/-/media/cal-advocates-website/files/press-room/reports-and-analyses/230824-public-advocates-distribution-grid-electrification-model-study-and-report.pdf.

The DGEM study goes on to state, "(G)ood forecasting and planning are key parts of achieving this downward pressure on rates. Utility forecasts must be accurate and not lead to infrastructure over-building. If overbuilding occurs, electrification could cause *upward pressure* on rates."³¹⁹ This is exactly what TURN is concerned about with SCE's TEGR. In Exhibit TURN-07 and above in Sections 11.1.1, TURN has demonstrated that SCE's TEGR forecast is based on flawed assumptions, is likely to be inaccurate, and lead to distribution grid infrastructure over-building. The potential for downward pressure on rates in the 2030s could be completely wiped out by infrastructure over-building or building in the wrong locations authorized in this GRC. Further, even if the downward pressure does materialize, it may be too late for low and medium-income residential ratepayers who are struggling to afford SCE's rates now. The impacts of the rate increases proposed to support the TEGR however will be felt by SCE's ratepayers as soon as the authorized rates are implemented.

11.1.4 Private Market Parties Claims of Energization Delays are Not Due to SCE's Lack of Capital

The Commission should not be swayed by claims from TeraWatt and the Joint Truck OEMs that the entire TEGR forecast cost (or more, in the case of TeraWatt) should be approved because of energization delays³²⁰ they have experienced, or uncertainties about SCE's ability to provide a firm energization dates.³²¹ TURN does not dispute that energization timeliness is important, but SCE has failed to establish that more funding will eliminate energization delays.

³¹⁹ DGEM Study & Report, p. 44.

³²⁰ Ex. Terawatt-01E, pp. 19-20.

³²¹ Ex. Joint Truck OEMs-01, p. 8.

There is a disconnect between party complaints about energization delays and SCE's request. Distribution capacity upgrades take time to build, and no amount of funding is going to address the external issues that have delayed some currently pending energizations projects. In evidentiary hearings, SCE's load growth witness Mr. Esguerra was asked three separate times about the causes of its underspending in 2023 or energization delays; each time he cited pandemic or supply chain concerns, rather than insufficient capital funds.

Specifically regarding underspending in 2023, Mr. Esguerra stated "(A)nd as I mentioned during that time frame, we had the impacts of the pandemic, we had changes in some of our permitting policies, and we had supply chain shortages, which impacted our ability to execute on those load growth projects."³²² Regarding complaints about energization delays, SCE explained there are "a number of factors" that led to complaints of energization delays, including "supply chain delays" and "pandemic restrictions" as "some of the primary factors".³²³

Lastly, when Mr. Esguerra was asked if reallocating any of SCE's underspending for TSP projects in 2023 to those delayed energization projects some parties complained about would have resolved the energization delays, Mr. Esguerra answered, "No, largely because a lot of those projects are multi-year projects. They would have had to have started much earlier."³²⁴ Accordingly, while party complaints and Commission concerns about energization delays are valid, authorizing unnecessarily excessive capital spending is not the solution and cannot address the many factors influencing energization projects that are outside of the utilities', and even the Commission's control. The Commission should not let fear guide its decision here. TURN's

³²² Hearing Transcript, Volume 6, May 8, 2024, p. 589: 6-10 (Esguerra/ SCE).

³²³ Hearing Transcript, Volume 6, May 8, 2024, p. 594: 12-15 10-12 (Esguerra/ SCE).

³²⁴ Hearing Transcript, Volume 6, May 8, 2024, p. 623:10-12 (Esguerra/ SCE).

load growth capital expenditures proposal provides sufficient funding for thoughtful and wellplanned load growth projects necessary to support realistic increases in load growth.

11.1.5 TURN's Project Specific Forecast Reductions are Reasonable and Supported by the Record

Besides concerns with SCE's load forecasting and proposed spending based on the TEGR, TURN has identified multiple project-specific costs within the "Base" forecast and other project areas that SCE has not shown to be reasonable.³²⁵ Under the Distribution Substation Plan (DSP), Transmission Substation Plan (TSP), and other load growth spending categories, SCE's forecasts include duplicative or inaccurate cost estimates, lack details on scope of work, or fail to demonstrate an appropriate level of cost-effectiveness for ratepayer funding. As a result, TURN recommends that the Commission authorize a corresponding reduction to SCE's Base request in the test year, in addition to removal of all costs resulting from the TEGR supplemental forecast. Based on TURN's analysis, this amounts to a reduction of \$59.818 million over 2023-2025, including \$38.184 million in the test year, about 9% of SCE's 2025 request. TURN further finds this test year reduction to be appropriate given the similar (8-10%) project-specific reductions identified in each of the attrition years relative to SCE's revised forecast.

TURN's recommendation is a relatively modest reduction to SCE's non-TEGR requests, given the numerous issues identified in TURN's analysis and in contrast to the significantly higher Base reduction proposed by Cal Advocates.³²⁶ Issues with specific project planning, in combination with SCE's use of outdated forecasting inputs from the 2020 IEPR (as discussed

127

³²⁵ For discussion, see Ex. TURN-07E, Pages 27-36.

³²⁶ Cal Advocates recommends a \$82.703 million (54%) reduction to DSP and TSP base costs in the 2025 test year, based on an adjustment to reconcile SCE's base request with the 2022 IEPR system load forecast (Ex. SCE-13, Vol. 7, Table II-11, p. 63).

above),³²⁷ have inflationary impacts on SCE's cost forecasts. Thus, accepting these minimal proposed cuts to SCE's Base, the Commission should find it even more reasonable to deny all of SCE's request based on the TEGR.

11.1.5.1 The DSP Base Forecast Should be Revised

SCE's Distribution Substation Plan (DSP) includes Distribution Circuit Upgrades, DERrelated upgrades, New Circuits, and Substations, and comprises about half of SCE's load growth capital expenditures forecast.³²⁸ SCE's rebuttal position identifies capital expenditures of \$231.375 million for DSP in the test year, including \$77.659 million for TEGR and \$153.717 million for Base projects.³²⁹ SCE spent 35.5% (\$59 million) less than it was authorized for DSP projects in the 2021 test year.³³⁰ In evidentiary hearings, SCE highlighted that it recorded more than it forecast in 2023 for DSP projects;³³¹ but excluding carryover costs from projects in prior years that were not included in SCE's 2025 test year GRC application, SCE actually spent 8% (\$2.939 million) less than it forecast for DSP projects.³³²

TURN has recommended two reductions to SCE's DSP request. First, TURN removes costs associated with the DSP Substation Project PIN 8043 (increasing capacity of the Garnet 115/33kV Substation) from its recommendation, because SCE was previously authorized

³²⁷ SCE prepared its base forecast by disaggregating the CEC's 2020 IEPR forecast, in combination with local engineering knowledge (Ex. SCE-13, Vol. 7, Page 64).

³²⁸ Ex. SCE-02, Vol. 7, p. 69.

³²⁹ SCE's recommendations are summarized in Ex. SCE-13, Vol. 7, Table II-6, p. 7, and Table II-7, p. 9. ³³⁰ Ex. SCE-02, Vol. 7, p. 14.

³³¹ Hearing Transcript, Volume 6, May 8, 2024, p. 622 (Esguerra/ SCE).

³³² Not including emergent projects (\$214,863) and carryover costs for projects from prior years (\$10.407 million). Ex. TURN-300, SCE Response to Data Request TURN-SCE-104.

funding for this project in the 2021 GRC to meet an in-service date of November 2021.³³³ SCE forecast \$9.830 million for the project over 2023-2024,³³⁴ and recorded spending \$9.647 million in 2023.³³⁵ SCE contends that the project's delay was due to unforeseen electrical and civil work, that it had jurisdiction to reprioritize those funds elsewhere, and that the project should still be entered into the permanent rate base once in service.³³⁶ While this project may be eligible for future recovery, there must be a limitation on duplicative authorizations for the same capital expenditures in subsequent rate cases. Customers should not be expected to repeatedly shoulder the same costs while SCE exhibits a trend of over-forecasting and underspending on DSP projects, reprioritizing funds to other investment areas, and later suggesting that load growth capital forecasts have been insufficient to complete distribution system upgrades to fulfill energization requests in a timely manner.³³⁷

Second, TURN opposes SCE's request for DSP DER capital expenditures because of the way in which SCE has estimated costs for the procurement of energy storage devices.³³⁸ SCE requests \$10 million in 2025, and \$119.633 million over the 2023-2028 period for this project area.³³⁹ While TURN is generally supportive of temporary and relocatable alternatives to traditional grid investments to meet new load, utilities should make these investments at least cost feasible to ratepayers. Based on cost estimates provided by SCE, renting rather than

³³³ Ex. TURN-07E, p. 28.

³³⁴ Ex. SCE-13, Vol. 7, p. 74.

³³⁵ Ex. TURN-300, SCE Response to Data Request TURN-SCE-104.

³³⁶ Ex. SCE-13, Vol. 7, p. 75.

³³⁷ Hearing Transcript, Volume 6, May 8, 2024, p. 588: 23-34 (Esguerra/ SCE).

³³⁸ Ex. TURN-07E, pp. 29-30.

³³⁹ Ex. SCE-02, Vol. 7, p. 48.
purchasing would provide a savings to ratepayers if energy storage devices are in use for fewer than four months each year, or continuously over five or fewer years.³⁴⁰

The utility cannot argue it both ways, as it seems to do in rebuttal.³⁴¹ If these devices are short-term, "stop-gap"³⁴² measures that do not defer traditional wires solutions, then they are strong candidates for leasing or renting. If these devices are long-lasting investments that should provide value to ratepayers for the entirety of their service lives, then SCE should demonstrate that value. As highlighted by Cal Advocates, SCE has yet to identify locations for future deployment of these mobile units.³⁴³ Lacking these details, at a minimum, the utility should be able to quantify the costs to "transport, interconnect, monitor, maintain, and disconnect" devices to compare the full ownership expenses with alternative operating models.³⁴⁴ The utility's potential for capital returns and a 30% Investment Tax Credit benefit³⁴⁵ should not motivate procurement over more affordable options, at ratepayers' expense.

Given these uncertainties, TURN removes the DSP DER cost forecast from its cost recommendations. The Commission should direct SCE to perform a cost-benefit analysis of ratepayer spending for distributed energy resources, including energy storage devices and mobile substations when data becomes available. A memorandum account with specific parameters regarding the types of costs that can be recorded in it, may be a better mechanism to record these costs, given the urgency of these investments to meet demand in the short-term and mitigate

³⁴⁰ Ex. TURN-07E, p. 30.

³⁴¹ Ex. SCE-13, Vol. 7, pp. 92-93.

³⁴² Ex. TURN-Atch1 (SCE Response to DR TURN-SCE-050, Q10.a-b).

³⁴³ Ex. CA-07, p. 13.

³⁴⁴ Ex. TURN-Atch1 (SCE Response to DR TURN-SCE-050, Q10.a-b).

³⁴⁵ Ex. TURN-07E, p. 27, as first mentioned in SCE Response to DR TURN-SCE-079, Question 1b.

potentially misplaced traditional infrastructure investments, if costs are shown to be reasonable for ratepayers.

11.1.5.2 The DER-Driven Grid Reinforcement Memorandum Account Should be Closed

Prior to filing its GRC application, SCE had not recorded any costs to the DER-driven grid reinforcement (DER-DGRPMA) memorandum account, which it was authorized to establish in the 2021 GRC decision.³⁴⁶ SCE has not forecast any related capital expenditures for this GRC period and has also failed to demonstrate any progress towards performing a grid reinforcement needs analysis to identify potential projects as part of its distribution planning process.³⁴⁷ Accordingly, TURN recommended that the Commission close the DER-DGRPMA memorandum account.³⁴⁸ The utility has since recorded \$0.197 million to the memorandum account, which SCE claims is "demonstrating that SCE is doing work in this area." ³⁴⁹ However, utilization of the memorandum account alone does not demonstrate that SCE has made progress in developing specific software and analytical processes. Therefore, TURN continues to recommend eliminating the memorandum account unless and until SCE has made progress in its tool development. SCE is requesting funding for the same types of distribution projects (circuit upgrades, new circuits etc.,) that it considers appropriate for this account³⁵⁰ through its Distribution Substation Plan and can also make use of DER-specific funding avenues such as an account pursuant to SB 410 if the need arises. Should the Commission decide to retain the memo

³⁴⁶ D.21-08-036, p. 653.

³⁴⁷ Ex. SCE-2 Vol. 7, p. 66.

³⁴⁸ Ex. TURN-07, p. 31.

³⁴⁹ Ex. SCE-18, Vol. 1, pp. 25-26.

³⁵⁰ Ex. SCE-2 Vol. 7, pp. 67-69.

account, it should direct that recorded costs may only be found reasonable for rate recovery if they make use of a DER-driven needs analysis that is conducted as part of SCE's distribution planning process.

11.1.5.3 The Transmission Substation Plan (TSP) Base Forecast Should be Revised

SCE's Transmission Substation Plan (TSP) includes the Subtransmission Lines, A-Bank, and Subtransmission VAR Plans, and comprises about a third of SCE's load growth capital expenditures forecast.³⁵¹ SCE's rebuttal position identifies capital expenditures of \$143.991 million for TSP in the test year, including \$89.709 million for TEGR and \$54.282 million for Base projects.³⁵² Similar to the DSP, SCE has exhibited notable work setbacks and underspending in this project area in recent years. While the utility overspent on TSP projects in its 2021 test year, this was due to exceptional costs associated with sporting events; SCE shares that overall, TSP projects were delayed relative to expectations due to environmental review holdups and supply chain issues.³⁵³ In 2023, SCE spent \$49.6 million, or 85%, less than it forecast in 2023 for TSP projects (91% less than forecast excluding carryover costs for prior year projects).

TURN recommends a reduction of SCE's request due to the cancellation of a planned A-Bank substation, TSPABank35796 (Rector), of \$8.442 million in the test year, with forecast costs of \$40.262 million over the 2023 to 2028 period.³⁵⁴ SCE suggests that because there was

³⁵¹ Ex. SCE-02, Vol. 7, p. 69.

³⁵² SCE's recommendations are summarized in Ex. SCE-13, Vol. 7, Table II-6, p. 7, and Table II-7, p. 9.
³⁵³ Ex. SCE-02, Vol. 7, p. 14.

³⁵⁴ Ex. TURN-07, p. 32.

an anticipated need for TSPABank35796 (Rector) at the point in time when its forecast was developed, and because smaller upgrade projects may be needed to take the project's place, funding is reasonable even though the project has been found unnecessary.³⁵⁵ TURN objects to SCE's arguments on multiple grounds.

As an intervenor, TURN makes recommendations based on the best information available over the course of this proceeding and does not have complete insight into which resources SCE may have had access to at a given time. It is simply unreasonable to authorize substantial funding for a large capital project that will not be carried out, particularly in an area of capital expenditures where SCE has historically underspent relative to its forecast due to external and internal obstacles. This is particularly true given that seven of the twelve proposed A-Bank projects in SCE's request are awaiting licensing and marked as "under review".³⁵⁶ Considering the size of this funding request, it is further illogical to authorize funding on the assumption that un-scoped and unplanned smaller projects will arise in its place, with similar overall costs. While imprecision may be an inevitable aspect of forecast based ratemaking, it should not be considered a preferred ideal.

SCE has not shown that the costs of its new proposed climate-driven circuit ties program are reasonable, and the full request for this initiative should be denied.³⁵⁷ The program includes \$19.714 in test year capital expenditures and a forecast of \$80.03 million from 2025 to 2028, which will support nine identified projects and four to eight projects that have not yet been scoped. TURN and Cal Advocates agree that SCE should not be authorized the \$6.3 million

³⁵⁵ Ex. SCE-13, Vol. 7, p. 79.

³⁵⁶ Including environmental review, per Ex. SCE-02 Vol. 7, p. 81E.

³⁵⁷ Ex. TURN-07, pp. 34-36.

(32%) of its request in the test year for these undefined projects.³⁵⁸ The wide range in project costs (\$1.9 to \$13.4 million), uncertain number of additional projects, and newness of this program area are all causes for concern. While SCE has since identified an additional two projects, their costs also vary significantly (\$0.701 million and \$5.258 million).³⁵⁹

Further, TURN found that for all proposed projects, SCE has failed to evidence good value for ratepayer money. SCE contends that Risk Spend Efficiency scores (RSEs) and costbenefit ratios are not required for all risk mitigation decisions, and that the CAVA proved the necessity of these investments.³⁶⁰ In the absence of a metric such as an RSE, the utility should still demonstrate the cost-worthiness of a given investment for recovery through rates. The information provided by SCE fails to meet this standard; project costs range from \$163 to \$11,445 per potentially impacted customer.³⁶¹ Furthermore, SCE's claim that "the quantitative analysis in CAVA"³⁶² proves the necessity of a climate ties program does not comport with a response to TURN that "SCE did not perform quantitative evaluation of impacts on customers from the alternatives to climate-driven circuit ties."³⁶³ While the CAVA identified future risks, this specific program's value cannot be determined without a thorough evaluation of alternatives such as vegetation management, fire wrapping poles, replacing pad mounted switches with submersible equivalents, and no mitigation, which includes the costs for emergency response

³⁵⁸ Ex. CA-7, p. 21 and TURN-07, p. 34.

³⁵⁹ Ex. SCE-13, Vol. 7, p. 97.

³⁶⁰ Ex. SCE-13, Vol. 7, pp. 100-102.

³⁶¹ Ex. TURN-07-Atch1 (SCE Response to DR TURN-SCE-050, Q21, "TURN-SCE-050 Q21a-c 240122_Nine_Identified_Projects.xlsx") and Ex. SCE-13, Vol. 7, p. 97.

³⁶² Ex. SCE-13, Vol. 7, p. 99.

³⁶³ Ex. TURN-07-Atch1 (SCE Response to DR TURN-SCE-006, Q18).

crews to re-energize customers in the aftermath of climate events.³⁶⁴ As Ms. Ashford testified: "risk reduction activities are potentially infinite, and ratepayer funds are not."³⁶⁵ TURN maintains its recommendation that SCE's request for this untested program be denied, and at a minimum, no costs authorized for the projects labelled "unplanned" in its application.

11.2 Transmission Projects

11.3 Engineering O&M

12. NEW SERVICE CONNECTIONS AND CUSTOMER REQUESTED SYSTEM MODIFICATIONS

12.1 New Service Connections

SCE's GRC forecast includes capital expenditures associated with new service connections work for each customer class or load type. SCE bases these capital forecasts on its forecast of new meter sets.³⁶⁶ SCE uses the terms "new meters" and "gross meters" interchangeably in its testimony, as does TURN here.³⁶⁷ To derive the capital expenditures forecast, SCE multiplied the gross meter set forecast for each customer class by the 2018-2022 five-year average recorded unit cost for that class.³⁶⁸

TURN recommends lower gross meter set forecasts for the residential, commercial, and agricultural customer classes, and therefore, lower capital forecasts, as explained in TURN's testimony and summarized below. As explained below, TURN also accepts SCE's unit cost

³⁶⁴ Ex. TURN-07, p. 35.

³⁶⁵ Ex. TURN-07, p. 35.

³⁶⁶ Ex. SCE-02V08, p. 5 (referring to the gross meter set forecasts presented in Ex. SCE-07V01, Section IV.B.2).

³⁶⁷ Ex. TURN-400 (SCE Response to TURN DR 124), p. 8.

³⁶⁸ Ex. SCE-02V08, p. 13 (residential capital expenditures); p. 20 (commercial capital expenditures); p. 25 (agricultural capital expenditures).

forecasts for each customer class, so the only difference between TURN's position and SCE's is the number of gross meter sets.

TURN withdraws some of its recommendations following consideration of SCE's rebuttal testimony. TURN highlights these changes here before addressing TURN's current recommendations.

First, TURN originally proposed to change the escalation rates used to forecast residential meter set unit costs. TURN assumed a 2.5% inflation rate starting in 2023, whereas SCE applied the same escalation rates used throughout its GRC testimony.³⁶⁹ TURN withdraws this recommendation. TURN now recommends that historical costs be escalated to test year 2025 dollars using standard escalation, consistent with SCE's position. Because TURN recommends that attrition year capital costs be determined pursuant to the post-test year mechanism adopted by the Commission in this GRC, escalation to post-test year dollars will flow from that mechanism.³⁷⁰

Second, TURN originally proposed the creation of a new one-way balancing account for new connections costs but withdraws that recommendation.³⁷¹

Finally, TURN withdraws the following procedural recommendations regarding SCE's showing in the next GRC: (1) the Commission should direct SCE to provide in its testimony (a) any alternative models and variables it considered in its forecast model selection process; and (b) a quantification of the historical accuracy of the forecast methodology it proposes; and (2)

³⁶⁹ Ex. TURN-08 (McGovern), pp. 20-21.

³⁷⁰ Ex. TURN-17 (Yap), p. 2 (recommending a standard capital attrition mechanism for all cost categories other than wildfire mitigation capital).

³⁷¹ Ex. TURN-08 (McGovern), pp. 3, 29.

prohibit SCE from using data that has been demonstrated to be repeatedly biased.³⁷² TURN no longer recommends that the Commission take these specific actions.

The following tables compare TURN's and SCE's gross meter set 2025-2028 forecasts and associated capital expenditures forecasts for Test Year 2025. TURN focuses on Test Year 2025 because no party has proposed budget based attrition for new connections capital expenditures in this GRC.

Residential - New Meter Connections					
	2025	2026	2027	2028	
SCE	33,421	36,084	36,768	36,154	
TURN	32,569	32,799	33,031	33,265	
TURN-SCE	(852)	(3,285)	(3,737)	(2,889)	
	Commercial -	New Meter	Connections		
	2025	2026	2027	2028	
SCE	3,918	3,918	3,918	3,918	
TURN	2,582	2,383	2,200	2,031	
TURN-SCE	(1,336)	(1,535)	(1,718)	(1,887)	
Agricultural - New Meter Connections					
	2025	2026	2027	2028	
SCE	184	184	184	184	
TURN	102	89	79	69	
TURN-SCE	(82)	(95)	(105)	(115)	

³⁷² Ex. TURN-08 (McGovern), pp. 3-4. TURN continues to recommend that the Commission require SCE to provide all raw data for its customers and meter forecast models in its workpapers supporting direct testimony, as explained below.

New Connections Capital Expenditures - Nominal (\$000)					
	Residential	Commercial	Agricultural		
Item	2025	2025	2025	Formula	
SCE Meter Sets	33,421	3,918	184	(a)	
TURN Meter Sets	32,569	2,582	102	(b)	
SCE Unit Cost	\$4.202	\$31.098	\$28.586	(c)	
SCE Total Cost**	\$141,863	\$134,697	\$5,354	(d)	
TURN Reduction	\$3,580	\$41,547	\$2,344	[(a-b)*c]	
TURN Total Cost	\$138,283	\$93,150	\$3,010	(d) - [(a-b)*c]	
** Ex. SCE-13V08, Tables II-3, II-4, II-5.					

12.1.1 The Commission Should Adopt TURN's Residential Gross Meter Set Forecast for the Test Year.

12.1.1.1 SCE's Residential Gross Meter Set Forecast is Flawed Because It Relies on an Overly Optimistic Housing Forecast and Underperforming Historical Regression Models.

There are two major components to SCE's residential meter forecast: (1) the historical regression model that projects the residential meter forecast into 2023-2028, and (2) the forecast housing data input into the extended model.³⁷³ The same is true for SCE's residential customer forecast.³⁷⁴ The main forecasted explanatory variable for both the residential new meter model and customer model is the average of January 2023 housing start forecast data provided by Moody's and S&P Global Market Intelligence (formerly IHS Markit and before that, Global Insight).³⁷⁵ According to SCE, "residential new meter installation activities are closely tied to activities in the residential construction sector, with lags of up to 12 months."³⁷⁶ SCE relied

³⁷³ Ex. TURN-08 (McGovern), p. 8.

³⁷⁴ Ex. TURN-08 (McGovern), p. 8.

³⁷⁵ Ex. SCE-02V08, p. 5; Ex. SCE-07V01, p. 99. SCE provides historical and forecast housing starts in Figure VI-4 on p. 91.

³⁷⁶ Ex. SCE-07V01, p. 98.

exclusively on Moody's housing start forecast in the 2021 GRC but used an average here to provide a more conservative approach.³⁷⁷ SCE explains that Moody's "predicts a more aggressive growth in the expansion of construction activities and the housing market," while S&P Global Market Intelligence "anticipates gradual to flat growth in the housing market."³⁷⁸ TURN demonstrates in testimony the weaknesses in both the regression model and the housing forecast relied on by SCE, summarized here.

SCE produces customer regression models for individual counties in its service territory, which are used to generate county-specific residential customer forecasts.³⁷⁹ The customer regressions are the first step in generating SCE's forecasts; they attempt to quantify a historical relationship between the monthly customer growth and various seasonal and economic factors including housing.³⁸⁰ SCE then totals the individual forecasts for a residential customer forecast.³⁸¹ SCE's individual residential customer regression models have varying levels of accuracy, between 58.5% and 89.4% explanatory power, and therefore the resulting aggregate residential customer regression model has mediocre explanatory power.³⁸² As TURN explains in its testimony, SCE's "regression results, which explain the relationship between housing starts and new residential customers, are often off, even if all the data is perfectly accurate."³⁸³ In addition, TURN pointed out that in the Santa Barbara/Ventura customer forecast model, the

³⁷⁷ Ex. SCE-02V08, p. 5.

³⁷⁸ Ex. SCE-02V08, p. 5.

³⁷⁹ Ex. TURN-08 (McGovern), p. 4.

³⁸⁰ Ex. TURN-08, (McGovern), p. 6.

³⁸¹ Ex. TURN-08 (McGovern), p. 4.

 ³⁸² Ex. TURN-08 (McGovern), p. 4; Ex. SCE-18V01, Appendix A, pp. A42-A43 (SCE-TURN-005, Q1).
 ³⁸³ Ex. TURN-08 (McGovern), p. 4.

housing starts are not even statistically significant, raising the question of whether housing starts are relevant to the forecast of residential customers in that region at all.³⁸⁴

Beyond the shortcomings in SCE's regression models, SCE's forecasts are heavily impacted by the housing start forecasts it relies on. As noted above, SCE uses a blend of the proprietary, confidential housing start forecasts provided by two third party vendors: Moody's and S&P Global Intelligence. These vendors' forecasts in other GRCs have been overly optimistic. SCE has relied on Moody's housing start forecasts in multiple GRCs.³⁸⁵ In the 2021 GRC, the Commission recognized that Moody's forecasts have repeatedly been high.³⁸⁶ In this GRC, TURN showed that Moody's housing start forecasts from the 2021 GRC were also much higher than actual housing starts from 2019 through 2022.³⁸⁷ SCE has not used housing start forecasts from S&P Global Intelligence in recent GRCs, but TURN pointed to its analysis in the still-pending SDG&E 2024 GRC, where TURN demonstrated that its forecasts have also shown upward bias.³⁸⁸ SDG&E used housing forecasts from both S&P Global Intelligence (then called IHS Markit) and Moody's in its customer connections forecast.³⁸⁹

³⁸⁴ Ex. TURN-08 (McGovern), p. 4.

³⁸⁵ D.21-08-036, pp. 142-144.

³⁸⁶ D.21-08-036, pp. 143-144.

³⁸⁷ Ex. TURN-08 (McGovern), pp. 9-10 (comparing SCE's housing starts forecast in the 2021 GRC for 2019-2022 in Figure 7 to recorded housing starts for those years in Figure 8).

³⁸⁸ Ex. TURN-08 (McGovern), p. 24;

³⁸⁹ Ex. TURN-08 (McGovern), p. 11, fn. 27.

TURN's observations regarding SCE's model made TURN cautious about using SCE's model in this GRC.³⁹⁰ TURN therefore developed an alternative methodology that does not rely on SCE's model.³⁹¹

12.1.1.2 TURN Recommends an Alternative Forecast Based Entirely on Historical Recorded Data to Promote Transparency and Eliminate the Risk of Bias.

TURN recommends a forecast of residential new meters based on the most recent 10-year average growth rate in meters, not based on housing forecasts.³⁹² TURN bases its forecast on historical data instead of proprietary third party, confidential data, as forecasts based on recent history are unbiased, relevant, transparent, and verifiable.³⁹³ TURN calculates the meter growth rate by comparing the number of new meters added in one year to the total number of residential meters at the end of the prior year.³⁹⁴ For example, the growth rate for 2022 would be calculated as follows (based on SCE's recorded data):

<u>Gross Meters in 2022 / Total 2021 Year-End Residential Meters = Growth Rate</u> 31,201 Gross Meters / 4,502,538 Existing Meters = 0.69% 2022 Growth Rate

³⁹⁰ Ex. SCE-18V01, Appendix A, pp. A42-A43 (SCE-TURN-005, Q1).

³⁹¹ Ex. SCE-18V01, Appendix A, pp. A42-A43 (SCE-TURN-005, Q1).

³⁹² Ex. TURN-08 (McGovern), p. 11.

³⁹³ Ex. TURN-08 (McGovern), p. 2.

³⁹⁴ Ex. SCE-18V01, Appendix A, pp. A44-A45 (SCE-TURN-005, Q4-Q5).

As TURN explained in response to a data request from SCE, TURN decided to use a 10year historical average growth rate to forecast new meters and customers for several reasons.³⁹⁵ As a general matter, historical averages are routinely used by utilities to forecast future work and costs in GRCs, particularly where historical values have fluctuated up and down overtime.³⁹⁶ Using an average will capture that historical variation. Historical averages also rely on transparent, recorded data, thus avoiding the complexity inherent in utility models that rely on proprietary third party data. In this case, TURN noticed that beginning in 2017, the growth rate of residential meters in the SCE's service territory began to stabilize and then drop, and therefore determined that projecting increasing rates of growth beyond historical growth rates would be inappropriate.³⁹⁷ TURN's witness additionally explained:

Further, history demonstrates that approximately once every 10 years in the US, a recession occurs. Therefore, a period shorter than 10 years may disproportionally emphasize higher or lower growth years. For instance, a 5-year average that picked up a recession would give recessionary impacts more weight than statistically appropriate, and the same would be true for a 5-year average without a recession. On the other hand, a longer historical period, such as 20 years, will pick up conditions that are no longer relevant, like earlier norms and public policies regarding housing development, density, building codes, and zoning. For an activity driven by econometric forces outside of the company's control, like customer growth, a relatively contemporary historical period like 10-years reasonably captures fluctuations and upward/downward trends without relying on outdated (or less relevant) earlier time periods.³⁹⁸

³⁹⁵ Ex. SCE-18V01, Appendix A, pp. A42-A43 (SCE-TURN-005, Q1).

³⁹⁶ See D.04-07-022 (SCE 2003 GRC), pp. 15-16 ("For those accounts which have significant fluctuations in recorded expenses from year to year, or which are influenced by weather or other external forces beyond the control of the utility, an average of recorded expenses over a period of time (typical four years) is a reasonable base expense for the [] test year.").

³⁹⁷ Ex. SCE-18V01, Appendix A, pp. A42-A43 (SCE-TURN-005, Q1).

³⁹⁸ Ex. SCE-18V01, Appendix A, pp. A42-A43 (SCE-TURN-005, Q1). Dr. McGovern has a B.A. in Math and Economics and a Master of Science and PhD in Economics. (Ex. TURN-08-Atch1, Statement of Qualifications of Dr. Jaime McGovern).

TURN originally calculated the 10-year average growth rate for 2013-2022, which equals 0.68%.³⁹⁹ In rebuttal testimony, SCE argued that TURN's forecast for residential new meters, as well as residential customers, "must be increased to reflect SCE's actual 2023 residential customer counts data."⁴⁰⁰ In 2023, SCE recorded 33,668 residential new meter connections and a total of 4,574,337 residential meters at year-end, as well as 4,578,185 residential customers.⁴⁰¹ TURN accepts SCE's recommended modification and has updated its 10-year average growth rate to include years 2014-2023. As the figure below shows, this update increases the 10-year average growth rate for new meter connections from 0.683% to 0.707%.⁴⁰²

³⁹⁹ Ex. TURN-08 (McGovern), p. 14; Ex. SCE-18V01, p. 82.

⁴⁰⁰ Ex. SCE-18V01E, p. 82 (2023 recorded residential new meter connections and customers); Appendix B, p. B108E (2023 recorded residential new meter connections and 2023 year-end recorded residential total meters).

⁴⁰¹ Ex. SCE-18V01, p. 82.

⁴⁰² Calculation: 2013 New Connections / 2012 Total Meters = 2013 % Meter Growth. SCE provides these same calculations in Ex. SCE-18V01E, p. B108E.

Residential Meters - Updated Growth Rate				
		New	% Meter	
Year	Total Meters	Connections	Growth	
2012	4,318,366			
2013	4,340,206	21,840	0.506%	
2014	4,364,839	24,339	0.561%	
2015	4,389,286	26,423	0.605%	
2016	4,416,513	32,231	0.734%	
2017	4,445,176	34,489	0.781%	
2018	4,474,407	34,759	0.782%	
2019	4,503,097	34,685	0.775%	
2020	4,535,250	32,828	0.729%	
2021	4,502,538	30,143	0.665%	
2022	4,542,098	31,201	0.693%	
2023	4,574,337	33,668	0.741%	
2013-2022 Average Growth Rate0.683%				
2014-2023 Average Growth Rate 0.707				

As a result of updating the 10-year average growth rate to 0.707%, TURN's forecast of test year 2025 residential new meter connections is 32,569. The following figure shows the derivation of this updated forecast.⁴⁰³

Residential Meters - Updated Test Year 2025 Forecast					
		New	% Meter		
Year	Year Total Meters Connections				
2023R	4,574,337	33,668	0.741%		
2014-2023 A	0.707%				
2024F	4,606,678	32,341	0.707%		
2025F		32,569	0.707%		

⁴⁰³ Calculation: 2023 Recorded Total Meters * 2014-2023 Average Growth Rate = 2024 Forecast New Connections.
2024 Forecast Total Meters = 2023 Recorded Meters + 2024 Forecast New Connections.
2024 Forecast Total Meters * 2014-2023 Average Growth Rate = 2025 Forecast New Connections.

As shown in TURN's testimony, TURN's original forecast, which applied the 2013-2022 average growth rate of 0.68% to forecast 2023-2028 residential new meter connections, produced a lower 2025 forecast of 31,452.⁴⁰⁴ Of note, TURN's updated methodology produces a 2025 forecast that is almost identical to the 5-yr average of 2019-2023 recorded new meter connections (32,505), which is the methodology SCE uses for commercial and agricultural meters.⁴⁰⁵ TURN also provides attrition year new connections forecasts using this updated 10year average meter growth rate for illustrative purposes, while noting that no party has proposed budget-based capital attrition adjustments for new connections capital.

TURN Forecast of Residential New Connections (Updated)				
Year New Connections				
2025F	32,569			
2026F	32,799			
2027F	33,031			
2028F	33,265			

12.1.1.3 SCE's Criticisms of TURN's Approach Are Misplaced.

SCE argues that TURN's new meter connections forecast "is based on a flawed methodology" because TURN compares the number of new meter connections added in one year to the number of existing meters at the end of the prior year to determine the growth rate.⁴⁰⁶ SCE asserts that the average annual growth rate should instead be calculated to capture the year-over-year change in the number of new meter connections and illustrates its alternative "growth rate"

⁴⁰⁴ Ex. TURN-08 (McGovern), p. 15, Figure 12.

⁴⁰⁵ Calculation: (34,685 + 32,828 + 30,143 + 31,201 + 33,668) / 5 = 32,505.

⁴⁰⁶ Ex. SCE-18V01, p. 81.

calculation in its rebuttal testimony.⁴⁰⁷ SCE's growth rate provides the proportional change from year to year in the number of new residential meter connections, expressed as a comparison between the number of gross meters added in the current year to the number of gross meters added in the prior year. For example, SCE calculates a growth rate of 11.44% for 2014, which reflects how the number of new meters in 2014 compares to the number of new meters in 2013 (it is 11.44% higher than the 2013 new meters), without regard to the total number of meters.⁴⁰⁸ SCE's calculated annual growth rates vary widely over the 2014-2023 time period, ranging from a high of 21.98% in 2016 to a low of -8.18% in 2021.⁴⁰⁹

SCE's approach provides a less useful perspective than TURN's because it tracks volatility, not the absolute volume of new meter connections. As shown in the figure above ("Residential Meters – Updated Growth Rate"), the years in the 2013-2023 time period with by far the highest numbers of new meter connections are 2017, 2018, and 2019 (with 34,489, 34,759, and 34,685 new meter connections, respectively). TURN's growth rate methodology likewise recognizes those three years as having the highest three growth rates in the 2013-2023 time period (with 0.781%, 0.782%, and 0.775%, respectively). In contrast, SCE's methodology assigns those three years growth rates of 7.01%, 0.78%, and -0.21%. Moreover, SCE's highest growth rate, 21.98%, belongs to a year with 32,231 new meters (2016), which is fewer new meters than recorded in 2017, 2018, 2019, 2020, and 2023. SCE's second highest growth rate, 11.44%, is for 2014, which had the lowest number of new meter connections, 24,339, among the years for which SCE calculated its growth rate. TURN submits that SCE's growth rate

⁴⁰⁷ Ex. SCE-18V01, p. 81 (referring to Appendix B (Average Growth Rate from 2013-2022)).

⁴⁰⁸ Ex. TURN-400 (TURN-SCE-124), Q4.

⁴⁰⁹ Ex. SCE-18V01E, Appendix B, p. B108E.

methodology, while interesting, does not produce a meaningful 10-year average growth rate for use in forecasting the level of new meter connections in this GRC.

SCE also criticizes TURN for using a denominator in its growth rate calculations -- the year-end total active residential meter count -- that captures changes in the number of meters from circumstances other than simply new customer connections, such as retirements, replacements, and temporary meter installations.⁴¹⁰ When TURN asked SCE to provide data that would enable TURN to isolate the impacts of retirements, replacements, and temporary meters on total year-end meters, SCE was unable to provide sufficient information to enable TURN to adjust its calculations.⁴¹¹ In any case, the recorded change in year-end total active residential meters from 2012-2023 was equal to or smaller than the number of new meter connections in all but two years, 2014 and 2022, as shown in the table below.⁴¹²

⁴¹⁰ Ex. SCE-18V01, pp. 81-82.

⁴¹¹ Ex. TURN-400 (TURN-SCE-124), Q1.

⁴¹² "Total Meters" and "New Connections" come from Ex. SCE-18V01E, Appendix B, p. B108E.
"Change in Total Meters" = difference between Total Meters in one year and the prior year. "Difference" = "New Connections" – "Change in Total Meters," such that a negative "Difference" indicates that there were fewer New Connections in that year than the Change in Total Meters.

Annual Change in Total Meters vs. New Connections				
		Change in	New	
Year	Total Meters	Total Meters	Connections	Difference
2012	4,318,366			
2013	4,340,206	21,840	21,840	-
2014	4,364,839	24,633	24,339	(294)
2015	4,389,286	24,447	26,423	1,976
2016	4,416,513	27,227	32,231	5,004
2017	4,445,176	28,663	34,489	5,826
2018	4,474,407	29,231	34,759	5,528
2019	4,503,097	28,690	34,685	5,995
2020	4,535,250	32,153	32,828	675
2021	4,502,538	(32,712)	30,143	62,855
2022	4,542,098	39,560	31,201	(8,359)
2023	4,574,337	32,239	33,668	1,429

This means that TURN's methodology has a generous bias resulting from this imprecision. TURN's methodology applies the 10-year average growth rate to 2023 recorded year-end total meters to produce a forecast of 2024 new meter connections, and then adds the 2024 new meter connections to 2023 recorded total meters to produce the forecast of 2024 total meters. That forecast of 2024 total meters is likely generous given historical trends, but TURN nonetheless uses it to generate the forecast of 2025 new meter connections.

And as shown above, TURN's growth rate calculations provide a logical approach to capturing the relative level of new connections each year. The Commission should reject SCE's contention that TURN's calculation is fundamentally flawed and produces "meaningless" results.⁴¹³

⁴¹³ Ex. SCE-18V01, p. 82.

12.1.1.4 The Commission Should Direct SCE to Use Forecasted Housing Completions Rather than Housing Starts in its Models in the Next GRC if SCE Elects to Base Its Customer and New Meter Forecasts on Housing Once Again.

SCE's asserts that "a change in the number of new meter connections or new customers is typically a result of a change in the number of housing starts that occurred up to 12 months earlier."⁴¹⁴ As a result, SCE's residential customer models "are constructed on the basis that new customers are determined primarily by housing starts (with a lag extending from zero up to 24 months depending on the region)."⁴¹⁵

TURN recommended in testimony that SCE should use housing completion data, rather than lagged housing starts, if it continues to forecast customers and new connections based primarily on housing.⁴¹⁶ The use of housing completion data as a residential meter and residential customer forecast input would have two impacts. It would (1) eliminate the need for lagged variables which would permit analysis by a greater range of software packages, and (2) reduce the error introduced by the lagged variable when housing is completed either quicker or more slowly than the specified lag.⁴¹⁷ Additionally, TURN notes that SDG&E made this switch in its 2024 GRC because, as SDG&E explained to TURN, "[u]sing housing completions, which is closely related to electric customer gains, allows SDG&E to eliminate the need to use a lagged housing start variable in its model."⁴¹⁸

⁴¹⁴ Ex. SCE-07V01 WP Bk. A, p. 167.

⁴¹⁵ Ex. SCE-07V01 WP Bk. A, p. 167.

⁴¹⁶ Ex. TURN-08 (McGovern), p. 11.

⁴¹⁷ Ex. TURN-08 (McGovern), p. 12.

⁴¹⁸ Ex. SCE-18V01, Appendix A, p. A46 (SCE-TURN-005, Q8).

In rebuttal testimony, SCE questioned whether TURN's suggestion would have a material impact on SCE's forecasts. But SCE expressed a willingness to "explore the impact of using housing completion as an alternative explanatory variable in SCE's customer and new meter forecast models in its next GRC cycle."⁴¹⁹

The Commission should accordingly direct SCE to including housing completion in its models in the next GRC (if relevant to SCE's chosen forecast methodology in that case). TURN understands that SCE may also choose to present an additional forecast based on housing starts. Even if SCE ultimately recommends a forecast based on housing starts, having the comparison will be instructive as parties evaluate SCE's forecast methodology and consider future refinements.

12.1.2 The Commission Should Adopt TURN's Commercial Gross Meter Set Forecast for the Test Year.

SCE describes its commercial gross meter forecast methodology as using "a simple linear trend projection leveraging most recent installation history as well as data and knowledge from SCE's local planning and operation organizations."⁴²⁰ SCE elsewhere suggests that it used the simple five-year average of gross meter sets in 2018-2022 for its 2023-2028 forecast, which is consistent with its calculations.⁴²¹

While SCE forecasts an increase in new commercial meters over 2022 recorded, TURN recommends an alternative forecast due to the declining trend in the number of new commercial meters from 2016-2022 and commercial real estate market conditions.⁴²² TURN's forecast for

⁴¹⁹ Ex. SCE-18V01, p. 86.

⁴²⁰ Ex. SCE-07V01, p. 99.

⁴²¹ Ex. SCE-02V08, pp. 20-21.

⁴²² Ex. TURN-08 (McGovern), pp. 17-19.

2025-2028 is based on the trendline for new commercial meters from 2016-2022.⁴²³ TURN's forecast compared to SCE's is as follows:

Commercial - New Meter Connections					
	2025	2026	2027	2028	
SCE	3,918	3,918	3,918	3,918	
TURN	2,582	2,383	2,200	2,031	
TURN-SCE	(1,336)	(1,535)	(1,718)	(1,887)	

12.1.3 The Commission Should Adopt TURN's Agricultural Gross Meter Set Forecast for the Test Year.

SCE used the five year 2018-2022 average of recorded agricultural meter sets installed to calculate an annual gross meter set forecast for 2023-2028.⁴²⁴ While SCE forecasts an increase in new agricultural meters over 2022 recorded, TURN recommends an alternative forecast due to the steadily declining trend in the number of new agricultural meters from 2018-2022 and continuing through 2023.⁴²⁵ TURN's forecast for 2025-2028 is based on the trendline for new agricultural meters from 2012-2023.⁴²⁶

In rebuttal testimony, SCE argues that TURN's methodology, "while grounded in historical precedent, may not fully account for the variable nature of agricultural demands and external influences such as economic conditions, technological advancements, and climate change."⁴²⁷ But the same can be said of SCE's 5-year average.

⁴²³ Ex. TURN-08 (McGovern), p. 17.

⁴²⁴ Ex. SCE-02V08, pp. 25-26.

⁴²⁵ Ex. TURN-08 (McGovern), pp. 15-17.

⁴²⁶ Ex. TURN-08 (McGovern), p. 17.

⁴²⁷ Ex. SCE-18V01, p. 84.

The Commission should find that SCE's forecast of a near-term rebound in the number of agricultural new connections is unsupported and adopt TURN's more conservative forecast.

Agricultural - New Meter Connections				
	2025	2026	2027	2028
SCE	184	184	184	184
TURN	102	89	79	69
TURN-SCE	(82)	(95)	(105)	(115)

TURN's forecast compared to SCE's is shown below.

12.1.4 The Commission Should Direct SCE to Include All Information and Raw Data Supporting Its Customer and New Service Connections Forecasts in Its Direct Testimony Workpapers in Future GRCs.

SCE did not include in its GRC direct testimony workpapers the input data used in its new connections regression and forecast models, such as historical and forecasted housing starts and historical meter counts, nor does it include the historical customer count data used in the development of its customer forecast model.⁴²⁸ SCE only makes that data available to intervenors through discovery, and in the case of housing data, only subject to a non-disclosure agreement.⁴²⁹ In contrast, SDG&E and GRC, who have used the same vendors as SCE, provide all of this information on a public basis in their GRC workpapers.⁴³⁰ TURN witness McGovern explained in testimony that these practices exacerbate the information asymmetry between SCE and intervenors and undermine an intervenor's ability to conduct a thorough analysis, including, e.g., considering the utility's current GRC showing in light of information provided in the last GRC, including confidential information provided only for use in that previous GRC.⁴³¹

⁴²⁸ Ex. TURN-08 (McGovern), pp. 23-25.

⁴²⁹ Ex. TURN-08 (McGovern), pp. 23-25.

⁴³⁰ Ex. TURN-08 (McGovern), pp. 23-28.

⁴³¹ Ex. TURN-08 (McGovern), p. 24.

The Commission should resolve this challenge by directing SCE to include in its future GRC direct testimony workpapers all information and raw data supporting its customer and new service connections forecasts and forecasting models. SCE should provide the required information on a non-confidential basis, unless SCE explains why it cannot publicly disclose the same information that SDG&E and SoCalGas have been able to share. It is reasonable to require SCE to include this information in its GRC direct testimony workpapers because it is fundamental to SCE's new meter and customer forecasts and is a best practice employed by SDG&E and SoCalGas in their GRCs.

In rebuttal testimony, SCE argues that it "already provides the raw data used in forecasting in workpapers, as TURN confirmed in response to a data request," making TURN's request unnecessary.⁴³² In fact, TURN confirmed that it did NOT receive SCE's modeling workpapers with SCE's initial GRC filing and testimony. Instead, TURN received those files in October 2023 in response to a data request.⁴³³ SCE also offered to contact its vendors to see if it may produce the data it uses in its forecasting models publicly in the next GRC, given the practices of SDG&E and SoCalGas.⁴³⁴ This step should position SCE well to comply with a Commission order to include the data it uses in its forecasting models in its testimony workpapers in future GRCs (either as public information or with a more examined explanation of why it must treat that information as confidential).

⁴³² Ex. SCE-18V01, p. 86.

⁴³³ Ex. SCE-18V01, Appendix A, p. A47 (SCE-TURN-005, Q10).

⁴³⁴ Ex. SCE-18V01, pp. 86-87.

12.2 Customer Requested System Modifications

12.2.1 Rule 20 Conversions

TURN sponsored testimony addressing Rule 20A Conversions in Ex. TURN-10.

Following SCE's rebuttal testimony, TURN conferred with SCE, and TURN agrees that SCE's revised forecast for Rule 20A Conversions in its rebuttal testimony is reasonable.

13. POLES

13.1 Poles O&M

13.2 Poles Capital

14. VEGETATION MANAGEMENT

SCE's Vegetation Management expense forecast represents 54%, the single largest portion, of its Grid Activities forecast.⁴³⁵ The proposed forecast of \$654 m in Vegetation Management expenses in 2025 is an increase of approximately 55% over the recorded Vegetation Management expenses in 2023, \$421.046 m in constant 2023 dollars (i.e. 2022 dollars).⁴³⁶ As described in TURN's equity testimony, California customers are increasingly energy insecure and additional rate increases can lead to evictions and homelessness.⁴³⁷ This is especially an issue in Black, Indigenous, and other People of Color (BIPOC) communities which are often disproportionately housing and energy burdened.⁴³⁸ SCE's proposed increase in

⁴³⁵ Ex. TURN-09-E, p. 1.

⁴³⁶ Ex. SCE-11, p. A-3.

⁴³⁷ Ex. TURN-03, p. 5.

⁴³⁸ Ex. TURN-03, p. 5 ("In addition to increasing homelessness generally, SCE's proposed rate increase will particularly harm Black, Indigenous, and other People of Color (BIPOC) communities which comprise a disproportionate share of the homeless population and have higher average housing and energy burdens than their white counterparts.").

vegetation management costs is incompatible with affordable rates and the Commission should reject the SCE forecast in favor of the TURN proposed alternative of \$430.408 m in 2025.

The Commission's consideration of SCE's proposed Vegetation Management portfolio of projects must bear in mind that the utility has made a massive investment, over \$4 billion, in covered conductor deployment.⁴³⁹ This investment has reduced wildfire risk in SCE's territory by 80%.⁴⁴⁰ Section 15 of this brief addresses the continued system hardening work that is expected in the SCE territory. At the same time, TURN acknowledges that the Commission has not addressed whether and how certain regulations and requirements for electrical line maintenance are adjusted where covered conductor is installed.⁴⁴¹ TURN similarly acknowledges that the utility must meet current compliance requirements where covered conductor is installed. However, as discussed further below, the extensive deployment of covered conductor should inform the Commission's consideration of discretionary programs.

TURN makes two major adjustments to SCE's proposals, with additional reductions of certain programs. First, TURN's adjustments to the SCE forecast removes the impacts of SCE's reliance on inflated escalation rates when developing program forecasts. The escalation rates relied on by SCE, 10% for Routine Line Clearing and the Hazard Tree Program⁴⁴² and 15% for inspections,⁴⁴³ reflect neither the Federal Reserve's Inflation Targets of 2%⁴⁴⁴ or the actual

⁴³⁹ Ex. TURN-12, p. 8, T. 2.

⁴⁴⁰ Ex. TURN-12, p. 1:19-23.

⁴⁴¹ SCE, Terry Ohanian, 13 Tr. 1346:20-1347:1 (SCE Ohanian).

⁴⁴² Ex. TURN-09-E, p. 8: SCE used a 10% rate for routine line clearing; Ex. TURN-09-E, p. 11: SCE used a 10% rate for Dead, Dying and Diseased trees; Ex. SCE-13, Vol. 10, p. 21: SCE includes a 15% increase for inspections costs.

⁴⁴³ Ex. TURN-09-E, p. 14.

⁴⁴⁴ Ex. TURN-09-E, p. 8; Ex. TURN-09-E, p. 11.

inflation rates which, between 2022-2024, were 6.2%, 4.8% and 4% (average of 5%).⁴⁴⁵ Regardless of whether the Commission determines it is more important to rely on the target rates or actuals, the 10% and 15% relied on by the utility are inappropriately high. To correct for this, TURN's alternative forecasts rely on either 2% or 5%.⁴⁴⁶

Second, TURN proposes that certain costs for discretionary work be removed from the forecast because the proposed program is not an efficient use of limited ratepayer dollars. TURN recommends that the Commission reject certain discretionary programs as the program benefits do not exceed the costs.

Ultimately, the utility has not demonstrated that its proposed forecast is reasonable and that the resulting budget is consistent with affordable utility service. TURN recommends that the Commission make the following adjustments to SCE's proposed Vegetation Management request:

- As discussed in Section 38.3 of this brief, the Commission should reject the SCE request for a two-way balancing account for Vegetation Management in favor of a one-way account.
- The Commission should adjust inflated unit costs relied on when developing the Routine Line Clearing, Dead, Dying and Diseased Tree and Inspections programs.
- The Commission should reject the SCE budget to fund two types of simultaneous inspections and instead direct the utility to more efficiently manage the transition to Remote Sensing.
- The Commission should reject certain discretionary spending for the Routine Line Clearing and Hazard Tree Management Programs.

⁴⁴⁵ Ex. SCE-13, Vol. 10, p. A18.

⁴⁴⁶ Ex. TURN-09-E, p. 8, 11; Ex. SCE-13, Vol. 10, p. 20.

• The Commission should reject expansion of the Dead, Dying and Diseased Tree Removal Program beyond current maintenance levels.

As explained further below, the Commission must reject the SCE Vegetation Management proposed budget and should instead adopt TURN's proposed budget adjustments to SCE's 2025 Vegetation Management O&M expenses which result in a budget of \$430.408 m. While the detail of the recorded costs provided for 2023 is not sufficiently granular to compare specific programs, the SCE proposal represents an increase of \$223 m over 2023 recorded costs. TURN's proposal still represents an increase of \$9 m over 2023 costs.

TURN recommends that the Commission should find, based on the information below, that the TURN forecast is sufficient to fund Vegetation Management consistent with safe, reliable, and affordable utility service.

14.1 Inspections Program: The Commission Should Ensure that the Utility Deliver the Promised Benefits of Remote Sensing to its Ratepayers.

SCE proposes \$85.146 m for inspections in 2025 including \$28.084 m for Traditional Ground Inspections and \$55.713 m in Remote Sensing.⁴⁴⁷ This compares to \$40.706 m in Traditional Ground Inspections in 2022 recorded dollars.⁴⁴⁸ TURN recommends the Commission significantly reduce SCE's proposed forecast to protect ratepayers from paying twice for inspections and to correct for SCE's reliance on an inflated escalation factor. Instead, the Commission should adopt TURN's proposed budget of \$52.122 m for 2025, subject to the posttest year (PTY) escalation adopted in this proceeding.⁴⁴⁹

⁴⁴⁷ Ex. SCE-13, Vol 10, p. 8.

⁴⁴⁸ Ex. SCE-13, Vol. 10, p 8.

⁴⁴⁹ Ex. TURN-09-E, p. 12.

Ultimately, the utility must determine how to best complete inspections with budget the Commission provides the utility.⁴⁵⁰ Although TURN testimony recommends a budget of \$47.122 m for ground inspections and \$5 m for remote sensing inspections, TURN is neutral as to which type of inspections SCE pursues, provided that the chosen method efficiently and effectively identifies the work that needs to be done and that ratepayers are delivered the promised benefits. TURN's \$52 m proposed forecast provides the utility a significant increase over 2022 costs, sufficient to fund the utility's preferred inspection technique.

14.1.1 SCE's Proposal Will Charge Ratepayers for both Ground and Remote Sensing Inspections with No Timeline for Delivering the Efficiencies Promised by LiDAR Technology.

SCE's test year (TY) 2025 forecast would have ratepayers pay for both Traditional Ground Inspections and remote sensing inspections relying on Light Detecting and Ranging (LiDAR) throughout the rate case period, with no clear timeline of when the efficiency benefits of remote sensing will be delivered to customers. SCE proposes that it will complete a full scope of Traditional Ground Inspections as well as LiDAR in 2025.⁴⁵¹ In later years of the Rate Case Cycle, the utility will rely on LiDAR and Traditional Ground Inspections but with a decreasing portion of the territory inspected by ground each year.⁴⁵² In the last year of the GRC cycle, the utility will complete LiDAR as well as 20% of a full round of Traditional Ground Inspections.⁴⁵³

⁴⁵⁰ Ex. TURN-09-E, p. 14 ("It is ultimately the company's decision how it will allocate resources within the authorized budget adopted in this GRC.").

⁴⁵¹ Ex. SCE-13, Vol. 10, p. 10.

⁴⁵² Ex. SCE-13, Vol. 10, p. 10.

⁴⁵³ Ex. SCE-13, Vol. 10, p. 10.

SCE boasts that LiDAR is ready to be deployed and that it is the preferred method of inspections.⁴⁵⁴ The utility emphasizes that LiDAR is not a pilot program, stating "SCE does not view Remote Sensing as being in a pilot stage. To the contrary, SCE views Remote Sensing as a tested and effective means of conducting inspections for vegetation management, having used, improved, and expanded its use of LiDAR since 2019."⁴⁵⁵ In comparing LiDAR with Traditional Ground Inspections, SCE states that LiDAR offers more advancements than Traditional Ground Inspections.⁴⁵⁶ Additionally, SCE has a "high level of confidence" in LiDAR due to "its established processes, mature vendor relationships, and solid expertise in this space."⁴⁵⁷

The utility explains:

With the implementation of wide-scale remote sensing, SCE expects a reduced need for ground inspections. In 2025, the full scope of ground inspections will be used to begin validating the remote sensing results on tree-to-conductor clearance and mapping the remote sensing data to SCE's tree inventory. Thereafter, the ground inspections will be necessary primarily to identify hazard tree conditions, conduct ground inspections at locations that are blocked from aerial views necessary for remote sensing (e.g., overhanging tree limbs), and respond to emergent concerns raised by customers.⁴⁵⁸

However, SCE provides no timeline or proof of when/if Traditional Ground Inspections will be

completely phased out or how the utility determined the appropriate amount of ground

inspections required.⁴⁵⁹ SCE states that LiDAR will lead to a more efficient system, while also

⁴⁵⁴ Ex. SCE-13, Vol. 10, pp. 13-16.

⁴⁵⁵ Ex. SCE-13, Vol. 10, p. 20.

⁴⁵⁶ Ex. SCE-13, Vol. 10, p. 16.

⁴⁵⁷ Ex. SCE-13, Vol. 10, p. 13.

⁴⁵⁸ Ex. SCE-02, Vol. 10, p. 23.

⁴⁵⁹ Ex. SCE-13, Vol. 10, p. 10 ("In 2025, SCE envisages using Remote Sensing to inspect its entire network of approximately 60,000 circuit miles for distribution and transmission. In this year, SCE plans to concurrently employ Traditional Ground Inspections across its territory to facilitate the verification of Remote Sensing data. In 2026-2028, SCE anticipates gradually reducing its use of Traditional Ground Inspections, with 2028 estimated to require 20% of the 2025 forecast for this work.").

requesting that ratepayers simultaneously also fund "less efficient" Traditional Ground Inspections. The longer ratepayers fund two types of inspections, the less value ratepayers receive from any potential efficiencies provided by LiDAR.

Despite this confidence in LiDAR, SCE still argues that it is necessary to complete Traditional Ground Inspections on the same circuits. SCE states, "[o]nly by completing both Remote Sensing and Traditional Ground Inspections for the full network over a single annual inspection cycle can SCE take full advantage of ground inspectors' validation of remote sensing data, which would facilitate SCE's development of a reliable and verified digital inventory baseline."⁴⁶⁰ The utility further states that, "predictive models require training through multiple iterations, alongside verification of the models' results by ground crews."⁴⁶¹ SCE, however, is not clear how many iterations or how much training is required in order to optimize the use of LIDAR. With each iteration and its attendant costs on ratepayers, any benefits or efficiencies are lost.

If anything, SCE has demonstrated that only one full cycle of dual inspections is required. There is no discussion in SCE's testimony regarding why more than one cycle of both remote sensing and Traditional Ground Inspection is required in certain locations or why there will still be ground inspections in the last year of this rate cycle. Perhaps most problematic, SCE provides no timeline for a full phase out of ground inspections. This is not the first instance of SCE overpromising and under-delivering ratepayer benefits. In its last rate case, SCE argued that expanded line clearing was a short-term cost, stating "[o]nce the initial deeper trims are complete in 2019-2020, trimming activities are expected to decrease in 2021 because there is less effort

⁴⁶⁰ Ex. SCE-13, Vol. 10, p. 18.

⁴⁶¹ Ex. SCE-13, Vol. 10, p. 18.

involved in maintaining already established trims."⁴⁶² However, as discussed in Section 14.2 below, these higher costs are now treated as the new status quo.⁴⁶³

In light of the ratepayer experiences with deep trims, SCE's promises of efficiencies are hard to believe especially when ratepayers are paying for "efficient" LiDAR in addition to Traditional Ground Inspections. Because the utility has failed to meet its burden of proof that the extent of the duplicative work requested by the utility is necessary, the Commission should provide a more limited inspections forecast–ensuring the benefits of LiDAR are delivered to ratepayers.

14.1.2 SCE's "Alternative" Proposal Relies on Inflated Escalation Costs to Create an Artificially High Forecast

Rather than justify a short-term increase in costs to launch remote sensing, to make its double inspections forecast appear reasonable, SCE proposes an artificially high alternative forecast for Traditional Ground Inspections without Remote Sensing. SCE proposes a 2025 alternative forecast amount of \$79.598 m.⁴⁶⁴ This compares to \$40.706 m in Traditional Ground Inspections recorded in 2022 and is five million less than the LiDAR forecast for 2025.⁴⁶⁵

In its rebuttal testimony, SCE explained that its alternative forecast uses a "15% increase to reflect initial bids in the new contract cycle as well as the overall competitiveness of this area given other utilities' wages and collective bargaining terms."⁴⁶⁶ It is unreasonable for the utility to rely on a 15% escalation rate when inflation targets are 2% and actual inflation rates over

⁴⁶² Ex. TURN-09-Atch1, p. 21: A.19-08-013, Ex. SCE-02, Vol. 6, p. 21.

⁴⁶³ See TURN-09-E, pp. 5-6 discussion of A.19-08-013 routine line clearing forecast.

⁴⁶⁴ Ex. SCE-13, Vol. 10, p. 9.

⁴⁶⁵ Ex. SCE-13, Vol. 10 p. 8.

⁴⁶⁶ Ex. SCE-13 Vol. 10, p. 21.

2022-2024 were 6.2%, 4.8% and 4% (on average, 5%).⁴⁶⁷ Further, SCE has since identified the outcome of contract negotiations in 2023. SCE has seen an "approximately 6% increase comparing previous lump sum and current unit rate contracts, completed negotiations Q4 2023."⁴⁶⁸

Especially given the outcome of contract negotiations, coming in almost 10% lower than the assumed escalation, the utility alternative inspections forecast seems designed solely to make the proposal that ratepayers pay for double inspections more reasonable. Because the alternative is unjustified itself, it cannot be used to justify SCE's preferred Remote Sensing alternative, If the utility believes that Remote Sensing is the best path forward for ratepayers, it should do so without unnecessarily raising rates. TURN's proposal provides a significant increase in Vegetation Management budget which the utility can then determine how best to spend, consistent with its responsibilities to provide safe and reliable service.

14.2 Routine Line Clearing: Inflated Escalation and Discretionary Costs Should be Removed from SCE's Routine Line Clearing Budget.

Routine Line Clearing comprises of the largest portion of SCE's proposed Vegetation Management forecast.⁴⁶⁹ SCE's rebuttal testimony requests \$344.159 m for its routine Line Clearing work.⁴⁷⁰ TURN recommends that the Commission reject the SCE proposal and instead adopt an adjusted forecast of \$213.776 m.⁴⁷¹

⁴⁷⁰ Ex. SCE-13, Vol. 10, p. 23, T. 11-9

⁴⁶⁷ Ex. SCE-13, Vol. 10, p. A18.

⁴⁶⁸ Ex. 602, pp. 1-2.

⁴⁶⁹ Ex. SCE-13, Vol. 10, p. 23.

⁴⁷¹ Ex. TURN-09-E, p. 2.

Although SCE may refer to this work as routine, routine does not mean that the work is required under CPUC general orders for maintenance of the utility distribution system. SCE, however, uses the word routine to refer to both their required compliance activities as well as discretionary work.

As discussed above, TURN acknowledges that compliance requirements have not been adjusted to account for the deployment of covered conductor. The utility must continue to complete compliance work consistent with legal and regulatory requirements, regardless of what type of wire is installed. The discretionary work that SCE proposes, however, may not be the best use of scarce ratepayer dollars.

TURN is concerned at the rate of growth of Routine Line Clearing costs. The Routine Line Clearing program relies on a trim rate that is constant regardless of the amount that is trimmed, meaning trimming to compliance distances or beyond compliance distances costs the same amount. It has not always been the case that trims were priced this way. In SCE's last rate case, SCE advocated for a higher Routine Line Clearing forecast, based on the need for more costly deeper trims.⁴⁷² In this rate case, SCE has abandoned this argument in favor of a universal trim rate, regardless of the amount trimmed.

SCE also argued that the deeper trim rates it advocated for in its last rate case were necessary and temporary costs, and that those costs would return to maintenance levels in the future.⁴⁷³ Not only have ratepayers seen even higher Routine Line Clearing costs that do not differentiate between compliance and beyond-compliance trimming, ratepayers have also not received relief in this rate case from those promised temporarily elevated deep trim costs.

⁴⁷² Ex. TURN-09-E, p. 6, Tables 4, 5.

⁴⁷³ Ex. TURN-09-E, pp. 5-6 discussion of A.19-08-013 routine line clearing forecast and Tables 4, 5.

TURN's reductions to the routine line clearing forecast help deliver the promised benefits to utility customers.

14.2.1 TURN Adjusts Routine Line Clearing Unit Costs to Remove Unnecessarily High Escalation.

SCE's Routine Line Clearing forecast includes a market escalation factor of 10% to adjust for contract cost increases.⁴⁷⁴ TURN's alternative forecast relies, instead, on a market escalation factor of 2%, a level that is consistent with the Federal Reserve Inflation Target Rates.⁴⁷⁵ It is unreasonable for the utility to ask for, and receive, a 10% escalation rate when inflation targets are 2% and actual inflation rates over 2022-2024 were 6.2%, 4.8% and 4% (on average, 5%).⁴⁷⁶

The 10% escalation rate was intended to account for contract negotiation uncertainties. However, the Routine Line Clearing contracts have now been completed and the actual amounts are known. Trim rates, now a blended cost rather than differentiated for maintenance and deep trims, increased 4.4%.⁴⁷⁷ Contracts do see a 65% increase in unit costs for removal of trees.⁴⁷⁸ However, only 5% of trees are slated for removal, meaning that the 65% increase in costs for removal only applies to a small portion of total costs. ⁴⁷⁹

⁴⁷⁴ Ex. TURN-09-E, p. 11.

⁴⁷⁵ Ex. TURN-09-E, p. 8.

⁴⁷⁶ Ex. SCE-13, Vol. 10, p. A18.

⁴⁷⁷ Ex. TURN-602.

⁴⁷⁸ Ex. TURN-602.

⁴⁷⁹ Ex. WP SCE-02, Vol. 10, p. 50.

While the increase in costs may exceed the 2% escalation proposed by TURN, it comes nowhere near the 10% assumed by SCE. TURN recommends that the Commission rely on TURN's number as more aligned with observed contract escalation.

14.2.2 TURN Adjusts Unit Costs Downward to Account for the Utility Relying on Elevated Costs the Commission Approved as a Temporary Measure.

SCE's rebuttal testimony states that TURN's testimony "embeds a significant amount of Expanded Line Clearance."⁴⁸⁰ This is not accurate. TURN is concerned that SCE has not delivered the cost reductions promised in the last general rate case to ratepayers, and that instead the cost of deeper trims has increased the cost of all trims in this rate case. TURN is also concerned that Expanded Line Clearing, a line item of approximately \$8 m has a low Risk Spend Efficiency (RSE).

Unlike the previous rate case, SCE contracts rely on a single blended cost for trims, regardless of the amount of vegetation trimmed. Trims to be completed at this single blended cost include 36,000 deep trims and 790,000 total trims.⁴⁸¹ This blended "unit costs are calculated using 2022 data as inputs."⁴⁸²

In the 2021 rate case, the utility expected that there would be a 25% increase in deep trims required with the costs doubling for those deep trims in "High Fire Impact" Areas.⁴⁸³ Meanwhile in non-high fire impact zones, there was a 12.5% increase in deep trims required with

⁴⁸⁰ Ex. SCE-13, Vol. 10, p. 27.

⁴⁸¹ Ex. SCE-13, Vol. 10, p. 28, fn. 27.

⁴⁸² Ex. SCE-13, Vol. 10, p. 23.

⁴⁸³ Ex. TURN-09-E, p. 6, T. 5.
a 1.25 cost multiplier.⁴⁸⁴ While SCE requested these increased costs, the utility assured the Commission that "trimming activities are expected to decrease in 2021 because there is less effort involved in maintaining already established trims."⁴⁸⁵

Table 7: SCE's Table II-9: Routine Line Clearing

2018-2022 Recorded/2025

Forecast Summary of SCE, Cal Advocates and TURN Positions.⁴⁸⁶

Line #	Routine Line Clearing	SCE Recorded					2025 Forecast				Variance from SCE Original Position		Variance from SCE Rebuttal Position	
							SCE Original	SCE Rebuttal	Cal		Cal		Cal	
		2018	2019	2020	2021	2022	Position	Position	Advocates	TURN	Advocates	TURN	Advocates	TURN
1	Labor	\$ 9,623	\$ 15,156	\$ 26,916	\$ 24,728	\$ 25,134	\$ 16.658	\$ 16.658	\$ -	\$ -	\$ (16,658)	\$ (16,658)	\$ (16,658)	\$ (16,658)
2	Non-Labor	\$ 122.220	\$ 263,269	\$ 351.688	\$ 324,079	\$ 303,430	\$ 331.056	\$ 327,501	\$ 307,800	\$ 213,776	\$ (23,256)	\$ (117,280)	\$ (19,702)	\$ (113,725)
3	Total	\$ 131,843	\$ 278,425	\$ 378,603	\$ 348,807	\$ 328,564	\$ 347.714	\$ 344,159	\$ 307,800	\$ 213,776	\$ (39,914)	\$ (133,938)	\$ (36,360)	\$ (130,384)

As the table above shows, while there was a limited decrease in costs in 2022. Costs remained significantly higher than 2019 and especially 2018 levels. The decrease is insufficient to deliver any efficiencies promised to ratepayers, and instead it is an elevated basis from which to derive reasonable unit costs looking forward.

Because of the blended unit cost approach relied on by the utility, it is difficult to determine with specificity and complete confidence exactly how the workload impacts of the deep trims impacted the contract prices. The deep trims were completed concurrently with line clearing costs increases incorporating the impacts of SB 247.⁴⁸⁷ Contract prices were then used to determine the unit cost relied on in this case. Considering these changed circumstances, TURN endeavored to unravel the influence to identify a forecast in this case that would deliver the promised cost efficiencies to customers. TURN determined the unit costs without the

⁴⁸⁴ Ex. TURN-09-E, p. 6, T.5.

⁴⁸⁵ Ex. TURN-09-E, p. 5, citing A.19-08-013, SCE-02, Vol. 6, p. 21.

⁴⁸⁶ Ex. SCE-13, Vol. 10, p. 23.

⁴⁸⁷ Ex. SCE-10, Vol. 10, p. 43.

"temporary" cost by using the approach relied on by SCE in the 2021 GRC to estimate a new unit cost. The result is an adjusted unit cost which TURN used for the purposes of developing a forecast for SCE. While it may reflect a different approach from SCE's, it provides an estimate of unit costs without the upward influence of "temporary" deeper trims.

14.2.3 Expanded Line Clearing Costs Should be Removed from the Forecast.

The Routine Line Clearing program proposed by SCE includes \$8.347 m in discretionary work to complete "Expanded Line Clearing."⁴⁸⁸ TURN recommends reducing the Routine Line Clearing program to remove the costs of Expanded Line Clearing because of its extraordinarily low RSE.⁴⁸⁹

As an initial matter, TURN addresses SCE's general position on the information provided by the RSE. SCE notes that, "a comprehensive wildfire risk mitigation should not be based solely on RSEs."⁴⁹⁰ To support its position SCE quotes the Commission's Safety and Enforcement Division (SED) which states that it would be "suboptimal from an aggregate risk portfolio standpoint," to focus on RSE as the sole isolated factor in selecting mitigation methods.⁴⁹¹ The SED language referenced is from a 2018 SED report on a PG&E filing that was made in 2017. Not only has the calculation of RSEs developed since this report, in the interim

⁴⁸⁸ Ex. TURN-601, p. 4.

⁴⁸⁹ Ex. TURN-09-E, p. 8.

⁴⁹⁰ Ex. SCE-13, Vol. 10, p. 34.

⁴⁹¹ Ex. SCE-13, Vol. 10, p. 43, fn. 92; *See* California Public Utilities Commission, Risk and Safety Aspects of Risk Assessment and Mitigation Phase Report of Pacific Gas and Electric Company Investigation 17-11-003 (March 30, 2018) via URL: https://www.cpuc.ca.gov/-/media/cpuc-website/divisions/safety-policydivision/

reports/sed_ramp_evaluation_pge_033018a.pdf, p. 18 ("A potential pitfall with looking at risks and mitigations in isolation based on the RSE scores is that the allocation of risk mitigation spending to the different risks may be suboptimal from an aggregate risk portfolio standpoint.").

the Commission has adopted a settlement approach to calculating and presenting quantitative risk analysis (D.18-12-014) as well as an update to the settlement approach (D.22-12-027). The RSEs calculated by SCE in this case were calculated consistent with the settlement approach which has been vetted by the CPUC.⁴⁹²

The SED report on SCE's current approach to calculating risk and the RSE does not include any similar language related to RSE as was included in the 2018 Report on PG&E. Instead, SED recommends that the utility model compliance actions related to vegetation management to determine if the Commission should reassess current compliance requirements:

Routine Vegetation Management activities are performed to maintain clearances around poles and equipment on the distribution and transmission systems, to comply with current regulations and Commission recommendations. SPD recommends that risk modeling be conducted for Routine Vegetation Management precisely because it is a compliance activity that reduces risk. If they are determined to have low RSE values, the Commission should re-evaluate associated regulatory requirements.⁴⁹³

This more recent SED reflects on the role and power of RSEs and demonstrates greater confidence in RSE scores and the information they provide. Therefore, SCE's assertions regarding the role of RSE based on the 2018 Report should be dismissed by the Commission.⁴⁹⁴

As noted above, the Routine Line Clearing program includes \$8.347 m in discretionary work for Expanded Line Clearing. SCE clarified that the utility treats compliance activities as "regulatory requirements from General Order (GO) 95 and other statutes."⁴⁹⁵ Expanded Line

⁴⁹² Ex. TURN-804, p. 11.

⁴⁹³ Ex. TURN-804, p. 32.

⁴⁹⁴ TURN provides an additional response to SCE's attempts to downplay the usefulness of RSEs in Section 5.3 above.

⁴⁹⁵ Ex. TURN-601, p. 2.

Clearing was applied to 36,000 trees.⁴⁹⁶ SCE Witness Ohanian clarified that SCE typically trims its trees to meet the [Grid Resiliency Clearance Distance (GRCD)] clearances recommended, but not required, by General Order 95, Rule 35.⁴⁹⁷ The 36,000 trees reflect the number of occurrences where the utility was unable to obtain the GRCD and "if we were able to trim these [trees] to the GRCD, it would cost [SCE] the \$8 million."⁴⁹⁸

The RSE of Expanded Line Clearing, as calculated by the utility, remains a 4,⁴⁹⁹ as compared to 125,431 for Routine Line Clearing otherwise.⁵⁰⁰ When SCE's RSE is converted to a cost-benefit ratio (CBR), the CBR of this program is only 0.1,⁵⁰¹ which means that Expanded Line Clearing would provide \$10 benefit for every \$100 spent.⁵⁰² Considering the Expanded Line Clearing budget reflects the costs to extend the trim from the required distance to the recommended distance, the RSE and CBR reflect the cost-effectiveness of expanded clearances. The stark comparison between the RSE for Expanded Line Clearing and Routine Line Clearing suggests that expanded line clearing does not reflect a reasonable investment of ratepayer dollars. Affordability requires that the CPUC reject the program as not cost-effective because of its low RSE. Additionally, the low RSE suggests, consistent with the SED language in the RAMP

⁴⁹⁶ Ex. SCE-02, Vol. 10, p. 44.

^{497 13} TR 1379:23-1380:7 (SCE Ohanian).

⁴⁹⁸ 13 TR 1386:1-9 (SCE Ohanian).

⁴⁹⁹ Ex. TURN-4, App. A, Table 3, p. A-26; Ex. SCE-02, Vol. 10, p. 44, T, II-14.

⁵⁰⁰ In Ex. TURN-4, p. A-18, SCE gave its "Distribution and Transmission Routine Vegetation Management" program a RSE score of 125,341. Whereas, in Ex. TURN-601, Table II-7, SCE gave its compliance Routine Line Clearing program a score of 125,431. Given the proximity of these two numbers and the fact that Ex. TURN-601 separated the Routine Line Clearing program into compliance and non-compliance, it is likely that the utility simply transposed numbers to reach the number cited in Ex. TURN-601.

⁵⁰¹ Ex. TURN-4, App. A, Table 3, p. A-26.

⁵⁰² Ex. TURN-09-E, p. 8.

report, that the Commission's GO, and particularly its recommended clearances, should be reconsidered.

SCE argues that factors beyond the RSE support continuing Expanded Line Clearing work pointing to the reduction in Tree-Caused Circuit Interruptions (TCCI), a potential "precursor to an ignition."⁵⁰³ While a reduction in TCCIs does reduce the risk for ignition, only a limited number of TCCIs, however, have led to ignitions. In 2022, three TCCIs caused by living trees, resulted in ignitions.⁵⁰⁴ SCE acknowledges that in addition to covered conductor, the utility deploys "fast curve settings and asset inspections" across the high fire risk area (HFRA).⁵⁰⁵ There is not information provided on the impact of these programs, in particular fast curve settings, on the reducing the potential for a TCCI to become an ignition. However, SCE's witness Fugere testified that fast curve has a standalone mitigation effectiveness of 40 to 50%.⁵⁰⁶ Given the low RSE of Expanded Line Clearing and the failure to demonstrate unique and additional mitigation of risk, there is an insufficient basis to approve Expanded Line Clearing.

In addition to the reductions in unit costs requested by TURN above, the Commission should also remove the \$8.347 m of costs of expanded line clearing from the SCE request.

⁵⁰³ Ex. SCE-13, Vol. 10, p. 31.

⁵⁰⁴ Ex. SCE-36.

⁵⁰⁵ Ex. SCE-36.

⁵⁰⁶ 10 TR 924:16 – p. 925:7 (SCE Fugere).

14.3 Dead, Dying, Diseased Tree Removal: The Dead, Dying, and Diseased Tree Removal Budget Should be Reduced to Remove Inflated Contract Inflation and Expansion of the Program.

SCE requests \$30.204 m for its Dead and Dying Tree Removal Program.⁵⁰⁷ TURN recommends that the Commission adjust for overinflated contract escalation and adopt a forecast of \$25.108 m.

First, consistent with the adjustments to other vegetation management programs and for the reasons described above, TURN adjusts the contract escalation cost from SCE's proposed 10% to 2% consistent with the Federal Reserve Target Inflation rate.⁵⁰⁸ The 2% relied on by TURN better reflects the actual contractual increase of 4.4% observed by SCE.⁵⁰⁹

Second, SCE asserts that it will need to remove an increasing number of trees each year because of "anticipated drought conditions."⁵¹⁰ However, the utility provides no evidence of anticipated drought conditions or the impact of said drought on tree mortality to justify the anticipated increase. TURN accordingly also adjusts the forecast to reflect the maintenance level of removals observed in 2023. The result of these two adjustments is TURN's recommended \$5.096 reduction in the Dead and Dying Tree Management Program.

⁵⁰⁷ Ex. SCE-02, Vol. 10, p. 79.

⁵⁰⁸ Ex. TURN-09-E, p. 11.

⁵⁰⁹ Ex. TURN-602.

⁵¹⁰ Ex. SCE-02, Vol. 10, p. 79.

14.4 Hazard Tree Management Program: SCE Fails to Demonstrate that the Hazard Tree Management Program Is a Cost-Effective and Reasonable Use of Ratepayer Funds.

SCE requests \$44.202 m for Hazard Tree Management Program (HTMP).⁵¹¹ TURN recommends that the Commission reject all costs for HTMP.⁵¹²

According to SCE, the HTMP "mitigates ignition and thus wildfire risk stemming from live trees and/or their parts that may appear healthy, but that could fall or blow into SCE's lines."⁵¹³ SCE's program targets trees "that are not at risk of growing into the regulatory clearance distance (RCD) as defined by Routine Vegetation Management."⁵¹⁴ This program is discretionary and does not provide benefits sufficient to justify its costs.

SCE calculated a low RSE score of 8 for the HTMP.⁵¹⁵ When the RSE is converted to a

CBR, the CBR is either 0.2 using SCE's discount rates, or 0.1 using TURN's discount rates.⁵¹⁶

HTMP would produce only \$10 to \$20 of benefit for every \$100 dollars spent.⁵¹⁷ Given the

limited value of this program to ratepayers, TURN recommends the Commission not fund the HTMP.

The Utility suggests that because the HTMP is included in an approved Wildfire Mitigation Plan (WMP) the program is now a legal requirement. SCE erroneously conflates an

⁵¹¹ Ex. SCE-13, Vol. 10, p. 38.

⁵¹² Ex. TURN-09-E, p. 9.

⁵¹³ Ex. SCE-13, Vol. 10, p. 38.

⁵¹⁴ Ex. TURN-09, p. 9 (Citing Ex. SCE-02, Vol. 10, p. 61).

⁵¹⁵ Ex. SCE-02, Vol. 10, p. 69, T. II-26. When SCE's questionable discount rates are adjusted, SCE's RSE falls to 4. Ex. TURN-4, App. A, Table 3, p. A-25.

⁵¹⁶ Ex. TURN-04, App. A, Table 3, p. A-25.

⁵¹⁷ Ex. TURN-8-E, p. 10; Ex. TURN-04, App. A, Table 3, p. A-25. Table 3 also shows that this program ranks in the bottom 20% of all SCE programs based on SCE's RSE.

approved WMP with compliance with CPUC general orders and statutory requirements. This is inconsistent with precedent. The Commission has previously addressed the meaning of WMP approval, and the role of the CPUC over programs included in an approved WMP:

The Commission has made it abundantly clear that it does not consider cost recovery when reviewing a utility's WMP; rather, the issue of whether WMP costs are just and reasonable is left to an electrical corporation's GRC or application permitted by Pub. Util. Code § 8386.4(b)(2).797 Therefore, the Commission's ratification of the Office of Energy Infrastructure Safety's approval of specific activities included within a WMP does not indicate the costs of those activities are just and reasonable, nor does it preclude the Commission from determining the appropriate costs for recovery based on the expected pace or scope of a utility's forecasted WMP activities.⁵¹⁸

The CPUC retains the jurisdiction to determine that certain projects or programs otherwise included in the WMP are not consistent with just and reasonable rates. It does not meet the Utility's burden of proof for demonstrating a project is just and reasonable to simply point to WMP approval. SCE must demonstrate to the CPUC that the program represents the program, pace, and scope to address the identified risk.

SCE also draws a comparison to more recent Office of Energy Infrastructure Safety (Energy Safety) approval of PG&E's Hazard Tree Program: "[Energy Safety] advocated for continuing a hazard tree program similar to SCE's HTMP."⁵¹⁹ SCE argues that the PG&E program identified by Energy Safety is "similar to SCE's HTMP."⁵²⁰ The approval of a "similar" program is an insufficient basis for the Commission to approve the \$44 m HTMP. First, SCE provides no information to the record describing PG&E's hazard tree program and provides no evidence that the two programs are similar. Second, SCE provides no evidence that the grid and

⁵¹⁸ D.21-08-036, pp. 251-52.

⁵¹⁹ Ex. SCE-13, Vol. 10, p. 42.

⁵²⁰ Ex. SCE-13, Vol. 10, p. 42.

vegetation conditions between the two utilities are similar and, therefore, the justification for the program in each territory is not comparable. Third, SCE does not demonstrate that the reasons for the adoption of the program by Energy Safety in PG&E's territory are present for SCE. This entire argument should be given no weight by the Commission.

SCE contends that the HTMP is required because it is the best mitigation to address the unique risk of blow in and fall in of live trees.⁵²¹ As illustrated in Ex. SCE-35 there are TCCIs caused by live trees beyond the compliance zone (i.e. the trees that would be mitigated by HTMP).⁵²² However, only a limited number of the TCCIs result in ignitions. Only three TCCIs caused by living trees, inside or outside compliance zones, resulted in ignitions in 2022.⁵²³ These three ignitions caused by live trees, especially without additional context and information on alternative mitigations, do not justify that the \$44 m dollar program is a reasonable use of ratepayer dollars. The utility must show that its program is appropriately tailored to the mitigate the risk faced by the utility program, and that the mitigation is the most cost-effective means of address this risk. SCE did not provide alternatives to the HTMP–either alternative mitigations or an alternative scope of the program. For instance, SCE maintains a "Tree Risk Calculator to conduct Level 2 Assessments and recommend mitigations based on the risk score."⁵²⁴ Indeed, a smaller program with a better-defined Tree Risk Calculator scores may be able to accurately

 ⁵²¹ Ex. SCE-13, Vol. 10, p. 43 ("Only HTMP addresses the primary root cause of TCCIs, with live, visibly healthy trees or tree parts known to be linked to tree-caused ignition events.").
 ⁵²² Ex. SCE-35.

⁵²³ Ex. SCE-35; Ex. SCE-36.

⁵²⁴ Ex. SCE-37.

capture the limited trees that threaten an ignition at a more reasonable budget. In the absence of a reasonably scoped program the Commission is required to reject SCE's proposed HTMP.

14.5 Seasonal Patrols/AOC/Emergent Work

- 14.6 Structure Brushing
- 14.7 Environmental Support For Vegetation Management
- 14.8 Wildfire Mitigation Vegetation Management Technology Solutions

15. WILDFIRE MANAGEMENT

15.1 Overview

15.2 Grid Hardening

15.2.1 Summary of SCE's Proposal and TURN's Recommendations

For the 2025-2028 rate case period, SCE proposes a drastic change to its wildfire mitigation strategy by dramatically increasing its reliance on undergrounding. Over this period, SCE proposes to underground 580 overhead miles with at a forecast four-year cost of \$3.27 billion, an average of \$815 million per year.⁵²⁵ By contrast, SCE spent an average of \$16 million per year on undergrounding in 2020-2023.⁵²⁶ In this GRC, undergrounding consumes more than half of SCE's capital forecast for grid hardening,⁵²⁷ whereas, in SCE's prior GRC, over 90% of SCE's wildfire management capital forecast was for covered conductor.⁵²⁸

⁵²⁵ Ex. SCE-04, Vol. 5, Part 2A, p. 10.

⁵²⁶ Based on Ex. SCE-04, Vol. 5, Part 2A, p. 9, Figure I-1. The 2023 figure is a forecast.

⁵²⁷ Ex. SCE-15, Vol. 5, Part 2, p. 3, Table I-2. Undergrounding is 53 percent of SCE's 2023-2028 grid hardening forecast.

⁵²⁸ D.21-08-036, p. 187. Undergrounding was not even discussed in SCE's 2021 GRC decision.

According to the utility, the driver of the shift in SCE's grid hardening policy is its classification of certain miles as "Severe Risk Areas" (SRA), which SCE describes as the "riskiest locations."⁵²⁹ Out of SCE's total 580-mile undergrounding proposal, 570 of those miles are in the remaining unhardened miles in locations SCE designates as SRAs.⁵³⁰ As will be discussed below, SCE's qualitative and opaque criteria for determining SRAs do not correspond with the CPUC's prescribed S-MAP methodology for calculating risk scores in D.18-12-014 and include a significant proportion of miles that are relatively low risk under the CPUC's risk scores. This mismatch between "severe" risk miles classified as SRAs and the miles with the highest risk scores under the S-MAP framework occurs even though the S-MAP risk scores include quantification of many of the same criteria used by SCE to designate an SRA.

TURN recommends major revisions to SCE's proposal that would achieve the same amount of risk reduction at \$2 billion less cost to ratepayers.⁵³¹ Covered conductor should continue to be the centerpiece of SCE's grid hardening efforts, with undergrounding serving a more limited role targeted to the truly riskiest locations where this much higher cost mitigation is warranted. The details of TURN's recommendation, including how it compares with SCE's proposal, are presented in Section 15.2.6 below.

⁵²⁹ Ex. SCE-15, Vol. 5, Part 2, p. 20.

⁵³⁰ Id.

⁵³¹ Ex. TURN-12-E (Borden/TURN), p. 2.

15.2.2 Covered Conductor Reduces Significant Wildfire Risk, Can Be Enhanced with Emerging Technologies, Is Much More Cost-Effective than Undergrounding, and Can Be Deployed Much More Quickly

SCE has achieved significant risk reduction with covered conductor in a short time period.⁵³² Covered conductor should remain the primary grid hardening tool because it is significantly more cost-effective than undergrounding and can be deployed more quickly.

Notably, SCE concurs with this overall assessment of the benefits of covered conductor, describing it as a "prudent and cost-effective mitigation" that "can buy down risk in a relatively short amount of time."⁵³³ Most importantly, as SCE states, covered conductor "mitigates the risk drivers that tend to cause the largest fires."⁵³⁴

SCE also touts the benefits of its installed covered conductor in reducing the number and duration of PSPS events because of its ability to reduce the risk of contact from foreign objects. SCE states that it was able to raise wind-speed de-energization thresholds from National Weather Service "advisory" levels, 46 mph gust speed, to "warning levels" of 58 mph gusts on 2,300 miles, including 64% of the Tier 3 HFTD miles where it has installed covered conductor.⁵³⁵ As required by D.21-08-036, SCE also provides data showing dramatic reductions in PSPS activations, minutes, and affected customers because of expedited installation of covered conductor.⁵³⁶

⁵³² This point is discussed further in Section 15.2.3 below.

⁵³³ Ex. SCE-04, Vol. 5, Part 2A, pp. 38-39.

⁵³⁴ Ex. SCE-04, Vol. 5, Part 2A, p. 52.

⁵³⁵ Ex. SCE-04, Vol. 5, Part 2A, p. 46.

⁵³⁶ SCE reports that for 70 high PSPS frequency circuits where it expedited covered conductor in 2022, PSPS de-energizations fell from 60 in 2020 and 2021 to 4 in 2022, affected customers fell from 180,000 to 1,900, and customer minutes of interruption (CMI) fell from 50,000,000 to 800,000. Ex. SCE-04, Vol. 5, Part 2A, p. 45.

SCE calculates the overall mitigation effectiveness of covered conductor as 73%.⁵³⁷ While this is less than the 98% effectiveness SCE reports for undergrounding, SCE notes that covered conductor can approximate the effectiveness of undergrounding when teamed with complementary emerging technologies such as REFCL and spacer cable.⁵³⁸ SCE has determined that REFCL has significant potential risk reduction benefits per dollar⁵³⁹ and is already forecasting significant use of REFCL, which it calls a "measured deployment" while continuing to validate its mitigation effectiveness, forecasting \$240 million in capital expenditures for this rate case period, at a unit cost of \$100,000 per overhead mile.⁵⁴⁰ SCE states that spacer cable is a particularly useful enhancement to covered conductor where tree fall-in risk is high and that the utility has initiated a pilot, which if successful, would support use of spacer cable as an additional mitigation.⁵⁴¹ Thus, covered conductor delivers high mitigation effectiveness more quickly than undergrounding, and in the longer term, can be supplemented with other technologies that, when combined with covered conductor, appear likely to approximate undergrounding's effectiveness.

Covered conductor delivers its significant benefits at a unit cost per overhead mile that is *six times less* than the cost of undergrounding on a weighted average basis, according to SCE's forecast.⁵⁴²

⁵³⁷ Ex. SCE-15, Vol. 5, Part 2, p. 30.

⁵³⁸ Ex. SCE-04, Vol. 5, Part 2A, pp. 16-17.

⁵³⁹ Ex. SCE-04, Vol. 5, Part 2A, p. 82.

⁵⁴⁰ Ex. SCE-04, Vol. 5, Part 2A, pp. 83-84; Ex. TURN-12-E, p. 24 and fn. 29.

⁵⁴¹ *Id.*, pp. 16-17.

⁵⁴² Ex. TURN-12-E, p. 23.



 Figure 5: SCE Forecast Unit Cost of Undergrounding vs. Covered Conductor

 per Overhead Circuit Mile (\$ Thousands)⁵⁴³

Relatively low unit costs and high mitigation effectiveness make covered conductor far more cost-effective than undergrounding. SCE's modeling in connection with its direct testimony indicates covered conductor is about 67 percent more cost-effective, on average, than targeted undergrounding,⁵⁴⁴ a point SCE did not challenge in rebuttal.⁵⁴⁵ However, this measure of covered conductor's superiority is actually a significant understatement, because SCE

⁵⁴³ *Id.*, p. 24.

⁵⁴⁴ Ex. TURN-12-E, p. 25.

⁵⁴⁵ TURN's testimony discussed in this section relied on the modeling data SCE provided in connection with its direct showing in this case, which TURN had sufficient opportunity to review, analyze, and address in its testimony. SCE's rebuttal testimony did not challenge any of TURN's findings and conclusions showing the enormous cost-effectiveness advantage of covered conductor over undergrounding. Instead, SCE's rebuttal presented a complex new analysis. Even though TURN had limited time to review that new analysis, TURN was able to identify serious problems that invalidated its results, as discussed in Section 15.2.7 below.

calculated cost-effectiveness based on its proposal rather than by an apples-to-apples comparison of the two alternatives if deployed *to the same circuit segments*.⁵⁴⁶

Using the data supporting SCE's direct showing, TURN was able to fill this gap in the record by performing two apples-to-apples comparisons, both of which show an even greater cost-effectiveness advantage for covered conductor. Neither were challenged by SCE. The first figure below shows that, when the two alternatives are compared on the same circuit segments, covered conductor is actually *between 260 and 288 percent* more cost-effective than undergrounding.



Figure 6: Risk Spend Efficiency of Undergrounding vs. Covered Conductor⁵⁴⁷

The following figure shows relative cost-effectiveness based on more granular risk tranches, using a slightly different methodology that aggregates circuit segments together at the project

⁵⁴⁶ Ex. TURN-12-E, p. 25.

⁵⁴⁷ Ex. TURN-12-E, p. 25, Figure 13.

level (rather than the circuit segment level, shown above) and sorts from highest to lowest undergrounding RSE.⁵⁴⁸



Figure 7: Risk Spend Efficiency of Undergrounding vs. Covered Conductor⁵⁴⁹

For each risk percentile shown in the figure above, covered conductor is more cost-effective than undergrounding. At the even more granular individual circuit segment level, covered conductor is more cost-effective than undergrounding on 99.6 percent of circuit segment miles (1,821 of 1,828).⁵⁵⁰

Each of TURN's cost-effectiveness comparisons discussed in this section relied on the modeling data SCE provided in connection with its direct showing in this case. SCE's rebuttal testimony did not challenge any of TURN's findings and conclusions based on that data showing the enormous cost-effectiveness advantage of covered conductor over undergrounding.

⁵⁴⁸ Ex. TURN-12-E, p. 26.

⁵⁴⁹ Id.

⁵⁵⁰ Id.

15.2.3 SCE's Proposed Massive Investment in Undergrounding Would Be a Poor Use of Ratepayer Funds In Light of the Significant Risk Reduction that Has Already Been Accomplished With Covered Conductor, at Ratepayer Expense

SCE's significant ratepayer-funded investment in covered conductor has resulted in massive risk reduction, leaving a relatively small percentage of risk remaining on SCE's system. The figure below shows SCE's estimate of risk reduced due to grid hardening (covered conductor and undergrounding) and fast curve settings, which cut off power when vegetation or another object come in contract with a powerline. As shown, by the end of 2024, SCE estimates a 72% reduction⁵⁵¹ in wildfire risk. The majority of this risk reduction, 77 percent, is due to grid hardening, almost entirely covered conductor.⁵⁵²

⁵⁵¹ Based on the figure below: 100% initial wildfire risk minus 27.76% (rounded to 28%) equals 72%.

⁵⁵² Ex. TURN-12-E, p. 19. Risk reduction from 2023 to 2024 is calculated based only on SCE's expected risk reduction percentage solely from grid hardening (primarily covered conductor). *Id*.



Figure 8: Wildfire Risk Remaining After Grid Hardening and Fast Curve Settings (2018-2028)⁵⁵³

Because of the significant risk reduction that has already been achieved, SCE anticipates much less risk reduction in the 2025-2028 rate case period, only 12%,⁵⁵⁴ compared to the 72% risk reduction from 2018 to 2024. Incongruously, to achieve this additional 12% risk reduction, SCE proposes to *increase* its grid hardening spending from \$3.5 billion in 2021-2024 to \$4.2 billion in 2025-2028.⁵⁵⁵

In light of the significant and successful grid hardening efforts that ratepayers have already funded and the higher cost-effectiveness of covered conductor, the Commission's Safety Policy Division's (SPD) RAMP Evaluation Report questioned the wisdom of SCE's proposed massive investment in undergrounding:

⁵⁵³ *Id.*, p. 21.

⁵⁵⁴ Based on the figure above: 28% (rounded) remaining wildfire risk in 2024 minus 16% remaining risk in 2028 (rounded to 28%) equals 12%.

⁵⁵⁵ Ex. TURN-12-E, p. 20.

Given the lower RSE, SPD staff question the appropriateness of substantial investment of ratepayer funds for TUG [Targeted Undergrounding] after the large-scale implementation of the CC [Covered Conductor] program has been underway for years. The [covered conductor program] was supposed to prioritize and install CC on the highest-risk circuit segments in the program's early years. Hence, there is no widespread need for TUG since the highest-risk circuit segments have CC installed.⁵⁵⁶

TURN could not agree more. In assessing whether SCE's broad-scope undergrounding proposal is appropriate, the Commission should account for the huge previous investment and significant risk reduction achieved with covered conductor, given that there is now significantly lower absolute risk than when SCE embarked on grid hardening measures in 2018 and the fact that ratepayers face the burden of ever-increasing rates and bills.

15.2.4 SCE Wrongly Relies on Its Flawed Designation of So-Called 'Severe Risk Areas' (SRA) to Define the Scope of Its Undergrounding Proposal

15.2.4.1 The Scope of SCE's Undergrounding Proposal Hinges on Its Definition of SRAs

SCE bases its grid hardening forecast largely on its designation of certain portions of its service territory as "Severe Risk Areas" (SRAs), where SCE states overhead powerlines must be undergrounded in most instances to limit or eliminate risk. The upshot is to make undergrounding the default choice in SRAs, rather than determining the best alternative based on location-specific conditions.

⁵⁵⁶ SPD Staff Evaluation Report on SCE's 2022 RAMP, A.22-05-013, Nov. 10, 2022, p. 35, made a part of the record of this case by ALJ ruling. Tr. Vol. 10, p, 996:20 – p. 997:3.

SCE's SRAs are based on the following criteria, only one of which need be present to qualify a location as an SRA:⁵⁵⁷

- 1. Population egress constraints, high fire frequency, and burn-in buffer into egress locations.⁵⁵⁸
- 2. Significant fire consequence Acres burned consequence greater than 10,000 over an 8-hour unsuppressed model simulation.
- 3. High winds Locations, which if fully covered with covered conductor, would still be subject to high PSPS likelihood.
- 4. Communities of Elevated Fire Concern (CEFCs) Smaller geographic areas where terrain, construction, and other factors could lead to smaller, fast-moving fires threatening populated locations under benign (normal) weather conditions.

Notably, as discussed below, SCE can and does take all of these factors into account in its S-

MAP required quantitative risk modeling. In order to justify the complication and expense of

undergrounding, SCE uses opaque qualitative criteria to label circuits as "severe risk," regardless

of how the S-MAP risk score of these circuits compares with others.

Based on these criteria, SCE identifies 3,226 miles of its service territory that are SRAs.

However, most of these miles are already hardened or will be by 2024 (primarily with covered

conductor); only 590 miles of SRA remain that will not be hardened through 2024, almost all of

which the utility seeks to underground in the GRC period.⁵⁵⁹

SCE proposes that it decide which grid hardening alternative to deploy based on the

following decision tree.

⁵⁵⁷ SCE-04, Vol. 5, Part 1A, pp. 23-24.

⁵⁵⁸ SCE states in SCE-04, Vol. 5, Part 1A, p. 50 fn. 48: "Burn-in buffers are areas adjacent to the egress constrained areas such that if the fire originated there, could enter the egress constrained area." ⁵⁵⁹ SCE-04, Vol. 5, Part 1A, Table II-7, p. 44.



Figure 9: SCE Decision Process for Grid Hardening⁵⁶⁰

Notable in this decision tree is the fact that whether undergrounding is deployed hinges on whether a location is an SRA. The only situation in which unhardened SRA miles will not be undergrounded is when SCE deems undergrounding not "feasible," which SCE says is based on "factors such as terrain, cost and customer constraints."⁵⁶¹ SCE clearly sees such infeasibility as a rare occurrence, as SCE proposes that 580 of the 590 SRA miles to be hardened in 2025-2028 would be undergrounded.⁵⁶² At a high *average* forecasted unit cost of over \$5 million per mile, it is not clear what, if any, cost threshold would be required for SCE to determine a project is "infeasible."

⁵⁶⁰ *Id.*, p. 45, Figure II-19.

⁵⁶¹ *Id.*, fn. 4 to Figure II-19.

⁵⁶² SCE-04, Vol. 5, Part 1A, Table II-7, p. 44.

Importantly, SCE's decision tree gives no consideration to a comparison of costeffectiveness values for undergrounding and overhead hardening at the location in question. Thus, even though each location is different with different risk drivers and cost considerations, SCE would rather use a default-to-undergrounding approach in SRAs rather than performing a meaningful comparison of alternatives.

15.2.4.2 SCE's SRA Designations Amount to an End Run Around the S-MAP Risk Assessment Framework It Agreed to and Is Required to Use

SCE's expansive undergrounding proposal relies on its claim that, even after six years of extensive grid hardening, supposedly in the highest risk locations, 580 remaining unhardened miles have such "severe" risk as to require undergrounding. As this section will show, the S-MAP risk modeling required by D.18-12-014 does not support this conclusion.

Instead, SCE resorts to qualitative criteria that use *a different definition of risk* to support the company's desired undergrounding plan. In the settlement adopted in D.18-12-014, SCE and the other large utilities agreed that risk should be defined as likelihood of risk event (LoRE) times consequence of risk event (CoRE).⁵⁶³ However, as SCE does not dispute,⁵⁶⁴ its SRA criteria completely ignore the likelihood side of the equation and only consider consequences, a fundamental departure from the definition of risk SCE is required to use to assess its current level of risk throughout its system. Moreover, SCE's decision-making methodology treats all

⁵⁶³ Ex. TURN-12-E, p. 8.

⁵⁶⁴ Ex. SCE-15, Vol. 5, Part 2, p. 27 (SCE's SRA analysis "does not include quantitative LoRE values . . .")

580 circuit miles that meet *any* of the proposed criteria as effectively the same, instead of the risk ranking by location that is afforded by the S-MAP risk assessment framework.⁵⁶⁵

As a result of the disconnect between the SRA criteria and the S-MAP risk framework, many of the circuit segments that SCE paints broadly as having "severe" risk actually have very low relative and absolute risk measured by S-MAP risk scores, as shown in the figure below.



Figure 10: Absolute Risk per Mile Values of Circuit Segments in "Severe Risk Areas"⁵⁶⁶

The figure ranks SRA circuit segments from highest to lowest risk per mile by segment according to SCE's own S-MAP risk analysis. The x-axis shows the percent of cumulative risk, while the y-axis shows the risk per mile of the circuit segment. The figure not only shows that

⁵⁶⁵ Ex. TURN-12-E, p. 8.

⁵⁶⁶ Ex. TURN-12-E, p. 9, Figure 5.

most of the SRA segments have very low risk per mile, but that there is enormous variation in risk among the broad swath of segments SCE uniformly classifies as "severe" risk areas.⁵⁶⁷

Other analyses conducted by TURN highlight the lack of rigor in SCE's categorization approach. SCE's risk data at the circuit segment level showed that 554 out of the total of 588 miles that SCE classifies as SRA are in the bottom 50 percent of risk calculated under the S-MAP framework; 404 miles are in the bottom **10%**.⁵⁶⁸ SCE's approach also misses some highrisk miles -- 44% of unhardened miles in the top 50% of risk as measured by S-MAP -- that might warrant at least being considered for undergrounding.⁵⁶⁹

SCE's rebuttal does not dispute TURN's analysis showing the mismatch between SCE's SRAs and the company's S-MAP risk scores. In fact, SCE acknowledges that 70% of the miles in its undergrounding program would not address the top 50% of risk as measured by the S-MAP risk scores.⁵⁷⁰ Instead, SCE quibbles that some miles in a circuit may be high risk and others may be low risk.⁵⁷¹ However, when the illustrative map that SCE presents was revised based on "isolatable circuit segments" that aggregate segments to the project level (i.e., the level at which projects are implemented, which is what TURN's proposal is based on),⁵⁷² the maps show, as expected, that high and low risk segments are not interspersed --- high risk segments are grouped in one area and low risk segments in another.⁵⁷³

⁵⁶⁷ *Id.*, p. 9.

⁵⁶⁸ *Id.*, p. 11.

⁵⁶⁹ *Id.*, pp. 9-10.

⁵⁷⁰ Ex. SCE-15, Vol. 5, Part 2, p. 19, fn. 48.

⁵⁷¹ *Id.*, p. 11.

⁵⁷² Ex. TURN-12-E, p, 27.

⁵⁷³ Ex. TURN-801, SCE response to TURN DR 113, question 4, Figures 2 and 3. TURN's proposal thus addresses SCE's concern regarding "project level" considerations by using much more aggregated data

The upshot is to validate TURN's point that SCE's SRAs include an astonishingly high percentage of circuit miles that would justify SCE performing undergrounding projects in areas in the bottom 50% of risk as measured by the S-MAP framework.⁵⁷⁴ Undergrounding lines in areas with relatively low risk, without due consideration of viable alternatives like covered conductor, is a poor use of ratepayer funds.

15.2.4.3 SCE's SRA Criteria Are a Poor Fit for Explaining When Undergrounding May Be Superior to Overhead Hardening

The SRA criteria identified by SCE do not provide a sound basis for choosing undergrounding over covered conductor or covered conductor combined with emerging technologies such as REFCL. Undergrounding may be a superior alternative to covered conductor in locations, for example, where the risk drivers are not aligned with the best use case of covered conductor, where undergrounding can be accomplished for a reasonably comparable cost, or where the absolute risk (based on likelihood and consequence per the S-MAP risk scores) is relatively high, among other possible scenarios. The following table, presented in TURN's testimony, shows the mismatch between SCE's SRA criteria and reasons that would make undergrounding the superior alternative.⁵⁷⁵

⁵⁷⁵ Ex. TURN-12-E, pp. 12-13.

provided by SCE itself in response to SPD's request in the RAMP. SCE's argument is therefore irrelevant to TURN's proposal.

⁵⁷⁴ SCE's rebuttal points to Table II-6 on p. 15 showing that risk per mile is higher in areas designated as SRA. (Ex. SCE-15, Vol. 5, Part 2-E, p. 15) But the table also shows that SRAs only capture about half of the risk calculated under the S-MAP framework. Thus, SCE's table reinforces TURN's point that the risk in SRAs is not homogenously "severe" and does not justify a uniform and highly expensive response of undergrounding almost all miles that SCE identified as SRA.

SRA Criteria	Does this demonstrate undergrounding is the preferred solution?
 Population egress constraints, high fire frequency, and burn- in buffer into egress locations. 	These criteria do not necessarily support undergrounding over covered conductor. They are not related to the location-specific drivers that determine the effectiveness of overhead hardening compared to undergrounding and otherwise, do not show why undergrounding is superior. SCE does not explain why its SRA designation better explains where undergrounding is preferable compared to its circuit segment risk scores, which quantitatively capture many of the risk factors in these criteria.
 Significant fire consequence Acres burned consequence greater than 10,000 over an 8- hour unsuppressed model simulation. 	A high consequence score does not necessarily mean risk drivers would not be significantly mitigated by covered conductor.
 High winds – Locations, which if fully covered with covered conductor, would still be subject to high PSPS likelihood. 	Of the four criteria, this one best explains why undergrounding may be preferable. Still, from this criterion, it is not clear what the likelihood or frequency of these high wind events are, nor whether this likelihood justifies the high expense of undergrounding. Further, undergrounding is by far the most expensive and likely least cost-effective mitigation to address PSPS risk. Most other programs intended to mitigate PSPS risk would be significantly more cost-effective and in the interest of ratepayers: customer care programs – batteries and generator back- up, sectionalization, switching plans, mobile generator deployment, and likely others.
 4. Communities of Elevated Fire Concern (CEFCs) – Smaller geographic areas where terrain, construction, and other factors could lead to smaller, fast-moving fires threatening populated 	Same problems as identified for the first and second criteria.

SCE's rebuttal testimony does not challenge any of these points.

15.2.4.4 SCE's Justification for Attempting to Circumvent the CPUC's Mandated Risk Assessment Framework Is Without Merit

SCE defends its use of a process that ignores both the CPUC's definition of risk and the risk scores from the CPUC's required framework on the grounds that its approach is "holistic" and uses both qualitative and quantitative analysis.⁵⁷⁶ SCE claims that its approach is superior because it mitigates gaps in data and considers a wider range of risk factors.⁵⁷⁷ SCE's rebuttal names a litany of factors that supposedly are only considered under its SRA approach.⁵⁷⁸

The fatal problem with SCE's argument is that all of the factors it lists can and should be part of the S-MAP risk analysis as needed to yield the utility's best estimate of risk. As noted, the S-MAP framework mandates that risk scores be based on likelihood of the risk event (LoRE) multiplied by the consequences of the risk event (CoRE).⁵⁷⁹ Utilities are to provide their best estimates of these values at "as deep a level of granularity as reasonably possible."⁵⁸⁰ These

192

⁵⁷⁶ Ex. SCE-15, Vol. 5, Part 2, p. 7.

⁵⁷⁷ *Id.* Similarly, SCE argues that the most useful data for quantitative risk analysis is not always up to date. *Id.*, p. 10.

⁵⁷⁸ *Id.*, pp. 8, 14-15. The listed factors include: egress constraints, engineering reviews, risk management evaluation, meteorologic studies, geospatial and topographical analysis, construction practicality and feasibility, photographic evidence, local knowledge, fuel loading, potential risk drivers, wind patterns, geographic features, and proximity to open wildlands.

 ⁵⁷⁹ D.18-12-014, adopting S-MAP Settlement, Att. A., Row 13. See Section 5.3.1 above.
 ⁵⁸⁰ Id., Row 14.

estimates are to be based on data "whenever practical and appropriate."⁵⁸¹ However, the S-MAP framework is clear that "the available data should not restrict the application of the risk assessment methodologies."⁵⁸² Where the utility believes there are data limitations or gaps, "SME judgment should be used if the methodologies require use of data that is not available."⁵⁸³

The bottom line is that, under the S-MAP framework, utilities should use whatever data they believe will provide the best estimates of risk in a given situation. Where the utility is dissatisfied with the available data – because the utility believes the data has gaps or is not up to date -- it can, and indeed must, use SME judgment to supplement the data. Thus, every factor that SCE claims must be assessed to accurately evaluate risk should have been considered in SCE's quantitative risk analysis. In response to a TURN data request concerning its rebuttal contentions, SCE had no choice but to concede this point. It acknowledged that the S-MAP framework is capable of addressing all the factors SCE lists and correctly noted that the S-MAP framework "recognizes the need to supplement MAVF . . . through the use of SME judgment."⁵⁸⁴ Moreover, SCE states that it incorporates both egress risk and PSPS risk – two of the key SRA criteria -- into its wildfire risk modeling and risk scores at the circuit segment level.⁵⁸⁵

In sum, SCE's criticisms of the supposed shortcomings of the S-MAP framework are completely unfounded. If SCE is dissatisfied with its data, it should either work harder at improving and updating the data or, where not possible, supplement the data with SME

⁵⁸¹ *Id.*, Row 31.

⁵⁸² Id.

⁵⁸³ Id.

⁵⁸⁴ Ex. TURN-802, SCE's response to TURN DR 113, Question 5(a).

⁵⁸⁵ SCE-04, Vol. 5, Part 1A, pp. 15, 17.

judgment. The answer is not to invent a whole new way of assessing risk on its system that bypasses the CPUC's decade-long effort to develop a detailed and comprehensive uniform risk assessment methodology,⁵⁸⁶ one that SCE and the other utilities agreed to.

Moreover, SCE's criticisms are a disservice to the obviously concerted efforts of SCE's subject matter experts to develop the best possible risk estimates under the S-MAP framework. Those S-MAP estimates are at odds with the coarse, broad brush results using the SRA criteria and do not justify SCE's expansive undergrounding proposal.

SCE's attempt to ignore its quantitative results in lieu of a qualitative framework serves the company's incentive to increase rate base to increase profits. Recognizing and correcting for this well-known incentive of investor-owned utilities is the foundation of the Commission's responsibility to protect the public interest.

15.2.5 SCE Places Inadequate Emphasis on Cost-Effectiveness in Its Proposal for How to Spend Ratepayer Money

The foregoing has shown that, in most (perhaps all) locations, covered conductor will be a more cost-effective use of ratepayer funds than undergrounding. Yet, even though covered conductor has been an extremely successful mitigation for SCE, even in its highest risk areas, SCE seeks the Commission's blessing for an approach that would make undergrounding – the most expensive alternative -- the default choice for almost all areas SCE labels as SRAs. SCE admits that its grid hardening proposal was not created with any thresholds for affordability in mind.⁵⁸⁷

⁵⁸⁶ See Section 5.3.2 above.

⁵⁸⁷ Ex. TURN-12-E, pp. 5-6, citing SCE Response to DR TURN-SCE-039, question 2a: "There are no affordability thresholds or constraints specific to the planned grid hardening scope in the referenced volume of testimony."

SCE claims that its approach is justified because of its extreme view that relative risk is irrelevant and only absolute risk should be considered. The extremeness of SCE's position is evident in its response to SPD's pragmatic suggestion in its RAMP evaluation that SPD consider revamping its proposal by focusing 40% of its spending to achieve 85 percent of the proposed risk reduction. SCE labeled this recommendation "irrelevant," claiming that "it is the remaining (i.e. residual) absolute risk that is relevant, not the amount of relative risk that can be bought down by making the limited investment SPD recommends."⁵⁸⁸

SCE's position is out of synch with the Commission's decision-making framework and basic common sense. It is also profoundly insensitive to the importance of ensuring the affordability of essential electric service, a factor that was central to the Commission's recent decision on PG&E's undergrounding proposal, as discussed below.⁵⁸⁹

The Commission has made clear that examination of the tradeoff between costs and risk reduction benefits is central to the Commission's review and obligations in GRCs. In its decision on SCE's 2021 test year GRC, the CPUC explained that "[o]ne of the central tasks in this proceeding is to balance safety and reliability risks with the associated cost to mitigate those

⁵⁸⁸ Ex. SCE-04, Vol. 5, Part 1A, p. 54. TURN's testimony identifies another example of SCE's extreme position, a response to a Cal Advocates data request in which SCE challenged Cal Advocates' question as improperly focused on relative risk, not absolute risk. TURN's expert Mr. Borden explains the fallacy in SCE's response: "In SCE's example, the highest risk circuit mile would have risk that is 9,900 percent greater than all remaining risk combined. Yet, SCE would have the CPUC compel ratepayers to fund the same type of work on all of the other 999 miles if these were considered 'SRA'. This is precisely why cost-effectiveness matters – not to say that no mitigation should ever take place on any of those remaining 999 circuit miles but, in considering whether and how to mitigate risk for those miles, it is necessary to examine cost-effectiveness tradeoffs that allow for both safe and affordable electric service." Ex. TURN-12-E, pp. 14-15 (emphasis in original).

⁵⁸⁹ D.23-11-069, p. 278. Even if the Commission finds merit to SCE's emphasis on absolute risk reduction over cost-effective risk reduction, we note that TURN's recommendation reduces the same amount of risk as SCE's proposal, for \$2 billion less.

risks."⁵⁹⁰ This balance requires prioritizing where and how ratepayer funding is targeted. As the Commission stated in PG&E's 2014 GRC decision:

Virtually everything a utility does [has]some nexus to safety and can be deemed to have some safety impact, *but the emphasis should be on those initiatives that deliver the optimal safety improvement in relation to the ratepayer dollars spent*.⁵⁹¹

In the decade since that decision, the Commission has devoted considerable resources to requiring utilities to provide RSE calculations, which it has found are "critical for determining whether utilities are effectively allocating resources to initiatives that provide the greatest risk reduction benefits per dollar spent, thus ensuring responsible use of ratepayer funds."⁵⁹²

In its most recent GRC decision, D.23-11-069, the Commission put these tools and principles into practice in significantly reducing PG&E's undergrounding proposal and directing PG&E to increase the use of covered conductor, reducing the cost of PG&E's grid hardening proposal by \$1.72 billion.⁵⁹³ The Commission explained that PG&E's more expensive proposal would "present challenges for customers regarding affordability"⁵⁹⁴ and that the adopted plan "strikes a balance between risk reduction, feasibility, timeliness, and cost containment."⁵⁹⁵ By

⁵⁹⁵ *Id.*, p. 273.

⁵⁹⁰ D.21-08-036, p. 30.

⁵⁹¹ D.14-08-032, p. 28 (emphasis added).

⁵⁹² D.21-08-036, p. 38, quoted in PG&E's 2023 GRC decision, D.23-11-069, p. 44. D.23-11-069 also noted the Commission's prior statement that one of the goals of the S-MAP settlement adopted in D.18-12-014 was to "use risk reduction per dollar spent to prioritize projects."

⁵⁹³ D.23-11-069, p. 273.

⁵⁹⁴ *Id.*, p. 278. The decision also stated on page 295: "Costs are a significant concern and . . . the Commission must examine alternatives to mitigate the burden to ratepayers"

directing PG&E to focus its grid hardening efforts on the highest risk locations,⁵⁹⁶ the adopted plan reduced more risk than PG&E's proposal at less cost.⁵⁹⁷

Thus, it is clear that SCE is out of step with the Commission's decision-making approach when it dismisses as irrelevant SPD's suggestion in its RAMP evaluation that SCE re-focus its grid hardening program to provide most of the risk reduction benefit at significantly reduced cost. SCE missed an opportunity to develop a more balanced grid hardening proposal that takes the relative benefits of risk mitigation alternatives into account and that recognizes risk reduction must be balanced with affordability.

15.2.6 TURN's Recommendation

15.2.6.1 TURN Recommends that Undergrounding Be Targeted Where It Will Produce the Greatest Benefit and that Covered Conductor Be Used in the Remaining Unhardened HFTD Locations

TURN's recommendation strikes a more reasonable balance between risk reduction and

affordability by achieving as much risk reduction as SCE's proposal at approximately \$2 billion

less cost.

Rather than using SCE's flawed SRAs to determine the appropriate scope of

undergrounding, TURN utilizes RSEs calculated under the S-MAP framework.⁵⁹⁸ Because of

⁵⁹⁶ *Id.*, p. 272.

⁵⁹⁷ *Id.*, p. 295.

⁵⁹⁸ As explained in TURN's testimony, TURN utilized SCE's risk modeling assumptions and outputs presented in its workpapers and aggregated SCE's extremely granular circuit segments (many of which are just a few feet long) into "isolatable segments" which is akin to how circuit segment assets are aggregated into a single "project." This provided a significantly more aggregated analysis that is more realistic for project planning purposes, reducing the approximately 38,000 circuit segment to around 1,900 "isolatable" sections. TURN then calculated RSEs for undergrounding on each isolatable circuit

the significant concentration of risk, these values show a precipitous drop-off in the costeffectiveness of undergrounding after the top 10-20% of risk, as shown in the figure below.



Figure 11: Risk Spend Efficiency of Undergrounding⁵⁹⁹

While these results would justify scoping undergrounding based on the top 20% of risk, TURN's proposal conservatively determines the appropriate scope of undergrounding based on the number of miles in the top **50%** of risk, which equates to 177 miles.⁶⁰⁰

Accordingly, TURN recommends that 177 miles serve as the maximum number of overhead miles to underground in the 2025-2028 rate case period.⁶⁰¹ The actual miles to be undergrounded up to this cap should be determined by the location-specific analysis discussed in Section 15.2.6.2 below. For the remainder of the HFRA miles that SCE proposes to harden, 1,651 miles in total, TURN recommends covered conductor deployment, in light of its

section and sorted from highest to lowest undergrounding RSE in order to target the most cost-effective circuits. Ex. TURN-12-E, pp. 27-28.

⁵⁹⁹ Ex. TURN-12-E, p. 28, Figure 15.

⁶⁰⁰ Ex. TURN-12-E, p. 28.

⁶⁰¹ *Id.*, p. 29.

significantly higher cost-effectiveness and the impressive risk reduction that SCE has already achieved with this mitigation.⁶⁰²

TURN's proposal does not rely solely on choosing the most cost-effective alternative; it recognizes that SCE may be able to make a case for deploying undergrounding in certain high risk locations even if undergrounding is not the most cost-effective choice.⁶⁰³ In addition, TURN supports extending grid hardening to the entire HFTD despite the fact that relative cost-effectiveness of covered conductor may be quite low for many of these areas, given that the HFTD designation indicates an elevated risk of a utility-caused wildfire.⁶⁰⁴ These features of TURN's proposal show that SCE incorrectly portrays TURN's recommendation as determined solely by RSEs and relative risk.⁶⁰⁵

TURN's proposal results in the *same* absolute risk reduction as SCE, as shown in the following figure, and a significantly more cost-effective grid hardening proposal overall.

⁶⁰² Id.

⁶⁰³ *Id.*, p. 28.

⁶⁰⁴ *Id.*, p. 15.

⁶⁰⁵ *E.g.*, Ex. SCE-15, Vol. 5, Part 2, p. 10, incorrectly claiming that TURN advocates "rigid adherence to model outputs."



Figure 12: Absolute Risk Reduction of Grid Hardening Proposals – TURN vs. SCE⁶⁰⁶

TURN's proposal achieves the same absolute risk reduction with many fewer miles because it targets the circuits with the very highest S-MAP risk scores for undergrounding, which SCE's SRA-based proposal does not.⁶⁰⁷

Overall, TURN's grid hardening proposal is 81 percent more cost-effective than SCE's – with an RSE of 1,603, compared to 888 for SCE.⁶⁰⁸ Furthermore, TURN's proposal costs approximately \$2 billion less than SCE's over the rate case period for the *same* absolute risk reduction by prioritizing high-risk miles for undergrounding, as determined by the S-MAP framework.⁶⁰⁹

⁶⁰⁶ Ex. TURN-12-E, p. 29, Figure 16.

⁶⁰⁷ *Id.*, p. 29.

⁶⁰⁸ *Id.*, p. 30.

⁶⁰⁹ Id.

15.2.6.2 SCE Should Justify Its Use of Undergrounding Through a Location-Specific Analysis

TURN's undergrounding recommendation provides a reasonable maximum number of miles where undergrounding can be expected to be the best system hardening alternative. However, even for the highest risk miles targeted by TURN's proposal, whether undergrounding proves to be the best alternative will depend on location-specific factors, such as the risk drivers and terrain present at the location, which affect the mitigation effectiveness of the competing alternatives and the cost.⁶¹⁰ These factors can vary significantly by project. As the Office of Energy Infrastructure Safety (Energy Safety) stated in its decision on SCE's 2023-2025 Wildfire Mitigation Plan (WMP):

Mitigation selection should consider a variety of location-specific factors, such as how long it takes to deploy the solution, effectiveness at mitigating particular ignition drivers in a given location, feasibility given terrain and access challenges, and the cost-benefit analysis.⁶¹¹

Consistent with Energy Safety's direction, SCE should be required to justify through a

location-specific analysis that undergrounding is the best alternative for that location. The

analysis should include consideration of the following:

- The location-specific RSE or Cost-Benefit Ratio (CBR)⁶¹² of undergrounding, based on location-specific costs, risk drivers, and risk reduction;
- The location-specific RSE/CBR of covered conductor, based on location-specific costs, risk drivers, and risk reduction and, where REFCL or spacer cable is feasible, the location-specific RSE/CBR of covered conductor with REFCL or spacer cable, based on the same location-specific factors.

⁶¹⁰ Ex. TURN-12-E, p. 31.

⁶¹¹ Decision on Southern California Edison Company's 2023-2025 Wildfire Mitigation Plan, Appendix A to Resolution SPD-17, p. 40.

⁶¹² SCE is required to transition to calculation of cost-benefit ratios under the Cost Benefit Approach set forth in D.22-12-027 in its next RAMP submission.
- The RSE/CBR analysis for undergrounding and covered conductor should reflect the "time value of risk" given that undergrounding is usually much more time intensive than covered conductor.
- Consideration of any other location-specific factors that would influence which mitigation is the best alternative.⁶¹³

Based on this location-specific analysis, SCE should determine the best system hardening alternative for each location where undergrounding is considered. In most cases, the best alternative will be the one with the highest RSE or cost-benefit ratio, as the RSE reflects a comprehensive analysis of the risk reduction benefits compared to the costs at the location in question.⁶¹⁴ However, if SCE chooses undergrounding even if it not the most cost-effective alternative, it should be required to explain and document the specific location-specific factor(s) that justify choosing undergrounding and why the RSE/CBR does not adequately reflect such factor(s). For example, SCE could explain that a high risk location has a particularly high tree fall-in risk and that the diminished effectiveness of covered conductor in addressing this driver is not sufficiently reflected in the RSE/CBR. For each undergrounding project that SCE elects to pursue, in the annual accountability report described below, it should document the RSE/BCRs it calculated in choosing among alternatives and, if undergrounding did not have the highest value, its explanation justifying its selection.⁶¹⁵

If, based on this analysis, SCE decides to pursue less than 177 overhead miles of undergrounding, overhead hardening should be deployed on the miles where undergrounding proved not to be the best alternative.⁶¹⁶

⁶¹³ Ex. TURN-12-E, pp. 31-32.

⁶¹⁴ See Section 15.2.4.4 above.

⁶¹⁵ *Id.*, p. 32.

⁶¹⁶ *Id*.

TURN notes that its proposal in this opening brief differs slightly from TURN's testimony in that TURN would no longer require all undergrounding projects to be in the top 50% of risk. Within the 177-mile cap, TURN would not oppose undergrounding if SCE can demonstrate that a project not ranking in the top 50% of risk warrants this mitigation alternative based on the location-specific analysis. Even though TURN suspects that few projects outside of the 50% threshold will succeed under such an analysis, this modification responds to SCE's claims that TURN's 50% limitation is arbitrary and would prevent SCE from implementing meritorious undergrounding projects.⁶¹⁷

The Commission should make explicit that no more than a total of 177 overhead miles from 2025-2028 (an average of 44 overhead miles of undergrounding per year) will be funded. If any miles are accomplished beyond this amount, they may not be recorded to any account (memorandum, balancing, etc.) nor may SCE request to establish such an account for this purpose.⁶¹⁸ To the extent that SCE performs less than 177 miles of undergrounding and replaces those miles with overhead hardening, at the end of the rate case period, the difference in costs, based on the forecast unit costs adopted in this decision, should be refunded to ratepayers. Therefore, the one-way balancing account that TURN recommends for all grid hardening expenditures⁶¹⁹ should have a separate one-way sub-account for undergrounding expenditures.⁶²⁰

⁶¹⁷ Ex. SCE-15, Vol. 5, Part 2, pp. 20-21.

⁶¹⁸ Ex. TURN-12-E., p. 32. As TURN's expert Mr. Borden explained, "the proliferation of memorandum and balancing accounts for expenditures that are foreseeable has . . . degraded the affordability of electric rates [for] all the IOUs and weakened Commission oversight of expenditures, particularly for wildfire mitigation" (*Id.*) See also Section 38.3 below.

⁶¹⁹ See Section 38.3 below.

⁶²⁰ Ex. TURN-12-E., pp. 32-33.

The scale of undergrounding proposed by TURN is a reduction from SCE's proposal, but still warrants additional scrutiny given the extraordinary cost and scale of this measure, particularly compared with historical deployment. To this end, SCE should be required to submit an annual accountability report, similar to the report PG&E was required to submit in D.23-11-069.⁶²¹ As discussed above, the report should require SCE to provide the results of its location-specific analysis for each undergrounding project it opted to pursue (including incomplete projects) in the preceding year. Consistent with the report PG&E is required to submit, SCE's annual report should also include information on completed undergrounding projects, including costs, unit costs, and overhead to underground conversion ratio information.⁶²²

15.2.6.3 Summary of TURN's Recommendations

In summary, TURN recommends the following:

• The Commission should approve a grid hardening forecast for the 2025-2028 period of 177 overhead miles converted to undergrounding and 1,651 miles insulated with covered conductor, as shown in the table below:

⁶²¹ D.23-11-069, p. 280.

⁶²² D.23-11-069, pp. 280-283. SCE contends (Ex. SCE-15, Vol. 5, Part 2, p. 22) that such a report is not needed because SCE's program is smaller than PG&E's. However, under TURN's proposal, SCE's undergrounding would still cost as much as \$1 billion, and SCE's proposal would cost \$3.3 billion. In either case, the factors that D.23-11-069 (p. 280) identified in support of an accountability report – the uncertainty associated with a large scale-up of undergrounding, the significance of the program as a risk reduction proposal, and the significant ratepayer costs involved – all apply here. SCE also argues (Ex. SCE-15, Vol. 5, Part 2, p. 22) that no accountability report is needed because it has provided documentation of its proposed undergrounding projects in its rebuttal testimony. However, that claimed documentation does not provide a comparison of alternatives to undergrounding, including RSEs/CBRs and does not explain why undergrounding should be deployed even when covered conductor would be more cost-effective. Nor does it provide most of the other information required in D.23-11-069.

	Undergrounding									
	2025	2026	2027	2028	Total / Weighted Average					
TURN Miles	44	44	44	44	177					
SCE Miles	60	150	200	170	580					
Unit Cost	\$ 5,083	\$ 5,677	\$ 5,717	\$ 5,687	\$ 5,632					
TURN Budget	\$ 224,903	\$ 251,227	\$ 252,984	\$ 251,633	\$ 980,746					
SCE Budget	\$ 304,954	\$ 851,620	\$1,143,432	\$ 966,727	\$ 3,266,733					
TURN-SCE	\$ (80,051)	\$ (600,392)	\$ (890,448)	\$ (715,095)	\$ (2,285,986)					
		0	Covered Condu	ctor						
	2025	2026	Covered Condu 2027	2028	Total / Weighted Average					
TURN Miles	2025	2026 413	2027 413	2028 413	Total / Weighted Average 1,651					
TURN Miles SCE Miles	2025 413 850	2026 413 300	2027 413 50	ctor 2028 413 50	Total / Weighted Average 1,651 1,250					
TURN Miles SCE Miles <u>Unit Cost</u>	2025 413 850 \$ 763	2026 413 300 \$ 778	2027 413 50 \$ 805	2028 2028 413 50 \$ 812	Total / Weighted Average 1,651 1,250 \$ 770					
TURN Miles SCE Miles Unit Cost	2025 413 \$50 \$763	2026 413 300 \$ 778	2027 2027 413 50 \$ 805	2028 2028 413 50 \$ 812	Total / Weighted Average 1,651 1,250 \$ 770					
TURN Miles SCE Miles Unit Cost TURN Budget	2025 413 \$ 763 \$ 314,921	2026 413 300 \$ 778 \$ 320,902	2027 2027 413 50 \$ 805 \$ 332,373	ctor 2028 413 50 \$ 812 \$ 335,247	Total / Weighted Average 1,651 1,250 \$ 770 \$ 1,303,442					
TURN Miles SCE Miles Unit Cost TURN Budget SCE Budget	2025 413 850 \$ 763 \$ 314,921 \$ 648,666	2026 413 300 \$ 778 \$ 320,902 \$ 233,289	2027 2027 413 50 \$ 805 \$ 332,373 \$ 40,271	ctor 2028 413 50 \$ 812 \$ 335,247 \$ 40,620	Total / Weighted Average 1,651 1,250 \$ 770 \$ 1,303,442 \$ 962,845					

 Table 8: Mileage and Costs of Grid Hardening (\$ thousands) – TURN vs. SCE

- For all 2025-2028 undergrounding projects, SCE should conduct the location-specific analysis described in Section 15.2.6.2 and should only implement projects where the analysis shows that undergrounding is the best alternative for that location.
- Ratepayers shall not be required to fund more than 177 overhead miles in 2025-2028. If SCE cannot justify undergrounding of 177 miles under the location-specific analysis and therefore undergrounds fewer miles, overhead hardening should be deployed on those miles. To the extent that SCE performs less than 177 miles of undergrounding and replaces those miles with overhead hardening, at the end of the rate case period, the difference in costs, based on the forecast unit costs adopted in this decision, should be refunded to ratepayers via the one-way balancing account that TURN recommends in Section 38.3 of this brief.
- The one-way balancing account that TURN recommends in Section 38.3 of this brief should have a separate one-way subaccount for undergrounding expenditures.
- The Commission should require SCE to submit an annual accountability report, similar to the report required in D.23-11-069. The report should require SCE to provide the results of its location-specific analysis for each undergrounding project it opted to pursue (including incomplete projects) in the preceding year. Consistent with the report required in D.23-11-069, SCE's annual report should also include

information on completed undergrounding projects, including costs, unit costs, and overhead to underground conversion ratio information.

15.2.7 Response to SCE's Rebuttal Testimony

This section responds to arguments presented in SCE's rebuttal testimony that were not addressed in the preceding sections. The need for this new section arises in large part due to the fact that SCE presented new analysis in its rebuttal testimony that was not part of SCE's direct showing.

15.2.7.1 The Results of SCE's New and Opaque 'BCR Analysis' Are Deeply Flawed and Unreliable

SCE's rebuttal testimony claims that its undergrounding proposal is supported by the results of a new, supposedly project-specific "Targeted Undergrounding BCR Analysis" that it presented for the first time in rebuttal testimony.⁶²³ SCE asserts that this analysis shows that, for more than 50% of its proposed undergrounding projects, undergrounding has a higher BCR than an alternative it calls "REFCL/CC ++"⁶²⁴ -- *i.e.*, covered conductor teamed with REFCL and certain other mitigations.

This result is based on unsupportable assumptions and collapses under even minimal scrutiny. *Even though SCE has successfully deployed covered conductor <u>without REFCL</u> in <i>what it perceived to be its highest risk areas from 2018 through 2023*, SCE's calculations include the assumption that the only variant of covered conductor that should be compared with undergrounding is covered conductor teamed with REFCL. SCE unreasonably assumed that REFCL would be included in every covered conductor project even when SCE estimated the

⁶²³ The output of the analysis is presented in Ex. SCE-15, Vol. 5, Part 2, App. B, pp. B1-B16.
⁶²⁴ Ex. SCE-15, Vol. 5, Part 2, p. 30.

costs to be infeasibly high.⁶²⁵ The REFCL costs that SCE included ranged from 0 to \$1.3 *billion* per overhead mile, a weighted average of \$5 *million* per mile, compared to SCE's unit cost for its proposed deployment of REFCL in this case of about \$100,000 per mile.⁶²⁶ By making the covered conductor alternative unreasonably costly for many projects, SCE skewed the cost-effectiveness comparison in favor of undergrounding.⁶²⁷

When TURN corrected just this assumption (i.e., leaving the rest of SCE's analysis unchanged) by limiting REFCL to projects where REFCL's unit cost was less than \$200,000 per mile (twice SCE's unit cost forecast in this GRC), covered conductor or covered conductor with REFCL had a higher BCR than undergrounding for 97.5 percent of the miles in SCE's undergrounding proposal – a completely different result from what SCE presented and consistent with TURN's findings in its opening testimony.⁶²⁸

Cross-examination revealed another significant flaw – the failure to incorporate SCE's well-established Fast Curve mitigation into the analysis. Fast Curve settings reduce fault energy by increasing the speed by which a relay reacts to faults;⁶²⁹ they have been widely installed in SCE's HFRA circuits.⁶³⁰ Fast Curve has a standalone mitigation effectiveness of 40 to 50 percent, is compatible with covered conductor, and does not need REFCL to operate.⁶³¹ Despite

⁶²⁵ Ex. TURN-20, pp. 1, 4-6.

⁶²⁶ Ex. TURN-20, pp. 4-6.

⁶²⁷ *Id.*, p. 5.

⁶²⁸ *Id.*, pp. 6-8. As discussed below, by presenting its complex new analysis for the first time in rebuttal testimony, SCE left TURN little time to scrutinize SCE's workpapers and make other necessary corrections. *Id.*, pp. 1, 3-4.

⁶²⁹ Ex. SCE-04, Vol. 5, Part 2A, p. 86.

⁶³⁰ Tr. Vol. 10, p. 923:9 – 924:5. (Fugere/SCE).

⁶³¹ *Id.*, p. 924:16 – 925:7.

Fast Curve's obvious usefulness as a way to significantly increase the effectiveness of covered conductor, SCE left Fast Curve out of the analysis entirely.⁶³² As a result, the analysis did not include a covered conductor plus Fast Curve option for the high percentage of locations where SCE's calculation of REFCL costs made REFCL infeasible. Had Fast Curve been included, the mitigation effectiveness of the covered conductor alternative would have been higher and TURN's alternative calculations would likely have shown the covered conductor alternative to be more cost effective than undergrounding for even more than 97.5 percent of miles.

Even though TURN had very limited time to examine SCE's complex new quantitative analysis, TURN identified numerous other problems that make the results entirely unreliable, including:

- SCE's rebuttal analysis used more granular cost estimates for covered conductor and REFCL, but for undergrounding SCE used just used an across-the-board average cost⁶³³ even though SCE's direct testimony stated that undergrounding costs can vary "significantly" based on factors such as "population density, topography, permitting and environmental clearances, paving and labor."⁶³⁴ SCE's failure to do an apples-to-apples comparison of project costs – when this was ostensibly the improvement it tried to make to its S-MAP analysis -- further compromises the reliability of its results.
- SCE's workpapers do not indicate whether the analysis incorporates the time benefit of covered conductor compared to the longer time required for undergrounding deployment.⁶³⁵
- SCE's workpapers do not provide calculations showing how SCE converted risk reduction values to dollars,⁶³⁶ including whether a reasonable value of statistical life (VSL) was used. SPD's RAMP Evaluation Report and TURN's RAMP

⁶³² *Id.*, p. 981:17-19. This omission is particularly glaring because SCE's direct testimony defined REFCL/CC ++, the alternative that was supposedly being compared with undergrounding, to include Fast Curve. Ex. SCE-04, Vol. 5, Part 1A, p. 46.

⁶³³ Ex. TURN-20., p. 3.

⁶³⁴ Ex. SCE-04, Vol. 5, Part 2A, p. 19.

⁶³⁵ Ex. TURN-20, p. 3.

⁶³⁶ Ex. TURN-20, p. 3.

comments were highly critical of the implied SVL in SCE's S-MAP analysis, which SPD found to be unfavorably high compared to accepted government figures.⁶³⁷ In future BCR analyses, utilities are required to use the US Department of Transportation VSL as the standard value.⁶³⁸

• SCE's analysis divided total REFCL costs only by SRA miles that would benefit from a given REFCL project, rather than dividing them by all miles that would benefit. The result was to unreasonably increase the unit costs of REFCL while not accounting for additional benefits of the project. This inaccurately added to the cost of the covered conductor alternative, once again skewing the analysis in favor or undergrounding.⁶³⁹

Notably, SCE did not use this rebuttal analysis or any similar comparative project-

specific analysis as part of its process for deciding the scope of its undergrounding proposal. In

fact, SCE admitted that its proposal is not based on any project-specific analysis examining

whether undergrounding was the most cost-effective alternative.⁶⁴⁰ Thus, the Commission

should give no weight to SCE's rebuttal analysis, not only because of its dubious assumptions

and results, but also because it is irrelevant to how SCE arrived at its undergrounding proposal.

SCE's rebuttal also cites another conclusion from its Undergrounding BCR Analysis, that

claims that benefits exceed costs for a majority of its proposed projects and 447 out of 580

proposed miles.⁶⁴¹ However, as discussed above, SCE's supposed project-specific analysis of

⁶³⁷ SPD Staff Evaluation Report on SCE's 2022 RAMP, A.22-05-013, Nov. 10, 2022, pp. 18-19, made a part of the record of this case by ALJ ruling. Tr. Vol. 10, p, 996:20 – p. 997:3 (hereinafter "SPD RAMP Evaluation). *See* Section 5.3.3 above.

⁶³⁸ D.22-12-027, p. 35.

⁶³⁹ Ex. TURN- 803, SCE response to TURN DR 128, Question 3(c)(i), which confirmed TURN's suspicion described in Ex. TURN-20, p. 5 and fn. 18.

⁶⁴⁰ Ex. TURN-12-E, p. 6 and fn. 6.

⁶⁴¹ Ex. SCE-15, Vol. 5, Part 2, p. 18. SCE (*id.*, pp. 17-18) also claims that the benefit-cost ratios (BCR) presented in Ex. TURN-4 show that SCE's undergrounding proposal would provide more benefits than costs. However, SCE ignores the cautions stated in TURN's testimony about how to use the BCR figures, including examining the particular calculation methodology to determine whether benefits are accurately estimated. (Ex. TURN-4, p. 15). As both SPD and TURN pointed out in the RAMP, SCE's S-MAP calculations exaggerate risk reduction benefits because they assume an unreasonably high value of a statistical life (SPD RAMP Evaluation, pp, 18-19 and Att. 3 thereto (TURN's comments), pp. 10-14), a

the benefits and costs of its undergrounding proposal is unreliable because SCE did not use project-specific undergrounding costs and likely used inflated values to measure the safety benefits of undergrounding, thereby exaggerating its benefits. Furthermore, since covered conductor is almost always *more* cost-effective than undergrounding, covered conductor also has benefits that exceed costs, and to a much greater degree than undergrounding.

Therefore, nothing in the record disputes that covered conductor is a highly effective mitigation and overwhelmingly more cost-effective than undergrounding. (See Section 15.2.2). Given this choice between two desirable alternatives, SCE should not burden ratepayers with a less cost-effective option unless SCE has provided a good reason for doing so based on location-specific considerations, an outcome that is only achieved under TURN's recommendations.

15.2.7.2 SCE Wrongly Argues that TURN's Proposal Deprives SCE of the Ability to Apply Subject Matter Expertise and Judgment

SCE asserts that TURN's recommendations would require SCE to "rely solely on the outputs of quantitative risk modeling data instead of also applying judgment, local knowledge, and subject matter expertise."⁶⁴²

SCE presents a blatantly false dichotomy. As TURN showed in Section 15.2.4.4 above, the S-MAP framework requires utilities to use their best subject matter expertise and judgment to

problem that SCE did not correct in its GRC calculations, but that SCE will have to fix in its next RAMP and GRC per D.22-12-027, p. 35. *See also* Section 5.3.3 above. Thus, it is unreasonable to conclude from SCE's RSEs (and CBRs derived from those RSEs) that SCE's undergrounding program is cost effective.

⁶⁴² Ex. SCE-15, Vol. 5, Part 2, p. 10. p. 16. In a similar vein, SCE (*id.*, pp. 10, 16) claims that TURN's recommendation is based solely on RSE. But then SCE contradicts itself and recognizes that TURN's proposal takes other factors into consideration. (*Id.*, pp. 16-17, fn. 36). As discussed in Section 15.2.6, while TURN recommends that relative cost-effectiveness be a key factor in determining which grid hardening alternative is deployed, TURN's proposal would allow SCE to make the case for undergrounding even when it is not the most cost-effective mitigation in a location.

supplement the available data as the utility finds necessary. Thus, it is simply incorrect for SCE to suggest that quantifying that expertise means that judgment is not being used. The important benefit of the S-MAP framework quantifications is to provide a transparent vehicle for the CPUC and parties to evaluate the data *and judgment* that inform the utility's risk assessments. Without such quantifications, SCE's decisions are based on a virtually unreviewable black box exercise of discretion by SCE managers, who work for an investor-owned utility with a strong incentive to build rate base. The quantification effort required by the S-MAP framework are an important way to ensure that utility choices are accountable and consistent with the public interest rather that promoting the shareholders' financial interest.

Moreover, TURN's recommendations would not dictate where SCE can move its wires underground.⁶⁴³ TURN would leave that choice to SCE, within the constraints of the 177-mile cap. As discussed, that cap is based on the output from SCE's S-MAP risk modeling showing that the top 50% of remaining risk on SCE's system is found in 177 miles. Provided that SCE can justify an undergrounding project through the location-specific analysis described in Section 15.2.6.2, SCE would determine where undergrounding is appropriate based on its expertise and judgment (documented in TURN's recommended accountability report), including if warranted in a location outside the top 50% of risk.

⁶⁴³ As noted in Section 15.2.6.2, TURN has modified its proposal in this opening brief to no longer require that <u>all</u> underground miles be in the top 50% of risk.

15.3 Emergent Technology And Inspections And Remediations

15.4 PSPS And Other Wildfire Activities

16. T&D OTHER COSTS AND OTHER OPERATING REVENUE

- 16.1 T&D Other Costs
- 16.2 T&D Other Operating Revenues

17. CUSTOMER SERVICE OPERATIONS

17.1 Billing And Payments

TURN sponsored testimony addressing Billing and Payments in Ex. TURN-10. TURN has since reached a mutually agreeable resolution for the forecasts related to Billing and Payments. TURN's recommendations are reflected in the "Stipulation of TURN, Cal Advocates, and SCE on Billing Services GRC Activity, Credit and Payment Services GRC Activity, and Billing and Payments Capital," which has been admitted into evidence as Ex. SCE-25. TURN urges the Commission to adopt all of the individual recommendations included in Ex. SCE-25, recognizing that they constitute an integrated agreement supported by TURN, Cal Advocates, and SCE in its entirety.

17.2 Customer Contacts

TURN sponsored testimony addressing Customer Contacts in Ex. TURN-10. TURN has since reached a mutually agreeable resolution for the forecasts related to Customer Contacts. TURN's recommendations are reflected in the "Stipulation of TURN, Cal Advocates, and SCE on Customer Contacts BPE," which has been admitted into evidence as Ex. SCE-29. TURN urges the Commission to adopt all of the individual recommendations included in Ex. SCE-29, recognizing that they constitute an integrated agreement supported by TURN, Cal Advocates,

212

and SCE in its entirety.

- 17.3 Customer Service Re-Platform
- 17.4 Customer Service-Related Other Operating Revenues
- 17.5 Billing Practices And Policies

18. BUSINESS CUSTOMER SERVICES

18.1 Business Customer Services

TURN sponsored testimony addressing Business Customer Services in Ex. TURN-10. TURN has since reached a mutually agreeable resolution for the forecasts related to Business Customer Services. TURN's recommendations are reflected in the "Stipulation of TURN, Cal Advocates, Walmart, and SCE on Business Customer Services BPE and Communication, Education, and Outreach BPE," which has been admitted into evidence as Ex. SCE-26. TURN urges the Commission to adopt all of the individual recommendations included in Ex. SCE-26, recognizing that they constitute an integrated agreement supported by TURN, Cal Advocates, Walmart, and SCE in its entirety.

18.2 Communications, Education, And Outreach

19. CUSTOMER PROGRAMS AND SERVICE

19.1 Customer Experience Management

19.2 Customer Programs Management

TURN sponsored testimony addressing Customer Programs Management in Ex. TURN-

10. TURN has since reached a mutually agreeable resolution for the forecasts related to Customer Programs Management. TURN's recommendations are reflected in the "Stipulation of TURN, Cal Advocates, and SCE on Customer Programs Management GRC Activity," which has been admitted into evidence as Ex. SCE-28. TURN urges the Commission to adopt all of the individual recommendations included in Ex. SCE-28, recognizing that they constitute an integrated agreement supported by TURN, Cal Advocates, and SCE in its entirety.

20. BUSINESS CONTINUATION

20.1 Planning, Continuity, And Governance

20.2 All Hazards Assessment, Mitigation and Analytics -- Seismic Resiliency for Non-Electric Facilities

The Commission should adopt TURN's forecast for SCE's seismic retrofitting activities at non-electric facilities.

SCE's "All Hazard Assessment, Mitigation and Analytics" capital expenditures forecast includes the Seismic Resiliency Program that covers, among other things, the seismic retrofits for the utility's "non-electric facilities," primarily offices and operational buildings supporting power delivery.⁶⁴⁴ SCE's forecasts for the Test Year 2025 GRC period include assessment activities that the utility contends are based on historical costs incurred between 2016 and 2021,⁶⁴⁵ and the mitigation activities at specific facilities. For the latter category, SCE states that its "project costs are based on a per square foot unit estimate provided by a third party engineering firm considering the requisite materials, construction and supporting activities by building type," with the resulting estimate then applied to the planned mitigation projects at non-

⁶⁴⁴ Ex. SCE-04, Vol. 1, pp. 21-23 and 46.

⁶⁴⁵ *Id.*, p. 40.

electric facilities from 2023-2028.⁶⁴⁶ On this basis, SCE forecasts \$23.286 million for 2023, \$28,400 million for 2024, and \$31.400 million for 2025 for the costs of assessing and retrofitting non-electric facilities.⁶⁴⁷

TURN's testimony recommended that the Commission determine that SCE has failed to demonstrate the reasonableness of the project-specific forecasts that underlie the utility's overall forecast.⁶⁴⁸ Though SCE cites several factors that it claims fed into each project's forecast, and its overall forecast purports to be a sum of numerous project-specific forecasts, its testimony and workpapers did not include any material explaining the basis for each of the project-specific forecasts. Instead, SCE relies heavily on a National Institute of Standards and Technology (NIST) model that it claims permitted it to develop a "better" cost estimate for each model based on its determination that "average seismic retrofit costs can be expected to be \$147.00 per square foot."⁶⁴⁹

There are numerous problems with SCE's purported reliance on the NIST model as set forth in the "working paper" SCE provided as the basis for the model's application here. For example, SCE's direct testimony described the NIST model as supporting the assumption that "average seismic retrofit costs can be expected to be \$147.00 per square foot."⁶⁵⁰ But the working paper that sets forth the NIST model results describes two different average cost figures that would apply given SCE's portfolio of buildings: \$91 per square foot to retrofit them to the

215

⁶⁴⁶ Id.

⁶⁴⁷ *Id.*, p. 46, Table II-8.

⁶⁴⁸ Ex. TURN-15-E2, p. 32.

⁶⁴⁹ Ex. SCE-04, Vol. 1, p. 30.

⁶⁵⁰ Ex. SCE-04, Vol. 1, p. 50.

"life safety" standard, and \$147 per square foot to retrofit them to the "immediate occupancy" standard.⁶⁵¹ Most of SCE's seismic retrofit projects performed in recent years and forecasted for the TY 2025 GRC period are being done to the lower cost "life safety" performance objective.⁶⁵²

The working paper's rather candid assessment of its own limitations is also an indication that it is not an appropriate basis for adopting SCE's forecast here. SCE's consultant tested the NIST model against three SCE retrofit projects, and described its findings as follows:

The estimates were reasonably close: the equation estimated all three square-foot costs (at least as far as SCE currently knows them) within \pm 34%. That modest margin is probably partly due to luck, but provides some validation anyway. SCE could realistically use the NIST model ... for preliminary budgeting purposes.⁶⁵³

Later in the working paper, SCE's consultant raised additional "cautions" about the underlying NIST model becoming less reliable as the underlying data grow older (it is pre-1993 data), and stated, "The limited tests we performed of the model's accuracy are <u>somewhat</u> <u>reassuring, but the agreement could be accidental</u>."⁶⁵⁴ Given the consultant's rather tepid endorsement of its own product, the Commission should find that SCE has not demonstrated the reasonableness of using the average square foot results calculated using the NIST model for purposes of developing a forecast for non-electric seismic retrofit costs here.

SCE's testimony states that the utility has completed 28 seismic retrofits of non-electric facilities since its 2021 GRC.⁶⁵⁵ The average recorded cost of these projects is approximately

⁶⁵¹ Ex. SCE-04, Vol 1-WP, p. 76.

⁶⁵² Ex. TURN-15-Atch1, Attachment 14 (SCE Responses to TURN DR 63-1 and 63-2).

⁶⁵³ Ex. SCE-04, Vol. 1-WP, p. 76 (emphasis added)

⁶⁵⁴ Ex. SCE04, Vol. 1-WP, p. 78 (emphasis added).

⁶⁵⁵ Ex. SCE-04, Vol. 1, p. 50.

\$57 per square foot.⁶⁵⁶ TURN recommends that the Commission should rely on the recorded average cost of these completed projects rather than the much higher estimated average cost figures set forth in SCE's testimony and workpapers, or the \$84 per square foot implied by SCE's proposed forecasts.⁶⁵⁷ TURN's testimony explains how the use of the \$57 per square foot figure results in the forecasts below for 2023, 2024 and 2025, figures that are approximately one-third lower than SCE's forecasts.⁶⁵⁸

\$,000	2023 forecast	2024 forecast	2025 forecast
SCE Forecast	\$23,296	\$28,400	\$31,400
TURN Forecast	\$15,966	\$19,549	\$21,587
Difference	\$7,330	\$8,851	\$9,813

SCE's recorded 2023 costs for the seismic resiliency program for its non-electric

facilities amounted to \$15.222 million, a figure much closer to the forecast developed under

TURN's method (\$15.966 million) than to SCE's 2023 forecast (\$23.296 million).⁶⁵⁹

SCE's rebuttal presented no substantive or well-founded challenge to TURN's

recommendation. The utility doubled-down on its reliance on the NIST model and SPA Risk

⁶⁵⁷ Id.

⁶⁵⁶ Ex. TURN-15-E2, p. 33.

⁶⁵⁸ *Id.* For these activities, TURN recommends that the authorized amounts for purposes of establishing the post-test year revenue requirements be determined consistent with TURN's recommendations for post-test year ratemaking as set forth in Ex. TURN-17 (Testimony of Cathy Yap) and Section 41 of this brief, rather than based on project-specific budgets as SCE has proposed.

⁶⁵⁹ Ex. TURN-102, p. 1 (2023 recorded figure); Ex. TURN-15-E2, p. 33 (for TURN 2023 forecast); and Ex. SCE-04, Vol. 1, p. 46, Table II-8 (for SCE 2023 forecast).

LLC's review thereof, but without acknowledging, much less addressing the significant caveats the analyst had raised in its working paper.⁶⁶⁰

SCE also asserted that "TURN's funding levels would negatively affect the safety of our workforce and reliability of service to our customers."⁶⁶¹ The Commission should dismiss such claims as contradicted by SCE's own 2023 spending level. Surely SCE is not suggesting that by spending slightly <u>below</u> TURN's 2023 forecast (and \$8 million below its own 2023 forecast), the utility was negatively affecting safety and reliability.⁶⁶² SCE's claim is also belied by the memorandum account that was adopted in the 2021 GRC, but has remained unused since then.⁶⁶³ SCE cannot seriously expect the Commission to believe that adopting TURN's forecast here (with a unit cost of approximately double that adopted in the 2021 GRC) is a threat to safety or reliability, when SCE's spending in the 2021-2024 period stayed within Commission-authorized amounts despite those amounts being far less than the utility's 2021 GRC forecasts for that period.

Similarly, the Commission should disregard SCE's criticism of TURN for having derived a single average cost per square foot rather than one that reflects differences between "structural" and "non-structural" mitigations.⁶⁶⁴ SCE's direct testimony cited a single cost per square foot,

⁶⁶⁴ Ex. SCE-15, Vol. 1, pp. 11-12.

⁶⁶⁰ Ex. SCE-15, Vol. 1, pp. 13-15.

⁶⁶¹ *Id.*, p. 10.

⁶⁶² Again, SCE forecasted \$23.296 million for 2023, but spent \$15.222 million.

⁶⁶³ In D.21-08-036 the Commission adopted a unit cost of \$28.66 per square foot for seismic retrofits at non-electric facilities, which reduced SCE's forecast by approximately \$10.745 million. The Commission also provided SCE with a new memorandum account to track above-authorized spending should there be any. D.21-08-036, p. 332-333. There were none: In the current GRC, SCE reports that it has recorded no incremental amounts in this memorandum account through the end of 2023, and does not anticipate recording any amounts in 2024. Ex. SCE-18, Vol. 1, p. 45.

claiming "average seismic retrofit costs can be expected to be \$147.00 per square foot."⁶⁶⁵ And though it claimed the utility includes both "structural and non-structural mitigations" in developing its "unique project cost estimations," the project cost estimations provided in the workpapers included no information that would permit the Commission to distinguish between structural and non-structural mitigations. Rather, for the projects specified for 2023 or 2024 work, a single annual amount is provided without any information explaining how the specific amount was derived. And for most of the 2024 forecast and all of the 2025 forecast (other than assessment work), the forecast is a single lump sum not attributable to any particular project.⁶⁶⁶ SCE should not be heard to criticize TURN for a lack of project-level specificity in the development of its forecast when TURN's forecast is not materially different in this regard than the material SCE included in its direct showing.

⁶⁶⁵ Ex. SCE-04, Vol. 1, p. 50.

⁶⁶⁶ Ex. SCE-04, Vol. 1-WP, p. 63.

21. EMERGENCY MANAGEMENT

- 21.1 Training, Drills And Exercises
- 21.2 Emergency Preparedness And Response
- 21.3 Storm Response

22. CYBERSECURITY

- 22.1 Cybersecurity Delivery
- 22.2 Grid Modernization Cybersecurity
- 22.3 Software License & Maintenance

23. PHYSICAL SECURITY

24. **GENERATION**

24.1 Hydro

24.1.1 Hydro Capital

TURN's testimony recommended a series of adjustments to SCE's proposed hydro capital expenditure forecast. These adjustments include moving the assumed date of issuance for FERC licenses to later years in order to reflect SCE's revised expectations, removing capital additions from the Results of Operations model for projects that are delayed, harmonizing start dates for Big Creek area recreation projects with a 2007 Settlement Agreement, assuming a two year delay in capital expenditures relating to the Borel project, and adjusting the cost of the Big Creek 4 Unit 1 generator rewind to reflect a lower escalation in materials costs.⁶⁶⁷ In addition, TURN's testimony recommended that ratepayer responsibility for San Gorgonio decommissioning costs should be reduced by at least \$10 million since ratepayers have

⁶⁶⁷ Ex. TURN-13-E, pp.23-25.

repeatedly paid for forecasted activities over the last five rate cases that were approved but not undertaken.⁶⁶⁸

SCE's rebuttal testimony made a series of concessions to TURN's positions with respect to delays in FERC licensing and decommissioning activities and Big Creek generator rewind costs.⁶⁶⁹ These concessions result in a forecast for 2024-2028 that is identical to TURN. The following table provides a comparison of the original and revised capital expenditure forecasts provided by TURN and SCE:⁶⁷⁰

HYDRO CAPITAL EXPENDITURES									
2023 2024 2025 2026 2027 2028 TOTAI									
SCE initial application	78,701	79,536	67,191	75,119	77,279	67,363	471,520		
SCE rebuttal + errata to 2023 recorded	52,051	94,103	83,503	73,785	43,254	59,382	406,077		
TURN direct/errata testimony	36,775	94,103	83,503	73,785	43,254	59,383	390,803		

Since the revised forecast by SCE and TURN for 2024-2028 is identical, the Commission need not resolve any of the underlying factual disputes related to these forecasts presented in testimony.

The remaining dispute relates to the treatment of capital expenditures at San Gorgonio. TURN's testimony noted that SCE's forecast of decommissioning costs for San Gorgonio increased from \$6.6 million (in the last GRC) to \$41.212 million (in the current GRC).⁶⁷¹ One driver of the increase was damage to a significant portion of the project in the Apple Fire of 2020, which occurred in the middle of litigating SCE's last GRC but was not brought to the Commission's attention until that case was completed.⁶⁷² SCE first began asking for cost recovery for San Gorgonio decommissioning in the 2009 GRC, received authorization to charge

⁶⁶⁸ Ex. TURN-13-E, p.25.

⁶⁶⁹ Ex. SCE-16, pp.39, 43-44.

⁶⁷⁰ Ex. TURN-13-E, p. 6, Table 1; Ex. SCE-16, p.36; Ex. SCE-40, p.A43.

⁶⁷¹ Ex. TURN-13-E, p.45.

⁶⁷² Ex. TURN-13-E, p.46.

customers for these costs in five successive GRCs, and collected approximately \$10.6 million from ratepayers over this period for work that was not performed.⁶⁷³

To address the systematic overcollection of costs from customers since 2009 that were not actually incurred, TURN recommends that the Commission reduce overall ratepayer cost responsibility for San Gorgonio decommissioning by \$10 million.⁶⁷⁴ This disallowance would not penalize SCE for poor performance but instead enforces a modicum of ratepayer fairness given the long history of forecasted costs being collected in rates, not spent on decommissioning, and instead retained by SCE's shareholders. If the Commission adopts TURN's recommendation, SCE would still be able to collect funding in this GRC to support actual anticipated decommissioning activities to be performed over the current cycle. But SCE would be forced to reimburse ratepayers for 15 years of overcollections that exclusively benefited its shareholders. The Commission should not allow SCE, or any utility, to repeatedly collect funds from its customers for projects that are repeatedly delayed and do not occur. The \$10 million disallowance would represent an equitable outcome given the historic benefits to shareholders and the costs already imposed on ratepayers.

24.1.2 Hydro O&M

SCE's application and direct testimony proposed an O&M revenue requirement of \$53.475 million in the 2025 test year.⁶⁷⁵ TURN's testimony recommended two separate adjustments to the SCE forecast. First, TURN proposed using data from 2016-2020 (instead of 2018-2020) to develop base year hydro non-labor O&M.⁶⁷⁶ This recommendation reduces test

⁶⁷³ Ex. TURN-13-E, p.46; D.21-08-036, p.353.

⁶⁷⁴ Ex. TURN-13-E, p.47.

⁶⁷⁵ Ex. SCE-05v1, p.48.

⁶⁷⁶ Ex. TURN-13-E, p.21

year O&M by \$0.622 million. Second, TURN recommended reducing test year O&M by an additional \$0.289 to reflect delays in new federal licenses for the Big Creek and Kaweah facilities.⁶⁷⁷ TURN's total adjustments proposed to Hydro O&M are \$0.911 million (\$2022).⁶⁷⁸

TURN's first recommendation is to use a 5-year historic period from 2016-2020 as the basis for SCE's non-labor O&M forecast rather than the 2018-2020 period relied upon by SCE. TURN witness Monsen explained that costs between 2018-2020 reflect the highest annual values when compared to a longer historical lookback (2014-2022) and therefore do not represent a reasonable starting point.⁶⁷⁹ As an alternative, TURN recommends a 5-year historic period that runs from 2016 through 2020 that yields a \$0.622 million reduction compared to SCE's proposal.⁶⁸⁰

In rebuttal testimony, SCE argues that the 2018-2020 historical timeframe is appropriate given storm events in 2021-2022 and asserts that any reliance on 2016-2017 recorded data for purposes of assessing the reasonableness of the test year forecast is "outside the scope of this GRC" and would require SCE to spend significant time to "fully analyze and conduct discovery on the data."⁶⁸¹ In support of this claim, SCE cites prior Commission decisions that authorize the GRC rate case plan.⁶⁸² None of these decisions have any bearing on the ability of an intervenor to propose that the Commission rely on a different historical time period to develop a base year forecast. Moreover, SCE had a fair opportunity to conduct discovery on TURN's recommendation but failed to do so.

⁶⁷⁷ Ex. TURN-13-E, pp.25-26.

⁶⁷⁸ Ex. TURN-13-E, p.6, Table 2.

⁶⁷⁹ Ex. TURN-13-E, p.22.

⁶⁸⁰ Ex. TURN-13-E, p.23.

⁶⁸¹ Ex. SCE-16, pp.23-24.

⁶⁸² Ex. SCE-16, p.24, footnote 69.

SCE further argues that the use of 2016 and 2017 data would be inappropriate because these years involved water runoff that "is lower than historical averages", resulted in less operation, and decreased the need for equipment repairs and replacement parts.⁶⁸³ Data provided by SCE in response to TURN discovery does not support this contention. The following table shows annual hydro generation between 2016-2022:⁶⁸⁴

SCE hydro generation (MWh) per year					
	Annual generation (MWh)				
2016	3,711,821,650				
2017	5,806,449,076				
2018	3,503,918,894				
2019	4,308,160,247				
2020	2,159,999,863				
2021	1,727,229,059				
2022	2,565,762,550				
TURN period average					
(2016-2020)	3,898,069,946				
SCE period average					
(2018-2020)	3,324,026,335				
Average (2016-2022)	3,397,620,191				
Average (2016-2017)	4,759,135,363				

As shown in this table, actual generation from the SCE hydro system over the period selected by TURN (2016-2020) is higher than during the period selected by SCE (2018-2020). Moreover, average generation during the 2016-2017 timeframe is greater than the long-term average (2016-2022), the average during SCE's proposed timeframe (2018-2020) and the longer time frame proposed by TURN (2016-2020). This data categorically disproves SCE's claim that

⁶⁸³ Ex. SCE-16, p.24.

⁶⁸⁴ Ex. TURN-709, SCE response to TURN Data Request 117, Q1.

its hydro facilities operated less during 2016 and 2017 than in subsequent years. The Commission should therefore reject SCE's unfounded critique and adopt TURN's recommended use of the 2016-2020 period for determining base year O&M.

TURN's second recommended adjustment reduces the adders SCE applies to its base labor O&M forecast for work relating to Big Creek and Kaweah, facilities that SCE assumed would be receiving new licenses from the Federal Energy Regulatory Commission (FERC) in 2023. As shown in TURN's testimony, SCE assumed that there would be additional O&M in 2025 to address relicensing conditions adopted in 2023.⁶⁸⁵ In response to discovery, SCE acknowledged delays that are expected to lead to the Big Creek license being issued in 2024 and the Kaweah license issued in 2025.⁶⁸⁶ SCE's response specifically states that the Kaweah license issuance date is 2025 and that "FERC is still completing their National Environmental Policy Act (NEPA) process and the License is expected within one year following the completion of the NEPA process."⁶⁸⁷

In rebuttal, SCE agrees that the Big Creek license is likely to be delayed to 2024 and makes a \$0.152 million reduction to its 2025 forecast of new license implementation activities.⁶⁸⁸ SCE contests TURN's proposal to reduce the 2025 forecast by an additional \$0.136 million based on two critiques. First, SCE asserts that its original testimony and workpapers forecasted that the Kaweah license will be issued in 2024 and there is "no reason to believe" the date will slip.⁶⁸⁹ This claim is flatly contradicted by SCE's response to TURN data requests acknowledging a likely 2025 license renewal date, a fact that SCE ignores and does not even

⁶⁸⁵ Ex. TURN-13-E, p.52.

⁶⁸⁶ Ex. TURN-13-E, p.28; Ex. TURN-13-Atch1, SCE response to TURN Data Request 52, Q2a.

 ⁶⁸⁷ Ex. TURN-13-Atch1, SCE response to TURN Data Request 52, Q2a, Attachment (Note 2).
 ⁶⁸⁸ Ex. SCE-16, p.25.

⁶⁸⁹ Ex. SCE-16, p.25, *citing* SCE's original workpapers.

address in its rebuttal.⁶⁹⁰ Second, SCE argues that some of the costs identified by TURN will not be affected by the delay in issuing new licenses for either Kaweah or Big Creek.⁶⁹¹ However, SCE's own workpapers show that some of these costs are attributable to "new license compliance for Kaweah" and that these costs can only be incurred after the license is received. ⁶⁹² This fact undermines the claim that delays in the receipt of a new FERC license will not affect the timing of these costs. The Commission should reject SCE's efforts to limit the downward adjustment to 2025 O&M costs given its acknowledgement regarding the Kaweah delays.

In light of these facts, the Commission should adopt TURN's full set of adjustments to the hydro O&M expense forecast for 2025.

24.2 Fossil Fuel (Including Mountainview and Peakers)

24.2.1 Mountainview Capital

TURN's testimony identified several capital projects proposed for the Mountainview generation plant that were not adequately supported by SCE in its application. TURN recommends adjusting the capital expenditure forecast through 2028 to remove three capital projects, reduce the recoverable cost for the Inlet Flow Distribution Grids by 25%, and recover costs for the Turbine Generator Improvement Program in a one-way balancing account with excess costs tracked in a memorandum account. TURN's proposals result in a net reduction, relative to SCE's forecast, for 2023-2028 of \$17.692 million.

A summary of TURN's adjustments is shown in the following table:⁶⁹³

 ⁶⁹⁰ Ex. TURN-13-Atch1, SCE response to TURN Data Request 52, Q2a, Attachment (Note 2).
 ⁶⁹¹ Ex. SCE-16, p.25.

⁶⁹² Ex. TURN-13-E, p.52, Figure 9 and p.54, Table 24 (See "License Compliance – WO Hydro (Recurring Studies, New License Compliance for Kaweah").

⁶⁹³ Ex. TURN-13-E, p.9, Table 3. The original table was modified to include recorded 2023 capital expenditures in place of SCE's forecast for that year. SCE provided recorded 2023 data for Mountainview in Ex. SCE-16, p.54.

Mountainview Capital Expenditures - TURN adjustments to SCE forecast								
	2023	2024	2025	2026	2027	2028	Total	
SCE Total Mountainview Capex	10,998	7,644	18,295	19,151	17,811	7,976	81,875	
less TURN Adjustments								
Disallow CO Catalyst Replacement	-	-	-	-	-	(1,900)	(1,900)	
Require Memo Account for Turbine/Generator Improvement Program	-	-	-	-	-	-	-	
25% Reduction in Costs for HRSG Inlet Flow Distribution Grids	-	(100)	(850)	(650)	-	-	(1,601)	
Disallow Turbine Control and BCS Project	-	-	-	-	(3,600)	(4,272)	(7,872)	
Disallow GR Variable Load Path Project	-	-	-	(800)	(5,519)	-	(6,319)	
Subtotal Impacts of TURN Recommendations	-	(100)	(850)	(1,450)	(9,119)	(6,172)	(17,692)	
SCE Total Mountainview Capex after TURN Adjustments 10,998 7,544 17,444 17,700 8,692 1,804 64,1					64,184			

The adjustments proposed by TURN are described in the following sections.

24.2.1.1 CO Catalyst Bed

SCE proposes to replace the Carbon Monoxide (CO) catalyst bed on Unit 3A and 3B at Mountainview in 2028. The catalyst is used to CO control emissions at the plant. SCE forecasts the costs of this project at \$1.9 million.⁶⁹⁴ While TURN agrees that Mountainview Unit 3 should meet all applicable emissions limits, SCE has not demonstrated that the catalysts are likely to require replacement within this GRC cycle. Due to this uncertainty, TURN recommends that the Commission decline to authorize this capital project and direct SCE to establish a memorandum account that can be used to track catalyst bed replacement costs if needed.⁶⁹⁵ Any costs tracked in the memorandum account can be submitted for review in the next GRC.

Although Mountainview is currently meeting emissions requirements, SCE justifies the project based on an assessment performed by outside consultants (Environex).⁶⁹⁶ A review of the Environex report does not support the claim that catalyst replacement is necessary, or even likely, in 2028. SCE asserts that replacement in 2028 is appropriate because the report finds that the Catalysts are only expected to "provide the required conversion through the next five

⁶⁹⁴ Ex. SCE-05v1, pp.232-233.

⁶⁹⁵ Ex. TURN-13-E, p.7.

⁶⁹⁶ Ex. SCE-05v1, p.232; Ex. TURN-13-E, p.58; Ex. TURN-13-E-Atch1, SCE response to TURN Data Request 80, Q9.

years".⁶⁹⁷ But the actual report provides life projections of ">5" years for the Catalysts in Units 3A and 3B and notes that projections "over five years have an increasing degree of uncertainty" due to a variety of factors.⁶⁹⁸ A plain reading the report shows that Environex found the Catalysts are expected to perform for <u>at least</u> five years, meaning that the catalysts could continue to operate successfully well beyond that time horizon. Environex further recommended remedial actions to improve catalyst performance and regular testing "to account for any changes in operation and monitor catalyst performance."⁶⁹⁹

During cross-examination, SCE witness Maddox claimed that the catalysts would be replaced in 2028 regardless of whether they are performing well.⁷⁰⁰ This position is not reasonable and could result in the unnecessary replacement of catalysts that are performing within desired specifications. If remedial actions and testing results demonstrate that the catalysts can operate past 2028, SCE should not be encouraged to prematurely replace these components.

TURN's recommendation to remove the catalyst replacement costs from the forecast is designed to prevent against wasteful spending. Recognizing that there is uncertainty, TURN proposes to allow SCE to establish a memorandum account that could be used to track these costs if replacement is necessary and prudent. This approach is reasonable in light of the dual objectives of safe operations and cost-minimization.

228

⁶⁹⁷ Ex. SCE-16, p.57.

⁶⁹⁸ Ex. TURN-13-Atch1, SCE response to TURN Data Request 28, Q. 24.c.iv, Attachment, p.30.

⁶⁹⁹ Ex. TURN-13-Atch1, SCE response to TURN Data Request 28, Q. 24.c.iv, Attachment, p.30. ⁷⁰⁰ Hearing Transcript, May 7, 2024, pp.483-484.

24.2.1.2 Turbine Generator Improvement Program

SCE's forecast includes costs for a major inspection and refurbishment of all four gas turbines at Mountainview, two steam turbines and their associated generators.⁷⁰¹ The cost of this project is estimated to be \$28.296 million and represents the largest capital project included in SCE's Mountainview forecast.⁷⁰² According to SCE witness Maddox, SCE accelerated the duration of this project from four years to slightly over two years after the submission of direct testimony in this proceeding but made no changes to the forecasted costs.⁷⁰³

TURN's testimony recommends establishing a one-way balancing account set at SCE's cost forecast and a memorandum account for tracking costs in excess of any authorized capital expenditures for this project.⁷⁰⁴ If costs are less than SCE's forecast, the overcollection would be returned to ratepayers. Since the work is now slated to occur over a shorter period of time (2+ years vs. original forecast of 4 years), total project costs could be lower than originally forecast. Approving an overly generous forecast would benefit SCE's shareholders through higher collections from ratepayers during the GRC cycle.

TURN's alternative would allow SCE to proceed with the project while ensuring that any cost savings are flowed through to customers during the current GRC cycle. To the extent that SCE's costs exceed the cap in the one-way balancing account, TURN's approach would allow SCE to track the overage and seek review of these costs in the next GRC. The Commission should adopt this treatment to better align the interests of ratepayers and shareholders.

24.2.1.3 HRSG Inlet Flow Distribution Grids

SCE proposes to install new Inlet Flow Distribution Grids in all 4 Heat Recovery Steam Generators (HRSGs) at Mountainview. The cost of this project is forecasted to be \$6.401 million with work occurring between 2024-2026.⁷⁰⁵ Although Mountainview originally had inlet flow distribution grids when it was built, SCE management decided to remove them in 2007 (shortly after the facility was completed) due the failure of key components of the system and chose not re-install new grids based on the mistaken assumption that the plant would operate primarily as a baseload resource with limited ramping.⁷⁰⁶ In 2019, Mountainview began experiencing an increase in HRSG tube failures which SCE decided could be remedied through the re-installation of the inlet flow distribution grids.⁷⁰⁷

TURN recommends that SCE shareholders be assigned 25% of the cost of new inlet flow distribution grids.⁷⁰⁸ This recommendation is based on the fact that SCE intentionally removed the original grids, did not attempt to repair them, and failed to demonstrate that it sought any remedies against the vendor that supplied the defective materials.⁷⁰⁹ Moreover, SCE ignored the ample evidence that state policy was driving the accelerated growth of intermittent renewable energy in a manner that would necessitate additional cycling of thermal generators over time. The Legislature enacted the California Renewable Portfolio Standard in 2002 and accelerated the targets in 2006 to require all retail sellers to achieve a 20% renewable portfolio by 2010.⁷¹⁰ In 2006, the Legislature also enacted the Global Warming Solutions Act which required California to reduce its Greenhouse Gas Emissions to 1990 levels by 2020 and was well understood to have

⁷⁰¹ Ex. SCE-05v1, p.221.

⁷⁰² Ex. TURN-13-E, p.60.

⁷⁰³ Transcript, May 7, 2024, p.487.

⁷⁰⁴ Ex. TURN-13-E, pp.60-61.

⁷⁰⁵ Ex. SCE-05v1, p.220, Table III-46.

⁷⁰⁶ Ex. TURN-13-E, p.61.

⁷⁰⁷ Ex. TURN-705, SCE response to TURN Data Request 117, Q17(e).

⁷⁰⁸ Ex. TURN-13-E, p.63.

⁷⁰⁹ Ex. TURN-13-E, p.62.

⁷¹⁰ SB 1078 (Sher, 2002); SB 107 (Simitian, 2006)

major implications for the power sector.⁷¹¹ Despite the clear evidence of increasing reliance on renewable and low carbon resources in the following years, SCE management chose not to repair or re-install the inlet flow distribution grids. This choice was at odds with the original plant design that included inlet flow distribution grids to provide greater operational flexibility.⁷¹²

SCE's actions fail the reasonable manager test that requires the Commission to evaluate whether utilities acted reasonably based on "facts that are known or should have been known at the time."⁷¹³ The Commission should recognize that SCE did not act prudently in 2007 when it failed to repair the existing system or seek remedies from the vendor. Asking ratepayers to bear 100% of the costs to re-install these components that were previously removed at the discretion of SCE management is not reasonable. TURN's cost sharing proposal would fairly allocate some costs resulting from this avoidable mistake to SCE's shareholders.

24.2.1.4 Turbine Control and BCS Project

SCE forecasts spending \$7.872 million on the Turbine Control System (TCS) and Baseline Security Center (BSC) project in 2027 and 2028.⁷¹⁴ The TCS entails a combination of hardware and control system applications used to manage the operational performance of the GE turbines and auxiliary equipment at Mountainview. The BSC is a set of hardware and applications that are integrated into the Turbine Control System and perform various cybersecurity functionalities to bring in enhanced security measures. SCE argues that the project

⁷¹¹ AB 32 (Nunez, 2006)

⁷¹² Transcript, May 7, 2024, p.491.

⁷¹³ D.90-09-088, pp.15-16.

⁷¹⁴ Ex. SCE-05v1, p.220, Table III-46.

represents a "five-year lifecycle refresh" of the Turbine Distributed Control System Upgrade project included in this GRC and scheduled for 2023.⁷¹⁵

As explained in TURN's testimony, SCE previously requested funds for this project and massively underspent relative to its forecasts. In the 2018 GRC, SCE requested \$1.1 million in 2019 and \$13.3 million in 2020. In the 2021 GRC, SCE requested \$6.0 million in 2021.⁷¹⁶ SCE spent \$47,000 in 2020, \$240,000 in 2021, and \$1.322 million in 2022.⁷¹⁷ In other words, SCE forecasted more than \$20 million in the prior 3 GRCs for this project but spent less than \$2 million. This historical pattern does not support the reasonableness of SCE's forecast in this GRC.

In rebuttal testimony, SCE argues that TURN's critiques regarding past actual spending compared to proposed spending are misplaced because prior requests for capital expenditures were forecasted to occur in attrition years.⁷¹⁸ Whether SCE's proposed expenditures for those projects in prior GRCs were to occur during attrition years or during the Test Years, SCE still proposed the projects and requested Commission approval for the proposed level of expenditures.

Finally, SCE does not "understand" TURN's objections to this project since it is forecast to occur during the attrition years of this proceeding.⁷¹⁹ This objection could apply to any capital project proposed by SCE outside of the test year. If there is no relevance to capital expenditures proposed during the attrition years, it is unclear why SCE is concerned about TURN's

⁷¹⁵ Ex. TURN-13-E, p.64; Ex. TURN-13-Atch1, SCE response to TURN Data Request 28, Q24.b.i (revised). The TCS project is projected to occur in 2023 at a cost of \$5.913 million.

⁷¹⁶ Ex. TURN-13-E, p.64. These requests were not only for the Turbine Control System and BCS Project but for other work as well.

⁷¹⁷ Ex. TURN-13-E, p.64.

⁷¹⁸ Ex. SCE-16, p.63.

⁷¹⁹ Ex. SCE-16, p.63.

objections. Given the unreasonable magnitude of the proposed capital expenditures for this project, and the historical pattern of overforecasting and underspending, TURN believes that it deserves greater scrutiny by the Commission.

For these reasons, TURN recommends that the Commission reject SCE's forecast in this GRC. Ratepayers should not have to repeatedly bear the costs of a project for which SCE has already received funding in past GRCs. However, if the Commission is convinced that this project is reasonable, then TURN would propose establishment of a one-way balancing account with a memorandum account to track overspending to ensure that SCE actually does pursue the project.

24.2.1.5 GE Variable Load Path project

SCE originally proposed to spend \$6.319 million in 2026 and 2027 on the GE Variable Load Path project.⁷²⁰ The project is a control system upgrade intended to improve plant efficiency at lower power levels and improve emissions performance during periods of cyclic operation.⁷²¹ SCE argues that the project is designed to allow Mountainview to "remain competitive" with newer generation power plants with higher efficiency.⁷²² TURN opposed funding for this project because SCE failed to demonstrate cost-effectiveness and has not done any engineering or economic analysis of the project.⁷²³

In rebuttal testimony, SCE noted its decision to remove this project from the capital forecast.⁷²⁴ Under cross examination, SCE witness Maddox stated that SCE is not planning to

⁷²⁰ Ex. SCE-05v1, p.220, Table III-46.

⁷²¹ Ex. TURN-13-E, p.65.

⁷²² Ex. SCE-05v1, p.228.

⁷²³ Ex. TURN-13-Ê, pp.66-67.

⁷²⁴ Ex. SCE-16, p.63.

pursue this project "at the present time."⁷²⁵ TURN and SCE therefore agree that no spending related to this project should be included in the capital forecast.

24.2.2 Peakers Capital

TURN's testimony identified one capital project proposed for SCE's fleet of gas-fired peaker plants that was not adequately supported by SCE in its application. SCE proposes to change out existing microprocessor-based relays at its peaking plants with newer-generation microprocessor-based relays. This project is forecast to cost \$4.0 million with expenditures of \$1 million per year from 2025-2028.⁷²⁶ TURN recommends reducing SCE's forecast by \$1 million in 2025 and \$1 million in 2026.⁷²⁷ The impacts of TURN's recommendations are shown in the following table:⁷²⁸

Peakers Capital Expenditures - TURN adjustments to SCE forecast								
	2023	2024	2025	2026	2027	2028	Total	
SCE Total Peakers Capex	320	502	1,852	1,052	3,572	4,102	11,402	
less TURN Adjustments								
Delay Changeout of Relays at Peakers Until at Least 2027			(1,000)	(1,000)			(2,000)	
Subtotal Impacts of TURN Recommendations for Chapter 5			(1,000)	(1,000)			(2,000)	
SCE Total Peakers Capex after Applying TURN Adjustments	320	502	852	52	3,572	4,102	9,402	

SCE notes that the existing relays will reach their designed service life of 20 years in 2027 and 2028 and argues that replacements should begin before the 20-year life has elapsed regardless of whether there are any failures.⁷²⁹ SCE asserts that these peakers provide needed Black Start capability for the California grid that cannot afford to be jeopardized in the event of a relay failure.⁷³⁰ SCE further claims that the existing relays are not compatible with current laptop

- ⁷²⁷ Ex. TURN-13-E, p.10, Table 4.
- ⁷²⁸ Ex. TURN-13-E, p.10, Table 4.
- ⁷²⁹ Ex. SCE-16, pp.69-70.

⁷²⁵ Transcript, May 7, p.492.

⁷²⁶ Ex. SCE-05v1, p.247, Table III-47.

⁷³⁰ Ex. SCE-16, p.70.

software used by technicians and failure to test the relays every six years would violate relevant compliance requirements.⁷³¹ Finally, SCE references an "economic analysis" that shows an economic benefit of the program.⁷³²

TURN identified a number of flaws and gaps in SCE's claims that raise questions about the need for new capital spending on the relays, particularly during the 2025 test year. First, TURN noted that the relays are designed to operate for 20 years and replacement should not be assumed to occur prior to that date absent a showing of reliability issues with the relays.⁷³³ Second, TURN noted that SCE's technicians were successfully able to use compliant laptops to test the existing relays in 2019, 2022 and 2023. In response to a TURN data request, SCE stated "to date, SCE has not experienced any known communication troubles with the applicable relays at the Peakers and has been able to meet all 6-year compliance testing requirements."⁷³⁴ Furthermore, SCE's proposed relay replacement schedule still assumes that existing relays at two facilities will be tested in 2025 with replacements occurring 1-2 years later.⁷³⁵

TURN further reviewed SCE's "economic analysis" which includes a hard-wired assumption that there is a 90% probability the plant will trip offline due to failure of relay on the date assumed for its replacement with a 3% annual growth in that failure probability.⁷³⁶ TURN's testimony notes that "SCE gives no indication how it came up with these parameters".⁷³⁷ SCE's rebuttal testimony offers no additional insights or information to support the validity of their analysis. In rebuttal testimony, SCE further claimed that relays should be replaced when there is

⁷³¹ Ex. SCE-05v1, p.248.

⁷³² Ex. SCE-05v1, p.249.

⁷³³ Ex. TURN-13-E, p.69.

⁷³⁴ Ex. TURN-13-Atch1, SCE response to TURN Data Request 28, Q25.b.i.

⁷³⁵ Ex. TURN-13-E, p.70, Table 25.

⁷³⁶ Ex. TURN-13-E, p.71; Ex. TURN-705, SCE response to TURN Data Request 117, Q23.

⁷³⁷ Ex. TURN-13-E, p.71.

a 63% probability of failure but SCE witness Maddox stated that this reference in SCE's rebuttal testimony "is a tad misleading" and clarified that this calculation "does not correlate to the need to replace protective relays" and does not relate to anything that could affect the reliable operation of the turbine,⁷³⁸ Additionally, SCE's responses to TURN data requests indicate that repairs are possible "using available parts" as needed.⁷³⁹ SCE witness Maddox explained that "there are spare parts available for some of the relays."⁷⁴⁰

Additionally, SCE's claims about the need for every peaker to continuously serve as a Black Start resource are overblown because they fail to consider the range of generators that provide this capability, do not identify the specific need for these resources on the CAISO system, and provide no analysis to support the claim that even a single peaker being offline for repairs would result in a violation or threaten system reliability. In response to a TURN data request, SCE provided a list of all generation capacity located south of Path 26 and north of SONGS on the CAISO grid that will provide black start capabilities by 2025. This list includes approximately 1,300 MW of Net Qualifying Capacity with Black Start capability in this geographic zone with SCE's entire peaker fleet representing approximately 18% of the total Black Start resource.⁷⁴¹ Each one of SCE's peaker units constitutes less than 4% of total Black

⁷³⁸ Transcript, May 7, pp.496-498. Subsequent to Mr. Maddox's appearance in hearings, SCE asked witness Billapati, who is not directly responsible for peaker operations, to provide an alternative perspective on the failure probability scores. Mr. Billapati claimed that the 63 percent replacement score is a manufacturer projection that "roughly centers around 15 years of operation of the relay" (Transcript, May 13, page 790) SCE did not explained why, if the existing relays are 63% likely to fail after 15 years of operation, they planned to delay replacements until a later date.

⁷³⁹ Ex. TURN-705, SCE response to TURN Data Request 117, Q23.

⁷⁴⁰ Transcript, May 7, p.499.

⁷⁴¹ Ex. TURN-704, SCE response to TURN Data Request 117, Q24.

Start capability in this area. SCE failed to demonstrate that any peaker availability would threaten system reliability or result in insufficient systemwide black start capability.

Finally, SCE was unable to provide basic information about the type, quantity and unit cost of replacement relays and could not even identify "which relays need to be replaced".⁷⁴² It is not clear how SCE was able to develop a capital expenditure forecast without this basic information.

Given these problems with SCE's justifications for the cost and timing of the project, TURN recommends delaying any forecasted capital expenditures until at least 2027 which is the end of the designated service lives for relays at Mira Loma, Grapeland, Barre and Center. This delay would result in a \$1 million reduction to capital expenditures forecast for 2025 and 2026.

24.2.3 Decommissioning Accruals and Contingency

SCE proposes revised decommissioning costs for Mountainview (\$14.036 million) and the Peakers (\$6.02 million) based on two new studies performed for this GRC.⁷⁴³ Both of the decommissioning studies use a 20% contingency. TURN recommends the use of a 15% contingency which would reduce the decommissioning estimates for Mountainview and the Peakers to \$13.167 million and \$6.020 million (2021\$), respectively.

TURN's proposal for a 15% contingency factor recognizes the fact that decommissioning is not expected to occur for many years. By contrast, TURN recommends a 10% contingency factor for the SPVP projects scheduled for decommissioning in 2025 and 2026.

The 20% factor proposed by SCE is inconsistent with the 15% contingency factor approved by the Commission in SCE's last GRC for purposes of decommissioning costs relating

⁷⁴² Ex. TURN-704, SCE response to TURN Data Request 117, Q29.

⁷⁴³ Ex. TURN-13-E, p.114.
to fuel cell projects.⁷⁴⁴ Additionally, a 15% contingency factor is comparable to assumptions used by Pacific Gas & Electric and San Diego Gas & Electric.⁷⁴⁵ Further, SCE uses a contingency of 10% or less for almost all of SCE's other electric generation-related capital projects.⁷⁴⁶

SCE's own experience demonstrates that decommissioning contingencies have rarely been used for generation projects. In response to a TURN data request, SCE identified five generating projects that have been decommissioned since 2000. In four of the five cases, the recorded costs used no contingency and were under the cost estimate.⁷⁴⁷ Three of the projects had no contingency in their original estimates and only one these (the Mohave coal-fired generating station) went overbudget.⁷⁴⁸ Also notable is the fact that SCE did not use any of its 15% contingency factor for the Perris SPVP facility or any of the contingency factor for the UC Santa Barbara Fuel Cell.⁷⁴⁹

In evaluating the reasonableness of SCE's contingency factors, the Commission should give weight to its recent decision in the California Water Company General Rate Case (A.21-07-002). In that decision, the Commission rejected the majority of contingency factors proposed by Cal Water for capital projects and instead assigned no contingency to that work. In doing so, the Commission rejected a "blanket approach" to contingency factors and further explained "it has

⁷⁴⁴ D.21-08-036, pp. 536-537.

⁷⁴⁵ D.21-08-036, p. 536; Ex. TURN-13-E, p.114.

⁷⁴⁶ Ex. TURN-13-E, p. 85; Ex. TURN-13-Atch1, SCE response to TURN Data Request 80, Q17.

⁷⁴⁷ Ex. TURN-704, SCE response to TURN DR 117, Q20.

⁷⁴⁸ Ex. TURN-704, SCE response to TURN DR 117, Q20. Since there was no contingency for Mohave, the fact that the costs were above the estimate only means that a 0% contingency factor would have been inadequate for that project. Mohave also involved a unique set of challenges given the nature of the facility. There is no evidence that similarly unique challenges are present for Mountainview and the Peakers.

⁷⁴⁹ Ex. TURN-704, SCE response to TURN DR 117, Q20 (The Perris facility is name "Dexus" in the table)

long been our practice, consistent with ratemaking policy, to disallow contingencies in order to motivate utilities to remain within their forecast budgets for their capital projects".⁷⁵⁰

In support of its preferred 20% contingency, SCE argues against TURN's proposal by claiming that "contingencies are typically higher on decommissioning than construction because of the higher level of uncertainty."⁷⁵¹ The source SCE cites for this finding is a trade press article discussing coal-fired power plants that includes one sentence with the exact words appearing in SCE's testimony but provides no data or other supporting material.⁷⁵² That article also suggests that utilities should "research the contingencies approved in prior regulatory proceedings".⁷⁵³ As described earlier in this section, TURN's research shows that other projects and utilities have used lower contingency factors. Finally, the article notes a series of decommissioning cost risks relating specifically to the environmental remediation of coal facilities including potential contamination associated with asbestos, mercury, lead paint, and coal ash ponds.⁷⁵⁴ None of these risks apply to the relatively new gas-fired plants at issue in this case. SCE's reliance on this trade industry publication article to support its proposed contingency factor should therefore be given little weight.

SCE's arguments are not persuasive in light of actual decommissioning cost experience, the use of lower contingency factors for other generation projects, and the Commission's increasing concern about the use of blanket contingency factors for capital projects. The Commission should adopt TURN's 15% recommendation as a reasonable alternative for Mountainview and the Peakers.

⁷⁵⁰ D.24-04-042, pp. 25, 27

⁷⁵¹ Ex. SCE-16, p.64.

⁷⁵² Ex. TURN-704, SCE response to TURN Data Request 17, Q19, Attachment (p.3)

⁷⁵³ Ex. TURN-704, SCE response to TURN Data Request 17, Q19, Attachment (p.3)

⁷⁵⁴ Ex. TURN-704, SCE response to TURN Data Request 17, Q19, Attachment (p.2)

24.3 Fuel Cell

SCE operated two fuel cell projects that were disconnected from the grid at the end of 2022. The larger project (1.4 MW) was located at the campus of California State University, San Bernadino (CSUSB) while the smaller project (0.2 MW) was located at the campus of the University of California, Santa Barbara (UCSB).⁷⁵⁵ The projects have a remaining book value of \$0.299 million.⁷⁵⁶

SCE proposes to earn a full rate of return (7.44%) on the unamortized net book value of these assets over a four-year period.⁷⁵⁷ SCE expects to earn approximately \$0.05 million in return associated with the unamortized book value.⁷⁵⁸ TURN recommends not allowing any debt or equity return on the unamortized book value because the assets are no longer operational. In particular, the CSUSB project was removed from service after 9 years and 3 months despite having a 10-year service life used for purposes of depreciation.⁷⁵⁹

SCE argues that a full rate of return should be granted when "the abandonment results in a net benefit to customers" and asserts that this condition has been met in this case.⁷⁶⁰ The Commission should reject this claim and enforce the longstanding line of precedents denying utilities the ability to receive a rate of return on assets that are no longer used and useful. For a review of the relevant legal framework and Commission precedents addressing this issue, TURN refers to the analysis provided in Section 24.4 for the Solar Photovoltaic Projects.

⁷⁵⁵ Ex. TURN-13-E, pp.100-101.

⁷⁵⁶ Ex. TURN-13-E, p.101; Ex. TURN-712, SCE response to TURN Data Request 103, Q13.

⁷⁵⁷ Ex. TURN-13-E, p.101.

⁷⁵⁸ Ex. TURN-712, SCE response to TURN Data Request 103, Q13.

⁷⁵⁹ Ex. TURN-703, SCE response to TURN Data Request 117, Q39

⁷⁶⁰ Ex. SCE-16, p.84.

With respect to the specific facts regarding the fuel cell projects, there are several salient points. First, SCE sought approval of the fuel cell projects in 2009 on its own initiative and made its own determination that utility ownership (rather than third party contracting) was appropriate.⁷⁶¹ SCE did not submit its original application in response to any official direction from the Commission relating to utility-owned fuel cells. Second, SCE did not conduct any costbenefit analysis relating to continued operation of the CSUSB or UCSB fuel cells either at the same site or at another location.⁷⁶² There is no basis for SCE to claim that ratepayer savings were the reason for its decision to remove these units from service. Third, the CSUSB facility was not officially placed into service until 9 months after its intended commercial operation date due to problems with the units that required the fuel cell module to be repaired by the manufacturer.⁷⁶³ This delay led to the unit being removed from service prior to the end of its 10-year depreciation life.

TURN believes that the facts justify the enforcement of the longstanding precedents governing the recovery of unamortized ratebase for prematurely retired facilities.

24.4 Solar

In 2022 and 2023, SCE deenergized its entire 80.6 MW portfolio of 23 rooftop solar power facilities and one ground mounted solar project developed under the Solar Photovoltaic Program (SPVP).⁷⁶⁴ One additional SPVP project decommissioned in 2019 was the subject of disputes between TURN and SCE regarding rate recovery in the last GRC. In this case, SCE seeks to justify its decision to de-energize the remaining portfolio of assets and requests revenue

⁷⁶¹ D.10-04-028, pages 5-7.

⁷⁶² Ex. TURN-703, SCE response to TURN Data Request 117, Q37; Transcript, May 7, p.502.

⁷⁶³ Ex. SCE-16, p.85.

⁷⁶⁴ Ex. TURN-13-E, p.84.

requirements for recovery of unamortized project ratebase, ongoing O&M, and decommissioning costs.

SCE requests \$4.347 million for O&M in 2025.⁷⁶⁵ TURN recommends that ratepayers should only be responsible for 50% of ongoing lease payments for the de-energized facilities and proposes adjustments to the projected escalation rate for lease payments to reflect more reasonable future inflation assumptions. If the Commission does not adopt TURN's primary recommendation, it should direct SCE to track lease payments in a one-way balancing account to reflect the uncertainty of escalation, the fact that some leases have already been terminated, and the prospect that others will be terminated in the coming years. The impact of TURN's primary proposal for O&M is shown in the following table:

Solar O&M - TURN adjustments to SCE forecast (\$ thousands)					
	2025				
SCE Total Solar O&M	4,347				
less TURN Adjustments					
Reduce Lease Payments to Annual Values with Lower Escalation	(1,224)				
Disallow 50% of lease payments	(1,526)				
Subtotal Impacts of TURN Recommendations	(2,750)				
SCE Total Solar O&M After Applying TURN Adjustments	1,597				

For capital, SCE seeks approval of its \$9.723 million in recorded 2023 costs, \$0.201 million forecasted for 2024 forecast and \$33.526 million for the 2025 test year.⁷⁶⁶ SCE also requests approval to recover the remaining book value of its retired projects at a full rate of return over the remaining 6.5 years of their anticipated service life.⁷⁶⁷ TURN recommends that SCE's capital forecast be reduced to reflect a 50% disallowance and the use of a 10%

⁷⁶⁵ Ex. SCE-16, p.92.

⁷⁶⁶ Ex. SCE-16, p.98. ⁷⁶⁷ Ex. TURN-13-E, p.94.

contingency for decommissioning costs.⁷⁶⁸ TURN further proposes that ratepayers be held responsible for only 50% of unamortized ratebase associated with the retired projects and that this amount be recovered without any rate of return over the course of six years.⁷⁶⁹ TURN further urges the Commission to require SCE to identify the amount of stranded distribution plant associated with the SPVPs and add this amount to the net book value subject to TURN's proposed solar ratemaking adjustment (50% disallowance, no rate of return). The impacts of TURN's proposals for capital and ratemaking disallowances are shown in the following tables:⁷⁷⁰

Solar Capital Expenditures - TURN adjustments to SCE forecast (\$ thousands)									
	2023	2024	2025	2026	2027	2028	Total		
SCE Total Solar Capex	11,136	201	33,526	33,109	-	-	77,972		
less TURN Adjustments									
Reduce Decommissioning Costs (10% Contingency)	(475)	(9)	(1,430)	(1,412)	-	-	(3,325)		
Disallow 50% of Decommissioning Costs	(5,331)	(96)	(16,048)	(15,849)	-	-	(37,324)		
Subtotal Impacts of TURN Recommendations	(5,805)	(105)	(17,478)	(17,260)	-	-	(40,648)		
SCE Total Solar Capex after TURN Adjustments	5,331	96	16,048	15,849	-	-	37,324		

Solar Ratemaking Adjustment (\$ thousands)									
	2025	2026	2027	2028	2029	2030	Total		
Disallow rate of return on ratebase for retired projects	(13,750)	(11,250)	(8,750)	(6,250)	(3,751)	(1,250)	(45,001)		
50% Disallowance to capital recovery	(13,870)	(13,711)	(13,567)	(13,416)	(13,263)	(13,107)	(80,934)		
Subtotal Impacts of TURN Recommendations	(27,620)	(24,961)	(22,317)	(19,666)	(17,014)	(14,357)	(125,935)		

TURN provides support for these recommendations in the following sections.

24.4.1 O&M costs

The primary driver of SCE's O&M forecast is lease payments for the SPVP sites hosting

projects that were disconnected and will soon be decommissioned. Most lease payments are

assumed to continue through 2031/2032 with one ending in 2030.771 SCE forecasts nominal

dollar lease costs for 2023 through the end of the lease period by applying an escalator based on

⁷⁶⁸ Ex. TURN-13-E, p.13.

⁷⁶⁹ Ex. TURN-13-E, pp.98-99.

⁷⁷⁰ The Solar Ratemaking Adjustment table does not include any stranded distribution plant since SCE failed to attempt to identify these amounts in its testimony.

⁷⁷¹ Ex. TURN-707, SCE response to CalCCA Data Request 5, Q5.1.b

actual escalation in lease payments between 2020-2022 for each SPVP project and then converting the annual nominal dollar lease payments for each project to real 2022 dollars.⁷⁷² SCE sums the individual lease payments in each year to calculate the annual total lease payment (in 2022\$) for 2023-2032 and then averages the total lease payments (in \$2022) for the years 2025-2028 to calculate a "normalized" forecast of lease payments over this timeframe. This normalized value is presented as the 2025 forecast.⁷⁷³ The average annual escalation applied to the entire portfolio of lease payments is 10.2% but the contract escalators for individual projects range from 3.35% to 15.69%.⁷⁷⁴ Using this approach, SCE developed a "normalized" lease payment for 2025 through 2028 of \$4.275 million (2022\$).⁷⁷⁵ SCE subsequently provided a revised forecast that makes certain changes to projected escalation rates for each project but SCE is not asking the Commission to use that revised forecast for purposes of establishing its O&M revenue requirement.⁷⁷⁶

Actual escalation for each lease is based on either the Consumer Price Index (CPI) for Los Angeles, Long Beach and Anaheim or an Annual Energy Use Charge (AEUC) that uses values from the Energy Information Administration's "Average Price of Electricity to Ultimate Customers by End-Use Sector" for the state of California.⁷⁷⁷ Out of 25 lease sites, 19 have escalation set at the higher of AEUC or CPI and the remaining 5 are tied exclusively to CPI.⁷⁷⁸

⁷⁷² Ex. TURN-13-E, p.88

⁷⁷³ Ex. TURN-707, Original SCE response to CalCCA Data Request 5, Q5.1.b

⁷⁷⁴ Ex. TURN-707, Original SCE response to CalCCA Data Request 5, Q5.1.b

⁷⁷⁵ Ex. TURN-707, Original SCE response to CalCCA Data Request 5, Q5.1.b.

⁷⁷⁶ Ex. TURN-13 Atch-1, SCE Revised response to TURN Data Request 80, Question 21c. ⁷⁷⁷ Ex. SCE-16, p.96.

⁷⁷⁸ Ex. TURN-13 Atch-1, SCE Revised response to TURN Data Request 80, Question 21c.

TURN identified a series of problems with SCE's forecasting approach that structurally bias the results to benefit shareholders. The first problem is that SCE assumes the extraordinarily high inflationary period that characterized 2021 and 2022 will continue in all future years. A review of historical AEUC data in the following table highlights the absurdity of SCE's approach:⁷⁷⁹

Average Price of Electricity to Ultimate Customers by End-Use Sector (US EIA) - California										
	Dec-14	Dec-15	Dec-16	Dec-17	Dec-18	Dec-19	Dec-20	Dec-21	Dec-22	Dec-23
Cents/kWh	14.46	14.41	14.8	15.2	16.1	16.05	17.26	19.41	22.07	24.16
% change (vs. prior year)		-0.3%	2.7%	2.7%	5.9%	-0.3%	7.5%	12.5%	13.7%	9.5%

For leases that escalate based on AEUC, SCE's forecasting approach conveniently relies exclusively on the two annual periods with anomalously high escalation values (2020-2021 and 2021-2022) shown in this table (highlighted in yellow). When asked why SCE did not consider a longer historical time horizon for purposes of developing an escalation forecast, SCE witness Billapati simply noted that "the team" chose not to include any prior years.⁷⁸⁰

SCE has presented no evidence to suggest that the extraordinarily escalation occurring during that period is likely to become the norm. For lease payments that change due to the Consumer Price Index, SCE similarly relies on an extraordinarily high inflationary period (2020-2022) shown in the following chart:⁷⁸¹

⁷⁷⁹ Ex. TURN-708, US EIA Electric Power Monthly reports for "Average Price of Electricity to Ultimate Customers by End-Use Sector" (Table 5.6.A) for California.

⁷⁸⁰ Transcript, May 13, p.817.

⁷⁸¹ Ex. TURN-707, US Bureau of Labor Statistics, Consumer Price Index, Los Angeles Area – March 2024.





TURN's testimony provided an alternative forecast of future lease escalation values through 2032 that relies on forecasted changes in CPI (from a third party forecaster) and AEUC (from the Energy Information Administration).⁷⁸² These forecasts show escalation expected to range from 2.0% to 4.5% (nominal) over the next 10 years.⁷⁸³ TURN's testimony recommends using the greater of the forecasted growth in CPI or AEUC to develop nominal dollar future lease payments that would be converted to 2022 dollars (using SCE's conversion factors) to determine annual payments for Test Year 2025. TURN's approach assumes that all SPVP project leases remain in force (and must be paid) through the GRC cycle.

SCE's rebuttal testimony takes issue with TURN's use of a national CPI forecast value (rather than one specific to Southern California) and an AEUC value for the entire Pacific Coast region.⁷⁸⁴ However, SCE made no effort to determine whether the changes in national CPI and

⁷⁸² Ex. TURN-13-E, pp.91-92, footnotes 114 and 115.

⁷⁸³ Ex. TURN-13-E, p.92, Table 29.

⁷⁸⁴ Ex. SCE-16, p.96.

Pacific Coast AEUC values have historically tracked with changes in the local CPI and California AEUC values included in the leases.⁷⁸⁵ During evidentiary hearings, SCE witness Billapati acknowledged that no comparison was performed to determine whether the national and local CPI values have been historically consistent.⁷⁸⁶ TURN acknowledges that its forecast uses CPI/AEUC values covering broader geographic areas, this approach is superior to SCE's decision to cherry-pick two of the highest inflationary years in recent history as the basis for calculating future lease payment escalation. The Commission should reject SCE's approach in favor of TURN's forecast method.

A second issue that undermines the validity of SCE's forecast is the likelihood that a significant number of SPVP leases will be terminated early and thereby reduce total payments SCE must make during the GRC cycle. This fact was omitted from SCE's prepared direct and rebuttal testimony. In response to TURN data requests, SCE acknowledged that the building owners at several SPVP sites have requested a re-roof of the facility "which provided SCE the opportunity to exercise its right to terminate the lease. Terminating those lease relieved SCE from future lease payments."⁷⁸⁷ During evidentiary hearings, SCE witness Billapati confirmed that any such request by the building owner allows SCE to terminate the lease and avoid any future payments.⁷⁸⁸

⁷⁸⁵ Ex. SCE-16, Appendix I, pp. I3 to I4 (Changes in the CPI/AEUC in a particular year relative to a base year are used to determine lease payments)

⁷⁸⁶ Transcript, May 13, p.821.

⁷⁸⁷ Ex. TURN-707, SCE response to TURN Data Request 117, Q53.

⁷⁸⁸ Transcript, May 13, p.809 ("if the building owner asks SCE to remove the panels and to reroof the entire facility, then the lease – the lease has to offer SCE the option to exercise its rights to terminate the lease. And it it's able to exercise the right to terminate, then SCE is not required to make the payments through the end of the lease.")

Despite this admission, SCE assumes that all leases will remain in force and incur payment obligations through the GRC cycle. However, three SPVP leases have already been terminated (Sites 13, 28, and 48) and will no longer require payments in 2025 and beyond.⁷⁸⁹ Although SCE will not incur any costs relating to these leases, SCE's forecasting methodology incorrectly assumes that total lease payments to these three sites will be \$0.974 million in 2025 and \$4.789 million between 2025-2028.⁷⁹⁰ These three leases comprise approximately 25% of the annual leasing costs forecasted by SCE for the entire SPVP portfolio over this time period.⁷⁹¹ There is no dispute that these costs will not be incurred by SCE and that any additional lease terminations would only add to the gap between forecasted and actual costs. When asked to confirm that SCE seeks to recover the entire forecast regardless of lease terminations that eliminate payment obligations, SCE witness Billapati stated "actual expenses on the leases could be lower than the original request. And that's the nature of forecasting",⁷⁹²

There is no justification for allowing SCE to collect amounts from ratepayers that will not be used to make lease payments and will instead be retained by shareholders. The beneficiaries of overforecasting are SCE's shareholders who would retain any excess revenues over this GRC period. The Commission should not allow SCE to overcollect given the information entered into the record during the course of this proceeding.

The adverse consequences for ratepayers are even more significant. As explained in the next section, TURN takes issue with SCE's cost-effectiveness analysis supporting the decision to

⁷⁸⁹ Ex. TURN-713, SCE response to TURN-SCE-Verbal-008, Q4

⁷⁹⁰ Ex. TURN-707, SCE response to CalCCA Data Request 5, Q5.1.b (values for 2025 for sites 13, 28 and 48)

⁷⁹¹ Ex. TURN-707, SCE response to CalCCA Data Request 5, Q5.1.b (values for 2025 for sites 13, 28 and 48 divided into the "total" values in 2025, 2026, 2027 and 2028)

⁷⁹² Transcript, May 13, p.815.

prematurely retire the SPVP facilities. One of the key inputs to the forecasted costs of continued operations (rather than retirement) is the assumption that, for all but one of the SPVP projects, the building owner will request a re-roofing that requires the projects to be entirely removed (and subsequently re-installed).⁷⁹³ SCE assumes that all these requests will occur during the current GRC cycle (between 2024-2027).⁷⁹⁴ Had SCE decided to continue operations at these facilities, rather than preemptively de-energizing the projects and committing to prompt decommissioning, future re-roofing requests (if they actually occur) would provide SCE with early termination rights and relief from ongoing lease payments for the remainder of the original lease term.

SCE's unilateral decision to retire and decommission these projects may foreclose the opportunity to benefit from early lease terminations since SCE would no longer have early termination rights once the systems are no longer on the building roofs. Instead, the full lease payments would be required through their original term. As a result, SCE's announcement of early retirement is likely to <u>increase</u> the total leasing costs relative to a scenario where SCE only retires an SPVP project if the building owner makes a re-roofing request (which would terminate the lease obligations). The Commission should take this "self-own" into account when assessing the reasonableness of SCE's overall actions.

Given these serious issues, TURN makes several recommendations for the O&M forecast for lease payments. First, any leases for which ratepayers are responsible should be assumed to escalate at the lower annual rates included in TURN's testimony. Second, ratepayers should only be responsible for 50% of forecasted lease payments to reflect both the fact that 25% of

⁷⁹³ Transcript, May 13, pp.799-800.

⁷⁹⁴ Ex. TURN-707, SCE response to TURN Data Request 103, Q1, "Ongoing O&M" attachment, "Remove and Reinstall year".

forecasted lease payments are associated with leases that have <u>already</u> been terminated, the fact that additional leases may be terminated in the coming years due to re-roofing requests, and the fact that SCE's decision to unilaterally announce the retirement of these projects will likely result in higher lease payment obligations (relative to a scenario where SCE waited to receive a reroofing request before deciding to decommission a facility).

If the Commission does not adopt TURN's primary recommendations, it should direct SCE to track lease payments in a one-way balancing account (the cap should be set using inflation forecast assumptions consistent with TURN's approach and remove the three terminated leases). TURN did not make this recommendation in direct testimony because the impact of reroofing requests on SCE's ability to request lease termination was only revealed in response to data requests relating to SCE's rebuttal testimony. TURN's alternative approach would ensure that SCE only recovers the actual costs of lease payments incurred through the GRC cycle. Disallowances for imprudence and unreasonable determinations could be addressed through a reduction in capital recovery discussed in the next section.

Ratepayers should not bear the costs of SCE's inflated (and demonstrably wrong) forecasting methodology or the adverse consequences of preemptively telling building owners seeking to reroof that they can continue to collect lease payments for the rest of the decade so long as they wait until SCE removes the installed solar project in the next few years. The Commission should assign ample responsibility to SCE and its shareholders for these easily-avoidable mistakes that constitute imprudence.

24.4.2 Capital costs

24.4.2.1 Reasonableness of SCE's management

24.4.2.1.1 SCE did not demonstrate prudent management of SPVP

The SPVP projects were installed between 2008 and 2013.⁷⁹⁵ When SCE originally asked for approval to invest in these facilities, it assumed a 20-year life for all projects and noted that panel warranties "range from 20-25 years" and the racking system should be expected to have a depreciable life of 30 years.⁷⁹⁶ SCE's original application seeking approval of the SPVP did not include any assumption that facilities may have to be prematurely removed due to re-reroofing requests or safety issues.⁷⁹⁷ The Commission should give little weight to SCE's current claim that the projects were essentially experimental and could not have been expected to operate for more than 12-15 years.⁷⁹⁸ This assertion is little more than a *post hoc* rationalization to justify a program that is characterized by massive stranded costs, poor performance and imprudent behavior by SCE. The SPVP program has been nothing short of a fiasco.

SCE engaged vendors and contractors to build the SPVP sites and relied on contractors to perform periodic inspections of the facilities.⁷⁹⁹ In testimony, SCE claims that the mounting system for the SPVPs was "largely untested" but, in response to data requests from TURN, was unable to identify the extent to which this system was in commercial use at the time the projects were installed.⁸⁰⁰ The notion that the mounting system involved new and untested technology was never referenced in SCE's original application seeking Commission approval.

⁷⁹⁵ Ex. SCE-05v1, p.268.

⁷⁹⁶ Ex. TURN-707, SCE testimony in A.08-03-015, pp.51-53.

⁷⁹⁷ Ex. TURN-706, SCE response to TURN Data Request 117, Q51.

⁷⁹⁸ Ex. SCE-16, p.108.

⁷⁹⁹ Ex. TURN-706, SCE response to TURN Data Request 103, Q8.

⁸⁰⁰ Ex. SCE-16, p.108; Ex. TURN-706, SCE response to TURN Data Request 117, Q48

The primary trigger for SCE's decision to evaluate the cost-effectiveness of continued operations was fire incidents at two SPVP sites (12 and 22) in 2021 caused by faulty panel connectors.⁸⁰¹ SCE asserts that the first time corrosion in the panel connections was noticed was after the ignition event at SPVP 012 in December 2021.⁸⁰² Upon noticing that many sites were at risk for similar fires, SCE decided to immediately de-energize the eight highest risk sites.⁸⁰³

There is ample reason to believe that SCE knew, or should have known, about the connector problems and other safety risks at SPVP facilities prior to the fire in December 2021. The Standard Cause Evaluation performed by SCE staff in 2022 noted that "hanging wires", "wires on the ground" and "bad connectors" were first identified as issues to be addressed at SPVP 12 in an inspection conducted in May of 2018.⁸⁰⁴ Additional inspections in mid-2021 found similar problems and identified the need for urgent repairs. However, no repairs were made and the facility experienced a fire in December 2021.⁸⁰⁵ A contributing cause identified in the report was "inadequate maintenance procedures" ⁸⁰⁶ The fact that SCE did not notice these same problems at other SPVP sites until there was a fire at SPVP 12 highlights a failure to conduct adequate regular maintenance and inspections.

Surprisingly, SCE was unable to identify the manufacturer of the defective parts or explain whether the defective parts were ordered by SCE or third-party contractors.⁸⁰⁷ SCE also

⁸⁰¹ Ex. TURN-13-Atch1, SCE response to TURN Data Request 28, Q29(d).

⁸⁰² Ex. TURN-706, SCE response to TURN Data Request 117, Q47.

⁸⁰³ Ex. SCE-16, p.107.

⁸⁰⁴ Ex. SCE-16, p.B16.

⁸⁰⁵ Ex. SCE-16, p.B16.

⁸⁰⁶ Ex. SCE-16, p.B16.

⁸⁰⁷ Ex. TURN-13-Atch1, SCE response to TURN Data Request 80, Q27.

failed to pursue claims against any vendor or manufacturer relating to design, manufacturing, or installation deficiencies with any SPVP project.⁸⁰⁸

The overall actions of SCE do not support a finding of prudent management. SCE's original application never mentioned any of the risks that subsequently materialized, assumed a standard 20-year project life, and never included the extremely relevant assumption that the building hosting every facility would need to be re-roofed prior to the end of the 20-year lease.⁸⁰⁹ SCE's delegation of procurement and installation to third parties with apparently minimal oversight or recordkeeping resulted in projects riddled with safety defects. SCE was on notice as to connector issues and bad wiring in 2018 but took no action until after a fire in December of 2021, after which SCE abruptly de-energized many facilities. The fact that SCE chose to de-energize all the remaining SPVP facilities approximately half-way through their expected operational lives based on safety concerns and questionable economic analysis does not represent reasonable performance or reflect typical industry experience. The Commission should find that SCE's management was imprudent and order disallowances of unrecovered capital as a remedy.

24.4.2.1.2 SCE's decision to prematurely retire all projects was driven by flawed analysis

SCE supports the decision to prematurely decommission its SPVP facilities in late 2022 based on an economic analysis performed in the third quarter of 2022.⁸¹⁰ This analysis purports to show that "decommissioning in 2025 and 2026 is the least-cost alternative to SCE customers".⁸¹¹ Specifically, the analysis finds that prematurely decommissioning the projects in

⁸⁰⁸ Ex. TURN-13-Atch1, SCE response to TURN Data Request 28, Q28(c) Supplemental.

⁸⁰⁹ Ex. TURN-706, SCE response to TURN Data Request 117, Q51.

⁸¹⁰ Transcript, May 13, pp.792-793.

⁸¹¹ Ex. SCE-05v1, p.271.

2025 and 2026 would cost \$295 million while continuing to operate them to the end of their original asset life would cost \$393 million.⁸¹² After reviewing the inputs to this analysis, only some of which were provided after repeated data requests, TURN is persuaded that structural flaws render its conclusions fundamentally incorrect. The analysis was designed to justify a choice that SCE had likely already made (prematurely decommission all SPVP facilities) and to hopefully convince the Commission not to enforce either a capital cost disallowance or deny SCE a return on sunk plant in a manner similar to the treatment for the Perris SPVP project adopted in the last GRC.

TURN's concerns fall into two basic categories. First, the valuation of SPVP project benefits is unreasonably low. The analysis finds only \$19 million in cumulative energy, generation capacity and Renewable Energy Credit benefits attributable to these projects through the end of their expected operational lives.⁸¹³

The Commission should decline to rely on this analysis. SCE refused to provide the generation profiles assumed for the SPVP projects in its cost-effectiveness analysis despite direct requests from TURN.⁸¹⁴ SCE relied on a "proprietary price forecasting mechanism" for future energy prices, declined to share these values with TURN, and refused to explain whether these values are consistent with the energy price forecast used for Integrated Resource Planning purposes by the Commission or the wholesale energy price forecast used for the Avoided Cost Calculator.⁸¹⁵ For capacity prices, SCE also relied on a September 2021 proprietary forecast that

⁸¹² Ex. SCE-05v1, p.271, Table IV-49.
⁸¹³ Ex. SCE-05v1, p.271, Table IV-49.

⁸¹⁴ Ex. TURN-706, SCE response to TURN Data Request 117, Q44.

⁸¹⁵ Ex. TURN-706, SCE response to TURN Data Request 117, Q45(a).

it refused to provide to TURN.⁸¹⁶ As a result, it is impossible for TURN or the Commission to assess whether SCE's input values and calculations are reasonable.

TURN's testimony provided the results of a preliminary analysis of the value of SPVP projects using values from the 2022 Avoided Cost Calculator (ACC) which, for stand-alone solar projects, are driven by forecasted wholesale energy prices and some credit for generation capacity during peak hours. TURN's analysis found that the operation of these projects would provide \$80-100 million (nominal) in ratepayer benefits through 2032.⁸¹⁷ Unlike SCE's proprietary modeling, TURN's use of the ACC is transparent and relies on Commission-approved values. Given the lack of transparency with SCE's forecast methodology, and the extraordinarily low benefits attributed to the operation of these projects, the Commission should decline to find that \$19 million in benefits from ongoing operation is reasonable.

The second (and bigger) problem with the analysis is the assumed cost for "ongoing O&M" assigned to the scenario where the SPVP facilities are <u>not</u> prematurely decommissioned. "Ongoing O&M" accounts for \$164 million under the continued operations scenario but only \$12 million under the premature decommissioning scenario.⁸¹⁸ Given that the overall cost difference between the two scenarios is \$98 million, and the difference in Ongoing O&M between the two scenarios is \$154 million, the basis for the "ongoing O&M" calculation should be carefully examined. Out of the \$164 million calculated for "ongoing O&M", approximately \$142 million (NPV) is attributable to "remove and reinstall".⁸¹⁹ This category calculates the cost of removing the entire solar system from a rooftop and then reinstalling the system after re-

⁸¹⁶ Ex. TURN-706, SCE response to TURN Data Request 117, Q45(b).

⁸¹⁷ Ex. TURN-13-E, p.98.

⁸¹⁸ Ex. SCE-05v1, p.271, Table IV-49.

⁸¹⁹ Ex. TURN-707, SCE response to TURN Data Request 103, Q1(b), "Ongoing O&M" attachment.

roofing has occurred.⁸²⁰ SCE provides no basis for the costs needed to perform this work at each site in its testimony or workpapers. The values for each project are simply hardcoded into the spreadsheet provided to parties seeking workpapers.⁸²¹

According to SCE, these "remove and reinstall" costs would be incurred if the building owner needs to re-roof the building.⁸²² Of the 24 SPVP installations, SCE's analysis assumes that 8 are removed/reinstalled in 2024, 8 removed/reinstalled in 2025, 5 are removed/reinstalled in 2026, and 2 are removed/reinstalled in 2027.⁸²³ Under the model logic, earlier removal/reinstall dates have a larger impact on the total NPV cost than later dates. After its own witness completed testifying, SCE provided additional written responses acknowledging that its model inaccurately included remove/re-install costs for a ground mounted installation (Site 42).⁸²⁴ This mistake inflated costs under the continued operation scenario by \$13 million (NPV) but is not shown in the results of SCE's analysis provided in testimony.

SCE provides no specific support for the assumption that every single building owner will seek to re-roof during this time period. SCE witness Billapati admitted that the dates selected in the analysis are "not based on a specific request from the building owner to remove the panel in order to perform the reroofing or any major refurbishment".⁸²⁵ SCE did not consider an alternative where only some building owners seek to re-roof. Instead, SCE relied on a forecast based on an opaque methodology that was never disclosed or explained in testimony or data

⁸²⁰ Transcript, May 13, p.799.

⁸²¹ Ex. TURN-707, SCE response to TURN Data Request 103, Q1(b), "Ongoing O&M" attachment, "remove and reinstall (nominal)"

⁸²² Transcript, May 13, p.799.

⁸²³ Ex. TURN-707, SCE response to TURN Data Request 103, Q1(b), "Ongoing O&M" attachment. 1 facility (Site 28) is assumed not to require removal/reinstallation.

⁸²⁴ Ex. TURN-713, SCE response to TURN-SCE-Verbal-008, Q3

⁸²⁵ Transcript, May 13, p.802.

responses. This forecast conveniently assumes that almost all building owners would initiate reroofing within a three-year period (2024-2026) at the front end of the current GRC cycle, thereby maximizing the cost impacts under the continued operation scenario. The arbitrary re-roofing assumption is intentionally constructed to support the outcome that SCE intended the exercise to produce.

Additionally, SCE does not consider a scenario where projects are refurbished and continue to operate until a building owner decides to re-roof. As explained in the discussion of O&M costs, SCE's decision to prematurely retire SPVP facilities appears to have locked in significant lease payment obligations through 2032 that could have otherwise been avoided. SCE acknowledges that the obligation to make lease payments through 2032 is unaffected by the decision to prematurely decommission projects.⁸²⁶ Had SCE waited until receiving a re-roofing request to decommission a project, all future lease payments associated with that site could have been avoided.

SCE's own analysis shows that refurbishing the projects now and operating them until a reroofing request is made would be cost-effective. SCE's workpapers reveal that the incremental costs of refurbishment and new telemetry at every SPVP site (total of \$15.8 million) would be more than offset by SCE's avoided lease payment obligations and the ongoing electric production benefits even if all building owners decide to reroof between 2024-2028.⁸²⁷ This obvious alternative approach was not considered by SCE in its analysis.

⁸²⁶ Ex. TURN-706, SCE response to TURN Data Request 117, Q52.

⁸²⁷ Ex. TURN-707, SCE response to CalCCA Data Request 5, Q5.1.b (avoiding lease payments for all SPVP projects starting in 2028 would yield \$25 million in savings); Ex. TURN-707, SCE response to TURN Data Request 103, Q1(b), "Ongoing O&M" attachment (total refurbish and telemetry costs for all SPVP projects assumed to be \$15.8 million); Ex. TURN-707, SCE response to TURN Data Request 103, Q1(a), "Benefits" attachment (total solar benefits from 2023-2033 assumed to be \$26.17 million).

SCE's approach is patently unreasonable. There is no credible basis for SCE to assume that all building owners hosting SPVP facilities would seek to re-roof prior to the end of the leases and specifically during a compressed period of a few years at the front end of this GRC cycle. SCE did not survey building owners or base this assumption on specific requests from the owners. Instead, SCE arbitrarily selected near-term dates, applied them to all but one of the SPVP projects, and then generated large cost values that could support its claim that premature retirement was cost-effective. SCE ignored potential savings from avoided lease payments if the owner decides to reroof, savings that are lost if SCE proactively removes the system, and undervalued the benefits of continued operations.

24.4.2.2 Treatment of unrecovered ratebase 24.4.2.2.1 TURN's proposal

For the unrecovered net book value of the SPVP assets that were removed from service, TURN proposes that ratepayers be held responsible for only 50% of unamortized ratebase associated with the retired projects. TURN further recommends that any allowable recovery of this sunk capital exclude any rate of return on debt or equity and be amortized over the course of six years.⁸²⁸ According to SCE, all SPVP projects were disconnected from the grid in March and April of 2023. At that time, the net book value for these assets was \$103.048 million.⁸²⁹ TURN recommends enforcing its ratemaking treatment based on the net book value for these facilities at the time they were removed from service.

TURN's recommendation to deny a rate of return on unrecovered capital does not rely on a finding of imprudence by the Commission. In the last GRC, the Commission adopted TURN's

⁸²⁸ Ex. TURN-13-E, pp.98-99.

⁸²⁹ Ex. TURN-712, SCE response to TURN Data Request 117, Q64.

proposal to deny SCE any return on the unrecovered capital associated with the Perris SPVP and recover the remaining net plant over six years.⁸³⁰ There was no finding of imprudence in that proceeding. Even if the Commission finds no unreasonable actions by SCE, the same treatment should be applied to all SPVP capital deemed eligible for recovery in this case.

TURN's recommended disallowance of unrecovered capital is based on both SCE's mismanagement of the SPVP projects and its decision to prematurely retire the assets in reliance on an intentionally flawed cost-effectiveness analysis. SCE's imprudent actions resulted in a large stranded cost obligation, much of which was avoidable, that should not be born exclusively by ratepayers. Although the Commission declined to enforce a capital disallowance for the Perris SPVP facility in the last GRC (D.21-08-026), TURN's recommendation in this proceeding is based on different facts.

For the Perris facility, SCE was forced to remove the project due to a reroofing request by the building owner and determined that it was uneconomic to reinstall the project. After the issuance of D.21-08-026, SCE made a unilateral decision to proactively deenergize and decommission all the remaining SPVP sites. SCE justifies this decision through a flawed economic analysis that assumes every remaining project will be subject to a hypothetical future re-roofing request (conveniently occurring in 2024-2027 timeframe covered by the next GRC). The analysis is wrong and SCE's decision to prematurely decommission the remaining facilities was an unreasonable mistake of its own making. A decision to continue operations at facilities not subject to actual re-roofing requests would have benefited ratepayers. SCE's unreasonable action also jeopardized its ability to benefit from lease terminations that would relieve ratepayers

⁸³⁰ D.21-08-036, p.534.

of the obligation to make payments for the remainder of the decade. These consequences merit a remedy that protects ratepayers rather than rewarding SCE's shareholders.

Furthermore, the disclosure of safety issues and poor maintenance/inspection practices at the remaining SPVP sites did not occur until this proceeding. That issue was not available for the Commission to consider in the last GRC or in any prior proceeding where capital recovery was at issue. The issue is therefore reviewed for the first time in this GRC. TURN believes that SCE's poor maintenance and inspection practices should serve as an additional basis for the disallowance of capital.

24.4.2.2.2 Commission precedents governing recovery of ratebase that is no longer "used and useful" are clear and consistent

SCE seeks to realize a full return on the unrecovered capital investment in its entire fleet of SPVP facilities despite the fact that these projects are no longer "used and useful" and were retired up to 10 years before the end of their expected operational and depreciable lives. TURN opposes this treatment as inconsistent with long established precedents governing abandoned plant and prematurely retired generation facilities. In a wide array of litigated situations involving shutdown generating facilities, the Commission has repeatedly denied any return on capital regardless of whether utility actions are demonstrated to be prudent.

The longstanding ratemaking treatment for prematurely retired facilities was not adopted to punish the utility for imprudence, mismanagement or poor planning. In circumstances where imprudence or mismanagement is involved, the Commission may disallow direct investment rather than merely denying a return on the investment (as proposed by TURN for SPVP assets in this case). By contrast, the denial of a return on capital has been characterized by the Commission as a fair balancing of shareholder and ratepayer interests.⁸³¹ Furthermore, the US Supreme Court has repeatedly rejected the claim that denying a regulated utility recovery of capital investments that are no longer used and useful constitutes an impermissible taking under the Constitution.⁸³² The Commission is therefore well within its rights, and acting consistent with decades of precedents, in adopting TURN's recommended ratemaking treatment even if it finds that SCE's management decisions were reasonable and prudent.

TURN relies on a number of precedents that have guided the Commission's

determinations with respect to prematurely retired facilities. In the case of Humboldt Bay Unit 3, the Commission denied any return on unrecovered capital for a nuclear plant that operated for 13 years before being prematurely retired by PG&E. The Commission explained that

in the case of a premature retirement, the ratepayer typically still pays for all of the plant's direct cost even though the plant did not operate as long as was expected. The shareholder recovers his investment but should not receive any return on the undepreciated plant. This is a fair division of risks and benefits.⁸³³

The Commission similarly denied any return on capital for several SDG&E-owned

facilities (Encina 1, Silvergate and Station B power plants) removed from service because they

were no longer needed after the commissioning of the Southwest Powerlink transmission line.834

The Commission denied any return on capital at several retired LNG facilities in the same rate

⁸³¹ D.85-08-046, p.22 ("in the case of a premature retirement, the ratepayer typically still pays for all of the plant's direct cost even though the plant did not operate as long as was expected. The shareholder recovers his investment but should not receive any return on the undepreciated plant. This is a fair division of risks and benefits")

⁸³² Duquesne Light Co. et al. v. Barasch et al. (1989) 488 U.S. 299, 302, 109 S. Ct. 609, 102 L. Ed. 2d 646, 1989 U.S. LEXIS 313, 57 U.S.Lw. 4083, 98 P.U.R.4th 253 ("a state scheme of utility regulation does not "take" property simply because it disallows recovery of capital investments that are not "used and useful in service to the public."")

⁸³³ D.85-08-046, p.22.

⁸³⁴ D.85-12-108, 1985 Cal. PUC LEXIS 1112, *57.

case.835

The same treatment was applied to PG&E's request for the recovery of costs for Geysers 15, a prematurely shutdown utility-owned geothermal generating facility. The Commission explained that

we once again endorse our longstanding regulatory principle that shareholders should earn a return only on used and useful plant. We note that DRA's recommendation does provide that ratepayers pay PG&E's shareholders for the entire remaining unamortized plant balance on Geysers 15, but simply not pay a return. We believe our decision is consistent with the Legislature's directives in PU § 455.5, and is fully supported by the record before us.⁸³⁶

In SCE's 2012 General Rate Case, the Commission refused to allow any return on \$90 million in unrecovered capital and decommissioning costs for the prematurely shutdown Mohave Generating Station. The Commission relied upon the Humboldt 3 precedent and concluded that "shareholders should not receive a rate of return on the undepreciated, non-operational plant or decommissioning expenses."⁸³⁷

In SCE's 2021 General Rate Case, the Commission affirmed the relevance of these precedents and applied the same treatment to the unrecovered capital associated with the prematurely retired Perris SPVP facility. In doing so, the Commission stated "we agree with TURN that it is inappropriate for SCE to continue to receive a return on the Perris investment because it has been decommissioning and is no longer used and useful."⁸³⁸ The Commission found no basis for an exception since the "impetus for the non-used and useful status was utility actions rather than Commission desires or actions."⁸³⁹ Moreover, the Commission concluded that

⁸³⁵ D.85-12-108, 1985 Cal. PUC LEXIS 1112, *64.

⁸³⁶ D.92-12-057, 1992 Cal. PUC LEXIS 971, *83, *84

⁸³⁷ D.12-11-051, pp.652-653

⁸³⁸ D.21-08-036, p.531

⁸³⁹ D.21-08-036, p.533

there was "no demonstration that the premature retirement results in net benefits to ratepayers."⁸⁴⁰ This conclusion is notable because SCE claimed that premature retirement of Perris was less expensive than continued operation – this rationale did not satisfy the Commission since the same facts would be true for any uneconomic asset that is prematurely retired. A mere showing that an asset is uneconomic to continue operating does not provide a waiver from the application of the "used and useful" rule to the disallowance of any rate of return on sunk capital. In every relevant precedent addressing this issue, the retired assets were uneconomic and retirement was more economic for ratepayers yet no return was authorized.

In each of these decisions, the Commission emphatically rejected the notion that prematurely retired plant should receive any return on debt or equity. The Commission should remain mindful of the unaltered and "longstanding regulatory principle" that return on capital is only available for plant that remains in service.⁸⁴¹ Importantly, none of these decisions rely upon a finding that imprudence was the cause of the early retirement and none characterized the denial of a return as attempts to punish the utility for bad behavior. Instead, the Commission has explained this outcome as "a fair division of risks and benefits."⁸⁴² The Commission need not reach any conclusion with respect to prudence to deny the utilities a return on their retired plant. In the event of imprudence or negligence, the Commission has other remedies such as reductions to ratebase, disallowance of outage costs or expenses, or denying the recovery of capital additions.

To the extent that the Commission has previously allowed any return on abandoned plant

⁸⁴⁰ D.21-08-036, p.533

⁸⁴¹ D.92-12-057, 1992 Cal. PUC LEXIS 971, *83, *84

⁸⁴² D.85-08-046, p.22.

or retired facilities, there are special circumstances at issue and no prior decision has authorized the full rate of return requested by SCE in this proceeding. For example, the Commission approved a settlement relating to the San Onfore Nuclear Generating Station Unit 1 that allowed unamortized capital to earn a return set at the embedded cost of debt.⁸⁴³ Since this outcome was included in a Settlement, and involved an associated commitment by SCE to permanently retire the facility, it cannot be considered precedential. The Commission has expressly declined to consider D.92-08-036 as a relevant precedent.⁸⁴⁴

In D.11-09-017 the Commission authorized the Golden State Water Company to recover sunk costs of a facility retired due to a settlement at the utility's cost of debt while explicitly denying any return on equity on the basis that "it is not reasonable for ratepayers to pay a return on equity as if Hill Street were still used and useful or capable of providing adequate service."⁸⁴⁵ In that case, the resolution again occurred via settlement that is not precedential.

In the case of SCE's legacy electromechanical meter retired prematurely due to the installation of advanced Smart Meters, the Commission allowed a reduced return on equity and explained that the outcome deviated "from the general principle of excluding a rate of return on the net plant balance of assets that are no longer used and useful.⁸⁴⁶ A similar decision issued for PG&E explains that the treatment for electromechanical meters is based on special circumstances even though, as a general matter, "The Commission has determined that plant which is not used

⁸⁴³ D.92-08-036.

⁸⁴⁴ D.05-12-040, Finding of Fact #65 ("In D.92-08-036, the Commission addressed the recovery of remaining undepreciated plant investment for Unit 1, which was shut down before the end of its license life. The Commission adopted a settlement that allowed a four-year amortization of the remaining unrecovered plant investment. It also allowed a return equal to the embedded cost of debt on the unamortized balance during the amortization period. Since this decision adopted a settlement, it did not set a precedent.")

⁸⁴⁵ D.11-09-017, p.6; D.10-06-031.

⁸⁴⁶ D.12-11-051, pp.649-650.

and useful should be excluded from rate base (and therefore excluded from earning a rate of return).⁸⁴⁷ The special circumstance in both proceedings was the Commission's directive for each utility to install Smart Meters which required the early retirement of legacy meters. There are no similar special circumstances presented by SCE in the current proceeding.

These precedents demonstrate the basic presumption that, for any prematurely retired facility, the utility may not earn any return on unrecovered capital. Even in cases involving special circumstances, the Commission has authorized less than a full rate of return. By contrast, SCE's position in this case would result in a full return for abandoned plant. Consistent with decades of relevant precedents addressing similar situations, the Commission must deny this request and enforce the longstanding prohibition on a utility earning a return on plant that is no longer "used and useful".

24.4.2.2.3 There is no relevant difference between the treatment of Perris in D.21-08-036 and the situation presented in this proceeding

In the last GRC, TURN successfully argued that SCE should not earn any return on the unrecovered investments in the Perris SPVP facility at the time it was prematurely retired. The Commission stated that "we do not find it consistent with Commission precedent or a fair division of risks and benefits for ratepayers to also pay for the return on the undepreciated plant balance" and noted that "such ratemaking treatment is consistent with past treatment the Commission has adopted for similar circumstances."⁸⁴⁸ SCE now urges the Commission to ignore that outcome and adopt the opposite treatment with respect to the remaining SPVP

⁸⁴⁷ D.11-05-018, p.55.

⁸⁴⁸ D.21-08-036, p.534.

facilities, arguing that the current situation is entirely different and justifies an exception from the rule. There is no basis for reaching the opposite conclusion in this case given the clear and consistent precedents governing the recovery of capital that is no longer used and useful. The Commission should reject SCE's self-serving arguments and apply the same treatment to the remaining SPVP facilities that was adopted in D.21-08-036. TURN addresses each of SCE's arguments in turn.

First, SCE claims that the retirement of the SPVP portfolio was caused by events beyond its control including defective components and lower value of solar energy in wholesale markets.⁸⁴⁹ The fact that SCE procured defective materials, and failed to identify these defects until over a decade after the program began, should not justify rewarding the shareholders with profits for stranded capital. With respect to the lower market value of solar energy, TURN disputes SCE's valuation methodology and notes that SCE refused to provide its own energy and capacity values to TURN citing a proprietary approach. Regardless, there is no exception to the "used and useful" rule that applies when an asset is no longer economic in the marketplace. Indeed, the "used and useful" rule is typically applied because generation plant becomes stranded due to changes in market forces.

Second, SCE claims that the SPVP program involved new technologies and risks "understood by the Commission when it approved SCE's SPVP program" and that denying a return on stranded capital would "discourage future investment".⁸⁵⁰ SCE cites no evidence that these risks were identified or discussed in the original Commission decision approving the SPVP, or that the Commission indicated a desire to provide additional ratemaking protections to

⁸⁴⁹ Ex. SCE-16, pp.102-103.

⁸⁵⁰ Ex. SCE-16, p.103

SCE in exchange for ratebasing new solar projects.⁸⁵¹ Furthermore, this same justification would apply to the Perris SPVP project addressed in the last GRC. All the SPVP projects were authorized by the same Commission decision. The Commission did not make an exception in that case and should not make a new exception in this case.

Third, SCE claims that the decision to prematurely retire the Perris facility is "completely different" because it was "based on the landlord's request to reroof the facility (which would have resulted in excessive cost if SCE elected to remove and reinstall the components to continue operations)".⁸⁵² As TURN explained in a prior section of this brief, SCE's primary justification for prematurely retiring the remaining SPVP facilities is its analysis showing that continued operations would be uneconomic. That analysis assumes that almost all of the remaining SPVP facilities will need to be removed and reinstalled due to future landlord reroofing requests. Eliminating that assumption would result in the analysis showing net benefits for continued operations (rather than benefits for premature retirement). The fact that SCE relies on the same cost driver (re-roofing requests) in this case highlights the comparability of the two cases.

Taken together, SCE's arguments fall entirely flat. There is no material difference between the circumstances surrounding the premature retirement of Perris and the remaining SPVP facilities. Any differences are irrelevant because the longstanding ratemaking approach described in the prior section is applicable in a wide range of situations with only a small number of exceptions resulting from specific circumstances not present in this case.

⁸⁵¹ D.09-06-049.

⁸⁵² Ex. SCE-16, p.104. The Commission addressed this issue in D.21-08-036, p.529.

24.4.2.2.4 Treatment of Distribution plant

SCE's original testimony supporting approval of the SPVP identified incremental investments in distribution plant as part of the cost of each installed project. SCE noted that distribution plant installed in connection with an SPVP facility "is the equipment needed to connect the solar PV generation to the distribution grid. The plant may include overhead conductor, underground conduit conductor, disconnect switches, distribution line transformers, services and other distribution equipment."⁸⁵³ SCE assumed that this plant has an average service life of 30-55 years.⁸⁵⁴ However, SCE's calculation of the sunk costs of SPVP facilities in this GRC excludes any distribution plant dedicated to the projects.⁸⁵⁵ SCE acknowledges that distribution plant may be retired when the SPVP projects are removed from service but insists that these unrecovered costs should continue to be collected in ratebase.⁸⁵⁶ TURN disagrees.

These costs are clearly attributable to the SPVP installation (as demonstrated by SCE's 2008 testimony), some of the plant is being removed as part of decommissioning, and other portions of the plant will remain unused. Allowing SCE to earn a full rate of return on this prematurely retired plant is unreasonable. The Commission can address this issue by requiring SCE to provide a full accounting of stranded distribution plant installed in connection with SPVP projects, disallowing 50% of the unrecovered book value, and declining to authorize any ongoing rate of return on that investment. Alternatively, the Commission could factoring this concern into the overall disallowance of SPVP capital proposed by TURN.

⁸⁵³ Ex. TURN-707, SCE testimony in A.08-03-015, p.53.

⁸⁵⁴ Ex. TURN-707, SCE testimony in A.08-03-015, p.53.

⁸⁵⁵ Ex. TURN-713, SCE response to TURN-SCE-Verbal-008, Q2

⁸⁵⁶ Ex. TURN-713, SCE response to TURN-SCE-Verbal-008, Q2

24.4.2.3 Decommissioning costs

SCE estimates decommissioning costs of \$77.697 for its portfolio of SPVPs and forecasts decommissioning will occur in 2025 and 2026.⁸⁵⁷ This estimate includes a 15% contingency factor applied to a set of standard assumptions that use both unit costing methods and site specific factors. The unit costs are derived from both the Perris decommissioning and specific quotes received from contractors. TURN recommends a 10% contingency factor be applied to the decommissioning cost estimates. This recommendation would reduce the total estimate from \$77.967 million to \$74.643 million.⁸⁵⁸

TURN's recommendation is based on several considerations. First, the near-term dates for decommissioning (2025-2026) limit the potential for unexpected long-term developments to increase overall costs.⁸⁵⁹ Second, SCE uses a contingency of 10% or less for almost all of SCE's other electric generation-related capital projects.⁸⁶⁰ SCE has not provided a specific rationale for deviating from this practice with respect to SPVP projects. Third, SCE's own experience demonstrates that decommissioning contingencies have rarely been used for generation projects. In response to a TURN data request, SCE identified five generating projects that have been decommissioned since 2000. In four of the five cases, the recorded costs used no contingency and were under the cost estimate.⁸⁶¹ Three of the projects had no contingency in their original estimates and only one these (the Mohave coal-fired generating station) went overbudget.⁸⁶² Also

⁸⁵⁷ Ex. TURN-13-E, p.84, *citing* Ex. SCE-05v1, p.271-272.

⁸⁵⁸ Ex. TURN-13-E, p.86.

⁸⁵⁹ Ex. TURN-13-E, p.85.

⁸⁶⁰ Ex. TURN-13-E, p.85; Ex. TURN-13-Atch1, SCE response to TURN Data Request 80, Q17.

⁸⁶¹ Ex. TURN-704, SCE response to TURN DR 117, Q20.

⁸⁶² Ex. TURN-704, SCE response to TURN DR 117, Q20.

notable is the fact that SCE did not use any of its 15% contingency factor for the Perris SPVP facility or any of the 35% contingency factor for the UC Santa Barbara Fuel Cell.⁸⁶³

In evaluating the reasonableness of SCE's contingency factors, the Commission should give weight to its recent decision in the California Water Company General Rate Case (A.21-07-002). In that decision, the Commission rejected the majority of contingency factors proposed by Cal Water for capital projects and instead assigned no contingency to that work. In doing so, the Commission rejected a "blanket approach" to contingency factors and further explained "it has long been our practice, consistent with ratemaking policy, to disallow contingencies in order to motivate utilities to remain within their forecast budgets for their capital projects".⁸⁶⁴

In defense of its 15% contingency factor, SCE argues that the use of cost data from the Perris decommissioning in 2019 for purposes of the remaining SPVP cost estimates should be assumed to be too low due to post-COVID inflation.⁸⁶⁵ However, SCE witness Billapati noted that some of the cost data was newly developed and did not rely on Perris costs.⁸⁶⁶ Specifically, Mr. Billapati explained that the cost of panel removal "was based on the recent quote from a vendor to perform the removal work."⁸⁶⁷ A review of the decommissioning cost estimates for two SPVP sites shows that panel removal costs comprise between 40-50% of the total cost estimate.⁸⁶⁸ This use of recent vendor quotes undermines SCE's claim that the Perris data is unreliable and requires a contingency.

⁸⁶³ Ex. TURN-704, SCE response to TURN DR 117, Q20 (The Perris facility is name "Dexus" in the table)

⁸⁶⁴ D.24-04-042, pp. 25, 27

⁸⁶⁵ Ex. SCE-16, p.102.

⁸⁶⁶ Transcript, May 13, pp.828-829.

⁸⁶⁷ Transcript, May 13, p.829.

⁸⁶⁸ Ex. TURN-707, SCE response to TURN Data Request 117, Q43, Attachments showing decommissioning cost estimates for 1464 Merrill Avenue and 9415 Kaiser Way.

SCE further claims that a higher contingency is appropriate because "SCE may also face claims from landlords relating to the removal work."⁸⁶⁹ However, SCE witness Billapati stated that "SCE has not faced any claims associated with the removal scope of work".⁸⁷⁰ The Commission should not allow for higher contingency factors based on pure speculation about theoretical liability risks that have not, in practice, ever occurred. To the extent that there are any liability issues that arise, the Commission should also evaluate whether SCE contributed to those costs through imprudent or otherwise unreasonable actions.

SCE's arguments are not persuasive in light of actual decommissioning cost experience, the use of 10% contingency factors for other generation projects, and the Commission's increasing concern about the use of blanket contingency factors for capital projects. The Commission should adopt TURN's recommendation as a reasonable alternative.

24.5 Catalina

SCE proposes several capital projects at the Pebbly Beach Generating Station (PBGS) on Catalina Island that are of concern to TURN. These projects include \$2.358 million solar carports and a \$1 million battery control system replacement/upgrade project.⁸⁷¹ Additionally, SCE initially proposed a \$0.5 million capital project relating to repurposing the microturbine space at PGBS. In response to discovery by TURN, SCE indicated that it would withdraw that project from this GRC.⁸⁷² This expenditure should be removed from the capital forecast.

⁸⁶⁹ Ex. SCE-16, p.101.

⁸⁷⁰ Transcript, May 13, p.830.

⁸⁷¹ Ex. SCE-05v1, pp.258, 261.

⁸⁷² Ex. TURN-13-E, pages 81-82, *citing* Ex. TURN-13-Atch1, SCE response to TURN Data Request 28, Q23(e).

TURN opposes SCE's request to ratebase the solar carport project due to SCE's violations of the express terms of the settlement agreement adopted in D.22-11-007 by failing to consider third-party ownership. The remedy for this violation should be a removal of the project costs from the capital forecast, a prohibition on placing these costs into ratebase and a requirement that project costs be borne by SCE's shareholders. If the Commission does not agree to TURN's primary recommendation, the project should not be allowed to earn any return on invested capital. Additionally, the project timeline should be moved to reflect a likely online date of January 2026.

TURN also opposes authorizing any expenditures on the Battery Control System (BCS) project in this GRC given the uncertain lifespan for the existing battery and SCE's failure to provide any assurances that the BCS would be compatible with a future battery system. If the Commission does not agree to TURN's primary recommendation, it should clarify that removal of the BCS from service after a short period of time would render it a stranded asset ineligible for a return on unamortized capital investment.

Catalina capital expenditures - TURN adjustments to SCE forecast									
	2023	2024	2025	2026	2027	2028	Total		
SCE Total Catalina Capex	1,998	1,079	1,000	1,000	500	500	6,077		
less TURN Adjustments									
Disallow Solar Carport Costs Due to Violation of Settlement	(1,279)	(1,079)	-	-	-	-	(2,358)		
Disallow BCS for NaS Battery	-	-	-	(1,000)	-	-	(1,000)		
Remove Repurpose MicroTurbine Space	-	-	-	-	(500)	-	(500)		
Subtotal Impacts of TURN Recommendations	(1,279)	(1,079)	-	(1,000)	(500)	-	(3,858)		
SCE Total Catalina Capex after TURN Adjustments	719	(0)	1,000	-	-	500	2,219		

The following table provides a summary of TURN's recommendations:

Support for TURN's recommendations is provided in the following sections.

24.5.1 Solar Carport project

SCE proposes to install two solar carports at PBGS pursuant that would be used to charge electric vehicles and offset onsite electrical loads at the main building.⁸⁷³ The project is being developed pursuant to an abatement order issued by the South Coast Air Quality Management District (SCAQMD) in September 2022 relating to a violation of particulate matter emissions at PBGS Unit 15. SCAQMD initially directed SCE to investigate the feasibility of installing a 100-400 kW solar system at PBGS and then subsequently directed SCE to install the generation by January 31, 2026.⁸⁷⁴ The estimated cost of the two solar carports is \$2.358 million.⁸⁷⁵

24.5.1.1 SCE's violation of the Catalina Settlement Agreement merits a remedy in this proceeding

In the last GRC, SCE sought authorization to pursue the Catalina Repower project that would have involved the replacement of six existing diesel generators at PBGS with new diesel generators.⁸⁷⁶ TURN opposed that request based on uncertainty surrounding the timing and scope of the overall project and SCE's unreasonable commitment to continue near-total reliance on diesel to generate power for Catalina island. The Commission found that additional scrutiny of the proposal was appropriate and directed SCE to file a stand-alone application for evaluation of an updated proposal.⁸⁷⁷

TURN participated actively in the proceeding dedicated to SCE's stand-alone Catalina Repower Project (A.21-10-005) and engaged with SCE and Cal Advocates to reach a settlement

⁸⁷³ Ex. SCE-05v1, p.258.

⁸⁷⁴ Ex. SCE-05v1, p.258.

⁸⁷⁵ Ex. SCE-05v1, p.258.

⁸⁷⁶ D.21-08-036, p.360.

⁸⁷⁷ D.21-08-036, p.362.
of contested issues that was adopted in D.22-11-007. The settlement provided a comprehensive framework governing the development of additional resources both at PBGS and at other locations on Catalina Island. Under the settlement, SCE committed to take specific actions relating to any new resources developed at the PBGS facility and elsewhere on Catalina. Section 6.2 of the settlement, which addresses the "resolution of Unit 15 Particulate Matter Noncompliance," requires SCE to consider third-party ownership for any non-diesel generation that SCAQMD requires be developed at the PBGS site. The language reads as follows:

If the SCAQMD requires SCE to install non-diesel generation at Pebbly Beach Generation Station (PBGS), SCE will be required to consider ownership of that generation by a third party. SCE will be required to demonstrate that its proposed solution (whether owned by SCE or a third party) is appropriate based on the following factors: cost-effectiveness, reliability, safety, physical security, cybersecurity, and operational viability.⁸⁷⁸

This section of the Settlement specifically applies to any direction provided by SCAQMD pursuant to the Unit 15 abatement order including replacement, retrofit or retirement of Unit 15 along with "other compliance options".⁸⁷⁹ SCAQMD has required SCE to install a solar carport (i.e., "non-diesel generation") at PBGS as part of compliance with its abatement order. Therefore, this provision of the settlement explicitly applies to the project. Under the plain language of this settlement provision, SCE is obligated to consider third-party ownership of the solar carport and is required to demonstrate that the proposed solution is cost-effective.

Despite the settlement, SCE has not taken any steps to consider third-party ownership of the solar carports.⁸⁸⁰ SCE agrees that the abatement order does not require the solar carport to be

⁸⁷⁸ Ex. TURN-13-Atch1, SCE, TURN, Cal Advocates Settlement Agreement, A.21-10-005, Section 6.2 (Phase 1B: Resolution of Unit 15 Particulate Matter Noncompliance)

⁸⁷⁹ Ex. TURN-13-Atch1, SCE, TURN, Cal Advocates Settlement Agreement, A.21-10-005, Section 6.2 (Phase 1B: Resolution of Unit 15 Particulate Matter Noncompliance)

⁸⁸⁰ Transcript, May 7, page 530.

utility owned and acknowledged that "SCE and SCAQMD did not discuss ownership of the required onsite solar at PBGS."⁸⁸¹ Instead, SCE claims that the third-party ownership provision of Section 6.2 of the Settlement does not apply to the solar carport project and would only be relevant if SCAQMD were to order SCE to install non-diesel generation at PBGS as an alternative to Unit 15.⁸⁸²

SCE's description of the Settlement provision is incorrect and adds new conditions that are not to be found in the plain language of Section 6.2. The specific language in Section 6.2 applies to any non-diesel projects installed at the PBGS site pursuant to an abatement order issued by SCAQMD. This provision was negotiated by TURN based on the understanding at that time that SCAQMD might require some onsite solar generation (as had been previously proposed) in addition to any diesel replacement or refurbishment of Unit 15.⁸⁸³ There is nothing in the settlement indicating that this provision applies only if SCAQMD requires the retirement of Unit 15 and replacement by non-diesel generation at the PBGS site. There is no evidence of sufficient land available at the cramped PBGS site to support a full replacement of Unit 15 with renewable generation. SCE's argument that the provision would only apply to SCAQMD-ordered renewable generation located at PBGS that completely replaces Unit 15 is neither reasonable nor plausible.

SCE further claims that the third-party ownership provisions in Section 6.2 "were clearly intended to apply to major projects like those anticipated from the Catalina Clean Energy RFO not to a tiny non-dispatchable project within the PBGS fence line."⁸⁸⁴ This claim is similarly

⁸⁸¹ Ex. TURN-702, SCE response to TURN Data Request 117, Q35; Transcript, May 7, page 534.

⁸⁸² Transcript, May 7, page 530.

⁸⁸³ Ex. TURN-13-E, page 78.

⁸⁸⁴ Ex. SCE-16, p.77.

unfounded. Section 6.2 applies to Phase 1B which involves the resolution of Unit 15 Noncompliance at the PBGS site and the installation of non-diesel resources within the fence line of PBGS pursuant to an abatement order. The Settlement includes <u>entirely different provisions</u> for Phase 2 which relate to the evaluation of third-party ownership of projects procured under the Clean Energy RFO.⁸⁸⁵ SCE's attempts to connect the requirements of Section 6.2 to the Clean Energy RFO constitutes a brazen misreading of the Settlement that is illogical on its face.

SCE further argues that, since parties to the settlement were aware that the January 2022 abatement order directed SCE to investigate the potential to install solar generation at PBGS, and there was no express reference to the solar carports in the Settlement, Section 6.2 was only intended to apply to a future abatement order issued by SCAQMD.⁸⁸⁶ This argument should be rejected. At the time the parties submitted the Settlement (April 29, 2022), the latest SCAQMD abatement order (issued January 4, 2022) directed SCE to <u>investigate</u> the potential to add new solar generation at PBGS.⁸⁸⁷ The settling parties (including TURN) were aware of the abatement order during the course of settlement negotiations. Since SCAQMD had not yet directed SCE to install any renewable generation at PBGS, the Settlement was crafted to ensure that any future abatement order requiring renewable generation at PGBS would be subject to an evaluation of third-party ownership. The evidentiary record of A.21-10-005 was submitted to the Commission on August 22, 2022.⁸⁸⁸ The second abatement order <u>requiring</u> the installation of solar generation at PBGS was issued on September 10, 2022.⁸⁸⁹ SCE's claim that the settlement provision should

⁸⁸⁵ Ex. TURN-13-Atch1, SCE, TURN, Cal Advocates Settlement Agreement, A.21-10-005, Section 6.3.1 (Phase 2: All-Source Request for Offers)

⁸⁸⁶ Ex. SCE-16, p.78.

⁸⁸⁷ D.22-11-007, p.4; Ex. SCE-16, p.78.

⁸⁸⁸ D.22-11-007, p.5

⁸⁸⁹ Ex. SCE-16, p.78.

only apply to abatement orders issued after the Commission's final decision in the proceeding (November 4, 2022) is not persuasive. SCE's narrative assumes that the parties knew, at the time when the settlement was submitted (April 29, 2022), that the Commission would not issue a final decision until November and intentionally created a gap in the Settlement obligations that would not apply to any SCAQMD abatement order issued prior to a final Commission decision. There is nothing in the Settlement or the Commission decision approving the Settlement that references a "future" abatement order or creates the loophole desired by SCE.

Finally, SCE argues that the Solar Carport project is not covered by Section 6.2 because it is "not dispatchable, only serves the EV load at PBGS and SCE's PBGS building, and is in no way an alternative to diesel generation."⁸⁹⁰ Once again, SCE inserts new limitations into the Settlement that do not appear in the actual agreement and are illogical on their face. Section 6.2 does not suggest that the requirements for the evaluation of third-party ownership only apply to "dispatchable" generation – this word and concept do not appear anywhere in the Settlement. Such a limitation would create a major loophole since most renewable generation, including solar, does not typically function as a "dispatchable" reesource. The fact that the solar carports would only serve EV charging and onsite building loads is equally irrelevant. Section 6.2 does not apply its requirements exclusively to renewable generation at PBGS that exports power to the rest of the island. Furthermore, since all loads on Catalina island are served exclusively by diesel generation at PBGS, any load at PBGS served by solar would reduce amount of load needed to be served by the diesel units at PBGS. In other words, any new solar generation on Catalina

⁸⁹⁰ Ex. SCE-16, p.80.

island (regardless of whether it exports to the larger grid) results in a reduction in diesel generation at PBGS.

SCE's intentional violation of the Settlement is egregious, flagrant and unacceptable. TURN entered into the Settlement with the assumption that SCE would act in good faith with repect to the binding provisions. The Commission should not allow SCE to ignore or evade the terms of the adopted all-party settlement. Since the Catalina proceeding is closed, and SCE seeks to ratebase assets covered by the Settlement in the GRC, this proceeding represents the only open and logical docket where SCE's compliance can be enforced.

TURN's primary recommendation is to exclude the carport project from ratebase with the costs borne entirely by SCE shareholders.⁸⁹¹ This outcome would ensure that SCE faces consequences for its violation and provide proper incentives for future compliance. If the Commission does not adopt TURN's primary recommendation, SCE could be permitted to move forward with the carport project but subject to a disallowance of any rate of return on the invested capital.⁸⁹² Given that SCE would still retain some of the tax benefits that are subject to the normalization of the federal Investment Tax Credit, this outcome would still permit SCE to realize financial benefits for its shareholders.⁸⁹³

If the Commission does not wish to enforce any financial penalty for SCE's violation of the settlement, it could instead order SCE to conduct an RFO for third-party offers to build and own the solar carport and sell the output to SCE under a Power Purchase Agreement. That approach would reflect the intent of the Settlement and ensure the lowest possible costs for

⁸⁹¹ Ex. TURN-13-E, p.79.

⁸⁹² Ex. TURN-13-E, p.72

⁸⁹³ Ex. TURN-13-E, p.79.

ratepayers. However, the absence of any financial consequences for SCE could fail to deter future noncompliance with other provisions of the Settlement.

24.5.1.2 Third-party ownership of the solar carports would reduce ratepayer costs

TURN's concern over SCE's faithful implementation of the Settlement with respect to third-party ownership is driven by the fact that utility-owned solar generation is lucrative for SCE's shareholders and generally harmful to the interests of ratepayers. In evaluating TURN's concerns, the Commission should be mindful of the disastrous experience with SCE's Solar Photovoltaic Program (SPVP) involving utility-owned solar projects that suffered safety problems, were prematurely retired, and resulted in significant stranded costs that may become the responsibility of ratepayers. TURN discusses the SPVP fiasco in Section 24.4 of this brief. With respect to the solar carport project, SCE's witness was unable to identify the expected life of the project or the duration of any warranties that would apply to procured equipment.⁸⁹⁴ SCE's witness was similarly unable to predict whether long-term ownership of the carport project would require future refurbishment or replacement costs to be collected from ratepayers.⁸⁹⁵

By comparison with a utility-owned solar project, third-party ownership locks in longterm fixed pricing, places the risk of poor performance on the project owner, and prevents ratepayers from serving as the financial backstop if a project fails to operate properly, does not have sufficient warranties for defective equipment, or is prematurely retired. These benefits are relevant to assessing which ownership model favors ratepayer interests.

⁸⁹⁴ Transcript, May 7, pp.538-539.

⁸⁹⁵ Transcript, May 7, p.539.

TURN's testimony outlines another set of important economic benefits to ratepayers from third-party ownership of solar resources.⁸⁹⁶ Any solar generating resource is eligible for the federal Investment Tax Credit (ITC). Federal law requires that utilities normalize the ITC while third-party project owners are not subject to this constraint and can flow through the full value in the form of price reductions in a Power Purchase Agreement.⁸⁹⁷ This differential treatment makes utility-owned solar projects significantly more expensive for ratepayers than third-party ownership. TURN's testimony and opening brief in A.21-10-005 explains the issue as follows:

> TURN witness Dowdell explained that both bonus depreciation and the Investment Tax Credit are normalized for utility-owned assets (as opposed to being flowed through) which lowers their value to ratepayers. Additionally, SCE's net operating losses and tax credit carryovers have already zeroed out the utility's tax liabilities in the coming years, making it very difficult for these tax incentives to yield meaningful value for either the utility or its ratepayers. Finally, third party projects are typically financed with more lower-cost debt and less high-cost equity, making the overall weighted average cost of capital lower than for comparable utility-owned projects.⁸⁹⁸

Under normalization, the ITC value is not flowed through as generated but applied evenly *pro rata* across the useful life of the project. However, for book accounting purposes, utilities recognize ITC and accelerated depreciation as generated, thereby increasing the near-term cash available for shareholders. The value of the ITC is effectively shared between ratepayers and utility shareholders, providing an effective boost to profits at the expense of ratepayers. In A.21-10-005, TURN provided a comparative cost analysis of the same solar project owned by an IOU and a third party that accounted for both the ITC value along with different capital structure and

⁸⁹⁶ Ex. TURN-13-E, pp. 76-78.

⁸⁹⁷ Ex. TURN-13-E, p.77

⁸⁹⁸ Ex. TURN-13-E, p.77, *citing* TURN opening brief, A.21-10-005, July 15, 2022, pp.10-11.

financing assumptions for each owner. This analysis found that the levelized cost (on a \$/kWh basis) could be reduced by approximately 50% under third-party ownership.⁸⁹⁹

The significant savings available under third-party ownership was a key issue raised by TURN in A.21-10-005 and served as the basis for the inclusion of the relevant provision in the Settlement agreement relating to any non-diesel generation installed at PBGS. SCE's testimony in the GRC entirely ignores the differential benefits of the ITC to ratepayers under various ownership scenarios despite TURN identifying and explaining this issue in direct testimony.

SCE's refusal to consider third-party ownership will deprive ratepayers of lower-cost alternatives and place ratepayers on the hook for cost overruns, performance problems and premature retirement. While SCE is not motivated to seek third-party ownership, the Commission should recognize the legitimacy of TURN's concerns.

24.5.1.3 SCE's schedule for the Solar Carport Project is implausible

SCE's original testimony forecasted all capital spending on the solar carports to occur in 2023 and 2024.⁹⁰⁰ In response to TURN's direct testimony criticizing the unrealistic timing of forecasted capital expenditures, SCE admitted that the original forecast "is inconsistent with the current status of the project" and modified its forecast by moving spending to 2024 and 2025.⁹⁰¹ Based on a review of likely constraints on the development of this project, TURN does not believe that it is likely the project will be completed until early 2026. The Commission should therefore adopt the assumption that the costs of this project, if allowed to be recovered from ratepayers, will not result in completed capital additions in the test year.

⁸⁹⁹ Ex. TURN-13-E, p.77, *citing* Testimony of Jennifer Dowdell, A.21-10-005, page 10, Table 2.
⁹⁰⁰ Ex. SCE-05v1, p.257.

⁹⁰¹ Ex. SCE-16, p.80.

TURN's testimony notes that the project is currently in the "conceptual design phase," that is expected to be completed in the second quarter of 2024.⁹⁰² During hearings, SCE witness Hernandez stated that this work was ongoing was expected to be complete "by the end of June of this year."⁹⁰³ Once conceptual design is complete, SCE envisions a six month design engineering phase.⁹⁰⁴ During this phase, SCE intends to apply for all relevant permits to be issued by the City of Avalon and the California Coastal Commision.⁹⁰⁵ SCE was unable to identify which permits may be needed or the timing involved in obtaining necessary approvals.⁹⁰⁶ As noted in TURN's testimony, SCE intends to order equipment during the design engineering phase occurring in the second half of 2024 and assumes 12 month leadtime for procuring the necessary equipment.⁹⁰⁷ SCE also assumes that construction will take 6 months.⁹⁰⁸

TURN's testimony uses these timing requirements to forecast the likely schedule for project completion.⁹⁰⁹ TURN's forecast assumed the completion of conceptual design in the first quarter of 2024 while SCE now admits that this task will not be complete until the end of the second quarter. TURN's forecast, which includes <u>no time delays</u> associated with the acquisition of relevant permits (an extremely optimistic assumption), shows construction being complete in early 2026.⁹¹⁰ The SCAQMD deadline for completion of this project is January 31, 2026.⁹¹¹ SCE's rebuttal testimony does not either respond to TURN's timeline or present any evidence to

⁹⁰² Ex. TURN-13-E, p.74.

⁹⁰³ Transcript, May 7, p.535.

⁹⁰⁴ Ex. TURN-13-E, p.74; Transcript, May 7, p.535.

⁹⁰⁵ Ex. TURN-13-E, p.74; Transcript, May 7, p.536.

⁹⁰⁶ Ex. TURN-702, SCE response to TURN Data Request 98, Q20; Transcript, May 7, p.536.

⁹⁰⁷ Ex. TURN-13-E, p.74; Ex.WPSCE05V1, p. 231.

⁹⁰⁸ Ex. TURN-13-E, p.74

⁹⁰⁹ Ex. TURN-13-E, p.74

⁹¹⁰ Ex. TURN-13-E, p.74, Table 26.

⁹¹¹ Ex. TURN-13-E, p.75

support an earlier online date. Given the realities of project development, and the delays that have already occurred, the Commission should direct SCE to assume, for purposes of determining when any allowable costs would be recorded to ratebase, that the project will be online in January 2026.

24.5.2 Battery Control System Upgrade

SCE requests approval of its forecast to spent \$1 million in 2026 to replace the Battery Control System (BCS) on the Sodium Sulfide (NaS) battery located at PBGS.⁹¹² The replacement is justified by the fact that the existing switchgear electronics have been experiencing performance issues.⁹¹³ At the time SCE filed its application in early 2023, the NaS battery was assumed to have five years of remaining useful life, meaning that the battery may need to be replaced in 2028.⁹¹⁴

TURN opposes SCE's request as premature given the uncertainty as to whether the NaS battery will operate beyond 2028. In response to TURN data requests, SCE could not state whether the new BCS to be procured in 2026 would be compatible with a new battery that may be needed to replace the NaS unit as early as 2028.⁹¹⁵ If the BCS is not compatible, ratepayers would be forced to pay for a new control system that operates for only two years and then becomes a stranded asset. This outcome is unreasonable, inefficient and unfair to ratepayer interests.

There are significant unresolved questions as to whether a new BCS is actually needed at PBGS. While SCE claimed in rebuttal testimony that the existing system "has reached the end of

⁹¹² Ex. SCE-05v1, p.257.

⁹¹³ Ex. SCE-05v1, p.261.

⁹¹⁴ Ex. TURN-13-Ê, p.80.

⁹¹⁵ Ex. TURN-13-E, p.80; Ex. TURN-13-Atch1, SCE response to TURN Data Request 52, Q21(c).

life and requires replacement^{"916}, SCE witness Hernandez clarified during hearings that "we still believe there is some remaining useful life of the BCS system as a whole".⁹¹⁷ Mr. Hernandez further explained that SCE recently procured parts to repair the BCS in order to conduct a complete diagnostic test and determine the likely remaining life of the entire NaS battery.⁹¹⁸ Based on this testimony, it is difficult to assess whether a new BCS is actually needed in the current GRC cycle, how long the NaS battery may continue to operate, and what alternatives may be under consideration in the coming years.

In rebuttal testimony, SCE argues that the failure of the battery could result in a violation of the SCAQMD Title V permit.⁹¹⁹ A violation would only occur if the battery is not functioning and less than 50% of the onsite microturbines are simultaneously operable.⁹²⁰ Assuming SCE reasonably maintains the microturbines, delays in resolving the long-term future of the NaS battery will not result in a permit violation.

Given these uncertainties, the Commission should reject SCE's request to authorize \$1 million for a new BCS and instead direct SCE present a comprehensive plan regarding the NaS battery system in the next GRC.⁹²¹ If SCE needs to replace the BCS or the entire NaS battery system prior to the next Test Year, SCE should request establishment of a memorandum account for the NaS system replacement, which would allow SCE the opportunity to have the reasonableness of its actions approved in the next GRC.

⁹¹⁶ Ex. SCE-16, p.81.

⁹¹⁷ Transcript, May 7, p.543.

⁹¹⁸ Transcript, May 7, p.544.

⁹¹⁹ Ex. SCE-16, p.82.

⁹²⁰ Ex. TURN-702, SCE response to TURN Data Request 117, Q32(a).

⁹²¹ Ex. TURN-13-E, p.81.

If the Commission decides to approve SCE's proposal in this GRC, it should clarify that the removal of the BCS from service only a few years into its operational life would result in the BCS no longer being used and useful and subject to the standard ratemaking treatment for such assets (no return on ratebase, recovery amortized over 4-6 years). Contrary to SCE's claim that such treatment would be "punitive" to SCE⁹²², adopting TURN's recommendation would properly motivate SCE to pursue long-term solutions that benefit ratepayers rather than prioritizing short-term capital expenditures designed solely to build ratebase and benefit shareholders. Even under TURN's recommendation, SCE would still be allowed to recover stranded capital and would only be denied a return on that investment.

24.6 Nuclear / Palo Verde Nuclear Generating Station

SCE proposes to collect \$83.104 for O&M costs in 2025 relating to the Palo Verde Generating Station (PVGS). Based on a review of SCE's testimony and discovery responses, TURN recommends two adjustments to the 2025 forecast. First, TURN proposes to reduce nonlabor O&M by 6% to correct for sustained historic overforecasting of PVGS O&M costs. TURN also recommends that PVGS costs be tracked in a balancing account with overspending limited to 110% of the forecast value to ensure that only recorded costs are recovered from ratepayers. Second, TURN opposes SCE's efforts to overturn the Commission's longstanding requirement that shareholders bear 50% of the costs of Nuclear Energy Institute (NEI) trade association dues.

The combined impacts of TURN's adjustments are shown in the following table:

⁹²² Ex. SCE-16, p.82.

Palo Verde O&M - TURN adjustments to SCE forecast (\$ thousands)		
	2025	
SCE PVGS O&M Revenue Requirements (non-labor)		
Non-labor	82,780	
Labor	324	
less TURN Adjustments		
Reduce Recovery of NEI Dues by 50%	(132)	
Reduce Non-Labor O&M by 6% for overcollection	(4,967)	
Subtotal Impacts of TURN Recommendations	(5,098)	
SCE PVGS O&M After TURN Adjustments	78,006	

TURN provides support for these recommendations in the following sections.

24.6.1 Historical overforecasting justifies adjustments to PVGS ratemaking

SCE recovers practically all PVGS O&M costs as non-labor O&M since the facility is operated by Arizona Public Service on behalf of a consortium of owners (including SCE). In each GRC, SCE provides a forecast for the test year that is escalated in attrition years using the Commission-adopted Post Test Year Ratemaking mechanism.⁹²³ As noted in TURN's original and errata testimony, SCE has historically recovered far more in customer revenues than it has spent on PVGS non-labor O&M.⁹²⁴

Based on SCE testimony and data responses, TURN calculates that total overcollections between 2018-2023 amount to \$29.198 million (nominal) or 6.44% in excess of actual expenses. The following table shows overcollections (or undercollections) by year:⁹²⁵

⁹²³ Ex. TURN-13-E, p.107.

⁹²⁴ Ex. TURN-13-E, p.107.

⁹²⁵ Ex. TURN-13-E, pp.107-108, Table 34. This table has been slightly modified to adjust the amounts collected in rates for 2019 and 2020 based on responses provided by SCE (Ex. TURN-103, SCE response to TURN Data Request 103, Q16).

PVGS non-labor O&M - authorized revenues v. recorded costs				
Year	Collected in Rates	Actual Expenses	Actual - Collected (nominal k\$)	Percent Overcollection (%)
2018	82,716	77,085	-5,631	7.30%
2019	84,397	76,146	-8,251	10.84%
2020	86,753	73,106	-13,647	18.67%
2021	71,992	73,138	1,146	-1.57%
2022	75,184	74,700	-484	0.65%
2023	81,779	79,446	-2,333	2.94%
Total	482,821	453,621	-29,200	6.44%

The historic mismatch between SCE's forecast-based revenue requirements and its actual costs resulted in SCE shareholders retaining \$29.198 million in excess funds. This gross overcollection should not be allowed to continue.

In rebuttal testimony, SCE asserts that the overcollections in 2018-2020 were the result of planned headcount attrition combined with the impacts of the COVID-19 pandemic.⁹²⁶ It is not clear why SCE failed to accurately incorporate planned headcount reductions into the forecast applicable to 2018 and 2019 and whether similarly poor forecasting could recur over the current GRC cycle. The impacts of COVID on PVGS costs would only occur starting in 2020. Regardless of the reason, SCE was able to reap a windfall for its shareholders over the course of this entire period. SCE also tries to minimize its over-collection of \$1.644 for 2021-2023 and implies that overcollections will not recur during the current GRC cycle but fails to provide any evidentiary basis for the Commission to reach this conclusion.⁹²⁷

⁹²⁶ Ex. SCE-16, p.129.

⁹²⁷ Ex. SCE-16, p.131.

In rebuttal, SCE attempts to introduce new 2024 "recorded" data to show the potential for an undercollection in the current year.⁹²⁸ However, this information does not reflect recorded 2024 costs but rather the 2024 "budget" adopted by the PVGS co-owners prior to the start of the year.⁹²⁹ In a data response to TURN, SCE admitted that "actual Palo Verde O&M Non-Labor expenses incurred during 2024 may vary relative to the budgeted expense."⁹³⁰ The Commission should give little weight to the 2024 budget especially since current year recorded cost data is not being evaluated in this proceeding.

TURN recommends that the Commission address these consistent overcollections by reducing SCE's forecasted non-labor expense for PVGS by 6%.⁹³¹ This adjustment will bring SCE's cost recovery into line with actual PVGS O&M costs. As an alternative, the Commission could establish a balancing account to track actual PVGS operating costs and revenue collection related to PVGS non-labor O&M and refund any overcollections to customers.⁹³² To ensure that ratepayers have a degree of protection against unreasonable PVGS cost escalation, SCE should also be limited to recovery of no more than 110% of the forecasted costs for PVGS in any one year. If SCE's share of PVGS costs exceeds the 110% cap, SCE would have the opportunity to come to the Commission in the next GRC to demonstrate the reasonableness of the costs above the cap.

⁹²⁸ Ex. SCE-16, p.132, Table V-18.

⁹²⁹ Ex. TURN-700, SCE response to TURN Data Request 117, Q59.

⁹³⁰ Ex. TURN-700, SCE response to TURN Data Request 117, Q59.

⁹³¹ Ex. TURN-13-E, p.108.

⁹³² Ex. TURN-13-E, p.108.

24.6.2 The Commission should continue to require shareholders to pay for 50% of Nuclear Energy Institute dues

SCE asks the Commission to reverse its longstanding policy on trade association dues by allowing full rate recovery of dues paid to the Nuclear Energy Institute (NEI). SCE pays NEI dues through its partial ownership of PVGS and in its role as a majority owner of the San Onofre Nuclear Generating Station (SONGS). NEI dues payments relating to PVGS are authorized in the GRC while NEI dues relating SONGS are authorized in the Nuclear Decommissioning Cost Triennial Proceeding (NDCTP). PVGS's total dues for NEI are \$1.812 million and SCE's allocation of that amount (a 15.8% share) is \$0.287 million.⁹³³

For PVGS, SCE originally requested 100% of its share of NEI dues in this case, apparently forgetting that it had argued in the last GRC that it should not recover the small portion of NEI dues classified as "lobbying expenses" along with "voluntary payments to the Foundation for Nuclear Studies".⁹³⁴ SCE only recalled its prior position supporting exclusion of this portion of NEI dues from rate recovery after receiving data requests from TURN seeking clarifications on prior NEI dues payments.⁹³⁵ For SONGS, SCE recently signed onto a settlement agreement that commits its shareholders to cover 50% of NEI dues attributable to the San Onofre Nuclear Generating Station (SONGS).⁹³⁶

⁹³³ Ex. TURN-13-E, p.104; Ex. SCE-WPSCE05V01S, p. 34.

⁹³⁴ Ex. TURN-13-E, p.104; Ex. SCE-05v1, p.288; D.21-08-036, pp.365-366.

⁹³⁵ Ex. TURN-13-Atch1, SCE response to TURN Data Request 80, Q32.

⁹³⁶ Ex. TURN-13-E, p.104; Ex. TURN-13-Atch1, Joint Motion for Approval of Settlement Agreement Between SCE, SDG&E, A4NR, CalPA and TURN, A.22-02-016, Filed May 3, 2023, Section 6 (SONGS NEI fees). On May 29, 2024, the Commission issued a Proposed Decision that would adopt the settlement without modification.

TURN recommends that the Commission enforce its longstanding policy of requiring nuclear utilities to remove half the costs of NEI dues from rates. This adjustment would reduce PVGS non-labor O&M by \$0.144 million in the 2025 Test Year.⁹³⁷

24.6.2.1 Historical Commission treatment of NEI dues

For the past two decades, the Commission has consistently adopted a 50/50 sharing of NEI dues between ratepayers and shareholders based on the recognition that the organization has a dual role of promoting nuclear power and working to cut industry costs. In D.06-05-016, the Commission first adopted TURN's recommendation to assign 50% of NEI dues to shareholders based on the fact that "the principle focus on NEI appears to be the advocacy of nuclear power, both nationally and globally."⁹³⁸ The Commission found that "there are many aspects of such furtherance of the nuclear industry that may not be appropriate for ratepayer funding" and noted that SCE failed to provide information in its prepared testimony "on specific activities and related benefits that accrue to the company and/or ratepayers."⁹³⁹

Despite SCE's claims in that proceeding that all advocacy costs were included in separately disclosed lobbying expenditures, the Commission explained that "we are not convinced that all public policy advocacy costs are reflected as lobbying and excluded from SCE's forecast."⁹⁴⁰ In the event that a different allocation of NEI dues is requested in a future GRC, the Commission directed SCE to "provide more detailed descriptions of the activities, the associated costs, and the resulting company and ratepayer benefits."⁹⁴¹ Absent such details, the

⁹³⁷ Ex. TURN-13-E, p.107.

⁹³⁸ D.06-05-016, p.35.

⁹³⁹ D.06-05-016, p.35.

⁹⁴⁰ D.06-05-016, p.35.

⁹⁴¹ D.06-05-016, p.35.

Commission determined that a 50/50 split of NEI dues between shareholders and ratepayers was reasonable.⁹⁴²

The following year, the Commission affirmed this treatment in a PG&E General Rate Case by approving a settlement with a 50/50 split of NEI dues.⁹⁴³ In the intervening years since these two Decisions, no utility has come forward with a "detailed description" of NEI activities, costs and benefits to ratepayers that has resulted in a change to the 50/50 assignment of NEI dues.

This approach was further affirmed in SCE's 2009 and 2015 GRCs.⁹⁴⁴ In the 2009 GRC,

the Commission explained that

SCE fails to establish that all the benefits of its NEI membership go to its customers. For instance, NEI engages in work that furthers the interests of the nuclear industry. Such work (for example, public relations and image advertising) may not be appropriate for ratepayer funding. SCE estimates that approximately 15% of membership fees are for these types of activities. Other work performed by NEI may benefit the industry rather than ratepayers. For example, DRA points out that "ratepayers should not be paying . . .to support NEI as it goes about '[s]tudying nuclear energy's intrinsic economic value to promote a general understanding of the value of nuclear power by policymakers and the public; and [b]uilding the next generation of nuclear power plants and technologies."" SCE fails to address the amount of resources allocated to these types of studies. Accordingly, while SCE made further efforts to describe how the work performed by NEI benefits ratepayers, the extent to which NEI work benefits ratepayers versus the members of the nuclear generation industry remains unclear. We adopt DRA's recommendation to continue our policy set forth in D.06-05-016 of authorizing SCE to recover half of its share of NEI fees, \$268,000.⁹⁴⁵

In SCE's 2018 GRC, the Commission found that SCE's submission of an Edison Electric

Institute membership invoice providing guidance for allocating dues payments between

shareholders and ratepayers was "insufficient evidence to establish the portion of the invoice

⁹⁴² D.06-05-016, Finding of Fact 10.

⁹⁴³ D.07-03-044, p.106.

⁹⁴⁴ D.09-03-025, pp.12-13, D.15-11-021, p.16.

⁹⁴⁵ D.09-03-025, pp.12-13.

which should be recovered from ratepayers."⁹⁴⁶ Absent a clear demonstration of the portion of dues that support beneficial services provided to ratepayers, the Commission found that SCE failed to satisfy its burden of proof and limited ratepayer recovery to 50% of the costs.⁹⁴⁷

In its 2021 GRC, SCE again asked the Commission to reverse its policy and grant full recovery of NEI dues (net of lobbying expenditures) for PVGS. After reviewing TURN's opposition, the Commission denied SCE's request and reaffirmed the 50/50 split between ratepayers and shareholders.⁹⁴⁸ In explaining the rejection of SCE's request, the Commission noted that "NEI engages in advocacy activities that extend beyond the activities classified as lobbying under Section 165(e)(1). It is unclear what portion of NEI membership dues fund these advocacy activities. It is also unclear to what extent ratepayers as opposed to the industry benefit from these advocacy activities."⁹⁴⁹ In response to SCE's claim that NEI provides "substantial cost-savings benefits for customers", the Decision found that "SCE fails to establish that all the benefits of NEI membership go to ratepayers."⁹⁵⁰

There is no reason to deviate from the Commission's historic policy. As explained in the following sections, SCE fails to provide sufficient evidence to satisfy the standards laid out in prior Decisions and is unable to show the portion of NEI dues used to support various activities that may provide any ratepayer benefit. Evidence provided by TURN shows that NEI's commitment to industry promotion and nuclear power advocacy continues to be a major focus of its work.

⁹⁴⁶ D.19-05-020, p.250.

⁹⁴⁷ D.19-05-020, p.250.

⁹⁴⁸ D.21-08-036, Findings of Fact 448, 449, 450.

⁹⁴⁹ D.21-08-036, p.367.

⁹⁵⁰ D.21-08-036, p.366.

24.6.2.2 SCE fails to identify the portion of PVGS NEI dues used to support different types of activities including advocacy and promotion

In this case, SCE makes another attempt to persuade the Commission to allow full recovery of NEI dues by offering a list of past NEI activities that may have yielded ratepayer benefits. SCE's direct testimony acknowledges that "many NEI members benefit from its lobbying and public policy advocacy functions" while asserting that NEI also provides many "functions that support nuclear plant operations", identifying some high-level categories of NEI work, and citing three historic initiatives undertaken by NEI that produced cost savings for nuclear facility owners.⁹⁵¹ SCE also provides a copy of the "NEI Member Value Overview" document prepared by NEI that lists past activities claimed to have yielded cost savings for various nuclear facility owners including activities "dedicated to supporting members".⁹⁵²

Amongst the benefits of NEI membership touted by SCE is access to the Personal Access Data System (PADS) that "is used to support decisions to grant, deny, or revoke unescorted access to the protected areas of operating nuclear power plants".⁹⁵³ SCE fails to mention that access to PADS involves a supplemental contribution (\$14,691 in 2024 for SCE share) collected outside of the base NEI dues.⁹⁵⁴ Because this discrete expenditure clearly benefits PVGS and may lower costs for ratepayers, TURN does not oppose that supplemental contribution being fully recovered in rates.

TURN does not dispute that NEI provides services to its members, advocates for less burdensome federal regulations, and undertakes other activities intended to lower costs for

⁹⁵¹ Ex. SCE-05v1, pp.289-290.

⁹⁵² Ex. SCE-WP05v1, pp.341-345.

⁹⁵³ Ex. SCE-05v1, p.290.

⁹⁵⁴ Ex. TURN-700, SCE response to TURN Data Request 117, Q56

nuclear facility owners. However, NEI also devotes substantial resources to activities that are focused on industry growth, market development, export promotion, investor outreach and grassroots advocacy in favor of nuclear energy. SCE witness Cameron agreed that NEI devotes resources to public promotion of the benefits of nuclear energy, stating "I believe that's articulated on their website and in their mission statement".⁹⁵⁵

SCE did not provide a budget of NEI activities showing an allocation of costs by program area, a summary of membership dues received by different classes of members, or any other information that could be used to assess the relative use of resources for different functional organizational areas. When asked whether NEI provides any detailed budget information indicating expenditures by program area, SCE witness Cameron stated "I don't believe so, no."⁹⁵⁶ Mr. Cameron further acknowledged that the only detailed information regarding NEI's budget provided to SCE comes in the form of its annual invoice of membership dues.⁹⁵⁷ As shown in TURN's testimony, this invoice does not contain any budget or cost breakdown apart from showing total dues, voluntary contributions to the Foundation for Nuclear Studies, and the percentage of dues that NEI estimates are attributable to lobbying expenses pursuant to 6033(e) of the Internal Revenue Code.⁹⁵⁸

In rebuttal testimony, SCE pointed to the value provided by NEI in the form of "actively advocating for a waiver process...to allow Russian fuel currently in the supply chain to be used."⁹⁵⁹ It appears that NEI's primary role in this effort was centered around the passage of Congressional legislation that banned Russian uranium imports but allows the Department of

⁹⁵⁵ Transcript, May 7, p.460.

⁹⁵⁶ Transcript, May 7, p.459

⁹⁵⁷ Transcript, May 7, p.460

⁹⁵⁸ Ex. TURN-13-Atch1, SCE response to TURN Data Request 80, Q34, Attachment.

⁹⁵⁹ Ex. SCE-16, p.123.

Energy to issue waivers under certain circumstances.⁹⁶⁰ TURN does not believe that NEI's advocacy relating to Congressional legislation should be treated as a ratepayer-funded cost.

In rebuttal testimony, SCE acknowledges that NEI undertakes an array of activities on behalf of the overall nuclear power and technologies industry but asserts that "each NEI member only pays for the types of services it receives through its NEI membership" and references "a tiered schedule of membership dues" designed to accomplish this goal.⁹⁶¹ No additional information about "tiered" dues was provided by SCE. In response to TURN data requests, SCE conceded that NEI does not actually provide any breakdown of the services and benefits associated with each tier of membership dues.⁹⁶² Under cross examination, SCE witness Cameron (who authored SCE's rebuttal testimony) could not answer key questions regarding the system of "tiered dues", stated "I don't really have visibility to all the different sorts of memberships" and declined to "speculate" about the different tiers of membership.⁹⁶³

SCE further states that NEI dues for PVGS "do not pay for lobbying or advocacy that benefit other NEI constituencies such as advanced reactor design or nuclear medicine technologies."⁹⁶⁴ SCE provided no specific documentation to support this claim. When TURN sought relevant written materials provided by NEI, SCE conceded that "NEI does not have any written prohibitions against using dues from one membership constituency to benefit another membership constituency."⁹⁶⁵

 ⁹⁶⁰ Ex. TURN-701, NEI Press Release ("NEI CEO Maria Korsnick on the Prohibiting Russian Uranium Imports Act"), Nuclear Newswire article ("Bill to Ban Russian Uranium Imports Heads to White House")
 ⁹⁶¹ Ex. SCE-16, p.125

⁹⁶² Ex. TURN-700, SCE response to TURN Data Request 117, Q57(a).

⁹⁶³ Transcript, May 7, pp.462-463.

⁹⁶⁴ Ex. SCE-16, p.126.

⁹⁶⁵ Ex. TURN-700, SCE response to TURN Data Request 117, Q57(b).

SCE's testimony falls far short of the standard established by the Commission in past Decisions. Notably, SCE has not demonstrated what portion of NEI dues fund non-lobbying advocacy activities, the extent to which the benefits of NEI membership accrue to ratepayers versus the nuclear industry and investors, and a showing that "all the benefits of NEI membership go to ratepayers."⁹⁶⁶ SCE's inability to provide this showing, along with opaque nature of NEI's programs and budgets, fails to justify a change in Commission ratemaking policy.

24.6.2.3 NEI's own materials demonstrate a significant commitment to ongoing advocacy outside the scope of "lobbying" that does not benefit ratepayers

A review of both public and confidential NEI materials demonstrates substantial ongoing advocacy that falls outside the limited scope of activities classified by the IRS as "lobbying." The portion of SCE's NEI dues attributable to "lobbying expenses" only includes spending covered by Internal Revenue Code §162(e)(1).⁹⁶⁷ The IRS definition of lobbying is limited to activities designed to directly influence legislation, support a candidate for elected office, influence election outcomes, or involve direct communications with senior executive branch officials regarding agency actions.⁹⁶⁸ The limited scope of activities classified as "lobbying" does not include any general advocacy outside of the specifics referenced in §162(e)(1). The NEI membership invoice for PVGS, which includes a percentage of dues attributable to "lobbying", does not demonstrate the portion of NEI's budget devoted to advocacy and industry promotion.

⁹⁶⁶ D.21-08-036, p.366.

⁹⁶⁷ Ex. TURN-13-E, p.105; Ex. TURN-13-Atch1, SCE response to TURN Data Request 91, Q3(a) in A.19-08-013.

⁹⁶⁸ 26 USC §162(e)(1).

TURN's testimony and attachments identify a range of NEI activities that focus on building support for the new and expanded use of nuclear energy through outreach to a range of organizations, efforts to engage the public, and strategies for shaping the overall narrative regarding the benefits of nuclear power.⁹⁶⁹ NEI's website includes one section devoted to "the advantages of nuclear energy"⁹⁷⁰ and another section describing NEI's efforts to promote the development and deployment of new ("advanced nuclear") plants.⁹⁷¹ Another focus of NEI's work is outreach to the financial community with the goal of encouraging investment in nuclear vendors and new plants.⁹⁷² The "Advocacy" portion of NEI's website focuses on efforts to promote exports of nuclear technology manufactured by domestic companies that are presumably NEI members.⁹⁷³ NEI also funds a podcast named "Fissionary" that produces episodes explaining how nuclear energy can "solve the climate crisis and secure our energy independence".⁹⁷⁴ Additionally, NEI recently announced a new "Generation" advertising campaign⁹⁷⁵ designed to shift public opinion on nuclear energy and urging viewers to "choose nuclear energy".⁹⁷⁶

⁹⁶⁹ Ex. TURN-13-E, pp.105-106; Ex. TURN-13-Atch1, Section II (Nuclear Energy Institute Documents), PDF pp.228-290.

⁹⁷⁰ Ex. TURN-13-E, p.105 (referencing https://www.nei.org/advantages)

⁹⁷¹ Ex. TURN-13-E, p.105; Ex. TURN-13-Atch1, Section II (Nuclear Energy Institute Documents), PDF pp.245-249.

⁹⁷² Ex. TURN-13-E, p.105; Ex. TURN-13-Atch1, Section II (Nuclear Energy Institute Documents), PDF pp.235-240.

⁹⁷³ Ex. TURN-13-E, p.105; Ex. TURN-13-Atch1, Section II (Nuclear Energy Institute Documents), PDF pp.229-231.

⁹⁷⁴ Ex. TURN-13-E, p.106; Ex. TURN-13-Atch1, Section II (Nuclear Energy Institute Documents), PDF p.232.

⁹⁷⁵ Ex. TURN-13-E, p.106; Ex. TURN-13-Atch1, Section II (Nuclear Energy Institute Documents), PDF pp.233-234.

⁹⁷⁶ Ex. TURN-13-E, p.106 (link to https://www.youtube.com/watch?v=EmFtgoVk0fA)



NEI's 2023 Annual Plan provides additional insights into the organization's basic

⁹⁷⁷ Ex. TURN-13-Atch2C, NEI 2023 Annual Plan Summary, page 3.
⁹⁷⁸ Ex. TURN-13-Atch2C, NEI 2023 Annual Plan Summary, page 3.
⁹⁷⁹ Ex. TURN-13-Atch2C, NEI 2023 Annual Plan Summary, page 3.
⁹⁸⁰ Ex. TURN-13-Atch2C, NEI 2023 Annual Plan Summary, page 4.

981
982
·
983
984

⁹⁸¹ Ex. TURN-13-Atch2C, NEI 2023 Annual Plan Summary, page 4.
⁹⁸² Ex. TURN-13-Atch2C, NEI 2023 Annual Plan Summary, page 4.
⁹⁸³ Ex. TURN-13-Atch2C, NEI 2023 Annual Plan Summary, page 4.
⁹⁸⁴ Ex. TURN-13-Atch2C, NEI 2023 Annual Plan Summary, page 4.

Nothing in the annual plan supports the Commission's requirement that SCE demonstrate that "all the benefits of NEI membership go to ratepayers."⁹⁸⁶

985

NEI's disclosures to the IRS show significant amounts of money spent annually on grants to advocacy groups and sponsorships of various events where policymakers gather.⁹⁸⁷ NEI is a major contributor to Nuclear Matters (approximately \$2 million per year in direct grants) which describes itself as "a national coalition of grassroots advocates, working to inform the public and policymakers about the clear benefits of nuclear energy."⁹⁸⁸ Expenditures on Nuclear Matters are outside the definition of "lobbying" used by NEI. NEI's website urges visitors to join Nuclear Matters "to get updates and action alerts on how to preserve nuclear energy for future generations."⁹⁸⁹ NEI also sponsors many conferences and provides direct grants to an array of organizations in an effort to gain support for the industry or obtain informal access to decisionmakers.⁹⁹⁰ These grants and sponsorships are not included in the IRS definition of "lobbying" but are clearly an extension of NEI's advocacy work to promote the industry it represents which include vendors, manufacturers and plant owners (like SCE) that earn profits from their investments in existing facilities.

NEI's public and private materials demonstrate that its primary focus is the promotion of nuclear power, both domestically and abroad, and activities that are designed to develop a

⁹⁸⁵ Ex. TURN-13-Atch2C, NEI 2023 Annual Plan, page 38.

⁹⁸⁶ D.21-08-036, p.366.

⁹⁸⁷ Ex. TURN-13-E, p.106; Ex. TURN-13-Atch1, Section II (Nuclear Energy Institute Documents), PDF pp.250-290.

⁹⁸⁸ Ex. TURN-13-E, p.106; Ex. TURN-13-Atch1, Section II (Nuclear Energy Institute Documents), PDF pp.241-242.

⁹⁸⁹ Ex. TURN-13-E, p.106; Ex. TURN-13-Atch1, Section II (Nuclear Energy Institute Documents), PDF pp.243-244.

⁹⁹⁰ Ex. TURN-13-E, p.106.

positive image for the industry. These activities are a central feature of NEI's operations, appear to be growing in scope and scale, and should be assumed to comprise a large portion of its budget. NEI does not provide a breakdown explaining the portion of its budget devoted to various types of advocacy, outreach, and public awareness activities. Neither SCE nor NEI have identified the portion of dues paid by individual nuclear facility owners that are allocated to the various programmatic areas of activity. Absent a far more comprehensive showing that answers these questions, there is no justification for abandoning the longstanding practice of a 50/50 split of NEI dues between shareholders and ratepayers.

25. ENERGY PROCUREMENT

- 25.1 Energy Procurement O&M
- 25.2 Energy Procurement Capital

26. ENTERPRISE TECHNOLOGY

- 26.1 Technology Planning, Design, And Support
- 26.2 Technology Delivery
- 26.3 Digital And Process Transformation
- 26.4 Service Management Office And Operations

27. OPERATING UNIT CAPITALIZED SOFTWARE

28. ENTERPRISE PLANNING AND GOVERNANCE (NON-INSURANCE)

28.1 Financial Oversight And Transactional Processing

28.2 Legal

28.3 Business And Financial Planning

Business and Financial Planning supports SCE's efforts "to develop, coordinate, and implement policies and practices that address federal and state regulatory and cost recovery requirements and related goals, as well as developing and managing business and financial operating plans and goals."⁹⁹¹ The Business Planning function within Business and Financial Planning "encompasses functions to perform integrated planning and financial forecasting for the enterprise," including "strategic planning, operations performance management, financial planning, and regulatory finance and economic forecasting."⁹⁹² SCE requests \$36.532 million for the Business Planning function, including \$28.196 million for labor and \$8.336 for non-labor, which is an increase of \$9 million over 2022 recorded costs (\$27.520 million).⁹⁹³ TURN takes issue only with SCE's non-labor forecast for Business Planning and recommends using 2022 last recorded year for non-labor (\$5.263 million).⁹⁹⁴ For the reasons provided below, the Commission should adopt TURN's Business Planning forecast of \$33.459 million, which is \$3.073 million less than SCE's request.

⁹⁹¹ Ex. SCE-17V02, p. 41.

⁹⁹² Ex. SCE-06V03, p. 73.

⁹⁹³ Ex. SCE-17V02, pp. 42, 47 (explaining that SCE reduced its original labor forecast by \$0.701 million because it identified 5 positions that will no longer be needed).

⁹⁹⁴ Ex. TURN-11 (Defever), pp. 3-4.

As TURN pointed out in testimony, SCE has consistently underspent on Business Planning non-labor since 2018, and SCE's recorded non-labor costs declined every year from 2018-2022.⁹⁹⁵ The following table illustrates both of these trends.⁹⁹⁶

Business Planning - Non Labor (Constant 2022 \$000s)					
	2018	2019	2020	2021	2022
Authorized	\$16,126	\$16,126	\$16,126	\$12,430	\$12,430
Actual	\$15,361	\$13,137	\$10,762	\$9,750	\$5,263
Underspent	\$765	\$2,989	\$5,364	\$2,680	\$7,167

SCE has struggled to accurately forecast Business Planning non-labor costs in the past. In the 2018 GRC, the Commission reduced SCE's forecast for Business Planning by \$8 million (2018 Constant \$) to account for the wide variation in outside services consulting costs from 2011-2015.⁹⁹⁷ In 2018, prior to the issuance of the 2018 GRC decision, D.19-05-020, SCE similarly incurred Business Planning expenses that were \$8.0 million (in 2018 Constant \$) or 18% less than SCE's GRC forecast.⁹⁹⁸ In the 2021 GRC, the Commission authorized SCE's uncontested forecast for Business Planning, including \$12.430 million for non-labor.⁹⁹⁹ SCE underspent its forecast in 2021 primarily because the level of outside consultants was \$4.575

⁹⁹⁵ Ex. TURN-11 (Defever), p. 3.

⁹⁹⁶ Ex. TURN-11 (Defever), p. 3. As noted in fn. 2 on page 3, the authorized amounts shown are test year authorizations without the increases provided through the post-test year adjustment mechanism authorized in the 2018 GRC for 2019 and 2020, and in the 2021 GRC for 2022.

⁹⁹⁷ Ex. TURN-401 (Excerpt from SCE's 2021 GRC Testimony, SCE-06V02), p. 77 (discussing recorded costs for Business Planning in 2018 relative to amounts authorized in the 2018 GRC decision); D.19-05-020, pp. 251-252.

⁹⁹⁸ Ex. TURN-401 (Excerpt from SCE's 2021 GRC Testimony, SCE-06V02), p. 77.

⁹⁹⁹ Ex. SCE-06V03, p. 7 D.21-08-036, p. 386.

million lower than planned, which SCE attributes to "the general uncertainty in the COVID environment."¹⁰⁰⁰

Given SCE's history of overforecasting non-labor costs for this activity from 2018-2022, the Commission should question the accuracy of SCE's forecast for 2025. Adopting the 2022 last recorded year as the non-labor forecast for Business Planning, as proposed by TURN, is reasonable in light of SCE's forecasting history and the consistent decline in non-labor spending from 2018-2022.¹⁰⁰¹

In rebuttal testimony, SCE argues that TURN overlooks the merits of SCE's adjustments to last recorded year in developing SCE's non-labor forecast. Those adjustments include an upward adjustment to maintain variable corporate consulting spend at 3-year historic levels as 2022 was atypically low, a reduction for reduced reliance on contingent workers, and an increase for consultants to support SCE's "reimagining the grid" and sustainability efforts.¹⁰⁰² To the contrary, TURN considered these adjustments in its testimony but concluded that SCE's forecast was unreasonable because of its history of underspending and declining costs.¹⁰⁰³

Moreover, it is unclear why SCE needs to retain a historical level of non-labor corporate consulting costs and add additional consulting support when SCE has requested to add

¹⁰⁰⁰ Ex. SCE-06V03, p. 77. SCE's forecasting inaccuracies took a different turn in 2023. SCE forecast non-labor costs of \$8.156 million in 2023.¹⁰⁰⁰ This forecast was wildly inaccurate, as SCE spent \$18.124 million in 2023 driven by \$13.660 million in corporate consulting, which is \$10 million higher than its forecast and exceeds non-labor expenditures every year since at least 2018. (Ex. SCE-06V03 WP, p. 199; Ex. SCE-11, Appendix A (2023 Recorded O&M); Ex. SCE-06V03 WP, p. 199; Ex. SCE-17V02, p. 46.).

¹⁰⁰¹ D.04-07-022 (SCE 2003 GRC), p. 15 (quoting D.89-12-057) ("If recorded expenses in an account have shown a trend in a certain direction over three or more years, the [last recorded year] level is the most recent point in the trend and is an appropriate base estimate for [the test year].").

¹⁰⁰² Ex. SCE-17V02, pp 45-46, referring to the adjustments discussed in Ex. SCE-06V03 at pp. 84-85. ¹⁰⁰³ Ex. TURN-11 (Defever), p. 4.

significant additional capacity and expertise to the Business Planning staff. SCE's labor request includes <u>thirteen new positions</u> in Business Planning, including <u>nine</u> additional staff who will provide "a commensurately resolute level of capabilities to fulfill our regulatory requirements, help ensure prudent resource management, and enable SCE to address evolving safety, reliability, financial, and compliance challenges in an increasingly risk-informed decision-making framework," and <u>four</u> additional staff "targeting longer term affordability, reliability and clean energy goals."¹⁰⁰⁴ This additional staffing, unopposed by TURN, should reduce the need for both types of consulting support reflected in SCE's upward forecast adjustments.¹⁰⁰⁵

The Commission should find SCE's request excessive for all of these reasons.

Accordingly, the Commission should adopt TURN's forecast of \$33.459 million for Business

Planning, which is comprised of SCE's forecast of \$28.196 for labor and TURN's non-labor

forecast of \$5.263 million.

¹⁰⁰⁴ Ex. SCE-06V03, pp. 82-84; Ex. SCE-06V03 WP, p. 203. In rebuttal testimony, SCE reduced its labor request after identifying five positions "that will no longer be needed" due to "operational efficiencies." (Ex. SCE-17V02, p. 44). These positions do not appear to be among the thirteen new positions discussed in SCE's direct testimony and workpapers, as SCE has not revised its extensive discussion of these thirteen new positions through errata.

¹⁰⁰⁵ See Ex. SCE-17V02, pp. 84-85, as modified on p. 85-E4 ("The first adjustment relates to targeted use of consultants on complex strategic issues (including issues where the consultants' knowledge of and experience with other energy companies or wider best practices are invaluable) to address emerging issues. Historically, these costs have varied on a year-by-year basis. ... Accordingly, the forecast for this category of non-labor spending is based on a 3-year average of \$3.991 million."); p. 85 ("Incremental costs of \$0.857 million for external consultants and EIX staff support are required to provide analytical support, company-wide coordination and expert knowledge on specialty areas that support California's key priorities around affordability, reliability and climate for electric customers, such as income-varying energy burden analysis, cost-benefit calculations for customer-adopted technologies, assessment of evolving reliability approaches and metrics, and analysis of new energy technologies. In addition, they will provide writing, editing and benchmarking support to prepare our annual sustainability report, assess the adoption of sustainability targets/metrics, and provide specialized services to track, manage and document sustainability data for compliance, reporting and operational purposes.").

28.4 Supply Chain Management And Supplier Diversity And Development

29. INSURANCE

29.1 Liability Insurance (Wildfire)

In D.23-05-013, the Commission adopted an agreement submitted in SCE's test year 2021 GRC proceeding to implement a self-insurance alternative for SCE's wildfire liability insurance coverage. TURN was a sponsoring party to that agreement, having reached the conclusion that under current market conditions a commitment to a self-insurance structure would better serve ratepayer interests than would continuing to rely on third-party commercial insurance products. TURN's prepared testimony here supported maintaining the self-insurance structure for wildfire liability claims rather than purchasing insurance products procured from third parties through the 2025 GRC period.¹⁰⁰⁶ On March 25, 2024, TURN joined with SCE and Cal Advocates in a motion requesting that the Commission approve and adopt an early decision in this proceeding extending the wildfire self-insurance program through the 2025 GRC period. On July 1, 2024, the Commission issued a proposed decision that would grant the joint motion without modification and is on the consent agenda for the July 11, 2024 meeting.

TURN very much appreciates the efforts of the ALJs and supporting staff whose efforts enabled preparation and issuance of this decision. With its issuance, there appear to be no further issues related to wildfire liability insurance that need to be addressed at this time in this proceeding.

¹⁰⁰⁶ Ex. TURN-15-E2, pp. 30-31.

29.2 Liability Insurance (Non-Wildfire)

TURN's prepared testimony challenged SCE's proposed forecast of \$79.2 million for non-wildfire liability insurance expense in the 2025 test year, and instead recommended adoption of a forecast of \$48.1 million based on the 2022 recorded expense.¹⁰⁰⁷ During the period between service of rebuttal testimony and the scheduled cross-examination on insurance-related issues, TURN, Cal Advocates and SCE successfully negotiated a proposed stipulation that, if adopted, would resolve all disputed non-wildfire insurance issues, including the non-wildfire liability insurance forecast amount.¹⁰⁰⁸ The proposed stipulation addresses the funding for nonwildfire liability and property insurance together, and would adopt a forecast of \$82.27 million for all such costs (as compared to the combined forecast of approximately \$104 million proposed by SCE, \$67.8 million proposed by TURN, and \$78.2 million proposed by Cal Advocates). The funding would also be subject to balancing account treatment through a new two-way General Liability & Property Insurance Balancing Account (GL&PBA).

The Commission should find the stipulation reasonable and adopt it without modification. While the forecast reduction as compared to SCE's request is not as large as TURN's testimony had called for, it represents a reasonable compromise of competing positions supported in the evidentiary record. And though TURN has called on the Commission to reduce its reliance on balancing and memorandum accounts as a general matter, TURN believes this new account is acceptable under the circumstances here and as part of the several compromises reflected in the stipulation.

¹⁰⁰⁷ Ex. TURN-15-E2, pp. 28-30.

¹⁰⁰⁸ Ex. SCE-34 (Stipulation of TURN, Cal Advocates and SCE on Non-Wildfire Insurance).

29.3 Property Insurance

TURN's prepared testimony challenged SCE's proposed forecast of \$25.2 million for property insurance expense in the 2025 test year, and instead recommended adoption of a forecast of \$19.6 million based on the 2022 recorded expense escalated by 10% per year.¹⁰⁰⁹ As described in the preceding section, the property insurance expense forecast is covered by the stipulation supported by TURN, Cal Advocates and SCE.¹⁰¹⁰ For the reasons set forth in the preceding section, TURN urges the Commission to find the stipulation reasonable and adopt it without modification.

30. EMPLOYEE BENEFITS, TRAINING AND SUPPORT

The Commission should adopt TURN's adjustments to SCE forecasts for its Short-Term Incentive Program (STIP), Long-Term Incentives Program (LTIP), 401(k), Medical benefits programs, and training programs, which are summarized in the following table.

¹⁰⁰⁹ Ex. TURN-15-E2, pp. 26-28.¹⁰¹⁰ Ex. SCE-34.

 Table 9: Summary of TURN Employee Benefits & Training Recommendations

 (\$1,000s)

Catagowy	2025			
Category	SCE	TURN	SCE > TURN	
STIP*	120.406	73.447	46.958	
LTIP	22.017	0	22.017	
401(k) (Nominal\$)*	132.041	126.895	5.146	
Medical Programs (Nominal)*	153.788	132.919	20.869	
Training Seat Time	37.023	28.511	8.512	
Training Delivery	23.198	17.872	5.326	
* In an anna (1 (02/21/24)). COE atota datat TUDN abased darata dara farada				

* In an email (02/21/24), SCE stated that TURN should rely on the fourth version of the RO Model for the Compensation and Benefits forecasts that it is including in the GRC. Here, tht includes the forecasts for STIP, 401(k), and Medical Programs, for which the reported SCE forecast is different than in SCE's Direct Testimony (see Ex. TURN-14-E2, pp. 9, 29, and 33).

30.1 Employee Support

TURN, Cal Advocates and SCE reached a stipulation that reflects a complete resolution

of disputed Employee Support issues, presented in Exhibit SCE-31. The Commission should

find that this stipulation is reasonable in light of the testimony submitted, consistent with law,

and in the public interest.

30.2 Employee Benefits & Programs

30.2.1 Short-Term Incentive Program (STIP) and Executive Incentive Compensation Program (EICP)

SCE's forecast for the combined, Short-Term Incentive Program (STIP) and Executive

Incentive Compensation Program (EICP) is \$120.406 million.¹⁰¹¹ SCE refers to the STIP and

¹⁰¹¹ Ex. TURN-14-E2, Prepared Testimony of Garrick Jones Addressing Employee Benefits, Training & Support, Errata filed April 30, 2024, p. 9 (referencing RO Model (v. 1.4), O1) O&M Dashboard', 'STIP'
EICP interchangeably and forecasts the two programs as STIP. For consistency, TURN will refer to the programs and forecast as STIP in this brief. TURN's forecast reduces the STIP forecast by \$46.958 million, for a forecast of \$73.447 million. Please note that this 39.1% reduction should be applied to the RO Model calculation, given that SCE's STIP forecast will ultimately depend on reductions that the Commission makes to the labor force in its GRC decision.

Table 10: Summary of TURN's Recommendation Regarding SCE's STIP Cost Forecast (\$1,000s)

Category	SCE	TUDN	SCE >	
	SCE	IURN	TURN	
STIP	120,406	73,447	46,958	

STIP is the annual variable pay program that gives employees and executives (through

the EICP) an opportunity to earn a cash award based on achieving company goals and individual

performance¹⁰¹² (with the second depending on the score achieved within the Individual

Performance Modifier (IPM)¹⁰¹³). As explained in Exhibit TURN-14,

(T)he amount that SCE pays in STIP in a given year, i.e., the STIP pool, is the sum of each eligible employee's STIP opportunity percentage multiplied by the employee's eligible earnings.¹⁰¹⁴ The STIP pool is determined and fixed by the opportunity percentage (after possible modification based on the company's performance against STIP goals, called the Company Multiplier)¹⁰¹⁵) and eligible earnings. The STIP pool is not affected by IPM machinations.¹⁰¹⁶

¹⁰¹⁴ Id.

¹⁰¹⁵ Id.

¹⁰¹⁶ Id.

tab. In an email (02/21/24), SCE stated that TURN should rely on the fourth version of the RO Model for the Compensation and Benefits forecasts that it is including in the GRC.)

¹⁰¹² Ex. SCE-06 Vol. 4, p. 63:2-3.

¹⁰¹³ Ex. TURN-14-E2, p. 10, referencing 2023 Short-Term Incentive Plan (pp. 2-3), provided as an attachment to DR TURN-SCE-009-1, provided in Ex. TURN-14-Atch1.

SCE also plans to convert a portion of its STIP compensation to Base Pay. TURN addresses the impact of this proposal Section 30.2.1.4 and Section 30.2.3.1 below.

30.2.1.1 Consistent with the Majority of Past Commission GRC Decisions, SCE Shareholders Should be Responsible for a Portion of STIP Costs because they Benefit from the Program

SCE's ratepayers should only be required to fund the STIP measures that primarily benefit ratepayers. If STIP measures are focused on shareholder benefits, shareholders should pay for the associated costs. STIP measures that provide benefits to both ratepayers and shareholders, should be paid for by both ratepayers and shareholders. This approach to STIP has been implemented by the Commission in numerous GRCs, including in Decision 14-08-032 where the Commission found it appropriate that "ratepayers bear reasonable costs for funding [ICP] metrics in relation to the benefits derived."¹⁰¹⁷ In recent GRC Decisions, the Commission rejects the utilities' argument that ratepayers should fund 100% of their short-term incentive program as part of cost of service. As noted in Exhibit TURN-14, in SCE's last GRC (TY 2021), SCE argued that variable pay should be included in rates because it is an important element of an overall total compensation package that is at market.¹⁰¹⁸ The Commission disagreed with SCE, explaining:

[T]he Commission has repeatedly rejected arguments that cost-of-service ratemaking principles require ratepayers to fully fund incentive compensation where elements of the program essentially benefit shareholders without a clear demonstrable benefit to ratepayers, including in cases where the utility has argued that the total compensation package was at market.[citation omitted] The Commission has explained that "the sharing of cost responsibility promotes a reasonable matching of costs with benefits experienced both by ratepayers and

¹⁰¹⁷ D.14-08-032, p. 522.

¹⁰¹⁸ Ex. TURN-14-E2, p. 12, citing D.21-08-036, p. 428.

shareholders."[citation omitted] The Commission has also noted that it is within SCE management's discretion to target incentive compensation to achieve ratepayer benefits.¹⁰¹⁹

TURN's Testimony, Exhibit TURN-14, addressed how certain STIP measures partially,

primarily, or exclusively benefit shareholders and accordingly should be wholly or partially

funded by shareholders.¹⁰²⁰ TURN's recommendations for ratepayer funding levels for SCE's

various STIP/EICP goals in summarized in the following table.

	SCE		TURN				
			GRC-				
STIP Goals		Forecasted	Recoverable	Forecasted			
	2023 Target	Payout	Forecast	Payout			
	Weight ¹	(\$1,000s)	Weight ²	(\$1,000s)			
Safety & Resiliency	55.0%	66,223	47.0%	56,591			
Covered Conductor	6.0%	7,224	3.0%	3,612			
Capital Deployment	5.0%	6,020	0.0%	-			
Remaining Safety & Resiliency Goals	44.0%	52,978	44.0%	52,978			
Performance Mgmt & Op. Excellence	45.0%	54,183	14.0%	16,857			
Core Earnings	25.0%	30,101	0.0%	-			
Clean Energy Transition	4.0%	4,816	0.0%	-			
Operational Excellence (Catalyst)	4.0%	4,816	2.0%	2,408			
Remaining Performance Mgmt & Op.	12.0%	14,449					
Excellence Goals			12.0%	14,449			
Total	100.0%	120,406	61.0%	73,447			
SCE > TURN (\$1,000s) => 46,958							
¹ WPSCE06V04BkB, p. 10.							

Table 11: Comparison of STIP/EICP Forecast, SCE versus TURN1021

² Justification for the reductions to the GRC-recoverable percentage can be found in the foregoing testimony.

¹⁰¹⁹ Id. Similar conclusions can be found for Sempra's 2012 GRC (*see* D.13-05-010 at p. 882) and PG&E's 2014 GRC (*see* D.14-08-032 at pp. 520-524).

¹⁰²⁰ See Ex. TURN-14-E2, pp. 13-22.

¹⁰²¹ Ex. TURN-14-E2, Table 9, p. 23. See Table 10 for a summary of TURN's reductions based on intermediate steps included in TURN's overall STIP/EICP Recommendation.

30.2.1.1.1 Ratepayers Should Not Fund Goals that Primarily Benefit Shareholders

Core Earnings Goal (Performance Management and Operational Excellence Category, 25% of total STIP Target – TURN Ratepayer Funding Recommendation of 0%)

SCE's main argument in support of ratepayer funding for its Core Earnings Goal is "the financial health of the company is imperative to ensure SCE is able to attract investors and have access to capital for the direct benefit of its customers."¹⁰²² While this may be true, SCE has not quantified this benefit, and it is not the metric by which achievement of this goal is measures. The Commission has also evaluated virtually the same argument before in the Sempra TY 2019 GRC, and found it unpersuasive:

Applicants argue that the financial metrics provide benefits to ratepayers in the form of lower interest rates but we find that this is not substantiated or quantified by the evidence presented. We also find any benefit resulting from achieving Applicants' financial goals to be incidental and secondary to what we consider as the primary goal of the financial metrics which is to reach a certain level of income or earnings. After all, achieving a target interest level for borrowing is not one of the metrics that triggers the award.¹⁰²³

As Decision 15-11-021 succinctly explains, "[The Commission] agree[s] with [utilities

in] that financial performance may benefit ratepayers, [but] the ratepayer benefit is much less direct than the shareholder benefit. Further, in some instances, financial performance may be achieved at the detriment of ratepayers."¹⁰²⁴ TURN's testimony enumerates various ways earnings can be approved to the detriment of ratepayers, including:¹⁰²⁵

¹⁰²² Ex. SCE-06, Vol. 4, p. 65 & Ex. SCE-17, Vol. 3, p. 30.

¹⁰²³ D.19-09-051, p. 543.

¹⁰²⁴ D.15-11-021, p. 261.

¹⁰²⁵ Ex. TURN-14-E2, p. 14.

- the reduction of costs through work curtailment between rate cases accompanied by a subsequent attempt to convince the regulator to provide additional money for deferred work at the next rate case;
- development of infrastructure programs which may be of questionable value to ratepayers, but raise rate base;
- the election of tax-timing changes between rate cases that may benefit shareholders in early years at the expense of ratepayers in later years.

TURN also explains that "operating efficiencies" that can increase core earning are also not always directly beneficial to ratepayers,

"... in cases where sustained and realized efficiencies are the means of improving earnings, shareholders benefit until the next rate case, at which point ratepayers presumably begin to benefit from such efficiencies. That is, ratepayers benefit to the extent such efficiencies and the resulting cost savings are included in the forecast for the next GRC, which is not always the case."¹⁰²⁶

Further, as noted in Exhibit TURN-14, the Commission has long held that financial-based

STIP goals, such as SCE's Core Earnings STIP goal, are inappropriate for inclusion GRC-based

revenues. In SCE's last GRC (TY 2021), the Commission agreed with TURN that 100% of

financial goals should be disallowed, the Commission stated,

"As in past GRCs, we continue to find that this goal is primarily intended to benefit shareholders.[citation omitted] The goal may or may not result in secondary benefits to ratepayers since a goal of "achieving core earnings" does not always align shareholder and ratepayer interests."¹⁰²⁷

The Commission also disallowed ratepayer funding for the "financial goals metric" in D. 23-11-

069 addressing PG&E's TY 2023 GRC.¹⁰²⁸ SCE's Rebuttal Testimony does not offer any new

arguments or facts to justify the Commission modifying its long standing policy that financial-

based STIP goals should be funded by shareholders.

¹⁰²⁶ Id.

¹⁰²⁷ D.21-08-036, pp. 431-432.

¹⁰²⁸ D.23-11-069, p. 609.

Capital Deployment Goal (Safety & Reliability Category, 5% of total STIP Target – TURN Ratepayer Funding Recommendation of 0%)

SCE describes the Capital Deployment STIP goal as being related to the execution of grid, technology, electrification, and other improvements to deliver safe, reliable, clean, and affordable energy for customers, while achieving CPUC and FERC jurisdictional capital improvement plan execution, consistent with appropriate regulatory direction.¹⁰²⁹ As noted in Exhibit TURN-14, SCE has an authorized Return on (Common) Equity (ROE) of 10.05% and its total, authorized rate of return is 7.44%.¹⁰³⁰ As a result of this rate of return on capital spending, this goal directly benefits shareholders, regardless of the whether the investment is "consistent with appropriate regulatory direction".

Further, this goal is largely duplicative of the financial goal of the STIP and the LTIP, as it incentivizes capital expenditure, which is the primary driver of EIX share performance. As discussed in TURN's testimony, EIX states that the 2025 Core earnings per share (EPS) growth estimate is driven by the 11–14% 2025 rate base growth.¹⁰³¹ Further, the existence of such a goal could potentially encourage undue focus on capital-intensive solutions in both SCE's regulatory and asset-management approaches, which could be detrimental to ratepayers, as ratebase growth is one of the key factors leading to the current utility rate affordability crisis.¹⁰³²

¹⁰²⁹ Ex. WP SCE-06, Vol. 4, BkB, p. 10.

¹⁰³⁰ Decision 22-12-031, p. 1.

¹⁰³¹ Ex. TURN-14-E2, p. 18. Referencing *Fourth-Quarter And Full-Year 2023 Financial Results* (Presentation) (p. 15), February 22, 2024. Presentation is in included in Ex. TURN-14-Atch1.

¹⁰³² Ex. TURN-02E, pp. 4-6 & FN 19 referencing CPUC, "Utility Costs and Affordability of the Grid of the Future: An Evaluation of Electric Costs, Rates, and Equity Issues Pursuant to P.U. Code Section 913.1" (May 2021), p. 7.

Exhibit TURN-14 provided some examples of how this incentivization of capital spending could occur as a result of this goal:

Management may be incentivized to prefer capital-intensive solutions at the expense of O&M solutions when recommending solutions to such issues as wildfire before the Commission and when selecting management solutions in the field (e.g., a preference for undergrounding of facilities or covered conductors at the expense of a well-managed vegetation management solution when evaluating how to approach the wildfire implications of a particular wire).¹⁰³³

For the foregoing reasons, ratepayers should not fund the cost of the Capital Deployment STIP goal. The Commission should apply its long standing policy that financial-based STIP goals should be funded by shareholders to this goal.

Clean Energy Transition (Performance Management and Operational Excellence Category, 4% of total STIP Target – TURN Ratepayer Funding Recommendation of 0%)

SCE describes the Clean Energy Transition as a goal to "[a]dvance electric technology adoption to enable emissions reductions across economic sectors [by a]dvanc[ing] SCE's clean energy pathway objectives [of] Transportation Electrification (TE) charging port installations ... and medium/heavy duty electric vehicle conversions ... [and] Building Electrification (BE) heat pump installs"¹⁰³⁴ Similar to the Capital Deployment goal discussed above, shareholders derive more benefits from this capital-deployment-focused goal than ratepayers. While supporting TE and BE is consistent with California's climate change mitigation targets, the investments incentivized in this goal come with significant capital spending that SCE's shareholders will earn a rate of return on for decades to come.¹⁰³⁵ Exhibit TURN-07 also

¹⁰³³ Ex. TURN-14-E2, p. 17.

¹⁰³⁴ Ex. TURN-14-E2, p. 18. Referencing *Fourth-Quarter And Full-Year 2023 Financial Results* (Presentation) (p. 15), February 22, 2024. Presentation is in included in Ex. TURN-14-Atch1.

¹⁰³⁵ See Ex. TURN-07E (Ashford) for a discussion of the significant capital expenditures SCE forecasts to support transportation electrification load growth.

highlights issues with SCE's over-forecasting for new TE load, indicating that SCE employees do not necessarily need to be incentivized to meet this goal.¹⁰³⁶ Further, as addressed in the Section 4.2.2 of this brief, much of the TE related spending is designed to support medium/heavy duty electric vehicle fleets owned by large corporations who will eventually reap financial benefits from electrification; but the grid upgrades are funded by all SCE customers, including residential ratepayers, many of whom are struggling to afford necessary electric service. Asking SCE ratepayers to fund incentive payments for SCE executives to encourage more capital spending adds to this inequity.

Exhibit TURN-14 addresses other policy issues with this goal.¹⁰³⁷ For the foregoing reasons, the Commission should find this goal primarily benefits shareholders, and is not appropriate for ratepayer funding.

30.2.1.1.2 The Cost of the Following Goals Should be Reduced Based on their Benefits to Ratepayers

TURN identifies the following goals that are constructed in such a way that they are contrary to, or do not primarily or only partially support ratepayer goals, and accordingly should not be fully funded by ratepayers.

Covered Conductor (Safety and Resiliency Category, 6% of total STIP Target – TURN Ratepayer Funding Recommendation of 3%)

The Covered Conductor STIP goal is based on the annual number of covered conductor circuit miles installed.¹⁰³⁸ Similar to the Capital Deployment goal discussed above, it is

¹⁰³⁶ Ex. TURN-07-E, pp. 6-20.

¹⁰³⁷ Ex. TURN-14-E2, p. 18.

¹⁰³⁸ Ex. WP SCE-06, Vol. 4, BkB, p. 10.

generally inappropriate for ratepayers to fund capital-deployment-focused goals as these goals provide significant benefits to shareholders. However, the Covered Conductor STIP goal is more appropriate for some ratepayer funding than the other capital deployment goals (i.e., general Capital Deployment and Clean Energy Transition) as "it is combined with a suite of other Wildfire-related goals that comprise a results-oriented goal (i.e., CPUC reportable ignitions) and several O&M-related goals that seem to work in concert with the strictly capitalrelated goal of Covered Conductors."¹⁰³⁹

Exhibit TURN-14 discusses examples of SCE touting the rate base growth benefits of

wildfire mitigation work to its shareholders:

It is important to note that the utility views the capital investment in wildfire (and clean-energy transmission) with enthusiasm in its earnings calls, stating, "Projected ~6–8% rate base growth 2023–2028; substantial additional investment opportunities offer upside[.] Strong rate base growth driven by wildfire mitigation and important grid work to support California's leading role in clean energy transition. ... 2025 Core EPS growth primarily driven by rate base earnings." Furthermore, on a page on which SCE discussed its double-digit total return potential with investors, the only specific utility activities that SCE raised were wildfire mitigation and electrification, both capital-intensive programs and the only programs for which Edison codifies capital-program specific, STIP goals.¹⁰⁴⁰

SCE does not specifically address this evidence in its rebuttal testimony, and generally argues

that in regards to capital deployment goals, "... shareholder and customer interests are aligned

¹⁰³⁹ Ex. TURN-14-E2, p. 19.

¹⁰⁴⁰ Id. Referencing *Fourth-Quarter And Full-Year 2023 Financial Results* (Presentation) (p. 12 & 18), February 22, 2024. Presentation is in included in Ex. TURN-14-Atch1 (Attachment 2).
¹⁰⁴¹ Ex. SCE-17, Vol. 3, p. 32.

For the Covered Conductor goal, TURN agrees with SCE that shareholder and ratepayers both benefit, and accordingly shareholders and ratepayers should <u>evenly share</u> the cost of the Covered Conductor goal. This cost sharing acknowledges the shareholder benefits of this goal which SCE touts to investors, and the wildfire mitigation benefits the goal provides to both shareholders and ratepayers.

Operational Excellence (Performance Management and Operational Excellence Category, 4% of total STIP Target – TURN Recommendation of 2%)

SCE describes the Operational Excellence goal as a goal to "[e]xecute continuous improvement efforts for Catalyst Program [by i]mplementing ... planned improvement projects."¹⁰⁴² It is appropriate to share the cost of payments on this goal at 50%/50% between ratepayers and shareholders. In theory any operational efficiencies that are implemented on a lasting basis benefit ratepayers to a certain degree, however, TURN's testimony raised two issues with this goal that require further consideration.

First, shareholders reap the rewards for efficiencies found after the GRC application is filed. While such efficiencies will eventually be recognized in the revenue requirement when the subsequent GRC is filed, shareholders will reap the rewards for up to four years during the GRC cycle. Exhibit TURN-14 noted that this concept is exemplified in SCE's Chief Financial Officer (CFO)'s prepared remarks in the 2023 Fourth Quarter earnings report,

"[O]ur operational excellence initiatives are off to a solid start, and we are seeing this translate into higher operating efficiency throughout the business. This was reflected in better-than-expected SCE operational variances."¹⁰⁴³

¹⁰⁴² Ex. WP SCE-06, Vol. 4, BkB , p. 10.

¹⁰⁴³ Ex. TURN-14-E2, p. 20, referencing Prepared Remarks of Edison International CEO and CFO Fourth Quarter and Full Year 2023 Earnings Teleconference February 22, 2024, 1:30 p.m. (PT) (p. 4), The

The second issue raised by TURN is that "O&M efficiencies create new opportunity for management to offset higher capital spending, perhaps making such expenditures more palatable for customers and the regulator."¹⁰⁴⁴ Exhibit TURN-14 presents examples of this, including multiple remarks for SCE executives regarding the value of the operational excellence program to shareholders.¹⁰⁴⁵ Accordingly, shareholders and ratepayers should evenly share the cost of the Operational Excellence (Catalyst Program) goal, recognizing that shareholders benefit at least as much as ratepayers as the result of operational efficiency.

30.2.1.2 Any Reduction that the Commission Adopts for STIP Should be Fully Recognized and Not Cut in Half

SCE observes and argues the following in its Direct Testimony regarding how much of

the reductions that the Commission adopts for any particular STIP goal should be reflected in a

reduced forecast:

The STIP or EICP payout equals the target payout for the employee times the corporate multiplier and the employee's individual performance multiplier. Since the individual performance multiplier has the same impact on the STIP or EICP payout as the corporate multiplier, any disallowance for a particular Corporate Goal should be cut in half to reflect that the weighting of the Corporate Goal only applies to the corporate multiplier and not to the individual performance multiplier [(IPM)]. Any further reduction on the incorrect assumption that SCE employees' daily work can somehow be balkanized into customer-versus-shareholder benefit devalues and distorts the work that SCE's hard-working employees accomplish.¹⁰⁴⁶

prepared remarks and the associated slideshow from the fourth-quarter and full-year 2023 earnings call are included in Ex. TURN-14-Atch1 (Attachment 2).

¹⁰⁴⁴ Ex. TURN-14-E2, p. 20.

¹⁰⁴⁵ Ex. TURN-14-E2, pp. 20-22.

¹⁰⁴⁶ Ex. SCE-06, V04, p. 53.

As found in Exhibit TURN-14-E2, SCE's recommended approach, that the Commission cut any reduction to STIP funding in half in order to avoid balkanizing employees' daily work into customer-versus-shareholder benefit, is fundamentally flawed.¹⁰⁴⁷ The amount that SCE pays in STIP in a given year, i.e., the STIP pool, is wholly determined by the company's performance against STIP goals and is not affected by IPM machinations.¹⁰⁴⁸ In fact, as noted in TURN-14-E2, the IPM is a zero-sum game across the company.¹⁰⁴⁹ SCE's STIP specifically states:

Once the company allocates an organization's STIP award pool, that pool is fixed. This means that if an organization's leadership wants to fund IPM amounts more than 100 percent for its high performing exempt Eligible Employees, such as those in Pay Zones 6 and 5, they must shift funds away from exempt Eligible Employees in lower Pay Zones. Depending on the number of high performers in an organization, this could mean that exempt Eligible Employees in Pay Zone 4 and below may receive less than 100 percent of their STIP Opportunity.¹⁰⁵⁰

As such, given that the total STIP payout is directly and only affected by the STIP Pool,

which is, in turn, a product of the STIP goals multiplied by employee Base Pay,¹⁰⁵¹ the

Commission would be correct to adopt reductions to STIP based on a 100% accounting of any

reductions it makes to individual goal percentages.

SCE's attempt to rebut TURN's position adds little new information – the company

essentially simply repeats the contention that because of the presence of the IPM at the

individual employee level, the Commission should simply disregard the fact that the STIP pool

¹⁰⁴⁷ Ex. TURN-14-E2, p. 24.

¹⁰⁴⁸ Id.

¹⁰⁴⁹ Id.

¹⁰⁵⁰ Id., referencing pp. 3-4 of SCE's 2023 Short-Term Incentive Plan.¹⁰⁵¹ Id.

as a whole is determined by company performance against company STIP goals¹⁰⁵² – except for its acknowledgement that TURN is correct in its assessment that the IPMs are a zero-sum game across the company.¹⁰⁵³ SCE's conclusion regarding how to treat the zero-sum nature of the IPMs, however, is incorrect. The utility reiterates its hope that the Commission will focus on the fact that the IPM can significantly impact the STIP payout calculation for the individual employee and ignore the fact that the STIP payout, generally – the amount that the GRC is intended to forecast – is based solely on the SCE performance against company-wide goals. The IPM, as the name suggests, is nothing more than a modifier for each individual employee and in no way impacts the amount of STIP that the company will payout to the body of employees as a whole in a given year. The foregoing makes it clear that the Commission should disregard SCE's argument that any disallowance for a particular Corporate Goal should be cut in half.

30.2.1.3 The Commission Should Make its Reductions to SCE's Forecast Based on 2023 STIP Goals

SCE requests in its Rebuttal Testimony that the Commission "make any reductions [to the STIP forecast] based on the 2024 SCE goals and allocations ... [rather than the 2023 goals and allocations, based on the idea that] STIP goals for TY 2025 will be developed using the 2024 goals as a starting point... [and will t]herefore ... be more reflective (as compared to the 2023 goals) of what the 2025 goals are likely to be".¹⁰⁵⁴ Parties were not privy to SCE's STIP goals and allocations during the pendency of SCE's case in chief. Instead, the workpapers that support

¹⁰⁵² Ex. SCE06, V04, pp. 36-38.

¹⁰⁵³ Id., p. 38.

¹⁰⁵⁴ Ex. SCE17, V03, pp. 38-39.

SCE's Direct Testimony, the workpapers that TURN relied upon,¹⁰⁵⁵ identified the goals and allocations for 2023.¹⁰⁵⁶ Information from 2023, therefore, is the only information that TURN was able to evaluate during its case in chief and should be the information upon which the Commission bases its STIP deliberations in this case. Precedence for TURN's approach comes from SCE's 2021 GRC, where the Commission found the following:

Our review of the STIP/EIC goals is based on SCE's 2019 goals, which SCE presented in its direct testimony in support of its funding request and which intervenors had the opportunity to analyze and address in their testimony. SCE notes that it subsequently revised its goals for 2020. Because management has the discretion to change the goals and weightings each year, it is unclear that the 2020 goals would necessarily be more representative of the goals for 2021-2023. Moreover, since SCE presented these revised goals in rebuttal testimony, other parties did not have the opportunity to present testimony on the revised goals and there is a lack of detail in the record regarding the 2020 goals compared to the 2019 goals. (Footnote omitted)¹⁰⁵⁷

TURN submits that the Commission should use analogous reasoning in the instant case

and use 2023 goals and allocations, contrary to SCE's request to use those from 2024.

30.2.1.4 The Commission Should Adopt a Base Pay Conversion Reporting Mechanism

Regarding SCE's plan to convert a portion of its STIP compensation to Base Pay, TURN

originally recommended a reporting and clawback mechanism to track and return funds to

ratepayers.¹⁰⁵⁸ In Exhibit SCE 17, Volume 3, SCE argued that the clawback mechanism was

unnecessary because "SCE has already completed two of the three phases for its STIP to base

pay transition and has extensively communicated to employees that the third phase will go into

¹⁰⁵⁵ Ex. TURN-14-E2, p. 23.

¹⁰⁵⁶ Ex. WP SCE-06, Vol. 04 BkB, p. 10.

¹⁰⁵⁷ D.21-08-036, p. 431.

¹⁰⁵⁸ Ex. TURN-14-E2, p. 25.

effect the first pay period of 2025."¹⁰⁵⁹ In light of this information, TURN agrees the clawback mechanism is unnecessary and rescinds that proposal. SCE indicated that it does not oppose TURN's reporting mechanism proposal to submit "an advice letter to show the conversion occurred and to inform the Commission to any additional adjustments to SCE's STIP targets."¹⁰⁶⁰

SCE disagrees with TURN's recommendation that a Tier 2 advice letter be used for this reporting mechanism. However, TURN continues to recommend a Tier 2 advice letter be adopted for this reporting mechanism.

30.2.2 Long-Term Incentive Program (LTIP)

The Commission should again deny SCE's request to have ratepayers fund any portion of its Long-Term Incentive Plan (LTIP) costs. SCE's LTIP provides stock options and performance shares,¹⁰⁶¹ to executives and "non-executive principal-level employees, attorneys, and some project managers."¹⁰⁶² The Commission has consistently denied SCE's request for rate recovery of costs associated with LTIP since at least the 2009 GRC.¹⁰⁶³

In its most recent rejection of SCE's request, noting SCE's standard defense that its overall compensation is at market, the Commission stated,

We continue to find that LTI is primarily designed to reward SCE employees for promoting shareholder interests. ... LTI is closely tied to the stock performance of EIX since LTI awards take the form of equity in EIX. ... SCE's arguments that that reconsideration of this issue is merited in light of AB 1054 are not convincing. ... [W]e see no reason to discontinue our longstanding policy of

¹⁰⁵⁹ Ex. SCE-17, Vol. 3, p. 43.

¹⁰⁶⁰ Id. at p. 44.

¹⁰⁶¹ Ex. SCE-17, Vol. 3, p. 53.

¹⁰⁶² Ex. SCE-17, Vol. 3, p. 67.

¹⁰⁶³ D.09-03-025, pp. 134-135; D.12-11-051, p. 452; D.15-11-021, p. 266; D.19-05-020, p. 188.

denying ratepayer recovery for LTI. Therefore, SCE's request to include these costs in rates is denied.¹⁰⁶⁴

As TURN addressed in Testimony, SCE's justifications for LTIP rate recovery in the present GRC primarily rely on the same arguments which the Commission has consistently found uncompelling in the past to support ratepayer funding,¹⁰⁶⁵ including SCE's repeat attempt to rely on AB 1054 to support the inclusion of its LTIP request in rates.¹⁰⁶⁶

The Commission should once again find SCE's arguments regarding AB 1054 unconvincing. There is nothing in the statutes implementing AB 1054, or SCE's proposed approach to responding to its requirements, that warrants a change to the Commission's longestablished policy of disallowing utilities' LTIP. Accordingly, the Commission should deny SCE's \$22.017 million LTIP forecast.¹⁰⁶⁷

30.2.3 401(k) Savings Plan

SCE forecasts a test-year expense of \$132.041 million (nominal \$) for the 401(k) Savings Plan costs.¹⁰⁶⁸ To prevent the unsupported upward adjustment to the 401(k) Savings Program owing to the STIP-to-Base Pay conversion, TURN removes the upward adjustment that results from the 2022 Labor (i.e., Base Pay) restatement (after further adjusting it for escalation) from the 2025 Labor (i.e., Base Pay) forecast and multiplies the result by the 2022, 401(k)-to-Labor

¹⁰⁶⁴ D.21-08-036, pp. 423-424.

¹⁰⁶⁵ Ex. TURN-14-E2, p. 29.

¹⁰⁶⁶ Ex. SCE-06, Vol. 4, p. 70.

¹⁰⁶⁷ Ex. SCE-06, Vol. 4E3, p. 67-E3.

¹⁰⁶⁸ Ex. TURN-14-E2, p. 29 (references "RO Model O1) O&M Dashboard", "401(k)" tab. This amount is different than the amount stated in SCE's testimony (i.e., \$122.609 million, per Ex. SCE06V04, p. 100:7-8). SCE states that TURN should rely upon the value from the workpapers (email from SCE, dated February 21, 2024)).

(i.e., Base Pay) ratio of 11.54%.¹⁰⁶⁹ This results in a downward adjustment to the 401(k) forecast of \$5.146 million for a forecast of \$126.895 million.¹⁰⁷⁰ In constant-dollar terms, TURN's forecast is \$113.297, which represents a \$4.595 million reduction to SCE's \$117.892 million forecast.¹⁰⁷¹ Please note that this 4% reduction¹⁰⁷² should be applied to the RO Model calculation, given that SCE's 401(k) Program forecast will ultimately depend on reductions that the Commission makes to the labor force in its GRC decision.

30.2.3.1 The STIP-to-Base Pay Conversion Causes an Unreasonable and Unsupported Increase to SCE's Total Compensation

The 401(k) Savings Plan is a defined-contribution savings plan, which provides employees an opportunity to defer current income to save for future financial needs. The cost of the program to SCE is the result of the company's policy of matching employee contributions on a dollar-for-dollar basis up to 6% of the employees' Base Pay.¹⁰⁷³ As identified in SCE's Direct Testimony, SCE proposes to move a portion of the STIP compensation to Base Pay for various policy reasons.¹⁰⁷⁴

As noted in Exhibit TURN 14-2E, the nature of SCE's policy for matching 401(k) contributions on a dollar-for-dollar basis up to 6% of Base Pay is mechanical and arbitrary. SCE derives the forecast for its contributions to the 401(k) program by multiplying the test-year, Base

¹⁰⁶⁹ "RO Model O1) O&M Dashboard", "401(k)" tab.

¹⁰⁷⁰ Ex. TURN-14-E2, pp. 29-30.

¹⁰⁷¹ Id., p. 30.

¹⁰⁷² I.e., \$5.146M / \$126.895M.

¹⁰⁷³ Ex. SCE06V04, p. 101:6-7.

¹⁰⁷⁴ Ex. SCE06V04, pp. 45-49.

Pay forecast by the ratio of 2022 recorded program costs to 2022 recorded labor (i.e., Base Pay) costs.¹⁰⁷⁵ Doing so, however, increases the total compensation level that it pays employees because the recorded, 2022 ratio is multiplied by a higher base pay because of the STIP-to-Base Pay Conversion, which results in an increase of \$44.591 million increase to the 401(k)-contribution forecast.¹⁰⁷⁶ Therefore, while the STIP-to-Base Pay conversion is ostensibly a simple conversion of pay from one component to another, it has the effect of causing an increase to SCE's total compensation based on the mechanics that SCE uses to effect the change.

The evidence of the increase to the 401(k) forecast as a result of the STIP-to-Base Pay Conversion is not in dispute.¹⁰⁷⁷ The only question before the Commission is whether the increase is reasonable. TURN submits that SCE has not provided any justification that overall compensation should increase, as noted in Exhibit TURN-14-E2.¹⁰⁷⁸ In addition, Edison's Total Compensation Study results show that the company's compensation is already at market, which further shows that an increase is unnecessary.¹⁰⁷⁹

The method for calculating the 401(k)-contribution forecast is arbitrary and, as described in Exhibit TURN-14-E2, can and should be adjusted such that the absolute amount of 401(k) contribution, and therefore the total compensation, remains consistent with the levels prior to the STIP-to-Base Pay conversion. As such, TURN reduces the 401(k) contribution forecast so that it is consistent with the STIP target percentages prior to the STIP compensation transfer to the

¹⁰⁷⁵ Id., p. 109:1-2. What SCE refers to as the "labor" forecast can also be referred to as the "Base Pay" forecast.

¹⁰⁷⁶ Ex. TURN-14-E2, p. 30.

¹⁰⁷⁷ Ex. SCE06, V04, p. 103:FN 181 and SCE17, V03, pp. 66-67.

¹⁰⁷⁸ Ex. TURN-14-E2, p. 32.

¹⁰⁷⁹ Ex. TURN-14-E2, p. 32.

Base Pay compensation component, as discussed in Exhibit TURN-14-E2.¹⁰⁸⁰ As illustrated in Exhibit TURN-14-E2, this is a ratemaking adjustment and does not affect SCE's employees' total compensation when the pre- and post-compensation-conversion compensation levels are compared. SCE is free to reduce the company's 401(k) match percentages, as it sees fit, but, if it were to reduce the percentage and maximum match, the action would maintain, rather than reduce employee total compensation vis-à-vis the total compensation prior to the STIP-to-Base Pay conversion.¹⁰⁸¹

To counter the fact that the mechanical and arbitrary nature of the 401(k) contribution in combination with the STIP-to-Base Pay conversion contributes to an unreasonable increase to SCE's compensation level, SCE argues in its Rebuttal Testimony that "as it has in the past, [the company] used the Results of Operation (RO) Model to forecast the 401(k) company contribution" and it "reflects the "employee compensation program"... ."¹⁰⁸² However, the simple fact that SCE uses the RO model to forecast the 401(k) company contribution does not mean that SCE's policy for matching the 401(k) contribution at 6% is not mechanical and arbitrary or that, in combination with the proposed STIP-to-Base Pay Conversion, such fact contributes to an overall increase in compensation. SCE further argues in its Rebuttal Testimony that it is "perfectly reasonable that the 2025 labor forecast used to calculate 2025 401(k) costs

¹⁰⁸² Ex. SCE17, V03, p. 66.

¹⁰⁸⁰ Ex. TURN-14-E2, p. 31-33.

¹⁰⁸¹ Id., p. 31. TURN understands that by the time this issue is decided by the Commission that the company likely would have implemented the compensation transfer, likely without a reduction of 401(k) matching percentages. This is correctly viewed as a matter that is internal to SCE (and could have been avoided with a more proactive and transparent methodology to make the conversion-component transfer truly just a transfer). The likelihood that SCE would already have implemented the conversion should not be relied on as reason not to apply the correct methodology for the transfer given the evidence that SCE has provided.

includes the base pay increase from the STIP to base pay transition"¹⁰⁸³ because, in part, "the 2025 401(k) forecast reflects the costs resulting from market average base pay."¹⁰⁸⁴

SCE further argues that the company's STIP-to-Base Pay conversion simply fixes what is currently a competitive disadvantage and that SCE's proposal strives to help the company "pay market average base pay".¹⁰⁸⁵ Just because base pay may be less than market does not mean that the company should arbitrarily increase total compensation when it is not warranted on the basis of the Total Compensation Study (TCS). As aforementioned, the TCS, which was conducted prior to the STIP-to-Base Pay Conversion, found that SCE's total-compensation package is already at market. The competitive disadvantage that SCE complains about is relative to Base Pay, not overall compensation,¹⁰⁸⁶ which renders the ancillary impact to total compensation owing to an increase to 401(k) compensation on the basis of an increase to Base Pay unsupported and unreasonable. SCE contends that the total compensation is 0.6 percent below the market average, and that the increase in 401(k) company contributions from STIP to base would not have brought SCE's actual total compensation to the market average.¹⁰⁸⁷ However, the TCS, itself, already noted that SCE's compensation was at market, despite being 0.6 percent below market, given that it was within +/- 10% of market.¹⁰⁸⁸ In any case, SCE hardly mentioned, and did not attempt to justify the forecasted increase to the 401(k)-contribution cost as the result of

¹⁰⁸³ Id.

¹⁰⁸⁴ Id.

¹⁰⁸⁵ Id.

¹⁰⁸⁶ Ex. SCE-06, V04, p. 47.

¹⁰⁸⁷ Ex. SCE-17, V03, p. 67.

¹⁰⁸⁸ Ex. TURN-14-E2, p. 35.

the STIP-to-Base Pay Conversion. The Commission should not allow this unsupported increase in total compensation.

30.2.4 Medical Program

SCE contributes to employee medical coverage under Medical Programs.¹⁰⁸⁹ TURN summarized SCE's forecasting methodology in Exhibit TURN-14.¹⁰⁹⁰ TURN takes issue with SCE's proposal to add 16% to its forecast to account for a new Premium-Sharing Redesign.¹⁰⁹¹ The new premium-sharing design includes a reduction to the employee share of (i) healthcare premiums across all medical plans and (ii) select medical-plan co-pays and out-of-pocket costs, as well as, (iii) implementation of a standard/closed prescription drug formulary for the pharmacy program offered by Express Scripts.¹⁰⁹²

SCE justifies the premium-sharing redesign by comparing its current contribution to employee medical benefits to PG&E's¹⁰⁹³ and asserting that the "changes are necessary to bring employee costs back within market benchmarks and stay competitive in a tight labor market."¹⁰⁹⁴ SCE's assertion that its Medical Programs contribution is below market ignores the fact that its benefits program, generally, is at 20% above market, well above what other market participants receive and its overall compensation program has been found to be at-market.¹⁰⁹⁵ SCE has provided no justification for an increase to its contribution to medical costs in light of the

¹⁰⁸⁹ Ex. SCE- 06, Vol. 4, p. 114.

¹⁰⁹⁰ Ex. TURN 14-E2, pp. 33-34.

¹⁰⁹¹ Ex. SCE- 06, Vol. 4, p. 120.

¹⁰⁹² *Id.*, p. 118.

¹⁰⁹³ *Id*.

 $^{^{1094}}$ Id.

¹⁰⁹⁵ Ex. SCE- 06, Vol. 4, pp. 44-45.

existing, generous overall benefits package and a total compensation package that SCE and its compensation consultant has found to be at-market. Accordingly, the Commission should remove the 16% upward adjustment that SCE includes for the Premium-Sharing Redesign, which results in a reduction of \$20.866 million to SCE's \$153.788 million forecast on a nominal basis. On a constant-dollar basis TURN's forecast recommendation is \$113.198 million for a reduction of \$17.770 million. Please note that this 16% reduction should be applied to the RO Model calculation, given that SCE's Medical Programs forecast will ultimately depend on reductions that the Commission makes to the labor force in its GRC decision.

30.2.5 Pensions

The Commission should maintain the "historical funding policy" and authorize a pension expense of \$17 million, rather than adopt the "new funding policy" SCE proposes, with its \$44.9 million forecast.

SCE's "historical funding policy" has been in place since at least 1982.¹⁰⁹⁶ In recent years, the combination of investment returns and ongoing ratepayer-funded contributions (the Pension Plan's only funding sources¹⁰⁹⁷), have produced a pension asset that is slightly overfunded – that is, the asset exceeds the present value of all benefits earned to date. In the 2021 GRC, using the historical funding policy resulted in a authorized pension cost forecast of \$84.3 million per year.¹⁰⁹⁸ Maintaining the historical funding policy through this test year 2025 GRC period would result in an authorized pension cost forecast of \$17 million per year.¹⁰⁹⁹

¹⁰⁹⁶ Ex. SCE-06, Vol. 4, p. 88.

¹⁰⁹⁷ *Id.*, p. 86.

¹⁰⁹⁸ *Id.*, p. 93.

¹⁰⁹⁹ Ex. TURN-18, p. 1, citing Response to TURN DR 37, Question 9(b) (included in Ex. TURN-18-Atch, p. 16).

The historical funding method is not one the Commission can rely upon indefinitely. SCE closed its pension plan to new participants after December 31, 2017.¹¹⁰⁰ This changes the underlying math, as the number of working employees eligible to participate in the plan and the pension-eligible payroll will shrink over time until each eventually reaches zero.¹¹⁰¹ But the amount of pension-eligible payroll is projected to be \$1.1 billion in the 2025 test year, and \$977 million in 2028, the last year of this rate case cycle, and SCE estimates the pension-eligible payroll will not reach zero for another 45 years, or 2068.¹¹⁰²

SCE's "new funding policy" is intended to "fix the mechanical issue with the current policy," and the utility contends it will achieve better alignment of contributions with benefit accruals and limit restrictions or costs associated with being underfunded.¹¹⁰³ It comes with a higher direct cost: The "new funding policy" includes an "actuarial forecast" of \$44.9 million for each year of the 2025 GRC period.¹¹⁰⁴ In addition, SCE's proposal would reduce and perhaps eliminate the pension cost adjustments that have occurred through the operation of the Pensions Cost Balancing Account (PCBA) in recent years. Instead of using above-expected asset returns to provide near-term credits to ratepayers, under the new funding policy would be "accumulated" and addressed in the test year 2029 GRC period.¹¹⁰⁵

¹¹⁰⁰ Ex. SCE-06, Vol. 4, p. 81.

¹¹⁰¹ *Id.*, p. 82.

¹¹⁰² Ex. TURN-114 Response to TURN DR 37, Question 5.

¹¹⁰³ Ex. SCE-06, Vol. 4, p. 91.

¹¹⁰⁴ *Id.*, p. 95.

¹¹⁰⁵ Ex. SCE-17, Vol. 3, p. 58; Ex. TURN-111, p. 1 (Response to TURN DR 127, Question 3.a.). The credit that flowed through the PCBA due to 2021 recorded amounts was \$59.4 million; the credit for 2022 recorded amounts was \$84.9 million. Ex. TURN-113 (Excerpts of SCE ERRA testimony), pp. 9 and 24. Investment returns are a very significant part of the "non-service" costs that resulted in the 2021 and 2022 credit amounts. Patel, SCE, 1244, 1. 16 to 1245, 1. 8.

TURN urges the Commission to retain the historical funding policy for the test year 2025 GRC period, and defer adoption of any new policy until the test year 2029 GRC. SCE's approach would have its customers pay approximately \$110 million of higher revenue requirement during the 2025 GRC cycle in order to mitigate risks that are neither as time-sensitive nor as dire as the utility makes them out to be. According to the calculations SCE presented from its actuary, the additional funding would serve to avoid having the plan underfunded by approximately 1%, and then only in 2028, the fourth year of the GRC cycle.¹¹⁰⁶ TURN explains below why that 1% underfunded figure is overstated, given 2023 recorded returns. But even if it were a reasonable forecast, the Commission should still conclude that, on balance, having SCE's customers pay approximately \$110 million less during this GRC cycle is the more reasonable outcome.

30.2.5.1 The Commission Should Adopt a Forecast of \$17 Million for Pension Costs Based on the Historical Policy, Rather than the \$44.9 Million SCE Proposes Under its New Funding Policy.

SCE forecasts \$44.9 million for its Pension Plan costs for each of the four years in the 2025 test year GRC period.¹¹⁰⁷ This forecast is premised on the Commission adopting SCE's "new funding policy." If the Commission were instead to retain the "historical funding policy" for this GRC period, the forecast would be \$17 million, per SCE's calculations.¹¹⁰⁸ TURN

¹¹⁰⁶ Ex. TURN-18, p. 11.

¹¹⁰⁷ Ex. SCE-06, Vol. 4, p. 81.

¹¹⁰⁸ SCE's direct testimony did not include the GRC forecast figure that would result were the historical funding policy maintained for the test year 2025 GRC period. TURN obtained that figure through discovery - \$17 million. Ex. TURN-18, p. 1, citing Response to TURN DR 37, Question 9(b) (included in Ex. TURN-18-Atch, p. 16).

recommends adoption of the \$17 million figure derived by continuing the longstanding historical policy, rather than the \$44.9 million figure SCE seeks under its new funding policy.

30.2.5.2 The Changed Conditions Associated with SCE's Pension Plan Being Closed to New Employees Do Not Require An Immediate Fundamental Change of Funding Policy With a \$110 Million Price Tag.

SCE describes its proposed new funding policy as intending "to fix structural issues with the legacy funding policy arising from the Pension Plan having closed to new employees after December 31, 2017."¹¹⁰⁹ TURN agrees that, with the closing of the plan to new participants, pension-eligible payroll will reduce over time. However, as noted above, there is no immediate urgency to revising the current policy. The actual recorded decline in eligible payroll is approximately 3.6% over the entirety of the 2018-2023 period.¹¹¹⁰ SCE projects a steeper pace of decline over the 2024-2028 period, but the remaining pension-eligible payroll would still be nearly \$1 billion in 2028, the last year of the test year 2025 GRC period.¹¹¹¹ Thus, while TURN does not dispute the need to modify the plan at some point in the foreseeable future, there is not such urgency that the plan must be modified at this time, rather than, for example, in the next SCE GRC.

Additionally, one of the issues SCE has identified as a basis for switching to its new funding policy is the possibility of an "unfunded liability" amount that would need to be

¹¹⁰⁹ Ex. SCE-06, Vol. 4, p. 81.

¹¹¹⁰ The pension-eligible payroll was \$1.315 billion in 2018 (the first year after the plan was closed to new employees), and \$1.268 billion in 2023, a decline of approximately 3.6% over the five-year period. Ex. TURN-114 (Response to TURN DR 37, Question 5).
¹¹¹¹ Id

amortized over a continually decreasing time period.¹¹¹² One of the touted features of the new funding policy is its element specific to recovering any "pension deficits" over an eight-year period tied to SCE's GRCs.¹¹¹³ But there is at present no such deficit, so under SCE's proposal this feature will not be deployed during the 2025 GRC period.¹¹¹⁴ Thus, waiting until the 2028 GRC to switch to a new policy would not reduce the amount of time available to amortize any "unfunded liability," should one arise during the 2025 GRC period.¹¹¹⁵

The Commission should find that SCE has not presented evidence that action must be taken as of the start of the 2025 test year, rather than as of the start of the next ensuing GRC, for example.

30.2.5.3 SCE's Attempts to Establish Potential Near-Term Funding Shortfalls Under Continuation of the Historical Policy are Unpersuasive.

The Commission should find that SCE's attempts to illustrate potential funding shortfalls during the 2025 GRC period from retention of the historical policy were overstated and unpersuasive.

First, there is Figure III-17 as presented in SCE's direct testimony. The graph purports to illustrate that continuation of the historical funding policy risks an outcome where the "projected funded position" would be at or above 100% in all but the last year of the test year 2025 GRC

¹¹¹² Ex. SCE-06, Vol. 4, p. 90.

¹¹¹³ *Id.*, p. 92.

¹¹¹⁴ *Id.*, p. 92. SCE states that there is no projected shortfall as of January 1, 2025, and "therefore the pension deficit amount included in the funding policy contribution is currently \$0."

¹¹¹⁵ As explained more fully below, TURN submits that the Commission should find that it is unlikely that such an unfunded liability will arise during the 2025 GRC period.

period, when it would be at 99%.¹¹¹⁶ Nowhere does SCE explain why the Commission should find attaining a 100% funded level for three of four years, and a one-year period at 99% funded, to be such a problematic outcome that it warrants adopting a revenue requirement that is higher by approximately \$28 million per year (or a cumulative \$110 million over the four-year GRC period) in order to avoid it. The more reasonable conclusion the Commission should draw from Figure III-17 as presented by SCE is that the utility's Pension Plan would be adequately well-funded during the 2025-2028 GRC period even if the Commission retains the historical funding policy as recommended by TURN.¹¹¹⁷

Then there is SCE's Figure III-17 as modified in TURN's direct testimony. The calculations leading to the percentages graphed by SCE include a forecasted "investment return" figure that ranges between approximately \$195 million to \$217 million per year for the historical funding policy, and \$201 million to \$217 million for the new funding policy.¹¹¹⁸ But in 2023, the actual investment return recorded by SCE was \$349 million, rather than the \$217 million SCE had forecasted for that year. If the 1/1/24 data point in SCE's Figure III-17 were updated to reflect the recorded amount reported in the utility's 10-K report filed with the Securities Exchange Commission (SEC) in early 2024, the "% Funded" figure under the historical policy increases from approximately 102.5% to 104%.¹¹¹⁹ As TURN's testimony illustrated with its

¹¹¹⁹ Ex. TURN-18, p. 8.

¹¹¹⁶ Ex. SCE-06, Vol. 4, p. 97, Figure III-17.

¹¹¹⁷ Ex. TURN-18, p. 7. SCE's illustrative graph also establishes that the plan would be somewhat <u>over</u>-funded throughout that period under SCE's proposed new funding policy. The "New Policy" data points are slightly above 101% in each year from 2025 through 2028. Ex. SCE-06, Vol. 4, p. 97, Figure III-17.

¹¹¹⁸ Ex. TURN-112, pp. 3 and 4, line A.5. This "investment return" figure is different than and distinct from the "excess investment return" figure SCE used to derive its revenue requirement forecast of \$44.9 million. Ex. SCE-06, Vol. 4, p. 93.

revised version of Figure III-17, adding the modified "% Funded" point based on the 2023 recorded data strongly suggests a substantially reduced likelihood of the plan being underfunded during the 2025-2028 period even with retention of the historical funding policy.¹¹²⁰

Exhibit TURN-112 puts a finer point to it. TURN had asked SCE to provide a recalculated Figure III-17 that made a single change -- substituting the recorded 1/1/24 figure for the forecasted figure SCE had included in the table version that appeared in its direct testimony. The utility had declined to do so, asserting that it "is not obligated to conduct new studies or analyses in response to data requests."¹¹²¹ Therefore TURN used the Excel version of the SCE version of the calculations behind the utility's graph to prepare the modified version of the calculations, the task the utility refused to perform.

TURN's cross-examination of SCE's witness reviewed in some detail the calculations underlying Figure III-17 as it appeared in SCE's testimony.¹¹²² The cross-examination also established that TURN's two additional tables had revised a single input figure for SCE's

¹¹²⁰ *Id*.

¹¹²¹ Ex. TURN-112, p. 2 (Response to TURN DR 127, Question 1.b.). SCE's refusal to perform such a calculation in a GRC is a relatively recent development. The test year 2018 GRC decision describes the utility making a "single change to it regression model" in order to enable a comparison of "SCE's forecast with TURN's modification to that forecast." D.19-05-020, p. 274. As TURN demonstrated during the evidentiary hearing, the calculations underlying Figure III-17 were far more straightforward than a regression model.

¹¹²² TURN first established that the "Investment Return" figures included in SCE's calculations reflected an assumed return of 6.5% in every year. Patel, SCE, 12 RT 1223, l. 23 to 1224, l. 17. TURN then methodically reviewed with the witness each of the categories of costs or credits included in the utility's calculation of the "Fair Value of Assets" and the Projected Benefit Obligation (PBO), and the use of the total from each of those two categories to derive the "Funded Status" as a percentage. *Id.*, at 12 RT 1225, l. 21 to 1229, l. 8. TURN then discussed how the outcomes from 2023 influenced the calculations for 2024. *Id.*, at 12 RT 1229, l. 9 to 1230, l. 12. TURN also confirmed with SCE's witness that 1) the figures for 2023 and 2024 in the two tables as prepared by SCE were identical, as were the amounts for "Benefit payments" and "Administrative expenses from trust" for the years 2025 through 2028, but 2) starting in 2025 the higher amounts collected in rates under the New Funding Policy would increase the "Employer contributions" figure and the "Investment return" figure (as the 6.5% assumed return was applied to a slightly higher balance due to the higher Employer contributions). *Id.*, 1231, l. 4 to 1234, l. 7.

calculations (the Investment return figure for 2023), and held the other input figures constant.¹¹²³ In TURN's version, the Fair Value of Assets figures for the end of 2023 and for each year thereafter increased, the mathematical result of reflecting the higher investment return for 2023, as did the "Funded Status" for each year.¹¹²⁴ And where SCE's Figure III-17 had shown a Funded Status of 99% in 2028, the same calculation with the same assumptions other than replacing SCE's 2023 "Investment return" figure with the recorded 2023 investment return produces a "Funded Status" of 103% in 2028.¹¹²⁵

SCE's re-direct examination of the utility's pension witness seemed to suggest it viewed TURN's illustration as putting the Commission at risk of drawing undue conclusions off of a single year's asset return performance, or structuring the pension funding policy based on a single year's returns.¹¹²⁶ But TURN is not proposing to change the amount of returns, either on a percentage or a dollar basis, for 2024 to 2028 – TURN's calculations used the same return figures for those years as did SCE. Furthermore, SCE pointed to its testimony's table of the pension fund's recorded return rates for the period from 2008 through 2022 for the proposition that market returns were very volatile during that period.¹¹²⁷ TURN submits that the historical data better serve to indicate that the return estimate that SCE uses in its calculations is relatively conservative, as the reported returns exceeded 6.5% in ten of the fifteen reported years.¹¹²⁸ In

¹¹²³ Ex. TURN-112, p. 5; Patel, SCE, 12 RT 1235, ll. 2-21; 1236, l. 23 to 1237, l. 19; and 1238, ll. 8-13.

¹¹²⁴ Ex. TURN-112, p. 3 (SCE-prepared table for "Historical Funding Policy"), lines A.1, A.6, and C.1; compare to p. 5 (TURN-prepared table for "Historical Funding Policy"), lines A.1, A.6, and C.1.

¹¹²⁵ Ex. TURN-112, p. 5, line C.1.

¹¹²⁶ Patel, SCE, 12 RT 1268, l. 9 to 1269, l. 24.

¹¹²⁷ Patel, SCE, 12 RT 1269, Il. 3-14, referring to Ex. SCE-06, Vol. 04, p. 85, Table III-10.

¹¹²⁸ Ex. SCE-06, Vol. 04, p. 85, Table III-10. If one were to look at the recorded return figures from 2004-2007 as well, the 6.5% estimate here was exceeded in fourteen of the nineteen years in the 2004-2022 period. Ex. TURN-116 (SCE TY 2021 GRC Testimony), p. 78, Table III-10. And given the much

addition, SCE cannot be permitted to elevate its assumptions for the 2023-2028 period to a sacrosanct status, such that the funding status estimates it developed based entirely on those assumptions are treated as inherently more reliable than near-identical funding status estimates that rely on the same assumptions, save a single recorded year's amount for the return. If SCE's model is so fragile that it goes from reliable to unreliable with the change of a single input from an estimated figure to a recorded figure, the problem is not with TURN's proposed change, but with the model itself.

To be clear, TURN's recalculation of the "projected pension funding status" that SCE included in Figure III-17 of its testimony is <u>not</u> intended to suggest that the Commission should assume that investment returns in 2024-2028 will be at a level higher than 6.5% for any year other than 2023, as reflected in the reported 1/1/24 figure for 2023. Rather, TURN asks the Commission to treat the results of the recalculation presented in Ex. TURN 112 as further confirmation of why it should reject SCE's undue urgency for a new policy: Adoption of a GRC funding level of \$17 million consistent with the historical funding policy will likely lead to a funding status of greater than 100%, rather than the 99% figure that SCE calculated.¹¹²⁹

higher investment return recorded in 2023 as compared to SCE's forecast, the Commission can reasonably infer that the 6.5 % figure was exceeded that year as well.

¹¹²⁹ Ex. TURN-18, p. 9.

30.2.5.4 SCE's New Funding Policy Would Postpone Credits from Higher than Forecasted Market Returns to the 2029 GRC Period, Rather than Permitting those Credits to Provide Some Amount of Near-Term Rate Relief.

Investment returns and the amount SCE collects in rates for its pension expenses "are the Pension Plan's only funding sources."¹¹³⁰ SCE's testimony described the circumstance where investment returns are lower than expected, requiring additional "contributions" collected from ratepayers.¹¹³¹ But over the last fifteen years, the pension fund performance has "compared favorably to market benchmarks," despite weak market performance in four of those fifteen years. And during the 2021-2024 GRC period (a period containing one of the four weak market years), SCE expects to see the actual contributions collected from ratepayers under the historical funding policy be <u>less</u> than the authorized amount due to the "stronger longer-term market performance."¹¹³²

Under the historical funding policy, SCE ratepayers see their actual contributions reduced during periods of relatively strong market performance, with at least part of that reduction achieved through the annual adjustment made in the Pensions Cost Balancing Account (PCBA). To the extent returns earned by pension trust assets in a given year exceeded expected amounts, some but not all of that excess return is credited to ratepayers.¹¹³³ These excess return amounts constitute a "significant" portion of the "non-service costs" tracked in the PCBA,¹¹³⁴ and

¹¹³⁰ Ex. SCE-06, Vol. 4, p. 86.

¹¹³¹ *Id*.

¹¹³² *Id*.

¹¹³³ Ex. TURN-111 (Responses to TURN DR 127), p. 1 (Response to TURN DR 127-3b); Patel, SCE, 12 RT 1248, ll. 1-7.

¹¹³⁴ Patel, SCE, 12 RT 1244, l. 16 to 1245, l. 8.

contributed to substantial credits benefiting ratepayers in recent years. In 2021, the "net gain" from the "non-service costs" was \$87.6 million, and was the main factor contributing to an overcollection of \$59.4 million that was credited to SCE's ratepayers at the start of 2022.¹¹³⁵ In 2022, the net gain amount was \$107.1 million, and the overcollection credited at the start of 2023 was \$84.9 million.¹¹³⁶ Thus, under the historical funding policy, the above-anticipated returns recorded during the 2021 GRC period serve to reduce the amounts collected in rates during that same GRC period, with at least some of that reduction occurring through the PCBA.

SCE's "new funding policy" would fundamentally change this element of current ratemaking. The amount of higher investment returns to be reflected in rates in each year of the 2025 GRC period would be capped for the entirety of the period at an amount SCE seems to have selected based on its comfort level.¹¹³⁷ Should the actual returns exceed that set amount, the excess returns would no longer be credited back to ratepayers during the 2025 GRC period, but instead "would be accumulated and increase the 'excess investment return' credit starting with the next [test year, here 2029]."¹¹³⁸ Thus, amounts that under the historical policy would have been available at least in part to be credited to SCE ratepayers during the test year 2025 GRC cycle would, under the proposed new funding policy, <u>not</u> be credited during this cycle and instead would be "accumulated" until at least 2029.

¹¹³⁵ Ex. TURN-113, pp. 9-10 (SCE 2022 ERRA testimony excerpt, pp. 63-64).

¹¹³⁶ *Id.*, pp. 24-25 (SCE 2023 ERRA testimony excerpt, pp. 68-69).

¹¹³⁷ Ex. SCE-06, Vol. 4, p. 93. SCE would set the "amount of excess investment return expected" at \$36 million. As noted earlier, in 2023, the amount of excess investment return recorded was \$132 million. Ex. TURN-18, p. 9.

¹¹³⁸ Ex. SCE-17, Vol. 3, p. 58; Patel, SCE, 12 RT 1246, ll. 3-9.

TURN submits SCE did not sufficiently emphasize this aspect of its "new funding policy." The utility's direct testimony referenced the new policy's treatment of a Projected Benefit Obligation (PBO) <u>shortfall</u> in several places.¹¹³⁹ There was no similarly direct reference to how investment returns that <u>exceed</u> expected levels would be treated differently as compared to under the historical policy. Instead, SCE repeatedly stated that it was seeking "continuation of the existing [PCBA],"¹¹⁴⁰ with "no change to the clear and well-defined operation of the pension balancing account or its application to ERRA as approved in [GRCs since 2006]."¹¹⁴¹ It may be technically true that the PCBA will not change, in that the key provisions of its tariff language may remain the same. But SCE's rebuttal testimony, discovery on that rebuttal testimony, and cross-examination of SCE's witness made clear that the PCBA will <u>not</u> operate in the same way under the new funding policy because the returns in excess of the annual forecast amounts would no longer flow through it to provide near-term rate relief to SCE's customers.¹¹⁴²

30.2.5.5 The Commission Should Consider Adopting the "Annual Check-up" Process SCE Proposed in its 2021 GRC.

Even though SCE has failed to present a compelling showing in favor of adopting its proposed "new funding policy" rather than maintaining the "historical funding policy" for the 2025 GRC cycle, TURN recognizes that circumstances can change, even over a four-year GRC period. In its 2021 GRC testimony, SCE proposed an annual review of the current funding

¹¹³⁹ Ex. SCE-06, Vol. 04, pp. 81, and 91-92.

¹¹⁴⁰ *Id.*, p.

¹¹⁴¹ Ex. SCE-17, Vol. 3, p. 59.

¹¹⁴² *Id.*, p. 58, Ex. TURN-111, p. 1 (Response to TURN-127, Question 3.a.); Patel, SCE, 12 RT 1247, ll. 4-17.

policy to enable the possibility of updating that policy if shown to be necessary before the next GRC cycle.¹¹⁴³ To be clear, TURN does not believe SCE has established that any such update is required at this time, or likely to be warranted during the 2025 GRC cycle. However, if the Commission wishes to create an informal process to check that changed circumstances do not warrant a different approach on relatively short notice, it could direct SCE to work with other interested parties to develop and propose such an informal process.

30.2.6 Post-Retirement Benefits Other than Pensions (PBOPs)

For test year 2025, SCE forecasts \$0.0 million for its Post-Retirement Benefits Other than Pensions (PBOP) costs.¹¹⁴⁴ This is in large part due to the fact that the PBOP trust fund assets had reached \$2.21 billion as of the end of 2022, and are invested such that the ongoing returns contribute to covering the PBOP costs.¹¹⁴⁵ As reported in SCE's ERRA testimony, the figure at the end of 2022 meant the entire PBOP plan was overfunded by approximately \$885 million.¹¹⁴⁶ As of the end of 2023, the overfunded amount had grown to \$1.5 billion.¹¹⁴⁷

TURN supports adopting a funding level of \$0 for the test year 2025 GRC period, as making sure overfunded amount does not continue to grow in part due to amounts collected from ratepayers is an important first step.¹¹⁴⁸ But TURN urges the Commission to recognize the degree of overfunding presents two areas warranting further inquiries: Is the overfunded amount sufficiently protected from being put to uses that are not in the interest of SCE's ratepayers? And

¹¹⁴³ Ex. TURN-116 (SCE 2021 GRC Testimony), p. 84 (ll. 3-12).

¹¹⁴⁴ Ex. SCE-06, Vol. 04, p. 103.

¹¹⁴⁵ *Id.*, p. 109.

¹¹⁴⁶ Ex. TURN-113, p. 29 (SCE 2023 ERRA testimony excerpt, p. 73).

¹¹⁴⁷ Ex. TURN-18, pp. 2 and 17, citing SCE's 10-K filing for 2023.

¹¹⁴⁸ *Id.*, p. 2.

are there alternative uses for the overfunded amount that might be permitted and would serve to further the interest of SCE's ratepayers?

The record establishes that SCE has already begun the process of identifying opportunities to put the overfunded PBOP trust fund amount to other purposes. For example, in the ERRA testimony revealing the \$885 million overfunded status as of the end of 2022, the utility stated, "SCE has been seeking ways to address this funding imbalance and believes that there will soon be a path forward for doing so."¹¹⁴⁹ TURN's testimony raised the concern about opportunities for the company to sell or otherwise transfer the obligations and the fund in a manner that might shortchange SCE ratepayers, such that they ultimately receive a reduced or discounted value from the trust.¹¹⁵⁰

SCE asserted that the funds are "locked up" while in the trust, and subject to "very tight guidelines."¹¹⁵¹ But in a "Funding Policy Discussion" document provided in mid-2022 to SCE by its pension and PBOP actuary, the "PBOP Funding Policy" included references to the possibility of amending the trusts to cover "PAYGO" benefits that are "not covered by existing trusts." The document also includes the heading "No need for CPUC input (?)," followed by a note indicating that SCE, or at least its actuary, thought that such amendments of the trust may not "have GRC implications."¹¹⁵² The following page includes the statements, "\$71 million of 'pay-as-you-go' benefits are not currently backed by any trust assets and associated benefit

¹¹⁴⁹ Ex. TURN-113, p. 29 (SCE 2023 ERRA testimony excerpt, p. 73).

¹¹⁵⁰ Ex. TURN-18, pp. 18-19.

¹¹⁵¹ Patel, SCE, 12 RT 1258, ll. 24-25.

¹¹⁵² Ex. TURN-115 ("SCE Funding Policy Discussion" dated July 13, 2022), p. 12.

payments of ~\$10m annually are paid from corporate assets," and "Expectation is most pay-asyou-go obligations would be covered by 401(h) via plan amendment."¹¹⁵³

It may be that the use of trust assets to cover amounts not currently "backed" by those assets can be structured in such a way as to be entirely in ratepayers' interest. But the Commission has no way to know this based on the minimal information SCE has put forward to date on the subject. And the utility's internal document suggests changes could take place without any need for Commission review or approval. In order to ensure that any such discussion includes not only the utility's actuary but also its regulator, the Commission should direct SCE to, no later than its next GRC, present a showing regarding its efforts seeking ways to address the funding imbalance, and any "paths forward" it has identified for protecting ratepayers' interest in these funds.¹¹⁵⁴ The Commission should further direct SCE to report on any use of trust assets to date to fund benefits that had not previously been backed by trust assets, and on the results of any further inquiries or investigation it has made into the "need for CPUC input" regarding such matters.

30.3 Employee Training

30.3.1 Training Seat Time and Delivery – Transmission and Distribution

TURN recommends that the Commission reduce SCE's test-year expense forecast of \$60.222 million for Training Seat Time and Delivery – Transmission and Distribution (T&D) to a more reasonable forecast of \$46.383 million. The utility's forecast is 62% higher than the base

¹¹⁵³ *Id.*, p. 13.

¹¹⁵⁴ Ex. TURN-18, p. 19.
year expense of \$37.058 million¹¹⁵⁵ and SCE has failed to demonstrate that the increase is just and reasonable.

The Training Seat Time and Delivery – T&D is divided into a (i) Training Seat Time forecast of \$37.023 million, for which TURN's recommendation includes a reduced recommendation of \$28.511 million; and (ii) Training Delivery forecast of \$28.511 million, for which TURN's recommendation includes a reduced recommendation of \$17.872 million.¹¹⁵⁶

As noted below, TURN's Training Seat Time recommendation includes, among other items, a reduction for TURN Witness Jones's observation that labor rates should be separated between new hires and existing employees rather than using a single, monolithic rate for all employees who attend a certain type of training.¹¹⁵⁷ However, SCE's Rebuttal Testimony states,

The itemized model by individual class and standard labor rate by job classification expected to attend those classes already accounts for the variance in labor rates by roles, as well as the new employee/existing employee dichotomy that TURN is attempting to address with its proposed forecasting model.¹¹⁵⁸

While no evidence of this assertion in SCE's rebuttal has been presented, TURN nevertheless allows for the possibility that SCE's model includes differential Seat Time charge rates that, if true, would account for TURN's recommendation to reduce the average training compensation rate to address the dichotomy of new vs. existing employee costs. As such, TURN introduces an alternative forecast for the Commission to consider, one that uses TURN's primary recommendation regarding Training Seat Time hours, but relies on SCE's forecast of rates.

¹¹⁵⁵ Ex. TURN 14-E2, pp. 36-37.

¹¹⁵⁶ Id., p. 38. *See also*, Ex. TURN-14 WP.

¹¹⁵⁷ Id., p. 45.

¹¹⁵⁸ Ex. SCE-17, Vol. 3, pp. 97-98.

TURN's alternative forecast is 29.204 million¹¹⁵⁹ – rather than the 28.511 million that is suggested under TURN's primary recommendation, as identified above.

Regardless of the Commission's decision on whether the particular forecast is reasonable, the Commission should order SCE to, either manually or dynamically within the RO Model, reduce both the Training Delivery and Training Seat Time activities if the Commission makes reductions to any training-impacted personnel on a prorated, percentage basis, that starts with the 2025 test year forecasts that the Commission adopts, as recommended by TURN Witness Jones.¹¹⁶⁰ The Training activities are mostly volume based – for example, the incremental costs within Training Delivery primarily owe to increased training demand, which in turn derives primarily from increased hiring in the field to support projects within the main organizational units of T&D – and the Training Seat Time is by definition volume based.¹¹⁶¹ The Commission's adoption of this recommendation would help ensure that any reductions to SCE's personnel counts that the Commission adopts will be reflected by the Training Seat Time- and Delivery-activity forecasts. This recommendation appears to be unrebutted by SCE.¹¹⁶²

¹¹⁵⁹ Calculated in the same manner as TURN's original calculation in TURN's Workpaper, Ex. TURN-14 WP, tab "Del_SeatTime-NHExiSpt_TURNCalcs", but with the application of SCE's average compensation rate of \$56.89 for all Training hours with the corresponding change to the Non-Labor Forecast, which is calculated by SCE as 17% of the Labor Forecast (according to WPSCE06V04BkC, p. 249, as: [the sum of TURN-recommended New Hires and Existing Employees training hours (358,000 hrs., per Ex. TURN-14-E2, p. 44) x the average rate according to SCE (\$56.89/hr., per WPSCE06V04BkC, p. 249)] + [the sum of TURN-recommended New Hires and Existing Employees training hours (358k) x the ratio of SCE's Employee Compensation Benefits value (\$688k/\$26.892M, per WPSCE06V04BkC, p. 249)] + [the remaining Labor items (\$4.327M, per Ex. TURN-14-E2, p. 47)] + [the Non-Labor Forecast (\$3.518M, per TURN's updated Labor Forecast of \$20.367M (i.e., 358,000 hrs. x \$56.89/hr. x SCE's 17% Labor-to-Non-Labor ratio (from WPSCE06V04BkC, p. 249)] + [the remaining Non-Labor items (\$3.888M, per Ex. TURN-14-E2, p. 47)].

¹¹⁶⁰ Id., pp. 38-39.

¹¹⁶¹ Id.

¹¹⁶² Ex. SCE-17, Vol. 3, pp. 93-98 and 102-103.

30.3.1.1 T&D Training Seat Time

TURN's forecast for Training Seat Time is primarily based on the 2023 recorded unit count for new-hire (or new role)-related training and an average of the 2018-2019 recorded unit count, the ongoing training of existing employees.¹¹⁶³ On this point, SCE's Rebuttal Testimony appears to misinterpret the basis of TURN's Seat Time forecast, claiming that TURN's forecast is "based on the 2023 Training Time hours planned against the recorded value."¹¹⁶⁴ While true that TURN's forecast of the new-hire-related portion of the forecast is based on the 2023 recorded value, the ongoing existing employee-related portion is based on the average of the recorded 2018-2019 forecast.¹¹⁶⁵

Prior to accounting for the additional items of Safety Training Seat Time Labor, Employee Compensation Benefit (i.e., STIP-to-Base Pay conversion), and Organizational Training Labor, the combined effect of reducing training time and separately calculating the existing and new-hire labor rates yields a Labor cost of \$19.789 million, compared to \$26.892 million under SCE's forecast methodology. After accounting for the additional items, with the Employee Compensation Benefit component of the forecast adjusted for a lower Labor amount, TURN's total Labor forecast for Training Seat Time is \$24.623 million. This is \$7.285 million lower than SCE's forecast of \$31.908 million.¹¹⁶⁶

¹¹⁶³ Ex. TURN-14-E2, p. 44.

¹¹⁶⁴ Ex. SCE-17, V03, p. 92.

¹¹⁶⁵ Ex. TURN-14-E2, p. 45.

¹¹⁶⁶ Id., pp. 46-47.

Other than two small items (Organizational Training Non-Labor and Employee-Led

Efficiency Savings), SCE's Non-Labor forecast for the Training Seat Time activity is calculated

as a percentage of the Labor forecast. Therefore, the adjustments that TURN.

The following table shows the high-level similarities and differences between SCE and

TURN's proposals.

Table 12: Itemized Comparison of SCE and TURN's Training Seat Time Forecast (1,000s of Constant 2022\$)1167

Category	SCE	TURN	SCE>TURN
Labor (New Hires and Existing) (473k hours x Labor Rate = \$56.89)	26,892	N/A	N/A
Labor (New Hires) (310k hours x Labor Rate = 46.73/hr)	N/A	14,475	N/A
Labor (Existing) (48k Labor Rate = \$110/hr)	N/A	5,314	N/A
Labor (TOTAL)	26,892	19,789	7,103
Employee Compensation Benefits	688	507	182
Remaining Labor Items	4,327	4,327	-
LABOR SUBTOTAL	31,908	24,623	7,285
Non-Labor	4,645	3,418	1,227
Remaining Non-Labor Items	470	470	-
NON-LABOR SUBTOTAL	5,115	3,888	1,227
TOTAL	37,023	28,511	8,512

However, if the Commission were to agree with SCE regarding the labor rate used in the Training Seat Time forecast, TURN's alternative forecast is \$29.204 million – rather than the \$28.511 million that is suggested under TURN's primary recommendation, as noted above.

30.3.1.1.1 The Basis of TURN's Forecast is Reasonable

SCE claims two sources of seat-time demand: demand for the training of new hires (or

the transition by existing employees to new roles) and ongoing training of employees in

incumbent roles.¹¹⁶⁸ TURN's recommendation provides SCE with a reasonable forecast of

¹¹⁶⁷ Ex. TURN-14-E2, p. 47.

¹¹⁶⁸ Id., p. 43, referencing SCE Response to DRs TURN-SCE-008-3.a, -7.a, and -9.a (included in Exhibit TURN-14-Atch1).

Training Seat Time hours, which contains the number of new-hire-related Seat Time hours that were *recorded* in 2023, the year that SCE claims is a reasonable proxy of 2025.¹¹⁶⁹ This will allow SCE to address the main drivers of the hiring-generated training needs and the incremental Training Seat Time labor forecast, such as Safety and Reliability Investment Incentive Mechanism (SRIIM) and Service Planning organization.^{1170,1171}

Regarding the ongoing training of existing employees, TURN removes the effects of the COVID rebound by recommending that the Commission adopt the two-year, 2018-2019 average of pre-pandemic, recorded Seat Time hours related to the on-going training of existing employees, given that the COVID rebound will clearly be in the rear-view mirror by 2025.¹¹⁷² Not only are these measures reasonable, but they are designed to conservatively reduce SCE's forecast of an activity that the company almost always over-forecasts and serve as a counterweight to a forecast that SCE has rendered impossible to test or audit.¹¹⁷³

30.3.1.1.2 SCE's Showing Regarding the Training Seat Time Hours Forecast is Problematic

SCE based its 2025 Training Seat Time forecast on the number of *planned* 2023 units because it expects 2025 to have similar training demand across the two years.¹¹⁷⁴ SCE's Rebuttal Testimony emphasizes the reasonableness of the company's total 2025 Seat Time-hours

¹¹⁶⁹ Ex. TURN-14-Atch1, SCE Response to DR TURN-SCE-008-03.h.

¹¹⁷⁰ Ex. SCE-06, V04, pp. 183-185.

¹¹⁷¹ Ex. TURN-14-E2, pp. 44-45.

¹¹⁷² Id., p. 41.

¹¹⁷³ Id., pp. 41-42.

¹¹⁷⁴ Ex. TURN-14-Atch1, SCE Response to DR TURN-SCE-008-03.h.

forecast on the basis that the 2023 *recorded* hours are significantly higher than previous, historical years.¹¹⁷⁵ However, a more appropriate starting point for establishing the reasonableness of the company's 2025 forecast is to compare the 2023 seat-time hours plan against the recorded hours.¹¹⁷⁶ The utility planned for 471,000 total Seat Time hours in 2023, but recorded just 391,000 which is 83% of planned.¹¹⁷⁷ Moreover, SCE has historically recorded much less training than planned – the recorded training hours for the historical period (save for the COVID-affected 2020) average 80% of planned with the pre-pandemic, with the recorded 2019 value, at 67% of planned, being substantially lower.¹¹⁷⁸

Such facts cast doubt on SCE's 2025 forecast and the Commission could reasonably assume that the 2025 recorded value will fall short of the forecast. In the face of such facts, however SCE would now have the Commission approve a forecast that is based on a Seat Time-hours forecast that is about 80% *higher* than the average in the non-COVID period of 2018-2019 and 2021-2022.¹¹⁷⁹ Contrary to what SCE attempts to portray in its Rebuttal Testimony¹¹⁸⁰, while TURN's Training Seat-hour forecast is not as extravagant as SCE's, it *is* conservatively higher than the relevant historical period suggests is necessary – i.e., TURN's forecast of 358,000 compared to an average of 263,000 in non-pandemic, historical years (i.e., 2018-2019 and 2021-2022), and also compared to 290,000 in 2022.¹¹⁸¹

¹¹⁷⁵ Id., p. 94.

¹¹⁷⁶ Ex. TURN-14-E2, p. 43.

¹¹⁷⁷ Id., pp. 43-44.

¹¹⁷⁸ Ex. TURN-14-E2, p. 45, Table 13.

¹¹⁷⁹ Id.

¹¹⁸⁰ Ex. SCE-17, Vol. 3, pp. 96-97.

¹¹⁸¹ Ex. TURN 14-E2, p.43.

SCE's Direct Testimony identified the COVID rebound as the explicit and sole reason for the increase to training costs, stating, "As the impacts of COVID-19 continue to decline, it is expected that training volume will continue to increase as a result of increased hiring, returning to normal training levels and/or continued make-up from prior deferrals."¹¹⁸² TURN does not object to an increase to incremental training hours from "increased hiring" relative to the base year, but simply recommends that the 2025 forecast be based on the 2023, *recorded* unit count (rather than the planned count), given SCE's contention that 2023 unit-count levels would be the same as 2025 levels. The recorded, 2023 level would allow SCE to address the increased hires for SRIIM and such organizations as Service Planning, as well as the company's request for funding for the multi-year impacts for programs in extended and/or multi-year duration, such as the three-year Apprenticeship programs,¹¹⁸³ given that SCE contends that the demand in 2023 and 2025 will be the same.¹¹⁸⁴

SCE's argument in Rebuttal Testimony that, in using the 2023 *recorded* values as the basis for the 2025 Seat Time-hours forecast rather than the 2023 *planned* values, TURN failed to realize that the 2023 recorded value was not available when the utility's initial testimony was developed,¹¹⁸⁵ is a curious argument. Why should the Commission not rely on a recorded value, regardless of whether it was known at the time of the utility's forecast development, as a way to evaluate the reasonableness of SCE's forecast? SCE explicitly tied the activity level in *2025* to

¹¹⁸² Ex. SCE-06, Vol. 4, p. 179.

¹¹⁸³ Id., p. 172.

¹¹⁸⁴ Id., p. 181.

¹¹⁸⁵ Ex. SCE-17, Vol. 3, p. 96.

that in *2023*, and offered no other rebuttal against the use of 2023 recorded values to forecast the new-hire-related portion of the Seat Time count.¹¹⁸⁶

As for the forecasted increase to base-year spending as the result of "returning to normal levels", this is exactly what TURN's forecast is consistent with. TURN's forecast for the ongoing training of existing employees is based on the average of 2018-2019 recorded values. The pre-pandemic level, not a level indicated by COVID-19 recovery, would be the "normal" level for ongoing training for existing employees.

As for the forecasted increase to base-year spending as the result of "continued make-up from prior deferrals", the claim is already tenuous enough when contemplating a 2025 test year – some four to five years on from the height of the pandemic – and impossible to verify with the dearth of information forthcoming from the company, as explained by TURN Witness Jones.¹¹⁸⁷ Despite the use of what SCE dubs a "bottoms-up itemized forecast methodology" for forecasting both Seat Time and Delivery, the utility was unable to produce a model that was anything other than a tautological absurdity comprising hardcoded Training Seat Time hours with no detail regarding the development of the underlying Seat Time durations and occurrences, as identified by TURN Witness Jones¹¹⁸⁸ and admitted to by SCE in rebuttal.¹¹⁸⁹ This is despite TURN's repeated attempts to obtain such simple information as the identification of the training hours,¹¹⁹⁰ how many training sessions were cancelled and/or deferred, how many remain unexecuted, or

¹¹⁸⁶ Id., pp. 93-98.

¹¹⁸⁷ Ex. TURN-14-E2, p. 41.

¹¹⁸⁸ Id., pp. 40-41.

¹¹⁸⁹ Ex. SCE-17, Vol. 3, pp. 93-94.

¹¹⁹⁰ Ex. TURN-14-E2, pp. 40-41 and Ex. TURN-14-Atch1 (Attachment 3 and SCE Responses to DR TURN-SCE-008-04.a, DR TURN-SCE-008-04.a Supplemental Answer and corresponding attachment, TURN-SCE-008-04.a.Supplemental).

how many cancelled and deferred sessions will be executed in 2023 and beyond.¹¹⁹¹ Without such information, the values remain hardcoded with no derivation, justification, or explanation as to their provenance. This leaves the "bottoms-up model" as a simply untestable proposition, a "give us the number of hours that we say we need because that is the number of hours that we say we need."

SCE attempted to justify the hardcoded values in the so-called, bottoms-up model and the general lack of support for its request in its Rebuttal Testimony, stating, "The quantity of Training Seat Time occurrences is derived through subject matter expertise, standard baseline class sizes where applicable, and conversations with organizational leadership to align training plans with expectations."¹¹⁹² While perhaps true, the cloak of "subject matter expertise", "conversations with leadership", and the like, provides no way of testing the hardcoded values, as proved by TURN's repeated attempts to obtain elementary information that could reasonably be used to test SCE's forecast.¹¹⁹³ Further, SCE has not established that these factors that it derives its forecast from are any better, or even different, than those that informed its significant over-forecasts in prior years.

Moreover, SCE's Rebuttal Testimony then alleged that there are "multiple dynamic, complex, and influencing factors that impact training volume beyond the scope of training operations," and attempted to rehabilitate the company's showing by highlighting "pandemic[-related]...factors outside of training [that] may have had a downstream effect to training with

¹¹⁹¹ Ex. TURN-14-E2, p. 42, supported by Ex. TURN-14-Atch1 (SCE Response to DR TURN-SCE-008-03).

¹¹⁹² Ex. SCE-17, Vol. 3, pp. 93-94.

¹¹⁹³ TURN-14-E2, pp. 40-41.

longer-term impacts.²¹¹⁹⁴ However, the company provided no concrete examples for such effects, no way to test the assertion, and no reason to believe that such effects might last until 2025, 4-5 years after the height of the pandemic.¹¹⁹⁵ SCE further alleged in its Rebuttal Testimony, while remaining silent in its Direct Testimony, "[F]actors unrelated to the pandemic also influence training demand..., [although] SCE does not track specific, increase or decrease for each unique and specific driver, especially in the case of an unprecedented global pandemic.²¹¹⁹⁶ This last point is undermines SCE's position. First, the company pinned the increase in expected Seat Time hours entirely to the pandemic in its Direct Testimony, and then reversed its position in Rebuttal by stating that there are other, non-pandemic drivers at play, drivers that not only does the company not track, but that are apt to vary from year to year even from a historical perspective.¹¹⁹⁷ If the company does not track such activities and those activities are given to vary from year to year, how would the company know that the demand for such activities would increase and how is the Commission supposed to audit the reasonableness of the allegation?

Finally, SCE claims in Rebuttal Testimony, "Not being authorized sufficient funds for 2025 and beyond would require SCE to reduce training hours."¹¹⁹⁸ The Commission should not succumb to such a strawman argument. TURN's recommendation allows SCE sufficient funds

¹¹⁹⁴ Ex. SCE-17, Vol. 3, p. 94.

¹¹⁹⁵ Id.

¹¹⁹⁶ Id., p. 94.

¹¹⁹⁷ Ex. SCE-17, Vol. 3, p. 95.

¹¹⁹⁸ Id., p. 96.

to do the training that it will need to do in the future and does not constitute a mandate for training reduction.

30.3.1.2 T&D Training Delivery

SCE's Training Delivery labor forecast consists of adjunct, internal SCE employees from other organizations to facilitate training, in addition to T&D Training staff and personnel, including permanent instructors, back-office support, analytical support, administrative support, and leadership.¹¹⁹⁹ TURN recommends a T&D Training Delivery Labor forecast (excluding the newly formed Safety Training) of \$12.422 million, which is \$3.980 million less than SCE's forecast of \$16.402 million,¹²⁰⁰ which is a 24.3% reduction.¹²⁰¹ TURN, likewise, reduces the quantity input for the Non Labor calculator by 24.3%, given that estimated rates and quantities of Non Labor expense anticipated to be needed to accommodate delivery to the forecasted Seat Time training hours are used to forecast Non-Labor expenses.¹²⁰² This reduction simply comports with the way SCE calculates Non Labor, but accounts for the lower Labor forecast that TURN recommends. Finally, TURN adjusts the Employee Compensation Benefits component of the forecast in similar fashion to the way that it adjusts for the Training Seat Time.^{1203,1204}

TURN's recommendation for the Labor forecast is supported by SCE's Direct Testimony, which indicated that the "Training Delivery ... and Training Seat Time [are] based

¹¹⁹⁹ Ex. SCE-06, Vol. 4, p. 182.

¹²⁰⁰ Ex. TURN-14-E2, p. 49.

¹²⁰¹ Ex. TURN-14-E2, p. 48.

¹²⁰² Id.

¹²⁰³ Id.

¹²⁰⁴ See Ex. TURN-14-E2, p. 49, Table 16 for an itemized comparison of SCE and TURN's Training Delivery Forecast, which includes Labor and Non-Labor components.

on "the volume of expected training hours to be delivered and standard labor rates averaged by class type for...staff and personnel associated with delivering and operating training programs."¹²⁰⁵ It is, therefore, reasonable to reduce the quantity input to the Labor-forecast calculation by the percentage of any reductions to the seat time quantity made in the Training Seat Time Labor calculation.

SCE claims to use a bottoms-up analysis that is based on the volume of expected training hours to be delivered and standard labor rates averaged by class type for the employee job classifications planned to attend training and/or staff and personnel associated with delivering and operating training programs, as previously noted. TURN has not received an analysis that shows how the Labor hours for Training Delivery are derived, despite repeated attempts, as documented above in the Seat Time section above.

SCE contended in Rebuttal Testimony that TURN's approach should not be relied upon because, "while some Training Delivery forecasting is dependent on increased training volumes, it is not a one-to-one relationship, [and] specific delivery expenses are not being forecasted in an itemized manner meaning that some items will be greatly over forecasted while others are greatly underforecasted." SCE further alleges that TURN's approach is flawed because (1) Training Seat Time and Training Delivery are different activities with different cost drivers and assumptions; (2) the blanket-percentage reduction that TURN applies ignores the standard labor rates of the employees charging to the activity; and (3) TURN's model ignores the possibility that non-labor expenses, quantities, and associated cost assumptions can be different than they are for labor and may or may not be dependent up on Training Seat Time – T&D volume.¹²⁰⁶

¹²⁰⁵ Ex. SCE-06, Vol. 4, pp. 180-181.

¹²⁰⁶ Ex. SCE-17, Vol. 3, p. 102.

TURN submits that SCE has failed to meet its burden to establish the reasonableness of its forecast when the only workpapers it provided are full of hardcoded inputs with little to no information about the provenance of the inputs. SCE's approach severely limits the Commission's ability to audit a so-called, bottoms-up forecast. While there might not be a one-for-one relationship between Training Delivery costs and Training volume, it is the best and most transparent evidence that any party has produced in this proceeding. It is SCE, after all, that suggested, "…Training Delivery… [is] based on the volume of expected training hours to be delivered and standard labor rates averaged by class type for the employee job classifications planned to attend training and/or staff and personnel associated with delivering and operating training programs."¹²⁰⁷

Finally, based on its objection to TURN's testimony regarding the Labor forecast for Training Delivery, SCE objects to TURN's reduction to the Non-Labor and Employee Compensation Benefits-adjustment forecast.¹²⁰⁸ Given that TURN continues to recommend the Labor reduction, its Non-Labor forecast and adjustment to the increase for Employee Compensation Benefits are reasonable and should be adopted.

¹²⁰⁷ Ex. SCE-06, Vol. 4, pp. 180-181.

¹²⁰⁸ Ex. SCE-17, Vol. 3, p. 102.

31. TOTAL COMPENSATION STUDY

32. ENVIRONMENTAL SERVICES

32.1 Environmental Services O&M

SCE's forecast for Environmental Services O&M includes two components, (1) Environmental Management and Development, and (2) Environmental Programs.¹²⁰⁹ TURN addressed SCE's forecast for Environmental Programs in its testimony.¹²¹⁰

SCE originally forecast \$22.694 million for Environmental Programs, including \$1.335 million in labor and \$21.359 million in non-labor.¹²¹¹ SCE subsequently modified its labor forecast to \$1.329 million, bringing its total forecast to \$22.689 million.¹²¹² TURN took issue with SCE's request for an \$8.07 million increase in non-labor. TURN pointed to SCE's failure to support this requested increase and also the consistent decline in non-labor costs in every year from 2018-2022.¹²¹³ TURN recommended a forecast of last recorded year, consistent with the Commission's guidance in prior GRC decisions that last recorded year is appropriate to use as a forecast of \$14.845 million, including \$1.553 in labor and \$13.292 million in non-labor.¹²¹⁵ Cal

¹²⁰⁹ Ex. SCE-06V06, p. 5.

¹²¹⁰ Ex. TURN-11 (Defever), pp. 4-6.

¹²¹¹ Ex. SCE-06V06, p. 20.

¹²¹² Ex. SCE-06V06E5, p. 20E5.

¹²¹³ Ex. TURN-11 (Defever), pp. 5-6.

¹²¹⁴ Ex. TURN-11 (Defever), p. 6.

¹²¹⁵ Ex. TURN-11 (Defever), pp. 5, 7; Ex. SCE-06V06, p. 20 (Figure II-7).

Advocates also recommended a reduction to SCE's non-labor forecast for different reasons than TURN, proposing a total of \$15.528 million for Environmental Programs O&M.¹²¹⁶

As explained in Ex. SCE-30, TURN, Cal Advocates, and SCE now jointly recommend a forecast of \$19.270 million for Environmental Programs O&M, including \$1.329 million in labor and \$17.941 million in non-labor.¹²¹⁷ This forecast is \$3.418 million less than SCE's non-labor request, which was the area disputed by TURN and Cal Advocates. Exhibit SCE-30 also presents the forecast for Environmental Management and Development now jointly recommended by TURN, Cal Advocates, and SCE, of \$18.539 million, including \$2.566 million in labor and \$15.973 million in non-labor.¹²¹⁸

During evidentiary hearings, SCE presented a statement of counsel to clarify the separation in this GRC between the Environmental Services activities and forecast and SCE's forecast for environmental support for vegetation management, which is a change from the 2021 GRC. SCE attorney Kris Vyas stated, "The stipulation in Exhibit SCE-30 relates to Exhibit SCE-06, Volume 6 on environmental services. It does not address revenues that are tracked in the Vegetation Management Balancing Accounting. Environmental services and the vegetation management [sic] are two separate areas in the rate case."¹²¹⁹ TURN appreciates this clarification and agrees that it is important for the GRC decision to make clear that the costs and activities covered by the authorized forecast for Environmental Services (Exhibit SCE-06, Volume 6) will not be treated as Vegetation Management costs. This clarification will avoid any

¹²¹⁶ Ex. SCE-30 (Environmental Services Stipulation), p. 2.

¹²¹⁷ Ex. SCE-30 (Environmental Services Stipulation), p. 2.

¹²¹⁸ Ex. SCE-30 (Environmental Services Stipulation), pp. 1-2.

¹²¹⁹ 17 RT 1556: 8-13 (SCE/Vyas).

confusion regarding which costs SCE is authorized to record to the Vegetation Management Balancing Account or the other wildfire mitigation-related accounts in which SCE may track costs related to vegetation management.¹²²⁰

In this GRC, Environmental Programs include activities to support compliance with environmental requirements relating to storm water management, air quality permitting, environmental clearance, hazardous waste management, spill prevention control and countermeasures, hazardous materials management and marine mitigation programs.¹²²¹ Unlike the 2021 GRC, SCE's Environmental Programs request in this GRC does not include the costs associated with Environmental Support for Vegetation Management."¹²²² Instead, SCE presents environmental support for vegetation management costs with its Vegetation Management request in Ex. SCE-02V10.¹²²³ As SCE explains, "In this 2025 GRC, SCE is separating environmental work that supports Vegetation Management activities from environmental work that supports other areas of the company (as described in exhibit SCE-06, Vol. 6, Section II)."¹²²⁴ As a result, SCE has charged costs for environmental support for Vegetation Management to the Vegetation Management BPE, and those costs are not reflected in SCE's Environmental Services BPE testimony.¹²²⁵ However, SCE's Environmental Management and Development request does include labor costs for "oversight of increased O&M work in the field primarily for vegetation

¹²²⁰ See Ex. SCE-07V01, pp. 32-33 (presenting SCE's proposal for the Vegetation Management Balancing Account); D.24-03-008, pp. 7-9 (discussing SCE's Wildfire Mitigation Plan Memorandum Account and Fire Risk Mitigation Memorandum Account).

¹²²¹ Ex. SCE-06V06, p. 17.

¹²²² D.24-03-008, p. 43; Ex. SCE-06V06, p. 7.

¹²²³ Ex. SCE-02V10, p. 19, Table II-79, pp. 93-103.

¹²²⁴ Ex. SCE-02V10, p. 94.

¹²²⁵ Ex. SCE-06V06, p. 7.

management" and otherwise to support "a significant increase in O&M work to support strategic grid resiliency and green energy initiatives including vegetation management," among other activities.¹²²⁶

Consistent with SCE's testimony and statement of counsel, and to avoid any confusion, TURN recommends that the Commission clarify that the forecast authorized for Environmental Services O&M covers the costs and activities included in SCE's Environmental Services request presented in Exhibit SCE-06V06, none of which are eligible for tracking in SCE's VMBA or other wildfire mitigation accounts for potential future cost recovery. This clarification will ensure that the full GRC forecast authorized for activities subject to the VMBA or other wildfire mitigation accounts is known and can be used to assess the incrementality of any additional costs that SCE may seek to recover in the future.¹²²⁷ It will also ensure that SCE has the appropriate incentive to manage its GRC-funded Environmental Services O&M costs.

32.2 Environmental Services Capital

TURN did not address SCE's forecasted Environmental Services capital expenditures. However, as explained in Exhibit SCE-30, TURN, Cal Advocates, and SCE now jointly recommend 2023-2025 capital expenditures of \$7.375 million, which reflects a compromise between the recommendations of Cal Advocates and SCE.¹²²⁸

¹²²⁶ Ex. SCE06V06, pp. 14-15.

¹²²⁷ In D.24-03-008, the Commission adopted specific requirements for SCE's showing of incrementality in future reasonableness reviews of wildfire mitigation-related costs, including "detailed information regarding program or activity level approved costs," with "page numbers, compared to actual expenditures and expenses, at the program and activity level." (D.24-03-008, pp. 19, 73).

¹²²⁸ Ex. SCE-30 (Environmental Services Stipulation), p. 3.

32.3 SDG&E Request For SONGS-Related Cost Recovery Re: Marine Mitigation

33. AUDIT SERVICES

34. ETHICS AND COMPLIANCE

35. SAFETY PROGRAMS

36. ENTERPRISE OPERATIONS

36.1 Transportation Services Department

36.2 Facilities And Land Operations

SCE forecasts 2023-2028 capital expenditures of roughly \$1.5 billion (nominal dollars) for Facilities and Land Operations, including projects falling into five categories: Infrastructure Upgrades, Facility Repurpose Projects, Substation Reliability Upgrades, Facility Management Capital Projects, and the Land Operations Program.¹²²⁹ TURN addressed five of SCE's proposed capital projects in testimony, including two Infrastructure Upgrades, the Edison Training Academy and Vehicle Maintenance Facilities, two Facility Repurpose Projects, Alhambra Regional Operations Facility Renovations and Westminster Combined Facility Renovations, and one Land Operations project, the San Jacinto Laydown Yard.¹²³⁰ SCE previously received funding in at least one prior GRC for the first four of these projects without completing the work, and continues to be behind schedule on all five.

¹²²⁹ Ex. SCE-06V07, p. 51. A list of all capital projects included in SCE's request appears in Table II-4 at SCE-06V07, p. 52.

¹²³⁰ Ex. TURN-11 (Defever), pp. 6-14.

Under similar circumstances in SCE's 2021 GRC, the Commission addressed SCE's renewed funding requests for several Facility and Land Operations capital projects. The Commission recounted, "[W]hile the Commission has on numerous occasions reduced or disallowed costs of activities that were requested and included in prior GRC authorizations, the question of whether to approve a renewed funding request is fact-specific and must be evaluated on a case-by-case basis."¹²³¹ As explained below, TURN recommends zero funding for these five projects because SCE previously received funding in at least one prior GRC, if not two, without completing the work, and/or SCE has not demonstrated that the projects will become operational during this GRC.¹²³² For each project, TURN also offers an alternative recommendation in the event the Commission concludes that the project should be included in this GRC. In that case, TURN recommends that the Commission disallow the contingency included in SCE's capital forecast because contingencies are not known and measurable. TURN notes that Cal Advocates also takes issue with SCE's request for each of these five projects but proposes different remedies than TURN.¹²³³

The following table summarizes the differences between TURN's primary and alternative recommendations and SCE's requests.

¹²³¹ D.21-08-036, pp. 451-452.

¹²³² SCE did not receive prior funding for the San Jacinto Laydown Yard.

¹²³³ Ex. CA-22 (Weaver), pp. 16 (recommending no funding for the Edison Training Academy with costs recorded to a memorandum account for recovery when the project is completed); 16 (recommending a 20% reduction in funding for Vehicle Maintenance Facilities); 24 (recommending a 20% reduction in funding for the Alhambra Regional Operations Facility Renovations); 25 (recommending a 20% reduction in funding for the Westminster Combined Facility Renovations); 29 (recommending a 20% reduction in funding for the San Jacinto Laydown Yard).

Facilities and Land Operations - 2023-2028 Capital (Nominal \$000)						
		SCE -		TURN -		
Project	SCE - Total	Contingency	TURN	Alternative		
Edison Training Academy (incl. T&D Equipment)	\$277,199	\$11,000	\$0	\$266,199		
Vehicle Maintenance Facilities	\$36,563	\$2,000	\$0	\$34,563		
Alhambra Regional Operations Facility Renovations	\$93,494	\$4,810	\$0	\$88,684		
Westminster Combined Facility Renovations	\$62,900	\$3,126	\$0	\$59,774		
San Jacinto Laydown Yard	\$22,857	\$1,000	\$0	\$21,857		
Total TURN Adjustments			\$493,013	\$21,936		

36.2.1 Edison Training Academy (formerly, T&D Training Center)

• The Commission should deny SCE's third request for funding for the Edison Training Academy, given prior funding authorizations in the 2018 and 2021 GRCs and ongoing project delays. Alternatively, if the Commission concludes the funding for this project is appropriate, the Commission should disallow SCE's requested \$11 million contingency.

SCE seeks funding to relocate and consolidate its T&D training operations from Chino,

Westminster, and Alhambra, creating the Edison Training Academy. In the 2021 GRC, SCE proposed to relocate and consolidate its T&D training operations at its Rancho Vista site, located in the City of Rancho Cucamonga. Due to conditions mandated by the City of Rancho Cucamonga, SCE has since decided not to pursue the Rancho Vista site and is instead pursuing an SCE-owned site in Corona, California for the consolidated T&D training site.¹²³⁴ SCE now

requests \$277.199 million for the Edison Training Academy (2023-2028), including a risk

¹²³⁴ Ex. SCE-06V07, pp. 53-54.

contingency factor of 9.5% (\$11.0 million) specific to this project.¹²³⁵ The Commission should deny this funding, or alternatively, at least disallow the contingency.

36.2.1.1 The Commission Should Deny Funding for this Project.

This is the third GRC in which SCE has requested funding for this project, formerly called the "T&D Training Center". SCE did not spend the authorized funding in the prior two GRCs. In the 2018 GRC, the Commission authorized \$92 million for the T&D Training Center.¹²³⁶ From 2017 to 2018, SCE unsuccessfully attempted to purchase land for the project.¹²³⁷ Through 2019, SCE had only spent \$2.132 million on the T&D Training Center.¹²³⁸ In the 2021 GRC, SCE requested \$45.285 million for the T&D Training Center. The Commission approved "SCE's 2019 recorded and 2020-2021 capital expenditure forecast for the T&D Training Center" with the expectation that the project would "move forward as planned."¹²³⁹ The Commission noted, "There also does not appear to be any reason to suspect this project will continue to be delayed, since SCE has now secured a site for the new training center and has commenced planning and engineering work for the project." ¹²⁴⁰ Nonetheless,

¹²³⁵ Ex. SCE-06V07, p. 53; Ex. TURN-11 (Defever), p. 7. This cost estimate includes the Edison Training Academy plus related IT Infrastructure and Equipment, which SCE requests in Ex. SCE-06V07, plus T&D Equipment costs for the Edison Training Academy included in Ex. SCE-02V05.

¹²³⁶ D.21-08-036, p. 448 (discussing the authorization in the 2018 GRC).

¹²³⁷ Ex. TURN-11 (Defever), p. 7.

¹²³⁸ D.21-08-036, p. 449.

¹²³⁹ D.21-08-036, p. 454.

¹²⁴⁰ D.21-08-036, p. 453.

SCE spent only \$3.6 million during 2018-2022, despite the second round of funding authorized in the 2021 GRC.¹²⁴¹

Considering this history, TURN witness Defever observed, "The Company's pattern of requesting funds for the project, collecting funds from ratepayers for the project, spending only a fraction of those funds on the project, and again requesting funds for the project decreases confidence that the project will be completed as requested."¹²⁴² Because SCE has not provided sufficient reason to believe that the requested funds would be spent on the authorized project, TURN recommended in testimony that the Commission deny SCE additional funding in this GRC for the Edison Training Academy.¹²⁴³

In rebuttal testimony, SCE asserts that project delays arose from circumstances beyond its control impacting the City of Corona's permitting process and timelines, but it has prioritized the work it could accomplish during the 2021 GRC cycle given the disruptions caused by the COVID-19 pandemic.¹²⁴⁴ SCE also states it has now "received the requisite permits, initiated Phase Zero and Phase 1 construction, and the project remains on track for completion on or before December 31, 2028."¹²⁴⁵ SCE notes that it recorded \$4.5466 million on activities to further this project in 2022-2023 and has spent a total of \$6.638 million so far.¹²⁴⁶

¹²⁴¹ Ex. TURN-11 (Defever), p. 7 (also noting that SCE spent \$0.200 million prior to 2018 for this project, bringing the total to \$3.8 million).

¹²⁴² Ex. TURN-11 (Defever), p. 8.

¹²⁴³ Ex. TURN-11 (Defever), p. 9.

¹²⁴⁴ Ex. SCE-17V05, pp. 12-13.

¹²⁴⁵ Ex. SCE-17V05, p. 14.

¹²⁴⁶ Ex. SCE-17V05, p. 13.

TURN recognizes that SCE recorded \$2.801 million for this project in 2023, bringing the total spending to \$6.6 million.¹²⁴⁷ But this is a fraction of what was already collected from the 2018 GRC and 2021 GRCs. Moreover, SCE's 2023 spending was only 1/3 of SCE's forecast for 2023 in this GRC of \$8.430 million.¹²⁴⁸ Given this continued delay and in consideration of the funding ratepayers have previously provided in the past two GRCs, the Commission should question the reliability of SCE's projected 2028 completion date and exclude funding for the Edison Training Center from this GRC. SCE can seek cost recovery for this project in its next GRC.

36.2.1.2 Alternatively, the Commission Should Disallow the Contingency.

SCE included a 9.5% contingency in its forecast for the Edison Training Academy, based on the risk assessment conduct by SCE in consultation with Cumming Management Group, Inc. (CMGI), the construction cost consulting firm SCE engages to create planning estimates for facility capital projects.¹²⁴⁹ According to SCE, the "project contingency accounts for risk of the need for additional expenditures."¹²⁵⁰ The Commission has considered utility requests for contingencies in cost forecasts in a number of proceedings over time and when challenged, has generally reviewed contingencies with skepticism because they reduce utility incentives to control costs. As the Commission recounted most recently in D.24-03-042, "It has long been our practice, consistent with ratemaking policy, to disallow contingencies in order to motivate

¹²⁴⁷ Ex. SCE-17V05, p. 9 (Table II-9).

¹²⁴⁸ Ex. SCE-17V05, p. 9 (Table II-9)(2023 Recorded); Ex. SCE-06V07, p. 53, Table II-6 (2023 Forecast for Edison Training Academy plus IT Infrastructure and Equipment).

¹²⁴⁹ Ex. SCE-06V07, p. 44; Ex. TURN-11 (Defever), p. 7.

¹²⁵⁰ Ex. SCE-06V07, p. 46.

utilities to remain within their forecast budgets for their capital projects, and wherever possible to "do it for less" as a way to benefit both ratepayers and shareholders."¹²⁵¹

For example, in the SCE 2021 GRC, the Commission considered SCE's request for contingency factors in the Seismic Assessment and Mitigation Program, which includes assessment of SCE's electric and non-electric facilities, generation infrastructure, and telecommunications/IT infrastructure to identify necessary seismic mitigations, followed by implementation of necessary retrofits and improvements.¹²⁵² SCE's forecast included, in pertinent part, a 35 percent contingency for the assessment and retrofit of 16 "Mechanical Electrical Equipment Rooms" within the Transmission Substation Mitigation category and a 1.5 percent contingency rate for the Non-Electric Facilities category.¹²⁵³ The Commission rejected contingencies for these projects, despite SCE's claim that use of contingency factors "is an industry standard practice to account for unknown or unforeseen conditions," as well as SCE's suggestion that the proposed seismic mitigation projects at transmission substations carry a particular risk of unforeseen field conditions during the construction phase.¹²⁵⁴ The Commission reasoned:

As explained in D.19-05-020, budgeting for contingencies is not necessarily appropriate in the context of a general rate case, where the utility must demonstrate the reasonableness of every dollar in its forecasted revenue requirement. Since contingency allowances are, by SCE's own admission, intended to cover "unforeseen conditions," these amounts are also unpredictable,

¹²⁵¹ D.24-03-042 (Cal Water 2023 GRC), p. 25.

¹²⁵² D.21-08-036, pp. 325-326.

¹²⁵³ D.21-08-036, pp. 327-328 (discussing Cal Advocates' and TURN's objections to SCE's requests for contingencies).

¹²⁵⁴ D.21-08-036, pp. 330-331.

and therefore, we find that SCE has not established these costs to be reasonable.¹²⁵⁵

The Commission further emphasized that disallowing all of SCE's requested contingencies – including the 35 percent and 15 percent contingencies – "should motivate SCE to remain within its forecast budgets for these projects.¹²⁵⁶ Finally, the Commission noted, "If additional funds become necessary SCE may seek to establish that necessity in the next GRC."¹²⁵⁷

Likewise in SCE's TY 2018 GRC, the Commission considered, and rejected, SCE's request for a range of contingencies for Capitalized Software projects. The Commission looked beyond SCE's claims that inclusion of contingencies in project cost estimates for IT is routine and in line with industry standards, focusing instead on whether contingencies are appropriate in general rate cases. The Commission considered SCE's responsibility to "demonstrate the reasonableness of every dollar in its revenue requirement" in a GRC, a showing absent given SCE's justification that "contingencies are necessary for the 'uncertainties and variables that are unknown'."¹²⁵⁸ The Commission further considered the balance of ratepayer and shareholder interests in the GRC framework, where "any savings the utility can generate between general rate cases belong to the shareholders," and "in exchange for this opportunity, the shareholders take on the burden of added expenses it may incur during a rate case cycle."¹²⁵⁹ The Commission explained why requiring SCE to "forecast what it projects to be a reasonable

¹²⁵⁵ D.21-08-036, p. 331.

¹²⁵⁶ Id.

¹²⁵⁷ Id.

¹²⁵⁸ D.19-05-020, p. 150.

¹²⁵⁹ D.19-05-020, p. 151.

expense" balances the interests of ratepayers and shareholders in the GRC framework.¹²⁶⁰ The Commission then concluded, "Consistent with ratemaking policy, disallowing these contingencies should motivate SCE to remain within its forecast budgets for these projects. If additional funds become necessary, SCE may seek to establish that necessity in the next GRC."¹²⁶¹

On the other hand, the Commission has authorized contingencies under certain circumstances. For instance, in D.06-11-048, the Commission granted a five percent contingency factor for PG&E on total project cost for the Humboldt powerplant.¹²⁶² Similarly, the Commission authorized a five percent contingency factor for SCE on total project cost for the Mountainview powerplant.¹²⁶³ In D.10-02-032, the Commission granted PG&E a 7.9 percent contingency on its new Advanced Metering Infrastructure ("AMI") system and a 12.9 percent contingency on its Smart Meter upgrade project, but refused any contingency amount for PG&E's Dynamic Pricing programming project.¹²⁶⁴ In D.19-09-051, the Commission authorized SoCalGas's request for Pipeline Safety Enhancement Plan (PSEP) project contingencies, but limited contingencies to an average of 15 percent.¹²⁶⁵

¹²⁶⁰ D.19-05-020, pp. 151-152.

¹²⁶¹ D.19-05-020, p. 152. *See also* D.19-09-025 (PG&E 2019 Gas Transmission and Storage Rate Case), pp. 229-230 (agreeing with TURN that PG&E's request for a contingency factor in its forecast for the Right-of-Way Maintenance program for gas transmission system pipeline facilities should be denied); D.14-08-032 (PG&E 2014 GRC), pp. 130, 135-136 (denying PG&E's requests for two contingency factors: a \$7.2 million contingency for constructing training facilities and an 18 percent contingency factor for the Electric Distribution (Electric and Gas) and Workforce Mobilization and Scheduling project).

¹²⁶² D.06-11-048, pp. 21-22.

¹²⁶³ D.03-12-059, p. 49.

¹²⁶⁴ D.10-02-032, pp. 128-129.

¹²⁶⁵ See D.19-09-051, pp. 205 (hydrotest projects), 215 (capital PSEP projects).

Generally speaking, where stakeholders have challenged utility requests for contingencies, the Commission's approach has been one of caution. The Commission has properly sought to avoid disturbing the balance of interests between shareholders and ratepayers embedded in cost of service ratemaking.¹²⁶⁶ This same balance should guide the Commission here in considering SCE's proposed contingencies. With capital project costs included in GRC base rates, the larger the contingency factor, the less risk to the shareholders because the ratepayers are covering more of the identified uncertainties. Shifting more risk to ratepayers reduces the utility's incentive to control costs. As such, the Commission should deny SCE's request for a 9.5% contingency for the Edison Training Academy if it authorizes any funding for the project in this GRC.

In rebuttal testimony, SCE argues that the use of contingencies is "an industry-standard practice that accounts for unforeseen conditions arising during the construction phase."¹²⁶⁷ However, the Commission made clear in the 2018 GRC and 2021 GRC that the question is not whether the inclusion of contingencies is routine and in line with industry standards.¹²⁶⁸ Rather, the question is whether contingencies are appropriate in general rate cases given the utility's burden and the balance of ratepayer and shareholder interests in the GRC framework, where "any savings the utility can generate between general rate cases belong to the shareholders," and "in exchange for this opportunity, the shareholders take on the burden of added expenses it may incur during a rate case cycle."¹²⁶⁹

¹²⁶⁶ See D.24-03-042, pp. 25-26 (quoting D.19-05-020, which in turn quoted D.85-03-042).

¹²⁶⁷ Ex. SCE-17V05, p. 14.

¹²⁶⁸ See D.19-05-020, pp. 150-152; D.21-08-036, pp. 330-331.

¹²⁶⁹ D.19-05-020, p. 151.

SCE further argues that TURN's challenge to contingencies should be rejected "since SCE is using the same estimating methodology previously found reasonable by the Commission" in the 2021 GRC decision.¹²⁷⁰ In the 2021 GRC, TURN challenged the sufficiency of SCE's total cost estimate for each project; TURN did not challenge the inclusion of a contingency specifically. As the Commission recounts in D.21-08-036, "TURN also asserts that SCE failed to meet its burden to justify the cost of each project: in response to a request for additional supporting documentation, SCE provided a single page cost summary from CCMI without any specific bids, contracts, invoices, or other supporting documentation."¹²⁷¹ The Commission addressed the concern TURN raised, finding that SCE's estimates were "sufficiently detailed and supported, and the estimated level of costs reasonable."¹²⁷² The Commission did not resolve any dispute over a contingency, as none was raised.

In this GRC, TURN has specifically challenged SCE's inclusion of a contingency for facilities construction projects, and the Commission must consider the merits of that challenge as presented here. For the foregoing reasons, the Commission should find that SCE's requested contingency should be denied.

36.2.2 Vehicle Maintenance Facilities

• The Commission should deny SCE's third request for funding for the Vehicle Maintenance Facilities project, given prior funding authorization in the 2018 and

https://docs.cpuc.ca.gov/PublishedDocs/Efile/G000/M347/K127/347127665.PDF.

¹²⁷⁰ Ex. SCE-17V05, pp. 15-16.

¹²⁷¹ D.21-08-036, p. 449. See also TURN Opening Brief, filed in A.19-08-013 on 9/11/20, pp. 234-235 (challenging SCE's support for T&D Training Center cost estimate), 235-236 (challenging SCE's support for Vehicle Maintenance Facilities cost estimate), 237 (challenging SCE's support for Devers Maintenance and Test Building cost estimate), 238 (challenging SCE's Rector Maintenance and Test Building cost estimate) available at

¹²⁷² D.21-08-036, pp. 453-454 (T&D Training Center), 455 (Devers and Rector Maintenance and Test Buildings).

SCE's failure to meet the Commission's requirements for additional funding in the 2021 GRC. Alternatively, if the Commission concludes the funding for this project is appropriate, the Commission should disallow SCE's requested \$2 million contingency.

SCE forecasts total 2023-2028 expenditures of \$36.563 million to renovate the vehicle maintenance facilities at the Orange Coast and Montebello service centers and demolish and rebuild the vehicle maintenance facility at the Ventura service center.¹²⁷³ SCE's forecast includes a risk contingency factor of 7.1% (\$2.0 million) specific to this project.¹²⁷⁴ The Commission should deny this funding, or alternatively, at least disallow the contingency.

36.2.2.1 The Commission Should Deny Funding for this Project.

This is SCE's third request for funding to renovate these three vehicle maintenance facilities. In the 2018 GRC, SCE requested and the Commission authorized \$22.374 million.¹²⁷⁵ SCE then spent \$0 from 2018-2020.¹²⁷⁶ SCE again requested \$22.646 million in the 2021 GRC, but the Commission denied that request, explaining that it would "not authorize additional funding for this project without some showing that progress has been made."¹²⁷⁷ The Commission noted that the delays associated with the project were "entirely within SCE's control."¹²⁷⁸

¹²⁷³ Ex. SCE-06V07, pp. 52, 67. This amount includes both the Vehicle Maintenance Facilities project and related IT Infrastructure and Equipment.

¹²⁷⁴ Ex. TURN-11 (Defever), p. 10.

¹²⁷⁵ Ex. TURN-11 (Defever), p. 9.

¹²⁷⁶ Ex. TURN-11 (Defever), p. 9.

¹²⁷⁷ D.21-08-036, pp. 449, 454.

¹²⁷⁸ D.21-08-036, p. 454.

The Commission should find that SCE has failed to meet its burden of demonstrating the reasonableness of additional funding at this time. As TURN explained in testimony, TURN asked SCE to explain in detail what specific progress has been made since the last GRC, and SCE's response provided a bulleted list of tasks it has completed. ¹²⁷⁹ However, in 2021 and 2022, SCE spent only \$120,000 and \$246,000, respectively, a fraction of what was already collected from the 2018 GRC.¹²⁸⁰ As such, it does not appear significant progress has been made. Further, SCE anticipates that the project will not be completed until 2028 and vehicles will not be sent to the three new facilities until 2029, the year after this GRC cycle ends. For this reason, SCE did not include any cost reductions in this GRC from no longer needing to send vehicles to alternative service facilities.¹²⁸¹ Finally, it is unclear whether funds authorized in this GRC would be used for this project, as SCE acknowledges that other pending Facility and Land Operations projects are higher priority.¹²⁸²

In rebuttal testimony, SCE explains that it spent \$720,000 in 2023, more than its 2023 estimate of \$500,000, and maintains that the projects are on schedule to be completed by the end of this GRC cycle, despite permitting process backlogs, the need to solicit construction bids, and, of course, the realities of construction.¹²⁸³ While the proverbial ball may now be rolling, it is rolling very slowly. SCE has spent just over \$1 million on this project, first presented and

¹²⁷⁹ Ex. TURN-11 (Defever), p. 10.

¹²⁸⁰ Ex. TURN-11 (Defever), p. 10.

¹²⁸¹ Ex. TURN-11 (Defever), p. 10.

¹²⁸² Ex. SCE-06V07, p. 65.

¹²⁸³ Ex. SCE-17V05, pp. 19-20.

funded in the 2018 GRC. The Commission should find that SCE is yet to make sufficient progress for additional funding, as required by D.21-08-036.¹²⁸⁴

36.2.2.2 Alternatively, the Commission Should Disallow the Contingency.

SCE's forecast includes a risk contingency factor of 7.1% (\$2.0 million) for the Vehicle Maintenance Facilities project, based on the risk assessment conduct by SCE in consultation with CMGI, the construction cost consulting firm SCE engages to create planning estimates for facility capital projects.¹²⁸⁵ According to SCE, the "project contingency accounts for risk of the need for additional expenditures."¹²⁸⁶

TURN addresses the inappropriateness of requiring ratepayers to pay for contingencies in responding to SCE's proposed contingency for the Edison Training Academy above. As TURN explains there, the Commission has previously sought to avoid disturbing the balance of interests between shareholders and ratepayers embedded in cost of service ratemaking when stakeholders have challenged contingencies. The Commission has been mindful of the fact that contingencies shift risk for identified uncertainties from shareholders to ratepayers and reduce the utility's incentive to control costs. To avoid disturbing this balance, the Commission should deny SCE's request for a 7.1% contingency for the Vehicle Maintenance Facilities if it authorizes any funding for the project in this GRC.

In rebuttal testimony, SCE repeats the same arguments in defense of its contingency as for the Edison Training Center, specifically that the Commission found SCE's cost estimating

¹²⁸⁴ D.21-08-036, p. 454.

¹²⁸⁵ Ex. SCE-06V07, p. 44; Ex. TURN-11 (Defever), p. 10.

¹²⁸⁶ Ex. SCE-06V07, p. 46.

methodology reasonable in the last GRC and should accordingly reject TURN's challenge to the contingency here.¹²⁸⁷ TURN has already addressed these arguments above and does not repeat them here.

36.2.3 Alhambra Regional Operations Facility Renovations

• The Commission should deny SCE's second request for funding for the Alhambra Regional Operations Facility Renovations, given the prior funding authorization in the 2021 GRC and ongoing project delays. Alternatively, if the Commission concludes the funding for this project is appropriate, the Commission should disallow SCE's requested \$4.810 million contingency.

SCE forecasts total 2023-2028 capital expenditures of \$93.494 million to renovate 6 of

the 12 buildings located at the Alhambra Regional Operations Facility.¹²⁸⁸ SCE claims that the

project is necessary as the average building age is 87 years old and the average FCI scores of the

six buildings is 19% (Poor Condition).¹²⁸⁹ SCE's forecast includes a risk contingency factor of

8.9% (\$4.810 million) specific to this project.¹²⁹⁰ The Commission should deny this funding, or

alternatively, at least disallow the contingency.

36.2.3.1 The Commission Should Deny Funding for this Project.

In the 2021 GRC, SCE forecasted \$58.608 million for this project, which was

uncontested and approved by the Commission.¹²⁹¹ However, SCE spent only \$3.815 million

¹²⁸⁷ Ex. SCE-17V05, pp. 21-22.

¹²⁸⁸ Ex. SCE-06V07, pp. 52, 107. This amount includes both the Alhambra Regional Operations Facility Renovations project and related IT Infrastructure and Equipment.

¹²⁸⁹ Ex. SCE-06V07, p. 108.

¹²⁹⁰ Ex. TURN-11 (Defever), p. 12.

¹²⁹¹ Ex. TURN-11 (Defever), p. 13.

from 2018-2022.¹²⁹² Based on SCE's underspending on this project, TURN questioned whether SCE would spend the requested funds on this project if approved and recommended that the Commission disallow the costs for this project.¹²⁹³

In rebuttal testimony, SCE argues that it recorded \$6.380 million in 2023 "to further this project's activities," bringing the total recorded spend to \$10.389 million.¹²⁹⁴ SCE claims that this spending shows SCE's commitment to completing this project as forecast during this GRC cycle and meeting the in-service date of December 31, 2028.¹²⁹⁵ But SCE fell much further behind its anticipated schedule in 2023, when SCE forecast spending \$22.913 million "to obtain permits and begin construction of the prefabricated warehouse and the new parking layout, Building D demolition and, planning for the Building AD battery storage expansion."¹²⁹⁶ SCE explains that permitting delays at the City of Alhambra have slowed this project down. SCE also cites the need for state and federal approval of soil contamination remediation plans as the cause of a nine month delay in 2023 to the schedules for the Building D demolition and new warehouse construction.¹²⁹⁷

Given the years of compounding delays, it is unclear whether SCE will be able to finish this project on time, particularly given its construction schedule which runs through 2028.¹²⁹⁸ Based on this history, the Commission should decline to provide SCE with additional funding in

¹²⁹² Ex. TURN-11 (Defever), p. 13.

¹²⁹³ Ex. TURN-11 (Defever), p. 13.

¹²⁹⁴ Ex. SCE-17V05, p. 42.

¹²⁹⁵ Ex. SCE-17V05, p. 42.

¹²⁹⁶ Ex. SCE-06V07, p. 107.

¹²⁹⁷ Ex. SCE-17V05, p. 43.

¹²⁹⁸ Ex. SCE-06V07, pp. 110-111.

this GRC. SCE can include the Alhambra Regional Operations Facility Renovations in its next GRC.

36.2.3.2 Alternatively, the Commission Should Disallow the Contingency.

SCE's forecast includes a risk contingency factor of 8.9% (\$4.810 million) for the Alhambra Regional Operations Facility Renovations, based on the risk assessment conduct by SCE in consultation with CMGI, the construction cost consulting firm SCE engages to create planning estimates for facility capital projects.¹²⁹⁹ According to SCE, the "project contingency accounts for risk of the need for additional expenditures."¹³⁰⁰

TURN addresses the inappropriateness of requiring ratepayers to pay for contingencies in responding to SCE's proposed contingency for the Edison Training Academy above. As TURN explains there, the Commission has previously sought to avoid disturbing the balance of interests between shareholders and ratepayers embedded in cost of service ratemaking when stakeholders have challenged contingencies. The Commission has been mindful of the fact that contingencies shift risk for identified uncertainties from shareholders to ratepayers and reduce the utility's incentive to control costs. To avoid disturbing this balance, the Commission should deny SCE's request for a 8.9% contingency for the Alhambra Regional Operations Facility Renovations if it authorizes any funding for the project in this GRC.

In rebuttal testimony, SCE repeats the same arguments in defense of its contingency as for the Edison Training Center, specifically that the Commission found SCE's cost estimating methodology reasonable in the last GRC and should accordingly reject TURN's challenge to the

¹²⁹⁹ Ex. SCE-06V07, p. 44; Ex. TURN-11 (Defever), p. 12.
¹³⁰⁰ Ex. SCE-06V07, p. 46.

contingency here.¹³⁰¹ TURN has already addressed these arguments above and does not repeat them here.

36.2.4 Westminster Combined Facility Renovations

• The Commission should deny SCE's second request for funding for the Westminster Combined Facility Renovations, given the prior funding authorization in the 2021 GRC and ongoing project delays. Alternatively, if the Commission concludes the funding for this project is appropriate, the Commission should disallow SCE's requested \$3.216 million contingency.

SCE forecasts total 2023-2028 capital expenditures of \$62.900 million to renovate the

Westminster Combined Facility to enhance safety, compliance, and efficiency.¹³⁰² SCE's

forecast includes a risk contingency factor of 8.0% (\$3.126 million) specific to this project.¹³⁰³

The Commission should deny this funding, or alternatively, at least disallow the contingency.

36.2.4.1 The Commission Should Deny Funding for this Project.

In the 2021 GRC, SCE forecasted \$26.653 for this project, which was uncontested and

approved by the Commission.¹³⁰⁴ However, SCE spent hardly more than \$3 million from 2018-

2022.¹³⁰⁵ Based on SCE's underspending on this project, TURN questioned whether SCE would

spend the requested funds on this project if approved and recommended that the Commission

disallow the costs for this project.¹³⁰⁶

¹³⁰¹ Ex. SCE-17-V05, pp. 43-44.

¹³⁰² Ex. SCE-06V07, pp. 52, 111-E3-112. This amount includes both the Westminster Combined Facility Renovations project and related IT Infrastructure and Equipment.

¹³⁰³ Ex. TURN-11 (Defever), p. 11.

¹³⁰⁴ Ex. TURN-11 (Defever), p. 11.

 ¹³⁰⁵ Ex. TURN-11 (Defever), p. 11 (citing \$3.015 million in 2018-2022 spending, based on TURN-SCE-031, Q.1); Ex. SCE-06V07, p. 111-E3 (showing \$3.133 million in pre-2023 spending).
 ¹³⁰⁶ Ex. TURN-11 (Defever), p. 11.

In rebuttal testimony, SCE argues that it recorded \$5.787 million in 2023 "to further this project's activities," bringing the total recorded spend to \$8.9 million.¹³⁰⁷ SCE claims that this spending shows SCE's commitment to completing this project as forecast during this GRC cycle and meeting the in-service date of December 31, 2028.¹³⁰⁸ But SCE fell further behind its anticipated schedule in 2023. SCE forecast spending \$7.217 million in 2023 to "continue construction of the yard configuration and begin planning for the MSS and LARS buildings."¹³⁰⁹

Given SCE's track record, the Commission should decline to provide SCE with additional funding in this GRC. SCE can include the Westminster Combined Facility Renovations in its next GRC.

36.2.4.2 Alternatively, the Commission Should Disallow the Contingency.

SCE's forecast includes a risk contingency factor of 8.0% (\$3.126 million) for the Westminster Combined Facility Renovations, based on the risk assessment conduct by SCE in consultation with CMGI, the construction cost consulting firm SCE engages to create planning estimates for facility capital projects.¹³¹⁰ According to SCE, the "project contingency accounts for risk of the need for additional expenditures."¹³¹¹

TURN addresses the inappropriateness of requiring ratepayers to pay for contingencies in responding to SCE's proposed contingency for the Edison Training Academy above. As TURN explains there, the Commission has previously sought to avoid disturbing the balance of interests

¹³⁰⁷ Ex. SCE-17V05, p. 47.

¹³⁰⁸ Ex. SCE-17V05, pp. 47-49.

¹³⁰⁹ Ex. SCE-06V07, p. 111-E3, Table II-25; p. 114.

¹³¹⁰ Ex. SCE-06V07, p. 44; Ex. TURN-11 (Defever), p. 11.

¹³¹¹ Ex. SCE-06V07, p. 46.
between shareholders and ratepayers embedded in cost of service ratemaking when stakeholders have challenged contingencies. The Commission has been mindful of the fact that contingencies shift risk for identified uncertainties from shareholders to ratepayers and reduce the utility's incentive to control costs. To avoid disturbing this balance, the Commission should deny SCE's request for a 8.0% contingency for the Westminster Combined Facility Renovations if it authorizes any funding for the project in this GRC.

In rebuttal testimony, SCE repeats the same arguments in defense of its contingency as for the Edison Training Center, specifically that the Commission found SCE's cost estimating methodology reasonable in the last GRC and should accordingly reject TURN's challenge to the contingency here.¹³¹² TURN has already addressed these arguments above and does not repeat them here.

36.2.5 San Jacinto Laydown Yard

• The Commission should deny SCE's request for funding for the San Jacinto Laydown Yard, given ongoing project delays. Alternatively, if the Commission concludes the funding for this project is appropriate, the Commission should disallow SCE's requested \$1 million contingency.

SCE forecasts total 2023-2028 capital expenditures of \$22.857 million to purchase and improve approximately 20 acres to construct a distribution laydown yard for storage and staging in the San Jacinto region.¹³¹³ SCE currently leases two laydown yards in the San Jacinto district, which are inadequate to meet the area's present and future materials laydown demands.¹³¹⁴ Additionally, SCE states that those leases expire in 2025 and may not be renewed by the

¹³¹² Ex. SCE-17-V05, pp. 50-51.

¹³¹³ Ex. SCE-06V07, pp. 52, 158.

¹³¹⁴ Ex. SCE-06V07, p. 159.

property owner.¹³¹⁵ SCE plans to purchase land, perform due diligence, secure entitlements, improve the site, and construct a building.¹³¹⁶ SCE's forecast includes a risk contingency factor of 8.3% (\$1.0 million) specific to this project.¹³¹⁷ The Commission should deny this funding, or alternatively, at least disallow the contingency.

36.2.5.1 The Commission Should Deny Funding for this Project.

A significant portion of the costs for this project are for the purchase of the land (more than 25%).¹³¹⁸ However, as of January 2, 2024, the Company had not found or identified the land that it intends to purchase.¹³¹⁹ As the identity and cost of the land are not known, this cost is not known and measurable. It is also unknown when or if this purchase will even occur. For example, as discussed above, SCE received funding in the 2018 GRC to purchase land for the T&D Training Center but ultimately did not purchase that land.¹³²⁰ Without the land purchase, there will be no site improvements or building construction for the laydown yard. For these reasons, TURN recommended the disallowance of the entire amount, a reduction of \$22.857 million.¹³²¹

In rebuttal testimony, SCE recounted its "due diligence for numerous properties," including submitting Letters of Intent to purchase properties in August 2023 and December

¹³¹⁵ Ex. SCE-06V07, p. 159.

¹³¹⁶ Ex. SCE-06V07, p. 160.

¹³¹⁷ Ex. TURN-11 (Defever), p. 13.

¹³¹⁸ Ex. SCE-06V07 WP Book C, p. 85.

¹³¹⁹ Ex. TURN-11 (Defever), p. 13.

¹³²⁰ Ex. TURN-11 (Defever), p. 7.

¹³²¹ Ex. TURN-11 (Defever), p. 14.

2023, but the respective property owners withdrew their offers to sell during the due diligence process.¹³²² SCE clarified that it "located a suitable parcel in Hemet with an existing warehouse" in October 2023, submitted a Letter of Intent in April 2024, and has now commenced due diligence.¹³²³ While SCE "expects to occupy the property later this year," the property owners could backout, as happened after SCE's last two Letters of Intent, or SCE could determine that the property will not meet its needs.¹³²⁴

Despite these continuing challenges, SCE suggests that its "efforts to purchase a suitable parcel remain ongoing and consistent with the forecast and schedule presented in its direct testimony."¹³²⁵ Yet SCE recorded no expenditures in 2023, despite forecasting \$508,000 related to site acquisition.¹³²⁶ Additionally, SCE stated, "In the absence of the acquisition of suitable land by *early 2024*, SCE will need to seek lease extensions from property owners and obtain agreements from local jurisdictions to allow for temporary laydown yard use while a permanent solution is sought."¹³²⁷

The Commission should find that the status of this project continues to be worrisome, despite SCE's efforts. As such, the Commission should deny the requested funding. SCE can include the San Jacinto Laydown Yard in its next GRC if the project comes to fruition.

¹³²² Ex. SCE-17V05, p. 60.

¹³²³ Ex. SCE-17V05, pp. 59, 61.

¹³²⁴ Ex. SCE-17V05, p. 59.

¹³²⁵ Ex. SCE-17V05, p. 61.

¹³²⁶ Ex. SCE-06V07, p. 160; Ex. SCE-17V05, p. 58.

¹³²⁷ Ex. TURN-11 (Defever), Appendix B, TURN-SCE-031, Q. 42 (emphasis added).

36.2.5.2 Alternatively, the Commission Should Disallow the Contingency.

SCE's forecast includes a risk contingency factor of 8.3% (\$1.0 million) for the San Jacinto Laydown Yard, based on the risk assessment conduct by SCE in consultation with CMGI, the construction cost consulting firm SCE engages to create planning estimates for facility capital projects.¹³²⁸ According to SCE, the "project contingency accounts for risk of the need for additional expenditures."¹³²⁹

TURN addresses the inappropriateness of requiring ratepayers to pay for contingencies in responding to SCE's proposed contingency for the Edison Training Academy above. As TURN explains there, the Commission has previously sought to avoid disturbing the balance of interests between shareholders and ratepayers embedded in cost of service ratemaking when stakeholders have challenged contingencies. The Commission has been mindful of the fact that contingencies shift risk for identified uncertainties from shareholders to ratepayers and reduce the utility's incentive to control costs. To avoid disturbing this balance, the Commission should deny SCE's request for a 8.3% contingency for the San Jacinto Laydown Yard if it authorizes any funding for the project in this GRC.

In rebuttal testimony, SCE repeats the same arguments in defense of its contingency as for the Edison Training Center, specifically that the Commission found SCE's cost estimating methodology reasonable in the last GRC and should accordingly reject TURN's challenge to the

¹³²⁸ Ex. SCE-06V07, p. 44; Ex. TURN-11 (Defever), p. 13.
¹³²⁹ Ex. SCE-06V07, p. 46.

contingency here.¹³³⁰ TURN has already addressed these arguments above and does not repeat them here.

37. POLICY, EXTERNAL ENGAGEMENT, AND RATEMAKING

37.1 Develop And Manage Policy And Initiatives

37.2 Education, Safety, And Operations

The Education, Safety and Operations GRC activity consists of work performed within SCE's Local Public Affairs organization.¹³³¹ SCE's forecast for Education, Safety and Operations in rebuttal testimony is \$7.630 million, a reduction to its original request of \$7.723 million due to incremental savings associated with SCE's "Operational Excellence" efforts.¹³³² SCE's forecast is an increase of nearly \$1.5 million over 2022 recorded costs of \$6.193 million.¹³³³ This increase covers labor for 3 new positions and filling vacancies plus non-labor for increased stakeholder workshops, increases in costs that had decreased due to COVID-19, and increased distribution work including a specialized consultant for community outreach.¹³³⁴ The Commission should instead adopt TURN's forecast of \$6.193 million, which is the 2022 last recorded year.

As TURN demonstrated in testimony, SCE has consistently underspent on Education, Safety and Operations since 2018, and SCE's recorded costs declined every year from 2019-

¹³³⁰ Ex. SCE-17V05, pp. 62-63.

¹³³¹ Ex. SCE-17V06, p. 5.

¹³³² Ex. SCE-17V06, pp. 6 (Table II-5), 8 (explaining that SCE reduced its forecast by \$0.093 million to account for eliminating two of the vacant administrative assistant positions in this department).

¹³³³ Ex. SCE-17V06, p. 6 (Table II-5); Ex. SCE-06V08, p. 23.

¹³³⁴ Ex. SCE-06V08, pp. 23-25.

2022.¹³³⁵ In fact, both of these trends continued in 2023, as the table below illustrates, despite SCE's forecast in this GRC that spending would increase in 2023 to \$6.795 million.¹³³⁶

Education, Safety and Operations (Constant 2022 \$000s)						
	2018	2019	2020	2021	2022	2023
Authorized	\$9,761	\$9,761	\$9,761	\$8,056	\$8,056	\$8,056
Actual	\$7,519	\$8,131	\$7,819	\$6,319	\$6,193	\$6,001
Underspent	\$2,242	\$1,630	\$1,942	\$1,737	\$1,863	\$2,055

SCE underspent relative to both authorized labor and non-labor amounts in each year from 2018-2023, not only total spending.¹³³⁷

Given SCE's history of overforecasting for this activity, in years preceding and following

the COVID-19 pandemic, the Commission should dismiss SCE's claims that it must increase

spending in 2025 so significantly above recent levels. Authorizing a forecast of 2022 last

recorded year is reasonable in light of the consistent decline in spending from 2019 through

2023.¹³³⁸ It is also higher than SCE's actual spending in 2023 for both labor and non-labor.¹³³⁹

Accordingly, the Commission should adopt TURN's forecast of \$6.193 million for Education,

Safety and Operations, including \$5.615 million for labor and \$0.578 for non-labor.

¹³³⁵ Ex. TURN-11 (Defever), pp. 14-15. As noted in fn. 44 on page 14, the authorized amounts shown are test year authorizations without the increases provided through the post test year adjustment mechanism authorized in the 2018 GRC for 2019 and 2020, and in the 2021 GRC for 2022 and 2023.

¹³³⁶ Ex. TURN-11 (Defever), p. 15; Ex. SCE-06V08, p. 22 (Figure II-5); Ex. SCE-11, Appendix A (2023 Recorded O&M).

 $^{^{1337}}$ Ex. SCE-06V08, p. 22 (Figure II-5, recorded L, NL 2018-2022); Ex. SCE-11, Appendix A (2023 Recorded O&M – L, NL); Ex. TURN-11 (Defever), Appendix B, TURN-SCE-033, Q.04 (L, NL authorized in 2018 and 2021 GRCs).

¹³³⁸ D.04-07-022 (SCE 2003 GRC), p. 15 (quoting D.89-12-057) ("If recorded expenses in an account have shown a trend in a certain direction over three or more years, the [last recorded year] level is the most recent point in the trend and is an appropriate base estimate for [the test year].").

¹³³⁹ Ex. SCE-06V08, p. 22 (Figure II-5, recorded L, NL 2018-2022); Ex. SCE-11, Appendix A (2023 Recorded O&M – L, NL).

37.3 Professional Education And Development

SCE forecasts \$2.113 million for Professional Education and Development, which consists of dues and memberships for seven professional organizations.¹³⁴⁰ In TURN's testimony, TURN opposed part of SCE's request for Edison Electric Institute (EEI) dues, recommending an adjustment of \$0.770 million.¹³⁴¹ TURN additionally opposes SCE's request for \$0.042 million in California Taxpayers Association (CalTax) dues, an issue not addressed in TURN's testimony. With these adjustments, TURN's forecast is \$1.301 million. The Commission should adopt TURN's forecast for Professional Education and Development for the reasons explained below.

37.3.1 The Commission should reduce SCE's forecast for EEI dues.

SCE's Professional Education and Development forecast includes \$1.844 million for Edison Electric Institute (EEI) dues.¹³⁴² The EEI forecast includes \$15,000 for the Restoration, Operations, and Crisis Management Program, which TURN supports. SCE removed from its request the portion of EEI dues identified in the EEI invoice as related to lobbying and charitable activities, including 13% of expenses classified as "Regular Activities of Edison Electric Institute," 20% of "Industry Issues" activities, and 100% of "Contribution to the Edison Foundation."¹³⁴³ TURN recommends a further reduction of \$770,000 because SCE has not met the Commission's clear requirements for demonstrating the reasonableness of requested funding

¹³⁴⁰ Ex. SCE-06V08 WP, p. 27.

¹³⁴¹ Ex. TURN-11, p. 17.

¹³⁴² Ex. SCE-06V08 WP, p. 27. SCE elsewhere refers to its request of \$1.893 million for EEI dues. Ex. SCE-06V08, p. 38; Ex. SCE-17V06, p. 9.

¹³⁴³ Ex. SCE-06V08, pp. 38-39.

for EEI dues.¹³⁴⁴ With TURN's reduction, ratepayers would fund 50% of EEI dues, plus the full amount for Restoration, Operations, and Crisis Management Program.

The Commission has addressed the extent to which ratepayers should pay for a utility's EEI dues in a number of recent GRCs. For example, in D.15-11-021, issued in SCE's 2015 GRC, the Commission reiterated the specific types of activities conducted by EEI for which ratepayers should not pay. Those activities include the following six cost categories, as defined by the National Association of Regulatory Utility Commissioners (NARUC): (1) Legislative Advocacy, (2) Legislative Policy Research, (3) Regulatory Advocacy, (4) Advertising, (5) Marketing, and (6) Public Relations.¹³⁴⁵ In that case, SCE had removed costs labeled "Lobbying" plus "Advertising, Marketing, and Public and Media Relations."¹³⁴⁶ The Commission concluded that the "Lobbying" category overlaps with NARUC's "Legislative Advocacy" category but does not include the categories of "Legislative Policy Research" and "Regulatory Advocacy," which must also be excluded because they are political in nature.¹³⁴⁷ In D.21-08-036, issued in SCE's 2021 GRC, the Commission similarly recounted, "The Commission has specifically barred ratepayer funding of membership activities such as: legislative advocacy, legislative policy research, regulatory advocacy, advertising, marketing, and public relations."¹³⁴⁸ There the Commission noted that SCE had not provided "a breakdown of EEI's membership activities or dues that would enable the Commission to determine how

¹³⁴⁴ Ex. TURN-11 (Defever), p. 17.

¹³⁴⁵ D.15-11-021, pp. 363-366.

¹³⁴⁶ D.15-11-021, p. 365.

¹³⁴⁷ D.15-11-021, pp. 365-366.

¹³⁴⁸ D.21-08-036, p. 462. *See also* D.19-05-020, issued in the SCE 2018 GRC, p. 250 (rejecting SCE's reliance on the EEI invoice to establish which portion of EEI dues ratepayers should fund and concluding that SCE had not met is burden to establish any portion of the dues were recoverable from ratepayers).

much of the dues are attributable to activities the Commission has previously deemed improper for ratepayer recovery."¹³⁴⁹

Where utilities have not met their burden to demonstrate the reasonableness of requested EEI dues, the Commission has authorized a lower forecast. For instance, in SCE's 2015 GRC, the Commission authorized a forecast of \$1.0 million, roughly \$0.5 million less than SCE's request (\$1.462 million) and approximately half of the full dues amount (\$1.922 million).¹³⁵⁰ In SCE's 2018 GRC, the Commission found that SCE had not met its burden to establish that any portion of EEI dues are recoverable from ratepayers and denied recovery.¹³⁵¹ In the 2021 GRC, SCE devoted 7 pages of its direct testimony to describing the many benefits of EEI membership for customers falling into the following categories: disaster preparedness, grid resiliency, customer savings, information exchange, and miscellaneous activities that benefit SCE customers.¹³⁵² Finding SCE's showing sufficient to demonstrate that EEI membership confers "some ratepayer benefits," the Commission determined that "some ratepayer funding for SCE's EEI membership dues" should be approved.¹³⁵³ The Commission authorized the portion of dues for the Restoration, Operations, and Crisis Management Program, plus 50% of the remainder of dues, for a total forecast of \$0.983 million.¹³⁵⁴

In this GRC, SCE devoted 9 pages of its direct testimony to describing the same customer benefits from EEI membership as SCE detailed in its 2021 GRC (disaster preparedness, disaster

¹³⁴⁹ D.21-08-036, p. 462.

¹³⁵⁰ D.15-11-021, pp. 363, 366.

¹³⁵¹ D.19-05-020, p. 250.

¹³⁵² D.21-08-036, p. 461 (citing Ex. SCE-06, Vol. 6, pp. 19-25).

¹³⁵³ D.21-08-036, p. 462.

¹³⁵⁴ D.21-08-036, pp. 462-463.

preparedness, grid resiliency, customer savings, information exchange, and miscellaneous activities that benefit SCE customers).¹³⁵⁵ SCE also shared its conclusion, based on EEI's "2023 Lobbying, Advocacy, and Other Expenditures" report, that "the majority of SCE's EEI dues are going to activities that benefit ratepayers and are thus eligible for ratepayer funding, with the exception of those portions already excluded from SCE's request that are identified in the [EEI] invoice as pertaining to influencing legislation."¹³⁵⁶ SCE points to EEI's budget breakdown by "Business and Policy Issue" and well as by "Department" to support its claim.¹³⁵⁷

SCE overlooks the fact that EEI is "the trade association that represents all U.S. investorowned electric companies," which "provides public policy leadership, strategic business intelligence, and essential conferences and forums" for its member electric companies.¹³⁵⁸ If investor owned electric companies and their ratepayers had the same interests, there would be no need for ratepayer advocates. It is not self-evident that EEI's activities provide concrete benefits to ratepayers. Why should the Commission assume that SCE's ratepayers benefit from the activities of EEI's General Counsel's Office, Government Relations Department, Political & External Affairs Department, or State & Federal Regulatory Affairs Department, for example, when EEI is a utility trade association?¹³⁵⁹ Similarly, the Commission cannot simply assume that SCE's ratepayers benefit from EEI's advocacy related to the specific policy issues addressed by EEI, such as Fuel Diversity and Clean Energy, Grid Investment & Modernization, Finance

¹³⁵⁵ Ex. SCE06V08, pp. 27-34.

¹³⁵⁶ Ex. SCE06V08, pp. 39-40.

¹³⁵⁷ Ex. SCE06V08, pp. 39-40; Ex. SCE-17V06, p. 12 (citing Appendix B, EEI's 2023 Lobbying, Advocacy, and Other Expenditures Report).

¹³⁵⁸ Ex. SCE-17V06, Appendix B, p. B2.

¹³⁵⁹ See Ex. SCE-17V06, Appendix B, p. B4 (EEI Core Budget Expenditures by Department).

and Taxes, and Customer Solutions.¹³⁶⁰ The Commission has disallowed funding for utilities in these same issue areas (among others) when ratepayer advocates demonstrate that the utility has failed to demonstrate the reasonableness of its request.¹³⁶¹ Given the nature of EEI, the Commission's default assumption has been, and should continue to be, that ratepayer funding for EEI dues *should be denied* "unless a utility provides sufficient evidence to establish clear ratepayer benefits."¹³⁶² SCE's reliance on the EEI budget breakdown is misplaced.

In rebuttal testimony, SCE faults TURN for failing to propose a way to calculate the portion of EEI dues that are attributable to the NARUC categories the Commission has previously deemed improper for ratepayer recovery, which, according to SCE, leaves the EEI invoice as "the best evidence available."¹³⁶³ SCE suggests that adopting SCE's forecast, "in the absence of any reasonable alternative from intervenors, would also be consistent with what the Commission did under similar circumstances" in SDG&E's 2019 GRC, citing D.19-09-051.¹³⁶⁴ The Commission should dismiss these errant contentions for two reasons.

First, SCE would inappropriately shift the burden of demonstrating the amount of reasonable funding for EEI dues to TURN, when "It is SCE's burden to establish that requested funds are eligible for rate recovery."¹³⁶⁵ For this reason, the Commission has repeatedly directed

¹³⁶⁰ See Ex. SCE-17V06, Appendix B, p. B3 (EEI Core Budget Expenditures by Issue Area).

¹³⁶¹ See, e.g., D.24-01-004 (denying SCE's application for approval of building electrification programs); D.19-05-020, pp. 370-371 (approving less funding than requested by SCE for Grid Modernization capital expenditures); D.21-10-025, pp. 27-29 (denying, among other things, SCE's requested financing order for O&M expenses and Uncollectibles); D.21-08-036, Section 19.1.1.1 (reducing funding for SCE's Billing Services).

¹³⁶² D.21-08-036, p. 461.

¹³⁶³ Ex. SCE-17V06, pp. 13-14.

¹³⁶⁴ Ex. SCE-17V06, p. 14 (citing D.19-09-051, p. 583).

¹³⁶⁵ D.12-11-051, issued in SCE's 2012 GRC, p. 507 (addressing SCE's request for EEI dues).

SCE to establish that its GRC request excludes all 6 of the NARUC categories and emphasized that the EEI invoice is insufficient to that end, as noted above. Moreover, SCE is an EEI member and as such, has access to information from EEI that TURN does not.

Second, in D.20-07-038, the Commission modified the treatment of EEI dues in D.19-09-051. While the Commission had initially authorized SDG&E's request for EEI dues less the percentage identified on the EEI invoice as for lobbying, the Commission concluded in D.20-07-038 that this treatment was in error because the EEI invoice provided "nothing to indicate whether any other portion of SDG&E's dues was allocated for other activities we have deemed improper for ratepayer funding."¹³⁶⁶ The Commission accordingly modified D.19-09-051 to further limit funding for SDG&E's EEI dues.¹³⁶⁷

Given SCE's showing, which fails to establish that the requested dues will not fund any activities the Commission has deemed improper for ratepayer funding, the Commission should take the same approach it took in SCE's 2021 GRC. The Commission should authorize funding for Restoration, Operations, and Crisis Management Program, plus 50% of the remainder of EEI non-charitable dues, as recommended by TURN. Based on TURN's calculation, this adjustment yields a reduction of \$0.770 million to SCE's request.¹³⁶⁸

¹³⁶⁶ D.20-07-038, p. 7.

¹³⁶⁷ D.20-07-038, p. 7.

¹³⁶⁸ Ex. TURN-11 (Defever), p. 18).

37.3.2 The Commission should disallow ratepayer funding for the California Taxpayer Association.

SCE's forecast includes \$42,156 for CalTax dues.¹³⁶⁹ According to SCE, CalTax's "mission is to protect taxpayers from unnecessary taxes and to promote government efficiency."¹³⁷⁰ SCE claims that CalTax "helps SCE reduce corporate tax liability" and "allows SCE to reduce tax expenses borne by customers."¹³⁷¹

The Commission should disallow recovery of CalTax dues here for the same reason as in prior GRC decisions. In D.12-11-051, issued in SCE's 2012 GRC, the Commission noted that CalTax is "focused on tax policy, not the delivery of electrical service, and ratepayers may disagree with their views or even be adversely affected by them." ¹³⁷² The Commission disallowed funding for CalTax dues because "advancing policies of tax reduction is inherently political and ratepayers should not fund SCE's membership dues in political organizations, regardless of some attenuated potential rate benefit."¹³⁷³ Accordingly, the Commission should reduce SCE's forecast for Professional Development and Education by \$42,156.

¹³⁷³ D.12-11-051, p. 507.

¹³⁶⁹ Ex. SCE-06V08 WP, p. 27. This amount excludes 14% of total dues which CalTax identified as related to lobbying. Ex. SCE-06V08, p. 34.

¹³⁷⁰ Ex. SCE-06V08, p. 34.

¹³⁷¹ Ex. SCE-06V08, pp. 34-35.

¹³⁷² D.12-11-051, p. 507. *See also* D.19-05-020, issued in SCE's 2018 GRC, p. 250 (disallowing funding for California Taxpayer Association, among other organizations, because "SCE has not established the ratepayer benefits of supporting these organizations.").

37.4 Ratemaking Cost Recovery Business Planning Element

38. RESULTS OF OPERATIONS

- **38.1** Results Of Operations
- **38.2** CPUC-Jurisdictional Revenue Requirement
- 38.3 GRC Ratemaking Proposals, Including Memorandum And Balancing Accounts

38.3.1 Memorandum and Balancing Accounts

38.3.1.1 The Commission Should Continue to Rein In the Reliance on Memorandum and Balancing Accounts.

TURN urges the Commission to continue here on the path from the recent PG&E GRC decision, by limiting opportunities for the utility to recover above-authorized costs and, where such an opportunity is provided, relying on an application-based reasonableness review rather than a lesser review or, as SCE has proposed in several key areas, effectively no review whatsoever.

It is more important than ever that Commission-adopted ratemaking mechanisms provide SCE with an effective cost control incentive. Similarly, where the utility is provided an opportunity to recover spending in excess of the forecast adopted here, there must be effective review processes to ensure that the cost control incentive worked. SCE's customers today already face substantial and growing affordability challenges from currently-authorized rates and the further increases likely to result from other pending proceedings. Clearly, the rate impacts will go from bad to worse with this GRC decision, with the only question being how much worse.¹³⁷⁴ The Commission needs to ensure that the utility's management acumen is deployed in full force to not only meet the growing and changing requirements for the scope and scale of its operations, but to do so in a least-cost manner.

The Commission has long recognized that forecast-based cost of service ratemaking serves to give "utility management an opportunity and incentive to find ways to conduct operations for less than projected."¹³⁷⁵ This incentive to manage costs at or below the authorized amount was recognized as essential to successful cost of service ratemaking:

If ratemaking ever becomes so conceptually upside down that utility management loses the economic incentive to exercise its business acumen, California will be in a sad posture and will suffer under utility management which is lethargic with a 'cost plus' mentality.¹³⁷⁶

To be clear, TURN is <u>not</u> arguing here that SCE has completely lost the economic incentive to control its costs. But the Commission should be very concerned with the severe erosion of that incentive in recent years with the growth of memorandum and balancing accounts, both in number and in the extent of utility and operations and spending now subject to those ratemaking mechanisms. If cost-of-service ratemaking is to include appropriate and meaningful incentives to ensure least-cost approaches from a utility, reliance on balancing and memorandum accounts should be more the exception, rather than the expectation.

¹³⁷⁴ SCE's GRC request seeks a \$1.76 billion increase for 2025, followed by increases in the \$593 million to \$627 million range for each of the three years following. The Commission can also anticipate separate applications in the relatively near future for an AMI 2.0 project and a Next Gen ERP project, both of which are certain to have very substantial costs.

¹³⁷⁵ D.96-12-066 (PG&E application seeking extra-GRC costs), 1996 Cal. PUC LEXIS 1111, *5, *quoting* D.85-03-072.

¹³⁷⁶ *Id.*, *5-6, *quoting* D.85-03-072.

Furthermore, the Commission should recognize that the fact that memorandum or balancing account treatment was adopted in the past for a category of costs does not mean that treatment should be maintained forever for such costs. As the utility gains experience with programs and activities, its management should be able to develop forecasts of sufficient accuracy and certainty and operate subject to those forecasts. And with that experience in hand, balancing or memorandum account treatment should no longer be necessary or appropriate due to any purported difficulty of forecasting.¹³⁷⁷

The prevalence of memorandum and balancing account ratemaking also undermines the transparency of the Commission's regulatory process, particularly the confidence the Commission can have in the bill impacts it uses to assess a GRC's impact on customers.¹³⁷⁸ Simply put, the Commission knows from experience that any estimate of the overall rate impact that is tied to the forecasts authorized in a GRC decision is likely to prove significantly understated once the above-authorized spending subject to memorandum or balancing account treatment is factored in. The difference can be quite substantial (\$300 million of above-authorized spending for vegetation management in 2021 alone, for example).¹³⁷⁹ At the time the Commission issued the 2021 GRC decision, it calculated the overall authorized revenue requirement increase over present rates as approximately \$489 million.¹³⁸⁰ Once the full impact

¹³⁷⁷ Ex. TURN-15-E2, pp. 5-6. In the recent PG&E GRC decision, the Commission replaced several key two-way balancing accounts with one-way balancing accounts in recognition of the utility's increased experience with the underlying activities. D.23-11-069, pp. 482-486, and 487-488. Similar outcomes are warranted here.

¹³⁷⁸ Ex. TURN-15-E2, p. 5.

¹³⁷⁹ The 2021 GRC decision authorized \$207 million for SCE's vegetation management programs; SCE recorded \$515 million for that work in 2021. Ex. SCE-02, Vol. 10, p. 4, Figure I-3.

¹³⁸⁰ D.21-08-036, Appendix B – CPUC RO Comparison.

of above-authorized 2021 spending in memorandum and balancing accounts is factored in, the effective increase is likely to be more than double that figure.

TURN urges the Commission to take reasonable steps to rein in the growth of reliance on memorandum and balancing accounts.¹³⁸¹ It should only permit the creation of new accounts or continuation of existing accounts where the utility demonstrates that such action is required by statute or compelling circumstances. And where memorandum or balancing accounts continue to be relied upon, the Commission <u>must</u> ensure an opportunity for close scrutiny of the reasonableness of any above-authorized amounts before rate recovery of those amounts is permitted. To the same end, the Commission must reject SCE's proposals to either do away altogether with reasonableness reviews of above-authorized spending, or to have that review take place either via advice letters or in ERRA proceedings. Without effective review of above-authorized costs for reasonableness, the utility faces little if any cost control incentive whatsoever.

The Commission should continue the progress displayed in the recent decision on PG&E's test year 2023 GRC. There the Commission modified the balancing accounts for wildfire mitigation and vegetation management program costs from two-way balancing accounts to a one-way balancing account going forward.¹³⁸² It also rejected a number of the "substantial and substantively significant revisions to numerous accounts" due to the need for additional time to fully review them.¹³⁸³ The decision also recognized that closing existing accounts "will

¹³⁸¹ Ex. TURN-15-E2, pp. 2-6.

¹³⁸² D.23-11-069 (PG&E TY 2023 GRC), pp. 485-486 and 487-488 (modifying the Wildfire Mitigation Balancing Account and the Vegetation Management Balancing Account to each become a one-way rather than a two-way balancing account).

¹³⁸³ D.23-11-069 (PG&E TY 2023 GRC), p. 728 and Finding of Fact 383.

promote transparency and simplicity."¹³⁸⁴ TURN urges the Commission to recognize that these outcomes represented progress, and the current GRC offers an opportunity to build on that progress.

38.3.1.2 The Commission Should Adopt a "Deductible" that Would Routinely Apply to New Memorandum Accounts.

In a recent decision addressing the Sempra Utilities' request for a new memorandum account that would permit recording of costs incurred due to upcoming changes in federal gas safety-related rules and regulations, the Commission authorized the new memorandum account, but subject to a deductible amount of \$10 million based on the then-effective "Z-Factor allowance" for those utilities.¹³⁸⁵ TURN recommends that the Commission apply this approach more routinely to memorandum account requests, starting with SCE's requests in this GRC.¹³⁸⁶ TURN believes this approach will have a number of beneficial impacts. First, it would stem at least somewhat the erosion in forecast-based ratemaking that has occurred due to the growth in recent years of reliance on memorandum accounts seemingly every time the utility identifies a potential new cost. Second, it should counteract at least somewhat the incentive SCE faces if it is positioned to avoid bearing any "deductible" if it creates a new memorandum account rather than seeking recovery through its existing Z Factor mechanism. TURN knows of no valid policy

¹³⁸⁴ *Id.*, Conclusion of Law 290. The Commission also noted that the requests to close accounts in that GRC were unopposed.

¹³⁸⁵ D.23-05-003, Ordering Paragraph 1. The Sempra Utilities had not proposed recording of the costs subject to their respective Z factor mechanisms, and the Commission did not direct recovery through the existing mechanisms. Rather, it applied a deductible at the level adopted for the Z factor to the new memorandum account.

¹³⁸⁶ Ex. TURN-15-E2, pp. 6-7.

purpose served by permitting SCE to face such a deductible only when Z factor treatment is applied. The Commission should apply the deductible to all memorandum account requests.

SCE opposes the \$10 million deductible, labeling it a vestige of the utility's short-lived experiment with performance-based ratemaking (PBR) that has no place in cost-based ratemaking.¹³⁸⁷ Missing from SCE's position is an acknowledgement that it was the utility that proposed maintaining that deductible when it returned to cost-of-service ratemaking after a six-year hiatus.¹³⁸⁸ The Z-factor deductible has been part of cost-of-service ratemaking since then, and SCE proposed its continuation here.¹³⁸⁹

38.3.1.3 SCE Proposed Changes to Existing Balancing Accounts and Memorandum Accounts

38.3.1.3.1 Wildfire Risk Mitigation Balancing Account/Grid Hardening Balancing Account

The Commission must reject SCE's proposals to adopt lesser levels of review for greater amounts of above-authorized grid hardening spending, and instead adopt TURN's recommended modifications.

SCE currently has a Wildfire Risk Mitigation Balancing Account (WRMBA) that was

authorized in its test year 2021 GRC. SCE had requested that the account cover a broad array of

¹³⁸⁷ Ex. SCE-18, Vol. 1, pp. 3-4.

¹³⁸⁸ D.04-07-022 (SCE TY 2003 GRC), p. 2 and 278-279. The Commission described how the deductible worked in the PBR context: "SCE is at risk for events that do not have a revenue requirement impact of more than \$10 million, and there is a \$10 million 'deductible' applied on a one-time basis to the first year's revenue requirement associated with any approved Z-Factors." After crediting SCE with the proposal to continue the Z-factor and the associated deductible as part of the post-test year ratemaking going forward, the Commission stated it was reasonable to do so "with the return to more conventional cost-of-service ratemaking." *Id.*, pp. 278-279.

¹³⁸⁹ Ex. SCE-07, Vol. 1, p. 34, fn. 46.

wildfire mitigation programs, with the review of above-authorized spending to occur in the utility's ERRA proceeding. As adopted by the Commission, the balancing account focused on a single program (the Wildfire Covered Conductor Program), with an application required for recovery of costs above 110% of the GRC-authorized amount.¹³⁹⁰

Here, SCE proposes to expand the scope of its existing WRMBA to include not just WCCP-related expenditures, but also costs of its Targeted Undergrounding Program (TUG), Rapid Earth Fault Current Limited (REFCL) activities, and Long Span Initiative (LSI) activities. SCE also proposes renaming the account as the Grid Hardening Balancing Account (GHBA) to reflect this broader scope of activities. For rate recovery issues, SCE's primary recommendation is to <u>eliminate</u> the reasonableness review of any amount of above-authorized spending. If the Commission feels some level of review is necessary, SCE proposes an increased threshold of 125%.¹³⁹¹

TURN's testimony recommended the Commission deny SCE's proposed modifications to the ratemaking associated with the WRMBA. TURN contended the utility's primary proposal to eliminate altogether reasonableness review of above-authorized spending is not to be taken seriously.¹³⁹² It is an extreme proposal: If SCE's recorded spending was twice the amount that had been authorized, there still would be no reasonableness review.¹³⁹³ As TURN explained, the same would be true even if the above-authorized spending had been caused by clear instances of imprudent or unreasonable action or inaction on the part of the utility – no reasonableness

¹³⁹⁰ D.21-08-036 (SCE TY 2021 GRC), pp. 247-250.

¹³⁹¹ Ex. SCE-07, Vol. 1, pp. 30-32.

¹³⁹² Ex. TURN-15-E2, pp. 7-9.

¹³⁹³ Id., p. 8, citing Attachment 2 (SCE Response to TURN DR 48-1.a.).

review, just rate recovery.¹³⁹⁴ TURN submits that such an approach cannot possibly satisfy the "just and reasonable" standard of Public Utilities Code 451. TURN also calls for rejection of SCE's alternative proposal to increase the threshold requiring a reasonableness showing from 110% to 125%. Instead, the Commission should adopt ratemaking modifications similar to those adopted for PG&E's wildfire mitigation programs in the test year 2023 GRC decision. As noted earlier, there the Commission authorized continuation of PG&E's WMBA, but as a one-way balancing account.¹³⁹⁵

SCE's direct testimony was served in mid-2023. In November 2023, the Commission issued its decision in PG&E's test year 2023 GRC, where it addressed that utility's proposals regarding its existing Wildfire Mitigation Balancing Account (WMBA). Instead of increasing from 115% to 125% the threshold above which reasonableness reviews were required of above-authorized spending as the utility had requested, the Commission revised the WMBA so it became a one-way balancing account (thereby eliminating the need for a threshold figure of any amount).¹³⁹⁶

SCE's proposed modifications run counter to the degree of utility discretion and control regarding the costs that would be recorded in the new account. The scope and scale of the grid hardening programs, their design, implementation and ongoing operation – every element is subject to management determinations and decisions. SCE is often heard to claim that balancing or memorandum account treatment is necessary due to costs being outside of its control. But for

¹³⁹⁴ *Id.*, p. 8.

 ¹³⁹⁵ As explained further in the Grid Hardening section of this brief (Section 15.2), TURN also recommends that the one-way balancing account for all grid hardening expenditures include a separate one-way sub-account with specific provisions applicable to undergrounding expenditures.
 ¹³⁹⁶ D.23-11-069, pp. 482-486.

⁴⁰²

its grid hardening activities (like its vegetation management activities), that trope should not get the utility very far. And when there is such a degree of utility influence and control over the amount of costs ultimately recorded, if the Commission chooses to permit an opportunity for rate recovery of above-authorized costs, there needs to be a rigorous reasonableness review showing required.

SCE's rebuttal testimony does not shore up its inadequate showing. In an attempt to be afforded treatment radically different from that adopted for PG&E just a few months ago, the utility cites distinctions that should not make a difference in this context.¹³⁹⁷ These are two regulated electric utilities operating in the same state, subject to the same laws and regulations, and operating wildfire mitigation programs of a very similar general nature. More galling is SCE's suggestion that its proposed two-way balancing account "mitigates risk for customers and investors."¹³⁹⁸ Under SCE's primary proposal, it would be permitted to spend uncapped amounts in excess of the forecast found reasonable based on the record of this GRC, and to recover those amounts subject to <u>no</u> reasonableness review. What risk is being mitigated for SCE's customers under that proposal?

38.3.1.3.2 Vegetation Management Balancing Account

As with the GHBA, the Commission must reject SCE's proposals to adopt lesser levels of review for greater amounts of above-authorized spending, and instead adopt TURN's recommended modifications.

¹³⁹⁷ Ex. SCE-18, Vol. 1, p. 14.
¹³⁹⁸ Id., p. 17.

In SCE's test year 2021 GRC, the utility's proposed ratemaking for its Vegetation Management Balancing Account (VMBA) was to have review of any above-authorized amounts recorded in the account occur in SCE's annual ERRA cost review proceeding. As a fallback position, SCE proposed a "soft cap" of 120%, with the reasonableness review of costs above that "cap" occurring through a Tier 3 Advice Letter.¹³⁹⁹ The Commission adopted a two-way balancing account as SCE requested, but rate recovery of costs in excess of 115% of the authorized amount would require a reasonableness review application. The Commission described its approach as "generally consistent" with the outcome adopted in the TY 2020 GRC of PG&E.¹⁴⁰⁰

Key elements of SCE's VMBA ratemaking proposal in this GRC are very similar to those of its proposal for the WRMBA/GHBA, discussed above. Most egregious is SCE's primary recommendation to <u>eliminate</u> the reasonableness review of any amount of above-authorized spending.¹⁴⁰¹ As explained in the previous section, the Commission should deny SCE's proposal to eliminate altogether reasonableness review of above-authorized spending. Again, under this new proposal, SCE could spend twice the amount authorized and there would still be no reasonableness review of the above-authorized spending.¹⁴⁰² The Commission should also reject SCE's alternative proposal to increase the threshold requiring a reasonableness showing from 115% to 125%.¹⁴⁰³ The utility has not demonstrated that such a change would do anything other

¹³⁹⁹ Ex. TURN-15-E2, p. 9, *citing* SCE's 2021 GRC testimony, Ex. SCE-18, Vol. 1, p. 17.

¹⁴⁰⁰ D.21-08-036, p. 186.

¹⁴⁰¹ Ex. SCE-07, Vol. 1, pp. 32-33.

¹⁴⁰² Ex. TURN-15-E2, p. 10, *citing* Attachment 3 (SCE Response to TURN DR 48-4.a.).

¹⁴⁰³ Ex. SCE-07, Vol. 1, pp. 32-33.

than increase the amount of potential costs that it could incur without ever having those costs subject to a meaningful determination of reasonableness.¹⁴⁰⁴ Instead, the Commission should modify SCE's VMBA so that it is a one-way balancing account, consistent with the approach adopted for PG&E's vegetation management mitigation programs in the test year 2023 GRC decision.¹⁴⁰⁵

SCE's rebuttal testimony does little beyond confirming how extreme and poorly supported its VMBA ratemaking proposal is. For example, SCE claims that a two-way balancing account is necessary because there is inherent variablility associated with vegetation management work, variability that is "in part due to exogenous factors."¹⁴⁰⁶ Even for costs due to truly exogenous factors, the utility will have some ability to control costs; a reasonableness review serves as an opportunity to ensure that the utility did, in fact, take appropriate cost minimization steps. And the obvious inference from SCE's description is that the purported variability is in part due to factors that are <u>not</u> exogenous, but rather within the utility's control or influence. Again, all the more reason why an effective reasonableness review showing must be required prior to recovery of any above-authorized costs. Furthermore, as the Commission recognized in the case of PG&E, the ratemaking treatment for vegetation management expenses should reflect the fact that the utility is "now well-experienced at an increased level of vegetation management" as compared to the level of activities and spending prior to the focus on wildfire mitigation work.¹⁴⁰⁷

¹⁴⁰⁴ Ex. TURN-15-E2, pp. 10-11.

¹⁴⁰⁵ D.23-11-069, pp. 487-488.

¹⁴⁰⁶ Ex. SCE-18, Vol. 1, p. 20.

¹⁴⁰⁷ D.23-11-069, p. 487 and Finding of Fact 233.

38.3.1.3.3 Risk Management Balancing Account

In D.23-05-013, the Commission adopted an agreement to implement a self-insurance alternative for SCE's wildfire liability insurance coverage. The settlement included a provision committing SCE to modify the Risk Management Balancing Account (RMBA) as necessary to shift to reliance on self-insurance and otherwise support the settlement. SCE has subsequently made such modifications to the RMBA. Here, SCE proposes to continue the RMBA as modified pursuant to D.23-05-013.¹⁴⁰⁸ TURN agrees with and supports this proposal. TURN also anticipates SCE will, in consultation with the other settling parties, present the Commission with proposed further modifications of the RMBA as needed to implement the final decision granting the motion to extend the self-insurance framework through the 2025 GRC period.¹⁴⁰⁹

38.3.1.3.4 Electric Vehicle Infrastructure Memorandum Account

SCE established a new Electric Vehicle Infrastructure Memorandum Account to record certain electric vehicle infrastructure related costs during the period from 2021 through 2024. Pursuant to Section 740.19(c) of the Public Utilities Code, costs incurred between January 1, 2021 and the utility's next GRC's implementation date are to be tracked in a memorandum account and "recovered, subject to a reasonableness review, in the decision adopting the next general rate case revenue requirement" for the utility. SCE contends that a "timing issue" means it will not be able to obtain a reasonableness review through this TY 2025 GRC for the amounts

¹⁴⁰⁸ Ex. SCE-07, Vol. 1, p. 34.

¹⁴⁰⁹ As described in Section 29.1, above, the Commission recently issued a proposed decision that would grant the unopposed motion.

it expects to spend in 2023 and 2024. Therefore, SCE proposes that the reasonableness review take place in the form of a Tier 3 advice letter, rather than in an application.¹⁴¹⁰

The Commission should reject SCE's proposal to have the reasonableness review of these costs in a Tier 3 advice letter, and instead continue to require that the review take place in an application proceeding. The statutory language is very specific; a reasonableness review is required, and it is expected to take place in the utility's next GRC. The Commission might have discretion to direct that the reasonableness review take place in a different application-based proceeding consistent with performing a reasonableness review. But the plain language of the statute does not contemplate the review taking place through the advice letter process. And it certainly would not permit the review to take place in an ERRA proceeding which, as the Commission told SCE in its 2021 GRC, is <u>not</u> a forum that constitutes a reasonableness review.¹⁴¹¹

Even if the Commission had discretion under the statute to substitute a Tier 3 advice letter for the required review in a GRC, it should reject SCE's proposal in recognition of the limitations of the advice letter process given the more compressed schedule, the lack of testimony, and no opportunity for hearings on disputed factual issues.¹⁴¹² TURN's testimony had explained that SCE's preliminary cost figures suggested a total cost figure of approximately

¹⁴¹² Ex. TURN-15-E2, p. 12.

¹⁴¹⁰ Ex. SCE-07, Vol. 1, p. 54.

¹⁴¹¹ SCE's rebuttal testimony suggests that if an advice letter is not an acceptable substitute, perhaps the ERRA application would be. Ex. SCE-18, Vol. 1, p. 27. In D.21-08-036, the Commission stated, "An annual compliance review ... in the ERRA proceeding, as proposed by SCE, would not entail a reasonableness review that considers such information [regarding the specific causes of above-authorized recorded amounts]." D.21-08-036, p. 404.

\$45 million.¹⁴¹³ SCE's rebuttal testimony argued that if one only considered the projects for which it has cost estimates at this time (77 of the 255 EVIMA-eligible projects), the requested review would likely involve \$15.8 million.¹⁴¹⁴ But the cited data request response does not reflect any any costs for the remaining EVIMA-eligible projects, all of which are "in flight" according to the utility. It merely says SCE only has cost estimates for approximately one third of them at this time.¹⁴¹⁵ Multiplying SCE's partial figure by three gets the Commission to TURN's estimated figure.

38.3.1.3.5 Z-Factor Memorandum Account

The Commission should reject SCE's proposal to extend its Z-Factor mechanism to the 2025 test year, and instead limit its availability to the PTYR period for this GRC cycle, as has been the case since the Z-Factor mechanism was first addressed in an SCE GRC application.¹⁴¹⁶ While SCE is correct that the Commission adopted a similar extension of the Sempra utilities' Z-Factor mechanism in their test year 2019 GRC, in doing so the agency specifically noted that it had not been presented with "any rationale" that might support limiting the Z-Factor mechanism to attrition years.¹⁴¹⁷ In the recent TY 2023 GRC decision for PG&E, the Commission provided just such a rationale:

Because the purpose of a general rate case is to provide a fairly precise forecast of the test year, the Commission does not adopt PG&E's proposal to apply the Z-Factor mechanism to the test year, 2023.¹⁴¹⁸

¹⁴¹³ Id., and Attachment 4 (SCE Response to TURN DR 101-1).

¹⁴¹⁴ Ex. SCE-18, Vol. 1, p. 27.

¹⁴¹⁵ Id., Appendix A, p. A-2 (SCE Response to TURN DR 101, Q. 1).

¹⁴¹⁶ Ex. TURN-15-E2, p. 13.

¹⁴¹⁷ D.19-09-051, p. 712.

¹⁴¹⁸ D.23-11-069, p. 717.

This fundamental purpose of a GRC is the same for SCE as it is for PG&E, and warrants adoption of the same ratemaking approach as the Commission adopted in D.23-11-069.

The Commission should also modify SCE's Z-Factor mechanism to require an application, rather than an advice letter, to seek recovery of costs tracked in its Z-Factor Memorandum Account. In the TY 2023 GRC decision, the Commission rejected PG&E's proposal to implement Z-Factor-related revenue requirement changes via advice letter, rather than application, finding "advice letters address ministerial matters and ... application of the Z-Factor mechanism is not simply ministerial...."¹⁴¹⁹ Similar logic should be applied to SCE going forward.

38.3.1.4 Proposed New Balancing Account – General Liability Insurance Balancing Account (GLIBA)

TURN's prepared testimony recommended rejection of SCE's proposal for a new General Liability Insurance Balancing Account (GLIBA).¹⁴²⁰ As discussed in Section 29.2, above, TURN has joined SCE and Cal Advocates in presenting a stipulation that, if adopted, would resolve the disputed forecast and ratemaking issues associated with the utility's general liability and property insurance. One element of the proposed stipulation would permit SCE to establish a new balancing account covering both general liability and property insurance forecasts and expenses, to be designated the General Liability & Property Insurance Balancing Account (GL&PBA).¹⁴²¹ Again, though TURN has called on the Commission to reduce its

¹⁴¹⁹ *Id*.

¹⁴²⁰ Ex. TURN-15-E2, pp. 14-16.

¹⁴²¹ Ex. SCE-34 (Stipulation of TURN, Cal Advocates and SCE on Non-Wildfire Insurance).

reliance on balancing and memorandum accounts as a general matter, TURN believes the new account is acceptable under the circumstances here and as part of the several compromises reflected in the stipulation.

38.3.1.5 New Memorandum Account Proposals

38.3.1.5.1 Next Gen ERP SAP Memorandum Account

SCE proposes to create a new NextGen ERP SAP Memorandum Account (NGESMA), with a January 1, 2024 effective date, to record the revenue requirements associated with O&M expenses and capital expenditures for activities related to the implementation phase of the NextGen ERP project.¹⁴²² The Commission should deny this request for a new memorandum account, based on the record evidence that any costs recorded before the end of 2024 should be deemed costs covered by the 2021 GRC revenue requirement, and due to SCE's failure to establish that the costs at issue are sufficiently substantial to warrant memorandum account treatment.

SCE's proposed NextGen ERP project is discussed in the utility's capitalized software testimony in this GRC.¹⁴²³ In SCE's test year 2021 GRC, the utility presented a "portfolio-based forecast" for its 2021-2023 capitalized software spending, without specified capital projects.¹⁴²⁴ Approximately 55% of the forecast was allocated to the utility's "Enterprise Support" group, covering a broad array of enterprise operations.

Future spend in these areas supports almost all business outcomes across SCE and drives operational and service excellence across the enterprise.

¹⁴²² Ex. SCE-07, Vol. 1, p. 45.

¹⁴²³ Ex. SCE-07, Vol. 1, p. 45, referring to Ex. SCE-06, Vol. 2.

¹⁴²⁴ Ex. TURN-104 (SCE testimony from test year 2021 GRC), p. 169.

Some of the key solutions being planned are improvements to our core, foundational SAP systems \dots^{1425}

SCE's core business data and functions all run or flow through SAP.¹⁴²⁶ In its test year 2021 GRC testimony, SCE stated "the current [SAP Enterprise Resource Planning] system ... is due to become obsolete in 2025."¹⁴²⁷ And in the testimony for the instant GRC, SCE states that it began planning in 2020 for the transformation from the current system to the "NextGen ERP" system.¹⁴²⁸ Given these circumstances, the Commission should not permit SCE a new memorandum account, but instead should deem the costs through the end of 2024 as subject to the currently authorized GRC revenue requirement. With its portfolio-based approach for 2021-2023 (and, 2024, consistent with the extension of the 2021 GRC cycle), SCE can identify few if any specific projects that were contemplated for 2024. And given the critical role SCE describes its SAP ERP system playing in key areas of its operations, and the utility's earlier recognition of the upcoming need for the work, it is reasonable to treat it as covered by the existing GRC revenue requirement.

The Commission should also deny SCE's request for a NGESMA because the capital costs associated with the project are highly unlikely to close to plant separately from the overall project costs, and the overall project costs will be addressed in an upcoming application. Thus,

¹⁴²⁵ Ex. TURN-104, pp. 177-178.

¹⁴²⁶ SCE Motion Seeking to Establish Three Memorandum Accounts (May 15, 2023), p. 5.

¹⁴²⁷ Ex. SCE-06, Vol. 1, Pt 2A from the test year 2021 GRC, p. 71. An excerpt of this testimony was included as an attachment to TURN's response to SCE's motion seeking new memorandum accounts (May 30, 2023).

¹⁴²⁸ SCE-06, Vol. 2 (Enterprise Technology – OU Capitalized Software), p. 73.

the new memorandum account seems to be addressing a non-existent cost recovery timing risk for capital costs.¹⁴²⁹

Prior to TURN's direct testimony, SCE had only referred to capital expenditures associated with the implementation phase of the NextGen ERP project. Therefore, TURN's testimony only addressed the expected capital expenditures.¹⁴³⁰ SCE's rebuttal testimony claims the utility has subsequently become aware of "a potential likelihood" of \$2-4 million it "could incur" in Implementation O&M costs in 2024.¹⁴³¹ TURN submits this late-breaking development should not alter the outcome, and the memorandum account request should still be denied. SCE has not demonstrated that this level of O&M funding is significant or substantial enough to warrant establishment of a new memorandum account.

If the Commission chooses to authorize this new memorandum account as of January 1, 2024, it should be subject to a deductible of \$10 million, consistent with TURN's earlierdescribed proposal to routinely apply a deductible in the amount of the Z factor deductible in effect at the relevant time, here 2024.

38.3.1.5.2 Advanced Metering Infrastructure 2.0 Memorandum Account

The Commission should deny SCE's request to establish a new Advanced Metering Infrastructure 2.0 Memorandum Account (AMIMA), and instead treat any O&M expenses associated with the pre-deployment base-level planning costs for a new AMI project as subsumed within the then-authorized 2021 GRC revenue requirement.

¹⁴²⁹ Ex. TURN-15-E2, pp. 16-17.

¹⁴³⁰ *Id.*, p. 17.

¹⁴³¹ Ex. SCE-17, Vol. 1, p. 89.

SCE is in the process of developing a new Advanced Metering Infrastructure (AMI) 2.0 project for providing metering services to its customers, and plans to file a stand-alone application "in the 2025 time frame" with a funding request for full AMI deployment.¹⁴³² The utility has included "base-level planning costs" from 2023 through 2027 in its GRC testimony. If the new AMIMA is approved, the utility would use it to record the "pre-Test Year O&M expenses associated with the planning costs."¹⁴³³ SCE anticipates incurring \$4.432 million of such O&M expenses in 2023, and \$0.585 million in 2024.¹⁴³⁴

TURN submit that there are several reasons for denying this memorandum account request. First, the "base-level planning costs" associated with ensuring SCE's customers receive adequate metering services should be deemed subsumed within the authorized 2021 GRC revenue requirement. Providing metering services to its customers is a fundamental part of SCE's operations. Costs incurred during the 2021 GRC period and associated with the replacement of existing meters, whether on a piecemeal basis or as the initial planning costs toward a wholesale changeout of equipment, should be treated as 2021 GRC-covered costs, even where the utility ostensibly chose not to include a forecast in the earlier GRC showing.¹⁴³⁵ In its motion seeking establishment of the memorandum accounts, SCE asserted that it did not include this subset of O&M costs in its 2021 GRC "given uncertainty in the magnitude, scope, and

¹⁴³² Ex. SCE-07, Vol 1, p. 46. The associated "base-level" capital expenditures in those years are included for review and recovery in SCE's GRC request, and would not be included in the AMIMA.
¹⁴³³ SCE Motion Seeking to Establish Three Memorandum Accounts (May 15, 2023), p. 3.
¹⁴³⁴ Ex. SCE-07, Vol 1, p. 46. The associated "base-level" capital expenditures in those years are included for review and recovery in SCE's GRC request, and would not be included in the AMIMA.
¹⁴³⁵ Ex. TURN-15-E2, pp. 18-19.

timing for these costs at the time that application was filed."¹⁴³⁶ The Commission considered and rejected a similar explanation from SoCalGas when it denied that utility's request for a new memorandum account under similar circumstances.¹⁴³⁷ SoCalGas had argued that the preconstruction costs associated with a major pipeline replacement project "were not ripe for inclusion" in its prior GRC.¹⁴³⁸ But the Commission relied more on SoCalGas's failure to present any argument or evidence that "it was prohibited, precluded, or otherwise incapable of including those costs in its 2016 GRC."¹⁴³⁹ Similarly, SCE has not presented any such evidence here.

Second, the record of the test year 2021 GRC proceeding establishes the appropriateness of treating the AMI 2.0 initial planning costs as covered by the authorized GRC revenue requirement through 2024. The GRC decision quotes SCE's testimony in describing the work done under the heading "Meter Activities," including "guards against the issues caused by technology obsolescence."¹⁴⁴⁰ In that testimony, SCE also described its "need to maintain its SmartConnect system by addressing the challenges created by the obsolescence of our existing metering system."¹⁴⁴¹ In the 2025 GRC, SCE initially described its proposed AMI 2.0 deployment as needed in order to address approaching technology obsolescence, ¹⁴⁴² but seemed to attempt to reverse course in its rebuttal testimony, asserting that the costs at issue "have

¹⁴³⁶ SCE Motion for a Ruling Authorizing Establishment of Three Memorandum Accounts to Track Certain Costs Beginning in 2023 and 2024 (May 15, 2023), p. 9.

¹⁴³⁷ D.18-04-012, p. 10.

¹⁴³⁸ *Id.*, p. 11.

¹⁴³⁹ Id.

¹⁴⁴⁰ D.21-08-036, p. 62, citing TY 2021 GRC Ex. SCE-02, Vol. 1, Pt. 3, p. 4.

 ¹⁴⁴¹ Ex. TURN-15-E2, pp. 18-19, citing Attachment 7 (TY 2021 GRC Ex. SCE-02, Vol. 1, Pt. 3, p. 26).
 ¹⁴⁴² Ex. SCE-02, Vol. 3, pp. 31-32.

nothing to do with ... addressing obsolescence issues associated with the existing meter fleet."¹⁴⁴³ TURN submits that SCE had it right the first time, and therefore the Commission should reasonably treat the AMI 2.0 initial planning costs for 2023 and 2024 as subsumed in the then-effective GRC revenue requirement.

Third, SCE has failed to establish that the approximately \$5 million of O&M expense is a substantial enough amount under the circumstances to warrant establishment of a new memorandum account. SCE forecasted \$4.432 million and \$0.585 million as the relevant O&M expenses in 2023 and 2024, respectively.¹⁴⁴⁴ This two-year total of approximately \$5 million compares to SCE's forecasted two-year total <u>net</u> operating revenue of approximately \$5.5 billion.¹⁴⁴⁵ The utility has failed to establish that \$5 million is an amount that warrants establishment of a new memorandum account.¹⁴⁴⁶

If the Commission chooses to authorize this new memorandum account, it should be subject to a deductible of \$10 million, consistent with TURN's earlier-described proposal to routinely apply a deductible in the amount of the Z factor deductible in effect at the relevant time, here in 2023 and 2024.

¹⁴⁴³ Ex. SCE-18, Vol. 1, p. 33.

¹⁴⁴⁴ Ex. SCE-02, Vol. 3, pp. 38 and 40.

¹⁴⁴⁵ Ex. SCE-18, Vol. 1, Appendix B, p. B9.

¹⁴⁴⁶ While SCE's rebuttal testimony argues that TURN presented no evidence that \$5 million is <u>not</u> a substantial amount under the circumstances (*Id.*, p. 34), the burden of proof is on the utility to establish that it <u>is.</u>

38.3.1.5.3 Cybersecurity Compliance Memorandum Account

The Commission should deny SCE's request for a new Cybersecurity Compliance Memorandum Account (CCMA). As originally proposed, the CCMA was explained as being necessary given the "variety of emerging mandatory cybersecurity standards that are at various stages of development and that would require additional investment to comply with."¹⁴⁴⁷ SCE originally sought to establish the CCMA with an effective date of May 12, 2023, but when no incremental costs of new cybersecurity regulations were recorded in 2023 and no "significant" amounts were expected for 2024, SCE modified its proposal and now seeks a January 1, 2025 effective date for the CCMA.¹⁴⁴⁸

TURN continues to oppose establishment of the new CCMA, even with a January 1, 2025 effective date. TURN understands SCE's contention that it has not included in its test year 2025 forecast costs associated with compliance with the "CMMC 2.0," and that the utility expects to incur "potentially significant" implementation costs at some point during the 2025 GRC cycle.¹⁴⁴⁹ Nonetheless, denial of the memorandum account request is necessary to achieve fundamental fairness. In SCE's 2021 GRC, the Commission authorized Cybersecurity Delivery O&M expenses of \$23.6 million for 2021; the utility recorded \$18.8 million that year, a difference of \$4.8 million.¹⁴⁵⁰ The recorded figure for 2022 was also below the GRC-authorized

¹⁴⁴⁷ Ex. SCE-07, Vol. 1, p. 47, largely repeating the assertion at Ex. SCE-04, Vol. 3, p. 18, and at *SCE Motion for a Ruling Authorizing SCE to Establish Three Memorandum Accounts to Track Certain Costs Beginning in 2023 and 2024* (May 15, 2023), p. 4.

¹⁴⁴⁸ Ex. SCE-18, Vol. 1, p. 39-40.

¹⁴⁴⁹ Ex. SCE-18, Vol. 1, p. 41.

¹⁴⁵⁰ Ex. SCE-4, Vol. 3, p. 34, Figure II-7.

amount.¹⁴⁵¹ And the Commission authorized capital expenditures of \$63.8 million for 2021; the utility recorded \$53.7 million, a \$10.1 million difference.¹⁴⁵² SCE should not be permitted to benefit from the below-authorized cybersecurity spending it recorded in 2021 and 2022, but then obtain memorandum account protection against the prospect that it might record above-authorized costs during the 2025 GRC cycle. The Commission should deny the requested CCMA.

If the Commission chooses to authorize this new memorandum account, it should be subject to a deductible of \$10 million, consistent with TURN's earlier-described proposal to routinely apply a deductible in the amount of the Z factor deductible in effect at the relevant time.

38.3.1.5.4 Historic Sporting Events Cost Tracking Memorandum Account

SCE seeks to establish a new Historic Sporting Events Cost Tracking Memorandum Account (HSECTMA) in which it would record costs and services "impacted" by the 2028 Summer Olympics and the 2026 World Cup. The utility suggests the recent Super Bowl serves as a rough indication of the order of magnitude of costs it can expect to incur due to these upcoming events.¹⁴⁵³ However, the 2022 Super Bowl costs were so insignificant that SCE did not bother tracking them in the absence of a cost recovery mechanism.¹⁴⁵⁴ TURN submits that if

¹⁴⁵¹ 2022 recorded spending was approximately \$19.9 million. Ex. SCE-04, Vol. 3 WP, p. 20.

¹⁴⁵² Ex. SCE-4, Vol. 3, p. 35, Figure II-8.

¹⁴⁵³ Ex. SCE-07, Vol. 1, p. 47. ["The anticipated (but yet unknown) order of magnitude is in the same range as the Super Bowl."]

¹⁴⁵⁴ When TURN asked for the amount of incremental costs SCE recorded due to the 2022 Super Bowl, the utility said that though it incurred costs, it could provide no quantification because there was no cost recovery mechanism in place. Ex., p. 22, citing Attachment 10 (SCE Response to TURN DR 61-2.a.i).
the utility expects the costs of the upcoming events to be at a similar level as the 2022 Super Bowl, the Commission should conclude the costs are unlikely to be substantial enough to warrant creation of yet another memorandum account.

SCE also suggests that having its customers bear all incremental costs associated with addressing the energy needs of these sporting events would "align with ... the expectation that all supporting institutions will be able to deliver services necessary to support these events even if these services are beyond what is typically provided."¹⁴⁵⁵ The source and the basis for this "expectation" are not clear in SCE's testimony. If SCE, as a supporting institution, wishes to subsidize the 2028 Summer Olympics or the 2026 World Cup, TURN is not aware of any restriction on its ability to use shareholder funds for that purpose. However, SCE's customers should not be treated as a "supporting institution" for such purposes, and the associated costs should not be recovered in rates.¹⁴⁵⁶

TURN recommends that the Commission deny SCE's request for the HSECTMA at this time. If the Commission chooses to authorize this new memorandum account, it should be subject to a deductible of \$10 million, consistent with TURN's earlier-described proposal to routinely apply a deductible in the amount of the Z factor deductible in effect at the relevant time.

¹⁴⁵⁵ Ex. SCE-07, Vol. 1, p. 47.

¹⁴⁵⁶ Ex. TURN-15-E2, p. 22.

38.3.1.6 Request for Recovery of Existing Memorandum Account Balances

38.3.1.6.1 Seismic Retrofit for Non-Electric Facilities Memorandum Account (SRNEFMA)

SCE's direct testimony explains that in its 2021 GRC, the Commission authorized the Seismic Retrofit for Non-Electric Facilities Memorandum Account (SRNEFMA) to track recorded costs in excess of the authorized amounts in that decision, with an opportunity to recover the recorded costs in its next GRC. SCE originally sought recovery here of the incremental costs recorded in the account, but merely provided a list of the recorded amounts, without any showing that might support a determination of reasonableness.¹⁴⁵⁷ TURN opposed such recovery based on the inadequacy of the utility's demonstration of reasonableness.¹⁴⁵⁸ In its rebuttal testimony, SCE withdrew its request because it no longer anticipated it would record costs of any amount to the account before the end of 2024.¹⁴⁵⁹

38.3.1.6.2 Service Center Modernization Projects Memorandum Account (SCMPMA)

The Commission should determine that SCE has failed to adequately establish the reasonableness of the amounts recorded in the Service Center Modernization Projects Memorandum Account (SCPMA) for projects in service before the end of 2024, and therefore deny the utility's request for rate recovery of any portion of the associated costs.

¹⁴⁵⁷ Ex. SCE-04, Vol. 1, pp. 49-50 and Table II-9.

¹⁴⁵⁸ Ex. TURN-15-E2, p. 23.

¹⁴⁵⁹ Ex. SCE-18, Vol. 1 p. 43.

In SCE's 2018 GRC the Commission agreed with a range of TURN criticisms regarding the evolution of and support for the utility's cost forecasts for service center modernization projects.¹⁴⁶⁰ The decision directed SCE to complete the projects identified in that GRC's testimony with the hope that such completion would be "within its forecasted budgets."¹⁴⁶¹ In addition, the Commission required establishment of a new memorandum account for recording the costs of six specified projects, with a future determination of "whether the expenditures recorded from January 1, 2018 … onward should be recovered in rates."¹⁴⁶² The Service Center Modernization Projects Memorandum Account (SCMPMA) was initially created to record the costs of the six projects, and in the 2021 GRC decision was expanded to also include costs for the Santa Barbara Service Center project.¹⁴⁶³

SCE now expects most but not all of the identified projects to be completed and in service by the end of 2024. The original estimate of \$24.291 million of revenue requirement through the end of 2024 was updated to \$46.622 million in SCE's update testimony. SCE seeks authority to recover that amount upon issuance of a final decision here.¹⁴⁶⁴

The Commission should deny SCE's request. SCE has failed to present the type of showing that might establish the reasonableness of the capital expenditures underlying the revenue requirement figures it now seeks to recover in rates. The utility's testimony in support of recovery of the SCMPMA amounts cross-referenced discussion of these projects in its

¹⁴⁶⁰ D.19-05-020 (SCE TY 2018 GRC), pp. 200-208.

¹⁴⁶¹ *Id.*, p. 202.

¹⁴⁶² *Id.*, p. 203.

¹⁴⁶³ Ex. SCE-07, Vol. 1, pp. 60-61.

¹⁴⁶⁴ Ex. SCE-40, p. 16.

Enterprise Operations testimony.¹⁴⁶⁵ When TURN's direct testimony challenged the sufficiency of that showing, SCE's rebuttal assured the Commission that if it looked at the Enterprise Operations testimony it would find "detailed explanations of the reasons for the variances" between the actual incurred costs and the forecasts originally presented in the 2018 GRC.¹⁴⁶⁶

The information in the Enterprise Operations testimony does not constitute an adequate demonstration of the reasonableness of the recorded amounts for each project. Rather than containing any "detailed explanations" of the cost variances, the material was more of an annotated invoice, listing for each project broad areas of costs and abbreviated statements of influencing factors.¹⁴⁶⁷ For example, consider the testimony's discussion of the Bishop Service Center Modernization project. SCE provides a bullet point level description of the scope of work completed as of the end of 2022 for the project, and of changes to the original scope of the project.¹⁴⁶⁸ It then presents a graph showing \$26.174 million as the 2018 request for the project, and indicating changes in four broad categories that led to the amount recorded through 2022, as well as an additional increment "forecast through completion."¹⁴⁶⁹ This is followed by six bullet points identifying issues that resulted in unanticipated cost increases.¹⁴⁷⁰ The testimony included a similar discussion of the other projects with costs recorded in the SCMPMA.¹⁴⁷¹ Neither the workpapers supporting the Enterprise Operations testimony nor the material in SCE's 2021 GRC

¹⁴⁶⁵ Ex. SCE-07, Vol. 1, p. 62.

¹⁴⁶⁶ Ex. SCE-18, Vol. 1, p. 48.

¹⁴⁶⁷ Ex. SCE-06, Vol. 7, pp. 16-36.

¹⁴⁶⁸ *Id*, pp. 17-18; Neal, SCE, 5 RT 564, l. 17 to 565, l. 7.

¹⁴⁶⁹ *Id.*, p. 19, Figure II-5; Neal, SCE, 5 RT 565, l. 8 to 567, l. 3. The \$26.174 million figure listed as the 2018 request for this project seems incorrect, as the figure from D.19-05-020 is \$20.054 million (p. 210). ¹⁴⁷⁰ *Id.*, pp. 19-20; Neal, SCE, 5 RT 567, ll. 10-13.

¹⁴⁷¹ Neal, SCE, 5 RT 567, ll. 14-19.

discussing costs recorded in the SCMPMA provided any further information that might support a determination of cost reasonableness.¹⁴⁷² There is nothing that might indicate SCE management awareness of the growing costs for these projects, or any actions taken to try to control those costs. Based on the inadequacy of the utility's reasonableness showing in support of the costs recorded in the SCMPMA, the Commission must deny rate recovery.

38.3.1.7 Wildfire Mitigation Plan Memorandum Account – The Commission Must Reject SCE's Attempt to Undo The 2021 GRC Denial of Rate Recovery for Fusing Mitigation Costs.

The Commission should deny SCE's request to recover approximately \$18.4 million as the December 31, 2024 revenue requirement related to the 2020 fusing mitigation capital costs.¹⁴⁷³ for which the Track 3 decision in the 2021 GRC denied rate recovery.

In SCE's 2021 GRC, the Commission denied SCE's request for rate recovery of certain capital costs associated with the utility's fusing mitigation program, based on its finding that SCE had failed to meet its burden of demonstrating the costs were reasonable and should be recovered from the utility's customers.¹⁴⁷⁴ In this GRC, SCE attempts to reverse the earlier decision's outcome. SCE apparently believes it is entitled to such a "second bite at the apple" because the Commission did not use the specific word "disallow" in describing its earlier denial of rate recovery, and because the costs had originally been recorded in a memorandum account.¹⁴⁷⁵

 ¹⁴⁷² Neal, SCE, 5 RT 568, l. 2 to 569, l. 18; Ex. TURN-101 (Excerpt of SCE TY 2021 GRC testimony).
¹⁴⁷³ Ex. SCE-40 (Update Testimony), p. 19.

¹⁴⁷⁴ D.22-06-032, p. 32.

¹⁴⁷⁵ Ex. SCE-04, Vol. 5, Pt. 2, pp. 114-115; Ex. SCE-07, Vol. 1, pp. 77-78.

The Commission must reject SCE's request for several reasons. First, it is inconsistent with the Track 3 decision, which gives no indication of the Commission intending to permit SCE a second opportunity to establish the reasonableness of the costs for which rate recovery was denied. SCE had sought a determination of reasonableness for \$24.6 million of incremental "fusing mitigation" capital expenditures.¹⁴⁷⁶ Instead, the Commission found "that SCE has failed to meet its burden of demonstrating" that the incremental capital costs associated with its fusing mitigation program were reasonable and should be recovered from ratepayers.¹⁴⁷⁷ Nothing in the decision suggests the Commission contemplated SCE having an opportunity to recover in this test year 2025 GRC any costs associated with the amount for which rate recovery was denied in the 2021 GRC.

Remarkably, SCE suggests reading the Track 3 decision as if it "did not disallow SCE's fusing mitigation program cost recovery amount, much less subject the capital costs to a permanent disallowance."¹⁴⁷⁸ While it is true that the Commission did not use the word "disallow" in its discussion of this cost recovery request, the language it <u>did</u> use leaves no doubt that it intended to deny rate recovery of at least some if not all of the costs. SCE here proposes an approach by which it would achieve rate recovery of all costs.

¹⁴⁷⁸ Ex. SCE-04, Vol. 5, Part 2, pp. 114-115.

¹⁴⁷⁶ The Track 3 decision describes the fusing mitigation work as covered by the GSRP settlement. *Id.*, p. 28. It is not clear from the decision whether the incremental fusing mitigation capital expenditures or the associated revenue requirement was originally recorded in the GSRPBA/MA, or in the WMPMA. SCE contends in the present GRC that it "is currently recording the capital-related revenue requirement associated with these incremental fusing mitigation capital costs in its [WMPMA]." Ex. SCE-07, Vol. 1, p. 77.

¹⁴⁷⁷ *Id.*, pp. 32-33. The Commission also noted that some of the costs for installation of the potentially defective fuses were already included in rates, and there was the possibility that SCE would receive some financial recovery from the fusing manufacturer. Therefore, the Track 3 decision called for a potential adjustment in this 2025 GRC to ensure that, "[t]o the extent ratepayers have funded some of these fuses, ratepayers should be credited their fair share of any recovery from the manufacturer or supplier." *Id.*

Which leads to the second reason the Commission should deny SCE's rate recovery request; it relies upon an apparent misuse of the memorandum account process. SCE's testimony acknowledges there would be a "retroactive ratemaking issue" but for the fact that the costs in question are currently recorded in a memorandum account.¹⁴⁷⁹ But the Commission has to ask why these costs were still recorded in a memorandum account after the Track 3 decision denied rate recovery. And it should be very concerned with the prospect that SCE is using the memorandum account ratemaking device to attempt to reverse previously adopted denials of rate recovery. Frankly, it never would have dawned on TURN that a rate recovery denial as clear as the one adopted in the Track 3 decision might have to be accompanied by a Commission directive to SCE to remove the associated costs from any memorandum or balancing account. Yet in the face of that rate recovery denial, SCE chose to simply leave the denied costs just where they were in its regulatory accounts, apparently convinced doing so would enable an opportunity to ask for them again and, if necessary, again.

TURN recommends that the Commission here make clear that the denial of cost recovery in the Track 3 decision is now a permanent disallowance of the fusing mitigation costs at issue. This response would appropriately signal SCE that it was inappropriate and unacceptable to attempt to use the memorandum account as a means of achieving the same level of rate recovery as would have occurred had the Track 3 decision found no deficiency in its showing. Alternatively, the Commission should deny rate recovery of the revenue requirement for an appropriate period given the circumstances. At minimum, this must include the revenue requirement through the end of the 2021 GRC period, which SCE estimates to be \$18.4 million

¹⁴⁷⁹ *Id.*, p. 115.

as of December 31, 2024.¹⁴⁸⁰ TURN recommends that the Commission also deny rate recovery of the revenue requirement for the 2025 GRC period, again as an appropriate signal to SCE that its approach here was inappropriate and unacceptable.

38.3.2 Other GRC Ratemaking Proposals

38.3.2.1 SCE's Proposed "True-up" of 2023 Recorded Capital Expenditures

The Commission must reject SCE's proposal to treat recorded 2023 capital expenditures as recoverable even where opposed, in favor of continuation of the established practice of relying on the recorded figure where there is no opposition to doing so.

The development of SCE's test year 2025 revenue requirement requires a determination of the reasonable amount of 2023 capital expenditures to include in the calculations.¹⁴⁸¹ In this GRC, 2023 is the "Base year +1," with costs initially presented as forecasts in the utility's direct testimony, but for which recorded amounts later become available and, as is typical, are included in the evidentiary record. For capital expenditures in 2024 and the 2025 test year, the Commission is resolving disputes that involve only forecasted figures. But for 2023 capital expenditures, the evidentiary record includes not only competing forecasts, but also SCE's recorded figures. However, because the recorded cost data were not provided until after intervenors had already served their prepared testimony, parties other than SCE had only very limited opportunity to review or conduct discovery on the 2023 recorded amounts.

¹⁴⁸⁰ Ex. SCE-40 (Update Testimony), p. 19.

¹⁴⁸¹ The amounts associated with 2023 capital expenditures to be collected in 2023 and 2024 are the product of the test year 2021 GRC and the post-test year attrition mechanism adopted in that proceeding.

SCE's rebuttal testimony asks the Commission to adopt a virtually automatic "true up" of its 2023 capital expenditures, which appears to entail replacing the forecasted 2023 figures from the utility's testimony and workpapers with the recorded 2023 capital expenditures figures that the utility shared with parties on March 11, 2024. SCE's position seems to largely be premised on the misguided notion that once the utility has recorded costs for 2023, the burden of proof shifts such that intervenors must demonstrate that the costs reflect imprudence on the utility's part. The Commission must reject SCE's request as overreaching, and its underlying logic as unfounded and unsupportable.

TURN does not suggest the Commission should never adopt the 2023 recorded figures for the capital expenditures of a specific program or project in this GRC. In many instances there may be no explicit or implicit dispute over the use of the recorded figures, particularly when SCE's actual 2023 spending was at a level below its forecast or, in some cases, even below the intervenor's lower forecast for that year. In the absence of opposition, the Commission may well see fit to substitute the 2023 recorded figures for SCE's original forecasts for that year. However, there may still be disputed 2023 spending amounts, particularly where SCE's 2023 recorded figures continue to be higher than an intervenor's forecast of the reasonable amount for that year. The Commission will need to resolve those continuing disputes in the same manner and subject to the same burdens of proof and productions as any other issue, with the 2023 recorded figure as another available data point, but not as a presumptively reasonable figure entitled to the "true-up" SCE proposes. That is, whether the 2023 capital expenditure figures are reviewed on a forecasted or a recorded basis, SCE still bears the same burden of establishing the reasonableness of the amount it seeks to establish as reasonable and therefore eligible for inclusion in its authorized revenue requirement.

SCE's proposed "true-up" seeks to impose a single approach to <u>all</u> recorded capital expenditure figures for 2023, even where an intervenor objects to applying that approach to a particular project or program, and in doing so apply a presumption of reasonableness simply due to the costs now being recorded rather than forecasted. According to the utility, "SCE's ongoing capital-related revenue requirement [associated with SCE's 2023 capital costs] should ultimately reflect the total, recorded capital expenditures that closed as capital additions to rate base."¹⁴⁸² In its view, any other outcome "would be inappropriate as it would constitute a disallowance of used and useful assets in service to customers."¹⁴⁸³ The utility describes its underlying view as "SCE is entitled to recover its investment on assets used and useful for customers, unless it can be shown that the spending was not reasonable."¹⁴⁸⁴

It is hard to know where to start in debunking SCE's claims. First, in the test year 2018 GRC, the Commission explicitly rejected SCE's very similar near-automatic "true-up" position when it was applied to amounts that had been forecasted in the previous GRC cycle.¹⁴⁸⁵ While SCE was permitted to seek to "true-up" its rate base during a GRC test year to reflect above-authorized spending from the previous GRC cycle, the Commission noted "it should not be presumed that the true up will be authorized following review by the Commission." And the review would not focus solely on the reasonableness with regard to the timing of the investment but could instead extend to whether the amount spent "is fair and reasonable to ratepayers." SCE fails to even mention this earlier outcome, much less explain why the position that was rejected

¹⁴⁸² Ex. SCE-18, Vol. 1, p. 115.

¹⁴⁸³ Ex. SCE-18, Vol. 1, p. 116.

¹⁴⁸⁴ *Id.*, p. 125.

¹⁴⁸⁵ D.19-05-020 (SCE test year 2018 GRC), pp. 342-343.

as applied to amounts forecasted in an earlier GRC should be adopted here when applied to amounts forecasted for the first year of this GRC period.

Second, the Commission has repeatedly rejected SCE's logic that the utility is entitled to recover costs of all "used and useful" plant when the utility took similar positions in previous GRCs.¹⁴⁸⁶ One of those rejections occurred in the context of SCE's rate base "true up" argument in the test year 2018 GRC.¹⁴⁸⁷ It is not at all clear how SCE saw fit to revive the argument here, again without any acknowledgment of these prior rejections.

Third, SCE's position is that once the utility has spent the money, the burden is on intervenors to establish imprudence and, absent such a showing, the Commission should permit the utility to include the recorded amount in rate base, even if it is substantially more than the utility forecasted for a given project or program.¹⁴⁸⁸ This position is legally and logically defective. The burden is on the utility to establish reasonableness and prudence of its recorded costs in circumstances such as these. "What critically matters is the prudency of the utility's actions, which the utility has the burden of proving, regardless of the testimonies of other parties."¹⁴⁸⁹ Here, SCE has provided no discussion of the prudency of its actions in 2023 with regard to the recorded amounts of capital expenditures; the utility's showing is, only slightly paraphrased, "we spent the money." For 2023 spending amounts that remain in dispute, a

¹⁴⁸⁶ D.15-11-021 (SCE test year 2015 GRC), p. 327 ["The fact that the new poles provide service to ratepayers and are used and useful is insufficient to prove that the expenditures to purchase and install the poles should be recovered form rates. That question turns on the prudency of the investment edecision."]. The Commission reiterated this approach in D.19-05-020 (SCE test year 2018 GRC), pp. 328-329. *See also* D.19-05-020, p. 332 ["SCE cannot establish reasonableness based simply on a claim that an expenditure was made and has resulted in an investment which is used and useful for SCE's customers."

¹⁴⁸⁷ D.19-05-020, pp. 342-343.

¹⁴⁸⁸ Ex. SCE-18, Vol. 1, pp. 116-117.

¹⁴⁸⁹ D.21-10-036 (Rehearing of PG&E test year 2019 GT&S rate case), p. 6.

"showing" that consists of nothing more than the recorded amount cannot meet the utility's burden of proving prudence with regard to that amount. SCE's approach also flies in the face of reality, as it would tacitly treat the 2023 recorded spending figures as if intervenors had a reasonable opportunity to review or address the prudence or reasonableness of figures that were first provided <u>after</u> their prepared testimony had been served.¹⁴⁹⁰ Similarly unsupported is the implication that having recorded cost figures should somehow resolve disputes over the 2023 cost forecasts. For example, TURN has recommended the Commission authorize 2023 spending for SCE's Overhead Conductor program at a level far below the amount SCE had originally forecasted, based partly on arguments that the scale of SCE's proposal and the failure to adequately consider less expensive alternatives rendered the forecast unreasonable. The fact that SCE now has a 2023 recorded spending figure for this program does not resolve any of those challenges, and nowhere does SCE explain why the Commission should treat the recorded figures as effectively eliminating such disputes.

Fourth, SCE relies on baseless claims that its approach is consistent with the modifications adopted to the Rate Case Plan in D.20-01-002. In that decision the Commission merely noted that having the Base Year +1 recorded cost data in the evidentiary record for a GRC is often helpful to the decision-making process.¹⁴⁹¹ There is nothing in the decision that would suggest the Commission intended to make the Base Year +1 recorded cost data determinative of anything, or to effectively shift the burden of proof among the parties.

¹⁴⁹⁰ Under the procedural schedule established for this proceeding, intervenor testimony was due on February 29, 2024. On March 11, 2024, SCE served its preliminary 2023 recorded capital expenditures figures. Thus, at all relevant times leading up to the presentation of their prepared testimony, in nearly all instances intervenors would not have had available to them even the preliminary recorded 2023 figures. ¹⁴⁹¹ D.21-01-002, pp. 61-62.

Finally, even if there were any legal or logical basis for the utility's position, the Commission should still decline to adopt it here due to the procedural defect of SCE choosing to wait until its rebuttal testimony to present its expanded proposal for a true-up. TURN found nothing in SCE's direct showing that even suggested the utility intended to include this proposal as part of its GRC request. Given that the proposal represents fundamental shifts in general GRC ratemaking practices in terms of the utility's burden of proving reasonableness and prudence, and would effectively reverse outcomes from prior GRCs when SCE's similar requests were denied, the Commission should recognize that waiting until rebuttal testimony to make the new request is an unfortunate and unacceptable tactic.

Again, TURN is not in any way suggesting that the 2023 recorded capital expenditure figures should serve no purpose to the Commission's decision-making process here. As the Commission has recognized in the past, "where a proposal or funding request has not been challenged by an intervenor, we generally adopt the utility's request as a practical reality of the decision-making process."¹⁴⁹² Here, for programs or projects where SCE's 2023 forecast was not challenged and no party raises a challenge to the 2023 recorded figure, the "practical reality" would likely be applicable. But in those instances where the 2023 spending level remains in dispute, the Commission must decide whether SCE's showing of reasonableness and prudence warrants authorizing recovery of any amount, whether it's the utility's 2023 forecast, the utility's recorded expenditures for 2023, the intervenor-sponsored forecast for 2023, or some other figure. For such a program or project, SCE's recorded figure for 2023 gives the Commission an additional data point, nothing more. It does <u>not</u> establish that the recorded spending amount is

¹⁴⁹² D.93-12-043 (SoCalGas Test Year 1994 GRC); 1993 Cal. PUC LEXIS 728, *12; 52 CPUC 2d 471.

reasonable, and it does <u>not</u> shift the burden to intervenors to disprove prudence, reasonableness, or any other element necessary for authorization of rate recovery.

38.4 Forecasts Of Sales, Customers, And New Meter Connections

In Section 12.1 above, TURN addresses most issues covered by SCE's direct and rebuttal testimony addressing the forecast of residential customers and new meter connections for residential, commercial, and agricultural customers (Ex. SCE-07V01 and Ex. SCE-18V01). For instance, TURN addresses the common shortcomings in SCE's forecasts of both residential new meters and residential customers, including issues with SCE's regression analyses linking housing starts to customers and new meters, as well as the reliability of SCE's housing start forecast, which is the primary explanatory variable in its forecast models. TURN also addresses many of SCE's criticisms of TURN's forecast methodology for residential customers and new meter connections, which entails using a 10-year average growth rate for residential customers to forecast future customers and a 10-year average growth rate for new residential meter connections to forecast new meter connections. TURN further addresses modifications to some of its procedural recommendations presented in testimony in consideration of SCE's rebuttal testimony. Finally, TURN explains that it has accepted SCE's recommendation that TURN update its 10-year historical average growth rates used to forecast both residential customers and new meter connections to include 2023 recorded data.

TURN's discussion here is limited to presenting TURN's updated residential customer forecast recommendation following TURN's incorporation of 2023 recorded data. As SCE recommended, TURN is willing to update its 10-year historical average growth rate for residential new customers to incorporate the 2023 recorded residential customer count

431

(4,578,185) provided by SCE in rebuttal testimony.¹⁴⁹³ Incorporating 2023 recorded data would impact the average growth rate by moving the series from 2013-2022 to 2014-2023 and impact the starting point for escalating customer count by the new growth rate.

However, TURN does not believe that SCE's recorded residential customer count for all years that are necessary to update this calculation, 2014 through 2023, is in the record; it may only be in TURN's confidential workpapers which SCE has but have not been admitted into evidence.¹⁴⁹⁴ As a result, TURN cannot officially provide an updated calculation. TURN unofficially indicates that the resulting 2014-2023 growth rate is higher than the 2013-2022 growth rate of 0.59% and the 2025 customer forecast is higher than SCE's forecast of 4,626,593. Accordingly, TURN no longer opposes SCE's 2025 residential customer forecast. TURN continues to recommend that customer growth be determined in a manner that is transparent, practical, and reasonable, like TURN's 10-year average historical growth rate. As such, TURN would not oppose the adoption of the updated forecast resulting from TURN's forecast methodology, should SCE be interested in seeking to amend the record to include the necessary data.

¹⁴⁹³ Ex. SCE-18V01, p. 82.

¹⁴⁹⁴ See Ex. SCE-18V01, Appendix A, p. A43 (SCE-TURN-005, Q3)(referring to a confidential workpaper that was provided as part of TURN's response but not attached by SCE to its rebuttal testimony). SCE provides 2021-2023 recorded residential customers in Ex. SCE-07V01, p. 98 and Ex. SCE-18V01, p. 76. Cal Advocates provides 2017-2022 recorded residential customers in Ex. CA-27, p. 2.

38.5 Present Rate Revenue

38.6 Cost Escalation

38.7 Other Operating Revenues (Excluding Non-Tariffed Products And Services)

38.8 Other Operating Revenues – Non-Tariffed Products And Services

In 1999, the Commission adopted a revenue sharing mechanism for non-tariffed products & services ("NTP&S") that awarded SCE shareholders 70% of the revenue or 90% if shareholders incur more than \$225,000 of expense.¹⁴⁹⁵ Since that time, SCE shareholders have been rewarded with more than \$1.342 billion of revenues while ratepayers received \$661.1

million,¹⁴⁹⁶ even though ratepayers paid for the assets *and* paid shareholders for the returns on the rate base.

38.8.1 SCE's Lack of Auditable Records Along with the Inherent Conflict of Interest in the Determination of Incremental Costs Should Result in the Finding that SCE Has Not Met Its Burden of Proof

SCE states that it uses the "but for" test to determine whether cost is incremental and thus charged to the shareholders.¹⁴⁹⁷ However, as TURN noted previously, 1) SCE alone conducts the "but for" test that determines which costs are incremental and should therefore be charged to shareholders, 2) SCE does not have a record of the "but for" tests, which renders an audit of these tests impossible, and 3) SCE does not keep a record or time log of its NTP&S Program's use of utility resources.

SCE claims that its NTP&S Program has been subject to several audits, including one by Baker Tilly in 2015 and audits by the State Controller's Office, and that there were no findings

¹⁴⁹⁵ D.99-09-070, Attachment A, pp. 3-4.

¹⁴⁹⁶ Ex. TURN-10, p. 12, citing DR TURN-SCE-90, Question 7.

¹⁴⁹⁷ Ex. SCE07V01, p. 124.

related to Rule VII.¹⁴⁹⁸ It is unclear what was the scope of the Baker Tilly audit in 2015 because SCE claims that it is unable to locate a copy of the audit "after a reasonable search" since it is no longer subject to document retention obligations,¹⁴⁹⁹ which TURN does not find credible. However, it is clear that the State Controller's Office was never able to audit whether incremental costs were properly captured and assigned to shareholders – it was only able to *trace the incremental costs and gross revenue to SAP ERP general ledger details to ensure accuracy of reporting*.¹⁵⁰⁰ In other words, due to the lack of auditable records, the State Controller's Office was not able to audit whether incremental costs were properly assigned to shareholders, since SCE alone performs the determination and there are no records that can be audited in terms of which costs SCE deemed to be incremental versus which costs it deemed not to be incremental.

To explore this further, TURN asked SCE to provide a sample of documents it provided to the auditors "to examine whether all incremental costs for NTP&S were properly assigned to shareholders."¹⁵⁰¹ SCE responded that it provided auditors with SCE's procedure for preparing its annual report, NTP&S training script, guidelines and procedure for recording incremental costs, list of accounting for SCE's reported incremental costs, and NTP&S training presentation.¹⁵⁰² Clearly, none of these documents would allow an auditor to determine whether incremental costs for NTP&S were properly assigned to shareholders – the documents are simply

¹⁴⁹⁸ Ex. SCE07V01, p. 126.

¹⁴⁹⁹ Ex. TURN-10, p. 13, citing DR TURN-SCE-090, Question 2.

¹⁵⁰⁰ Ex. TURN-10, p. 13, citing SCE Audit Report Affiliate Transaction Rules, December 2020, p. 14.

¹⁵⁰¹ Ex. TURN-10, p. 13, citing DR TURN-SCE-090, Question 2.

¹⁵⁰² Ex. TURN-10, p. 13, citing DR TURN-SCE-090, Question 2.

guidelines, training, and reports. The auditor does not get to see which costs SCE determined were incremental versus non-incremental, and whether SCE's determination was reasonable or proper. Again, the auditor can only audit the accounting trail *after* SCE has already determined which costs and how much are incremental.

The fact that SCE is the sole arbiter of which costs should be assigned to ratepayers versus shareholders is a clear conflict of interest.¹⁵⁰³ Furthermore, not only is there a clear conflict of interest, there are no auditable records to show that incremental costs have been properly assigned to shareholders, as discussed above. The combination of both should result in the Commission's finding that SCE has not met its burden of proof to demonstrate that incremental costs for NTP&S have been properly assigned to shareholders. SCE claims that it is too burdensome to keep these records, yet it also claims that these tests are consistently and correctly performed. Essentially, the public should just trust SCE. That does not give the public sufficient comfort, and it is not enough to meet SCE's burden of proof.

<u>Response to SCE's Rebuttal</u>. In its rebuttal, SCE claims that TURN nor any other party "has identified any NTP&S incremental costs that have been inadvertently included in this GRC revenue requirement,"¹⁵⁰⁴ and that "SCE's NTP&S and incremental costs have consistently been audited as part of the biannual Affiliate Transactions Audit conducted by the Energy Division or their external auditors."¹⁵⁰⁵ However, as obviously and thoroughly demonstrated by TURN during evidentiary hearings, it is impossible for TURN or other stakeholders to examine whether

¹⁵⁰³ Even though SCE attempted to argue in its rebuttal testimony (Ex. SCE18V01, p. 105) that SCE is not the sole arbiter of this decision, SCE's witness conceded on the stand that Edison alone makes the decision regarding which costs are considered incremental (12 RT 1167:8 – 1168:12).

¹⁵⁰⁴ Ex. SCE18V01, p. 94.

¹⁵⁰⁵ Ex. SCE18V01, p. 95.

NTP&S incremental costs have been included in the GRC revenue requirement precisely because SCE does not keep or maintain any records related 1) how much and why it determined certain costs were non-incremental,¹⁵⁰⁶ and 2) an accounting of costs that it determined to be non-incremental.¹⁵⁰⁷ In addition, because SCE does not keep any records, it is also impossible for the Commission or other auditors to conduct an audit of whether any incremental costs have been improperly classified as non-incremental.¹⁵⁰⁸ During evidentiary hearings, SCE absurdly suggested that despite there being no auditable records, the auditors should trust that SCE employees are doing the right thing based on its training and guidelines.¹⁵⁰⁹ When asked how the Commission or other stakeholders could find evidence that incremental costs were inappropriately classified as non-incremental if there is no record of non-incremental costs, SCE absurdly stated that the auditors should ask SCE employees, "Have you done anything inappropriate?"¹⁵¹⁰

38.8.2 The Commission Should Order SCE to Maintain Auditable "But For" Tests and Time Logs at Shareholder Expense

SCE states that a study by KPMG found that it would cost between \$4.36 to \$5.72 million annually to implement non-incremental resource tracking.¹⁵¹¹ SCE also provided alternates at lower costs, ranging from Time Sheets to Process Mining.¹⁵¹² However, it is unclear

¹⁵⁰⁶ 12 RT 1143:25 - 1145:19.

¹⁵⁰⁷ 12 RT 1155:7 – 1156:10.

¹⁵⁰⁸ 12 RT 1147:10 – 1149:6.

¹⁵⁰⁹ 12 RT 1149:7 – 1150:4.

¹⁵¹⁰ 12 RT 1150:7 – 1151:11.

¹⁵¹¹ Ex. SCE07V01, p. 134.

¹⁵¹² Ex. SCE07V01, p. 136.

whether these alternatives are as effective as maintaining "but for" logs, and whether these alternatives would lend themselves to auditing. Thus, the Commission should order SCE to maintain auditable "but for" tests and time logs at shareholder expense.

Proper record keeping and cost tracking are part of the costs of doing business, which is why shareholders should be responsible for these costs. Even using SCE's own "but for" test would result in the same conclusion – since SCE would not have incurred these record keeping costs *but for* the NTP&S Program, these costs should be deemed incremental and charged to shareholders.

TURN notes that even if these record keeping costs were charged to shareholders, SCE shareholders would still achieve an incredibly high level of return for the NTP&S Program. If we deduct the highest range in the cost of \$5.72 million per KPMG's study from SCE shareholders' after-tax net profit in 2022, it would result in \$9.28 million of after tax net profit from an investment of \$38.1 million,¹⁵¹³ or an outrageous return of 24.4%, which is 310% of SCE's authorized cost of capital.¹⁵¹⁴

Thus, it is more than reasonable for the Commission to order SCE to maintain auditable "but for" tests and time logs at shareholder expense if shareholders can still achieve a 24.4% return even assuming the highest range of the estimated costs.

¹⁵¹³ Ex. TURN-10, p. 15, citing DR TURN-SCE-090, Question 7.

¹⁵¹⁴ Ex. TURN-10, p. 15, citing SCE AL 5120-E, SCE's authorized rate of return is 7.87% for 2024.

38.8.3 The Unreasonable Sharing Mechanism and Arbitrary Determination of Incremental Costs Have Allowed Shareholders to Achieve Astronomical Levels of Profitability Under the NTP&S Program

The unreasonable sharing mechanism and SCE's ability to arbitrarily determine which costs should be borne by shareholders have allowed shareholders to achieve astronomical levels of profitability under the NTP&S Program. According to SCE's own records, SCE shareholders have achieved an average return of 39.6% from the NTP&S over the last 9 years:¹⁵¹⁵

	2014	2015	2016	2017	2018	2019	2020	2021	2022	Average
Cost to shareholders (\$M)	\$29.4	\$27.8	\$33.8	\$35.2	\$39.2	\$31.6	\$33.9	\$36.3	\$38.1	\$33.9
Shareholder After-tax Net										
Profit (\$M)	\$15.3	\$17.1	\$11.9	\$11.6	\$9.4	\$13.5	\$12.9	\$14.1	\$15.0	\$13.4
Shareholder Return (%)	52.0%	61.5%	35.2%	33.0%	24.0%	42.7%	38.1%	38.8%	39.4%	39.6%

Over the last three years from 2020 to 2022, the average return for shareholders was similarly at 38.8%.

SCE's current authorized cost of capital (or rate of return) is 7.87% per SCE AL 5120-E. Thus, a return of 39.6% would be an eye-popping 503.2% of SCE's authorized rate of return. Even if we use SCE's authorized cost of common equity of 10.75% as a comparison,¹⁵¹⁶ the shareholders were still able to achieve an outrageous 368.4% of its authorized cost of common equity under the NTP&S Program!

The Commission should recognize that this astronomical level of profitability for shareholders is unreasonable, especially if as noted above, there a clear conflict of interest in the

 ¹⁵¹⁵ Ex. TURN-10, p. 16, citing DR TURN-SCE-090, Question 7.
¹⁵¹⁶ Ex. TURN-10, p. 16, citing SCE AL 5120-E.

way that incremental costs are determined by SCE, and there are no auditable records to show that incremental costs have been properly assigned to shareholders.

Response to SCE's Rebuttal. In its rebuttal, SCE claims that over the life of SCE's GRSM, customers have received 72% of the net revenues (compared to 28% for shareholders), and that customers rate of return is "mathematically infinite" compared to the 39.6% of after-tax net return for shareholders. SCE's assertions are nonsensical and should be rejected by the Commission. First, a share of net revenues is meaningless without a comparison to the amount of investment or expenses borne by each party. For example, if Party A receives 90% of net revenues (compared to 10% for Party B), does that mean Party A is better off? Of course not, since one would need to consider the amount of investment by each party in order to calculate the rate of return. Party A could easily have a much higher percentage of the total net revenue but achieve a much lower rate of return. The same principle applies here when comparing ratepayers to shareholders. SCE concedes that ratepayers paid billions of dollars for the assets that are being used to provide NTP&S, and shareholders paid zero.¹⁵¹⁷ SCE further concedes that it would not be possible to offer these NTP&S if ratepayers did not pay billions of dollars for these assets.¹⁵¹⁸ Hence, the Commission should reject SCE's attempt to argue that ratepayer have somehow received a better deal under the GRSM structure because they've received 72% of the net revenues.

38.8.4 SCE's Usage of Ratepayer Funded Assets Raises Questions About Overbuilding of Capacity for Non-Utility Uses

SCE states that Edison Carrier Solutions ("ECS") uses "SCE's temporarily available

¹⁵¹⁷ 12 RT 1185:7 – 1187:9.

¹⁵¹⁸ 12 RT 1186:18 – 1187:2.

excess capacity on SCE's fiber network to provide commercial telecommunication non-voice services to non-residential customers."¹⁵¹⁹

SCE's record shows that as of 2023, SCE is only using 24.3% of rate-base fiber network capacity to provide energy utility operations.¹⁵²⁰ The remaining 75.7% is either used for NTP&S or unused. SCE's usage appears little changed from 2017, six years earlier, when it was only using 22.4% to provide energy utility operations,¹⁵²¹ which represents an increase of only 1.9% after six years. In fact, as of 2023, 68% of SCE's fiber network capacity remains unused,¹⁵²² which is an astonishing percentage. SCE's usage of ratepayer funded fiber network capacity raises questions about the overbuilding of capacity at ratepayer expense, potentially to enable SCE to provide NTP&S on the unused capacity.

In its rebuttal testimony, SCE attempts to argue that the "existence of capacity in excess of near-term electric utility operational needs is a prudent and natural consequence of the construction of fiber facilities to meet those utility needs" because it designs the fiber network to have enough capacity for the projected 15-20 year useful life.¹⁵²³ However, SCE's argument is belied by its own evidence. If that were the case, one would expect that over a six-year period, the excess capacity should reduce by 30% (6 divided by 20) to 40% (6 divided by 15). Yet, over a six-year period, SCE's excess fiber network capacity only reduced by 2.5%.¹⁵²⁴ Hence, SCE's claim does not add up.

¹⁵¹⁹ Ex. SCE07V01, p. 129.

¹⁵²⁰ Ex. TURN-10, p. 17, citing DR TURN-SCE-090, Question 3.

¹⁵²¹ Ex. TURN-10, p. 17, citing DR TURN-SCE-090, Question 3.

¹⁵²² Ex. TURN-10, p. 17, citing DR TURN-SCE-090, Question 3.

¹⁵²³ Ex. SCE18V01, p. 110.

^{1524 1.9% / 75.7% = 2.5%.}

This over-building was not the intent of the Commission when it adopted a sharing mechanism. The intent of the Commission was to provide a mechanism for the IOUs to enhance utilization of existing but temporarily underutilized utility assets.¹⁵²⁵ However, the unjust sharing mechanism awarding 90% to shareholders created a perverse incentive for SCE to take undue advantage of the opportunity and used ratepayer dollars to build far more fiber optic network capacity than necessary, which then used the extra capacity to enrich shareholders.

Both the Proposed Decision and the Alternate Proposed Decision in Application 17-02-001 agreed. The Proposed Decision stated that "[t]he record demonstrates that SCE's nontariffed fiber optic offering has increased to an inappropriate magnitude" and that "the rules permitting utilities to offer non-tariffed products and services and the 90/10 shareholder/ratepayer revenue sharing allocation established for SCE in D.99-07-070 were not intended to apply to this magnitude of overcapacity of utility assets."¹⁵²⁶ The Alternate Proposed Decision similarly stated, "the record demonstrates that SCE's non-tariffed dark fiber optic offering has reached a level far greater than that envisioned for non-tariffed product or service (D.97-12-088, as amended by D.98-08-035), and on which the 90/10 shareholder/ratepayer revenue sharing is based (D.99-09-070)."¹⁵²⁷ The Alternate Proposed Decision further noted that the Commission's intent when authorizing a sharing mechanism for Non-Tariffed Products & Services ("NTP&S") was that the products would "stem from only incidentally underutilized utility assets, not from a systematic build-up of assets funded by ratepayers."¹⁵²⁸

¹⁵²⁵ D.99-09-070, p. 58, Attachment A.

¹⁵²⁶ A.17-02-001, January 9, 2018 Proposed Decision, p. 20, Conclusions of Law #1.

¹⁵²⁷ A.17-02-001, July 5, 2018 Alternate Proposed Decision, p. 9.

¹⁵²⁸ A.17-02-001, July 5, 2018 Alternate Proposed Decision, p. 8.

This is an important question that needs to be addressed by the Commission as part of a comprehensive review of SCE's NTP&S Program, discussed further in the section below.

38.8.5 The Commission Needs to Perform a Comprehensive Review of the NTP&S Program, Including the Outdated Sharing Mechanism that Has Unreasonably Enriched Shareholders at Ratepayers' Expense

SCE asserts that the Commission has affirmed on numerous occasions that "any proposed changes to SCE's GRSM is subject to a separate rulemaking proceeding."¹⁵²⁹ However, that is no longer accurate with the recent issuance of PG&E's recent GRC decision, D.23-11-069. In that decision, the Commission approved PG&E's NTP&S program for only two years out of the four-year GRC cycle and noted that "longer-term continuation of this program, with funding by ratepayers, requires further information and consideration by the Commission."¹⁵³⁰ The Commission also ordered PG&E not to distribute NTP&S profits to shareholder but instead retain all profits in an interest-bearing account. If PG&E wishes to continue the program, PG&E is required to file an application to justify the continuation of the program, including "(1) details on the benefits to ratepayers, (2) how shareholders (or ratepayers) bear risks of potential loss, (3) information about its profit-sharing mechanism, and (4) how the program aligns with the Commission's Affiliate Transaction Rules," as well as detailed accounting, in addition to a comprehensive audit of the program.¹⁵³¹ Thus, SCE's statement that the sharing mechanism needs to be addressed in a rulemaking proceeding is no longer correct.

SCE's NTP&S Program is similarly due for a comprehensive review by the Commission, including issues TURN addressed above – conflict of interest, potential overbuilding of capacity,

¹⁵²⁹ Ex. SCE07V01, p. 124.

¹⁵³⁰ D.23-11-069, pp. 528-529.

¹⁵³¹ D.23-11-069, pp. 530-531.

accurate and auditable determination of incremental costs, astronomical levels of shareholder returns, and the revenue sharing mechanism. It is absurd that the dollar amount allocated 100% to ratepayers has never been adjusted after 25 years and remains the same amount (\$16.72 million) as 1999 – this means that adjusted for inflation, ratepayers are only receiving \$8.94 million today compared to \$16.72 million, a 46.5% decrease,¹⁵³² while shareholders continue to receive 70-90% above the threshold. This unjust and unreasonable result needs to be revisited by the Commission.

In its rebuttal testimony, SCE claimed that the \$16.72 million threshold amount was put in place to reflect the level of NTP&S and associated revenue at the time of SCE's 1995 GRC.¹⁵³³ SCE further claimed that "many of the services that made up the \$16.7 million Threshold Amount are no longer being offered by SCE," and that the \$16.7 million threshold should be either reduced significantly or even eliminated.¹⁵³⁴ Here, once again SCE's claim is belied by its own evidence. SCE's record shows that even though 13 areas of NTP&S that made up the \$16.7 million are no longer being offered in 2023,¹⁵³⁵ the revenue from the remaining NTP&S that made up the original \$16.7 million now totals to \$27.5 million in 2023,¹⁵³⁶ a 64.6% increase! This is further evidence that the outdated and inappropriate \$16.7 million threshold from 25 years needs to be revisited by the Commission.

When the Commission adopted the sharing mechanism in 1999, the Commission also

¹⁵³² Ex. TURN-10, p. 19; \$1 in 1999 = \$1.87 in 2024.

¹⁵³³ Ex. SCE18 V01, p. 106.

¹⁵³⁴ Ex. SCE18 V01, p. 106.

¹⁵³⁵ Ex. TURN-200, p. 4.

¹⁵³⁶ Ex. TURN-200, p. 4.

recognized that giving ratepayers 10% of revenues "may fall on the low side of the range of reasonableness."¹⁵³⁷ The most recent decision addressing revenue sharing mechanisms of which TURN is aware is the Sempra Utilities' 2012 general rate case decision. There, the Commission adopted a 25/75 shareholder/ratepayer revenue allocation for SDG&E's research and development activities (rather than the 40/60 shareholder/ratepayer allocation SDG&E proposed).¹⁵³⁸ Furthermore, in a recent application in March 2017, SDG&E also proposed allocating 25% of the revenue from sale or lease of Intellectual Property ("IP") rights to shareholders and 75% to ratepayers.¹⁵³⁹

In Application 17-02-001, after reviewing the record, Administrative Law Judge ("ALJ") Yacknin issued a Proposed Decision that adopted a 25/75 shareholder/ratepayer sharing mechanism.¹⁵⁴⁰ Separately, Commissioner Rechtschaffen issued an Alternate Proposed Decision that adopted a 50/50 shareholder/ratepayer sharing mechanism.¹⁵⁴¹ ALJ Yacknin's Proposed Decision *increased* ratepayer allocation by 650%, while Commissioner Rechtschaffen's Alternate Proposed Decision increased ratepayer allocation by 400%! Clearly, both determined that allocating only 10% of revenues to ratepayers was unreasonable and should be increased significantly.

SCE's unjust and unreasonable sharing mechanism needs to be revisited by the Commission. TURN recommends that the Commission order a comprehensive review of SCE's

¹⁵³⁷ D.99-09-070, p. 29.

¹⁵³⁸ D.13-05-010 (Sempra Utilities' 2012 GRC), pp. 600 and 1023-1024.

¹⁵³⁹ A.17-03-019, SDG&E Application, p. 6. The Commission adopted the proposed sharing in its Proposed Decision, but SDG&E ultimately withdrew the application.

¹⁵⁴⁰ A.17-02-001, January 9, 2018 Proposed Decision, p. 8.

¹⁵⁴¹ A.17-02-001, July 5, 2018 Alternate Proposed Decision, p. 10.

NTP&S Program as well as an audit, consistent with the recent Commission decision for PG&E's GRC. The Commission should authorize the NTP&S Program for two more years, and if SCE wishes to continue its NTP&S program, it should be required to file an application containing at a minimum the same information PG&E is required to submit. Furthermore, a comprehensive audit should be conducted by an independent auditor within 12 months of a Commission decision in this proceeding, consistent with D.23-11-069.

38.9 Operation And Maintenance Expense Forecast

38.10 Overhead Allocation

38.11 Reinvestments In Utility-Owned Generation Resources

39. RATE BASE

39.1 Plant In Service, Reserves, And Depreciation Expense

SCE's GRC forecast includes an estimate of rate base, which consists of three components: net plant-in-service, working capital, and accumulated deferred income taxes.¹⁵⁴² Net plant-in-service is SCE's electric plant-in-service and accumulated depreciation and amortization.¹⁵⁴³ SCE asks the Commission to find that its electric plant-in-service (plant) estimates are reasonable for purposes of determining GRC revenue requirements.

However, as TURN demonstrates in its testimony, SCE prematurely includes in its plant estimates certain costs recorded to memorandum accounts that have yet to be found reasonable for cost recovery by the Commission, including costs recorded to the Catastrophic Event Memorandum Account (CEMA) and Wildfire Mitigation Plan Memorandum Account

¹⁵⁴² Ex. SCE-07V02, p. 2.

¹⁵⁴³ Ex. SCE-07V02, p. 3.

(WMPMA).¹⁵⁴⁴ TURN identified \$883 million in plant costs in the RO Model associated with capital costs booked to CEMA and MWPMA that were, at the time of TURN's testimony, undergoing reasonableness review in other proceedings, plus \$41 million in plant associated with capital costs booked to CEMA for which SCE had yet to file a reasonableness review application.¹⁵⁴⁵ TURN recommended that these costs be excluded from plant within SCE's RO Model unless and until the Commission conducts a reasonableness review and approves cost recovery.

Since that time, the Commission has resolved two of the three reasonableness review proceedings considering costs SCE included in plant, while the third is still pending.¹⁵⁴⁶ In D.24-03-008, issued in A.22-06-003, the Commission authorized most of SCE's requested cost recovery but deferred consideration of \$21.09 million in WMPMA 2021 construction work-in-progress capital expenditures to a later proceeding.¹⁵⁴⁷ In D.24-05-037, issued in A.22-03-018, the Commission found reasonable and authorized recovery of all requested capital expenditures (and all but \$3.216 million in O&M expense).¹⁵⁴⁸ The third proceeding, A.23-10-001, has a proposed decision expected in the second quarter of 2025.¹⁵⁴⁹ SCE has also submitted a new application, A.24-04-005, seeking reasonableness review of costs recorded to CEMA for some,

¹⁵⁴⁸ D.24-05-037, p. 22.

¹⁵⁴⁴ Ex. TURN-19 (Yap), pp. 2-4.

¹⁵⁴⁵ Ex. TURN-19 (Yap), pp. 2 (Table 1) (identifying CEMA and/or WMPMA costs under review in A.22-03-018, A.22-06-003, and A.23-10-001), 4 (Table 2) (identifying CEMA costs for which SCE had yet to request a reasonableness review).

¹⁵⁴⁶ See Ex. TURN-19 (Yap), p. 2, Table 1.

¹⁵⁴⁷ D.24-03-008, as modified by D.24-06-025, p. 25. The Commission recounted SCE's statement that these costs are included in A.23-10-001. *Id*.

¹⁵⁴⁹ A.23-10-001, Email Ruling Modifying Procedural Schedule, 5/2/24.

but not all, of the CEMA events identified by TURN as not yet subject to a request for review by SCE.¹⁵⁵⁰ That proceeding is currently pending.

Even if the total dollar amount of unreviewed (and unauthorized) memorandum account balances is now smaller than at the time of TURN's testimony, the policy issue remains the same. The Commission should make clear that SCE may not include costs booked to memorandum accounts in plant in the RO Model unless the Commission has found those costs reasonable and authorized cost recovery. Additionally, the Commission should ensure that the GRC revenue requirements authorized in this proceeding for test year 2025 and attrition years 2026-2028 do not include capital costs prematurely included in plant by SCE.

39.1.1 A Utility Cannot Receive Cost Recovery for Capital Expenditures Recorded in Memorandum Accounts Until the Commission Reviews Those Costs for Reasonableness and Determines that They Were Prudently Incurred.

The Commission authorizes utilities to create memorandum accounts to track uncertain costs associated with specific activities that have yet to be found reasonable and necessary, such that the utility may apply for future cost recovery without violating the prohibition on retroactive ratemaking.¹⁵⁵¹ At issue in this proceeding are two memorandum accounts, CEMA and WMPMA. The Commission has made it abundantly clear that costs recorded in these memorandum accounts cannot be passed on to ratepayers without a Commission order approving cost recovery after a reasonableness review.

447

¹⁵⁵⁰ Ex. TURN-19 (Yap), p. 4, Table 2.

¹⁵⁵¹ *See, e.g.*, D.23-05-003, p. 6 (describing the role of memorandum accounts generally); Resolution E-3238, p. 2 (discussing the impetus for the creation of catastrophic event memorandum accounts).

In Resolution E-3238, the Commission authorized the utilities to create CEMA but cautioned that "authorizing the recording of costs associated with a disaster should not be construed as a prejudgment of the appropriateness of recovery of any amounts so accumulated."¹⁵⁵² The Commission clarified,

The Commission will examine closely all costs recovered in a utility's catastrophic event memorandum account before allowing their recovery in customers' rates. ... The costs recorded in the account will not be recoverable in rates without a request by the affected utility, a showing of their reasonableness, and approval by the Commission. Such a request must be made by formal application specifically for that purpose, by inclusion in a subsequent general rate case or other ratesetting application.¹⁵⁵³

Including CEMA costs in the RO Model that are yet-to-be reviewed and approved for cost recovery by the Commission will result in the prohibited inclusion of these costs in customer rates.

Similarly, in D.23-11-069, the Commission addressed PG&E's inclusion of costs in its GRC RO Model that were recorded to CEMA and WMPMA but not yet reviewed for reasonableness (as well as costs recorded to other memorandum accounts). The Commission concluded that such costs must be excluded from plant, explaining, "For amounts recorded in memorandum accounts, the Commission must first review those costs for reasonableness, and to include costs in rate base they must be both used and useful as well as prudently incurred."¹⁵⁵⁴

SCE asserts that it is appropriate to include these unapproved capital expenditures recorded to CEMA and WMPMA in its RO Model because these capital expenditures "represent used and useful net plant-in-service capital costs associated with expected rate base

¹⁵⁵² Resolution E-3238, p. 2.

¹⁵⁵³ Resolution E-3238, pp. 2-3.

¹⁵⁵⁴ D.23-11-069, issued in PG&E's 2023 GRC, p. 775 (citing Cal. Pub. Util. Code Section 451).

amounts."¹⁵⁵⁵ SCE is incorrect. In D.19-05-020, the Commission explained, "SCE cannot establish reasonableness based simply on a claim that an expenditure was made and has resulted in an investment which is used and useful for SCE's customers."¹⁵⁵⁶ Instead, SCE must first seek and receive a determination from the Commission that the costs are just and reasonable for recovery.

39.1.2 The Commission Should Direct SCE to Exclude from Plant Within its RO Model All Costs Recorded to Memorandum Accounts that the Commission Has Not Found Reasonable for Recovery.

As explained above, ratepayers should not pay for capital expenditures recorded to memorandum accounts unless the Commission first determines that SCE's costs were incremental and prudently incurred and then orders cost recovery. However, the unauthorized capital expenditures recorded to CEMA and WMPMA that SCE has included in plant in the RO Model will ultimately end up in GRC rates unless the Commission orders their removal.

In D.23-11-069, the Commission removed from PG&E's authorized GRC revenue requirement the test year and attrition year revenue requirements associated with the capital expenditures recorded in PG&E's memorandum accounts that the Commission had yet to find reasonable for recovery.¹⁵⁵⁷ The Commission pointed out that "PG&E will have the opportunity to seek recovery of such costs but must first request and obtain a determination from the Commission that the costs are just and reasonable."¹⁵⁵⁸ The same is true for SCE. Accordingly, the Commission should remove from SCE's test year and attrition year revenue requirements all

 ¹⁵⁵⁵ Ex. TURN-19, pp. 4-5 (quoting SCE's responses to TURN data requests); Ex. SCE-18V02A, pp. 2-3.
¹⁵⁵⁶ D.19-05-020, p. 333.

¹⁵⁵⁷ D.23-11-069, pp. 775-776.

¹⁵⁵⁸ D.23-11-069, p. 776.

costs recorded to memorandum accounts that the Commission has not found reasonable for cost recovery.

In rebuttal testimony, SCE suggests that if the Commission ultimately disallows any of the costs recorded to memorandum accounts in a reasonableness review, "the utility could then make a change to its GRC revenue requirement to reflect the disallowance."¹⁵⁵⁹ This adjustment will not make ratepayers whole unless SCE also refunds all capital-related revenue requirements already collected through GRC rates before the change. The simpler approach is to exclude the capital costs still subject to reasonableness review and only include them in GRC revenue requirements once the Commission has authorized cost recovery.

To facilitate this update to SCE's plant estimate, TURN recommends that the Commission direct SCE to provide an updated accounting of all capital costs in its plant figures in the RO Model that are recorded in memorandum accounts and have yet to receive a reasonableness determination and cost recovery authorization by the Commission. This exhibit should be provided as a late-filed exhibit with accompanying motion, and filed as late as possible but prior to the Commission's issuance of the proposed decision.

39.2 Working Capital (Excluding Customer Deposits)

39.2.1 Income Tax Lag

TURN's prior analysis in SCE's 2021 GRC recognized that due to net operating loss and other tax credit carryovers, SCE hadn't paid cash federal income taxes since 2009 and cash California income taxes since 2016.¹⁵⁶⁰ Furthermore, SCE was unlikely to have an actual tax

¹⁵⁵⁹ Ex. SCE-18V02A, p. 4.

¹⁵⁶⁰ D.21-08-036, p. 498.

burden during the 2021 rate case cycle.¹⁵⁶¹ In D.21-08-036, the Commission concluded that

SCE's forecast of lag days for state and federal income taxes were unreasonable because they

were unlikely to be representative of the lag days for the test year. As such, SCE's proposal was

not an appropriate basis for forecast ratemaking:

We do not find SCE's forecasted lag days for state and federal income taxes to be reasonable because SCE fails to demonstrate that they are likely to be representative of the lag days for the test year. SCE fails to justify going back to tax payment history for 2005-2009 and 2011-2016 to forecast lag days for 2021. We cannot ignore the reality that SCE last paid federal income taxes in 2009 and state income taxes in 2016. Moreover, SCE does not attempt to deny that its tax situation is unlikely to change in the upcoming GRC cycle. SCE generally agrees that it has incurred significant deductible tax costs over the past 10 years and that the deductibility of potential wildfire obligations could limit federal or state tax liabilities for the next few years.

Given that SCE has not paid federal income taxes for several GRC cycles and state income taxes since before the last GRC cycle and given the lack of evidence that SCE's tax situation is likely to change for this GRC cycle, we find TURN's proposal to use 365 lag days for both state and federal taxes to be reasonable for purposes of calculating the appropriate expense lag adjustment to working cash.¹⁵⁶²

Contrary to being a" flawed decision" as SCE asserts in its rebuttal,¹⁵⁶³ D.21-08-036 appropriately recognized that forecast ratemaking results in just and reasonable rates which reflect the cost of service when the forecast values are *reasonable*. If this fails to be the case, the Commission is correct in departing from standard practice to set rates. If the Commission were

¹⁵⁶¹ D.21-08-036, p. 499, citing Ex TURN OB at 279.

¹⁵⁶² D.21-08-036, p. 500.

¹⁵⁶³ Ex SCE-18, Vol 02A, p. 24.

tied to a methodology that resulted in an unreasonable forecast, it would have no mechanism by which to protect ratepayers from the negative effects of inflated utility forecasts in rates.

Further, D.21-08-036 explicitly notes that:

OII 24 "does not foreclose the possibility under extraordinary circumstances, it would be appropriate for the Commission to consider tax impacts associated with events outside the rate case in forecasting income tax expenses for ratesetting purposes. Circumstances under which a utility has not paid federal taxes for over a decade and state taxes for over a GRC cycle constitute such extraordinary circumstances that would warrant an alternative method.¹⁵⁶⁴

SCE has asserted in its testimony that due to alternative minimum tax requirements under the Inflation Reduction Act, it will be a cash taxpayer for federal taxes during this GRC cycle.¹⁵⁶⁵ SCE also asserts that it expects to be a cash state income taxpayer during the 2025 GRC cycle.¹⁵⁶⁶ However, SCE appears to misunderstand TURN's proposal when it states that "TURN suggests that working cash should be adjusted during the remaining GRC cycle if SCE ends up paying lower estimated taxes than forecast."¹⁵⁶⁷

TURN's proposal simply recommends that the Commission hold SCE to its word. Should SCE fail to owe cash federal and state taxes *again* during the 2025 GRC, SCE's proposed federal and state income tax lags of 54 days and 40 days respectively would continue to be unreasonable. Consequently, TURN proposes that the working cash allowance should be adjusted prospectively if this is the case. TURN notes that the tax lag associated with taxes represents a significant portion of SCE's working cash request. The difference in working cash

¹⁵⁶⁴ D.21-08-036, p. 501.

¹⁵⁶⁵ Ex TURN-02, p. 29., citing Ex. SCE 07 Vol 2., p. 39

¹⁵⁶⁶ Ex TURN-02, p. 29., citing Ex. SCE 07 Vol 2., p. 39

¹⁵⁶⁷ Ex. SCE 18 Vol. 02A., p. 24 citing Ex. TURN-02., p. 30.

requirements based on Cal Advocate's proposal which assumes no cash payment of federal taxes (365-day lag) and significantly lower than statutory levels of California state taxes (328.5-day lag) versus SCE's amended forecast (based on a 54 day federal tax lag, and a 40 day state tax lag) is \$447.262 million annually.¹⁵⁶⁸ Relative to SCE's opening testimony of \$1.223 billion, this adjustment in taxes represents more than one third¹⁵⁶⁹ of the total request. Should SCE continue its trend of owing no taxes into the 2025 GRC cycle contrary to its assertions, TURN believes that this would constitute an extraordinary circumstance of the kind contemplated by OII 24. As such, it would certainly warrant an alternative method for determining the appropriate working cash allowance for the remaining years of the rate case cycle. Should this prove to be the case, the Commission would be remiss not to prospectively adjust the authorized working capital to reflect a federal tax lag of 365 days and a state tax lag of 290 days consistent with SCE owing no cash taxes due.

39.3 Customer Deposits

The Commission has consistently treated SCE's customer deposits as a source of permanent working capital as an offset to rate base since SCE's 2003 GRC.¹⁵⁷⁰ In the 2012, 2015, 2018, and 2021 GRCs, SCE asked the Commission to reject this policy as it does again in this GRC.¹⁵⁷¹

 ¹⁵⁶⁸ Ex SCE 18 Vol 02, p. 24, Table III-11 (Taxes Based on Income Summary of Lag Days Proposals)
¹⁵⁶⁹ \$447.262/\$1,223 million =36.6%

¹⁵⁷⁰ Ex TURN-02, p. 30, citing D.15-11-021 (SCE 2015 GRC), p. 470.

¹⁵⁷¹ D.15-11-021, pp. 470-471, D.12-11-051 (SCE 2012 GRC), pp. 627-628.
SCE continues to argue that other utilities have different treatment for their customer deposits. However, as SCE itself points out, the Commission has "treated customer deposits inconsistently among the large utilities."¹⁵⁷² Furthermore, in D.04-07-044 (SCE's 2003 GRC), the Commission explicitly addressed the applicability of SP U-16 to the treatment of customer deposits, stating that "as the Commission previously held, U-16 is only a guide, and deviations are appropriate where circumstances warrant."¹⁵⁷³ TURN does not believe it is appropriate that SCE be allowed to pick and choose its "preferred treatment."¹⁵⁷⁴ While SCE's customer deposits have declined, even at the lowest forecast value recommended by TURN, they represent a significant source of working capital which does not have to be provided by other investors.

In this GRC, SCE argues that the Commission's treatment of its customer deposits as a permanent source of working capital is predicated on Commission comments that the SCE's customer deposits have remained high and stable over time.¹⁵⁷⁵ SCE appears to indicate that the reduction in customer deposits during Covid is a reason the Commission should change its historical treatment in this GRC. However, customer deposits continue to provide a significant portion of working capital for SCE. SCE's own forecasts project customer deposit to increase toward the end of this GRC cycle, with annual averages exceeding the 2023 levels by 2027 and reaching a high of \$197 million in 2028.¹⁵⁷⁶ In recognition of the unusual circumstances of Covid and recovery in its aftermath, TURN recommended that the Commission maintain its

¹⁵⁷² Ex SCE 18 Vol 02, p. 30.

¹⁵⁷³ D.04-07-044, p. 253.

¹⁵⁷⁴ Ex SCE 18 Vol 02, p. 31.

¹⁵⁷⁵ Ex SCE 18 Vol 02, p. 32.

¹⁵⁷⁶ Ex TURN-02, p. 32., citing TURN DR 089 and Ex SCE 07 Vol. 02, p. 49.

treatment for SCE's customer deposits and adopt the value of \$174 million, which represents the *lowest* annual level of customer deposits held as projected by SCE.¹⁵⁷⁷ Thus, adopting the \$174 million value would represent the most conservative approach to estimating the amount of customer deposits held by SCE throughout this GRC period.

Hence, the Commission should maintain its longstanding practice and treatment of SCE's customer deposits and adopt \$174 million as an offset to rate base.

39.4 Taxes

40. SCE ASSET DEPRECIATION STUDY

SCE seeks a \$313 million increase to its authorized depreciation expense for test year 2025, calculated based on year-end 2022 plant balances (a 14% increase).¹⁵⁷⁸ Of this amount, \$294 million is for depreciation of transmission and distribution assets, and represents the combined impact of SCE's proposed \$212 million increase for net salvage costs, with an additional \$82 million increase resulting from SCE's proposed service lives for those assets.¹⁵⁷⁹

TURN's testimony regarding these accounts continues to rely on the analytical methods used in recent GRCs where the Commission generally or entirely adopted outcomes consistent with TURN's recommendations. TURN's recommended net salvage rates include adjustments consistent with the principle of "gradualism," just as was the case in SCE's test year 2021 GRC

¹⁵⁷⁷ Ex TURN-02, p. 32.

¹⁵⁷⁸ Ex. TURN-16 (Prepared Testimony of David Garrett), pp. 5-6. The dollar figures discussed here as well as those set forth in TURN's depreciation testimony are based on plant balances as of December 31, 2022. This permits a direct comparison to the depreciation accruals presented in SCE's depreciation study, which was also based on plant balances as of December 31, 2022. The actual depreciation expense for the 2025 test year will reflect the adopted depreciation rates applied to the authorized plant balance for 2025, and will be higher to the extent the 2025 plant balances are higher than the end-of-year 2022 figures.

¹⁵⁷⁹ Ex. SCE-18, Vol. 3, p. 1, Table I-1.

when the Commission adopted TURN's net salvage recommendations in all disputed accounts. And TURN's recommended average service lives are based on a straightforward analysis without any need to rely on "statistically aged" or otherwise simulated data, just as was the case in PG&E's test year 2023 GRC when the Commission adopted TURN's service life recommendations in very nearly all disputed accounts.¹⁵⁸⁰

The Commission should limit any increase in depreciation expense to no more than is recommended in TURN's testimony and presented in this brief, based on the evidentiary record developed in this proceeding. TURN's proposed net salvage rates for SCE's transmission and distribution accounts are consistent with the Commission's commitment to "gradualism" in this area. TURN's proposed service live for several of the transmission and distribution plant accounts are more reasonable in light of the utility's recorded retirement data. The combined impact of TURN's positions and recommendations is a \$71.6 million increase as compared to the currently authorized depreciation parameters.¹⁵⁸¹

40.1 T&D Net Salvage

For a large number of its T&D mass property accounts, SCE proposes net salvage rates that are substantially more negative and thus lead to higher depreciation rates, all else equal. The utility's proposed changes to net salvage rates for T&D accounts would result in \$212 million of

¹⁵⁸⁰ TURN wishes to be clear that this brief's discussion of the topics of net salvage rates and average service lives is to varying degrees something of a cut-and-paste of briefs TURN submitted in each of these two relatively recent GRC proceedings. This approach seemed particularly appropriate where, as here, the utility's rebuttal testimony largely cut-and-pasted material from rebuttal testimony submitted on net salvage issues in the test year 2021 SCE GRC, and on average service life issues in the test year 2023 PG&E GRC. An obvious difference between the two approaches is that TURN is relying on material consistent with the outcomes the Commission adopted in those earlier proceedings.

¹⁵⁸¹ Ex. TURN-18, p. 6.

increased annual depreciation accrual (based on 2022 year-end plant balances).¹⁵⁸² TURN's depreciation analysis relied on the Commission's past commitment to "gradualism" and recommended smaller changes to the currently authorized net salvage rates.

40.1.1 General Principles of Net Salvage

The "net salvage rate" for a particular plant account represents the combined effect of the "gross salvage" the utility might obtain from an asset at the end of its useful life, and the "cost of removal" associated with removing the asset from service. For nearly every T&D mass property account of SCE, the net salvage rate is a negative figure, because the cost of removing the assets from service is expected to exceed the gross salvage value. When a negative net salvage rate is applied to the plant balance in an account to calculate the depreciation rate, it results in increasing the total depreciable base to be recovered over a particular period of time and, by extension, increases the depreciation rate. Therefore, a greater negative net salvage rate equates to a higher depreciation rate and expense, all else held constant.¹⁵⁸³

Net salvage rates are calculated by determining gross salvage and removal costs at the time of retirement as a percent of the original cost of the assets retired. In other words, salvage and removal costs are based on current dollars (when the assets are removed from service), while retirements are based on historical dollars, reflecting uninflated cost figures from years, and often decades earlier. Increasing labor costs associated with asset removal combined with the fact that original costs remain the same have contributed to increasing negative net salvage over time.¹⁵⁸⁴

¹⁵⁸² Ex. SCE-07, Vol. 03, p. 2, Table I-1.

¹⁵⁸³ Ex. TURN-16 (Garrett), pp. 34-35.

¹⁵⁸⁴ *Id.*, p. 35.

40.1.2 TURN's Net Salvage Recommendation – The Commission Should Limit Increases Consistent With The "Gradualism" Employed In Recent GRC Decisions.

In PG&E's test year 2014 GRC, the Commission expressed concern over the increasing

negative net salvage values reported by the utilities, and the impact the resulting requests for

increased depreciation expense could have on the utility's customers.¹⁵⁸⁵ To mitigate the impact

on the utility's rates, the Commission there described and relied on the concept of "gradualism":

In evaluating whether a proposed increase reflects gradualism, however, we believe the more appropriate measure is how the change affects customers' retail rates. The fact that PG&E previously proposed higher removal costs than adopted has no bearing on how a proposed change would impact current ratepayers. Accordingly, we apply the principle of gradualism based on how a proposed change in estimate compares to adopted costs reflected in current rates, irrespective of what PG&E may have forecasted in an earlier depreciation study.¹⁵⁸⁶

To achieve an outcome consistent with "gradualism," the Commission's general

approach was to "adopt no more than 25% of PG&E's estimated increases in the accrual

provisions for removal costs," in order to "temper[] the impacts on current ratepayers."¹⁵⁸⁷

Since the PG&E 2014 GRC, in GRCs where the Commission has seen fit to modify net

salvage values at all for the major energy utilities, it has limited any changes to no more than

25% of the estimated increase produced by the utility's depreciation study.

Most recently, in PG&E's 2023 GRC, the company proposed negative salvage rate increases for several of its distribution and transmission plant accounts. In response, TURN proposed the Commission limit the net salvage rate increases to no more than 25% of the increase requested by the utility, consistent with the Commission's policy regarding

¹⁵⁸⁵ D.14-08-032 (PG&E test year 2014 GRC), p. 597.

¹⁵⁸⁶ *Id.* at 598.

¹⁵⁸⁷ *Id.*, at 602.

gradualism.¹⁵⁸⁸ The Commission found that "PG&E's proposed net salvage values are inconsistent with principles of gradualism"¹⁵⁸⁹ and instead adopted TURN's estimates of net salvage percentages for the accounts in dispute.¹⁵⁹⁰

Gradualism has been a consistent element of the Commission's decisions in recent SCE GRC proceedings, with one exception where the Commission authorized no change to existing depreciation rates. In the test year 2021 GRC decision for SCE, the Commission stated that it had applied the gradualism principle in adopting net salvage rates in SCE's 2015 GRC, and continued to endorse the concept for the 2021 GRC cycle, limiting net salvage increases to 25% of SCE's requested increases.¹⁵⁹¹

SCE's depreciation study proposes increased (that is, more negative) figures for the net salvage rates for twelve T&D accounts, as addressed in TURN's testimony.¹⁵⁹² There appears to be no mention in the utility's direct testimony of the concept of "gradualism" as applied to the utility's net salvage analysis here, or any recognition that the Commission had relied on "gradualism" in adopting net salvage values in several of SCE's most recent GRCs.¹⁵⁹³ While TURN's study concluded that SCE's net salvage rates warranted some movement in the direction requested by the utility, TURN's proposed adjustments are based on the 25% cap originally described in the PG&E 2014 GRC decision, and reaffirmed in the more recent GRC

¹⁵⁸⁸ D.23-11-069 (in PG&E TY 2023 GRC), p. 672, Table 10-D.

¹⁵⁸⁹ Id.

¹⁵⁹⁰ Id. at 674-675.

¹⁵⁹¹ D.21-08-036 (in SCE TY 2021 GRC), p. 512.

¹⁵⁹² Ex. TURN-16, p. 8, Figure 3.

¹⁵⁹³ SCE's depreciation study mentions "gradualism" several times in the context of its average service life proposals, where application of the principle might limit reductions that could otherwise be made to depreciation rates.

decisions: For each account for which SCE proposed a more negative net salvage rate, TURN's adjustments limit the change to 25% of the utility's estimated increase. SCE calculates TURN's recommended net salvage rates would produce a \$52.4 million increase in the annual depreciation expense when viewed in isolation.¹⁵⁹⁴

40.1.3 After Ignoring the Concept of Gradualism In Its Depreciation Study, SCE Raised Oft-Rejected Or Baseless Arguments Against It in Rebuttal.

Coming into this GRC, SCE should have been well aware of the Commission's recent

practice of applying the concept of "gradualism" in setting the level of reasonable net salvage

increases where the Commission has determined that higher net salvage rates of any amount are

warranted. In the utility's test year 2021 GRC, the Commission's discussion of net salvage rates

concluded with the following:

We continue to endorse the concept of gradualism with respect to net salvage rates for this rate case cycle given that the overall cost increases at issue in this GRC (for both Track 1 and Track 2) are substantial and ratepayers are facing a great deal of economic uncertainties associated with the global COVID-19 pandemic. [cites omitted] Even SCE recognizes that its requested net salvage rate increase is significant. [citing SCE's rebuttal testimony]. In consideration of these factors and consistent with past Commission precedent, we find it reasonable to limit any net salvage increases to 25 percent of SCE's requested increases.¹⁵⁹⁵

And in SCE's test year 2018 GRC, the utility itself proposed a cap on its proposed net salvage rate increases in the name of such "gradualism." However, the Commission found such "little merit" in the net salvage elements of SCE's depreciation study that it declined to apply the

¹⁵⁹⁴ Ex. SCE-18, Vol. 3, p. 3, Table II-2.

¹⁵⁹⁵ D.21-08-036 (SCE TY 2021 GRC), p. 512.

concept of gradualism under the circumstances, and instead retained the then-authorized rates without any increase as the more reasonable outcome.¹⁵⁹⁶

Here, SCE made no mention of "gradualism" in the net salvage portion of its depreciation study,¹⁵⁹⁷ and instead proposed increased net salvage rates that, if adopted, would result in an annual depreciation expense increase of \$212 million when applied to 2022 plant balances (which would translate to a correspondingly larger increase if applied to authorized 2025 plant balances).¹⁵⁹⁸ TURN submits that the Commission's response to SCE's near-identical showing in support of the requested \$212 million increase here should be the same as was adopted in the prior GRC when the utility's requested increase was \$199 million; it should "instead find reasonable the consistent approach set forth in TURN's proposal."¹⁵⁹⁹

40.1.3.1 SCE's Ongoing Claims Of Deficient Depreciation Rates Continue To Be Inadequately Supported.

SCE has a longstanding practice of raising arguments regarding the purported deficiency of previously-adopted depreciation rates, arguments the Commission has regularly and uniformly rejected or chosen to leave unmentioned in past GRC decisions. SCE's arguments are based on its assumption that the amounts it has recorded as cost of removal, as well as its past and present proposals for depreciation accruals and calculations of future costs of removal, together represent a sacrosanct truth regardless of what the Commission has said about them in past decisions. And anything that would result in depreciation accruals at a level less than the utility has requested,

¹⁵⁹⁶ D.19-05-020 (SCE test year 2018 GRC), pp. 314 and 319.

¹⁵⁹⁷ Ex. SCE-07, Vol. 3, pp. 13-65.

¹⁵⁹⁸ Ex. SCE-18, Vol. 3, p. 1, Table I-1.

¹⁵⁹⁹ D.19-05-020, pp. 508 and 512.

whether here or in prior GRCs, represents a "deficit" that the utility finds compelling but the Commission has never embraced. For example, SCE contends that the adopted net salvage rates in recent GRCs "have been set below the levels justified in corresponding depreciation studies."¹⁶⁰⁰ But according to those GRC decisions, the Commission adopted net salvage rates at the levels determined to be reasonable given an evidentiary record that consisted not only of SCE's depreciation study, but material establishing the fundamental flaws of that study. In SCE's test year 2012 GRC, the utility claimed that its accumulated depreciation balance as of the end of 2009 should be \$2.7 billion higher than it was because previously authorized depreciation rates have not kept pace with removal costs, and sought a depreciation expense increase of \$59 million per year to address the purported deficit (in addition to the increase of \$511 million from the utility's proposed changes to depreciation parameters).¹⁶⁰¹ The Commission dismissed SCE's contention as relying on a self-fulfilling prophecy of the utility's own making:

Regarding documentation of the accumulated depreciation deficit, SCE's basis to change its rates, we recall the function of the reserve is to allocate cost recovery for the cost of installation and removal of a group of assets over the service life. The Commission previously adopted depreciation rates and service lives, and SCE has made the resulting cost allocations. The calculated "deficit" is the mathematical difference between what SCE asked for and what was authorized by the Commission.[¶] On the other hand, slightly different assumptions would significantly influence the sufficiency of the accumulated depreciation reserve. Thus, SCE's deficit argument is self-fulfilling because it presumes that its assumptions in prior GRC requests were correct, including constant escalation of COR, even though some assumptions were not adopted by the Commission or borne out by actual retirements. [¶] For purposes of this GRC, we do not determine whether the \$2.7 billion claimed deficit is an accurate number.¹⁶⁰²

¹⁶⁰⁰ Ex. SCE-18, Vol. 3, p. 4. SCE made the identical argument in its rebuttal testimony in the test year 2021 GRC. Ex. TURN-117, p. 5.

¹⁶⁰¹ D.12-11-051 (SCE TY 2012 GRC), pp. 658-659.

¹⁶⁰² *Id.*, pp. 671-672.

SCE made a similar claim in its TY 2015 GRC, and the Commission had a consistent

response:

Generally, SCE argues that its currently authorized depreciation rates are too low, thus shifting costs from current customers to future customers. SCE claims that its depreciation proposals reduce, but do not eliminate this cost shifting, while the TURN and ORA proposals would exacerbate it. As we noted in the last GRC decision, SCE's calculations of past depreciation "deficits" and ongoing or future "deferrals" are merely calculations reflecting the difference between SCE's proposals for depreciation parameters and Commission-adopted or party-proposed parameters. SCE's point that if ongoing depreciation expense is "too low," future customers will be required to pay more may be valid. However, we recognize that determining the "right" level of depreciation expense is a complex exercise of forecasting future costs and events. SCE's calculations of deficits and deferrals are only valid if we assume that SCE's past and present proposals are correct. We do not start with this assumption; instead, we remind SCE that it bears the burden of proof that its proposals are reasonable.¹⁶⁰³

The Commission continues to have good reason to be dubious of SCE's recorded net

salvage figures. The utility reports 5-year and 10-year averages for net salvage rates for some of its largest distribution plant accounts that, if they are to be taken at face value, suggest the Commission should expect that it would cost from three to nearly ten times as much to remove the plant in service than it originally cost to install the plant.¹⁶⁰⁴ And while SCE describes the cost of removal values as "recorded costs," the largest part of those recorded costs are the product of an allocation of the total costs of the underlying plant replacement project.¹⁶⁰⁵ In this way, SCE has substantial control over the amounts that it is reporting as "recorded" costs of

¹⁶⁰⁵ D.15-11-021 (SCE TY 2015 GRC), pp. 412-413.

¹⁶⁰³ D.15-11-021 (SCE TY 2015 GRC), pp. 394-395 (footnote citations omitted).

¹⁶⁰⁴ Ex. WP-SCE-07, Vol. 3, pp. 69 (for Account 364) and 90 (for Account 369). The -333% and -407% figures reported as the 5-year and 10-year average for Account 364 (Distribution Poles) indicate net salvage cost (primarily the cost of plant removal at the end of its life) of 3.3 to 4 times the original plant cost. For Account 369 (Services), the -972% and -696% averages indicate net salvage cost of 7 to nearly 10 times the original plant cost. For Account 364, SCE's recorded figures for cost of removal have exceeded -500% in five of the past ten years.

removal or net salvage costs. And these "recorded" costs are the fodder of its ongoing dire predictions of the consequences that would follow should the Commission fail to authorize increases in the amounts SCE requests.

40.1.3.2 SCE Falls Short with its Attempts To Illustrate The Inadequacy Of The Depreciation Rates Found Just and Reasonable in Past GRCs.

In its rebuttal testimony, SCE describes the GRC outcomes since 2009 as suggesting a pattern of the Commission adopting longer service lives while "looking skeptically at proposed increases to net salvage rates," resulting in lower overall depreciation rates "even if a gradual aggregate increase was warranted." But the associated figure in the utility's rebuttal illustrates that over the same period, SCE itself proposed a lower overall depreciation rate as compared to the overall then-authorized rate from the 2009 GRC to the 2018 GRC, and the requested amount for 2021 was the second-lowest of the five GRCs.¹⁶⁰⁶ In other words, SCE is criticizing the Commission for failing to authorize an increased overall depreciation rate in prior GRCs, even though SCE itself did not seek an increased overall depreciation rate. Furthermore, if the record in those prior proceedings convinced the Commission that it would be reasonable to adopt longer service lives, but left the agency unconvinced of the reasonableness of SCE's proposed net salvage changes, the utility should not now try to pin the blame for that on the Commission.

SCE's rebuttal testimony also includes an attempt to illustrate that adoption of longer service lives along with "stagnated net salvage rates [has led] to a growing and distressing gap between recorded costs and GRC-authorized costs for net salvage."¹⁶⁰⁷ Again, the "recorded"

¹⁶⁰⁶ Ex. SCE-18, Vol. 3, pp. 5-6 and Figure II-1.

¹⁶⁰⁷ Ex. SCE-18, Vol. 3, pp. 7-8 and Figure II-2.

costs are the SCE-produced figures the Commission did not adopt, whereas the GRC-authorized costs are the figures the Commission found reasonable based on the record of each of the GRCs reflected in SCE's graphic. And the illustration is telling, but not at all in the manner SCE suggests. From the 2009 GRC through the 2021 GRC, the SCE-proposed average service lives increased by approximately 25% overall (from 39 years to 50 years), while its 10-year average "recorded" figures for net salvage rates increased by approximately 80% (from -80% in 2009 to -145% in 2021). The GRC authorized net salvage rates increased by approximately 33% (from -54% to -72%) during that same period. TURN submits that SCE's table raises far more troubling questions about the pattern displayed by its "recorded" net salvage figures over this period than it does about the reasonableness of the Commission-adopted outcomes. Similarly, the comparison SCE makes between "unit costs" in this context actually demonstrates the reverse of the utility's apparent point.¹⁶⁰⁸ Here the comparison is between SCE's 2025 GRC version of the table and the 2021 GRC version.¹⁶⁰⁹ In the 2021 GRC, SCE calculated a unit cost based on its purported "Recent Cost to Remove," and compared it to a unit cost based on the "Authorized COR" (that is using 2018 GRC-authorized figures), and "TURN Proposed" and "SCE Proposed." SCE's calculations purported to show that the "Authorized COR" and "TURN Proposed" amounts were uniformly less than the "Recent Cost to Remove" for each of the four distribution accounts the utility selected. The 2025 GRC version of the table shows marked improvement, even though the "Authorized COR" increased not by the amount SCE sought, but the gradualism-tempered amount proposed by TURN. That is, the SCE-calculated difference between its Recent Cost to Remove" and the Authorized COR unit cost figures was substantially

¹⁶⁰⁸ Ex. SCE-18, Vol. 3, p. 9, Table II-3.

¹⁶⁰⁹ Ex. TURN-117 (SCE TY 2021 GRC Depreciation Rebuttal Excerpt), p. 11, Table II-3.

less in the 2025 GRC calculations than they were in the 2021 GRC calculations.¹⁶¹⁰ To be clear, TURN does not in any way agree that SCE's underlying calculations of its "Recent Cost to Remove" is valid. But by the utility's own calculations, even with the net salvage value increases adopted in 2021 being limited by gradualism, those increases significantly closed the gap. No doubt, the net salvage values adopted under gradualism may not have closed the gap as much as SCE would prefer. But a comparison of the two tables leaves no doubt that, even with gradualism being consistently applied, the movement is in the direction of greater depreciation accruals.

40.2 T&D Average Service Life

40.2.1 TURN's Analysis And Recommendations Are Firmly Based On SCE's Retirement Data and Produce Reasonable Curves and Lives.

TURN's recommended service lives are the product of a straightforward analysis that relies on the objective data recorded by SCE (rather than simulated data), and the employment of mathematical and visual curve fitting and expert judgment to derive a reasonable Iowa curve for each of the accounts in dispute.¹⁶¹¹ TURN proposes service life adjustments to seven of SCE's transmission and distribution accounts. TURN's service life analysis relied upon the "retirement rate method," the most common actuarial method used by depreciation analysts. The retirement rate method is ultimately used to develop an observed life table (OLT) which shows the percentage of property surviving at each age interval, yielding a pattern of property retirement described as a "survivor curve."

¹⁶¹⁰ For Account 365 (Overhead Conductor), the difference declined from \$1.70 to \$0.13; for Account 366 (Overhead Conduit), the difference declined from \$7.32 to \$2.94; for Account 367 (Underground Conductor), the difference declined from \$5.61 to \$3.98; and for Account 368 (Transformers), the difference declined from \$837 to \$212.

¹⁶¹¹ Ex. TURN-16 (Testimony of David Garrett), pp. 11-19.

Since the retirement data for an account typically does not provide a smooth or complete curve, the depreciation witness selects an Iowa curve to complete the curve-fitting process and to derive a recommended service life for each account.¹⁶¹² TURN's analysis used the aged property data provided by SCE to develop an OLT curve for each transmission and distribution plant account, then engaged in a curve-fitting process to select the Iowa curve that best fit the OLT curve. For the curve-fitting, TURN's analyst relied upon a combination of visual and mathematical techniques, as well as relying on his professional judgment. He first reviewed the OLT curve data to ensure the analysis reflected the more reliable data, without irregularities or erratic shifts. He then applied a mathematical curve fits, and observed the OLT against potential Iowa curves in order to determine how well the curve fits visually. This process might be repeated several times for any given account in order to ensure that the most reasonable Iowa curve is selected.¹⁶¹³

TURN's curve selection process does not rely exclusively on any single step of this analysis. For example, while mathematical fitting is an important part of the curve-fitting process because it promotes objective, unbiased results, TURN's analyst recognized it may not always yield the optimum result. Similarly, not every portion of the OLT curve should be given equal weight. Often the "tail end" of a curve may have less analytical value that other portions of the curve, and should be given less weight. The fitting process therefore focuses not only on

¹⁶¹² TURN's testimony included a more detailed explanation of how the Iowa curves are used in the actuarial analysis. Ex. TURN-16 (Garrett), Appendix C.

¹⁶¹³ Ex. TURN-16 (Garrett), pp. 11-13.

the entire OLT curve, but also the portion that presented the most significant part of the curve for certain accounts.¹⁶¹⁴

TURN's curve selection also does not rely on simulated data produced by SCE's "statistical aging" process. As explained in TURN's testimony, SCE maintains what is known as "aged data," which means a record is kept of the vintage year of a retired asset so that the age is known. In this case, SCE has aged retirement experience available dating back to 2002. That is, the utility's actual recorded retirement experience dating back to 2002 are real data, based entirely in fact. In the opinion of TURN's expert witness, twenty years of retirement experience data is sufficient for actuarial analysis without the need to supplement the data.¹⁶¹⁵ In PG&E's 2023 GRC, the Commission agreed, as it adopted nearly all of TURN's proposed service lives based on an analysis using only the 22 years of actual recorded retirement experience data, rather than the "statistically aged" data PG&E had presented in that case.¹⁶¹⁶

The following sections summarize TURN's showing on each of the seven accounts for which TURN proposes a life-curve that is different than SCE's proposal for the account.

40.2.1.1 Account 352 (Structures and Improvements)

TURN recommends a curve of R1-67, whereas SCE recommends R2-60.¹⁶¹⁷ For this account, the graph in TURN's testimony illustrates that the R1-67 curve TURN selected results in a closer fit to the OLT than does SCE's selected curve, and includes a truncation line demarking the point at which the data points in the OLT become less relevant. For mathematical

¹⁶¹⁴ *Id.*, pp. 13-14.

¹⁶¹⁵ *Id.*, pp. 15-16.

¹⁶¹⁶ D.23-11-069 (PG&E TY 2023 GRC), p. 670, Table 10-C.

¹⁶¹⁷ Ex. TURN-16 (Garrett), pp. 19-21.

curve fitting, the sum-of-squared-differences (SSD) is 5.9040 for SCE's proposed curve, but 2.0254 for TURN's, indicating the R1-67 curve is a better mathematical fit with SCE's historical data. The Commission should adopt the R1-67 curve as proposed by TURN.

40.2.1.2 Account 354 (Towers and Fixtures)

TURN recommends a curve of R4-76, whereas SCE recommends R5-70.¹⁶¹⁸ For this account, the graph in TURN's testimony illustrates that the R4-76 curve TURN selected results in a closer fit to SCE's OLT than does SCE's selected curve. For this account, TURN's witness applied his judgment to select a reasonably well-fitting curve rather than the best-fitting curve, as the latter would likely result in an unreasonably long life estimate. Nonetheless, this does not mean the data are irrelevant, particularly when SCE has not presented any compelling evidence outside the statistical data to support its proposed service life for this account. Both TURN's and SCE's proposed curves suggest an increase in the retirement rate going forward, but the R4-76 curve TURN recommends gives more credit to SCE's observed data. For mathematical curve fitting, the sum-of-squared-differences (SSD) is 0.3489 for SCE's proposed curve, but 0.1418 for TURN's, indicating the R4-76 curve is a better mathematical fit with SCE's historical data. The Commission should adopt the R4-76 curve as proposed by TURN.

40.2.1.3 Account 356 (Overhead Conductors and Devices)

TURN recommends a curve of R2.5-74, whereas SCE recommends R3-65. The R2.5-74 curve results in a good balance between giving due credit to the actual retirement experience in this account while being conservative to ensure the service life estimate is not unreasonably long.

¹⁶¹⁸ *Id.*, pp. 22-24.

If future depreciation studies indicate a shorter average service life, then life estimates can be adjusted accordingly. TURN's proposed curve again achieves a better mathematical fit to the OLT curve (an SSD of 2.0645 for SCE's, and 0.7608 for TURN's).¹⁶¹⁹ The Commission should adopt the R2.5-74 curve as proposed by TURN.

40.2.1.4 Account 357 (Transmission Underground Conduit)

TURN recommends a curve of R4-61, whereas SCE recommends R4-55. Again, the R4-61 curve results in a good balance between giving due credit to the actual retirement experience in this account while being conservative to ensure the service life estimate is not unreasonably long. TURN's proposed curve again achieves a better mathematical fit to the OLT curve (an SSD of 0.0963 for SCE's, and 0.0182 for TURN's).¹⁶²⁰ The Commission should adopt the R4-61 curve as proposed by TURN.

40.2.1.5 Account 366 (Distribution Underground Conduit)

For Account 366, TURN recommends a curve of R2.5-66, whereas SCE recommends R3.0-60. The graph included in TURN's testimony shows the vast majority of this OLT curve is statistically relevant. Both of the selected Iowa curves provide relatively close and similar fits to the OLT curve through age 40. After that point, the higher mode of SCE's selected R3 curve (i.e., a less flattened trajectory) causes it to diverge from the OLT curve relative to the lower-mode of the R2.5 Iowa curve TURN recommends. Since SCE has not presented any meaningful evidence beyond its historical retirement data for this account, then an estimated average life of

¹⁶¹⁹ Ex. TURN-16 (Garrett), pp. 24-26.

¹⁶²⁰ Ex. TURN-16 (Garrett), pp. 26-28.

66 years is more supported by the evidence. TURN's proposed curve again achieves a better mathematical fit to the OLT curve (an SSD of 0.1535 for SCE's, and 0.0392 for TURN's).¹⁶²¹ The Commission should adopt the R2.5-66 curve as proposed by TURN.

40.2.1.6 Account 367 (Underground Conductors and Devices)

TURN recommends a curve of L1-50, whereas SCE recommends R1.5-45. TURN's testimony explained that both recommended curves result in relatively close fits to the OLT curve and are both within a reasonable range for this account. TURN's proposed curve again achieves a better mathematical fit to the OLT curve (an SSD of 0.0239 for SCE's, and 0.0054 for TURN's).¹⁶²² The Commission should adopt the L1-50 curve as proposed by TURN.

40.2.1.7 Account 369 (Services)

TURN recommends a curve of R2.5-62, whereas SCE recommends R2-55. For this account, there is a more pronounced difference between the OLT curve reflecting SCE's actual retirement experience, which TURN relied upon, and the OLT curve to which SCE has added its statistically aged data to its actual retirement experience. The graph in TURN's testimony illustrates that the R2.5-62 curve is a much closer fit with SCE's actual retirement experience. This is confirmed by the mathematical fitting process, as SCE's proposed curve results in an SSD of 0.4946, while TURN's results in an SSD of 0.0711.¹⁶²³ The Commission should adopt the R2.5-62 curve as proposed by TURN.

¹⁶²¹ Ex. TURN-16 (Garrett), pp. 28-30.

¹⁶²² Ex. TURN-16 (Garrett), pp. 30-32.

¹⁶²³ *Id.*, pp. 32-34,

40.2.2 The Criticisms Raised in SCE's Rebuttal Lack Adequate Evidentiary or Analytical Support.

40.2.2.1 SCE's Defense of its Use of Statistically-Aged Data Misses the Point – Reliance on Recorded Data Should be Preferred over Reliance on Statistically-Aged Data.

This is the first case in which SCE has proposed to use statistically-aged data.¹⁶²⁴ According to the utility, it faced two choices – rely on 20 years of recorded aged data (from 2002 through 2021), or use statistically aged data for retirements prior to 2002.¹⁶²⁵ The difference between the two is that "SCE's recorded data" from 2002 forward includes the recorded vintage years, while that information does not exist on a recorded data for pre-2002 retirements.¹⁶²⁶ SCE argues that its process is not one of manufacturing data, because "[t]he recorded retirements in the statistically aged data are the retirement amounts recorded on SCE's books."¹⁶²⁷ But the retirement amounts for the pre-2002 data do not include "recorded vintage years."¹⁶²⁸ After the statistical aging process, the pre-2002 data DOES include such information, as statistical aging serves to "assign vintage years to data prior to 2002 in order to be included in the study."¹⁶²⁹ The vintage year data for pre-2002 plant did not exist prior to the statistical aging process deployed by SCE's consultant, but did exist after that process.

SCE appears to believe that the Commission should find that the utility's first-time reliance on simulated data here is reasonable because <u>PG&E</u> included such data in its 2017 GRC,

¹⁶²⁴ Ex. SCE-18, Vol. 3, p. 35.

¹⁶²⁵ *Id.*, p. 34.

¹⁶²⁶ Id., p. 36.

¹⁶²⁷ Id.

¹⁶²⁸ Id.

¹⁶²⁹ *Id.*, p. 33.

and TURN raised no objection to the practice then.¹⁶³⁰ The Commission should disregard such arguments. The fact that neither TURN nor any other party addressed the issue of PG&E's reliance on simulated data in an earlier GRC cannot reasonably be interpreted as assent to that method; the unfortunate truth is that in a proceeding of the breadth of a PG&E GRC, there are likely numerous issues that are simply missed by intervenors despite their best efforts.

Besides, one would think the more recent history on the topic is more relevant. SCE's depreciation testimony fails to mention that when the utility's witness earlier appeared as PG&E's depreciation witness in that utility's most recent 2023 GRC, TURN raised extensive objections to the PG&E's reliance on simulated data. The final decision in that GRC did not squarely address the merits of that specific dispute, but it strongly implied skepticism when it cited the concerns TURN had raised about PG&E's pre-1999 data (the period for which PG&E relied on statistical aging) and noted with favor that TURN had used "more recent experience bands" (that is, those bands that did not contain pre-1999 data) in developing its recommendations, which the Commission adopted for all but three of the accounts in dispute.¹⁶³¹

40.2.2.2 Twenty Years of Recorded Data are Sufficient.

SCE argues that its reliance on statistically aged data is necessary to enable reliance on "many more decades of data" rather than the 20 years of historical data available to the

¹⁶³⁰ Id.

¹⁶³¹ D.23-11-069 (PG&E TY 2023 GRC), pp. 669-670. The Commission adopted TURN's recommendation for twelve of the fourteen plant accounts for which TURN addressed the proposed Average Service Life in PG&E's GRC.

utility.¹⁶³² But the Commission had recently found reasonable TURN's application of only the more recent experience bands to fit survival curves in the PG&E GRC, rather than using simulated data through statistical aging in order to create longer experience bands.¹⁶³³ TURN's expert witness here testified that, in his experience, 20 years of retirement experience data is sufficient for actuarial analysis without the need to supplement the data.¹⁶³⁴ SCE's rebuttal testimony has made no attempt to explain why the 22 years of historical data would be sufficient for the development of average service lives for the majority of the accounts in dispute in the PG&E GRC, but 20 years should be found inadequate for SCE GRC purposes. Nor does SCE explain why it deems the \$3.0 billion of T&D asset retirements (representing "millions of assets") experienced in the <u>10</u>- year period from 2012-2021 a "reasonable basis for estimating future net salvage costs,"¹⁶³⁵ but <u>20</u> years is inadequate for development of average service lives in the same depreciation study. The Commission should find that 20 years is indeed sufficient for purposes of setting average service lives in this GRC, consistent with the outcome recently adopted in the test year 2023 PG&E GRC.

¹⁶³² Ex. SCE-18, Vol. 3, p. 33. SCE's rebuttal testimony refers to the data in the "2002-2021 experience band" as reflecting either 19 years or 22 years of data. *Id.*, pp. 34 (19 years) and 33 and 37 (22 years). The 19-year figure is a miscalculation – the period 2002-2021 encompasses 20 years, consistent with how PG&E's rebuttal testimony in its test year 2023 GRC calculated the 1999-2020 experience band as representing 22 years of data. The 22 years figure as it appears in the SCE GRC rebuttal appears to be a failure on the part of SCE's witness to make that correction when he cut-and-paste material from his PG&E GRC rebuttal for purposes here. *See, for example,* Ex. SCE-18, Vol. 3, p. 37, lines 1-19, and Ex. TURN-105 (Excerpt from PG&E Rebuttal Testimony from test year 2023 GRC), pp. 12-74, 1. 22 to 12-75, 1. 22.

¹⁶³³ D.23-11-069 (PG&E TY 2023 GRC), p. 669.

¹⁶³⁴ Ex. TURN-16, p. 16.

¹⁶³⁵ Ex. SCE-07, Vol. 3, p. 19.

40.2.2.3 The Role of Gradualism in Setting ASLs

SCE's rebuttal testimony argued that TURN had failed to apply the concept of "gradualism" to its service lives recommendations, whereas the utility had limited its proposed average service lives "more than five years in either direction."¹⁶³⁶ This portion of SCE's argument is, largely a "cut-and-paste" from PG&E's rebuttal testimony in that utility's test year 2023 GRC, where PG&E had the same witness addressing its service life proposals.¹⁶³⁷ If SCE saw fit to repeat here its argument from the prior GRC, the utility should also have indicated how the Commission resolved the disputed issue there. As noted earlier, in the 2023 GRC, the Commission adopted TURN's proposed average service life figures for twelve of fourteen accounts. And it did so even as it "confirm[ed] its interest in maintaining a gradual approach to changes in depreciation, which must be driven by specific aging analyses."¹⁶³⁸ The 2023 GRC decision includes a table comparing the then-current curve and the TURN-proposed curve the Commission adopted as reasonable.¹⁶³⁹ For the eight electric plant accounts, half reflect increased lives of more than 5 years, and seven reflect increased lives of 9-16% when calculated as a percentage of the then-current life. The Commission should find SCE's argument as lacking in support as it did in the PG&E GRC, and instead find again that TURN's proposed service lives are consistent with maintaining a gradual approach to changes in depreciation.

¹⁶³⁶ Ex. SCE-18, Vol. 3, pp. 29-30.

¹⁶³⁷ Ex. TURN-105 (Excerpt from PG&E Rebuttal Testimony from test year 2023 GRC), pp. 12-25 to 12-27.
¹⁶³⁸ D.23-11-069 (PG&E TY 2023 GRC), pp. 669-670 [citations omitted].
¹⁶³⁹ *Id.*, p. 670, Table 10-C.

40.2.2.4 SCE's General References to Electrification Policies Are Not Compelling.

SCE's direct testimony in support of its depreciation recommendations describes as an emerging issue the impact that California's "Net Zero by 2045" goal will have on the appropriate service lives for utility plant. SCE's witness described how the focus to date has been on how the natural gas industry and power generation will be impacted, but stated the expectation that there will be impacts on the electric industry as well. While he cited only impacts that would result service lives being shorter in the future than they have been in the past, "all else equal," SCE made no specific recommendations to shorten any service life due to these factors.¹⁶⁴⁰ The Commission should recognize this element of SCE's depreciation showing as a marginally revised version of the analysis the same witness sponsored in PG&E's Test Year 2023 GRC when that utility sought to adopt a "Units of Production" method of depreciation to replace the straight-line method for gas distribution plant accounts. In D.23-11-069, the Commission determined the approach focused on gas assets was not "a solution ready for adoption in this GRC," and noted a series of "fundamental questions to consider before deciding upon its implementation."¹⁶⁴¹

The Commission should reach a similar conclusion here, and reject SCE's suggestion that the potential impacts of achieving the Net Zero by 2045 goal warrants adopting lower average service lives for electric plant at this time. This is particularly appropriate where the utility has again sought to re-purpose the rebuttal testimony its witness sponsored in the PG&E GRC as rebuttal testimony here. For example, SCE criticizes TURN's testimony because the "proposals

¹⁶⁴⁰ Ex. SCE-07, Vol. 3, pp 73-74.

¹⁶⁴¹ D.23-11-069, pp. 658-661.

are instead primarily, if not entirely, based on the experience of the Company's assets in the past and do not give sufficient consideration of how the electric and gas industries of the future will be different from the past."¹⁶⁴² This is word-for-word the same critique the same witness raised in rebuttal testimony submitted on behalf of PG&E.¹⁶⁴³ In light of the similarity of the underlying issues and the utility arguments regarding those issues, the Commission should not hesitate to reach an outcome identical to that adopted in the recent PG&E GRC, and adopt TURN's proposed service lives.

40.3 Small Hydro Decommissioning

SCE proposes small hydro decommissioning accruals based on a probability-weighted calculation that assigns each facility a likelihood of being decommissioned, a future year in which the decommissioning would commence, and an estimated decommissioning cost.¹⁶⁴⁴ SCE then escalates the probability-adjusted decommissioning cost estimate to the year's dollars in which the decommissioning is assumed to begin and determines annual accruals beginning in 2025.¹⁶⁴⁵ Under this approach, SCE would accrue \$52.8 million in 2025 and beyond.¹⁶⁴⁶

TURN recommends that the Commission deny decommissioning accruals for any facility that has less than a 90% of being decommissioned, an approach that would limit accruals to three specific facilities that have a 90-100% chance of commencing decommissioning during the GRC

¹⁶⁴² Ex. SCE-18, Vol. 3, p. 16.

¹⁶⁴³ Ex. TURN-105 (Excerpt from PG&E Rebuttal Testimony from test year 2023 GRC), p. 12-13. And while SCE has edited the portion of rebuttal that appears under the heading "Forces Impacting the Service Lives of the Companies [*sic*] Assets," the argument is very similar to that included in PG&E's rebuttal under the heading "Factors Influencing PG&E's Future Service Lives." *Compare*, Ex. SCE-18, Vol. 3, pp. 18-23, with Ex. TURN-105, pp. 12-67 to 12-69.

¹⁶⁴⁴ Ex. SCE-07v3, pp.86-88.
¹⁶⁴⁵ Ex. TURN-13-E, p.111.
¹⁶⁴⁶ Ex. SCE-07v3, p.86.

cycle (San Gorgonio, Borel, Rush Creek/Agnew + Rush Meadows). This approach would result in annual accruals of \$30.8 million, or a \$22 million reduction relative to SCE's proposal.¹⁶⁴⁷

In the 2021 GRC, SCE made an identical proposal that was rejected by the Commission. After reviewing the arguments offered by SCE in favor of beginning accruals for facilities assigned 50% and 10% probabilities of decommissioning, the Decision found that, "given the degree of uncertainty regarding when SCE may initiate decommissioning of these plants, the Commission finds that SCE does not present sufficient justification to begin recovery of decommissioning costs for these plants at this time."¹⁶⁴⁸ The final Decision limits decommissioning accruals to facilities with at least a 90 percent probability of being decommissioned, the same position TURN takes in this GRC.

In this GRC, SCE repeats the same arguments rejected by the Commission in the last GRC. For example, SCE explains that, for the three facilities assigned a 50% probability, the economic analysis of whether to continue operation "does not point strongly in either direction."¹⁶⁴⁹ For facilities assigned a 10% probability, SCE "generally anticipates that relicensing will be economically preferable to decommissioning."¹⁶⁵⁰ These characterizations are, almost word for word, identical to the rationales provided in the last GRC and rejected in D.21-08-036.¹⁶⁵¹ The only new argument provided in this GRC is the unremarkable claim that all

 ¹⁶⁴⁷ Ex. TURN-13-E, p.113, Table 35. TURN's calculation includes the impact of several recommendations: (1) delaying assumed decommissioning for Borel, (2) using \$2028 instead of nominal dollars, and (3) excluding projects that have less than a 90% probability of occurrence.
 ¹⁶⁴⁸ D.21-08-036, p.525.

¹⁶⁴⁹ Ex. SCE-05v1, p.142.

¹⁶⁵⁰ Ex. SCE-05v1, p.143.

¹⁶⁵¹ D.21-08-036, p.525 ("With regard to the plants assigned a 50 percent probability, SCE explains that the financial and economic analyses of the costs to decommission versus the costs to continue operations do not point strongly in either direction. With regard to the plants assigned a 10 percent probability, 'SCE generally anticipates that relicensing will be economically preferable to decommissioning.")

small hydro assets "will at some point reach the end of their respective useful lives and require retirement."1652

The Commission should take note of the fact that several of the facilities included in SCE's probability matrix for decommissioning are currently undergoing federal license renewal. For example, Kaweah 3 has been assigned a 50% probability of beginning decommissioning in 2026 and Kaweah 1-2 has been assigned a 10% probability of beginning decommissioning in 2026.¹⁶⁵³ These probabilities are puzzling given SCE's insistence on seeking O&M costs driven by the relicensing of Kaweah expected to be completed in 2025.¹⁶⁵⁴ It is not reasonable for the Commission to approve O&M costs based on the assumption that the plant will be relicensed in 2025 and decommissioning accruals based on the assumption that the plant will be permanently retired in 2026. Similarly, SCE assigns a 10% probability of retirement to Bishop Creek in 2024 despite the fact that the plant is still operating in 2024 and is expected to receive a renewed federal license in 2026.1655

In rebuttal testimony, SCE expresses concern that failure to begin accruals for low probability plants in this GRC could require larger accruals in the 2029 GRC cycle (and cause "rate shock") if the economics of operating these plants becomes unfavorable.¹⁶⁵⁶ This claim is misleading and overblown. Using SCE's own accrual estimates, only \$19.2 million (or 31%) out of \$62.1 million in annual collections sought in this GRC involve projects with a decommissioning probability of 50% or lower.¹⁶⁵⁷ Moreover, one facility (Rush Creek Gem)

¹⁶⁵² Ex. SCE-05v1, p.143.

¹⁶⁵³ Ex. SCE-05v1, p.141, Table II-26.

¹⁶⁵⁴ Ex. TURN-13-E, p.28; Ex. TURN-13-Atch1, SCE response to TURN Data Request 52, Q2a; Ex. SCE-16, p.25 ¹⁶⁵⁵ Ex. SCE-05v1, p.141, Table II-26; Ex. TURN-13-E, p.28.

¹⁶⁵⁶ Ex. SCE-16, p.46.

¹⁶⁵⁷ Ex. TURN-13-E, p.110, Table V-30.

assigned a 50% probability of decommissioning in 2030 accounts for \$8.4 million of the \$19.2 million associated with lower probability facilities.¹⁶⁵⁸ SCE also expects that facility to receive a new federal license in 2029 that would enable many years of future operation.¹⁶⁵⁹

Missing from SCE's analysis is any consideration of the impacts on current ratepayers who are facing a crisis of affordability due to high electricity rates. Given the highly speculative nature of future decommissioning dates and probabilities, the Commission should favor proposals that minimize the burdens placed on current ratepayers. Adopting TURN's proposal to limit accruals to projects with at least a 90% probability of decommissioning, which mirrors the outcome adopted in D.21-08-036, strikes the correct balance.

40.4 Generation Decommissioning Escalation

SCE's proposal for accruing decommissioning costs for all generation units uses nominal dollars rather than constant dollars for the remaining years of plant life prior to decommissioning.¹⁶⁶⁰ This proposal was rejected by the Commission in the prior two GRCs in favor of an approach that calculates future decommissioning expense in constant dollars tied to the final year of the GRC cycle.¹⁶⁶¹ TURN's proposal in the current GRC mirrors the approach adopted by the Commission in these prior two cases by calculating accruals in this cycle based on constant \$2028.¹⁶⁶²

Although this issue has been litigated in two successive GRCs, SCE presses the Commission to reconsider its precedents by offering the same arguments that were used in the

¹⁶⁵⁸ Ex. TURN-13-E, p.110, Table V-30.

¹⁶⁵⁹ Ex. TURN-13-E, p.28.

¹⁶⁶⁰ Ex. SCE-18v3, pp.55-58.

¹⁶⁶¹ D.19-05-020, pp.324-325; D.21-08-036, pp.525-528.

¹⁶⁶² Ex. TURN-13-E, pp.111, 115.

prior cases. The primary argument relied upon by SCE is the claim that TURN's proposal "results in much lower accruals early in the asset's life that are made up for with much higher accruals at the end of the asset's life."¹⁶⁶³ Additionally, SCE argues that TURN's approach would result in accrual escalation growing "at many times the rate of inflation" in the final years prior to asset retirement.¹⁶⁶⁴ The Commission previously addressed these arguments.

In D.21-08-036, the Commission explained that the use of inflation-adjusted accruals "appropriately accounts for the time value of money and avoids the result of current ratepayers paying on a vastly overinflated expense."¹⁶⁶⁵ By SCE's own admission, its preferred approach would result in front-loaded customer contributions that decline (in real dollars) throughout the collection period.¹⁶⁶⁶ TURN's alternative is designed to adjust the burden on customers over time and reduce the real dollar total obligations to ratepayers over the entire period. In D.21-08-036, the Commission explained that, due to differences in real dollar impacts, "SCE incorrectly assumes that the total amount to be collected over a 20-year period under TURN's method would be the same as under the straight-line method."¹⁶⁶⁷ SCE's comparison of proposals in this case ignores the fact that different methods of collecting the same amount of nominal dollars over an extended period do not result in the same real dollar impacts on customers.¹⁶⁶⁸ Conversely,

¹⁶⁶³ Ex. SCE-18v3, p.58.

¹⁶⁶⁴ Ex. SCE-18v3, p.57.

¹⁶⁶⁵ D.21-08-036, page 526.

¹⁶⁶⁶ Ex. TURN-712, SCE response to TURN Data Request 117, Q62(e).

¹⁶⁶⁷ D.21-08-036, p.527.

¹⁶⁶⁸ Ex. SCE-18v3, p.57. Table V-10 claims to show that the same amount of nominal dollars would be collected under the SCE and TURN proposals but does not adjust the annual contributions to constant dollars to reflect the time value of money. This failure renders the comparison flawed. Applying a 7% discount rate to the annual values on that table, which approximates the utility cost of capital, results in a 7.8% reduction in real dollar obligations between 2022-2040 under the TURN approach. The use of a discount rate that exceeds inflation appropriately reflects the increased value that residential customers place on current year dollars.

SCE's approach would allow SCE to overcollect (in real dollars) and retain the time value benefits for its shareholders.

SCE's rebuttal testimony points out that TURN's proposal to use 2028 dollars for calculation of the accrual would result in an accrual for SCE's hydro that is \$2 million higher than proposed by SCE.¹⁶⁶⁹ This result is not surprising since SCE's proposed revenue requirement assumed that the San Gorgonio project would be fully decommissioned by the end of 2025¹⁶⁷⁰ and that Bishop Creek 2-6 would start decommissioning in 2024 and complete it in 2027. Both assumptions are flawed and result in an understatement of the decommissioning accrual for small hydro using SCE's nominal dollar approach.

SCE admits that decommissioning efforts for San Gorgonio have not yet commenced.¹⁶⁷¹ During cross examination, SCE witness Allen noted that major storms in 2023 and early 2024 left some areas inaccessible and, as of May 2024, prevented the development of damage assessments that could inform expectations of what projects could be completed in the coming years.¹⁶⁷² It is therefore unreasonable to assume that SCE could complete the decommissioning of San Gorgonio on the schedule assumed by SCE when it calculated its decommissioning accrual for that plant. Delay would result in increased nominal dollar costs for the decommissioning of San Gorgonio, which would increase SCE's assumed accrual for hydro.

SCE's assumption that Bishop Creek would be decommissioned by 2027 is also unreasonable since Bishop Creek is still operating in 2024 and is expected to receive a renewed federal license in 2026.¹⁶⁷³

¹⁶⁶⁹ Ex. SCE-18v3, Table V-9, p.55.

¹⁶⁷⁰ Ex. SCE-07v3WP, p.111.

¹⁶⁷¹ Ex. SCE-16, p.40.

¹⁶⁷² Transcript, May 15, pp.1039-1040.

¹⁶⁷³ Ex. SCE-05v1, p.141, Table II-26; Ex. TURN-13-E, p.28.

When more reasonable assumptions are used for the decommissioning dates for these two plants, the difference between TURN and SCE's accrual for small hydro becomes *de minimus*.

TURN urges the Commission to affirm the treatment adopted in the prior two GRCs and require that decommissioning accruals be collected using constant dollars. This approach correctly balances the interests of current and future ratepayers and prevents overcollections (in real dollars) that benefit utility shareholders.

40.5 Solar PY

TURN addresses decommissioning cost issues relating to the SPVP projects in Section 24.4.2.3 since this topic was included in the main body of SCE's generation rebuttal testimony (Ex. SCE-16).

40.6 Fuel Cell Generation

TURN addresses cost recovery issues relating to the Fuel Cell projects in Section 24.3 since this topic was primarily included in the main body of SCE's generation rebuttal testimony (Ex. SCE-16).

40.7 Miscellaneous/Other

41. POST TEST YEAR RATEMAKING

SCE and TURN both propose a two-part post-test year ratemaking (or "attrition") mechanism with separate annual revenue requirement adjustments for expense and capital during the 2026-2028 post-test years. The disputes between TURN and SCE boil down to four issues: (1) the index to use for escalating authorized test year O&M expense; (2) whether to use a trended historical average of capital additions for capital attrition or to escalate test year capital additions for the "basic" capital attrition mechanism; (3) how to calculate test year capital additions for purposes of determining attrition year capital additions; and (4) whether to exclude any cost categories from the general capital attrition mechanism, beyond wildfire mitigation, and instead to authorize post-test year budgets for those activities.

SCE proposes to escalate O&M expense using S&P Global Market Intelligence's (formerly IHS Markit's and before that, Global Insight's) proprietary utility-specific indices. TURN would escalate all O&M expense based on general inflation using CPI-U. SCE seeks to escalate test year capital additions using S&P Global Market Intelligence indices to determine basic capital attrition, while TURN would use a trended 7-year average of recorded capital additions for 2016-2022, excluding recorded additions for wildfire mitigation. SCE would modify the calculation of test year capital additions (that would then be escalated in its basic capital attrition mechanism) by adding an adjustment for the lag in some capital expenditures closing to plant, while TURN opposes this adjustment. And SCE would add attrition year budgets for four specific capital projects and wildfire mitigation, while TURN proposes budget-based attrition only for wildfire mitigation capital (specifically, Targeted Undergrounding and Covered Conductor).¹⁶⁷⁴

For the reasons explained below, the Commission should find that TURN's two-part attrition mechanism meets the objectives of attrition and more reasonably balances the interests of ratepayers and shareholders during the post-test year period than SCE's proposal.

¹⁶⁷⁴ See, e.g., Ex. SCE-07V04, pp. 26 (overall mechanism), 34 (adjusted calculation of test year capital additions); Ex. TURN-17 (Yap), pp. 1-2.

41.1 Attrition Is Intended to Provide Some Reasonable Relief to Shareholders Between General Rate Cases, Not Guarantee Company Earnings.

Attrition is a mechanism that the Commission has used to offset the financial risk experienced by the utilities between general rate cases.¹⁶⁷⁵ The Commission adopted the original attrition mechanism in 1980 during a period of very high inflation.¹⁶⁷⁶ By 1996, the Commission still characterized attrition mechanisms as an exception to the general strategy of examining one test year out of every three years and providing the utility an incentive to improve its productivity; the Commission only allowed attrition adjustments in years when inflation was high.¹⁶⁷⁷ Since that time, the Commission has continued to affirm its discretion to grant or deny attrition requests, and has maintained that the utilities are not automatically entitled to an attrition mechanism between rate cases.¹⁶⁷⁸

Moreover, the Commission has made it quite clear over the years that attrition does not

provide a guarantee of earnings but rather a reasonable offset to increasing costs:

Attrition is the year-to-year decline in a utility's earnings caused by increased costs that are not offset by increased rates or sales. In order to protect utility-shareholders from the effects of attrition to some extent, the Commission has-adopted a ratemaking mechanism called the Attrition Rate Adjustment (ARA). The ARA mechanism was designed to 'provide utilities with the reasonable opportunity of achieving their authorized rates of return during years in which they are not permitted under the Commission's rate case plan procedures to file for general rate relief but in which they still face volatile economic conditions.'¹⁶⁷⁹

Similarly, attrition allowances "are not intended to insulate utilities from economic pressures that

¹⁶⁷⁵ D.92497, Cal. PUC LEXIS 1024; 4 CPUC2d 725 (December 5, 1980) at *101.

¹⁶⁷⁶ Id.

¹⁶⁷⁷ D.00-02-046, p. 428 (discussing Re Pacific Gas and Electric Company (1996) 69 CPUC2d 691, 695). ¹⁶⁷⁸ See, e.g., D.21-08-036, p. 546.

¹⁶⁷⁹ D.04-05-055, p. 26 (citing D.85-12-076, Finding of Fact 1, 9 CPUC 2d 453,476).

all businesses experience."¹⁶⁸⁰ Unlike a test year cost of service analysis, an attrition mechanism serves "merely to mitigate economic volatility between test years to a reasonable degree so that a well-managed utility can provide safe and reliable service while maintaining financial integrity."¹⁶⁸¹

In a number of cases the Commission has specifically found that a utility's attrition proposal placed too great a burden on ratepayers and significantly reduced the authorized attrition amount.¹⁶⁸² While TURN is not proposing that the Commission deny attrition relief to SCE in the pending case, nonetheless the utility's need for increased funds to offset rising costs should be balanced against the burden that higher rates place on ratepayers.

41.1.1 SCE Confuses Attrition with Cost of Service Ratemaking.

SCE claims its proposed attrition mechanism is "based on the principles of cost-ofservice ratemaking" and asserts that adopting anything other than SCE's "cost-of-service PTYR mechanism" would be "inconsistent with SCE providing safe and reliable operations while maintaining financial integrity."¹⁶⁸³ Yet the Commission has stated previously that the attrition rate adjustment "is not intended to replicate a test year analysis, or to cover all potential cost changes so as to guarantee….rate of return."¹⁶⁸⁴

¹⁶⁸⁰ D.04-07-022, p. 270.

¹⁶⁸¹ D.14-08-032, pp. 652-653. *See also* D.20-01-002, p. 41 (affirming this language and accordingly rejecting SCE's request for assurance that under the new four-year rate case plan adopted in that decision, utilities would receive an attrition mechanism that "fully compensates the utility for its costs of service in the attrition years").

¹⁶⁸² See, e.g., D.09-03-025, pp. 305-306 and D.13-05-010, pp. 1009-1010.

¹⁶⁸³ Ex. SCE-18V04, pp. 2-3.

¹⁶⁸⁴ D.14-08-032, p. 652.

SCE asserts that it would be "unreasonable for the Commission to adopt a post-test year mechanism that does not allow the utility to continue operations" at the standard established by the Commission.¹⁶⁸⁵ SCE then presents a table illustrating the extent to which TURN's (and Cal Advocates') attrition proposals fall short of SCE's requests in terms of attrition year revenue requirements, O&M escalation, and capital additions.¹⁶⁸⁶ SCE warns of the consequences to SCE and its customers if SCE's attrition proposals are not adopted.¹⁶⁸⁷

It is important to recognize that SCE's table compares TURN's proposals to SCE's *requested* attrition adjustments, not SCE's post-test year *spending needs*. SCE's overall post-test year spending needs have not been vetted in this proceeding. On the other hand, SCE's test year needs have been vetted, and intervenors have recommended adjustments to many of the utility's O&M and capital proposals. TURN cannot recall a litigated GRC decision in which the Commission authorized the utility to proceed with 100% of its proposed O&M and capital investments over the rate case period. The Commission should not presume that the utility's post-test year O&M and capital investment plans are reasonable for the purpose of evaluating attrition mechanism proposals.

Moreover, while the Commission may establish performance standards and determine reasonable GRC revenue requirements, the Commission leaves it to SCE to determine how best to operate within the established parameters, subject to spending accountability oversight. SCE routinely claims it cannot operate safely and reliably without its proposed test year (or attrition year) revenue requirement, which reflects SCE's belief in its own claims about the need for

¹⁶⁸⁵ Ex. SCE-18V04, p. 3.

¹⁶⁸⁶ Ex. SCE-18V04, p. 4.

¹⁶⁸⁷ Ex. SCE-18V04, p. 5, Table I-1.

greater and greater amounts of revenue requirement. However, the Commission has previously granted SCE less revenue requirement than it had requested¹⁶⁸⁸ and has even reduced SCE's test year revenue requirement,¹⁶⁸⁹ despite SCE's claims that it could not operate safely and reliably at lower revenue requirements. SCE has nonetheless been able to operate successfully. GRC cost of service ratemaking does not prescribe a particular revenue requirement level in a specific posttest year period, but rather dictates a level of increase that balances concerns about the utility's financial health against other important factors, such as the need to encourage the utility to stretch into greater productivity and the need to consider whether ratepayers can absorb the corresponding rate increases.¹⁶⁹⁰

Finally, SCE points specifically to TURN's proposed attrition year budgets for wildfire mitigation capital, warning that TURN's proposal "is barely one-half of the amounts that SCE has proposed."¹⁶⁹¹ SCE states that TURN's attrition recommendations "would cause SCE to contemplate reductions in service to customers and significant reductions in system improvements and wildfire mitigation activities."¹⁶⁹² This argument is a red herring. TURN supports SCE's proposal to establish attrition year capital budgets for wildfire mitigation, as discussed below. To the extent the Commission finds TURN's proposed budgets more reasonable than SCE's, the "shortfall" would not be the fault of TURN's proposed attrition

¹⁶⁸⁸ See, e.g., D.09-03-025, pp. 5, 6.

¹⁶⁸⁹ In SCE's 2018 GRC, the Commission adopted a test year revenue requirement that was 7.53% (\$417 million) lower than SCE's request, where SCE had already requested a \$22 million revenue requirement *decrease*, and attrition increases that were approximately \$100 million lower than requested by SCE for each attrition year. D.19-05-020, pp. 2-3.

¹⁶⁹⁰ See e.g., D.14-08-032, p. 652 ("[W]e seek to promote [the utility's] incentive to stretch to achieve productivity between test years.").

¹⁶⁹¹ Ex. SCE-18V04, p. 3.

¹⁶⁹² Ex. SCE-18V04, p. 4.

mechanism. Rather, the shortfall would reflect the Commission's determination of the appropriate level of spending on wildfire mitigation in this GRC cycle.

41.2 O&M Attrition Should Be Based on General Inflation (CPI-U).

As noted above, SCE proposes to escalate authorized Test Year O&M expense in each year of the post-test year period on a highly tailored basis using S&P Global Market Intelligence's proprietary utility-specific indices.¹⁶⁹³ This is the same approach to forecasting escalation that SCE proposes for establishing test year expense levels.¹⁶⁹⁴

While SCE's utility-specific indices may be appropriate for projecting test-year expenses, which the Commission now establishes every four years, the complex and utility-specific indices are not the best choice for the attrition period. These complex indices simply pass along the costs of business-as-usual activities during the attrition period, providing little incentive for SCE to keep its costs down. Indeed, SCE acknowledges that its proposals are "based on the actual impact of inflation on the costs of providing utility service in the attrition years."¹⁶⁹⁵

However, as the Commission most recently recounted in D.20-01-002, an attrition mechanism

is not intended to replicate a test year analysis, or to cover all potential cost changes so as to guarantee [the utility's] rate of return [during the attrition years]," but "is merely to mitigate economic volatility between test years to a reasonable degree so that a well-managed utility can provide safe and reliable service while maintaining financial integrity."¹⁶⁹⁶

¹⁶⁹³ Ex. SCE-07V04, p. 26.

¹⁶⁹⁴ *Id*.

¹⁶⁹⁵ Ex. SCE-18V04, p. 1.

¹⁶⁹⁶ D.20-01-002, p. 41 (quoting TURN's comments which quoted and cited D.14-08-032, pp. 652-653).
Similarly in SCE's 2015 GRC decision, the Commission authorized an attrition mechanism to, in pertinent part, "give SCE an opportunity to offset *some* inflationary price increases," not to make SCE indifferent to inflation.¹⁶⁹⁷ The Commission similarly explained in D.14-08-032, "[W]e seek to promote [the utility's] incentive to stretch to achieve productivity between test years."¹⁶⁹⁸ Thus, SCE's proposal to incorporate each of the complex and proprietary indices used to establish test year expense levels is not the best choice to increase expense levels for the attrition period.

For the reasons provided below, the Commission should reject the utility's request and instead incorporate broad indices for all O&M escalation in the attrition years to provide the proper incentives to utility management to manage costs.

41.2.1 Using a Broad Index for O&M Escalation Reasonably Addresses Inflation-Related Cost Pressures While Providing the Utility an Incentive to Manage its Operations as Efficiently as Possible.

In D.04-05-055, the Commission summarized the attributes of the historical attrition

mechanism:

The traditional attrition mechanism provides for an advice letter filing, just prior to the attrition year, by the utility seeking increased rates based on the escalation of adopted TY GRC expense and rate base. A seven-year average of plant additions is used to account for rate base growth during the attrition period. The escalation rates are conventional indices such as the U.S. Department of Labor, Bureau of Labor Statistics' CPI, and DRI.¹⁶⁹⁹

¹⁶⁹⁷ D.15-11-021, p. 389.

¹⁶⁹⁸ D.14-08-032, p 652.

¹⁶⁹⁹ D.04-05-055, p. 27.

In this GRC, TURN recommends that the Commission apply the traditional attrition mechanism and base O&M escalation on a broad inflation index.¹⁷⁰⁰ TURN specifically proposes to escalate SCE's authorized test year 2025 O&M expenses by CPI-U to determine the appropriate amount for O&M expenses in attrition year (AY) 2026. SCE's AY 2026 O&M expense levels would similarly be escalated by CPI-U to determine AY 2027 O&M levels, with the same approach used to escalate AY 2027 O&M levels for AY 2028.

The choice of how to estimate inflationary effects for purposes of utility ratemaking is a policy matter. TURN's position is that use of S&P Global Market Intelligence's indices is generally not objectionable for setting test year cost of service revenue requirements (exceptions may apply). On the other hand, attrition year adjustments are not updates to cost of service; such updates occur with the next GRC. The Commission must consider whether SCE's proposed attrition mechanism, with its myriad of account specific escalators, best serves the Commission's purpose of promoting SCE to "stretch to achieve productivity" in attrition years 2026-2028. TURN submits that the PTY escalation index that SCE requests using, S&P Global Market Intelligence's various utility cost forecasts, is simply too protective of SCE to properly incent it to manage operations productively before its next GRC. Use of a broad index like CPI as a measure of inflation during the PTYs, rather than an index that more precisely tracks the escalation in utility costs (like S&P Global Market Intelligence), is a reasonable method of achieving the Commission's purpose in providing attrition adjustments.

TURN understands that the Commission has, on a number of recent occasions, relied on forecasts of escalation that are specific to utility sector costs for purposes of attrition instead of

¹⁷⁰⁰ Ex. TURN-17 (Yap), p. 14.

the CPI-U, while expressing its concern that CPI-U "does not specifically cover the prices of the typical goods [the utility] purchases."¹⁷⁰¹ The Commission reached this conclusion in many recent SCE GRCs.¹⁷⁰² However, such an approach is not required to fulfill the purpose of attrition. As noted above, an attrition mechanism "is not intended to replicate a test year analysis, or to cover all potential cost changes so as to guarantee [the utility's] rate of return [during the attrition years]," but "is merely to mitigate economic volatility between test years to a reasonable degree so that a well-managed utility can provide safe and reliable service while maintaining financial integrity."¹⁷⁰³

The Commission should not shy away from recognizing the utility's own responsibility to manage costs and improve efficiencies during the rate case cycle. Adopting an attrition approach that mimics test year cost of service ratemaking undermines the utility's incentive to do so. The Commission made this clear in its decision in the Sempra Utilities' Test Year 2012 GRC, D.13-05-010:

Having reviewed all of the testimony and arguments of the parties concerning the PTY proposals, we hesitate to adopt the proposal of SDG&E and SoCalGas to adopt their PTY ratemaking mechanisms. Their proposed mechanisms seek to include the use of two formulas which lean in their favor. These are the use of Global Insight's utility-specific cost index, and a California-specific health care cost index. Although these utility-specific indexes may be a better reflection of the PTY costs in a "business as usual" setting, such indexes, if adopted, will not provide the Applicants with an incentive to manage and reduce their costs during the PTY period.¹⁷⁰⁴

1704 D.13-05-010, p. 1008.

¹⁷⁰¹ See, e.g., D.04-07-022, p. 278.

¹⁷⁰² D.21-08-036, p. 547.

¹⁷⁰³ D.14-08-032, pp. 652-653. *See also* D.20-01-002, p. 41 (affirming this language and accordingly rejecting SCE's request for assurance that under the new four-year rate case plan adopted in that decision, utilities would receive an attrition mechanism that "fully compensates the utility for its costs of service in the attrition years").

The Commission further clarified that use of Global Insight's utility specific cost index in setting the test year revenue requirement "does not mean we should automatically use those same indexes for the PTY period."¹⁷⁰⁵ The Commission accordingly chose to replace the utility specific indices proposed by SoCalGas and SDG&E with an index based on CPI-U plus 75 basis points.¹⁷⁰⁶

Here, too, the Commission should weigh the importance of proper incentives for the utility during the PTY period more highly than the need to compensate the company exactly for any change in its expenses between test years.

Moreover, it is appropriate for the Commission to adopt an attrition mechanism that protects ratepayers from the "cumulative adverse burdens on customers of absorbing such large attrition increases on top of significant test year increases," as the Commission recognized in D.14-08-032 (quoted above).¹⁷⁰⁷ In SCE's 2018 GRC decision, the Commission concluded that "limiting the annual increase in SCE's revenue requirements during this GRC period to" the rate of growth in customer income is necessary to "begin to strive for greater affordability."¹⁷⁰⁸ Given that CPI-U is the index used by the Social Security Administration to adjust retirement benefits and Supplemental Security Income payment amounts each year, its use in escalating O&M expense in the post test years supports the Commission's intention to strive for greater affordability.¹⁷⁰⁹

¹⁷⁰⁵ D.13-05-010, p. 1009.

¹⁷⁰⁶ D.13-05-010, p. 1010.

¹⁷⁰⁷ D.14-08-032, p. 657.

¹⁷⁰⁸ D.19-05-020, p. 20.

¹⁷⁰⁹ Ex. TURN-17 (Yap), p. 15.

For all of these reasons, the Commission should conclude that the use of CPI-U to escalate O&M expense in the attrition years best balances the interests of shareholders and ratepayers in this GRC period.

41.3 Capital Attrition Should Be Based on Seven Years of Recorded Capital Additions With an Exception for Wildfire Mitigation.

TURN recommends that the Commission apply the time-tested "traditional" approach to capital attrition in this GRC for non-wildfire mitigation capital, despite that the Commission has not adopted this approach for SCE in the past 20 years.¹⁷¹⁰ That traditional approach involves trending a seven-year average of recorded plant additions.¹⁷¹¹ TURN excluded recorded additions for wildfire mitigation in calculating the seven-year average, as TURN recommends budget-based attrition for wildfire mitigation.¹⁷¹²

The Commission has adopted a modified version of this traditional capital attrition

mechanism for PG&E, SDG&E, and SoCalGas in recent years. In the PG&E 2014 GRC, the

Commission adopted a seven-year average combining recorded data with forecasted data (2008-

2014). The Commission highlighted the value of incorporating an historical average:

Use of an historical average is consistent with the approach applied in the past, and normalizes actual utility spending variations over time. Without conducting full-scale review of 2015 and 2016 capital spending requirements, reliance on historical averages offers a reasonable outcome.

We find insufficient basis to rely on PG&E's assumption that 100% of test year additions should also constitute the basis for trending each of the attrition years.¹⁷¹³

¹⁷¹⁰ Ex. TURN-17 (Yap), p. 11, Table 1.

¹⁷¹¹ D.04-05-055, p. 27.

¹⁷¹² Ex. TURN-17 (Yap), pp. 15-16.

¹⁷¹³ D.14-08-032, p. 657.

Similarly, in the SDG&E/SoCalGas 2019 GRC, the Commission adopted a seven-year average combining five years of recorded data with two years of forecasted capital additions (2013-2019). The Commission asserted that incorporating historical capital additions provides an important counterbalance to the utility forecast of capital additions.¹⁷¹⁴ The Commission extended that same mechanism for another two years in D.21-05-003 after conducting a detailed review.¹⁷¹⁵ Furthermore, SDG&E and SoCalGas have proposed trended capital additions as the basis for capital attrition in their currently pending 2024 GRC, A.22-05-015 et al.¹⁷¹⁶

While the Commission has most recently adopted trended capital additions combining recorded and forecast data, TURN urges the Commission to reconsider the use of forecasted capital additions in this GRC. Using the higher forecasted test year capital in the averaging formula instead of the lower recorded capital may bias the projected attrition-year capital values upward if the after-the-fact recorded figures do not match the forecast.¹⁷¹⁷ This bias could cut either way, as the authorized test year forecast will still be a forecast. For example, SCE's actual 2023 capital expenditures were \$262 million lower than its 2023 capital forecast in this GRC.¹⁷¹⁸ Using recorded values for all seven years avoids introducing this uncertainty and the associated risk to ratepayers.

¹⁷¹⁴ Ex. TURN-17 (Yap), p. 13; D.19-09-051, p. 709.

¹⁷¹⁵ D.21-05-003, Ordering Paragraph 2.

¹⁷¹⁶ Ex. TURN-17 (Yap), p. 13.

¹⁷¹⁷ See Ex. TURN-17 (Yap), p. 17, Figure 1.

¹⁷¹⁸ Ex. SCE-18V01, p. 114.

41.3.1 SCE's Critiques of TURN's Capital Attrition Mechanism for Non-Wildfire Should Be Dismissed.

First, SCE asserts that TURN "mispresents historical Commission decisions that address the capital portion of the major energy utility attrition mechanisms."¹⁷¹⁹ SCE compares TURN's statement that "the great majority of PTY decisions that adopted a two-part mechanism have determined the capital portion based on trending of recorded capital additions" with the decisions shown in Table 1 in TURN's testimony.¹⁷²⁰ SCE misunderstands TURN's statement.

When one reads the entire sentence from which SCE's excerpt is taken, it becomes clear that TURN was discussing the full history of attrition decisions since 1980 in referring to the "great majority" of decisions, not simply the decisions since 2004 included in Table 1.¹⁷²¹ TURN agrees that the Commission has adopted a capital attrition approach using trended recorded capital additions in 3 of the most recent 12 litigated GRC decisions, while escalating test year capital additions in 5 GRC decisions. Both are contemporary approaches to attrition, while using trended recorded capital additions is also the traditional approach generally applied by the Commission from 1980 until 2004.¹⁷²²

Next, SCE criticizes TURN's non-wildfire capital attrition mechanism because it relies on 2016-2022 recorded additions and thus "bears no relationship to Test Year authorized capital expenditures."¹⁷²³ SCE argues that historical data is important in GRCs "as it guides the

¹⁷¹⁹ SCE-18V04, pp. 2, 11-12.

¹⁷²⁰ SCE-18V04, p. 11.

¹⁷²¹ Ex. TURN-17 (Yap), p. 11 ("Furthermore, if we consider the full history of the PTY (attrition) mechanism as characterized in the excerpt from D.04-05-055 [which describes the "traditional" attrition mechanism], the great majority of PTY decisions that adopted a two-part mechanism have determined the capital portion based on trending of recorded capital additions.").

¹⁷²² D.04-05-055, pp. 27, 30-32.

¹⁷²³ Ex. SCE-18V04, pp. 2, 12.

development of a Test Year forecast, but the authorized revenue requirements should be established based on reasonable expectations of SCE's future cost of service, not its past."¹⁷²⁴ However, the Commission has made clear that a capital attrition adjustment mechanism need not cover all attrition year capital expenditures – either projected by the utility or actual.

In PG&E's 2014 GRC, PG&E argued that capital attrition based on a 7-year average would be inadequate because PG&E's projected attrition year capital investment levels would be "much higher than a seven-year historic average."¹⁷²⁵ The Commission dismissed PG&E's concern, noting that PG&E's criticisms mistakenly presupposed that its attrition year spending forecasts were reasonable, when those spending plans had not received close scrutiny (unlike test year spending).¹⁷²⁶ The Commission further acknowledged that a 7-year average would "not necessarily cover all attrition year capital expenditures that PG&E may ultimately find prudent."¹⁷²⁷ Even so, the Commission noted, "the Commission historically has not closely covered projected revenue requirement through an attrition mechanism," yet rate base has grown.¹⁷²⁸ The Commission also explained that a 7-year average would provide PG&E "a stronger incentive to find ways to curb the rate of spending growth."¹⁷²⁹ The Commission ultimately concluded that "use of a seven year average better balances the interests of both ratepayers as well as shareholders than does PG&E's methodology."¹⁷³⁰

¹⁷²⁴ Ex. SCE-18V04, p. 12.

¹⁷²⁵ D.14-08-032, p. 657. In that case, PG&E proposed to escalate authorized test year capital additions. (D.14-08-032, p. 656).

¹⁷²⁶ D.14-08-032, pp. 657-658.

¹⁷²⁷ D.14-08-032, pp. 658-659.

¹⁷²⁸ D.14-08-032, pp. 658-659.

¹⁷²⁹ D.14-08-032, p. 657.

¹⁷³⁰ D.14-08-032, p. 657.

41.3.2 Wildfire Capital Attrition Should Be Based on Attrition Year Budgets.

TURN supports SCE's proposal to use a budget-based forecast for wildfire mitigation capital additions during the attrition period. As TURN acknowledged in testimony, it would not be appropriate to simply trend recorded capital addition amounts for wildfire mitigation because the rate of change for wildfire mitigation capital additions is very different now than the rate of change in the previous GRC.¹⁷³¹ TURN witness Yap explained:

As shown previously, the magnitude of those wildfire mitigation capital expenditures are truly staggering. They are too large to be handled on a trended test year capital additions basis. Furthermore, the risk spend ratios for each wildfire mitigation activity need to be taken into account when determining the appropriate priority for the use of ratepayer dollars. Hence, careful consideration of the appropriate mix of wildfire mitigation capital programs as well as the proper level of wildfire mitigation capital expenditures is warranted for each year of the GRC cycle. TURN has provided such an evaluation of the appropriate capital expenditure levels for each year of the GRC cycle in TURN's testimony regarding wildfire capital expenditures [citing Ex. TURN-12 (Borden)].¹⁷³²

SCE takes issue with TURN's proposed budgets for wildfire mitigation, claiming

TURN's proposal "would result in insufficient funds to mitigate wildfire safety risks."¹⁷³³

TURN addresses the appropriate attrition year capital budgets for wildfire mitigation in Section

15 of this brief.

41.3.3 SCE's Other Budget Based Attrition Proposals Should Be Denied.

SCE proposes that four additional specific projects be added on a budget basis to the

trended capital additions and wildfire budgeted capital additions because these specific projects

¹⁷³¹ Ex. TURN-17 (Yap), p. 7.

¹⁷³² Ex. TURN-17 (Yap), p. 7.

¹⁷³³ Ex. SCE-18V04, p. 10.

have "uneven capital additions" over the PTY period.¹⁷³⁴ SCE points to the Commission's special treatment of SoCalGas' Pipeline Safety Enhancement Plan (PSEP) projects in D.19-09-051 as justification for this budget based treatment.¹⁷³⁵ The Commission should reject SCE's proposal for this budget based treatment of these specific projects for several reasons.

First, the bulk of the projects (\$660 million) are scheduled as capital additions in 2028, which is the last year of the GRC cycle, although a few of them (\$225 million) are scheduled for 2027.¹⁷³⁶ SCE is making a long term capital forecast based on information available in 2022 and early 2023 for activities that will occur four to five years later. Yet there is a very real possibility for delays in construction schedules due to problems with permitting or construction activities (unforeseen events, weather, etc.).¹⁷³⁷ In fact, TURN has opposed funding for one of these projects, the T&D Training Center (aka Edison Training Academy), because of persistent delays during the 2018 and 2021 GRC cycles, despite authorized funding in both cases for this project.¹⁷³⁸ These projects could miss the entire PTY period due to delays, but under SCE's proposal, ratepayers would have to pay anyway for the phantom capital additions.

Second, SCE attempts to draw a parallel between these non-wildfire related projects and the PSEP projects that the Commission considered in D.19-09-051, claiming that the Commission was focused somehow on uneven or insufficient capital investment. However, that comparison is completely faulty. If any comparison should be made to the PSEP projects, it

¹⁷³⁴ Ex. SCE-07V04, p. 37.

¹⁷³⁵ Ex. SCE-07V04, p. 37.

¹⁷³⁶ Ex. SCE-07V04, p. 37, Table III-18.

¹⁷³⁷ Ex. TURN-17 (Yap), p. 8.

¹⁷³⁸ Ex. TURN-11 (Defever), pp. 6-8. TURN addresses funding for the T&D Training Center / Edison Training Academy in Section 36 of this brief.

should be made with SCE's wildfire mitigation projects. Each of the proposed PSEP projects were subject to detailed project review.¹⁷³⁹ The flexibility that the Commission adopted for "PSEP capital-related costs not fully reflected in the TY2019 revenue requirement" by permitting them to "be included as part of the PTYs" was directed at a safety program for which the utility was for the first time receiving base rate treatment.¹⁷⁴⁰ Similarly, D.19-09-051 provided a memorandum account for PSEP, which was directed at ensuring that the safety programs were completed in a timely fashion and subject to reasonableness review.¹⁷⁴¹

This is not the same situation that SCE is facing with its proposals to enhance buildings or augment substations. SCE is simply looking for an opportunity to beef up its attrition earnings in the later years of the GRC cycle. The Commission has historically frowned upon budget-based attrition. It rejected SCE's request for budget-based capital attrition in every GRC from the 2006 GRC through the 2018 GRC and permitted only narrow exceptions in the 2021 GRC.¹⁷⁴² The Commission has adopted budget-based attrition in limited circumstances where the costs at issue are truly extraordinary.¹⁷⁴³ The Commission should deny SCE's inappropriate request to augment its capital attrition mechanism with budgeted amounts for non-wildfire mitigation projects.

¹⁷³⁹ D.19-09-051, pp. 198-207, 209-216.

¹⁷⁴⁰ D.19-09-051, pp. 215-216.

¹⁷⁴¹ D.19-09-051, pp. 218-219.

¹⁷⁴² D.19-05-020, pp. 283-285; D.21-08-036, pp. 548-549.

¹⁷⁴³ D.21-08-036, pp. 547-550.

41.4 If the Commission Adopts SCE's Basic Capital Attrition Mechanism, the Commission Should Apply the National Version of Capital Escalation Indices and Reject SCE's Test Year Capital Additions "Lag" Adjustment.

TURN understands that the Commission may conclude that test year capital additions should be escalated to determine attrition year capital revenue requirements, as SCE proposes. In that case, TURN recommends that the Commission reject two elements of SCE proposal.

41.4.1 Test Year Capital Additions Should Be Escalated Using Nationally-Based Rather Than Regionally-Based Capital Indices.

SCE proposes to tie its capital additions levels in the PTY period to an escalation of test year capital additions (except for wildfire and some other selected projects, as noted above). SCE proposes to escalate capital additions with an SCE-specific weighting of regional construction cost indices provided by S&P Global Market Intelligence, rather than the national version of capital indices.¹⁷⁴⁴ This is in contrast to the nationally based indices that SCE proposes to use for O&M escalation.¹⁷⁴⁵

If the Commission is inclined to adopt a capital index based on SCE specific weighting combined with S&P Global Market Intelligence indices, TURN strongly recommends that the Commission incorporate the national rather than regional basis for the capital indices that are applied to the weighting.¹⁷⁴⁶ Using regional indices makes the escalation too tailored to SCE's behavior. SCE is a large enough utility to influence the level of the regional indices.¹⁷⁴⁷

Comparing the utility to itself or to similarly situated businesses reduces the pressure on the utility to stretch. In fact, if the indices are fashioned too narrowly, the attrition simply

¹⁷⁴⁴ Ex. SCE-07V01, p. 114.

¹⁷⁴⁵ Ex. TURN-17 (Yap), p. 4.

¹⁷⁴⁶ Ex. SCE-07V01, p. 110.

¹⁷⁴⁷ Ex. TURN-17 (Yap), p. 4.

becomes a self-fulfilling prophecy where the more the utility expends, the more will be passed along in the future. As the Commission previously explained, "although these utility-specific indexes may be a better reflection of the PTY costs in a 'business as usual' setting, such indexes, if adopted, will not provide the Applicants with an incentive to manage and reduce their costs during the PTY period."¹⁷⁴⁸

The Commission has previously addressed this issue in depth. In designing performance based regulation ("PBR") mechanisms in the 1990s, the Commission reflected on the fact that the use of narrow escalators might reduce the incentive for utility efficiency:

To make this update of utility rates independent of the utility's costs, the price and productivity values should come from national or industry measures and not from the utility itself. The independence of the update rule from the utility's own costs allows PBR regulation to resemble the unregulated market where the firm faces market prices which develop independently of its own cost and productivity. In contrast, traditional regulation often updates rates through a review of the utility's own costs and productivity.¹⁷⁴⁹

TURN submits that incentivizing efficient utility behavior during the PTY period is critical. As discussed in TURN's affordability testimony (Ex. TURN-02), SCE's customers are already struggling to keep up with rapidly increasing electricity bills, and SCE's GRC request will likely further exacerbate those affordability challenges. Thus, if the Commission is inclined to adopt a capital index based on SCE-specific weighting combined with S&P Global Market Intelligence indices, TURN strongly recommends that the Commission incorporate the national version of these indices rather than the regional version when applying them to the SCE-based weighting factors.¹⁷⁵⁰

¹⁷⁴⁸ D.13-05-010 (SDG&E/SoCalGas 2012 GRC), p. 1008.

¹⁷⁴⁹ D.97-07-054, 1997 Cal. PUC LEXIS 751; 179 P.U.R.4th 237 (July 16, 1997) at *30. ¹⁷⁵⁰ Ex. SCE-07V01, p. 110.

41.4.2 SCE's Proposed Adjustment to Mitigate the "Lag" in Capital Expenditures Closing to Plant is Inappropriate.

On top of its request to escalate capital additions from the test year to be adopted as capital additions in during the PTY period, SCE asks to have its capital additions for the test year calculated in a more complex fashion where the lag in some capital expenditures closing to plant is reflected in the capital additions amount. SCE proposes to continue this lagging calculation into the PTY period. The Commission should reject SCE's proposal to calculate capital additions during the PTY period on a more complex basis.

An examination of the history of capital expenditures closing to capital additions demonstrates that SCE's projected 55 percent figure for plant closings in the test year may be somewhat low.¹⁷⁵¹ In any case, SCE's RO model already reflects the lag in capital expenditures when it calculates capital additions for the test year; the test year 2025 capital additions calculated by the model already include construction work in progress carried over from 2024 and earlier years. As TURN witness Yap explained:

Capital additions, which are generally referred to in Commission decisions, are derived from projected construction work in progress ("CWIP") at the end of the previous year, combined with proposed capital expenditures plus cost of retirement, overheads, and allowance for funds used during construction ("AFUDC"). Weighted average capital additions net of retirements are then added to the previous year's end of year plant in service to determine the current year's weighted average plant in service, which is the largest component of rate base.¹⁷⁵²

SCE has used this same approach to develop the test year capital additions for decades.¹⁷⁵³ There is no reason to further adjust post-test year capital additions to account for a lag in capital

¹⁷⁵¹ Ex. TURN-17 (Yap), p. 5.

¹⁷⁵² Ex. TURN-17 (Yap), pp. 3-4 (citing SCE RO Model).

¹⁷⁵³ Ex. TURN-17 (Yap), pp. 6.

expenditures closing to capital additions or to add this "lag" adjustment to escalated test year additions to determine attrition year capital costs.

While SCE points to the "lag" in capital expenditures closing to capital additions to justify its proposed methodology, the real culprit appears to be SCE's proposal to dramatically increase capital expenditures in test year 2025 over the prior year (from \$6.149 billion in 2024 to \$7.393 billion in test year 2025).¹⁷⁵⁴ This is a much larger increase than has been experienced in prior cases. For example, in the previous GRC, SCE proposed to increase the \$5.110 billion in 2020 capital expenditures to \$5.601 billion for the test year 2021.¹⁷⁵⁵ Thus, SCE is worried that the weighted average of capital additions based on 2024's lower capital expenditures and 2025's much high capital expenditures will not be large enough to accommodate SCE's extremely high budget projections that range from \$7.9 to \$8.4 billion during the attrition years.¹⁷⁵⁶ The Commission can eliminate this "problem" by adopting more moderate capital expenditure forecasts for the test year and tempering spending growth during the post-test year period, based on the robust record developed by TURN, Cal Advocates, and other intervenors in this proceeding.

In sum, the Commission should find that TURN's two-part attrition mechanism meets the objectives of attrition and more reasonably balances the interests of ratepayers and shareholders during the post-test year period than SCE's. Nonetheless, if the Commission adopts SCE's approach to capital attrition, the Commission should order the use of national rather than

¹⁷⁵⁴ Ex. TURN-17 (Yap), p. 6.

¹⁷⁵⁵ Ex. TURN-17 (Yap), p. 6.

¹⁷⁵⁶ Ex. TURN-17 (Yap), p. 6.

regional capital escalation indices and should reject SCE's proposed adjustment to test year capital additions to address the "lag".

41.5 The Z-Factor Threshold and Deductible Should Be Increased to Reflect Inflation Since Its Initial Adoption for SCE.

SCE's current post-test year ratemaking (PTYR) mechanism includes a "Z-Factor" mechanism that creates an opportunity for the utility to recovery in rates cost changes associated with exogenous events.¹⁷⁵⁷ The Z-Factor is subject to both a \$10 million "threshold" and a \$10 million deductible that applies on a one-time basis to the first year's revenue requirement associated with any approved Z-Factor.¹⁷⁵⁸

The Commission should increase the \$10 million threshold and deductible figures for Z-Factor purposes to reflect inflation that has occurred since the Z-Factor was first adopted for SCE.¹⁷⁵⁹ TURN submits that whatever incentive or ratepayer protection was created by \$10 million figures in the late 1990s, or even in 2004, when SCE sought to retain the Z-Factor as it shifted from performance-based ratemaking back to cost-of-service ratemaking, is significantly dampened when the same \$10 million figures are calculated in 2025 dollars. TURN proposes an increase of 80% (to \$18 million) to reflect general inflation since 2000, based on data reported by the Federal Reserve Bank of Saint Louis.¹⁷⁶⁰ If the Commission instead saw fit to tie the increase to inflation since SCE first sought to have the Z-Factor mechanism applied to cost-ofservice ratemaking, it could adopt a slightly lower increase to reflect cumulative inflation since

¹⁷⁵⁷ Ex. SCE-07, Vol. 4, pp. 37-38.

¹⁷⁵⁸ D.04-07-022 (SCE Test Year 2023 GRC), pp. 278-279.

¹⁷⁵⁹ Ex. TURN-15-E2, pp. 25-26.

¹⁷⁶⁰ The Federal Reserve Economic Data indicate a cumulative Consumer Price Index increase of approximately 83% for the period from 1/1/2000 through the end of 2023.

2004. The point is that twenty-plus years of inflation may reasonably be expected to dampen any incentive and risk-sharing effect the deductible is intended to serve, and the Commission should take action to remedy that.

The Commission should dismiss SCE's contention that TURN's proposal is "punitive."¹⁷⁶¹ The contention would seem to apply equally to SCE's escalation adjustments and post-test year ratemaking proposals and any of the numerous other ratemaking adjustments made to reflect the effect of cumulative inflation. Similarly baseless is SCE's suggestion that the Z Factor should be treated as an outdated element of performance-based ratemaking;¹⁷⁶² the Z-Factor has been part of SCE's GRC ratemaking since the utility first proposed to include it in the test year 2003 GRC.¹⁷⁶³

¹⁷⁶¹ Ex. SCE-18, Vol. 1, p. 24.

¹⁷⁶² *Id*.

¹⁷⁶³ D.04-07-022 (SCE Test Year 2023 GRC), pp. 278-279.

42. **RESIDENTIAL DISCONNECTIONS AND ARREARAGES**

43. **COMPLIANCE REQUIREMENTS**

44. **ACCESSIBILITY ISSUES**

RESULTS OF FINANCIAL EXAMINATION BY CAL ADVOCATES 45.

46. **GRC UPDATE PHASE**

47. **CONCLUSION**

Date: July 15, 2024

Respectfully submitted,

By:

____/s/_____ David Cheng

David Cheng, Staff Attorney Hayley Goodson, Managing Attorney Thomas Long, Director of Legal Strategy Robert Finkelstein, General Counsel Matthew Freedman, Staff Attorney Katy Morsony, Staff Attorney Elise Torres, Staff Attorney Isabella Montoya, Utility Justice Legal Fellow

THE UTILITY REFORM NETWORK

360 Grand Avenue, #150 Oakland, CA 94610 Phone: (415) 929-8876 E-mail: dcheng@turn.org